

Contributions to the relative trace formula of Guo-Jacquet

Huajie Li

► To cite this version:

Huajie Li. Contributions to the relative trace formula of Guo-Jacquet. Number Theory [math.NT]. Université Paris Cité, 2020. English. NNT: 2020UNIP7080 . tel-03226143

HAL Id: tel-03226143 https://theses.hal.science/tel-03226143

Submitted on 14 May 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





Université de Paris

École Doctorale de Sciences Mathématiques de Paris-Centre (ED 386) Institut de Mathématiques de Jussieu-Paris Rive Gauche (UMR7586)

Contributions to the relative trace formula of Guo-Jacquet

Par Huajie LI

Thèse de doctorat de mathématiques Dirigée par **Pierre-Henri CHAUDOUARD**

Présentée et soutenue publiquement le 29 septembre 2020

Devant le jury composé de :

Mme Anne-Marie AUBERT, Directrice de recherche, CNRS, Présidente

M. Raphaël BEUZART-PLESSIS , Chargé de recherche, CNRS, Examinateur

M. Pierre-Henri CHAUDOUARD , Professeur, Université de Paris, Directeur

M. Volker HEIERMANN, Professeur, Aix-Marseille Université, Rapporteur

M. Nadir MATRINGE, Maître de conférence HDR, Université de Poitiers, Examinateur

M. Jean-Loup WALDSPURGER, Directeur de recherche émérite, CNRS, Examinateur

M. Wei ZHANG, Professeur, Massachusetts Institute of Technology, Rapporteur

Except where otherwise noted, this is work licensed under https://creativecommons.org/licenses/by-nc-nd/3.0/fr/

Projet soutenu par l'attribution d'une allocation de recherche Région Ile-de-France

Université Paris-Diderot École Doctorale ED 386 Bâtiment Sophie Germain Case courrier 7012 8 place Aurélie Nemours 75205 Paris CEDEX 13

Institut de Mathématiques de Jussieu - Paris Rive Gauche UMR7586 Université Paris Diderot Bâtiment Sophie Germain Boite Courrier 7012 75205 PARIS Cedex 13

Title, abstract and key words (in English and French)

Title: Contributions to the relative trace formula of Guo-Jacquet

Abstract: We establish global and local trace formulae for infinitesimal symmetric spaces of Guo-Jacquet. We also prove several local results concerning the comparison of regular semi-simple terms which are noninvariant weighted orbital integrals. This thesis contains five chapters. In Chapter 1, we recall the motivations and state our main reults. Our work is inspired by a conjecture of Guo-Jacquet, which is an example in the relative Langlands programme, and analytic problems appearing in the relative trace formula approach. In Chapter 2, we establish an infinitesimal variant of Guo-Jacquet trace formula for the case of $(GL_{2n,D}, GL_{n,D} \times GL_{n,D})$. It is a kind of Poisson summation formula obtained by an analogue of Arthur's truncation. We describe regular semi-simple terms as explicit weighted orbital integrals. In Chapter 3, we estabilish a similar formula and have a similar description of regular semi-simple terms for the case of a central simple algebra containing a quadratic extension. Moreover, we state and prove the weighted fundamental lemma thanks to Labesse's work on the base change for GL_n . In Chapter 4, we establish an infinitesimal invariant local trace formula of Guo-Jacquet over a p-adic field by following works of Waldspurger and Arthur. During the proof, we also obtain an infinitesimal noninvariant local trace formula, Howe's finiteness for weighted orbital integrals and the representability of the Fourier transform of weighted orbital integrals. In Chapter 5, with the results in previous chapters, we adopt Waldspurger's strategy on the endoscopic transfer to prove some relations between Fourier transforms of invariant local weighted orbital integrals.

Key words: Guo-Jacquet trace formula, Arthur's truncation, weighted fundamental lemma, local trace formula, noninvariant transfer

Titre : Contributions à la formule des traces relative de Jacquet-Guo

Résumé : On établit des formules des traces globale et locale pour les espaces symétriques infinitésimaux de Jacquet-Guo. On prouve également quelques résultats locaux concernant la comparaison de termes semi-simples réguliers qui sont des intégrales orbitales pondérées non invariantes. Cette thèse contient cinq chapitres. Dans le chapitre 1, on rappelle les motivations et énonce nos principaux résultats. Notre travail s'inspire d'une conjecture de Jacquet-Guo, qui est un exemple dans le programme de Langlands relatif, et des problèmes analytiques apparaissant dans l'approche par la formule des traces relative. Dans le chapitre 2, on établit une variante infinitésimale de la formule des traces de Jacquet-Guo pour le cas de $(GL_{2n,D}, GL_{n,D} \times GL_{n,D})$. Elle est une sorte de formule sommatoire de Poisson obtenue par un analogue de la troncature d'Arthur. On décrit les termes semi-simples réguliers comme des intégrales orbitales pondérées explicites. Dans le chapitre 3, on établit une formule similaire et a une description similaire des termes semi-simples réguliers pour le cas d'une algèbre centrale simple contenant une extension quadratique. De plus, on énonce et prouve le lemme fondamental pondéré grâce aux travaux de Labesse sur le changement de base pour GL_n . Dans le chapitre 4, on établit une formule des traces locale invariante infinitésimale de Jacquet-Guo sur un corps p-adique en suivant les travaux de Waldspurger et Arthur. Au cours de la démonstration, on obtient également une formule des traces locale non invariante infinitésimale, la finitude de Howe pour les intégrales orbitales pondérées et la représentabilité de la transformée de Fourier des intégrales orbitales pondérées. Dans le chapitre 5, avec les résultats des chapitres précédents, on adopte la stratégie de Waldspurger sur le transfert endoscopique pour prouver certaines relations entre transformées de Fourier des intégrales orbitales pondérées locales invariantes.

Mots-clefs : formule des traces de Jacquet-Guo, troncature d'Arthur, lemme fondamental pondéré, formule des traces locale, transfert non invariant

Acknowledgement

I would like to express my deepest appreciation to my supervisor Prof. Pierre-Henri Chaudouard. It is him who has introduced me to this subject. He has kindly shared his insight with me and always offered me helpful advice from different aspects. He has also answered numerous questions during my preparation of this thesis. This thesis was finished during an unusual period caused by the Covid-19, but he has spent a lot of time discussing with me to help me overcome difficulties. I am also grateful for his long list of comments on earlier drafts of my thesis, which has greatly improved the exposition. This work would not have been accomplished without his guidance and support since my Master 2.

I am sincerely grateful to Prof. Volker Heiermann and Prof. Wei Zhang for kindly accepting to write a report for my thesis and carefully reading my manuscript during the summer. It is my pleasure to invite Prof. Jean-Loup Waldspurger to be a member of my jury, whose pioneering works have undoubtedly a huge influence on my thesis. I thank him especially for seriously reading my manuscript and correcting several errors. Apart from them, I really appreciate that the professors Anne-Marie Aubert, Raphaël Beuzart-Plessis and Nadir Matringe agreed to be part of the jury of my thesis defense. I am delighted to have my thesis presented with their presence.

I would like to express my gratitude to Prof. Luc Illusie. It was fortunate for me to meet him four years ago who introduced me to Prof. Chaudouard, which led to the beginning of this project.

During my preparation of this work, I have benefited from interesting discussion on related topics with Raphaël Beuzart-Plessis, Jiaming Chen, Marion Chommaux, Nicolas Duhamel, Ning Guo, Volker Heiermann, Spencer Leslie, Qirui Li, Wen-Wei Li, Jie Lin, Nadir Matringe, Yiannis Sakellaridis, Cheng Shu, Ye Tian, Jingwei Xiao, Liyang Yang, Hongjie Yu, Wei Zhang, Peng Zheng, Jialiang Zou, Jiandi Zou and Michał Zydor, which has helped me understand better my project or inspired some ideas in my work though some of them may have not realised it. Part of this work was presented in the seminars at Marseille and Poitiers, and I would like to thank the organisers and the audience for offering me these opportunities to make everything clearer. I would also like to thank Peiyi Cui, Ratko Darda, Yanbo Fang, Miao Gu, Zhizhong Huang, Xiaocheng Li, Yongqi Liang, Chenguang Liu, Zicheng Qian, Guanyang Wang, Hua Wang, Cong Xue, Ruotao Yang and others for their advice and help on my graduate studies.

It is my pleasure doing my PhD in the group "Formes Automorphes" at the IMJ-PRG with an excellent atmosphere and numerous experts in this field. I have attended the regular research seminar "Groupes Réductifs et Formes Automorphes" as well as reading seminars organised by Arnaud Mayeux, Benoît Stroh and Hongjie Yu on topics related to my research. I would like to thank all the organisers and participants from whom I have learnt a lot.

I was lucky enough to attend a wide range of courses in Orsay and Paris in these years which have improved my understanding of mathematics. I would like to thank all the professors: Christophe Breuil, Antoine Chambert-Loir, François Charles, Pierre-Henri Chaudouard, Huayi Chen, Laurent Clozel, Olivier Fouquet, Guy Henniart, Marc Hindry, Bruno Klingler, Emmanuel Letellier, Matthew Morrow, Benoît Stroh, Éric Vasserot and Tony Yue Yu. I would also like to thank Mr. Pascal Dingoyan, my tutor in Master 2, for his help during my first year in France.

I am thankful that there were many professors who made me prepared for research in mathematics when I was an undergraduate student at USTC. In particular, I would like to thank Ye Tian, Ke Chen and Yi Ouyang for raising my interest in number theory and giving valuable advice on my further studies in France.

I would like to thank the teams of "Français Langue Etrangère" at UPMC and Paris Diderot for teaching me French, which has made my research and life easier in France. I would like to thank the "Bureau des Relations Internationales" at Paris diderot as well for offering me an opportunity to live in the beautiful park CIUP, where a large part of this thesis was written.

I am happy to have a lot of colleagues at the IMJ-PRG and in Paris to share my PhD life. They include particularly my office mates in the Office 653: Léa Bittmann, Charles Fougeron, Omar Mohsen

and Peng Zheng. There are also many friends making my life in Paris more colourful such as my classmates from USTC and new friends in the CIUP. I would like to thank all of them though their names are too many to be listed here.

My special thanks go to the "Fondation Sciences Mathématiques de Paris" and the "Région Ile-de-France" for offering scholarships for my Master 2 via the programme "PGSM International" and my PhD via the programme "DIM RDM-IdF" respectively, which made my graduate studies in France possible.

Last but not least, I thank my family for their constant support.

Contents

Title, abstract and key words (in English and French)	3
Acknowledgement	5
Introduction en français 1. La conjecture de Jacquet-Guo et l'approche de la formule des traces relativ 2. Principaux résultats	9 9 10
Chapter 1. Introduction 1 Guo-Jacquet conjecture and the relative trace formula approach	17 17
2. Main results	18
 Chapter 2. An infinitesimal variant of Guo-Jacquet trace formula: the case of (GL_{2n,D}, GL_{n,D} × GL_{n,D}) 1. Introduction 2. Notation 3. The symmetric pair 4. Integrability of the modified kernel 5. Exponential polynomial distributions 6. Non-equivariance 7. An infinitesimal trace formula for Mataxa D ⊕ Mataxa D //GLa D × GLa D 	23 23 25 28 31 40 49 51
8. The second modified kernel Weilde Lither that for $\operatorname{Hat}_{p \times q, D} \oplus \operatorname{Hat}_{q \times p, D} / \operatorname{GL}_{p, D} \times \operatorname{GL}_{q, D}$	52 57
9. Weighted orbital integrals	57
 Chapter 3. An infinitesimal variant of Guo-Jacquet trace formula: the case of a consimple algebra containing a quadratic extension 1. Introduction 2. Notation 3. The symmetric pair 4. Integrability of the modified kernel 5. Polynomial distributions 6. Noninvariance 7. An infinitesimal trace formula for \$\sigma/H\$ 8. The second modified kernel 9. Weighted orbital integrals 10. The weighted fundamental lemma 	ntral 61 63 65 68 74 78 80 81 85 88 88
Chapter 4. A local trace formula for <i>p</i> -adic infinitesimal symmetric spaces: the ca Guo-Jacquet 1. Introduction	se of 97 97
2. Notation and preliminaries	99
3. Symmetric pairs 4. Weighted orbital integrals	101 109
5. The noninvariant trace formula	113
6. Howe's finiteness for weighted orbital integrals	120
7. Representability of the Fourier transform of weighted orbital integrals	125
8. Invariant weighted orbital integrals	134
9. The invariant trace formula	139
10. A vanishing property at infinity	144

Chapter 5. On certain identities between Fourier transforms of weighted orbital	
integrals on infinitesimal symmetric spaces of Guo-Jacquet	149
1. Introduction	149
2. Notation and preliminaries	150
3. Labesse's lemma	154
4. Matching of orbits	155
5. Statement of results	161
6. Limit formulae	164
7. Construction of test functions	164
8. The weighted fundamental lemma	166
9. Approximation of local data by global data	167
10. An infinitesimal variant of Guo-Jacquet trace formulae	167
11. Proof of Proposition 5.5	169
Bibliography	175

Introduction en français

Ce projet de thèse s'inspire d'une conjecture de Jacquet-Guo [23] qui généralise le théorème connu de Waldspurger [50] reliant les périodes toriques aux valeurs centrales des fonctions L automorphes pour GL_2 en dimensions supérieures. C'est l'un des premiers exemples intéressants du programme de Langlands relatif initié par Jacquet et ses collaborateurs et systématiquement proposé par Sakellaridis-Venkatesh [46]. Un outil efficace pour attaquer ce genre de problèmes est la formule des traces relative introduite par Jacquet [29] pour réprouver le résultat de Waldspurger. Les objectifs de cette thèse sous la direction de Pierre-Henri Chaudouard incluent le développement de variantes infinitésimales des formules des traces de Jacquet-Guo globale et locale et l'établissement des résultats locaux de comparaison nécessaires pour une étude plus approfondie de cette conjecture. On espère que certaines méthodes ici seront également utiles dans d'autres formules des traces relatives.

1. La conjecture de Jacquet-Guo et l'approche de la formule des traces relative

Rappelons brièvement la conjecture de Jacquet-Guo proposée dans [23]. Soient E/F une extension quadratique de corps de nombres et η le caractère quadratique attaché de $\mathbb{A}^{\times}/F^{\times}$, où \mathbb{A} est l'anneau des adèles de F. Soient $G := GL_{2n}$ et $H := GL_n \times GL_n$ son sous-groupe. Tous les groupes considérés sont définis sur F. Soit π une représentation automorphe cuspidale de $G(\mathbb{A})$ avec un caractère central trivial. On dit que π est H-distinguée si les deux formes linéaires (appelées "périodes") sur son espace

$$\mathcal{P}_H:\phi\mapsto\int_{H(F)Z(\mathbb{A})\backslash H(\mathbb{A})}\phi(h)dh$$

 et

$$\mathcal{P}_{H,\eta}:\phi\mapsto \int_{H(F)Z(\mathbb{A})\setminus H(\mathbb{A})}\phi(h)\eta(\det(h))dh$$

sont non nulles, où Z est le centre de G. Cette propriété est directement liée à la non-annulation de certaines valeurs centrales de fontions L grâce au travail de Friedberg-Jacquet [22]. On doit également considérer une autre paire de groupes. Notons X(E) l'ensemble des classes d'isomorphismes d'algèbres de quaternions D/F contenant E. Pour tout $D \in X(E)$, on note $G' = GL_{n,D}$ le groupe algébrique défini sur F dont le groupe de points sur F est $GL_n(D)$. Soit $H' = \operatorname{Res}_{E/F}GL_{n,E}$ son sous-groupe. Soit π' une représentation automorphe cuspidale de $G'(\mathbb{A})$ avec un caractère central trivial. On dit que π' est H'-distinguée si la forme linéaire sur son espace

$$\mathcal{P}_{H'}:\phi\mapsto \int_{H'(F)Z(\mathbb{A})\backslash H'(\mathbb{A})}\phi(h)dh$$

est non nulle, où le centre de G' s'identifie à Z. Une direction de la conjecture de Jacquet-Guo dit que si π' est H'-distinguée et correspond par la correspondance de Jacquet-Langlands à π , alors π est H-distinguée. On s'attend également à une réciproque au moins pour n impair. Pour n = 1, celles-ci étaient connues par Waldspurger [50] et réprouvées par Jacquet [29] via la formule des traces relative.

Maintenant, on décrit formellement l'approche par la formule des traces relative suivant Jacquet [29] à cette conjecture. Soit f_G une fonction lisse sur $G(\mathbb{A})$ à support compact. En tant qu'analogue de la formule des traces d'Arthur-Selberg, la formule des traces relative pour le cas de (G, H) indique grossièrement qu'il existe deux façons d'écrire l'intégrale (vue comme une distribution)

(1.0.1)
$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \mathbb{K}_{f_G}(x,y)\eta(\det(x))dxdy$$

où $G(\mathbb{A})^1$ est le sous-ensemble des éléments dans $G(\mathbb{A})$ dont les déterminants sont de valeur absolue 1 et $\mathbb{K}_{f_G}(x, y) := \sum_{\gamma \in G(F)} f_G(x^{-1}\gamma y)$. Le côté géométrique devrait être une somme d'intégrales orbitales relatives tandis que le côté spectral devrait être une expansion de périodes. On pourrait de même imaginer une autre formule pour le cas de (G', H'). Alors la comparaison des périodes sur les différents groupes prédite par la conjecture de Jacquet-Guo est réduite à la comparaison d'intégrales orbitales relatives, pour laquelle on a plus d'outils tels que le lemme fondamental de Guo [23] et le transfert lisse de Zhang [58].

Cette approche a été adoptée par Feigon-Martin-Whitehouse [21] pour obtenir des résultats partiels. Cependant, on a négligé la difficulté analytique dans la discussion ci-dessus. C'est-à-dire qu'une obstruction de l'approche par la formule des traces relative à la conjecture de Jacquet-Guo est le problème suivant.

PROBLÈME 1.1. L'intégrale double (1.0.1) et ses développements ne sont pas convergents.

Ce problème explique certaines conditions locales restrictives dans les principaux résultats de [21] basés sur une simple formule des traces relative. Si l'on espère supprimer ces restrictions apparemment artificielles et obtenir des informations sur toutes les représentations cuspidales, ce problème ne peut être ignoré. Par conséquent, il est nécessaire d'établir des formules des traces relatives valides et générales pour les cas (G, H) et (G', H') plutôt que des formules formelles ou simples. Ensuite, on doit comparer les nouveaux termes apparaissant dans ces formules en dehors des intégrales orbitales relatives ordinaires.

Un tel problème existait également dans la formule des traces d'Arthur-Selberg classique et Arthur a introduit un processus de troncature dans [3] et [4] pour le résoudre. Dans le cadre relatif, Jacquet-Lapid-Rogawski nous a fourni le premier exemple de la "troncature mixte" dans [30], dont l'idée grossière est de définir une troncature compatible avec les différents groupes concernés. Ces travaux et d'autres donnent de bons exemples pour notre étude du problème 1.1.

On se concentrera sur une variante infinitésimale du problème 1.1 et la comparaison de nouveaux termes impliqués dans cette thèse. Cela signifie que l'on va travailler sur l'espace tangent d'un espace symétrique (appelé espace symétrique infinitésimal). Cela serait intéressant pour au moins deux raisons. D'une part, il est proche du côté géométrique de la formule des traces relative originale et nous suggère une facon d'aborder le problème original, mais le côté spectral de la formule des traces relative est remplacé par la transformée de Fourier du côté géométrique où l'analyse harmonique est plus simple. D'autre part, une variante infinitésimale des formules des traces de Jacquet-Guo est également utile dans la comparaison de formules des traces relatives entre deux cas ; par exemple, sa version simple a été utilisée par Zhang [58] pour montrer le transfert lisse.

2. Principaux résultats

2.1. Une variante infinitésimale des formules des traces de Jacquet-Guo. On résout d'abord une variante infinitésimale du problème 1.1 dans le chapitre 2 et le chapitre 3. C'est le principal résultat global de cette thèse.

2.1.1. Le cas de (G, H). On remarque que le cadre du chapitre 2 est un peu plus général pour inclure le cas de [57], mais on se concentrera sur la paire (G, H) définie ci-dessus dans l'introduction. Notons S l'espace symétrique G/H. La double intégrale (1.0.1) peut être formellement écrite comme une intégrale seule

$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} K_{f_S}(x)\eta(\det(x))dx$$

où $f_S(x) := \int_{H(\mathbb{A})\cap G(\mathbb{A})^1} f_G(xy) dy$ définit une fonction lisse sur $S(\mathbb{A})$ à support compact et $K_{f_S}(x) := \sum_{\gamma \in S(F)} f_S(x^{-1}\gamma x)$. Si l'on remplace S par l'espace tangent $\mathfrak{s} \simeq \mathfrak{gl}_n \oplus \mathfrak{gl}_n$ en l'élément neutre, alors le problème 1.1 apparaît comme la divergence de l'intégrale

(2.1.1)
$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} k_f(x)\eta(\det(x))dx,$$

où f est une fonction de Bruhat-Schwartz sur $\mathfrak{s}(\mathbb{A})$ et $k_f(x) := \sum_{\gamma \in \mathfrak{s}(F)} f(x^{-1}\gamma x)$. On remplace $k_f(x)$ par une expression explicite $k_f^T(x)$ définie dans §4 du chapitre 2 pour rendre (2.1.1) absolument convergente, où $T \in \mathbb{R}^{2n}$ est un paramètre de troncature. Cette définition essentielle combine des idées de [28] [61] [17] pour la décomposition de $H(\mathbb{A})$ et [40] pour la décomposition de $\mathfrak{s}(\mathbb{A})$. Afin de décrire le développement géométrique, on définit une relation d'équivalence sur $\mathfrak{s}(F)$: deux éléments dans $\mathfrak{s}(F)$ sont équivalents si et seulement s'ils se trouvent dans la même fibre du quotient catégorique $\mathfrak{s}//H$. Notons \mathcal{O} l'ensemble des classes d'équivalence. Pour tout $\mathfrak{o} \in \mathcal{O}$, on définit de même $k_{f,\mathfrak{o}}^T(x)$ en remplaçant $\mathfrak{s}(F)$ par \mathfrak{o} . Le principal résultat du chapitre 2 est le théorème suivant qui donne le développement géométrique de (2.1.1) avec $k_f(x)$ remplacé par $k_f^T(x)$. THÉORÈME 2.1 (voir le théorème 4.11 du chapitre 2). Pour tout T appartenant à un cône convenable dans \mathbb{R}^{2n} ,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1}k_{f,\mathfrak{o}}^T(x)\eta(\det(x))dx$$

est absolument convergente.

Notons $J_{\mathfrak{o}}^{T}(\eta, f)$ l'intégrale (2.1.1) avec $k_{f}(x)$ remplacé par $k_{f,\mathfrak{o}}^{T}(x)$. On montre que c'est un polynômeexponentielle en T (voir le corollaire 5.6 du chapitre 2). Notons $J_{\mathfrak{o}}(\eta, f)$ son terme constant. Dans le cadre infinitésimal, le développement géométrique de la transformée de Fourier de f (notée \hat{f}) joue le rôle du côté spectral original (cf. [13]). Alors la variante infinitésimale de la formule des traces de Jacquet-Guo pour le cas de (G, H) ci-dessous est déduite de la formule sommatoire de Poisson.

THÉORÈME 2.2 (voir le théorème 7.1 du chapitre 2). Pour toute fonction de Bruhat-Schwartz f sur $\mathfrak{s}(\mathbb{A})$, on a l'égalité

$$\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(\eta,f)=\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(\eta,\hat{f}).$$

De plus, pour la plupart des classes \mathfrak{o} (c'est-à-dire les classes semi-simples régulières), on peut écrire $J_{\mathfrak{o}}(\eta, f)$ comme une intégrale orbitale pondérée explicite ; en particulier, pour les classes elliptiques \mathfrak{o} , on voit que $J_{\mathfrak{o}}(\eta, f)$ n'est rien d'autre que l'intégrale orbitale ordinaire. On dit qu'un sous-groupe parabolique semi-standard de G est ω -stable s'il contient $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$.

THÉORÈME 2.3 (voir le théorème 9.2 du chapitre 2). Soient $\mathfrak{o} \in \mathcal{O}$ une classe semi-simple régulière, P_1 un sous-groupe parabolique relativement standard ω -stable de G et $X_1 \in \mathfrak{o}$ un élément elliptique relatif à P_1 (défini dans la section 9.2 du chapitre 2). Pour toute fonction de Bruhat-Schwartz f sur $\mathfrak{s}(\mathbb{A})$, on a

$$J_{\mathfrak{o}}(\eta, f) = \operatorname{vol}([H_{X_1}]) \cdot \int_{H_{X_1}(\mathbb{A}) \setminus H(\mathbb{A})} f(x^{-1}X_1x) v_{P_1}(x) \eta(\det(x)) dx,$$

où H_{X_1} est le centralisateur de X_1 dans H, $vol([H_{X_1}])$ est son volume associé et $v_{P_1}(x)$ est le volume d'un enveloppe convexe.

Le poids $v_{P_1}(x)$ est exactement la restriction à $H(\mathbb{A})$ du poids d'Arthur dans [3] pour $G(\mathbb{A})$. Il est remarquable que notre intégrale orbitale pondérée a le même poids que celui de la formule des traces tordue (voir [39, p. 131]) pour l'espace tordu ($GL_n \times GL_n$) $\rtimes \sigma$, où $\sigma(x, y) := (y, x)$.

2.1.2. Le cas de (G', H'). On étudie encore un cadre plus général dans le chapitre 3 que la paire (G', H') ci-dessus. En fait, on considère une algèbre centrale simple \mathfrak{g}' sur F contenant E et le centralisateur \mathfrak{h}' de E dans \mathfrak{g}' . On définit G' et H' comme les groupes d'éléments inversibles respectivement dans \mathfrak{g}' et \mathfrak{h}' . Ce cas plus général s'inspire de la conjecture locale de Prasad et Takloo-Bighash [44, Conjecture 1] et est nécessaire pour compléter la réciproque de la conjecture de Jacquet-Guo pour n pair.

Quitte à conjuguer (G', H') par G'(F), la paire symétrique est réduite à l'un des deux cas ci-dessous (voir la proposition 3.7 du chapitre 3). Ces deux cas peuvent être traités de la même manière.

Cas I: $(G', H') \simeq (GL_{n,D}, \operatorname{Res}_{E/F}GL_{\frac{n}{2}, D\otimes_F E})$, où D est une algèbre à division centrale sur F contenant E, et $D' := \operatorname{Cent}_D(E)$ est le centralisateur de E dans D.

Cas II: $(G', H') \simeq (GL_{n,D}, \operatorname{Res}_{E/F}GL_{\frac{n}{2}, D\otimes_F E})$, où D est une algèbre à division centrale sur F telle qu'il n'y a pas de plongement $E \to D$ en tant que F-algèbres.

Remarquons que (G', H') est identique à (G, H) après le changement de base à une clôture algébrique de F contenant E. On a des définitions et des résultats similaires pour le cas de (G', H') comme suit. Notons $H'(\mathbb{A})^1$ le sous-ensemble des éléments dans $H'(\mathbb{A})$ dont les normes réduites sont de valeur absolue 1. On note \mathcal{O}' l'ensemble des classes d'équivalence sur $\mathfrak{s}'(F)$ définies par le quotient catégorique $\mathfrak{s}'//H'$. Soient f' une fonction de Bruhat-Schwartz sur $\mathfrak{s}'(\mathbb{A})$ et $\mathfrak{o} \in \mathcal{O}'$. Pour un paramètre de troncature $T \in \mathbb{R}^n$ et $x \in H'(x)$, on définit un noyau tronqué $k_{f',\mathfrak{o}}^T(x)$ dans §4 du chapitre 3.

THÉORÈME 2.4 (voir le théorème 4.2 du chapitre 3). Pour tout T appartenant à un cône convenable dans \mathbb{R}^n ,

$$\sum_{\mathfrak{o}\in\mathcal{O}'}\int_{H'(F)\backslash H'(\mathbb{A})^1}k_{f',\mathfrak{o}}^T(x)dx$$

 $est \ absolument \ convergente.$

Pour toute $\mathfrak{o} \in \mathcal{O}'$, on note $J^T_{\mathfrak{o}}(f')$ l'intégrale correspondante dans la dernière somme. On prouve que c'est un polynôme en T (voir le corollaire 5.3 du chapitre 3). On note $J_{\mathfrak{o}}(f')$ son terme constant. Notons \hat{f}' la transformée de Fourier de f'. Voici la variante infinitésimale de la formule des traces de Jacquet-Guo pour le cas de (G', H').

THÉORÈME 2.5 (voir le théorème 7.1 du chapitre 3). Pour toute fonction de Bruhat-Schwartz f' sur $\mathfrak{s}'(\mathbb{A})$, on a l'égalité

$$\sum_{\mathfrak{o}\in\mathcal{O}'}J_{\mathfrak{o}}(f')=\sum_{\mathfrak{o}\in\mathcal{O}'}J_{\mathfrak{o}}(\hat{f}').$$

Comme précédemment, on peut écrire les termes semi-simples réguliers comme des intégrales orbitales pondérées explicites avec les mêmes poids que ceux d'Arthur dans [3]. Pour $(G', H') = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,E})$, on obtient les mêmes poids apparaissant dans la formule des traces tordue (voir [39, p. 131]) pour l'espace tordu ($\operatorname{Res}_{E/F}GL_{n,E}$) $\rtimes \sigma'$, où σ' est la conjugaison galoisienne non triviale.

THÉORÈME 2.6 (voir le théorème 9.2 du chapitre 3). Soient $\mathfrak{o} \in \mathcal{O}'$ une classe semi-simple régulière, P'_1 un sous-groupe parabolique standard de H' et $Y_1 \in \mathfrak{o}$ un élément elliptique par rapport à P'_1 (défini dans la section 9 du chapitre 3). Pour toute fonction de Bruhat-Schwartz f' sur $\mathfrak{s}'(\mathbb{A})$, on a

$$J_{\mathfrak{o}}(f') = \operatorname{vol}([H'_{Y_1}]) \cdot \int_{H'_{Y_1}(\mathbb{A}) \setminus H'(\mathbb{A})} f'(\operatorname{Ad}(x^{-1})(Y_1)) v_{P'_1}(x) dx,$$

où H'_{Y_1} est le centralisateur de Y_1 dans H', $vol([H'_{Y_1}])$ est le volume associé à H'_{Y_1} et $v_{P'_1}(x)$ est le volume d'un envelope convexe.

Effectivement, ce cas est encore plus simple que celui de (G, H); par exemple, $J_o^T(f')$ est un polynôme pur en T au lieu d'un polynôme-exponentielle. Une des raisons de cette simplicité est que le tore central F-déployé maximal de H' est le même que celui de G'. De plus, il existe une bijection entre l'ensemble des sous-groupes paraboliques semi-standards de H' et l'ensemble des sous-groupes paraboliques relativement standards de G'. Cependant, il y a quelques problèmes de rationalité supplémentaires dans ce cas.

2.2. Quelques résultats locaux pour la comparaison de termes semi-simples réguliers. Les termes $J_{\mathfrak{o}}(\eta, f)$ and $J_{\mathfrak{o}}(f')$ ci-dessus associés aux classes semi-simples régulières mais non elliptiques $\mathfrak{o} \in \mathcal{O}$ et $\mathfrak{o} \in \mathcal{O}'$ respectivement, qui sont des intégrales orbitales pondérées globales, sont les premiers nouveaux termes que l'on doit étudier et comparer pour les applications des formules des traces de Jacquet-Guo. La principale difficulté pour étudier ces distributions globales par rapport à celles de [60] ou [61] est que on est confronté à des distributions non invariantes sous la conjugaison de $H(\mathbb{A})$ ou $H'(\mathbb{A})$. C'est proche de la situation de la formule des traces d'Arthur-Selberg classique. Alors que la procédure standard mais assez difficile est de rendre notre formule des traces invariante comme Arthur l'a fait dans [6] et [7], une autre manière proposée par Labesse dans [37] est de comparer directement les distributions non invariantes. On suivra cette dernière approche. Les intégrales orbitales pondérées locales correspondantes sont les principaux objets de §10 du chapitre 3, le chapitre 4 et le chapitre 5.

Soit E/F une extension quadratique de corps locaux non-archimédiens de caractéristique zéro. Soit η le caractère quadratique de F/NE^{\times} attaché à E/F, où NE^{\times} désigne la norme de E^{\times} . On définit des paires symétriques comme dans le cas global. Notons \mathfrak{s}_{rs} l'ensemble des éléments semi-simples réguliers dans \mathfrak{s} . On note $\mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ (resp. $\mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$) l'espace des fonctions localement constantes à support compact sur $\mathfrak{s}(F)$ (resp. $\mathfrak{s}'(F)$). Soit M un sous-groupe de Levi ω -stable de G, c'est-à-dire qu'il est un facteur de Levi d'un sous-groupe parabolique ω -stable. Soit $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Pour toute $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, on définit l'intégrale orbitale pondérée locale

$$J_M^G(\eta, X, f) := |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) \eta(\det(x)) v_M^G(x) dx$$

où $|D^{\mathfrak{s}}(X)|_F$ est le discriminant de Weyl et $v_M^G(x)$ est la fonction de poids locale correspondante. De même, soit M' un sous-groupe de Levi de H' et $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, où $\widetilde{M'}$ est le sous-groupe de Levi de G' tel que $\widetilde{M'} \cap H' = M'$. Pour toute $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$, on définit l'intégrale orbitale pondérée locale

$$J_{M'}^{H'}(Y,f') := |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} f'(\mathrm{Ad}(x^{-1})(Y)) v_{M'}^{H'}(x) dx,$$

où $|D^{\mathfrak{s}'}(Y)|_F$ est le discriminant de Weyl et $v_{M'}^{H'}(x)$ est la fonction de poids locale correspondante.

Il y a une application injective $M' \mapsto M$ de l'ensemble des sous-groupes de Levi de H' dans l'ensemble des sous-groupes de Levi ω -stables de G. On supposera que G et G' ont la même dimension n^2 . Comme $\mathfrak{s}//H \simeq \mathbf{A}^n \simeq \mathfrak{s}'//H'$, si $X \in \mathfrak{s}_{rs}(F)$ et $Y \in \mathfrak{s}'_{rs}(F)$ ont la même image dans \mathbf{A}^n , on dit qu'ils ont des orbites associées. Plus généralement, fixons une paire de sous-groupes de Levi associés M et M'respectivement de G et H'. On a la notion d'orbites M-associées définies par blocs pour $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ et $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$.

2.2.1. Le lemme fondamental pondéré. On obtient le premier résultat sur la comparaison non invariante des formules des traces de Jacquet-Guo dans §10 du chapitre 3. Dans la comparaison des côtés géométriques des formules des traces de Jacquet-Guo, un cas important est appelé le lemme fondamental. Il dit en gros que en presque toute place non ramifiée, certaines fonctions basiques sur G(F) et G'(F)devraient avoir des intégrales orbitales relatives locales associées sur des orbites associées. Guo [23] l'a prouvé pour les unités d'algèbres sphériques de Hecke à l'aide du lemme fondamental de changement de base pour les algèbres de Hecke sphériques complètes pour GL_n connu par Kottwitz [36, lemme 8.8] et Arthur-Clozel [10, théorème 4.5 du chapitre 1]. Une version infinitésimale [58, lemme 5.18] a été utilisée par Zhang pour prouver le transfert lisse des intégrales orbitales ordinaires pour les formules des traces de Jacquet-Guo en suivant la même philosophie du travail de Waldspurger [52] sur le transfert endoscopique.

Inspiré de [37, définition III.3.2], on définit dans §10 du chapitre 3 la notion de paires "fortement associées" de fonctions localement constantes à support compact sur $\mathfrak{s}(F)$ et $\mathfrak{s}'(F)$. En gros, deux fonctions sont fortement associées si elles ont des intégrales orbitales pondérées associées sur des orbites associées. En presque toute place non ramifiée, (G', H') est isomorphe à $(GL_{2n}, \operatorname{Res}_{E/F}GL_{n,E})$ et $\mathfrak{s}'(F) \simeq \mathfrak{gl}_n(E)$. Notons \mathcal{O}_F (resp. \mathcal{O}_E) l'anneau des entiers de F (resp. E). Soit f_0 et f'_0 les fonctions caractéristiques respectivement de $\mathfrak{s}(\mathcal{O}_F) \simeq (\mathfrak{gl}_n \oplus \mathfrak{gl}_n)(\mathcal{O}_F)$ et $\mathfrak{s}'(\mathcal{O}_F) \simeq \mathfrak{gl}_n(\mathcal{O}_E)$. Parce que l'on obtient les mêmes poids pour les cas de Jacquet-Guo que ceux des formules des traces tordues, on peut réduire la version pondérée de [58, lemme 5.18] ci-dessous au travail de Labesse [37] sur le changement de base pour GL_n . On montre qu'en presque toute place non ramifiée v, f_0 et f'_0 sont fortement associées.

THÉORÈME 2.7 (voir le théorème 10.9 du chapitre 3 pour un énoncé précis et général). Soient M et M' une paire de sous-groupes de Levi associés respectivement de G et H'. On a

(1) si $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ et $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ ont des orbites *M*-associées, alors

$$\kappa(X)J_M^G(\eta, X, f_0) = J_{M'}^{H'}(Y, f_0)$$

 $o\dot{u} \kappa(X)$ est un facteur de transfert;

(2) si
$$X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$$
 satisfait $\det(AB) \notin NE^{\times}$, alors
 $J_M^G(\eta, X, f_0) = 0.$

2.2.2. Une formule des traces locale infinitésimale. Notre prochain objectif est de prouver le transfert non invariant des intégrales orbitales pondérées en utilisant le lemme fondamental pondéré suivant la stratégie de [14] et [15] sur le changement de base stable. Pour y parvenir, on doit préparer quelques résultats sur l'analyse harmonique locale comme dans [51]. On prouve une variante infinitésimale des formules des traces locales invariantes pour le cas de Jacquet-Guo dans le chapitre 4 suivant [51] et [8].

On se concentre ici sur le cas de (G, H) pour l'illustration. On définit l'intégrale orbitale pondérée locale (H, η) -invariante $I_M^G(\eta, X, f)$ par le processus standard d'Arthur à partir de l'intégrale orbitale pondérée locale non invariante ci-dessus $J_M^G(\eta, X, f)$. Pour $f, f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, on définit (voir §3.2 du chapitre 4 pour les notations)

$$I^{G}(\eta, f, f') := \sum_{\substack{M \in \mathscr{L}^{G, \omega}(M_{0}) \\ I^{G}_{M}(\eta, X, \hat{f}) I^{G}_{G}(\eta, X, f') dX.}} |W^{GL_{n}|-1}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}}(\eta, X, f')|^{-1} dX.$$

Le principal résultat du chapitre 4 est la formule des traces locale invariante suivante.

THÉORÈME 2.8 (voir le théorème 9.1 du chapitre 4). On a l'égalité

$$I^G(\eta, f, f') = I^G(\eta, f', f).$$

Elle est déduite d'une formule des traces locale non invariante (voir le théorème 5.3 du chapitre 4), qui est essentiellement une conséquence de la formule de Plancherel combinée avec un processus de

troncature. Certains autres résultats locaux, y compris la finitude de Howe pour les intégrales orbitales pondérées et la représentabilité de la transformée de Fourier des intégrales orbitales pondérées, sont obtenus lors de la démonstration.

PROPOSITION 2.9 (voir le corollaire 6.6 du chapitre 4). Soient r un sous-groupe compact ouvert de $\mathfrak{s}(F), M$ un sous-groupe de Levi ω -stable de G et $\sigma \subseteq (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Notons $\mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/r)$ le sous-espace de $\mathcal{C}^{c}_{c}(\mathfrak{s}(F))$ constitué par les fonctions invariantes par translation de r. Supposons qu'il existe un sousensemble compact $\sigma_0 \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ tel que $\sigma \subseteq \operatorname{Ad}((M \cap H)(F))(\sigma_0)$. Alors il existe un sous-ensemble fini $\{X_i : i \in I\} \subseteq \sigma$ et un sous-ensemble fini $\{f_i : i \in I\} \subseteq \mathcal{C}^{\infty}_c(\mathfrak{s}(F)/r)$ tels que pour tout $X \in \sigma$ et toute $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/r)$, on a l'égalité

$$J_M^G(\eta, X, f) = \sum_{i \in I} J_M^G(\eta, X_i, f) J_M^G(\eta, X, f_i).$$

PROPOSITION 2.10 (voir la proposition 7.2 du chapitre 4). Soient M un sous-groupe de Levi ω -stable de G et $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Alors il existe une fonction localement constante $\hat{j}_M^G(\eta, X, \cdot)$ sur $\mathfrak{s}_{rs}(F)$ telle que

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), J^{G}_{M}(\eta, X, \hat{f}) = \int_{\mathfrak{s}(F)} f(Y) \hat{j}^{G}_{M}(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY.$$

On obtient également une propriété d'annulation à "l'infini" analogue à [14, proposition 2.2].

PROPOSITION 2.11 (voir la proposition 10.1 du chapitre 4). Soit $M \neq G$ un sous-groupe de Levi ω -stable de G. Soient $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ et $Y \in \mathfrak{s}_{rs}(F)$. Alors il existe $N \in \mathbb{N}$ tel que si $\lambda \in F^{\times}$ satisfait $v_F(\lambda) < -N$, on a î

$$^{G}_{M}(\eta, \lambda X, Y) = 0,$$

où $\hat{i}_M^G(\eta, X, \cdot)$ est l'analogue (H, η) -invariant de $\hat{j}_M^G(\eta, X, \cdot)$.

Ce sont des résultats d'intérêt indépendant. Des analogues de tous les résultats ci-dessus sont obtenus pour le cas de (G', H').

2.2.3. Certaines identités entre transformées de Fourier des intégrales orbitales pondérées. Comme mentionné ci-dessus, la prochaine étape de notre recherche est de comparer les termes semisimples réguliers dans les côtés géométriques des formules des traces de Jacquet-Guo comme dans [14] et [15]. Cela devrait servir d'exemple de comparaison non invariante dans le contexte relatif. On obtient des relations entre transformées de Fourier des intégrales orbitales pondérées dans le chapitre 5 qui généralisent certains principaux résultats de [58] et sont des analogues pour les formules des traces de Jacquet-Guo de [14].

Soient M un sous-groupe de Levi ω -stable de G et $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Notons $\hat{J}_M^G(\eta, X, \cdot)$ (resp. $\hat{I}_{M}^{G}(\eta, X, \cdot)) \text{ la transformée de Fourier de la distribution } J_{M}^{G}(\eta, X, \cdot) \text{ (resp. } I_{M}^{G}(\eta, X, \cdot)). \text{ Soit } \hat{j}_{M}^{G}(\eta, X, \cdot) \text{ (resp. } \hat{i}_{M}^{G}(\eta, X, \cdot)) \text{ la fonction localement constante sur } \mathfrak{s}_{\mathrm{rs}}(F) \text{ représentant } \hat{J}_{M}^{G}(\eta, X, \cdot) \text{ (resp. } \hat{I}_{M}^{G}(\eta, X, \cdot)).$ Pour un sous-groupe de Levi M' de H' et $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, on obtient de même les fonctions localement constantes $\hat{j}_{M'}^{H'}(Y, \cdot)$ et $\hat{i}_{M'}^{H'}(Y, \cdot)$ sur $\mathfrak{s}'_{\rm rs}(F)$.

Les fonctions $\hat{j}_M^G(\eta, X, \cdot)$ sont décomposées comme leurs analogues invariants $\hat{i}_M^G(\eta, X, \cdot)$ et les fonctions de poids v_M^G . La décomposition des fonctions $\hat{j}_{M'}^{H'}(Y, \cdot)$ est similaire. Afin d'obtenir des relations entre $\hat{j}_{M}^{G}(\eta, X, \cdot)$ et $\hat{j}_{M'}^{H'}(Y, \cdot)$, qui fait partie de la comparaison non invariante, on se concentrera sur les relations entre $\hat{i}_{M}^{G}(\eta, X, \cdot)$ et $\hat{i}_{M'}^{H'}(Y, \cdot)$ dans le chapitre 5.

Fixons une paire de sous-groupes de Levi associés M' et M respectivement de H' et G. Pour $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F), \text{ on note } \eta(X) := \eta(\det(AB)).$ Notre principal résultat dans le chapitre 5 est le suivant

THÉORÈME 2.12 (voir le corollaire 5.6 et la proposition 5.9 du chapitre 5). 1) Supposons que $X \in$ $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ et $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ ont des orbites *M*-associées. Supposons que $U \in \mathfrak{s}_{rs}(F)$ et $V \in \mathfrak{s}'_{rs}(F)$ ont des orbites associées. Alors on a l'égalité

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X)\kappa(U)\hat{i}_{M}^{G}(\eta,X,U) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{i}_{M'}^{H'}(Y,V),$$

où $\gamma_{\psi}(\mathfrak{h}(F))$ et $\gamma_{\psi}(\mathfrak{h}'(F))$ sont des constantes de Weil (voir la section 2.2 du chapitre 5). 2) Soient $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ et $U \in \mathfrak{s}_{rs}(F)$. Si $\eta(X) \neq \eta(U)$, alors

 $\hat{i}_M^G(\eta, X, U) = 0.$

$$M_M(\eta, X, U) = 0$$

Comme dans [58], on utilise la méthode globale de Waldspurger sur le transfert endoscopique [52] pour montrer 1) et une méthode locale pour montrer 2). Pour montrer 1), on définit une notion d'intégrales orbitales pondérées associées (voir la définition 5.2 du chapitre 5) et prouve que cette propriété commute sous certaines restrictions avec la transformée de Fourier (voir le théorème 5.3 du chapitre 5). Sa preuve peut être considérée comme une application de presque tous les résultats des chapitres précédents. Ensuite, on peut extraire les relations entre $\hat{i}_M^G(\eta, X, \cdot)$ et $\hat{i}_{M'}^{H'}(Y, \cdot)$ à l'aide du lemme de Labesse [37, lemme 1.7.1]. Ces étapes sont proches de celles de [14]. Cependant, alors que le lemme fondamental pondéré pour les formes intérieures est tautologique dans *loc. cit.*, la condition d'annulation du lemme fondamental pondéré est ici plus subtile. On utilise la cohomologie galoisienne abélianisée (voir [38]) pour résoudre quelques difficultés techniques (voir §4.3-4.4 et la preuve de la proposition 11.2 du chapitre 5).

CHAPTER 1

Introduction

This PhD project is inspired by a conjecture of Guo-Jacquet [23] which generalises Waldspurger's well-known theorem [50] relating toric periods to central values of automorphic L-functions for GL_2 to higher ranks. It is one of the first interesting examples in the relative Langlands programme initiated by Jacquet and his collaborators and systematically proposed by Sakellaridis-Venkatesh [46]. An effective tool for attacking such kind of problems is the relative trace formula which was first introduced by Jacquet [29] to reprove Waldspurger's result. The aims of this thesis under the supervision of Pierre-Henri Chaudouard include developing infinitesimal variants of global and local Guo-Jacquet trace formulae and establishing necessary local results of comparison for further study of this conjecture. We expect that some methods here would also be useful in other relative trace formulae.

1. Guo-Jacquet conjecture and the relative trace formula approach

Let us briefly recall the Guo-Jacquet conjecture proposed in [23]. Let E/F be a quadratic extension of number fields and η the quadratic character of $\mathbb{A}^{\times}/F^{\times}$ attached to it, where \mathbb{A} denotes the ring of adèles of F. Let $G := GL_{2n}$ and let $H := GL_n \times GL_n$ be its subgroup. All the groups considered are defined over F. Let π be a cuspidal automorphic representation of $G(\mathbb{A})$ with trivial central character. We say that π is H-distinguished if the two linear forms (called "periods") on it

$$\mathcal{P}_H:\phi\mapsto \int_{H(F)Z(\mathbb{A})\backslash H(\mathbb{A})}\phi(h)dh$$

and

$$\mathcal{P}_{H,\eta}:\phi\mapsto \int_{H(F)Z(\mathbb{A})\backslash H(\mathbb{A})}\phi(h)\eta(\det(h))dh$$

are both non-zero, where Z denotes the centre of G. This property is directly related to the nonvanishing of some central L-values by Friedberg-Jacquet's work [22]. We also need to consider another pair of groups. Denote by X(E) the set of isomorphic classes of quaternion algebras D/F in which Eembeds. For any $D \in X(E)$, let $G' = GL_{n,D}$ be the algebraic group defined over F whose F-points are $GL_n(D)$ and let $H' = \operatorname{Res}_{E/F} GL_{n,E}$ be its subgroup. Let π' be a cuspidal automorphic representation of $G'(\mathbb{A})$ with trivial central character. We say that π' is H'-distinguished if the linear form on it

$$\mathcal{P}_{H'}:\phi\mapsto \int_{H'(F)Z(\mathbb{A})\backslash H'(\mathbb{A})}\phi(h)dh$$

is not zero, where we identify the centre of G' with Z. One direction of the Guo-Jacquet conjecture says that if π' is H'-distinguished and π is deduced from π' by the Jacquet-Langlands correspondence, then π is H-distinguished. One may also expect a converse at least when n is odd. For n = 1, these were known by Waldspurger [50] and reproved by Jacquet [29] via relative trace formulae.

Now we formally describe the relative trace formula approach following Jacquet [29] to this conjecture. Let f_G be a smooth function on $G(\mathbb{A})$ with compact support. As an analogue of Arthur-Selberg's trace formula, the relative trace formula for the case of (G, H) roughly says that there are two ways to write the integral (viewed as a distribution)

(1.0.1)
$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \mathbb{K}_{f_G}(x,y)\eta(\det(x))dxdy,$$

where $G(\mathbb{A})^1$ denotes the subset of elements in $G(\mathbb{A})$ whose determinants are of absolute value 1 and $\mathbb{K}_{f_G}(x,y) := \sum_{\gamma \in G(F)} f_G(x^{-1}\gamma y)$. The geometric side is expected to be a sum of relative orbital integrals while the spectral side should be an expansion of periods. Similarly, one could imagine another formula for the case of (G', H'). Then the comparison of periods of different pairs of groups predicted by the Guo-Jacquet conjecture is reduced to the comparison of relative orbital integrals, for which we have more tools such as Guo's fundamental lemma [23] and Zhang's smooth transfer [58].

This approach has been adopted by Feigon-Martin-Whitehouse [21] to obtain some partial results. However, we have neglected analytic difficulty in the above discussion. That is to say, an obstruction of the relative trace formula approach to Guo-Jacquet conjecture is the following problem.

PROBLEM 1.1. The double integral (1.0.1) is not convergent and neither are two ways of its expansions.

The above problem accounts for some restrictive local conditions in the main results of [21] based on a simple relative trace formula. If one hopes to remove these seemingly artificial restriction and obtain information about all cuspidal representations, this problem can not be ignored. Therefore, it is necessary to establish valid and general relative trace formulae for the cases of (G, H) and (G', H') rather than formal or simple ones. Then we are supposed to compare new terms appearing in these formulae apart from ordinary relative orbital integrals.

Such a problem also existed in the classical Arthur-Selberg trace formula and Arthur introduced a truncation process in [3] and [4] to solve it. In the relative setting, Jacquet-Lapid-Rogawski povided us with the first example of the so-called "mixed truncation" in [30], whose rough idea is defining a truncation compatible with different groups concerned. These work and others set good examples for our study of Problem 1.1.

We shall focus on an infinitesimal variant of Problem 1.1 and the comparison of new terms involved in this thesis. It means that we shall work on the tangent space of a symmetric space (called an infinitesimal symmetric space). This would be of interest for at least two reasons. For one thing, it is close to the geometric side of the original relative trace formula and suggests us a way to tackle the original problem, but the spectral side of the relative trace formula is replaced by the Fourier transform of the geometric side where the harmonic analysis is simpler. For another, an infinitesimal variant of Guo-Jacquet trace formulae is also useful in the comparison of relative trace formulae between two cases; for example, its simple version has been used by Zhang [58] to prove the smooth transfer.

2. Main results

2.1. An infinitesimal variant of Guo-Jacquet trace formulae. We first solve an infinitesimal variant of Problem 1.1 in Chapter 2 and Chapter 3. It is the main global result in this thesis.

2.1.1. The case of (G, H). We remark that the setting in Chapter 2 is a bit more general to include the case in [57], but we shall focus on the pair (G, H) defined above in the introduction. Denote by S the symmetric space G/H. Notice that the double integral (1.0.1) can be formally written as a single integral

$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} K_{f_S}(x)\eta(\det(x))dx,$$

where $f_S(x) := \int_{H(\mathbb{A})\cap G(\mathbb{A})^1} f_G(xy) dy$ defines a smooth function on $S(\mathbb{A})$ with compact support and $K_{f_S}(x) := \sum_{\gamma \in S(F)} f_S(x^{-1}\gamma x)$. If one replaces S with the tangent space $\mathfrak{s} \simeq \mathfrak{gl}_n \oplus \mathfrak{gl}_n$ at the neutral element, then Problem 1.1 appears as the divergence of the integral

(2.1.1)
$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} k_f(x)\eta(\det(x))dx,$$

where f is a Bruhat-Schwartz function on $\mathfrak{s}(\mathbb{A})$ and $k_f(x) := \sum_{\gamma \in \mathfrak{s}(F)} f(x^{-1}\gamma x)$.

We replace $k_f(x)$ with some explicit $k_f^T(x)$ defined in §4 in Chapter 2 to make (2.1.1) absolutely convergent, where $T \in \mathbb{R}^{2n}$ is a truncation parameter. This key definition combines ideas from [28][61][17] for the decomposition of $\mathcal{H}(\mathbb{A})$ and [40] for the decomposition of $\mathfrak{s}(\mathbb{A})$. To describe the geometric expansion, we define a relation of equivalence on $\mathfrak{s}(F)$: two element in $\mathfrak{s}(F)$ are equivalent if and only if they lie in the same fibre of the categorical quotient $\mathfrak{s}//H$. Denote by \mathcal{O} the set of classes of equivalence. For all $\mathfrak{o} \in \mathcal{O}$, we define $k_{f,\mathfrak{o}}^T(x)$ similarly by replacing $\mathfrak{s}(F)$ with \mathfrak{o} . The main result of Chapter 2 is the following theorem which gives the geometric expansion of (2.1.1) with $k_f(x)$ replaced by $k_f^T(x)$.

THEOREM 2.1 (see Theorem 4.11 in Chapter 2). For T in a suitable cone in \mathbb{R}^{2n} ,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1}k_{f,\mathfrak{o}}^T(x)\eta(\det(x))dx$$

is absolutely convergent.

Denote by $J_{\mathfrak{o}}^T(\eta, f)$ the integral (2.1.1) with $k_f(x)$ replaced by $k_{f,\mathfrak{o}}^T(x)$. We prove that it is an exponential polynomial in T (see Corollary 5.6 in Chapter 2). Denote by $J_{\mathfrak{o}}(\eta, f)$ its constant term. In the infinitesimal setting, the geometric expansion of the Fourier transform of f (denoted by \hat{f}) plays the role of the original spectral side (cf. [13]). Then the infinitesimal variant of Guo-Jacquet trace formula for the case of (G, H) below is deduced from the Poisson summation formula.

THEOREM 2.2 (see Theorem 7.1 in Chapter 2). For all Bruhat-Schwartz function f on $\mathfrak{s}(\mathbb{A})$, we have the equality

$$\sum_{\mathfrak{o}\in\mathcal{O}} J_{\mathfrak{o}}(\eta, f) = \sum_{\mathfrak{o}\in\mathcal{O}} J_{\mathfrak{o}}(\eta, \hat{f}).$$

Additionally, for most (namely regular semi-simple) \mathfrak{o} , we can write $J_{\mathfrak{o}}(\eta, f)$ as an explicit weighted orbital integral; in particular, for elliptic \mathfrak{o} , we see that $J_{\mathfrak{o}}(\eta, f)$ is nothing but the ordinary orbital integral. We say a semi-standard parabolic subgroup of G is ω -stable if it contains $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$.

THEOREM 2.3 (see Theorem 9.2 in Chapter 2). Let $\mathfrak{o} \in \mathcal{O}$ be a regular semi-simple class, P_1 an ω -stable relatively standard parabolic subgroup of G and $X_1 \in \mathfrak{o}$ an elliptic element relative to P_1 (defined in Section 9.2 in Chapter 2). For a Bruhat-Schwartz function f on $\mathfrak{s}(\mathbb{A})$, we have

$$J_{\mathfrak{o}}(\eta, f) = \operatorname{vol}([H_{X_1}]) \cdot \int_{H_{X_1}(\mathbb{A}) \setminus H(\mathbb{A})} f(x^{-1}X_1x) v_{P_1}(x) \eta(\det(x)) dx,$$

where H_{X_1} denotes the centraliser of X_1 in H, $vol([H_{X_1}])$ is its associated volume and $v_{P_1}(x)$ is the volume of some convex hull.

The weight $v_{P_1}(x)$ is exactly the restriction to $H(\mathbb{A})$ of Arthur's weight in [3] for $G(\mathbb{A})$. It is interesting that our weighted orbital integral shares the same weight as in the twisted trace formula (see [39, p. 131]) for the twisted space $(GL_n \times GL_n) \rtimes \sigma$, where $\sigma(x, y) := (y, x)$.

2.1.2. The case of (G', H'). Again we study a more general setting in Chapter 3 than the pair (G', H') above. Actually we deal with a central simple algebra \mathfrak{g}' over F containing E and the centraliser \mathfrak{h}' of E in \mathfrak{g}' . We define G' and H' as the groups of invertible elements in \mathfrak{g}' and \mathfrak{h}' respectively. This more general case is suggested by the related local conjecture of Prasad and Takloo-Bighash [44, Conjecture 1] and is necessary for completing the converse direction of Guo-Jacquet conjecture for n even.

Up to conjugation by G'(F), the symmetric pair (G', H') is reduced to one of the two cases below (see Proposition 3.7 in Chapter 3). These two cases can be treated similarly.

Case I: $(G', H') \simeq (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,D'})$, where D is a central division algebra over F containing E, and $D' := \operatorname{Cent}_D(E)$ is the centraliser of E in D.

Case II: $(G', H') \simeq (GL_{n,D}, \operatorname{Res}_{E/F}GL_{\frac{n}{2}, D\otimes_F E})$, where D is a central division algebra over F such that there is no embedding $E \to D$ as F-algebras.

Notice that (G', H') is the same as (G, H) after the base change to an algebraic closure of F containing E. We have similar construction and results for the case of (G', H') as follows. Denote by $H'(\mathbb{A})^1$ the subset of elements in $H'(\mathbb{A})$ whose reduced norms are of absolute value 1. Denote by \mathcal{O}' the set of classes of equivalence on $\mathfrak{s}'(F)$ defined by the categorical quotient $\mathfrak{s}'//H'$. Let f' be a Bruhat-Schwartz function on $\mathfrak{s}'(\mathbb{A})$ and $\mathfrak{o} \in \mathcal{O}'$. For a truncation parameter $T \in \mathbb{R}^n$ and $x \in H'(x)$, we define some truncated kernel $k_{f',\mathfrak{o}}^T(x)$ in §4 in Chapter 3.

THEOREM 2.4 (see Theorem 4.2 in Chapter 3). For T in a suitable cone in \mathbb{R}^n ,

$$\sum_{\mathfrak{o}\in\mathcal{O}'}\int_{H'(F)\backslash H'(\mathbb{A})^1}k^T_{f',\mathfrak{o}}(x)dx$$

is absolutely convergent.

For $\mathfrak{o} \in \mathcal{O}'$, denote by $J^T_{\mathfrak{o}}(f')$ the summand of the last sum. We prove that it is a polynomial in T (see Corollary 5.3 in Chapter 3). Denote by $J_{\mathfrak{o}}(f')$ its constant term. Denote by \hat{f}' the Fourier transform of f'. Here is the infinitesimal variant of Guo-Jacquet trace formula for the case of (G', H').

THEOREM 2.5 (see Theorem 7.1 in Chapter 3). For all Bruhat-Schwartz function f' on $\mathfrak{s}'(\mathbb{A})$, we have the equality

$$\sum_{\mathfrak{o}\in\mathcal{O}'}J_{\mathfrak{o}}(f')=\sum_{\mathfrak{o}\in\mathcal{O}'}J_{\mathfrak{o}}(\hat{f}').$$

As before, we can write the regular semi-simple terms as explicit weighted orbital integrals with the same weights as Arthur's in [3]. For $(G', H') = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,E})$, we obtain the same weights appearing in the twisted trace formula (see [39, p. 131]) for the twisted space $(\operatorname{Res}_{E/F}GL_{n,E}) \rtimes \sigma'$, where σ' is the nontrivial Galois conjugation.

THEOREM 2.6 (see Theorem 9.2 in Chapter 3). Let $\mathfrak{o} \in \mathcal{O}'$ be a regular semi-simple class, P'_1 a standard parabolic subgroup of H' and $Y_1 \in \mathfrak{o}$ an elliptic element with respect to P'_1 (defined in Section 9 in Chapter 3). For all Bruhat-Schwartz function f' on $\mathfrak{s}'(\mathbb{A})$, we have

$$J_{\mathfrak{o}}(f') = \operatorname{vol}([H'_{Y_1}]) \cdot \int_{H'_{Y_1}(\mathbb{A}) \setminus H'(\mathbb{A})} f'(\operatorname{Ad}(x^{-1})(Y_1)) v_{P'_1}(x) dx,$$

where H'_{Y_1} denotes the centraliser of Y_1 in H', $vol([H'_{Y_1}])$ is the volume associated to H'_{Y_1} and $v_{P'_1}(x)$ is the volume of some convex hull.

In fact, this case is even simpler than that of (G, H) in some sense; for example, $J_{\sigma}^{T}(f')$ is a pure polynomial in T instead of an exponential polynomial. One reason for the simplicity is that the maximal F-split torus in the centre of H' is the same as that of G'. Moreover, there is a bijection between the set of semi-standard parabolic subgroups of H' and the set of so-called relatively standard parabolic subgroups of G'. However, there are some additional rationality issues in this case.

2.2. Some local results for comparison of regular semi-simple terms. The terms $J_{\mathfrak{o}}(\eta, f)$ and $J_{\mathfrak{o}}(f')$ above associated to regular semi-simple but not elliptic classes $\mathfrak{o} \in \mathcal{O}$ and $\mathfrak{o} \in \mathcal{O}'$ respectively, which are global weighted orbital integrals, are the first new terms that we need to study and compare for the application of Guo-Jacquet trace formulae. The main difficulty to study these global distributions compared to those in [60] or [61] is that we are facing noninvariant distributions under the conjugation of $H(\mathbb{A})$ or $H'(\mathbb{A})$. This is close to the situation of the classical Arthur-Selberg trace formula. While the standard but quite difficult procedure is to make our trace formula invariant as Arthur did in [6] and [7], a different way suggested by Labesse in [37] is to compare directly noninvariant distributions. We shall follow the latter approach. Their corresponding local weighted orbital integrals are the main objects in §10 in Chapter 3, Chapter 4 and Chapter 5.

Let E/F be a quadratic extension of non-archimedean local fields of characteristic zero. Let η be the quadratic character of F/NE^{\times} attached to E/F, where NE^{\times} denotes the norm of E^{\times} . We define symmetric pairs as in the global case. Denote by \mathfrak{s}_{rs} the set of regular semi-simple elements in \mathfrak{s} . Denote by $\mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ (resp. $\mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$) the space of locally constant and compactly supported functions on $\mathfrak{s}(F)$ (resp. $\mathfrak{s}'(F)$). Let M be an ω -stable Levi subgroup of G, i.e., it is a Levi factor of some ω -stable parabolic subgroup. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. For $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we define the local weighted orbital integral

$$J_{M}^{G}(\eta, X, f) := |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{H_{X}(F)\backslash H(F)} f(\mathrm{Ad}(x^{-1})(X))\eta(\det(x))v_{M}^{G}(x)dx$$

where $|D^{\mathfrak{s}}(X)|_F$ is the Weyl discriminant and $v_M^G(x)$ is the corresponding local weight function. Similarly, let M' be a Levi subgroup of H' and $Y \in (\widetilde{\mathfrak{m}}' \cap \mathfrak{s}'_{rs})(F)$, where \widetilde{M}' is the Levi subgroup of G' such that $\widetilde{M}' \cap H' = M'$. For $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$, we define the local weighted orbital integral

$$J_{M'}^{H'}(Y,f') := |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} f'(\mathrm{Ad}(x^{-1})(Y)) v_{M'}^{H'}(x) dx,$$

where $|D^{\mathfrak{s}'}(Y)|_F$ is the Weyl discriminant and $v_{M'}^{H'}(x)$ is the corresponding local weight function.

There is an injection $M' \mapsto M$ from the set of Levi subgroups of H' into the set of ω -stable Levi subgroups of G. We shall suppose that G and G' have the same dimension n^2 . Since $\mathfrak{s}//H \simeq \mathbf{A}^n \simeq \mathfrak{s}'//H'$, if $X \in \mathfrak{s}_{rs}(F)$ and $Y \in \mathfrak{s}'_{rs}(F)$ have the same image in \mathbf{A}^n , we say that they have matching orbits. More generally, fix a pair of matching Levi subgroups M and M' of G and H' respectively. We have the notion of M-matching orbits defined by blocks for $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$.

2.2.1. The weighted fundamental lemma. We provide the first evidence of noninvariant comparison of Guo-Jacquet trace formulae in §10 in Chapter 3. In the comparison of geometric sides of Guo-Jacquet trace formulae, an important case is the so-called fundamental lemma. It roughly says that at almost all unramified places, some basic functions on G(F) and G'(F) should have associated local relative orbital integrals on matching orbits. Guo [23] proved it for the units of spherical Hecke algebras with the help of the base change fundamental lemma for the full spherical Hecke algebras for GL_n known by Kottwitz [36, Lemma 8.8] and Arthur-Clozel [10, Theorem 4.5 in Chapter 1]. An infinitesimal version [58, Lemma 5.18] was used by Zhang to prove the smooth transfer of ordinary orbital integrals for Guo-Jacquet trace formulae following the same philosophy of Waldspurger's work [52] on the endoscopic transfer.

Inspired by [37, Definition III.3.2], we define in §10 in Chapter 3 the notion of "strongly associated" pairs of locally constant and compactly supported functions on $\mathfrak{s}(F)$ and $\mathfrak{s}'(F)$. Roughly speaking, two functions are strongly associated if they have associated weighted orbital integrals on matching orbits. For almost all unramified places, (G', H') is isomorphic to $(GL_{2n}, \operatorname{Res}_{E/F}GL_{n,E})$ and $\mathfrak{s}'(F) \simeq \mathfrak{gl}_n(E)$. Denote by \mathcal{O}_F (resp. \mathcal{O}_E) the ring of integers in F (resp. E). Let f_0 and f'_0 be the characteristic functions on $\mathfrak{s}(\mathcal{O}_F) \simeq (\mathfrak{gl}_n \oplus \mathfrak{gl}_n)(\mathcal{O}_F)$ and $\mathfrak{s}'(\mathcal{O}_F) \simeq \mathfrak{gl}_n(\mathcal{O}_E)$ respectively. Because we get the same weights for the cases of Guo-Jacquet as those in twisted trace formulae, we are able to reduce the following weighted version of [58, Lemma 5.18] to Labesse's work [37] on the base change for GL_n . We show that for almost all unramified places v, f_0 and f'_0 are strongly associated.

THEOREM 2.7 (see Theorem 10.9 in Chapter 3 for a precise and general statement). Let M and M' be a pair of matching Levi subgroups of G and H' respectively. We have

(1) if $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ have M-matching orbits, then

$$\kappa(X)J_M^G(\eta, X, f_0) = J_{M'}^{H'}(Y, f_0)$$

where $\kappa(X)$ is a transfer factor;

(2) if
$$X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$$
 satisfies $\det(AB) \notin NE^{\times}$, then $J_M^G(\eta, X, f_0) = 0.$

2.2.2. An infinitesimal local trace formula. Our next goal is to prove the noninvariant transfer of weighted orbital integrals using the weighted fundamental lemma following the strategy of [14] and [15] on the stable base change. To achieve this, we need to prepare some results of local harmonic analysis as in [51]. We prove an infinitesimal variant of invariant local trace formulae for the case of Guo-Jacquet in Chapter 4 following [51] and [8].

We focus on the case of (G, H) here for illustration. We define the (H, η) -invariant local weighted orbital integral $I_M^G(\eta, X, f)$ by Arthur's standard process from the above noninvariant local weighted orbital integral $J_M^G(\eta, X, f)$. For $f, f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we define (see §3.2 in Chapter 4 for notations)

$$I^{G}(\eta, f, f') := \sum_{\substack{M \in \mathscr{L}^{G, \omega}(M_{0}) \\ I^{G}_{M}(\eta, X, \hat{f}) I^{G}_{G}(\eta, X, f') dX.}} |W^{GL_{n}|^{-1}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W^{GL_{n}|^{-1}}(-1)^{\dim(A_{M}/A_{G})} |W^{GL_{n}|^{-1}}(-1)^{$$

The main result in Chapter 4 is the following invariant local trace formula.

THEOREM 2.8 (see Theorem 9.1 in Chapter 4). We have the equality

$$I^G(\eta, f, f') = I^G(\eta, f', f).$$

It is deduced from a noninvariant local trace formula (see Theorem 5.3 in Chapter 4), which is essentially a consequence of the Plancherel formula combined with a truncation process. Some other local results including Howe's finiteness for weighted orbital integrals and representability of the Fourier transform of weighted orbital integrals are given during the proof.

PROPOSITION 2.9 (see Corollary 6.6 in Chapter 4). Let r be an open compact subgroup of $\mathfrak{s}(F)$, M be an ω -stable Levi subgroup of G and $\sigma \subseteq (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Denote by $\mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/r)$ the subspace of $\mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ consisting of the functions invariant by translation of r. Suppose that there exists a compact subset $\sigma_{0} \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ such that $\sigma \subseteq \operatorname{Ad}((M \cap H)(F))(\sigma_{0})$. Then there exists a finite subset $\{X_{i} : i \in I\} \subseteq \sigma$ and a finite subset $\{f_{i} : i \in I\} \subseteq \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/r)$ such that for all $X \in \sigma$ and all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/r)$, we have the equality

$$J_M^G(\eta, X, f) = \sum_{i \in I} J_M^G(\eta, X_i, f) J_M^G(\eta, X, f_i).$$

PROPOSITION 2.10 (see Proposition 7.2 in Chapter 4). Let M be an ω -stable Levi subgroup of G and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Then there exists a locally constant function $\hat{j}_M^G(\eta, X, \cdot)$ on $\mathfrak{s}_{rs}(F)$ such that

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), J^{G}_{M}(\eta, X, \hat{f}) = \int_{\mathfrak{s}(F)} f(Y) \hat{j}^{G}_{M}(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY.$$

We also obtain a vanishing property at "infinity" analogous to [14, Proposition 2.2].

PROPOSITION 2.11 (see Proposition 10.1 in Chapter 4). Let $M \neq G$ be an ω -stable Levi subgroup of G. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in \mathfrak{s}_{rs}(F)$. Then there exists $N \in \mathbb{N}$ such that if $\lambda \in F^{\times}$ satisfies $v_F(\lambda) < -N$, we have

$$\hat{i}_{M}^{G}(\eta, \lambda X, Y) = 0,$$

where $\hat{i}_M^G(\eta, X, \cdot)$ is the (H, η) -invariant analogue of $\hat{j}_M^G(\eta, X, \cdot)$.

These results are of independent interest. Analogues of all the results above are obtained for the case of (G', H').

2.2.3. Certain identities between Fourier transforms of weighted orbital integrals. As mentioned above, the next step of our research is to compare regular semi-simple terms in the geometric sides of Guo-Jaquect trace formulae as in [14] and [15]. This should serve as an example of noninvariant comparison in the relative context. We obtain some relations between Fourier transforms of weighted orbital integrals in Chapter 5 which generalise some of the main results in [58] and are analogues for Guo-Jacquet trace formulae of [14].

Let M be an ω -stable Levi subgroup of G and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Denote by $\hat{J}_{M}^{G}(\eta, X, \cdot)$ (resp. $\hat{I}_{M}^{G}(\eta, X, \cdot)$) the Fourier transform of the distribution $J_{M}^{G}(\eta, X, \cdot)$ (resp. $I_{M}^{G}(\eta, X, \cdot)$). Let $\hat{j}_{M}^{G}(\eta, X, \cdot)$ (resp. $\hat{i}_{M}^{G}(\eta, X, \cdot)$) be the locally constant function on $\mathfrak{s}_{rs}(F)$ representing $\hat{J}_{M}^{G}(\eta, X, \cdot)$ (resp. $\hat{I}_{M}^{G}(\eta, X, \cdot)$). For a Levi subgroup M' of H' and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, we similarly obtain locally constant functions $\hat{j}_{M'}^{H'}(Y, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}'_{rs}(F)$.

The functions $\hat{j}_M^G(\eta, X, \cdot)$ is decomposed as their invariant analogues $\hat{i}_M^G(\eta, X, \cdot)$ and weight functions v_M^G . The decomposition for the functions $\hat{j}_{M'}^{H'}(Y, \cdot)$ is similar. In order to obtain relations between $\hat{j}_M^G(\eta, X, \cdot)$ and $\hat{j}_{M'}^{H'}(Y, \cdot)$, which is part of the noninvariant comparison, we shall focus on the relations between $\hat{i}_M^G(\eta, X, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ in Chapter 5.

Fix a pair of matching Levi subgroups M' and M of H' and G respectively. For $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$, we denote $\eta(X) := \eta(\det(AB))$. Our main result in Chapter 5 is as follows.

THEOREM 2.12 (see Corollary 5.6 and Proposition 5.9 in Chapter 5). 1) Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ have M-matching orbits. Let $U \in \mathfrak{s}_{rs}(F)$ and $V \in \mathfrak{s}'_{rs}(F)$ have matching orbits. Then we have the equality

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X)\kappa(U)\hat{i}_{M}^{G}(\eta,X,U) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{i}_{M'}^{H'}(Y,V)$$

where $\gamma_{\psi}(\mathfrak{h}(F))$ and $\gamma_{\psi}(\mathfrak{h}'(F))$ are Weil constants (see Section 2.2 in Chapter 5).

2) Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in \mathfrak{s}_{rs}(F)$. If $\eta(X) \neq \eta(U)$, then

$$\hat{i}_M^G(\eta, X, U) = 0.$$

As in [58], we use Waldspurger's global method on the endoscopic transfer [52] to show 1) and a local method to show 2). To show 1), we define a notion of matching weighted orbital integrals (see Definition 5.2 in Chapter 5) and prove that this property commutes with Fourier transform under some restriction (see Theorem 5.3 in Chapter 5). Its proof can be viewed as an application of almost all results in previous chapters. Then we may extract the relations between $\hat{i}_M^G(\eta, X, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ with the help of Labesse's lemma [37, Lemma 1.7.1]. These steps are close to those in [14]. However, while the weighted fundamental lemma for inner forms is tautological in *loc. cit.*, the vanishing condition of the weighted fundamental lemma here is more subtle. We use abelian Galois cohomology (see [38]) to deal with some technical difficulties (see §4.3-4.4 and the proof of Proposition 11.2 in Chapter 5).

CHAPTER 2

An infinitesimal variant of Guo-Jacquet trace formula: the case of $(GL_{2n,D}, GL_{n,D} \times GL_{n,D})$

We establish an infinitesimal variant of Guo-Jacquet trace formula for the case of $(GL_{2n,D}, GL_{n,D} \times GL_{n,D})$. It is a kind of Poisson summation formula obtained by an analogue of Arthur's truncation process. It consists in the equality of the sums of two types of distributions which are non-equivariant in general: one type is associated to rational points in the categorical quotient, while the other type is the Fourier transform of the first type. For regular semi-simple points in the categorical quotient, we obtain weighted orbital integrals.

1. Introduction

The Guo-Jacquet conjecture proposed in [23] is a possible generalisation in higher dimensions of Waldspurger's well-known theorem on central values of automorphic *L*-functions for GL_2 . We briefly recall it as follows. Let E/F be a quadratic extension of number fields and η the quadratic character of $\mathbb{A}^{\times}/F^{\times}$ attached to it, where \mathbb{A} denotes the ring of adèles of *F*. Consider the group $G = GL_{2n}$ and its subgroup $H = GL_n \times GL_n$ defined over *F*. Let π be a cuspidal automorphic representation of $G(\mathbb{A})$ with trivial central character. We say that π is *H*-distinguished if the two linear forms (called "periods") on it

$$\mathcal{P}_H:\phi\mapsto\int_{H(F)Z(\mathbb{A})\backslash H(\mathbb{A})}\phi(h)dh$$

and

$$\mathcal{P}_{H,\eta}:\phi\mapsto \int_{H(F)Z(\mathbb{A})\backslash H(\mathbb{A})}\phi(h)\eta(\det(h))dh$$

are both non-zero, where Z denotes the centre of G. This property is directly connected with the nonvanishing of some central L-values (see Friedberg-Jacquet's work [22]). We also need to deal with another pair of groups. Let X(E) denote the set of isomorphic classes of quaternion algebras D/F in which Eembeds. For any $D \in X(E)$, let $G_D = GL_{n,D}$ be the algebraic group defined over F whose F-points are $GL_n(D)$ and $H_D = Res_{E/F}GL_{n,E}$ be its subgroup. Let π_D be a cuspidal automorphic representation of $G_D(\mathbb{A})$ with trivial central character. We say that π_D is H_D -distinguished if the linear form on it

$$\mathcal{P}_{H_D}:\phi\mapsto\int_{H_D(F)Z(\mathbb{A})\setminus H_D(\mathbb{A})}\phi(h)dh,$$

is not zero, where we identify the centre of G_D with Z. One part of the Guo-Jacquet conjecture says that if π_D is H_D -distinguished and π is deduced from π_D by the Jacquet-Langlands correspondence, then π is H-distinguished. We can also expect a converse at least when n is odd. For n = 1, these were known by Waldspurger [50] and reproved by Jacquet [29].

Now we formally describe the approach of relative trace formulae following Jacquet [29]. This was adopted by Feigon-Martin-Whitehouse [21] to obtain some partial results. Let f^G be a smooth function on $G(\mathbb{A})$ with compact support. As an analogue of Arthur-Selberg trace formula, the relative trace formula for the case (G, H) roughly says that there are two ways to write the integral (viewed as a distribution)

$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \mathbb{K}_{f^G}(x,y)\eta(\det(x))dxdy,$$

where $G(\mathbb{A})^1$ denotes the elements in $G(\mathbb{A})$ with absolute-value-1 determinant and $\mathbb{K}_{f^G}(x, y) = \sum_{\gamma \in G(F)} f^G(x^{-1}\gamma y)$. The geometric side is expected to be a sum of relative (weighted) orbital integrals while the spectral side should be an expansion of periods. Similarly there is also another formula for the

case of (G_D, H_D) . Then the comparison of periods of different pairs of groups predicted by the Guo-Jacquet conjecture is reduced to the comparison of relative (weighted) orbital integrals, for which there are already some works such as Guo's fundamental lemma [23] and Zhang's transfer [58].

However, we have neglected analytic difficulty in the above discussion. That is to say, the double integral above is not convergent and neither are two ways of its expansions. This is the reason why some restrictive local conditions are needed in the main results of [21] though they seem kind of artificial. The aim of this article is to solve this kind of problem at the level of Lie algebras for the case of (G, H). Denote by S the symmetric space G/H. Notice that such double integral can be formally written as a single integral

$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} K_{f^S}(x)\eta(\det(x))dx,$$

where $f^{S}(x) = \int_{H(\mathbb{A})\cap G(\mathbb{A})^{1}} f^{G}(xy) dy$ defines a smooth function on $S(\mathbb{A})$ with compact support and $K_{f^{S}}(x) = \sum_{\gamma \in S(F)} f^{S}(x^{-1}\gamma x)$. Replacing S by its tangent space $\mathfrak{s} \simeq \mathfrak{gl}_{n} \oplus \mathfrak{gl}_{n}$ at the neutral element, we are faced with the divergence of the integral

$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} k_f(x)\eta(\det(x))dx,$$

where f is a Bruhat-Schwartz function on $\mathfrak{s}(\mathbb{A})$ and $k_f(x) = \sum_{\gamma \in \mathfrak{s}(F)} f(x^{-1}\gamma x)$.

Our main results can be described as follows.

First of all, as in [3], we replace $k_f(x)$ with some explicit $k_f^T(x)$ (see its definition in (4.0.1) and (4.2.1)) to make the last integral absolutely convergent, where $T \in \mathbb{R}^{2n}$ is a truncation parameter. Moreover, there is a relation of equivalence on $\mathfrak{s}(F)$ defined by the categorical quotient $\mathfrak{s}//H$; we denote by \mathcal{O} the set of classes of equivalence. For each class $\mathfrak{o} \in \mathcal{O}$, we define $k_{f,\mathfrak{o}}^T(x)$ and its integral similarly by replacing $\mathfrak{s}(F)$ with \mathfrak{o} . Then we have

$$k_f^T(x) = \sum_{\mathbf{o} \in \mathcal{O}} k_{f,\mathbf{o}}^T(x)$$

and prove the following theorem which gives the geometric expansion of

$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} k_f^T(x)\eta(\det(x))dx.$$

THEOREM 1.1 (see Theorem 4.11). For T in a suitable cone in \mathbb{R}^{2n} ,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1}k_{f,\mathfrak{o}}^T(x)\eta(\det(x))dx$$

is absolutely convergent.

Moreover, we see that each summand in the geometric expansion is a sum of products of polynomials and exponential functions in T. In fact, most (namely regular semi-simple) terms are simply polynomial distributions.

THEOREM 1.2 (see Corollary 5.6). For T in a suitable cone in \mathbb{R}^{2n} and each $\mathfrak{o} \in \mathcal{O}$,

$$J^{T}_{\mathfrak{o}}(\eta, f) := \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^{1}} k^{T}_{f, \mathfrak{o}}(x) \eta(\det(x)) dx$$

is an exponential polynomial in T. In particular, if \mathfrak{o} is regular semisimple, it is a polynomial in T.

This property allows us to take the constant term $J_{\mathfrak{o}}(\eta, f)$ of $J_{\mathfrak{o}}^{T}(\eta, f)$ to eliminate the truncation parameter. In the infinitesimal setting, the geometric expansion of the Fourier transform of f plays the role of the original spectral side (cf. [13]). Our infinitesimal variant of Guo-Jacquet trace formula equating the geometric developments of f and its Fourier transform (defined by (3.5.2) and denoted by \hat{f}) is the following, which essentially comes from the Poisson summation formula.

THEOREM 1.3 (see Theorem 7.1). For a Bruhat-Schwartz function f on $\mathfrak{s}(\mathbb{A})$, we have the equality

$$\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(\eta,f)=\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(\eta,\hat{f})$$

Such a formula should be of interest for at least two reasons. For one thing, it is close to but easier than its analogue for the symmetric space. For another, a simplified version of this formula (see [58, Theorem 8.4 and p. 1875]) has been used in Zhang's proof of the smooth transfer.

Unfortunately, the distributions $J_{\mathfrak{o}}(\eta, \cdot)$ on $\mathfrak{s}(\mathbb{A})$ that we obtained are non-equivariant under the conjugation of $H(\mathbb{A}) \cap G(\mathbb{A})^1$ in general, which is close to the situation in [3] and different from that in [61]. In fact, we have the following formula of non-equivariance.

PROPOSITION 1.4 (see Corollary 6.2). For a Bruhat-Schwartz function f on $\mathfrak{s}(\mathbb{A})$ and $y \in H(\mathbb{A}) \cap G(\mathbb{A})^1$, we denote $f^y(x) := f(yxy^{-1})$. Then

$$J_{\mathfrak{o}}(\eta, f^y) = \eta(\det(y)) \sum_Q J^Q_{\mathfrak{o}}(\eta, f^\eta_{Q,y})$$

where the sum on Q runs over all ω -stable relatively standard parabolic subgroups of G (defined in Section 5.2). Here $J^Q_{\mathfrak{o}}(\eta, \cdot)$ is an analogue of $J_{\mathfrak{o}}(\eta, \cdot)$ with G replaced by Q, and $f^{\eta}_{Q,y}$ is defined by (6.0.1) with s = 0.

Nevertheless, we can write regular semi-simple terms as explicit weighted orbital integrals whose weights are the restriction to $H(\mathbb{A})$ of Arthur's in [3] for $G(\mathbb{A})$.

THEOREM 1.5 (see Theorem 9.2). Let $\mathfrak{o} \in \mathcal{O}$ be a regular semi-simple class, P_1 an ω -stable relatively standard parabolic subgroup of G and $X_1 \in \mathfrak{o}$ an elliptic element relative to P_1 (defined in Section 9.2). For a Bruhat-Schwartz function f on $\mathfrak{s}(\mathbb{A})$, we have

$$J_{\mathfrak{o}}(\eta, f) = \operatorname{vol}([H_{X_1}]) \cdot \int_{H_{X_1}(\mathbb{A}) \setminus H(\mathbb{A})} f(x^{-1}X_1x) v_{P_1}(x) \eta(\det(x)) dx$$

where H_{X_1} denotes the centraliser of X_1 in H, $vol([H_{X_1}])$ is its associated volume and $v_{P_1}(x)$ is the volume of some convex hull.

This paper is organised in the following way. Section 2 and 3 are devoted to standard notation in Arthur's work on trace formulae and characterisation of \mathcal{O} in the specific symmetric pair that we consider respectively. We define the truncated kernel $k_{f,\mathfrak{o}}^T(x)$ and prove its integrability in Section 4. This key definition is partly inspired by [28] [61] [17] (for the decomposition of groups) and [40] (for the decomposition of linear spaces) apart from Arthur's pioneering work [3] and its Lie algebra variant [13]. Section 5 is about the quantitive behaviour of the distributions that we got with respect to the truncation parameter T. In Section 6, we study their variance under the conjugation of $H(\mathbb{A}) \cap G(\mathbb{A})^1$. In Section 7, the infinitesimal Guo-Jacquet trace formula for the case of $(GL_{2n}, GL_n \times GL_n)$ is given. Section 8 and 9 aim to express the regular semi-simple distribution as weighted orbital integrals.

Here are two final remarks. Firstly, actually we study the more general symmetric pair $(GL_{p+q,D}, GL_{p,D} \times GL_{q,D})$ instead of $(GL_{2n}, GL_n \times GL_n)$ and add an extra term $|\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s$ to the integrand in most of this article. Not only do we prefer more general results (including the case considered in [57] for instance) or possible applications (cf. [41] for the study of the first derivative of *L*-functions), but the study of the case where p = q and s = 0 also yields consideration on a more general setting. A simple reason for this comes from the structure of the intersection of *H* and semi-standard Levi subgroups of *G*. Secondly, there are some similarities between our case and the twisted trace formula (cf. [39]) for $(GL_n \times GL_n) \rtimes \sigma$ where σ exchanges two copies of GL_n . In fact, we obtain the same weights for regular semi-simple orbits. However, we shall see that more parabolic subgroups will be needed to define the truncation here. We shall return to its discussion at the end of this paper.

Acknowledgement. I would like to express my great appreciation to my PhD advisor Professor Pierre-Henri Chaudouard for introducing me to this problem and valuable suggestions during my preparation of this work. Part of this paper was revised during my visit to the Institute for Mathematical Sciences at the National University of Singapore and I would like to thank their hospitality. This work was supported by grants from Région Ile-de-France.

2. Notation

2.1. Roots and weights. Let F be a number field and G a reductive group defined over F. Denote by Z_G the centre of G. Fix a minimal Levi F-subgroup M_0 of G. All the following groups are assumed to be defined over F without further mention. We call a parabolic subgroup P of G semi-standard if $M_0 \subseteq P$. For any semi-standard parabolic subgroup P of G, we usually write M_P for the Levi factor

containing M_0 and N_P the unipotent radical. Denote by A_P the maximal F-split torus in the centre of M_P . Let $X(M_P)_F$ be the group of characters of M_P defined over F. Then define

$$\mathfrak{a}_P := \operatorname{Hom}_{\mathbb{Z}}(X(M_P)_F, \mathbb{R})$$

and its dual space

$$\mathfrak{a}_P^* := X(M_P)_F \otimes_{\mathbb{Z}} \mathbb{R},$$

which are both \mathbb{R} -linear spaces of dimension dim (A_P) . Notice that the restriction $X(M_P)_F \hookrightarrow X(A_P)_F$ induces an isomorphism

$$\mathfrak{a}_P^* \simeq X(A_P)_F \otimes_{\mathbb{Z}} \mathbb{R}.$$

Suppose that $P_1 \subseteq P_2$ are a pair of semi-standard parabolic subgroups of G. The restriction $X(M_{P_2})_F \hookrightarrow X(M_{P_1})_F$ induces $\mathfrak{a}_{P_2}^* \hookrightarrow \mathfrak{a}_{P_1}^*$ and its dual map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. Denote by $\mathfrak{a}_{P_1}^{P_2}$ the kernel of the latter map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. The restriction $X(A_{P_1})_F \twoheadrightarrow X(A_{P_2})_F$ induces $\mathfrak{a}_{P_1}^* \twoheadrightarrow \mathfrak{a}_{P_2}^*$ and its dual map $\mathfrak{a}_{P_2} \hookrightarrow \mathfrak{a}_{P_1}$. The latter map $\mathfrak{a}_{P_2} \hookrightarrow \mathfrak{a}_{P_1}$ provides a section of the previous map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. Thus we have decompositions $\mathfrak{a}_{P_1} = \mathfrak{a}_{P_2} \oplus \mathfrak{a}_{P_1}^{P_2}$

and

$$\mathfrak{a}_{P_1}^* = \mathfrak{a}_{P_2}^* \oplus (\mathfrak{a}_{P_1}^{P_2})^*.$$

When P_1 is a minimal semi-standard parabolic subgroup, since \mathfrak{a}_{P_1} (resp. A_{P_1}) and $\mathfrak{a}_{P_1}^{P_2}$ are independent of the choice of P_1 , we write them as \mathfrak{a}_0 (resp. A_0) and $\mathfrak{a}_0^{P_2}$ respectively.

For a pair of semi-standard parabolic subgroups $P_1 \subseteq P_2$ of G, write $\Delta_{P_1}^{P_2}$ for the set of simple roots for the action of A_{P_1} on $N_{P_1}^{P_2} := N_{P_1} \cap M_{P_2}$. Notice that $\Delta_{P_1}^{P_2}$ is a basis of $(\mathfrak{a}_{P_1}^{P_2})^*$. Let

$$(\widehat{\Delta}_{P_1}^{P_2})^{\vee} := \{ \varpi_{\alpha}^{\vee} : \alpha \in \Delta_{P_1}^{P_2} \}$$

be the basis of $\mathfrak{a}_{P_1}^{P_2}$ dual to $\Delta_{P_1}^{P_2}$. If B is a minimal semi-standard parabolic subgroup contained in P_1 , one has the coroot β^{\vee} associated to any $\beta \in \Delta_B^{P_2}$. For every $\alpha \in \Delta_{P_1}^{P_2}$, let α^{\vee} be the projection of β^{\vee} to $\mathfrak{a}_{P_1}^{P_2}$, where $\beta \in \Delta_B^{P_2}$ whose restriction to $\mathfrak{a}_{P_1}^{P_2}$ is α . Such α^{\vee} is independent of the choice of B. Define

$$(\Delta_{P_1}^{P_2})^{\vee} := \{\alpha^{\vee} : \alpha \in \Delta_{P_1}^{P_2}\}$$

which is a basis of $\mathfrak{a}_{P_1}^{P_2}$. Denote by

$$\widehat{\Delta}_{P_1}^{P_2} := \{ \varpi_\alpha : \alpha \in \Delta_{P_1}^{P_2} \}$$

the basis of $(\mathfrak{a}_{P_1}^{P_2})^*$ dual to $(\Delta_{P_1}^{P_2})^{\vee}$.

For a semi-standard parabolic subgroup P of G, set

$$\mathfrak{a}_P^+ := \{ T \in \mathfrak{a}_P : \alpha(T) > 0, \alpha \in \Delta_P^G \}.$$

For $P_1 \subseteq P_2$ as above, define $\tau_{P_1}^{P_2}$ and $\hat{\tau}_{P_1}^{P_2}$ as the characteristic functions of

$$\{T \in \mathfrak{a}_0 : \alpha(T) > 0, \alpha \in \Delta_{P_1}^{P_2}\}$$

and

$$\{T \in \mathfrak{a}_0 : \varpi(T) > 0, \varpi \in \widehat{\Delta}_{P_1}^{P_2}\}$$

respectively.

2.2. The functions H_P and F^P . Let \mathbb{A} be the ring of adèles of F and $|\cdot|_{\mathbb{A}}$ the product of normalised local absolute values on the group of idèles \mathbb{A}^* . Fix a maximal compact subgroup K of $G(\mathbb{A})$ that is admissible relative to M_0 in the sense of [5, p. 9]. In this paper, we choose the standard maximal compact subgroup for inner forms of GL_n (see [54, p. 191 and 199] for example). More concretely, suppose that $G(F) = GL_n(D)$, where D is a central division algebra over F. For every place v of F, fix an isomorphism $D \otimes_F F_v \simeq \mathfrak{gl}_{r_v}(D_v)$, where D_v is a central division algebra over F_v . Under this isomorphism, the completion at v of G(F) is $G_v \simeq GL_{n_v}(D_v)$, where $n_v = nr_v$. For v a finite place of F, let $K_v \simeq GL_{n_v}(\mathcal{O}_{D_v})$, where \mathcal{O}_{D_v} is the ring of integers of D_v ; for v an infinite place of F, we choose K_v to be the orthogonal group, unitary group and compact symplectic group (see [25, Chapter 1.2.8] for example) for $G_v \simeq GL_{n_v}(\mathbb{R})$, $GL_{n_v}(\mathbb{C})$ and $GL_{n_v}(\mathbb{H})$ respectively; let $K := \prod_v K_v$. Suppose that P is a semi-standard parabolic subgroup of G. If $m \in M_P(\mathbb{A})$, define $H_P(m) \in \mathfrak{a}_P$ by

$$\langle H_P(m), \chi \rangle = \log(|\chi(m)|_{\mathbb{A}}), \chi \in X(M_P)_F.$$

Write $M_P(\mathbb{A})^1$ for the kernel of H_P and A_P^{∞} for the neutral component for the topology of \mathbb{R} -manifolds of the group of \mathbb{R} -points of the maximal \mathbb{Q} -split torus in $\operatorname{Res}_{F/\mathbb{Q}}A_P$. Then any element $x \in G(\mathbb{A})$ can be written as x = nmak, where $n \in N_P(\mathbb{A})$, $m \in M_P(\mathbb{A})^1$, $a \in A_P^{\infty}$ and $k \in K$. We can define a continuous map $H_P : G(\mathbb{A}) \to \mathfrak{a}_P$ by setting $H_P(x) := H_P(a)$ with respect to this decomposition. Notice that H_P induces an isomorphism from A_P^{∞} to \mathfrak{a}_P . If $P \subseteq Q$ are a pair of semi-standard parabolic subgroups, write

$$A_P^{Q,\infty} := A_P^{\infty} \cap M_Q(\mathbb{A})^1.$$

Then H_P also induces an isomorphism from $A_P^{Q,\infty}$ to \mathfrak{a}_P^Q .

Denote by Ω^G the Weyl group of (G, A_0) . In the cases to be considered in this paper, for every $s \in \Omega^G$, we can always choose one representative $\omega_s \in G(F) \cap K$ such that ω_s normalises A_0 . In fact, we are dealing with the case of $G = GL_n$ or its inner forms, thus we can choose Ω^G to be the group of permutation matrices. For an F-subgroup H of G and $s \in \Omega^G$, we usually write $sH := \omega_s H \omega_s^{-1}$. Let P_1 and P_2 be a pair of semi-standard parabolic subgroups of G. Denote by $\Omega^G(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_2})$ the set (perhaps empty) of distinct isomorphisms from \mathfrak{a}_{P_1} to \mathfrak{a}_{P_2} obtained by restriction of elements in Ω^G . Denote by $\Omega^G(\mathfrak{a}_{P_1}; P_2)$ the subset (perhaps empty) of double classes in $\Omega^{M_{P_2}} \setminus \Omega^G / \Omega^{M_{P_1}}$ of elements $s \in \Omega^G$ such that $s(\mathfrak{a}_{P_1}) \supseteq \mathfrak{a}_{P_2}$. Suppose additionally that P_1 and P_2 contain a common minimal semi-standard parabolic subgroup P_0 of G. We can talk about positive roots with respect to P_0 . By [39, Lemme 1.3.6], all $s \in \Omega^G(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_2})$ admits a unique representative (still denoted by s) in Ω^G such that $s^{-1}\alpha > 0$ for all $\alpha \in \Delta_{P_0}^{P_2}$. By [39, Lemme 1.3.7], all $s \in \Omega^G(\mathfrak{a}_{P_1}; P_2)$ admits a unique representative (still denoted by s) in Ω^G such that $s^{-1}\alpha > 0$ for all $\alpha \in \Delta_{P_0}^{P_2}$.

From the reduction theory (see [3, p. 941]), we know that there exists a real number $t_0 < 0$ and a compact subset $\omega_B \subseteq N_B(\mathbb{A})M_0(\mathbb{A})^1$ for each minimal semi-standard parabolic subgroup B of G such that for any semi-standard parabolic subgroup P of G containing B, we have

$$G(\mathbb{A}) = P(F)\mathfrak{S}_B^P(\omega_B, t_0)$$

Here the Siegel set $\mathfrak{S}_B^P(\omega_B, t_0)$ is defined by

$$\mathfrak{S}_B^P(\omega_B, t_0) := \omega_B A_B^\infty(P, t_0) K,$$

where

$$A_B^{\infty}(P, t_0) := \{ a \in A_B^{\infty} : \alpha(H_B(a)) > t_0, \alpha \in \Delta_B^P \}$$

We shall fix such t_0 and ω_B . Additionally, we are authorised to assume that $\omega_{sB} = \omega_s \omega_B \omega_s^{-1}$ for $s \in \Omega^G$. Moreover, we require that $(M_P(\mathbb{A}) \cap \omega_B, M_P(\mathbb{A}) \cap K, B \cap M_P, t_0)$ will play the role of (ω_B, K, B, t_0) for any semi-standard parabolic subgroup P of G containing B.

Let $B \subseteq P$ and t_0 be as above. For $T \in \mathfrak{a}_0$, define the truncated Siegel set

$$\mathfrak{S}_B^P(\omega_B, t_0, T) := \omega_B A_B^\infty(P, t_0, T) K_S$$

where

$$A_B^{\infty}(P, t_0, T) := \{ a \in A_B^{\infty}(P, t_0) : \varpi(H_B(a) - T) \leq 0, \varpi \in \widetilde{\Delta}_B^P \}$$

Denote by $F_B^P(\cdot, T)$ the characteristic function of the projection of $\mathfrak{S}_B^P(\omega_B, t_0, T)$ to $P(F) \setminus G(\mathbb{A})$.

2.3. Bruhat-Schwartz functions and Haar measures. Write \mathfrak{g} for the Lie algebra of G. For an F-linear subspace \mathfrak{s} of \mathfrak{g} , denote by $\mathcal{S}(\mathfrak{s}(\mathbb{A}))$ the Bruhat-Schwartz space of $\mathfrak{s}(\mathbb{A})$, namely the \mathbb{C} -linear space of functions on $\mathfrak{s}(\mathbb{A})$ generated by $f_{\infty} \otimes \chi^{\infty}$, where f_{∞} is a Schwartz function on $\mathfrak{s}(F \otimes_{\mathbb{Q}} \mathbb{R})$ and χ^{∞} is the characteristic function of an open compact subgroup of $\mathfrak{s}(\mathbb{A}^{\infty})$, where we denote by \mathbb{A}^{∞} the ring of finite adèles of F.

Let P be a semi-standard parabolic subgroup of G. For every connected subgroup V of N_P (resp. every subspace \mathfrak{h} of \mathfrak{g}), choose the unique Haar measure on $V(\mathbb{A})$ (resp. on $\mathfrak{h}(\mathbb{A})$) such that $\operatorname{vol}(V(F)\setminus V(\mathbb{A})) = 1$ (resp. $\operatorname{vol}(\mathfrak{h}(F)\setminus \mathfrak{h}(\mathbb{A})) = 1$). We also take the Haar measure on K such that $\operatorname{vol}(K) = 1$.

Fix a Euclidean norm $\|\cdot\|$ on \mathfrak{a}_0 invariant by the group Ω^G and Haar measures on all subspaces of \mathfrak{a}_0 compatible with this norm. If $P \subseteq Q$ are a pair of semi-standard parabolic subgroups, we obtain the Haar measures on A_P^{∞} and $A_P^{Q,\infty}$ via the isomorphism H_P .

Denote by $\rho_P \in (\mathfrak{a}_P^G)^*$ the half of the sum of weights (with multiplicities) for the action of A_P on \mathfrak{n}_P . We choose compatible Haar measures on $G(\mathbb{A})$ and its subgroups by requiring that for any $f \in L^1(G(\mathbb{A}))$,

$$\begin{split} \int_{G(\mathbb{A})} f(x) dx &= \int_{N_P(\mathbb{A})} \int_{M_P(\mathbb{A})} \int_K f(nmk) e^{-2\rho_P(H_P(m))} dn dm dk \\ &= \int_{N_P(\mathbb{A})} \int_{M_P(\mathbb{A})^1} \int_{A_P^\infty} \int_K f(nmak) e^{-2\rho_P(H_P(a))} dn dm dadk \end{split}$$

3. The symmetric pair

Let F be a number field and D a central division algebra over F. Let d be the degree of D, i.e., $\dim_F(D) = d^2$. Denote by $GL_{n,D}$ the reductive group over F whose F-points are $GL_n(D)$. For $x \in GL_n(D)$, we write $\operatorname{Nrd}(x)$ for its reduced norm, $\operatorname{Trd}(x)$ for its reduced trace and Prd_x for its reduced characteristic polynomial. For $x \in GL_p(D) \times GL_q(D)$, denote by x_1 (resp. x_2) its projection to the first (resp. second) component. Until further notice, we shall work in a more general setting than that of Guo-Jacquet for later use, i.e., we shall study the case of $(GL_{p+q,D}, GL_{p,D} \times GL_{q,D})$ and add an additional term $|\operatorname{Nrd}(x_1)|_{\mathbb{A}}^{\mathbb{A}}$ in the integral of the modified kernel.

3.1. Groups and linear spaces. Let $G := GL_{p+q,D}$ and $H := GL_{p,D} \times GL_{q,D}$ its subgroup by diagonal embedding. Define an involution θ on G by $\theta(g) = \epsilon g \epsilon^{-1}$, where $\epsilon = \begin{pmatrix} 1_p & 0 \\ 0 & -1_q \end{pmatrix}$. Thus $H = G^{\theta}$, where G^{θ} denotes the θ -invariant subgroup of G.

Define an anti-involution ι on G by $\iota(g) = \theta(g^{-1})$. Denote by G^{ι} the ι -invariant subvariety of G. There is a symmetrization map

$$s: G \to G^{\iota}, s(g) := g\iota(g),$$

by which one can regard the symmetric space S := G/H as a subvariety of G^{ι} . We see that $H \times H$ acts on G by left and right translation and that H acts on G^{ι} by conjugation.

Let $\mathfrak{g} := Lie(G)$ and $\mathfrak{h} := Lie(H)$. Denote by $d\theta$ the differential of θ . Thus $\mathfrak{h} = \{X \in \mathfrak{g} : (d\theta)(X) = X\}$. Let \mathfrak{s} be the tangent space of S at the neutral element. We shall always view \mathfrak{s} as a subspace of \mathfrak{g} . Then $\mathfrak{s} = \{X \in \mathfrak{g} : (d\theta)(X) = -X\}$ and $\mathfrak{s}(F) = \left\{ \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} : A \in \operatorname{Mat}_{p \times q}(D), B \in \operatorname{Mat}_{q \times p}(D) \right\} \simeq \operatorname{Mat}_{p \times q}(D) \oplus \operatorname{Mat}_{q \times p}(D)$. There is an H(F)-action on $\mathfrak{s}(F)$ by conjugation, i.e., $(h_1, h_2) \cdot (A, B) = (h_1Ah_2^{-1}, h_2Ah_1^{-1})$.

3.2. Semi-simple elements. We say that an element $X \in \mathfrak{s}$ is semi-simple if the orbit $H \cdot X$ is Zariski closed in \mathfrak{s} . By a regular element $X \in \mathfrak{s}$, we mean that the stabiliser H_X has minimal dimension.

PROPOSITION 3.1. An element X of $\mathfrak{s}(F)$ is semi-simple if and only if it is H(F)-conjugate to an element of the form

$$X(A) := \left(\begin{array}{rrrrr} 0 & 0 & 1_m & 0\\ 0 & 0 & 0 & 0\\ A & 0 & 0 & 0\\ 0 & 0 & 0 & 0 \end{array}\right)$$

with $A \in GL_m(D)$ being semi-simple in the usual sense. More precisely, the set of H(F)-conjugacy classes of semi-simple elements of $\mathfrak{s}(F)$ is bijective to the set of pairs $(m, \{A\})$ where $0 \le m \le \min\{p, q\}$ is an integer and $\{A\}$ is a semi-simple conjugacy class in $GL_m(D)$. Moreover, X(A) is regular semisimple if and only if $m = \min\{p, q\}$ and A is regular semi-simple in $GL_{\min\{p, q\}}(D)$ in the usual sense.

PROOF. The case D = F is [31, Proposition 2.1 and Lemma 2.1] while the case p = q is [57, Proposition 5.2]. This proposition is nothing but a slightly more general one combining both cases, whose proofs are similar and still work here.

PROPOSITION 3.2. If $p \leq q$, an element $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}$ is regular semi-simple if and only if Prd_{AB} is separable and $\operatorname{Prd}_{AB}(0) \neq 0$. If p > q, an element $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}$ is regular semi-simple if and only if Prd_{BA} is separable and $\operatorname{Prd}_{BA}(0) \neq 0$.

PROOF. We only consider the case $p \leq q$ since the other case can be deduced by symmetry. We may study the proposition over an algebraic closure \overline{F} of F.

Suppose that Prd_{AB} is separable and $\operatorname{Prd}_{AB}(0) \neq 0$. Let $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} = X_s + X_n$ be the Jordan decomposition in \mathfrak{g} , where X_s is semi-simple, X_n is nilpotent and $X_s X_n = X_n X_s$. By the uniqueness of the Jordan decomposition, we see that $X_s, X_n \in \mathfrak{s}$. From Proposition 3.1, up to conjugation by H, we

the Jordan decomposition, we see that $X_s, X_n \in \mathfrak{s}$. From Proposition 5.1, up to conjugation by H, we may suppose that $X_s = \begin{pmatrix} 0 & 1_{dp} & 0 \\ C & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$, where $C \in GL_{dp}(\overline{F})$ is semi-simple. Since $X_sX_n = X_nX_s$, simple computation (cf. [31, Lemma 2.1]) shows that $X_n = \begin{pmatrix} 0 & D & 0 \\ DC & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$, where $D \in \mathfrak{gl}_{dp}(\overline{F})$ and DC = CD. But X_n is nilpotent, which forces D to be zero because C is invertible. Then $X_n = 0$ and $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} = X_s$. We deduce that Prd_C is separable and that $\operatorname{Prd}_C(0) \neq 0$. By linear algebra, C is regular semi-simple in $GL_{dp}(\overline{F})$. Hence, $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix}$ is regular semi-simple by Proposition 3.1.

The other direction is a direct consequence of Proposition 3.1.

3.3. Invariants. Denote by \mathfrak{c} the affine space $\mathbf{A}^{d\min\{p,q\}}$. Define a morphism $\pi:\mathfrak{s}\to\mathfrak{c}$ by mapping $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}$ to the coefficients of the reduced characteristic polynomial of AB. It is constant on H-orbits. Denote by \mathfrak{c}_{rs} the subset of $(c_i)_{0 \leq i \leq d \min\{p,q\}-1} \in \mathfrak{c}$ such that the polynomial

$$P(\lambda) := \lambda^{d\min\{p,q\}} + \sum_{i=0}^{d\min\{p,q\}-1} c_i \lambda^i$$

is separable and $c_0 \neq 0$. It is a principal Zariski open subset of \mathfrak{c} . Denote by \mathfrak{c}^{\times} the subset of $(c_i)_{0 \leq i \leq d \min\{p,q\}-1} \in \mathfrak{c}$ such that $c_0 \neq 0$. Then $\mathfrak{c}_{rs} \subseteq \mathfrak{c}^{\times}$.

PROPOSITION 3.3. The pair (\mathfrak{c}, π) defines a categorical quotient of \mathfrak{s} by H over F.

PROOF. It suffices to consider the case $p \le q$ since the case p > q can be obtained by symmetry.

We first extend the base field to an algebraic closure \overline{F} of F. Then $H_{\overline{F}} \simeq GL_{dp,\overline{F}} \times GL_{dq,\overline{F}}$ and $\mathfrak{s}_{\overline{F}} \simeq \operatorname{Mat}_{dp \times dq,\overline{F}} \oplus \operatorname{Mat}_{dq \times dp,\overline{F}}$. For $(c_i)_{0 \le i \le dp-1} \in \mathfrak{c}_{\overline{F}}$, denote by $A((c_i)_{0 \le i \le dp-1}) \in GL_{pd}$ its companion matrix

$$A((c_i)_{0 \le i \le dp-1}) := \begin{pmatrix} 0 & 0 & \cdots & 0 & -c_0 \\ 1 & 0 & \cdots & 0 & -c_1 \\ 0 & 1 & \ddots & \vdots & -c_2 \\ \vdots & \ddots & \ddots & 0 & \vdots \\ 0 & \cdots & 0 & 1 & -c_{dp-1} \end{pmatrix}.$$

Define a morphism $\mathfrak{c}_{\overline{F}} \to \mathfrak{s}_{\overline{F}}$ by mapping $(c_i)_{0 \le i \le dp-1}$ to

$$\left(\begin{array}{ccc} 0 & 1_{dp} & 0\\ A((c_i)_{0 \le i \le dp-1}) & 0 & 0\\ 0 & 0 & 0 \end{array}\right).$$

This is a section of π , so π is surjective. By Propositions 3.2 and 3.1, the fibre of any point in the non-empty open subset $\mathfrak{c}_{\overline{F},rs} \subseteq \mathfrak{c}_{\overline{F}}$ contains exactly one closed orbit. We may use Igusa's criterion (see [43, Theorem 4.13] and Remark 3.4 below) to show that the pair $(\mathfrak{c}_{\overline{F}}, \pi)$ defines a categorical quotient of $\mathfrak{s}_{\overline{F}}$ by $H_{\overline{F}}$.

The morphism $\pi : \mathfrak{s} \to \mathfrak{c}$ defined over F factors through the categorical quotient $\operatorname{Spec}(F[\mathfrak{s}]^H)$ of \mathfrak{s} by H over F. This induces a dual morphism $F[\mathfrak{c}] \to F[\mathfrak{s}]^H$ of F-algebras. We have shown that after the base change to \overline{F} , it is an isomorphism of \overline{F} -algebras. By Galois descent, we deduce that the morphism $F[\mathfrak{c}] \to F[\mathfrak{s}]^H$ is an isomorphism of F-algebras, i.e., the pair (\mathfrak{c}, π) defines a categorical quotient of \mathfrak{s} by H over F. \square

REMARK 3.4. We notice that $\mathfrak{c}_{\overline{F}}$ can be of dimension 1 (when D = F and $\min\{p, q\} = 1$) in the proof of Proposition 3.3 above, so the first condition in [43, Theorem 4.13] may not be satisfied. However, as is evident from the proof of Igusa's criterion, this condition can be replaced with the surjectivity of π .

The categorical quotient (\mathfrak{c}, π) defines a relation of equivalence on $\mathfrak{s}(F)$, where two elements are in the same class if and only if they have the same image under π . We denote by \mathcal{O} the set of equivalent classes for this relation. By Proposition 3.1, two semi-simple elements of $\mathfrak{s}(F)$ belong to the same class of \mathcal{O} if and only if they are conjugate by H(F). Denote by \mathcal{O}_{rs} the subset of \mathcal{O} with images in \mathfrak{c}_{rs} . By Proposition 3.2, each class in \mathcal{O}_{rs} is a regular semi-simple H(F)-orbit in $\mathfrak{s}(F)$. Denote by \mathcal{O}^{\times} the subset of \mathcal{O} with images in \mathfrak{c}^{\times} . Then $\mathcal{O}_{rs} \subseteq \mathcal{O}^{\times}$.

3.4. Relatively standard parabolic subgroups. Fix P_0 a minimal parabolic subgroup of H defined over F and M_0 a Levi factor of \tilde{P}_0 defined over F. Then M_0 is also a Levi subgroup of G defined over F. For a semi-standard parabolic subgroup P of G (namely $M_0 \subseteq P$), we say that P is "relatively standard" if $\tilde{P}_0 \subseteq P$, i.e., $P \cap H$ is a standard parabolic subgroup of H (namely $\tilde{P}_0 \subseteq P \cap H$). We shall suppose that $\omega_{\tilde{P}_0} \subseteq \omega_B$ for all relatively standard minimal parabolic subgroup B of G. Denote by K the standard maximal compact subgroup of $G(\mathbb{A})$ and by $K_H := H(\mathbb{A}) \cap K$ the maximal compact subgroup of $H(\mathbb{A})$. Up to conjugation by G(F), we may assume that M_0 is the subgroup of diagonal matrices in G and that \tilde{P}_0 is the product of groups of upper triangular matrices.

We can describe the embedding $H \hookrightarrow G$ via *D*-bimodules. Let $V := \langle e_1, \dots, e_p \rangle_D$ (resp. $W := \langle f_1, \dots, f_q \rangle_D$) be the free *D*-bimodule generated by the basis $\{e_1, \dots, e_p\}$ (resp. $\{f_1, \dots, f_q\}$). Set GL(V) to be the group of *F*-linear automorphisms on V, which acts on V on the left. Denote by $GL(V)_D$ the subgroup of GL(V) which respects the right *D*-module structure on V. Put $G := GL(V \oplus W)_D$ and $H := GL(V)_D \times GL(W)_D$. Then M_0 is the stabiliser in G (or in H) of the *D*-lines $\langle e_i \rangle_D, 1 \le i \le p$ and $\langle f_i \rangle_D, 1 \le i \le q$. Suppose that \widetilde{P}_0 is the direct product of the stabiliser in $GL(V)_D$ of the flag

$$0 \subsetneq \langle e_1 \rangle_D \subsetneq \langle e_1, e_2 \rangle_D \subsetneq \cdots \subsetneq \langle e_1, \cdots, e_p \rangle_D =: V$$

and the stabiliser in $GL(W)_D$ of the flag

Λ

$$0 \subsetneq \langle f_1 \rangle_D \subsetneq \langle f_1, f_2 \rangle_D \subsetneq \cdots \subsetneq \langle f_1, \cdots, f_q \rangle_D =: W$$

A relative standard parabolic subgroup P of G can be interpretated as the stabiliser in G of the flag

$$0 \subsetneq \langle e_1, \cdots, e_{p_1}, f_1, \cdots, f_{q_1} \rangle_D \subsetneq \langle e_1, \cdots, e_{p_1}, f_1, \cdots, f_{q_1}, e_{p_1+1}, \cdots, e_{p_1+p_2}, f_{q_1+1}, \cdots, f_{q_1+q_2} \rangle_D$$
$$\subseteq \cdots \subsetneq \langle e_1, \cdots, e_{p_1}, f_1, \cdots, f_{q_1}, \cdots, e_{p-p_l+1}, \cdots, e_p, f_{q-q_l+1}, \cdots, f_q \rangle_D =: V \oplus W,$$

where $\sum_{i=1}^{l} p_i = p$, $\sum_{i=1}^{l} q_i = q$ and we allow p_i or q_i to be zero. In particular, we have

$$M_P \simeq GL_{p_1+q_1,D} \times \cdots \times GL_{p_l+q_l,D}$$

and

$$I_{P_H} \simeq GL_{p_1,D} \times \cdots \times GL_{p_l,D} \times GL_{q_1,D} \times \cdots \times GL_{q_l,D}$$

PROPOSITION 3.5. Let P be a relative standard parabolic subgroup of G. For all $X \in (\mathfrak{m}_P \cap \mathfrak{s})(F)$ and $U \in (\mathfrak{n}_P \cap \mathfrak{s})(F)$, we have

$$\pi(X) = \pi(X + U).$$

PROOF. It is a consequence of [40, Lemma 2.1]. We can also give a direct proof as follows. Let \overline{F} be an algebraic closure of F. For $A \in \operatorname{Mat}_{dp \times dq}(\overline{F})$ and $B \in \operatorname{Mat}_{dq \times dp}(\overline{F})$, we see that

$$\det\left(\lambda I_{d(p+q)} - \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix}\right) = \lambda^{d(q-p)} \det(\lambda^2 I_{dp} - AB).$$

Then for any $X \in \mathfrak{s}(F)$, $\pi(X)$ is determined by the coefficients of the reduced characteristic polynomial of X regarded as an element of $\mathfrak{g}(F)$. The proposition follows from the easy fact: for $X \in \mathfrak{m}_P(F)$ and $U \in \mathfrak{n}_P(F)$, the reduced characteristic polynomial of X + U is equal to that of X. \Box

COROLLARY 3.6. Let P be a relative standard parabolic subgroup of G and $\mathfrak{o} \in \mathcal{O}$. For all subsets $S_1 \subseteq (\mathfrak{m}_P \cap \mathfrak{s})(F)$ and $S_2 \subseteq (\mathfrak{n}_P \cap \mathfrak{s})(F)$, we have $\mathfrak{o} \cap (S_1 \oplus S_2) = (\mathfrak{o} \cap S_1) \oplus S_2$.

3.5. Fourier transform. Fix a nontrivial unitary character Ψ of \mathbb{A}/F . Let $\langle \cdot, \cdot \rangle$ be the nondegenerate $H(\mathbb{A})$ -invariant bilinear form on $\mathfrak{s}(\mathbb{A})$ defined by

(3.5.1)
$$\forall X_1, X_2 \in \mathfrak{s}(\mathbb{A}), \langle X_1, X_2 \rangle := \operatorname{Trd}(X_1 X_2).$$

For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, its Fourier transform $\hat{f} \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ is defined by

(3.5.2)
$$\forall \widehat{X} \in \mathfrak{s}(\mathbb{A}), \widehat{f}(\widehat{X}) := \int_{\mathfrak{s}(\mathbb{A})} f(X) \Psi(\langle X, \widehat{X} \rangle) dX.$$

4. Integrability of the modified kernel

Fix a minimal semi-standard parabolic subgroup P_0 of G. For any semi-standard parabolic subgroup P of G and $T \in \mathfrak{a}_0$, denote by T_P the projection of sT in \mathfrak{a}_P , where s is any element in Ω^G such that $sP_0 \subseteq P$. Notice that this definition is independent of the choice of s. For a semi-standard parabolic subgroup P of G, $x \in H(\mathbb{A})$ and $T \in \mathfrak{a}_0$, define

$$F^{P}(x,T) := F^{P}_{sP_{0}}(x,T_{sP_{0}}),$$

where s is any element in Ω^G such that $sP_0 \subseteq P$. Note that this definition is independent of the choice of s since we choose all $\omega_s \in G(F) \cap K$. In fact, for any minimal semi-standard parabolic subgroup $B \subseteq P$, $F_{sB}^{sP}(x,T) = F_B^P(\omega_s^{-1}x,s^{-1}T)$ and $F_B^P(\cdot,T)$ is left $M_P(F)$ -invariant. Let $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, P be a relatively standard parabolic subgroup of G and $\mathfrak{o} \in \mathcal{O}$. Write $P_H := P \cap H$.

For $x \in M_{P_H}(F)N_{P_H}(\mathbb{A}) \setminus H(\mathbb{A})$, define

$$k_{f,P,\mathfrak{o}}(x) := \sum_{X \in \mathfrak{m}_P(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(X+U)x) dU,$$

and for $x \in H(F) \setminus H(\mathbb{A})$, define

$$(4.0.1) k_{f,\mathfrak{o}}^T(x) := \sum_{\{P:\tilde{P}_0 \subseteq P\}} (-1)^{\dim(A_P/A_G)} \sum_{\delta \in P_H(F) \setminus H(F)} \hat{\tau}_P^G(H_P(\delta x) - T_P) \cdot k_{f,P,\mathfrak{o}}(\delta x).$$

From [3, Lemma 5.1], we know that the sum over $\delta \in P_H(F) \setminus H(F)$ is finite.

4.1. Reduction theory. There is a $T_+ \in \mathfrak{a}_{P_0}^+$ such that [3, Lemma 6.4] holds for $T \in T_+ + \mathfrak{a}_{P_0}^+$. We shall fix such a T_+ and say that such T is sufficiently regular.

LEMMA 4.1. For all relatively standard parabolic subgroup Q of G, sufficiently regular T and $x \in$ $H(\mathbb{A})$, we have

$$\sum_{\{P:\tilde{P}_0\subseteq P\subseteq Q\}}\sum_{\delta\in P_H(F)\setminus Q_H(F)}F^P(\delta x,T)\tau_P^Q(H_P(\delta x)-T_P)=1.$$

This is an analogue of [61, Proposition 2.3] whose proof relies on [28, (2.5) in p. 674] (cf. Lemma 4.8 below). It is essentially a restricted form to H from [3, Lemma 6.4] for G. We can give a proof close to the steps in an early version of [61], which reflects that a main complexity of the truncation here arises from the fact that none of the Siegel sets of H is contained in any Siegel set of G, as mentioned in [28]. However, we shall adopt alternatively the point of view in [17] to give a more conceptual proof here, which might be useful in other relative trace formulae as well.

First we introduce a variant (see $[17, \S1.5]$) of some concepts and results in $[17, \S2]$ without reproducing proofs. We say that a semi-standard parabolic subgroup Q of G is standard if $P_0 \subseteq Q$. For $P \subseteq Q$ a pair of standard parabolic subgroups of G, denote by ρ_P^Q the half of the sum of weights (with multiplicities) for the action of A_P on $\mathfrak{n}_P \cap \mathfrak{m}_Q$. We denote by $\overline{\mathfrak{a}_{P_0}^+}$ the closure of $\mathfrak{a}_{P_0}^+$ in \mathfrak{a}_0 .

DEFINITION 4.2. For $g \in G(\mathbb{A})$, Q a standard parabolic subgroup of G and $T \in \overline{\mathfrak{a}_{P_0}^+}$, we define the degree of T-instability of g with respect to Q by the following formula

$$\deg_{i,T}^{Q}(g) := \max_{(P,\delta)} \langle \rho_{P}^{Q}, H_{P}(\delta g) - T \rangle$$

where (P, δ) runs over the pairs of a standard parabolic subgroup $P \subseteq Q$ and an element $\delta \in P(F) \setminus Q(F)$.

From [3, Lemma 5.1], we know that the supremum of $\langle \rho_P^Q, H_P(\delta g) - T \rangle$ in the definition is finite and attainable.

LEMMA 4.3 (cf. [17, Lemme 2.2.1]). Let $g \in G(\mathbb{A})$, Q be a standard parabolic subgroup of G and $T \in \overline{\mathfrak{a}_{P_0}^+}$. The following two conditions are equivalent:

- (1) $\deg_{i,T}^{Q}(g) \le 0;$
- (2) for all parabolic subgroup $P \subseteq Q$, all $\delta \in P(F) \setminus Q(F)$ and all $\varpi \in \widehat{\Delta}_P^Q$, we have $\langle \varpi, H_P(\delta g) T \rangle \leq C$

DEFINITION 4.4. Let $g \in G(\mathbb{A})$ and $T \in \overline{\mathfrak{a}_{P_0}^+}$. We say that a pair (P, δ) of a standard parabolic subgroup $P \subseteq G$ and an element $\delta \in P(F) \setminus G(F)$ is T-canonical for g if it satisfies the following two conditions:

(1) $\langle \rho_P^G, H_P(\delta g) - T \rangle = \deg_{i,T}^G(g);$

(2) for any parabolic subgroup $Q \supseteq P$ such that $\langle \rho_Q^G, H_Q(\delta g) - T \rangle = \deg_{i,T}^G(g)$, we have Q = P.

LEMMA 4.5 (cf. [17, Lemme 2.3.2]). Let $g \in G(\mathbb{A})$ and $T \in \overline{\mathfrak{a}_{P_0}^+}$. Then (P, δ) is a T-canonical pair for g if and only if it satisfies the following two conditions:

(1) $\deg_{i,T}^{P}(\delta g) \leq 0;$

(2) for any $\alpha \in \Delta_P^G$, we have $\langle \alpha, H_P(\delta g) - T \rangle > 0$.

PROPOSITION 4.6 (cf. [17, Proposition 2.4.1]). Let $g \in G(\mathbb{A})$ and $T \in \overline{\mathfrak{a}_{P_0}^+}$. Then there exists a unique *T*-canonical pair for *g*.

Let $T \in \mathfrak{a}_0$ and Q be a standard parabolic subgroup of G. Define $\widetilde{F}^Q(\cdot, T)$ as the characteristic function of $g \in G(\mathbb{A})$ such that $\deg_{i,T}^Q(g) \leq 0$.

PROPOSITION 4.7 (cf. [17, Proposition 2.5.1]). For $g \in G(\mathbb{A})$, Q a standard parabolic subgroup of G and $T \in \overline{\mathfrak{a}_{P_0}^+}$, we have

(1)

$$\sum_{\{P:P_0\subseteq P\subseteq Q\}}\sum_{\delta\in P(F)\setminus Q(F)}\widetilde{F}^P(\delta g,T)\tau_P^Q(H_P(\delta g)-T_P)=1;$$

(2)

$$\widetilde{F}^Q(g,T) = \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in P(F) \setminus Q(F)} \widehat{\tau}^Q_P(H_P(\delta g) - T_P)$$

Since we have similar formulae for $F^Q(\cdot, T)$ for sufficiently regular T (see [3, Lemma 6.4]), we know that $\widetilde{F}^Q(\cdot, T) = F_{P_0}^Q(\cdot, T)$ for such T. Now we can return to the proof of Lemma 4.1.

PROOF OF LEMMA 4.1. It is noticeable that the identity is reduced to its analogues for semistandard Levi factors of Q, which is a product of $GL_{p_i+q_i,D}$ whose intersection with H is $GL_{p_i,D} \times GL_{q_i,D}$. By induction on the rank of G, it suffices to prove the identity for Q = G.

For a standard parabolic subgroup P of G, fix a set of representatives $\Omega_{P,G}$ in $\{s \in \Omega^G | \tilde{P}_0 \subseteq s^{-1}P\}$ for the relation $s_1 \sim s_2$ if and only if $s_2 s_1^{-1} \in \Omega^{M_P}$. We can rewrite the equality in the lemma as

$$\sum_{\{P:P_0\subseteq P\}}\sum_{s\in\Omega_{P,G}}\sum_{\delta\in(s^{-1}P)_H(F)\backslash H(F)}F_{P_0}^P(\omega_s\delta x,T_{P_0})\tau_P^G(H_P(\omega_s\delta x)-T_P)=1.$$

In fact, this follows from

$$F^{s^{-1}P}(\delta x, T) = F^{s^{-1}P}_{s^{-1}P_0}(\delta x, T_{s^{-1}P_0}) = F^P_{P_0}(\omega_s \delta x, T_{P_0})$$

and

$$\tau^{G}_{s^{-1}P}(H_{s^{-1}P}(\delta x) - T_{s^{-1}P}) = \tau^{G}_{P}(H_{P}(\omega_{s}\delta x) - T_{P}).$$

Combining the double sums over s and δ , we claim that the equality above is equivalent to

$$\sum_{\{P:P_0\subseteq P\}}\sum_{\delta\in P(F)\setminus P(F)\Omega^G H(F)}F^P_{P_0}(\delta x, T_{P_0})\tau^G_P(H_P(\delta x) - T_P) = 1$$

In fact, for any $s \in \Omega_{P,G}$, consider the map

$$(s^{-1}P)_H(F) \setminus H(F) \to P(F) \setminus P(F)\Omega^G H(F), \delta \mapsto \omega_s \delta$$

Firstly, it is well-defined: if $\delta_1 = \omega_s^{-1} p \omega_s \delta_2$ with $p \in P(F)$, then $\omega_s \delta_1 = p \omega_s \delta_2$. Secondly, it is injective: if $\omega_s \delta_1 = p \omega_s \delta_2$ with $p \in P(F)$, then $\delta_1 = \omega_s^{-1} p \omega_s \delta_2$ with $\omega_s^{-1} p \omega_s = \delta_1 \delta_2^{-1} \in (s^{-1}P)_H(F)$. Thirdly, for $s_1 \neq s_2$ in $\Omega_{P,G}$, we have $\omega_{s_1} \delta_1 \neq p \omega_{s_2} \delta_2$ with $p \in P(F)$: otherwise, $s_1^{-1}P = (\delta_2 \delta_1^{-1})^{-1}(s_2^{-1}P)(\delta_2 \delta_1^{-1})$ with $\delta_2 \delta_1^{-1} \in H(F)$, so $(s_1^{-1}P)_H = (\delta_2 \delta_1^{-1})^{-1}(s_2^{-1}P)_H(\delta_2 \delta_1^{-1})$, and then $\delta_2 \delta_1^{-1} \in (s_1^{-1}P)_H(F) = (s_2^{-1}P)_H(F)$ for both of $(s_1^{-1}P)_H$ and $(s_2^{-1}P)_H$ are standard parabolic subgroups of H, which implies $s_1^{-1}P = s_2^{-1}P$ contradicting $s_1 \neq s_2$. Fourthly, any $\tilde{s} \in \Omega^G$ appears in the image of the map for some $s \in \Omega_{P,G}$: since $(\tilde{s}^{-1}P)_H = (s_0^{-1}(\tilde{s}^{-1}P))_H = ((\tilde{s}s_0)^{-1}P)_H$, i.e., $\tilde{s}s_0 \in \Omega_{P,G}$. To sum up, we finish the argument of the claim.

It suffices to prove an analogue of the last equality by replacing $F_{P_0}^P$ with \widetilde{F}^P for $T \in \overline{\mathfrak{a}_{P_0}^+}$, as they are identical for sufficiently regular T. That is to say, for $x \in H(\mathbb{A}) = G^{\theta}(\mathbb{A})$, if (P, δ) is the unique

T-canonical pair for x, we need to prove that $\delta \in P(F) \setminus P(F) \Omega^G H(F)$. Recall that $\theta(g) = \epsilon g \epsilon^{-1}$ for $g \in G(\mathbb{A})$, where $\epsilon = \begin{pmatrix} 1_p & 0 \\ 0 & -1_q \end{pmatrix}$. Since $\epsilon \in M_0(F) \cap K$, from Lemma 4.5, we deduce that (P, δ) is the unique T-canonical pair for $g \in G(\mathbb{A})$ if and only if $(P, \theta(\delta))$ is the unique T-canonical pair for $\theta(g)$. In particular, if (P, δ) is the unique T-canonical pair for $x \in H(\mathbb{A})$, we have $\delta = \theta(\delta)$. Denote by δ_0 a representative of $\delta \in P(F) \setminus G(F)$. Then $\delta_0 \epsilon \delta_0^{-1} \in P(F)$.

Suppose that $\delta_0 \epsilon \delta_0^{-1} = mu$, where $m \in M_P(F)$ and $u \in N_P(F)$. Both of mu and m are semi-simple in G(F) (in the classical sense) for $(mu)^2 = m^2 = 1$. Applying [3, Lemma 2.1] to the characteristic function of the singleton $\{u\}$, one obtains that mu is $N_P(F)$ -conjugate to mu' for some $u' \in N_P(F)$ such that mu' = u'm. Since both of mu' and m are semi-simple in G(F), by the uniqueness of Jordan decomposition, we have u' = 1, i.e., $\delta_0 \epsilon \delta_0^{-1}$ is $N_P(F)$ -conjugate to m. By linear algebra, m is $M_P(F)$ conjugate to a diagonal matrix with entries $\{\pm 1\}$ with expected multiplicities p and q respectively. In sum, $\delta_0 \epsilon \delta_0^{-1}$ is P(F)-conjugate to $\omega_s \epsilon \omega_s^{-1}$ for some $s \in \Omega^G$. Suppose that $p_0 \in P(F)$ satisfies $\delta_0 \epsilon \delta_0^{-1} = p_0(\omega_s \epsilon \omega_s^{-1})p_0^{-1}$. Then $\omega_s^{-1}p_0^{-1}\delta_0 \in G^{\theta}(F) = H(F)$, i.e., $\delta = P(F)\delta_0 \in P(F) \setminus P(F)\Omega^G H(F)$.

LEMMA 4.8. Let P be a relatively standard parabolic subgroup of G. For any $a \in A^{\infty}_{\widetilde{P}_0}(P_H, t_0)$, there exists a relatively standard minimal parabolic subgroup $B \subseteq P$ such that $a \in A_B^{\infty}(P, t_0)$.

PROOF. This is an analogue of [28, (2.5) in p. 674]. By induction on dim (A_P) , it suffices to prove this assertion for P = G.

Let $a \in A_{\widetilde{P}_0}^{\infty}(H, t_0)$. Then $a = \operatorname{diag}(a_1, \dots, a_{p+q})$, where $\frac{a_i}{a_{i+1}} > e^{t_0}$ for $1 \le i \le p-1$ and $p+1 \le i \le p+q-1$. In the definition of Siegel sets, we suppose that $t_0 < 0$, so $0 < e^{t_0} < 1$. Note that $A_{\widetilde{P}_0}^{\infty}(G, t_0) = \{\operatorname{diag}(b_1, \dots, b_{p+q}) | \frac{b_i}{b_{i+1}} > e^{t_0}, \forall 1 \le i \le p+q-1\}$. Thus we need to show that there exists a permutation $s \in \Omega^G$ such that $s \cdot a = \operatorname{diag}(a_{s^{-1}(1)}, \dots, a_{s^{-1}(p+q)})$ satisfies the following two conditions:

- (1) s(i) < s(i+1) for $1 \le i \le p-1$ and $p+1 \le i \le p+q-1$; (2) $\frac{a_{s^{-1}(i)}}{a_{s^{-1}(i+1)}} > e^{t_0}$ for $1 \le i \le p+q-1$.

Firstly, we show that one can move a_{p+1} to its left hand side in (a_1, \dots, a_{p+q}) such that both the first p+1 elements and the last q-1 ones in the new sequence are in "good" order (which means that the quotient of any consecutive pairs is $> e^{t_0}$), while keeping the original relative orders among (a_1, \dots, a_p) and among $(a_{p+1}, \dots, a_{p+q})$. If $\frac{a_p}{a_{p+1}} > e^{t_0}$, we are already done (one can take s = 1). In general, write

$$i_1 := \max\left\{0, \max\left\{1 \le i \le p \left| \frac{a_i}{a_{p+1}} > e^{t_0}\right\}\right\}\right\}.$$

When $1 \leq i_1 \leq p-1$, since $e^{t_0} < 1$, $\frac{a_{i_1+1}}{a_{p+1}} \leq e^{t_0}$ implies $\frac{a_{p+1}}{a_{i_1+1}} \geq e^{-t_0} > 1$; there is an $s \in \Omega^G$ such that $s \cdot a = \operatorname{diag}(a_1, \dots, a_{i_1}, a_{p+1}, a_{i_1+1}, \dots, a_p, a_{p+2}, \dots, a_{p+q})$. When $i_1 = 0$, which implies $a_{p+1} > a_1$, there is an $s \in \Omega^G$ such that $s \cdot a = \operatorname{diag}(a_{p+1}, a_1, \dots, a_p, a_{p+2}, \dots, a_{p+q})$.

Secondly, we consider moving a_{p+2} as before. One should check that a_{p+2} will not exceed the new place of a_{p+1} , which results from the fact that $\frac{a_{p+1}}{a_{p+2}} > e^{t_0}$. Thus one can move a_{p+1} and a_{p+2} to their place of a_{p+1} , which results from the fact that $\frac{a_{p+1}}{a_{p+2}} > e^{t_0}$. left hand side in (a_1, \dots, a_{p+q}) such that both the first p+2 elements and the last q-2 ones in the new sequence are in "good" order, while still keeping the original relative orders among (a_1, \dots, a_p) and among $(a_{p+1}, \cdots, a_{p+q})$.

We can finish the argument of our claim by induction on q.

PROPOSITION 4.9. Let B be a minimal semi-standard parabolic subgroup of G. Let P be a parabolic subgroup of G containing B. Suppose that T is sufficiently regular. If $m \in \omega_B \cap M_P(\mathbb{A}), a \in A_B^{\infty}(P, t_0)$ and $k \in K \cap M_P(\mathbb{A})$ satisfy $F_B^P(mak, T_B) = 1$, then $a \in A_B^{\infty}(P, t_0, T_B)$.

PROOF. It results from Lemma 4.3, since $\widetilde{F}_B^P(\cdot,T) = F_B^P(\cdot,T)$ for sufficiently regular T. Here we write $\widetilde{F}_{B}^{P}(\cdot,T)$ for $\widetilde{F}^{P}(\cdot,T)$ when B plays the role of P_{0} .

For a relatively standard parabolic subgroup P of G, denote by $\mathcal{P}(\widetilde{P}_0, P)$ the set of relatively standard minimal parabolic subgroups of G contained in P. For $B \in \mathcal{P}(\tilde{P}_0, P)$, write

$$A_B^{G,\infty}(P,t_0) := A_B^{\infty}(P,t_0) \cap G(\mathbb{A})^1$$

and

$$\forall T \in \mathfrak{a}_0, A_B^{G,\infty}(P, t_0, T) := A_B^{\infty}(P, t_0, T) \cap G(\mathbb{A})^1$$

COROLLARY 4.10. Let P be a relatively standard parabolic subgroup of G. For sufficiently regular T, the following subset of $M_{P_H}(\mathbb{A}) \cap G(\mathbb{A})^1$

$$\bigcup_{B \in \mathcal{P}(\tilde{P}_0, P)} (\omega_{\tilde{P}_0} \cap M_{P_H}(\mathbb{A})) \cdot (A^{\infty}_{\tilde{P}_0}(P_H, t_0) \cap A^{G, \infty}_B(P, t_0, T_B)) \cdot (K_H \cap M_{P_H}(\mathbb{A}))$$

projects surjectively on $\{m \in M_{P_H}(F) \setminus M_{P_H}(\mathbb{A}) \cap G(\mathbb{A})^1 | F^P(m,T) = 1\}.$

PROOF. This is an analogue of [61, Corollaire 2.5]. By Lemma 4.8, the following subset of $M_{P_H}(\mathbb{A}) \cap G(\mathbb{A})^1$

$$\bigcup_{B \in \mathcal{P}(\widetilde{P}_0, P)} (\omega_{\widetilde{P}_0} \cap M_{P_H}(\mathbb{A})) \cdot (A^{\infty}_{\widetilde{P}_0}(P_H, t_0) \cap A^{G, \infty}_B(P, t_0)) \cdot (K_H \cap M_{P_H}(\mathbb{A}))$$

projects surjectively on $M_{P_H}(F) \setminus M_{P_H}(\mathbb{A}) \cap G(\mathbb{A})^1$. Recall that $\omega_{\tilde{P}_0} \subseteq \omega_B$ for all $B \in \mathcal{P}(\tilde{P}_0, P)$ and that $K_H \subseteq K$ by our choices (see Section 3.4). Therefore, the statement to be proved follows from Proposition 4.9.

4.2. Integrability.

THEOREM 4.11. For all sufficiently regular T and all $s \in \mathbb{R}$,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1}|k_{f,\mathfrak{o}}^T(x)||\mathrm{Nrd}(x_1)|^s_{\mathbb{A}}dx<\infty,$$

where we write $x = (x_1, x_2) \in GL_{p,D}(\mathbb{A}) \times GL_{q,D}(\mathbb{A})$.

PROOF. Let $P_1 \subseteq P_2$ be a pair of relatively standard parabolic subgroups of G. Following [3, §6], for $T_1 \in \mathfrak{a}_{P_1}$, we define the characteristic function

$$\sigma_{P_1}^{P_2}(T_1) := \sum_{\{Q: P_2 \subseteq Q\}} (-1)^{\dim(A_{P_2}/A_Q)} \tau_{P_1}^Q(T_1) \hat{\tau}_Q^G(T_1),$$

and recall that for $P \supseteq P_1$ a relatively standard parabolic subgroup of G,

$$\tau_{P_1}^P(T_1)\hat{\tau}_P^G(T_1) = \sum_{\{P_2: P \subseteq P_2\}} \sigma_{P_1}^{P_2}(T_1).$$

Denote $P_{1,H} := P_1 \cap H$. For $x \in P_{1,H}(F) \setminus H(\mathbb{A})$, we put

$$\chi_{P_1,P_2}^T(x) := F^{P_1}(x,T)\sigma_{P_1}^{P_2}(H_{P_1}(x) - T_{P_1}),$$

and

$$k_{P_1,P_2,\mathfrak{o}}(x) := \sum_{\{P: P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} k_{f,P,\mathfrak{o}}(x).$$

Using Lemma 4.1 and the left invariance of H_P and $k_{f,P,\mathfrak{o}}$ by $P_H(F)$, we have

$$k_{f,\mathfrak{o}}^{T}(x) = \sum_{\{P_{1},P_{2}:\widetilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}} \sum_{\delta\in P_{1,H}(F)\setminus H(F)} \chi_{P_{1},P_{2}}^{T}(\delta x) k_{P_{1},P_{2},\mathfrak{o}}(\delta x).$$

Thus

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}|k_{f,\mathfrak{o}}^{T}(x)||\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx$$

$$\leq \sum_{\mathfrak{o}\in\mathcal{O}}\sum_{\{P_{1},P_{2}:\widetilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}}\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\chi_{P_{1},P_{2}}^{T}(x)|k_{P_{1},P_{2},\mathfrak{o}}(x)||\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx.$$

It suffices to prove that for any pair of relatively standard parabolic subgroups $P_1 \subseteq P_2$ of G,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_{1,H}(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)|k_{P_1,P_2,\mathfrak{o}}(x)||\mathrm{Nrd}(x_1)|^s_{\mathbb{A}}dx<\infty.$$

If $P_1 = P_2 \neq G$, by [3, Lemma 6.1], we have $\sigma_{P_1}^{P_2} = 0$ and then $\chi_{P_1,P_2}^T = 0$, so the integration is zero. If $P_1 = P_2 = G$, by Corollary 4.10, every $x \in H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$ with $F^G(x,T) = 1$ has a representative in the compact subset

$$\bigcup_{B \in \mathcal{P}(\widetilde{P}_0,G)} \omega_{\widetilde{P}_0} \cdot A_B^{G,\infty}(G,t_0,T_B) \cdot K_H,$$

so the integral is bounded by an integral of a continuous function over a compact subset and thus convergent. Therefore, we reduce ourselves to proving the following proposition. \Box

PROPOSITION 4.12. Let $f \in S(\mathfrak{s}(\mathbb{A}))$, $s \in \mathbb{R}$ and $P_1 \subsetneq P_2$ be two relatively standard parabolic subgroups of G. Fix any two positive real numbers ϵ_0 and N. Then there exists a constant C such that

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)|k_{P_{1},P_{2},\mathfrak{o}}(x)||\mathrm{Nrd}(x_{1})|^{s}_{\mathbb{A}}dx\leq Ce^{-N||T|}$$

for all sufficiently regular T satisfying $\alpha(T) \geq \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}^G$.

For $x \in H(F) \setminus H(\mathbb{A})$, define

$$k_{f,G}(x) := \sum_{\mathfrak{o} \in \mathcal{O}} k_{f,G,\mathfrak{o}}(x) = \sum_{X \in \mathfrak{s}(F)} f(x^{-1}Xx)$$

and

(4.2.1)
$$k_f^T(x) := \sum_{\mathfrak{o} \in \mathcal{O}} k_{f,\mathfrak{o}}^T(x)$$

COROLLARY 4.13. Let $f \in S(\mathfrak{s}(\mathbb{A}))$ and $s \in \mathbb{R}$. Fix any two positive real numbers ϵ_0 and N. Then there exists a constant C such that

$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} |k_f^T(x) - F^G(x,T)k_{f,G}(x)| |\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx \le Ce^{-N||T||}$$

for all sufficiently regular T satisfying $\alpha(T) \geq \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}^G$.

PROOF OF PROPOSITION 4.12. Let P be a relatively standard parabolic subgroup of G such that $P_1 \subseteq P \subseteq P_2$. For any $X \in \mathfrak{m}_P(F) \cap \mathfrak{o}$, there exists a unique relatively standard parabolic subgroup R of G such that $P_1 \subseteq R \subseteq P$ and $X \in (\mathfrak{m}_P(F) \cap \mathfrak{r}(F) \cap \mathfrak{o}) - \left(\bigcup_{P_1 \subseteq Q \subsetneq R} \mathfrak{m}_P(F) \cap \mathfrak{q}(F) \cap \mathfrak{o}\right)$. Write

$$\widetilde{\mathfrak{m}}_{P_1}^R := \mathfrak{m}_R - \left(igcup_{\{Q:P_1 \subseteq Q \subsetneq R\}} \mathfrak{m}_R \cap \mathfrak{q}
ight)$$

and

$$\mathfrak{n}_R^P := \mathfrak{n}_R \cap \mathfrak{m}_P.$$

By Corollary 3.6, we have

$$(\mathfrak{m}_P(F) \cap \mathfrak{r}(F) \cap \mathfrak{o}) - \left(\bigcup_{P_1 \subseteq Q \subsetneq R} \mathfrak{m}_P(F) \cap \mathfrak{q}(F) \cap \mathfrak{o}\right) = (\widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}) \oplus ((\mathfrak{n}_R^P \cap \mathfrak{s})(F)).$$

Hence

$$k_{f,P,\mathfrak{o}}(x) = \sum_{X \in \mathfrak{m}_P(F) \cap \mathfrak{o}} \int_{\mathfrak{n}_P \cap \mathfrak{s}(\mathbb{A})} f(x^{-1}(X+U)x) dU$$
$$= \sum_{\{R: P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{X \in (\mathfrak{n}_R^P \cap \mathfrak{s})(F)} \int_{(n_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi+X+U)x) dU.$$

Denote by \overline{P} the parabolic subgroup of G opposite to P and write

$$\overline{\mathfrak{n}}_R^P := \mathfrak{n}_{\overline{R}} \cap \mathfrak{m}_P$$

Note that the restriction of $\langle \cdot, \cdot \rangle$ (defined in (3.5.1)) to $((\mathfrak{n}_R^P \cap \mathfrak{s})(\mathbb{A})) \times ((\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(\mathbb{A}))$ is also non-degenerate. For any $\xi \in (\mathfrak{m}_R \cap \mathfrak{s})(\mathbb{A})$, applying the Poisson summation formula to the Bruhat-Schwartz function $\int_{(\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + \cdot + U)x) dU$, we get

$$\sum_{X \in (\mathfrak{n}_R^P \cap \mathfrak{s})(F)} \int_{(\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + X + U)x) dU = \sum_{\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}),$$
where the partial Fourier transform $\Phi_{\xi}^{x,R}$ of $\int_{(\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + \cdot + U)x) dU$ is defined by

$$\forall \widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(\mathbb{A}), \Phi_{\xi}^{x,R}(\widehat{X}) := \int_{(\mathfrak{n}_R^P \cap \mathfrak{s})(\mathbb{A})} \left(\int_{(\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + X + U)x) dU \right) \Psi(\langle X, \widehat{X} \rangle) dX.$$

Since $\langle U, \widehat{X} \rangle = 0$ for $U \in (\mathfrak{n}_P \cap \mathfrak{s})(\mathbb{A})$ and $\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(\mathbb{A})$, as well as $\mathfrak{n}_R = \mathfrak{n}_P \oplus \mathfrak{n}_R^P$, we have

$$\forall \widehat{X} \in (\overline{\mathfrak{n}}_{R}^{P} \cap \mathfrak{s})(\mathbb{A}), \Phi_{\xi}^{x,R}(\widehat{X}) = \int_{(\mathfrak{n}_{R} \cap \mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + U)x) \Psi(\langle U, \widehat{X} \rangle) dU,$$

whose expression is actually independent of P.

To sum up,

$$k_{f,P,\mathfrak{o}}(x) = \sum_{\{R: P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}).$$

Hence

$$\begin{aligned} k_{P_1,P_2,\mathfrak{o}}(x) &= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} k_{f,P,\mathfrak{o}}(x) \\ &= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \left(\sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}) \right) \\ &= \sum_{\{R:P_1 \subseteq R \subseteq P_2\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}} \left(\sum_{\{P:R \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \sum_{\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}) \right) \end{aligned}$$

For a relatively standard parabolic subgroup P_3 of G containing R, we write

$$(\overline{\mathfrak{n}}_R^{P_3})' := \overline{\mathfrak{n}}_R^{P_3} - \left(\bigcup_{\{Q:R\subseteq Q\subsetneq P_3\}} \overline{\mathfrak{n}}_R^Q\right).$$

Then

$$\sum_{\widehat{X} \in (\overline{\mathfrak{n}}_R^P \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}) = \sum_{\{P_3 : R \subseteq P_3 \subseteq P\}} \sum_{\widehat{X} \in ((\overline{\mathfrak{n}}_R^{P_3})' \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}).$$

We have

$$\sum_{\{P:R\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \sum_{\widehat{X}\in(\overline{\mathfrak{n}}_R^P\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X})$$

$$= \sum_{\{P:R\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \sum_{\{P_3:R\subseteq P_3\subseteq P\}} \sum_{\widehat{X}\in((\overline{\mathfrak{n}}_R^{P_3})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X})$$

$$= (-1)^{\dim(A_{P_2}/A_G)} \sum_{\{P_3:R\subseteq P_3\subseteq P_2\}} \sum_{\widehat{X}\in((\overline{\mathfrak{n}}_R^{P_3})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}) \sum_{\{P:P_3\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_{P_2})}$$

From [3, Proposition 1.1], we know that

$$\sum_{\{P:P_3 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_{P_2})} = \begin{cases} 1, & \text{if } P_3 = P_2; \\ 0, & \text{otherwise.} \end{cases}$$

We obtain

$$\sum_{\{P:R\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \sum_{\widehat{X}\in(\overline{\mathfrak{n}}_R^P\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}) = (-1)^{\dim(A_{P_2}/A_G)} \sum_{\widehat{X}\in((\overline{\mathfrak{n}}_R^{P_2})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}).$$

Thus

$$k_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_G)} \sum_{\{R: P_1 \subseteq R \subseteq P_2\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{\widehat{X} \in ((\overline{\mathfrak{n}}_R^{P_2})' \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{X}).$$

$$\begin{aligned} &\operatorname{Now} \\ &\sum_{\mathfrak{o}\in\mathcal{O}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) |k_{P_{1},P_{2},\mathfrak{o}}(x)| |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx \\ &\leq \sum_{\mathfrak{o}\in\mathcal{O}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \left(\sum_{\{R:P_{1}\subseteq R\subseteq P_{2}\}} \sum_{\xi\in\widetilde{\mathfrak{m}}_{P_{1}}^{R}(F)\cap\mathfrak{o}} \sum_{\widehat{X}\in((\overline{\mathfrak{n}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\widehat{X})| \right) |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx \\ &= \sum_{\{R:P_{1}\subseteq R\subseteq P_{2}\}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\xi\in(\widetilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\widehat{X}\in((\overline{\mathfrak{n}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\widehat{X})| |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx. \end{aligned}$$

We reduce ourselves to bounding

(4.2.2)
$$\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \chi^T_{P_1,P_2}(x) \sum_{\xi \in (\widetilde{\mathfrak{m}}^R_{P_1}\cap \mathfrak{s})(F)} \sum_{\widehat{X} \in ((\overline{\mathfrak{n}}^{P_2}_R)' \cap \mathfrak{s})(F)} |\Phi^{x,R}_{\xi}(\widehat{X})| |\operatorname{Nrd}(x_1)|^s_{\mathbb{A}} dx$$

for any fixed relatively standard parabolic subgroup R of G such that $P_1 \subseteq R \subseteq P_2$.

By the Iwasawa decomposition and our choice of measures, we have

$$\begin{split} &\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)\sum_{\xi\in(\tilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)}\sum_{\hat{X}\in((\tilde{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\hat{X})||\mathrm{Nrd}(x_{1})|^{s}_{\mathbb{A}}dx\\ &=\int_{K_{H}}\int_{M_{P_{1,H}}(F)\backslash M_{P_{1,H}}(\mathbb{A})\cap G(\mathbb{A})^{1}}\int_{N_{P_{1,H}}(F)\backslash N_{P_{1,H}}(\mathbb{A})}F^{P_{1}}(m_{1},T)\sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(m_{1})-T_{P_{1}})\\ &\cdot\sum_{\xi\in(\tilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)}\sum_{\hat{X}\in((\tilde{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{n_{1}m_{1}k,R}(\hat{X})|e^{-2\rho_{P_{1,H}}(H_{P_{1,H}}(m_{1}))}|\mathrm{Nrd}(m_{1,1})|^{s}_{\mathbb{A}}dn_{1}dm_{1}dk, \end{split}$$

where we write $m_1 = (m_{1,1}, m_{1,2}) \in GL_{p,D}(\mathbb{A}) \times GL_{q,D}(\mathbb{A}).$

By Corollary 4.10, the following subset of $M_{P_{1,H}}(\mathbb{A}) \cap G(\mathbb{A})^1$ $\left| \begin{array}{c} (\omega_{\tilde{n}} \cap M_{P_{1,H}}(\mathbb{A})) \cdot (A_{\tilde{n}}^{\infty}(P_{1,H},t_0) \cap A_{P}^{G,\infty}(P_{1},t_0,T_B)) \cdot (K_{H} \cap M_{P_{1,H}}(\mathbb{A})) \end{array} \right|$

$$\bigcup_{B\in\mathcal{P}(\tilde{P}_0,P_1)} (\omega_{\tilde{P}_0}\cap M_{P_{1,H}}(\mathbb{A})) \cdot (A_{\tilde{P}_0}^{\infty}(P_{1,H},t_0)\cap A_B^{\odot,\infty}(P_1,t_0,T_B)) \cdot (K_H\cap M_{P_{1,H}}(\mathbb{A}))$$

projects surjectively on $\{m_1 \in M_{P_{1,H}}(F) \setminus M_{P_{1,H}}(\mathbb{A}) \cap G(\mathbb{A})^1 | F^{P_1}(m_1,T) = 1\}$. Hence

$$\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\xi \in (\widetilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\widehat{X} \in ((\overline{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\widehat{X})| |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx \\
\leq c_{1} \sum_{B \in \mathcal{P}(\widetilde{P}_{0},P_{1})} \int_{K_{H}} \int_{[cpt \subseteq M_{\widetilde{P}_{0}}(\mathbb{A})^{1}]} \int_{A_{B}^{G,\infty}(P_{1},t_{0},T_{B})} \int_{[cpt \subseteq N_{\widetilde{P}_{0}}^{P_{2},H}(\mathbb{A})]} \int_{[cpt \subseteq N_{P_{2},H}(\mathbb{A})]} \sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a) - T_{P_{1}}) \\
\cdot \sum_{\xi \in (\widetilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\widehat{X} \in ((\overline{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{n_{2}namk,R}(\widehat{X})| e^{-2\rho_{\widetilde{P}_{0}}(H_{B}(a))} |\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{s} dn_{2} dn da dm dk,$$

where $c_1 = \operatorname{vol}(K_H \cap M_{P_{1,H}}(\mathbb{A}))$ is a constant independent of T, and all the compact subsets in the integrals are independent of T. (We use the notation $[cpt \subseteq *]$ for denoting a compact subset in *.)

LEMMA 4.14. Let $x \in H(\mathbb{A}), \xi \in (\mathfrak{m}_R \cap \mathfrak{s})(\mathbb{A})$ and $\widehat{X} \in (\overline{\mathfrak{n}}_R \cap \mathfrak{s})(\mathbb{A})$. Let $R \subseteq P_2$ be a pair of relatively standard parabolic subgroups of G. For $n_2 \in N_{P_{2,H}}(\mathbb{A})$, we have

$$\Phi_{\xi}^{n_2 x, R}(\widehat{X}) = \Phi_{\xi}^{x, R}(\widehat{X})$$

PROOF OF LEMMA 4.14. Let $U_2 := n_2^{-1} \xi n_2 - \xi$. Then

$$\begin{split} \Phi_{\xi}^{n_2 x, R}(\widehat{X}) &= \int_{(\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})} f(x^{-1} n_2^{-1} (\xi + U) n_2 x) \Psi(\langle U, \widehat{X} \rangle) dU \\ &= \int_{(\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})} f(x^{-1} (\xi + U_2 + n_2^{-1} U n_2) x) \Psi(\langle U, \widehat{X} \rangle) dU. \end{split}$$

Since both U_2 and $n_2^{-1}Un_2 - U$ belong to $(\mathfrak{n}_{P_2} \cap \mathfrak{s})(\mathbb{A})$, we have

$$\langle U_2 + n_2^{-1}Un_2 - U, \widehat{X} \rangle = 0,$$

so

$$\Phi_{\xi}^{n_{2}x,R}(\widehat{X}) = \int_{(\mathfrak{n}_{R}\cap\mathfrak{s})(\mathbb{A})} f(x^{-1}(\xi + U_{2} + n_{2}^{-1}Un_{2})x)\Psi(\langle U_{2} + n_{2}^{-1}Un_{2}, \widehat{X}\rangle)dU$$

Because the change of variables $U_2 + n_2^{-1}Un_2 \mapsto U$ does not change the Haar measure, we obtain

$$\Phi_{\xi}^{n_2 x, R}(\widehat{X}) = \Phi_{\xi}^{x, R}(\widehat{X}).$$

Using Lemma 4.14, we get

$$\Phi_{\xi}^{n_2namk,R}(\widehat{X}) = \Phi_{\xi}^{namk,R}(\widehat{X}) = \Phi_{\xi}^{aa^{-1}namk,R}(\widehat{X}).$$

By change of variables $a^{-1}Ua \mapsto U$, using the fact that

$$\langle U, \widehat{X} \rangle = \langle a^{-1} U a, a^{-1} \widehat{X} a \rangle,$$

we have

$$\Phi_{\xi}^{n_2 namk, R}(\widehat{X}) = e^{2\rho_{R, +}(H_B(a))} \Phi_{a^{-1}\xi a}^{a^{-1}namk, R}(a^{-1}\widehat{X}a)$$

where we denote by $\rho_{R,+}$ the half of the sum of weights (with multiplicities) for the action of A_0 on $\mathfrak{n}_R \cap \mathfrak{s}$. From the reduction theory (see [3, p. 944]), we know that for a satisfying $\sigma_{P_1}^{P_2}(H_{P_1}(a) - T_{P_1}) \neq 0$, $a^{-1}na$ belongs to a compact subset independent of T. In sum,

$$\int_{P_{1,H}(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\xi \in (\widetilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\widehat{X} \in ((\overline{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\widehat{X})| |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx \\
\leq c_{2} \sum_{B \in \mathcal{P}(\widetilde{P}_{0},P_{1})} \sup_{y \in \Gamma} \int_{A_{B}^{G,\infty}(P_{1},t_{0},T_{B})} e^{(2\rho_{R,+}-2\rho_{\widetilde{P}_{0}})(H_{B}(a))} \sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a)-T_{P_{1}}) \\
\cdot \sum_{\xi \in (\widetilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\widehat{X} \in ((\overline{\mathfrak{n}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{a^{-1}\xi a}^{y,R}(a^{-1}\widehat{X}a)| |\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{s} da,$$

where c_2 is a constant independent of T, and Γ is a compact subset independent of T.

Denote by \mathcal{O}_F the ring of integers of F. Fix an F-basis for each weight space for the action of A_0 on $\mathfrak{s}(F)$. Then we are authorised to talk about \mathcal{O}_F -points of such a weight space. Since the function $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ is compactly supported on finite places, there exists a positive integer N_1 independent of T such that the sums over $\xi \in (\widetilde{\mathfrak{m}}_{P_1}^R \cap \mathfrak{s})(F)$ and $\widehat{X} \in ((\overline{\mathfrak{n}}_R^{P_2})' \cap \mathfrak{s})(F)$ can be restricted to lattices $\frac{1}{N_1}(\widetilde{\mathfrak{m}}_{P_1}^R \cap \mathfrak{s})(\mathcal{O}_F)$ and $\frac{1}{N_1}((\overline{\mathfrak{n}}_R^{P_2})' \cap \mathfrak{s})(\mathcal{O}_F)$ respectively. In fact, N_1 can be made explicit as in [13, §1.9] by replacing \mathfrak{m}_R and \mathfrak{n}_R in *loc. cit.* with $\mathfrak{m}_R \cap \mathfrak{s}$ and $\mathfrak{n}_R \cap \mathfrak{s}$ respectively.

Fix a Euclidean norm $\|\cdot\|$ on the \mathbb{R} -linear space $\mathfrak{s}(F \otimes_{\mathbb{Q}} \mathbb{R})$. Consider a sufficiently large integer k > 0 to be described precisely at the end of the proof. There exists an integer $m \ge 0$, a real number $k_{\alpha} \geq 0$ for each $\alpha \in \Delta_B^{P_2}$, and a real number $c_3 > 0$ satisfying the following conditions (cf. [13, (4.10) in p. 372]):

(1) if
$$R = P_2, m = 0;$$

(2) for all
$$\alpha \in \Delta_B^{P_2} - \Delta_B^R, k_\alpha \ge k;$$

(3) for all $a \in A_B^\infty$,

(3) for all
$$a \in A^{\circ}_{T}$$

(4.2.3)

$$\sum_{\widehat{X} \in \frac{1}{N_1}((\overline{\mathfrak{n}}_R^{P_2})' \cap \mathfrak{s})(\mathcal{O}_F)} \|a^{-1}\widehat{X}a\|^{-m} \le c_3 \prod_{\alpha \in \Delta_B^{P_2}} e^{-k_\alpha \alpha(H_B(a))}.$$

We can choose a multi-index \vec{i} whose sum of components is m. We extend the differential operator $\partial \vec{i}$ on $\mathfrak{s}(F \otimes \mathbb{R})$ to $\mathfrak{s}(\mathbb{A})$ by defining $\partial^{\vec{i}}(f_{\infty} \otimes \chi^{\infty}) := (\partial^{\vec{i}}f_{\infty}) \otimes \chi^{\infty}$ (see Section 2.3). Write

$$\Phi_{\xi}^{x,R,\overrightarrow{i}}(\widehat{X}) := \int_{(\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})} (\partial^{\overrightarrow{i}} f)(x^{-1}(\xi+U)x)\Psi(\langle U,\widehat{X}\rangle)dU.$$

Invoking integration by parts, for $\hat{X} \neq 0$, we get

$$|\Phi_{a^{-1}\xi a}^{y,R}(a^{-1}\widehat{X}a)| = c_4(y) ||a^{-1}\widehat{X}a||^{-m} |\Phi_{a^{-1}\xi a}^{y,R,\overrightarrow{i}}(a^{-1}\widehat{X}a)|,$$

where $c_4(y)$ is a continuous function of y.

Denote by $\Phi(A_B, \mathfrak{m}_R \cap \mathfrak{s})$ the set of weights of A_B in $\mathfrak{m}_R \cap \mathfrak{s}$. For any $\mu \in \Phi(A_B, \mathfrak{m}_R \cap \mathfrak{s})$, let \mathfrak{m}_{μ} be the corresponding weight space. From [53, §41], we know that there exists a function $\phi_{\mu} \in \mathcal{S}(\mathfrak{m}_{\mu}(\mathbb{A}))$ for each

 $\mu \in \Phi(A_B, \mathfrak{m}_R \cap \mathfrak{s}) \text{ and a function } \phi_{\mathfrak{n}_R \cap \mathfrak{s}} \in \mathcal{S}((\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})) \text{ such that for all } \xi + U \in (\mathfrak{m}_R \cap \mathfrak{s})(\mathbb{A}) \oplus (\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})$ and $y \in \Gamma$,

$$|(\partial^{\overrightarrow{i}}f)(y^{-1}(\xi+U)y)| \le \left(\prod_{\mu\in\Phi(A_B,\mathfrak{m}_R\cap\mathfrak{s})}\phi_{\mu}(\xi_{\mu})\right)\phi_{\mathfrak{n}_R\cap\mathfrak{s}}(U),$$

where ξ_{μ} denotes the projection of ξ to $\mathfrak{m}_{\mu}(\mathbb{A})$.

Now

$$\begin{split} &\sum_{\xi \in (\tilde{\mathfrak{m}}_{P_{1}}^{R} \cap \mathfrak{s})(F)} \sum_{\hat{X} \in ((\bar{\mathfrak{m}}_{R}^{P_{2}})' \cap \mathfrak{s})(F)} |\Phi_{a^{-1}\xi a}^{y,R}(a^{-1}\hat{X}a)| \\ &= \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{P_{1}}^{R} \cap \mathfrak{s})(\mathcal{O}_{F})} \sum_{\hat{X} \in \frac{1}{N_{1}} ((\bar{\mathfrak{m}}_{R}^{P_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} |\Phi_{a^{-1}\xi a}^{y,R}(a^{-1}\hat{X}a)| \\ &= \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{P_{1}}^{R} \cap \mathfrak{s})(\mathcal{O}_{F})} \sum_{\hat{X} \in \frac{1}{N_{1}} ((\bar{\mathfrak{m}}_{R}^{P_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} c_{4}(y) ||a^{-1}\hat{X}a||^{-m} |\Phi_{a^{-1}\xi a}^{y,R,\overrightarrow{i}}(a^{-1}\hat{X}a)| \\ &\leq c_{5} \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{P_{1}}^{R} \cap \mathfrak{s})(\mathcal{O}_{F})} \left(\prod_{\mu \in \Phi(A_{B},\mathfrak{m}_{R} \cap \mathfrak{s})} \phi_{\mu}(\mu(a)^{-1}\xi_{\mu}) \right) \cdot \sum_{\hat{X} \in \frac{1}{N_{1}} ((\bar{\mathfrak{m}}_{R}^{P_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} ||a^{-1}\hat{X}a||^{-m} \\ &\leq c_{5}c_{3} \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{P_{1}}^{R} \cap \mathfrak{s})(\mathcal{O}_{F})} \left(\prod_{\mu \in \Phi(A_{B},\mathfrak{m}_{R} \cap \mathfrak{s})} \phi_{\mu}(\mu(a)^{-1}\xi_{\mu}) \right) \cdot \prod_{\alpha \in \Delta_{B}^{P_{2}}} e^{-k_{\alpha}\alpha(H_{B}(a))}, \end{split}$$

where $c_5 := \sup_{y \in \Gamma} c_4(y) \int_{(\mathfrak{n}_R \cap \mathfrak{s})(\mathbb{A})} \phi_{\mathfrak{n}_R \cap \mathfrak{s}}(U) dU$, and we have used (4.2.3) in the last inequality. Thus

$$\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\xi\in(\tilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(F)} \sum_{\hat{X}\in((\tilde{\mathfrak{m}}_{R}^{P_{2}})'\cap\mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\hat{X})||\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx$$

$$\leq c_{2}c_{5}c_{3} \sum_{B\in\mathcal{P}(\tilde{P}_{0},P_{1})} \int_{A_{B}^{G,\infty}(P_{1},t_{0},T_{B})} e^{(2\rho_{R,+}-2\rho_{\tilde{P}_{0}})(H_{B}(a))} \sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a)-T_{P_{1}})$$

$$\cdot \sum_{\xi\in\frac{1}{N_{1}}(\tilde{\mathfrak{m}}_{P_{1}}^{R}\cap\mathfrak{s})(\mathcal{O}_{F})} \left(\prod_{\mu\in\Phi(A_{B},\mathfrak{m}_{R}\cap\mathfrak{s})} \phi_{\mu}(\mu(a)^{-1}\xi_{\mu})\right) \cdot \prod_{\alpha\in\Delta_{B}^{P_{2}}} e^{-k_{\alpha}\alpha(H_{B}(a))}|\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{s} da.$$

From [13, p. 375], we know that for all $a \in A_B^{G,\infty}(P_1, t_0, T_B)$ satisfying $\sigma_{P_1}^{P_2}(H_{P_1}(a) - T_{P_1}) \neq 0$ and $\alpha \in \Delta_B^{P_2}$, we have $\alpha(H_B(a)) > t_0$. Denote by $\Sigma_B^{\mathfrak{m}_R \cap \mathfrak{s}}$ the positive weights of $\mathfrak{m}_R \cap \mathfrak{s}$ under the action of A_B . Consider the subsets S of $\Sigma_B^{\mathfrak{m}_R \cap \mathfrak{s}}$ with the following property: for all $\alpha \in \Delta_B^R - \Delta_B^{P_1}$, there exists $u \in C$ such that its a coordinate is ≥ 0 . Then $\mu \in S$ such that its α -coordinate is > 0. Then

$$\begin{split} & \sum_{\xi \in \frac{1}{N_1} \widetilde{\mathfrak{m}}_{P_1}^R \cap \mathfrak{s}(\mathcal{O}_F)} \left(\prod_{\mu \in \Phi(A_B, \mathfrak{m}_R \cap \mathfrak{s})} \phi_\mu(\mu(a)^{-1} \xi_\mu) \right) \\ & \leq \sum_S \left[\prod_{\mu \in S} \left(\sum_{\xi_- \in \frac{1}{N_1} \mathfrak{m}_{-\mu}(\mathcal{O}_F) - \{0\}} \phi_{-\mu}(\mu(a) \xi_-) \right) \right] \left[\prod_{\mu \in \Sigma_B^{\mathfrak{m}_R \cap \mathfrak{s}}} \left(\sum_{\xi_+ \in \frac{1}{N_1} \mathfrak{m}_\mu(\mathcal{O}_F)} \phi_\mu(\mu(a^{-1}) \xi_+) \right) \right] \\ & \cdot \left[\sum_{\xi_0 \in \frac{1}{N_1} \mathfrak{m}_0(\mathcal{O}_F)} \phi_0(\xi_0) \right]. \end{split}$$

As in [13, p. 373], for the first and third factors, we also have

$$\prod_{\mu \in S} \left(\sum_{\xi_- \in \frac{1}{N_1} \mathfrak{m}_{-\mu}(\mathcal{O}_F) - \{0\}} \phi_{-\mu}(\mu(a)\xi_-) \right) \le c_6 \prod_{\alpha \in \Delta_B^R - \Delta_B^{P_1}} e^{-k\alpha(H_B(a))}$$

`

and

$$\sum_{\xi_0 \in \frac{1}{N_1} \mathfrak{m}_0(\mathcal{O}_F)} \phi_0(\xi_0) \le c_7,$$

where c_6 and c_7 are constants independent of T and a. One may note that our bound for the second factor is slightly different from [13, (4.14) in p. 373]. Actually we get

$$\prod_{\mu \in \Sigma_B^{\mathfrak{m}_R \cap \mathfrak{s}}} \left(\sum_{\xi_+ \in \frac{1}{N_1} \mathfrak{m}_{\mu}(\mathcal{O}_F)} \phi_{\mu}(\mu(a^{-1})\xi_+) \right) \leq c_8 e^{(2\rho_{B,+} - 2\rho_{R,+})(H_B(a))},$$

where c_8 is a constant independent of T. However, we claim that this discrepancy will be unimportant when we follow the rest of the proof of [13, p. 375], as mentioned in [61, end of Theorem 3.7]. In fact, it suffices to add a factor

$$e^{(2\rho_{B,+}-2\rho_{\tilde{P}_{0}})(H_{B}(a))}|\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{s} = e^{(2\rho_{B,+}-2\rho_{\tilde{P}_{0}})(H_{B}(a))} \left(\frac{|\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{1/p}}{|\operatorname{Nrd}(a_{2})|_{\mathbb{A}}^{1/q}}\right)^{\frac{pq}{p+q}s}$$

in the form of $\prod_{\alpha \in \Delta_B^G} e^{c_{\alpha} \alpha(H_B(\alpha))}$ to [13, (4.17) in p. 375], where c_{α} are constant coefficients. This factor

only results in an extra factor $e^{c_9||T||+c_{10}t}$ to the integral in [13, (4.18) in p. 375], where c_9 and c_{10} are constant coefficients (here we have used [3, Corollary 6.2]). Since we can choose sufficiently large k, it does not matter. Hence we complete the argument of our claim and conclude.

5. Exponential polynomial distributions

Let T be sufficiently regular, $\mathfrak{o} \in \mathcal{O}$ and η be the quadratic character of $\mathbb{A}^{\times}/F^{\times}$ attached to a quadratic field extension E/F. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $s \in \mathbb{C}$, define

(5.0.1)
$$J^{G,T}_{\mathfrak{o}}(\eta, s, f) := \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} k^T_{f,\mathfrak{o}}(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|^s_{\mathbb{A}} dx$$

and

$$J^{G,T}(\eta, s, f) := \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} k_f^T(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx$$

where $k_{f,\mathfrak{o}}^T(x)$ and $k_f^T(x)$ are defined by (4.0.1) and (4.2.1) respectively, and we write $x = (x_1, x_2) \in GL_{p,D}(\mathbb{A}) \times GL_{q,D}(\mathbb{A})$. From Theorem 4.11, we know that $J_{\mathfrak{o}}^{G,T}(\eta, s, \cdot)$ and $J^{G,T}(\eta, s, \cdot)$ are well-defined distributions on $\mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and that

$$J^{G,T}(\eta,s,f) = \sum_{\mathfrak{o}\in\mathcal{O}} J^{G,T}_{\mathfrak{o}}(\eta,s,f),$$

which is an analogue of the geometric side of Arthur's trace formula.

5.1. A generalised case in the product form. Let Q be a relatively standard parabolic subgroup of G. Then

$$M_Q \simeq GL_{p_1+q_1,D} \times \cdots \times GL_{p_l+q_l,D}$$

and

$$M_{Q_H} \simeq GL_{p_1,D} \times \cdots \times GL_{p_l,D} \times GL_{q_1,D} \times \cdots \times GL_{q_l,D}$$

where $\sum_{i=1}^{l} p_i = p$, $\sum_{i=1}^{l} q_i = q$ and we allow p_i or q_i to be zero. The tangent space of M_Q/M_{Q_H} at the neutral element is

$$\mathfrak{m}_Q \cap \mathfrak{s} \simeq \bigoplus_{\{1 \le i \le l \mid p_i q_i \ne 0\}} \begin{pmatrix} 0 & \operatorname{Mat}_{p_i \times q_i, D} \\ \operatorname{Mat}_{q_i \times p_i, D} & 0 \end{pmatrix}.$$

The conjugate action of $M_{Q_H}(F)$ on $(\mathfrak{m}_Q \cap \mathfrak{s})(F)$ can be described as follows:

(1) if $p_i q_i \neq 0$, $\begin{pmatrix} GL_{p_i}(D) \\ GL_{q_i}(D) \end{pmatrix}$ acts on $\begin{pmatrix} 0 & \operatorname{Mat}_{p_i \times q_i}(D) \\ \operatorname{Mat}_{q_i \times p_i}(D) & 0 \end{pmatrix}$ by conjugation; (2) if $p_i q_i = 0$, $\begin{pmatrix} GL_{p_i}(D) \\ GL_{q_i}(D) \end{pmatrix}$ acts on 0 (viewed as a 0-dimensional vector space) trivially. We may generalise integrability in last section to the product setting here whose proof is similar.

Define a relation of equivalence on $(\mathfrak{m}_Q \cap \mathfrak{s})(F)$ which is similar to that on $\mathfrak{s}(F)$ on each component. We denote by $\mathcal{O}^{\mathfrak{m}_Q \cap \mathfrak{s}}$ the set of equivalent classes for this relation. For $\mathfrak{o} \in \mathcal{O}$, the intersection $\mathfrak{o} \cap \mathfrak{m}_Q(F)$ is a finite (perhaps empty) union of classes $\mathfrak{o}_1, \dots, \mathfrak{o}_t \in \mathcal{O}^{\mathfrak{m}_Q \cap \mathfrak{s}}$. Fix the minimal parabolic subgroup $\widetilde{P}'_0 := \widetilde{P}_0 \cap M_{Q_H} = \widetilde{P}_0 \cap M_Q$ of M_{Q_H} and its Levi factor M_0 . We say that a parabolic subgroup P'of M_Q is semi-standard (resp. relatively standard) if $M_0 \subseteq P'$ (resp. $\widetilde{P}'_0 \subseteq P'$). Notice that there exists a bijection from the set of semi-standard (resp. relatively standard) parabolic subgroups of Gcontained in Q to the set of semi-standard (resp. relatively standard) parabolic subgroups of M_Q given by $P \mapsto P \cap M_Q$, whose inverse is given by $P' \mapsto P'N_Q$.

Choose $\varsigma_Q \in \Omega^G$ (not unique) such that $\varsigma_Q P_0 \subseteq Q$. Fix the minimal semi-standard parabolic subgroup $P'_0 := (\varsigma_Q P_0) \cap M_Q$ of M_Q depending on the choice of ς_Q . For any semi-standard parabolic subgroup P' of M_Q and $T \in \mathfrak{a}_0$, denote by $T_{P'}$ the projection of sT in $\mathfrak{a}_{P'}$, where $s \in \Omega^{M_Q}$ such that $sP'_0 \subseteq P'$. For $s \in \Omega^{M_Q}$ and a semi-standard parabolic subgroup $P \subseteq Q$ of G, we see that $sP'_0 \subseteq P \cap M_Q$ if and only if $s\varsigma_Q P_0 \subseteq P$. Then $(\varsigma_Q T)_{P \cap M_Q} = T_P$ which is independent of the choice of ς_Q . This is also the reason why we introduce ς_Q . If $T \in \mathfrak{a}_{P_0}^+$ is sufficiently regular with respect to $P_0 \subseteq G$, then $\varsigma_Q T \in \mathfrak{a}_{P'}^+$ is sufficiently regular with respect to $P'_0 \subseteq M_Q$.

 $\varsigma_Q T \in \mathfrak{a}_{P'_0}^+$ is sufficiently regular with respect to $P'_0 \subseteq M_Q$. Let $f' \in \mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A})), P'$ be a relatively standard parabolic subgroup of M_Q and $1 \leq j \leq t$. Write $P'_H := P' \cap M_{Q_H} = P' \cap H$. For $x \in M_{P'_H}(F)N_{P'_H}(\mathbb{A}) \setminus M_{Q_H}(\mathbb{A})$, define

(5.1.1)
$$k_{f',P',\mathfrak{o}_j}^{M_Q}(x) := \sum_{X \in \mathfrak{m}_{P'}(F) \cap \mathfrak{o}_j} \int_{(\mathfrak{n}_{P'} \cap \mathfrak{s})(\mathbb{A})} f'(x^{-1}(X+U)x) dU.$$

For $T \in \mathfrak{a}_0$ and $x \in M_{Q_H}(F) \setminus M_{Q_H}(\mathbb{A})$, define

$$k_{f',\mathfrak{o}_{j}}^{Q,T}(x) := \sum_{\{P': \tilde{P}'_{0} \subseteq P'\}} (-1)^{\dim(A_{P'}/A_{M_{Q}})} \sum_{\delta \in P'_{H}(F) \setminus M_{Q_{H}}(F)} \widehat{\tau}_{P'}^{M_{Q}}(H_{P'}(\delta x) - T_{P'}) \cdot k_{f',P',\mathfrak{o}_{j}}^{M_{Q}}(\delta x).$$

For sufficiently regular $T \in \mathfrak{a}_{P_0}^+$ and $\{s_i\}_{1 \leq i \leq l} \in \mathbb{C}^l$, define

$$J^{Q,T}_{\mathfrak{o}_{j}}(\eta,\{s_{i}\},f') := \int_{M_{Q_{H}}(F)\backslash M_{Q_{H}}(\mathbb{A})\cap M_{Q}(\mathbb{A})^{1}} k^{Q,\varsigma_{Q}T}_{f',\mathfrak{o}_{j}}(x)\eta(\operatorname{Nrd}(x)) \prod_{1\leq i\leq l} |\operatorname{Nrd}(x_{i,1})|^{s_{i}}_{\mathbb{A}} dx_{i,1}|^{s_{i}} dx_{i,1}|^{s_{i}$$

where we write $x = (x_1, ..., x_l) \in GL_{p_1+q_1,D}(\mathbb{A}) \times \cdots \times GL_{p_l+q_l,D}(\mathbb{A})$ and $x_i = (x_{i,1}, x_{i,2}) \in GL_{p_i,D}(\mathbb{A}) \times GL_{q_i,D}(\mathbb{A})$. As explained above, $k_{f',\mathfrak{o}_j}^{Q,\varsigma_QT}$ and $J_{\mathfrak{o}_j}^{Q,T}$ are independent of the choice of ς_Q . Then we have well-defined distributions $J_{\mathfrak{o}_j}^{Q,T}(\eta, \{s_i\}, \cdot)$ on $\mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}))$. It only depends on the projection of $\varsigma_Q T$ to $\mathfrak{a}_{\varsigma_Q P_0}^Q$ and does not depend on T_Q . Now we define

(5.1.2)
$$J_{\mathfrak{o}}^{Q,T} := \sum_{j=1}^{t} J_{\mathfrak{o}_{j}}^{Q,T}$$

and

$$J^{Q,T} := \sum_{\mathfrak{o} \in \mathcal{O}} J^{Q,T}_{\mathfrak{o}}.$$

For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, define $f_Q^{\eta} \in \mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}))$ by

(5.1.3)
$$\forall X \in (\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}), f_Q^{\eta}(X) := \int_{K_H} \int_{(\mathfrak{n}_Q \cap \mathfrak{s})(\mathbb{A})} f(k^{-1}(X+V)k)\eta(\operatorname{Nrd}(k))dVdk.$$

5.2. ω -stable parabolic subgroups. In our case, we can embed G into \mathfrak{g} in the standard way. For any linear subspace \mathfrak{v} of \mathfrak{g} , we denote by \mathfrak{v}^{\times} the intersection of \mathfrak{v} and G in \mathfrak{g} . Assume that p = q. Let us denote n := p = q. Then $\mathfrak{s}^{\times}(F)$ is the union of classes in \mathcal{O}^{\times} . Let $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix} \in G(F)$. By the notation in Section 3.4, ω is the element in G exchanging e_i and f_i for all $1 \le i \le n$. Then $\omega \widetilde{P}_0 \omega^{-1} = \widetilde{P}_0$. We say that a semi-standard parabolic subgroup Q of G is " ω -stable" if $\omega Q \omega^{-1} = Q$. By Chevalley's theorem, this condition is equivalent to $\omega \in Q$. For a relatively standard parabolic subgroup Q of G, we see that Q is ω -stable if and only if $p_i = q_i$ for any $1 \le i \le l$; an illustrating example for l = 2 looks like

$$Q = \left(\begin{array}{cccc} * & * & * & * \\ 0 & * & 0 & * \\ * & * & * & * \\ 0 & * & 0 & * \end{array}\right)^{\times}$$

Notice that there is a bijection $P_n \mapsto \begin{pmatrix} \mathfrak{p}_n & \mathfrak{p}_n \\ \mathfrak{p}_n & \mathfrak{p}_n \end{pmatrix}^{\times}$ from the set of standard parabolic subgroups in $GL_{n,D}$ (namely containing the group of upper triangular matrices) to the set of ω -stable relatively standard parabolic subgroups in G. For $Q \subseteq R$ a pair of relatively standard parabolic subgroups of G, one sees that "Q is ω -stable" implies "R is ω -stable", but "R is ω -stable" does not imply "Q is ω -stable". For any relative standard parabolic subgroup Q of G, define

$$\overline{Q}^{\omega\text{-st}} := \bigcap_{\{R: Q \subseteq R, \omega R \omega^{-1} = R\}} R,$$

which is the minimal ω -stable parabolic subgroup of G containing Q.

Denote by $\rho_{Q,+}$ the half of the sum of weights (with multiplicities) for the action of A_0 on $\mathfrak{n}_Q \cap \mathfrak{s}$. We see that $\rho_{Q,+} = \rho_Q - \rho_{Q_H}$ and that for $Q \subseteq R$ a pair of relatively standard parabolic subgroup of G, the restriction of $(2\rho_{Q,+} - 2\rho_{Q_H})|_{\mathfrak{a}_Q}$ to \mathfrak{a}_R equals $(2\rho_{R,+} - 2\rho_{R_H})|_{\mathfrak{a}_R}$.

PROPOSITION 5.1. Assume that p = q = n. Let $\mathfrak{o} \in \mathcal{O}$. The following three conditions are equivalent: (1) $\mathfrak{o} \in \mathcal{O}^{\times}$;

- (2) for all relatively standard parabolic subgroup Q of G, if $\mathfrak{o} \cap \mathfrak{q}(F) \neq \emptyset$, then Q is ω -stable;
- (3) for all relatively standard parabolic subgroup Q of G, if $\mathfrak{o} \cap \mathfrak{m}_Q(F) \neq \emptyset$, then Q is ω -stable.

PROOF. The direction $(2) \Rightarrow (3)$ is trivial. We actually have $(2) \Leftrightarrow (3)$ from Proposition 3.5.

Next, we prove the direction $(1) \Rightarrow (2)$. We assume that $\mathfrak{o} \in \mathcal{O}^{\times}$ and that $\mathfrak{o} \cap \mathfrak{q}(F) \neq \emptyset$ for some relatively standard parabolic subgroup Q of G. If Q is not ω -stable, let k be the minimal integer such that $1 \leq k \leq l-1$ and that

$$\sum_{1 \le i \le k} p_i - \sum_{1 \le i \le k} q_i \neq 0$$

Without loss of generality, we may assume that

$$\sum_{\leq i \leq k} p_i - \sum_{1 \leq i \leq k} q_i < 0.$$

 $1 \le i \le k \qquad 1 \le i \le \kappa$ Let $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{o} \cap \mathfrak{q}(F)$. Then $A \in \mathfrak{gl}_n(D)$ is in the form of $\begin{pmatrix} * & * \\ 0 & * \end{pmatrix}$, where the size of the zero matrix in the lower left corner is at least $\left(\sum_{k+1 \le i \le l} p_i\right) \times \left(1 + \sum_{k+1 \le i \le l} p_i\right)$. Therefore, A is not invertible, which contradicts with $\mathfrak{o} \in \mathcal{O}^{\times}$. This establishes (1) \Rightarrow (2).

Finally, we prove the direction (3) \Rightarrow (1). We assume (3). Suppose that $\mathfrak{o} \notin \mathcal{O}^{\times}$. Let $P(\lambda) :=$ $\operatorname{Prd}_{AB}(\lambda)$, where $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix}$ is any element in \mathfrak{o} . By [56, Proposition 5], $P(\lambda) = \lambda^d R(\lambda)$, where $R(\lambda) = \operatorname{Prd}_C(\lambda)$ for some $C \in GL_{n-1}(D)$. Let Q be the relative standard parabolic subgroup of G containing elements of the form $\begin{pmatrix} * & * \\ 0 & * \end{pmatrix}^{\times}$, where the size of the zero matrix in the lower left corner is $1 \times (n-1)$.

Then
$$\begin{pmatrix} 0 & 0 & 0 & 0 \\ C & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \end{pmatrix} \in \mathfrak{o} \cap \mathfrak{m}_Q(F)$$
, which contradicts with (3). This shows (3) \Rightarrow (1).

LEMMA 5.2. Assume that p = q = n. Let Q be a relatively standard parabolic subgroup of G. For all $\varpi^{\vee} \in \widehat{\Delta}_Q^{\vee}$, we have $(2\rho_{Q,+} - 2\rho_{Q_H})(\varpi^{\vee}) \ge 0$. Moreover, $2\rho_{Q,+} - 2\rho_{Q_H}$ viewed as an element of $(\mathfrak{a}_Q^G)^*$ is zero if and only if Q is ω -stable.

PROOF. We use the notation in Section 3.4. Put $e_i^* \in \mathfrak{a}_0^*$ (resp. $f_i^* \in \mathfrak{a}_0^*$) to be the character of the action of A_0 on e_i (resp. f_i). Write $e_i^{\vee} \in \mathfrak{a}_0$ (resp. $f_i^{\vee} \in \mathfrak{a}_0$) to be the dual basis, i.e., $e_i^*(e_i^{\vee}) = \delta_{ij}$

(resp. $f_i^*(f_j^{\vee}) = \delta_{ij}$) for $1 \leq i, j \leq n$. A basis of \mathfrak{a}_Q is given by $h_i^{\vee} := e_{p_1+\cdots+p_{i-1}+1}^{\vee} + \cdots + e_{p_1+\cdots+p_i}^{\vee} + f_{q_1+\cdots+q_i}^{\vee} + 1 \leq i \leq l$. Write $h_i^* \in (\mathfrak{a}_Q)^*$ to be the dual basis. Denote

$$\varpi_k^{\vee} := \frac{\sum_{i=k+1}^l (p_i + q_i)}{2n} (h_1^{\vee} + \dots + h_k^{\vee}) - \frac{\sum_{i=1}^k (p_i + q_i)}{2n} (h_{k+1}^{\vee} + \dots + h_l^{\vee}).$$

Recall that

$$(\widehat{\Delta}_Q^G)^{\vee} = \{ \varpi_k^{\vee} | 1 \le k \le l-1 \}$$

is a basis of \mathfrak{a}_Q^G . We can also see that

$$2\rho_{Q,+}\big|_{\mathfrak{a}_Q} = \dim_F(D) \sum_{1 \le i < j \le l} (p_i q_j + q_i p_j) (h_i^* - h_j^*)$$

and that

$$2\rho_{Q_H}\Big|_{\mathfrak{a}_Q} = \dim_F(D) \sum_{1 \le i < j \le l} (p_i p_j + q_i q_j)(h_i^* - h_j^*),$$

 \mathbf{SO}

$$(2\rho_{Q,+} - 2\rho_{Q_H})\big|_{\mathfrak{a}_Q} = \dim_F(D) \sum_{1 \le i < j \le l} (p_i - q_i)(q_j - p_j)(h_i^* - h_j^*)$$

Since $\sum_{i=1}^{l} p_i = \sum_{i=1}^{l} q_i = n$, we have

$$(h_i^* - h_j^*)(\varpi_k^{\vee}) = \begin{cases} 0, & \text{if } k+1 \le i < j \le l \text{ or } 1 \le i < j \le k; \\ 1, & \text{if } 1 \le i \le k \text{ and } k+1 \le j \le l. \end{cases}$$

Then

$$(2\rho_{Q,+} - 2\rho_{Q_H})(\varpi_k^{\vee}) = \dim_F(D) \sum_{\substack{1 \le i \le k \\ k+1 \le j \le l}} (p_i - q_i)(q_j - p_j)$$
$$= \dim_F(D) \left(\sum_{1 \le i \le k} p_i - \sum_{1 \le i \le k} q_i \right) \left(\sum_{k+1 \le j \le l} q_j - \sum_{k+1 \le j \le l} p_j \right)$$
$$= \dim_F(D) \left(\sum_{1 \le i \le k} p_i - \sum_{1 \le i \le k} q_i \right)^2 \ge 0.$$

It is clear that $(2\rho_{Q,+} - 2\rho_{Q_H})(\varpi_k^{\vee}) = 0$ for all $1 \le k \le l-1$ if and only if $p_i = q_i$ for all $1 \le i \le l$. \Box

5.3. Exponential polynomials. Let $T_1, T_2 \in \mathfrak{a}_0$. Following [5, §2], define the function $\Gamma_P(T_1, T_2)$ inductively on dim (A_P/A_G) by setting

$$\hat{\tau}_{P}^{G}(T_{1} - T_{2}) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_{Q}/A_{G})} \hat{\tau}_{P}^{Q}(T_{1}) \Gamma_{Q}(T_{1}, T_{2})$$

for any relatively standard parabolic subgroup P of G. This definition can be explicitly given by [5, (2.1) in p. 13] and only depends on the projections of T_1, T_2 onto \mathfrak{a}_P^G . For $T = (t_1, ..., t_{p+q}) \in \mathfrak{a}_0$, we denote $\Sigma_1(T) := t_1 + ... + t_p$. If we use the notation in Section 3.4 and put $e_i^* \in \mathfrak{a}_0^*$ (resp. $f_i^* \in \mathfrak{a}_0^*$) to be the character of the action of A_0 on e_i (resp. f_i), it is equivalent to say that $\Sigma_1 = \sum_{1 \le i \le p} e_i^*$. For $T_2 \in \mathfrak{a}_Q$ and $s \in \mathbb{C}$, write

(5.3.1)
$$p_{Q,s}(T_2) := \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1)(T_1)} \Gamma_Q(T_1, T_2) dT_1.$$

When p = q = n, s = 0 and Q is ω -stable, it is reduced to

$$p_{Q,0}(T_2) = \int_{\mathfrak{a}_Q^G} \Gamma_Q(T_1, T_2) dT_1$$

by Lemma 5.2.

For $Q \subseteq R$ a pair of relatively standard parabolic subgroups of G, denote by $\mathbb{Z}(\widehat{\Delta}_Q^R)^{\vee}$ the lattice generated by $(\widehat{\Delta}_Q^R)^{\vee}$ in \mathfrak{a}_Q^R and by $\mathbb{Z}(\Delta_R^G)^{\vee}$ the lattice generated by $(\Delta_R^G)^{\vee}$ in \mathfrak{a}_R^G . Following [5, §2], for $\lambda \in \mathfrak{a}_{Q,\mathbb{C}}^* := \mathfrak{a}_Q^* \otimes_{\mathbb{R}} \mathbb{C}$, define

$$\widehat{\theta}^R_Q(\lambda) := \operatorname{vol}(\mathfrak{a}^R_Q / \mathbb{Z}(\widehat{\Delta}^R_Q)^{\vee})^{-1} \prod_{\varpi^{\vee} \in (\widehat{\Delta}^R_Q)^{\vee}} \lambda(\varpi^{\vee})$$

and

$$\theta_R^G(\lambda) := \operatorname{vol}(\mathfrak{a}_R^G/\mathbb{Z}(\Delta_R^G)^{\vee})^{-1} \prod_{\alpha^{\vee} \in (\Delta_R^G)^{\vee}} \lambda(\alpha^{\vee}).$$

PROPOSITION 5.3. Let Q be a relatively standard parabolic subgroup of G, $T_2 \in \mathfrak{a}_Q$ and $s \in \mathbb{C}$. The function $T_1 \mapsto \Gamma_Q(T_1, T_2)$ is compactly supported on \mathfrak{a}_Q^G . Moreover, the function $T_2 \mapsto p_{Q,s}(T_2)$ is an exponential polynomial in T_2 ; more precisely, there exists a polynomial $p_{Q,R,s}$ (not necessarily unique) on \mathfrak{a}_R^G of degree $\leq \dim(A_Q/A_G)$ for each relatively standard parabolic subgroup R containing Q such that

$$p_{Q,s}(T_2) = \sum_{\{R:Q\subseteq R\}} e^{(2\rho_{R,+}-2\rho_{R_H}+s\Sigma_1)(T_{2,R}^G)} p_{Q,R,s}(T_{2,R}^G),$$

where we write $T_{2,R}^G$ for the projection of $T_2 \in \mathfrak{a}_Q$ in \mathfrak{a}_R^G via the decomposition $\mathfrak{a}_Q = \mathfrak{a}_Q^R \oplus \mathfrak{a}_R^G \oplus \mathfrak{a}_G$. When p = q = n and s = 0, the purely polynomial term of $p_{Q,0}(T_2)$ is given by

$$\sum_{R:Q\subseteq R,\omega R\omega^{-1}=R\}} p_{Q,R,0}(T_{2,R}^G)$$

which is a homogeneous polynomial in T_2 of degree $\dim(A_{\overline{Q}^{\omega-st}}/A_G)$; in particular, if Q is ω -stable, then $p_{Q,0}(T_2)$ is a homogeneous polynomial in T_2 of degree $\dim(A_Q/A_G)$.

PROOF. The first statement is [5, Lemmas 2.1]. First let us prove the second one. From [5, Lemma 2.2], we know that the integral

$$\int_{\mathfrak{a}_Q^G} e^{\lambda(T_1)} \Gamma_Q(T_1, T_2) dT_1$$

is an entire function in $\lambda \in \mathfrak{a}_{Q,\mathbb{C}}^*$, and its value is given by

$$\sum_{\{R:Q\subseteq R\}} (-1)^{\dim(A_Q/A_R)} e^{\lambda(T_{2,R}^G)} \widehat{\theta}_Q^R(\lambda)^{-1} \theta_R^G(\lambda)^{-1}$$

when the latter expression makes sense.

Fix $\varepsilon \in \mathfrak{a}_{Q,\mathbb{C}}^*$ such that $\widehat{\theta}_Q^R(\varepsilon) \neq 0$ and $\theta_R^G(\varepsilon) \neq 0$ for all relatively standard parabolic subgroups R containing Q. Then for $t \in \mathbb{R}^{\times}$ whose absolute value is small enough, we also have $\widehat{\theta}_Q^R(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon) \neq 0$ and $\theta_R^G(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon) \neq 0$ for all relatively standard parabolic subgroups R containing Q. Let $\lambda = 2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon$ in the formula above, and we obtain

$$p_Q(T_2) = \lim_{t \to 0} \sum_{\{R: Q \subseteq R\}} (-1)^{\dim(A_Q/A_R)} e^{(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon)(T_{2,R}^G)} \widehat{\theta}_Q^R (2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon)^{-1} \\ \cdot \theta_R^G (2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1 + t\varepsilon)^{-1}.$$

Since the restriction of $2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1$ to \mathfrak{a}_R equals $2\rho_{R,+} - 2\rho_{R_H} + s\Sigma_1$, we get $e^{(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1)(T^G_{2,R})} = e^{(2\rho_{R,+} - 2\rho_{R_H} + s\Sigma_1)(T^G_{2,R})}.$

We can put $p_{Q,R,s}(T_{2,R}^G)$ to be the constant term of the Laurent series development around t = 0 of

$$t\mapsto (-1)^{\dim(A_Q/A_R)}e^{(t\varepsilon)(T^G_{2,R})}\widehat{\theta}^R_Q(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1+t\varepsilon)^{-1}\theta^G_R(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1+t\varepsilon)^{-1}.$$

Then $p_{Q,R,s}(T_{2,R}^G)$ is a polynomial in $T_{2,R}^G$ of degree $\leq \dim(A_Q/A_G)$. Hence we prove the existence in the second statement.

Now let p = q = n and s = 0. From Lemma 5.2, we know that the purely polynomial term of $p_{Q,0}$ is given by

$$\sum_{\{R:Q\subseteq R,\omega R\omega^{-1}=R\}} p_{Q,R,0}(T_{2,R}^G).$$

Next we compute the degree of $p_{Q,R,0}$ that we chose above for each ω -stable parabolic subgroup R containing Q. Denote

$$N_1 := \sharp \{ \varpi^{\vee} \in (\widehat{\Delta}_Q^R)^{\vee} : (2\rho_{Q,+} - 2\rho_{Q_H})(\varpi^{\vee}) = 0 \}$$

and

$$\mathbf{N}_{2} := \sharp \{ \alpha^{\vee} \in (\Delta_{R}^{G})^{\vee} : (2\rho_{Q,+} - 2\rho_{Q_{H}})(\alpha^{\vee}) = 0 \},\$$

where \sharp means the cardinality of a finite set. Then

1

$$\deg(p_{Q,R,0}) = N_1 + N_2.$$

Recall that both of $(\widehat{\Delta}_R^G)^{\vee}$ and $(\Delta_R^G)^{\vee}$ are bases of \mathfrak{a}_R^G . Since R is relatively standard and ω -stable, by Lemma 5.2, we have

$$N_2 = \dim(A_R/A_G).$$

Keep the notation as in the proof of Lemma 5.2 for Q. Since R is relatively standard and ω -stable, by Lemma 5.2, we may suppose that R is the stabiliser in G of the flag

 $0 \subsetneq \langle e_1, \dots, e_{r_1}, f_1, \dots, f_{r_1} \rangle_D \subsetneq \langle e_1, \dots, e_{r_1+r_2}, f_1, \dots, f_{r_1+r_2} \rangle_D \subsetneq \dots \subsetneq \langle e_1, \dots, e_{r_1+\dots+r_{l'}}, f_1, \dots, f_{r_1+\dots+r_{l'}} \rangle_D$. The fact that $Q \subseteq R$ tells us that both of the partitions (p_1, \dots, p_l) and (q_1, \dots, q_l) are refinements of the partition $(r_1, \dots, r_{l'})$ of n, and that every r_i is divided into the same number of segments in these two refinements. Then

$$\begin{aligned} (\widehat{\Delta}_Q^R)^{\vee} &= \left\{ \text{projection of } \varpi_k^{\vee} \in (\widehat{\Delta}_Q^G)^{\vee} \text{ to } \mathfrak{a}_Q^R \middle| 1 \le k \le l-1, \sum_{i=1}^k (p_i + q_i) \neq \sum_{i=1}^j 2r_i \forall 1 \le j \le l'-1 \right\} \\ &= \left\{ \text{projection of } \varpi_k^{\vee} \in (\widehat{\Delta}_Q^G)^{\vee} \text{ to } \mathfrak{a}_Q^R \middle| 1 \le k \le l-1, \nexists 1 \le j \le l'-1s.t. \sum_{i=1}^k p_i = \sum_{i=1}^k q_i = \sum_{i=1}^j r_i \right\}.\end{aligned}$$

Because the restriction of $2\rho_{Q,+} - 2\rho_{Q_H}$ to \mathfrak{a}_R equals $2\rho_{R,+} - 2\rho_{R_H}$ and R is relatively standard and ω -stable, by Lemma 5.2, we do not need the projection, i.e.,

$$(2\rho_{Q,+} - 2\rho_{Q_H}) \left(\text{projection of } \varpi_k^{\vee} \in (\widehat{\Delta}_Q^G)^{\vee} \text{ to } \mathfrak{a}_Q^R \right) = (2\rho_{Q,+} - 2\rho_{Q_H}) \left(\varpi_k^{\vee} \in (\widehat{\Delta}_Q^G)^{\vee} \right).$$

From the proof of Lemma 5.2, for any $1 \leq k \leq l$, we have $(2\rho_{Q,+} - 2\rho_{Q_H})(\varpi_k^{\vee}) = 0$ if and only if $\sum_{i=1}^k p_i = \sum_{i=1}^k q_i$. We can also see that $\overline{Q}^{\omega\text{-st}}$ is the ω -stable parabolic subgroup R containing Q with maximal $l' := \dim(A_R)$. To sum up, we have

$$N_1 = \dim(A_{\overline{O}^{\,\omega\text{-st}}}/A_R).$$

Hence for each ω -stable parabolic subgroup R containing Q,

$$\deg(p_{Q,R,0}) = N_1 + N_2 = \dim(A_{\overline{Q}^{\omega-\mathrm{st}}}/A_R) + \dim(A_R/A_G) = \dim(A_{\overline{Q}^{\omega-\mathrm{st}}}/A_G).$$

The assertion about the particular case where Q is ω -stable is [5, Lemma 2.2] combined with Lemma 5.2; it can also be read from the results above that we have proved.

5.4. Quantitive behaviour in T. For a relatively standard parabolic subgroup Q of G, let $\{s_i^Q\}_{1 \le i \le l} \in \mathbb{Z}^l$ be the explicit constants determined by

(5.4.1)
$$\forall x \in M_{Q_H}(\mathbb{A}) \cap M_Q(\mathbb{A})^1, \prod_{1 \le i \le l} |\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}^{s_i^Q} = e^{(2\rho_{Q,+} - 2\rho_{Q_H})(H_{Q_H}(x))}$$

where we write $x = (x_1, ..., x_l) \in GL_{p_1+q_1,D}(\mathbb{A}) \times \cdots \times GL_{p_l+q_l,D}(\mathbb{A})$ and $x_i = (x_{i,1}, x_{i,2}) \in GL_{p_i,D}(\mathbb{A}) \times GL_{q_i,D}(\mathbb{A})$. If $p_iq_i = 0$ for some $1 \leq i \leq l$, we shall take $|\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}^{s_i^Q} = 1$ and $s_i^Q = 0$ by convention. Then such constants are unique.

PROPOSITION 5.4. Let Q be a relatively standard parabolic subgroup of G. If $p_i q_i \neq 0$ for some $1 \leq i \leq l$, then

$$s_i^Q = 2d\left(\sum_{k < i} (p_k - q_k) + \sum_{k > i} (q_k - p_k)\right).$$

When p = q = n, if Q is ω -stable, then $s_i^Q = 0$ for all $1 \le i \le l$.

PROOF. Assume that $p_i q_i \neq 0$ for some $1 \leq i \leq l$. Let $x \in M_{Q_H}(\mathbb{A})$. We have

- (1) the contribution of $x_{i,1}$ to $e^{2\rho_{Q,+}(H_{Q_H}(x))}$ is the $d\left(\sum_{k>i} q_k \sum_{k< i} q_k\right)$ -th power of $|\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}$;
- (2) the contribution of $x_{i,1}$ to $e^{2\rho_{Q_H}(H_{Q_H}(x))}$ is the $d\left(\sum_{k>i} p_k \sum_{k< i} p_k\right)$ -th power of $|\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}$;
- (3) the contribution of $x_{i,2}$ to $e^{2\rho_{Q,+}(H_{Q_H}(x))}$ is the $d\left(\sum_{k>i} p_k \sum_{k<i} p_k\right)$ -th power of $|\operatorname{Nrd}(x_{i,2})|_{\mathbb{A}}$;
- (4) the contribution of $x_{i,2}$ to $e^{2\rho_{Q_H}(H_{Q_H}(x))}$ is the $d\left(\sum_{k>i} q_k \sum_{k< i} q_k\right)$ -th power of $|\operatorname{Nrd}(x_{i,2})|_{\mathbb{A}}$.

In sum, the contribution of x_i to $e^{(2\rho_{Q,+}-2\rho_{Q_H})(H_{Q_H}(x))}$ is the product of the $d\left(\sum_{k< i}(p_k-q_k)+\sum_{k>i}(q_k-p_k)\right)$ -th power of $|\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}$ and the $d\left(\sum_{k< i}(q_k-p_k)+\sum_{k>i}(p_k-q_k)\right)$ -th power of $|\operatorname{Nrd}(x_{i,2})|_{\mathbb{A}}$.

Now let $x \in M_{Q_H}(\mathbb{A}) \cap M_Q(\mathbb{A})^1$. Then $|\operatorname{Nrd}(x_{i,1})\operatorname{Nrd}(x_{i,2})|_{\mathbb{A}} = |\operatorname{Nrd}(x_i)|_{\mathbb{A}} = 1$. Therefore, the contribution of x_i to $e^{(2\rho_{Q,+}-2\rho_{Q_H})(H_{Q_H}(x))}$ is the $2d\Big(\sum_{k< i} (p_k-q_k) + \sum_{k>i} (q_k-p_k)\Big)$ -th power of $|\operatorname{Nrd}(x_{i,1})|_{\mathbb{A}}$.

We have proved the first statement.

The second statement is nothing but a special case of the first one, since we have $p_k = q_k$ for $1 \le k \le l$ in this case.

THEOREM 5.5. Let T' be sufficiently regular, $\mathfrak{o} \in \mathcal{O}$ and $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$. Then for all sufficiently regular T and $s \in \mathbb{C}$, we have

$$J^{G,T}_{\mathfrak{o}}(\eta,s,f) = \sum_{\{Q:\tilde{P}_0 \subseteq Q\}} p_{Q,s}(T_Q - T'_Q) e^{(2\rho_{Q,+} - 2\rho_{Q_H} + s\Sigma_1)((T')^G_Q)} J^{Q,T'}_{\mathfrak{o}}(\eta,\{s^Q_i + s\}, f^\eta_Q),$$

where we write $(T')_Q^G$ for the projection of $T'_Q \in \mathfrak{a}_Q$ in \mathfrak{a}_Q^G via the decomposition $\mathfrak{a}_Q = \mathfrak{a}_Q^G \oplus \mathfrak{a}_G$, the distributions $J_{\mathfrak{o}}^{G,T}$ and $J_{\mathfrak{o}}^{Q,T'}$ are defined by the formulae (5.0.1) and (5.1.2) respectively, and f_Q^{η} and $p_{Q,s}$ are defined by the formulae (5.1.3) and (5.3.1) respectively.

COROLLARY 5.6. Let $\mathbf{o} \in \mathcal{O}$, $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $s \in \mathbb{C}$. Then $J_{\mathbf{o}}^{G,T}(\eta, s, f)$ and $J^{G,T}(\eta, s, f)$ are exponential polynomials in T for sufficiently regular T, so we can extend them to all $T \in \mathfrak{a}_0$. When p = q = n and s = 0, their purely polynomial terms have degree $\leq n - 1$; in particular, if $\mathbf{o} \in \mathcal{O}^{\times}$ (e.g., $\mathbf{o} \in \mathcal{O}_{rs}$), $J_{\mathbf{o}}^{G,T}(\eta, 0, f)$ is a polynomial in T of degree $\leq n - 1$ for sufficiently regular T.

PROOF OF COROLLARY 5.6. It results from Theorem 5.5, Propositions 5.3 and 5.1.

REMARK 5.7. We may extend our result to the product form in Section 5.1 by similar argument. Let Q be a relatively standard parabolic subgroup of G. Let $\mathfrak{o} \in \mathcal{O}$, $f' \in \mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}))$ and $\{s_i\}_{1 \leq i \leq l} \in \mathbb{C}^l$. For sufficiently regular $T \in \mathfrak{a}_{P_0}^+$, $J_{\mathfrak{o}}^{Q,T}(\eta, \{s_i\}, f')$ and $J^{Q,T}(\eta, \{s_i\}, f')$ are exponential polynomials in T independent of T_Q , so we can extend them to all $T \in \mathfrak{a}_0$.

PROOF OF THEOREM 5.5. Let P be a relatively standard parabolic subgroup of $G, \delta \in P_H(F) \setminus H(F)$ and $x \in H(\mathbb{A}) \cap G(\mathbb{A})^1$. Substituting $T_1 = H_P(\delta x) - T'_P$ and $T_2 = T_P - T'_P$ in the definition of $\Gamma_P(T_1, T_2)$, we get

$$\hat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_{Q}/A_{G})} \hat{\tau}_{P}^{Q}(H_{P}(\delta x) - T_{P}') \Gamma_{Q}(H_{P}(\delta x) - T_{P}', T_{P} - T_{P}').$$

Then

$$\begin{split} J^{G,T}_{\mathfrak{o}}(\eta,s,f) &= \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \left(\sum_{\{P:\tilde{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta\in P_{H}(F)\backslash H(F)} \widehat{\tau}^{G}_{P}(H_{P}(\delta x) - T_{P}) \cdot k_{f,P,\mathfrak{o}}(\delta x) \right) \\ &\quad \cdot \eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_{1})|^{s}_{\mathbb{A}} dx \\ &= \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \sum_{\{P:\tilde{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta\in P_{H}(F)\backslash H(F)} \\ &\quad \left(\sum_{\{Q:P\subseteq Q\}} (-1)^{\dim(A_{Q}/A_{G})} \widehat{\tau}^{Q}_{P}(H_{P}(\delta x) - T'_{P})\Gamma_{Q}(H_{P}(\delta x) - T'_{P}, T_{P} - T'_{P}) \right) k_{f,P,\mathfrak{o}}(\delta x) \\ &\quad \cdot \eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_{1})|^{s}_{\mathbb{A}} dx. \end{split}$$

Exchanging the order of two sums over P and Q, and decomposing the sum over $P_H(F) \setminus H(F)$ into two sums over $P_H(F) \setminus Q_H(F)$ and $Q_H(F) \setminus H(F)$, we have

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f) = \sum_{\{Q: \tilde{P}_0 \subseteq Q\}} \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} \sum_{\{P: \tilde{P}_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in P_H(F) \setminus Q_H(F)} \sum_{\delta' \in Q_H(F) \setminus H(F)} \tilde{\tau}_P^Q(H_P(\delta\delta'x) - T'_P) \Gamma_Q(H_P(\delta\delta'x) - T'_P, T_P - T'_P) k_{f,P,\mathfrak{o}}(\delta\delta'x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx.$$

Combining the integral over $H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$ and the sum over $Q_H(F) \setminus H(F)$ into the integral over $Q_H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$, and using the fact that

$$P_H(F) \setminus Q_H(F) \simeq (P_H(F) \cap M_{Q_H}(F)) \setminus M_{Q_H}(F),$$

we obtain

$$J_{\mathbf{o}}^{G,T}(\eta, s, f) = \sum_{\{Q: \tilde{P}_0 \subseteq Q\}} \int_{Q_H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} \sum_{\{P: \tilde{P}_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P_H(F) \cap M_{Q_H}(F)) \setminus M_{Q_H}(F)} \hat{\tau}_P^Q(H_P(\delta x) - T'_P) \Gamma_Q(H_P(\delta x) - T'_P, T_P - T'_P) k_{f,P,\mathbf{o}}(\delta x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx.$$

By the Iwasawa decomposition and our choice of measures, we have

$$J_{\mathbf{o}}^{G,T}(\eta, s, f) = \sum_{\{Q: \tilde{P}_{0} \subseteq Q\}} \int_{K_{H}} \int_{M_{Q_{H}}(F) \setminus M_{Q_{H}}(\mathbb{A}) \cap M_{Q}(\mathbb{A})^{1}} \int_{A_{Q}^{G,\infty}} \int_{N_{Q_{H}}(F) \setminus N_{Q_{H}}(\mathbb{A})} \sum_{\substack{\{P: \tilde{P}_{0} \subseteq P \subseteq Q\}}} (-1)^{\dim(A_{P}/A_{Q})} \\ \sum_{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)} \widehat{\tau}_{P}^{Q} (H_{P}(\delta namk) - T'_{P}) \Gamma_{Q} (H_{P}(\delta namk) - T'_{P}, T_{P} - T'_{P}) \\ \cdot k_{f,P,\mathbf{o}}(\delta namk) \eta(\operatorname{Nrd}(mk)) |\operatorname{Nrd}(a_{1}m_{1})|_{\mathbb{A}}^{s} e^{-2\rho_{Q_{H}}(H_{Q_{H}}(am))} dn dadm dk.$$

Notice that

$$\widehat{\tau}_P^Q(H_P(\delta namk) - T'_P) = \widehat{\tau}_P^Q(H_P(\delta m) + H_P(a) - T'_P) = \widehat{\tau}_P^Q(H_P(\delta m) - T'_P),$$

and that

 $\Gamma_Q(H_P(\delta namk) - T'_P, T_P - T'_P) = \Gamma_Q(H_Q(\delta namk) - T'_Q, T_Q - T'_Q) = \Gamma_Q(H_Q(a) - T'_Q, T_Q - T'_Q).$ In addition, by change of variables, we see that

$$\begin{split} k_{f,P,\mathfrak{o}}(\delta namk) &= \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P} \cap \mathfrak{s})(\mathbb{A})} f((\delta namk)^{-1}(X+U)\delta namk) dU \\ &= \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P} \cap \mathfrak{s})(\mathbb{A})} f((\delta a^{-1}namk)^{-1}(X+a^{-1}Ua)\delta a^{-1}namk) dU \\ &= \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P} \cap \mathfrak{s})(\mathbb{A})} f((\delta a^{-1}namk)^{-1}(X+U)\delta a^{-1}namk) e^{2\rho_{Q,+}(H_{Q}(a))} dU \\ &= e^{2\rho_{Q,+}(H_{Q}(a))} k_{f,P,\mathfrak{o}}(\delta a^{-1}namk). \end{split}$$

Since $\delta a^{-1}na\delta^{-1} \in N_{Q_H}(\mathbb{A}) \subseteq N_{P_H}(\mathbb{A})$ and $k_{f,P,\mathfrak{o}}$ is left invariant by $N_{P_H}(\mathbb{A})$, we deduce that

$$k_{f,P,\mathfrak{o}}(\delta namk) = e^{2\rho_{Q,+}(H_Q(a))} k_{f,P,\mathfrak{o}}(\delta mk).$$

In sum, the integrand in $J_{\mathfrak{o}}^{G,T}(\eta, s, f)$ is independent of $n \in N_{Q_H}(F) \setminus N_{Q_H}(\mathbb{A})$. We can choose the Haar measure such that $\operatorname{vol}(N_{Q_H}(F) \setminus N_{Q_H}(\mathbb{A})) = 1$. Then

$$J_{\mathbf{0}}^{G,T}(\eta, s, f) = \sum_{\{Q: \tilde{P}_{0} \subseteq Q\}} \left(\int_{A_{Q}^{G,\infty}} |\operatorname{Nrd}(a_{1})|_{\mathbb{A}}^{s} e^{(2\rho_{Q,+}-2\rho_{Q_{H}})(H_{Q}(a))} \Gamma_{Q}(H_{Q}(a) - T_{Q}', T_{Q} - T_{Q}') da \right)$$

$$\int_{M_{Q_{H}}(F) \setminus M_{Q_{H}}(\mathbb{A}) \cap M_{Q}(\mathbb{A})^{1}} \sum_{\{P: \tilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)}$$

$$\hat{\tau}_{P}^{Q}(H_{P}(\delta m) - T_{P}') \left(\int_{K_{H}} k_{f,P,\mathfrak{o}}(\delta mk) \eta(\operatorname{Nrd}(k)) dk \right) \eta(\operatorname{Nrd}(m)) |\operatorname{Nrd}(m_{1})|_{\mathbb{A}}^{s} e^{-2\rho_{Q_{H}}(H_{Q_{H}}(m))} dm$$

By the definition of the Haar measure on $A_Q^{G,\infty}$, we have

$$\begin{split} &\int_{A_Q^{G,\infty}} |\mathrm{Nrd}(a_1)|_{\mathbb{A}}^s e^{(2\rho_{Q,+}-2\rho_{Q_H})(H_Q(a))} \Gamma_Q(H_Q(a) - T'_Q, T_Q - T'_Q) da \\ &:= \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_1)} \Gamma_Q(T_1 - T'_Q, T_Q - T'_Q) dT_1 \\ &= e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)((T')_Q^G)} \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_1)} \Gamma_Q(T_1, T_Q - T'_Q) dT_1 \\ &= e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)((T')_Q^G)} p_{Q,s}(T_Q - T'_Q). \end{split}$$

Since $\mathfrak{n}_P = \mathfrak{n}_P^Q \oplus \mathfrak{n}_Q$, by change of variables, we see that

$$k_{f,P,\mathfrak{o}}(\delta mk) = \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P}^{Q} \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_{Q} \cap \mathfrak{s})(\mathbb{A})} f((\delta mk)^{-1}(X+U+V)\delta mk)dV$$
$$= e^{2\rho_{Q,+}(H_{Q_{H}}(m))} \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P}^{Q} \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_{Q} \cap \mathfrak{s})(\mathbb{A})} f(k^{-1}((\delta m)^{-1}(X+U)\delta m+V)k)dV,$$

so we can write

$$\begin{split} \int_{K_{H}} k_{f,P,\mathfrak{o}}(\delta m k) \eta(\mathrm{Nrd}(k)) dk &= e^{2\rho_{Q,+}(H_{Q_{H}}(m))} \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P}^{Q} \cap \mathfrak{s})(\mathbb{A})} f_{Q}^{\eta}((\delta m)^{-1}(X+U)\delta m) dU \\ &= e^{2\rho_{Q,+}(H_{Q_{H}}(m))} \sum_{j=1}^{t} k_{f_{Q}^{\eta},P \cap M_{Q},\mathfrak{o}_{j}}^{M_{Q}}(\delta m) \end{split}$$

by (5.1.1). Now we can draw our conclusion by noting that

$$\begin{split} J_{\mathbf{0}}^{Q,T'}(\eta,\{s_{i}^{Q}+s\},f_{Q}^{\eta}) &= \sum_{j=1}^{t} \int_{M_{Q_{H}}(F)\setminus M_{Q_{H}}(\mathbb{A})\cap M_{Q}(\mathbb{A})^{1}} \sum_{\{P:\tilde{P}_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P\cap M_{Q}}/A_{M_{Q}})} \\ &\sum_{\delta\in((P\cap M_{Q})(F)\cap M_{Q_{H}}(F))\setminus M_{Q_{H}}(F)} \hat{\tau}_{P\cap M_{Q}}^{M_{Q}}(H_{P\cap M_{Q}}(\delta m) - (\varsigma_{Q}T')_{P\cap M_{Q}}) \\ &\cdot k_{f_{Q}^{\eta},P\cap M_{Q},\mathbf{0}_{j}}(\delta m)\eta(\operatorname{Nrd}(m))|\operatorname{Nrd}(m_{1})|_{\mathbb{A}}^{s} e^{(2\rho_{Q,+}-2\rho_{Q_{H}})(H_{Q_{H}}(m))}dm \\ &= \int_{M_{Q_{H}}(F)\setminus M_{Q_{H}}(\mathbb{A})\cap M_{Q}(\mathbb{A})^{1}} \sum_{\{P:\tilde{P}_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta\in(P_{H}(F)\cap M_{Q_{H}}(F))\setminus M_{Q_{H}}(F)} \\ &\hat{\tau}_{P}^{Q}(H_{P}(\delta m) - T'_{P}) \left(\sum_{j=1}^{t} k_{f_{Q}^{\eta},P\cap M_{Q},\mathbf{0}_{j}}(\delta m)\right) \eta(\operatorname{Nrd}(m))|\operatorname{Nrd}(m_{1})|_{\mathbb{A}}^{s} \\ &\cdot e^{(2\rho_{Q,+}-2\rho_{Q_{H}})(H_{Q_{H}}(m))}dm. \end{split}$$

5.5. Independence of constant terms. Let $J^G_{\mathfrak{o}}(\eta, s, f)$ and $J^G(\eta, s, f)$ be the constant terms of

5.5. Independence of constant terms. Let $J_{\mathfrak{o}}^{-}(\eta, s, f)$ and $J^{-}(\eta, s, f)$ be the constant terms of $J_{\mathfrak{o}}^{G,T}(\eta, s, f)$ and $J^{G,T}(\eta, s, f)$ and $J^{G,T}(\eta, s, f)$ respectively. We fix a common minimal Levi subgroup M_0 of H and G. Firstly, the distributions $J_{\mathfrak{o}}^{G}(\eta, s, f)$ and $J^{G}(\eta, s, f)$ are independent of the choice of the relatively standard minimal parabolic subgroup P_0 of G at the very beginning of last section. In fact, let P'_0 be another relatively standard minimal parabolic subgroup of G and $\sigma \in \Omega^G$ such that $P'_0 = \sigma P_0$. Denote by $J_{P'_0,\mathfrak{o}}^{G,T}(\eta, s, f)$ and $J_{P'_0,\mathfrak{o}}^{G}(\eta, s, f)$ the distributions obtained starting from P'_0 . Then if $T \in \mathfrak{a}_{P'_0}$, we have $J_{P'_0,\mathfrak{o}}^{G,T}(\eta, s, f) = J_{\mathfrak{o}}^{G,\sigma^{-1}T}(\eta, s, f)$, so $J_{P'_0,\mathfrak{o}}^G(\eta, s, f) = J_{\mathfrak{o}}^G(\eta, s, f)$.

Secondly, the distributions $J^G_{\mathfrak{o}}(\eta, s, f)$ and $J^G(\eta, s, f)$ are independent of the choice of the minimal parabolic subgroup \widetilde{P}_0 of H. In fact, let \widetilde{P}'_0 be another minimal parabolic subgroup of H and $\sigma \in \Omega^H$ such that $\widetilde{P}'_0 = \sigma^{-1}\widetilde{P}_0$. Put $P'_0 := \sigma^{-1}P_0$. Denote by $J^{G,T}_{\widetilde{P}'_0,\mathfrak{o}}(\eta, s, f)$ and $J^G_{\widetilde{P}'_0,\mathfrak{o}}(\eta, s, f)$ the distributions obtained starting from \widetilde{P}'_0 and P'_0 . We can apply the argument of [13, Proposition 4.6] after some minor modifications here to prove that $J^{G,T}_{\mathfrak{o}}(\eta, s, f) = J^{G,\sigma^{-1}T}_{\widetilde{P}'_0,\mathfrak{o}}(\eta, s, f)$, so $J^G_{\mathfrak{o}}(\eta, s, f) = J^G_{\widetilde{P}'_0,\mathfrak{o}}(\eta, s, f)$.

6. Non-equivariance

Let Q be a relatively standard parabolic subgroup of $G, s \in \mathbb{R}$ and $y \in H(\mathbb{A}) \cap G(\mathbb{A})^1$. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, define $f_{Q,s,y}^{\eta} \in \mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}))$ by

$$(6.0.1) \quad \forall X \in (\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}), f_{Q,s,y}^{\eta}(X) := \int_{K_H} \int_{(\mathfrak{n}_Q \cap \mathfrak{s})(\mathbb{A})} f(k^{-1}(X+V)k)\eta(\operatorname{Nrd}(k))p_{Q,s}(-H_Q(ky))dVdk,$$

where $p_{Q,s}$ is defined by the formula (5.3.1).

PROPOSITION 6.1. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $y \in H(\mathbb{A}) \cap G(\mathbb{A})^1$, we denote $f^y(x) := f(yxy^{-1})$. Then for all sufficiently regular $T, \mathfrak{o} \in \mathcal{O}$ and $s \in \mathbb{R}$, we have

$$J_{\mathbf{o}}^{G,T}(\eta, s, f^{y}) = \eta(\operatorname{Nrd}(y))|\operatorname{Nrd}(y_{1})|_{\mathbb{A}}^{s} \sum_{\{Q: \tilde{P}_{0} \subseteq Q\}} e^{(2\rho_{Q,+} - 2\rho_{Q_{H}} + s\Sigma_{1})(T_{Q}^{G})} J_{\mathbf{o}}^{Q,T}(\eta, \{s_{i}^{Q} + s\}, f_{Q,s,y}^{\eta}) = 0$$

where $J_{\mathfrak{o}}^{G,T}$ and $J_{\mathfrak{o}}^{Q,T}$ are defined by the formulae (5.0.1) and (5.1.2) respectively, $\{s_i^Q\}_{1\leq i\leq l}\in \mathbb{Z}^l$ are the explicit constants determined by (5.4.1), and we write T_Q^G for the projection of $T_Q\in\mathfrak{a}_Q$ in \mathfrak{a}_Q^G via the decomposition $\mathfrak{a}_Q=\mathfrak{a}_Q^G\oplus\mathfrak{a}_G$.

For $\mathfrak{o} \in \mathcal{O}$ and $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ (resp. $f' \in \mathcal{S}((\mathfrak{m}_Q \cap \mathfrak{s})(\mathbb{A}))$), thanks to Corollary 5.6 (resp. Remark 5.7), we may take the constant term $J^G_{\mathfrak{o}}(\eta, s, f)$ of $J^{G,T}_{\mathfrak{o}}(\eta, s, f)$ (resp. $J^Q_{\mathfrak{o}}(\eta, \{s_i\}, f')$ of $J^{Q,T}_{\mathfrak{o}}(\eta, \{s_i\}, f')$) for $s \in \mathbb{C}$ (resp. $\{s_i\}_{1 \leq i \leq l} \in \mathbb{C}^l$). When s = 0 (resp. $s_i = 0$ for all $1 \leq i \leq l$), denote $J^G_{\mathfrak{o}}(\eta, f) := J^G_{\mathfrak{o}}(\eta, 0, f)$ (resp. $J^Q_{\mathfrak{o}}(\eta, f') := J^Q_{\mathfrak{o}}(\eta, \{0\}, f')$).

COROLLARY 6.2. Assume that p = q = n. Let $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A})), y \in H(\mathbb{A}) \cap G(\mathbb{A})^1$ and $\mathfrak{o} \in \mathcal{O}$. We have

$$J^G_{\mathfrak{o}}(\eta, f^y) = \eta(\operatorname{Nrd}(y)) \sum_{\{Q: \widetilde{P}_0 \subseteq Q, \omega Q \omega^{-1} = Q\}} J^Q_{\mathfrak{o}}(\eta, f^\eta_{Q, 0, y}).$$

PROOF OF COROLLARY 6.2. We apply Proposition 6.1 to the case s = 0 and consider the constant terms of both sides. Because $J_o^{Q,T}$ is independent of T_Q , by Lemma 5.2, only ω -stable Q contribute to the purely polynomial term. Then we apply Proposition 5.4 to the case p = q = n to conclude.

PROOF OF PROPOSITION 6.1. By definition,

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^{1}} \left(\sum_{\{P: \widetilde{P}_{0} \subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta \in P_{H}(F) \setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \cdot k_{f^{y}, P, \mathfrak{o}}(\delta x) \right) \\ \cdot \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx,$$

where

$$k_{f^{y},P,\mathfrak{o}}(\delta x) = \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{P} \cap \mathfrak{s})(\mathbb{A})} f(y(\delta x)^{-1}(X+U)\delta xy^{-1})dU = k_{f,P,\mathfrak{o}}(\delta xy^{-1}).$$

By change of variables, we have

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^{1}} \left(\sum_{\{P: \tilde{P}_{0} \subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta \in P_{H}(F) \setminus H(F)} \hat{\tau}_{P}^{G}(H_{P}(\delta xy) - T_{P}) \cdot k_{f,P,\mathfrak{o}}(\delta x) \right) \\ \cdot \eta(\operatorname{Nrd}(xy)) |\operatorname{Nrd}(x_{1}y_{1})|_{\mathbb{A}}^{s} dx.$$

For $x \in H(\mathbb{A})$ and P a relatively standard parabolic subgroup of G, let $k_P(x)$ be an element in K_H such that $xk_P(x)^{-1} \in P_H(\mathbb{A})$. Then

$$\widehat{\tau}_P^G(H_P(\delta xy) - T_P) = \widehat{\tau}_P^G(H_P(\delta x) - T_P + H_P(k_P(\delta x)y))$$

Substituting $T_1 = H_P(\delta x) - T_P$ and $T_2 = -H_P(k_P(\delta x)y)$ in the definition of $\Gamma_P(T_1, T_2)$, we get

$$\widehat{\tau}_P^G(H_P(\delta xy) - T_P) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_Q/A_G)} \widehat{\tau}_P^Q(H_P(\delta x) - T_P) \Gamma_Q(H_P(\delta x) - T_P, -H_P(k_P(\delta x)y)).$$

Thus

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \sum_{\{P:\tilde{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta\in P_{H}(F)\backslash H(F)} \left(\sum_{\{Q:P\subseteq Q\}} (-1)^{\dim(A_{Q}/A_{G})} \widehat{\tau}_{P}^{Q}(H_{P}(\delta x) - T_{P}) \Gamma_{Q}(H_{P}(\delta x) - T_{P}, -H_{P}(k_{P}(\delta x)y)) \right) \\ \cdot k_{f,P,\mathfrak{o}}(\delta x) \eta(\operatorname{Nrd}(xy)) |\operatorname{Nrd}(x_{1}y_{1})|_{\mathbb{A}}^{s} dx,$$

Exchanging the order of two sums over P and Q, and decomposing the sum over $P_H(F) \setminus H(F)$ into two sums over $P_H(F) \setminus Q_H(F)$ and $Q_H(F) \setminus H(F)$, we obtain

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \sum_{\{Q: \widetilde{P}_{0} \subseteq Q\}} \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^{1}} \sum_{\{P: \widetilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in P_{H}(F) \setminus Q_{H}(F)} \sum_{\delta' \in Q_{H}(F) \setminus H(F)} \widehat{\tau}_{P}^{Q}(H_{P}(\delta\delta'x) - T_{P}) \Gamma_{Q}(H_{P}(\delta\delta'x) - T_{P}, -H_{P}(k_{P}(\delta\delta'x)y)) k_{f,P,\mathfrak{o}}(\delta\delta'x)\eta(\operatorname{Nrd}(xy)) \cdot |\operatorname{Nrd}(x_{1}y_{1})|_{\mathbb{A}}^{s} dx.$$

Combining the integral over $H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$ and the sum over $Q_H(F) \setminus H(F)$ into the integral over $Q_H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$, and using the fact that

$$P_H(F) \setminus Q_H(F) \simeq (P_H(F) \cap M_{Q_H}(F)) \setminus M_{Q_H}(F),$$

we have

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \sum_{\{Q: \tilde{P}_{0} \subseteq Q\}} \int_{Q_{H}(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^{1}} \sum_{\{P: \tilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)} \hat{\tau}_{P}^{Q} (H_{P}(\delta x) - T_{P}) \Gamma_{Q}(H_{P}(\delta x) - T_{P}, -H_{P}(k_{P}(\delta x)y)) k_{f,P,\mathfrak{o}}(\delta x) \eta(\operatorname{Nrd}(xy)) |\operatorname{Nrd}(x_{1}y_{1})|_{\mathbb{A}}^{s} dx.$$

By the Iwasawa decomposition and our choice of measures, we get

$$J_{\mathbf{0}}^{G,T}(\eta, s, f^{y}) = \sum_{\{Q: \tilde{P}_{0} \subseteq Q\}} \int_{K_{H}} \int_{M_{Q_{H}}(F) \setminus M_{Q_{H}}(\mathbb{A}) \cap M_{Q}(\mathbb{A})^{1}} \int_{A_{Q}^{G,\infty}} \int_{N_{Q_{H}}(F) \setminus N_{Q_{H}}(\mathbb{A})} \sum_{\{P: \tilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)} \widehat{\tau}_{P}^{Q}(H_{P}(\delta namk) - T_{P})\Gamma_{Q}(H_{P}(\delta namk) - T_{P}, -H_{P}(k_{P}(\delta namk)y)) + k_{f,P,\mathfrak{0}}(\delta namk)\eta(\operatorname{Nrd}(mky))|\operatorname{Nrd}(a_{1}m_{1}y_{1})|_{\mathbb{A}}^{s}e^{-2\rho_{Q_{H}}(H_{Q_{H}}(am))}dndadmdk.$$

As in the proof of Theorem 5.5, we see that

$$\hat{\tau}_P^Q(H_P(\delta namk) - T_P) = \hat{\tau}_P^Q(H_P(\delta m) - T_P),$$

and that

$$k_{f,P,\mathfrak{o}}(\delta namk) = e^{2\rho_{Q,+}(H_Q(a))}k_{f,P,\mathfrak{o}}(\delta mk).$$

In addition,

$$\begin{split} \Gamma_Q(H_P(\delta namk) - T_P, -H_P(k_P(\delta namk)y)) &= \Gamma_Q(H_Q(\delta namk) - T_Q, -H_Q(k_P(\delta namk)y)) \\ &= \Gamma_Q(H_Q(a) - T_Q, -H_Q(k_Q(\delta namk)y)) \\ &= \Gamma_Q(H_Q(a) - T_Q, -H_Q(ky)). \end{split}$$

To sum up, the integrand in $J^{G,T}_{\mathfrak{o}}(\eta, s, f^y)$ is independent of $n \in N_{Q_H}(F) \setminus N_{Q_H}(\mathbb{A})$. We can choose the Haar measure such that $\operatorname{vol}(N_{Q_H}(F) \setminus N_{Q_H}(\mathbb{A})) = 1$. Then

$$J_{\mathfrak{o}}^{G,T}(\eta, s, f^{y}) = \sum_{\{Q: \widetilde{P}_{0} \subseteq Q\}} \int_{K_{H}} \int_{M_{Q_{H}}(F) \setminus M_{Q_{H}}(\mathbb{A}) \cap M_{Q}(\mathbb{A})^{1}} \int_{A_{Q}^{G,\infty}} \sum_{\{P: \widetilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})}$$
$$\sum_{\substack{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)}} \widehat{\tau}_{P}^{Q} (H_{P}(\delta m) - T_{P}) \Gamma_{Q} (H_{Q}(a) - T_{Q}, -H_{Q}(ky))$$
$$\cdot e^{2\rho_{Q,+}(H_{Q}(a))} k_{f,P,\mathfrak{o}}(\delta mk) \eta (\operatorname{Nrd}(mky)) |\operatorname{Nrd}(a_{1}m_{1}y_{1})|_{\mathbb{A}}^{s} e^{-2\rho_{Q_{H}}(H_{Q_{H}}(am))} dadmdk.$$

First, let us compute the integral on $A_Q^{G,\infty}$, which is

$$\begin{split} &\int_{A_Q^{G,\infty}} |\mathrm{Nrd}(a_1)|_{\mathbb{A}}^s e^{(2\rho_{Q,+}-2\rho_{Q_H})(H_Q(a))} \Gamma_Q(H_Q(a) - T_Q, -H_Q(ky)) da \\ &:= \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_1)} \Gamma_Q(T_1 - T_Q, -H_Q(ky)) dT_1 \\ &= e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_Q^G)} \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_1)} \Gamma_Q(T_1, -H_Q(ky)) dT_1 \\ &= e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_Q^G)} \int_{\mathfrak{a}_Q^G} e^{(2\rho_{Q,+}-2\rho_{Q_H}+s\Sigma_1)(T_1)} \Gamma_Q(T_1, -H_Q(ky)) dT_1 \end{split}$$

Next, we consider the integral on K_H , which is

$$\int_{K_H} k_{f,P,\mathfrak{o}}(\delta m k) \eta(\operatorname{Nrd}(k)) p_{Q,s}(-H_Q(ky)) dk$$

As in the proof of Theorem 5.5, we see that

$$k_{f,P,\mathfrak{o}}(\delta mk) = e^{2\rho_{Q,+}(H_{Q_H}(m))} \sum_{X \in \mathfrak{m}_P(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_P^Q \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_Q \cap \mathfrak{s})(\mathbb{A})} f(k^{-1}((\delta m)^{-1}(X+U)\delta m+V)k) dV_{\mathcal{S}}(h) dU \int_{(\mathfrak{n}_Q \cap \mathfrak{s})(\mathbb{A})} f(k^{-1}((\delta m)^{-1}(X+U)\delta m+V)k) dV_{\mathcal{S}}(h) d$$

so we can write

$$\begin{split} &\int_{K_H} k_{f,P,\mathfrak{o}}(\delta m k)\eta(\mathrm{Nrd}(k))p_{Q,s}(-H_Q(ky))dk\\ =& e^{2\rho_{Q,+}(H_{Q_H}(m))}\sum_{X\in\mathfrak{m}_P(F)\cap\mathfrak{o}}\int_{(\mathfrak{n}_P^Q\cap\mathfrak{s})(\mathbb{A})} f_{Q,s,y}^\eta((\delta m)^{-1}(X+U)\delta m)dU\\ =& e^{2\rho_{Q,+}(H_{Q_H}(m))}\sum_{j=1}^t k_{f_{Q,s,y}^\eta,P\cap M_Q,\mathfrak{o}_j}^{M_Q}(\delta m) \end{split}$$

by (5.1.1). Hence

$$J_{\mathbf{o}}^{G,T}(\eta, s, f^{y}) = \eta(\operatorname{Nrd}(y))|\operatorname{Nrd}(y_{1})|_{\mathbb{A}}^{s} \sum_{\{Q:\tilde{P}_{0} \subseteq Q\}} e^{(2\rho_{Q,+}-2\rho_{Q_{H}}+s\Sigma_{1})(T_{Q}^{G})} \int_{M_{Q_{H}}(F)\setminus M_{Q_{H}}(\mathbb{A})\cap M_{Q}(\mathbb{A})^{1}} \sum_{\{P:\tilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in (P_{H}(F)\cap M_{Q_{H}}(F))\setminus M_{Q_{H}}(F)} \widehat{\tau}_{P}^{Q}(H_{P}(\delta m) - T_{P}) \left(\sum_{j=1}^{t} k_{f_{Q,s,y}^{\eta}, P\cap M_{Q}, \mathfrak{o}_{j}}(\delta m)\right) \eta(\operatorname{Nrd}(m))|\operatorname{Nrd}(m_{1})|_{\mathbb{A}}^{s} e^{(2\rho_{Q,+}-2\rho_{Q_{H}})(H_{Q_{H}}(m))} dm.$$

As in the proof of Theorem 5.5, we notice that

$$\begin{split} J_{\mathbf{0}}^{Q,T}(\eta, \{s_{i}^{Q}+s\}, f_{Q,s,y}^{\eta}) = & \int_{M_{Q_{H}}(F) \setminus M_{Q_{H}}(\mathbb{A}) \cap M_{Q}(\mathbb{A})^{1}} \sum_{\{P: \tilde{P}_{0} \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta \in (P_{H}(F) \cap M_{Q_{H}}(F)) \setminus M_{Q_{H}}(F)} \\ & \quad \hat{\tau}_{P}^{Q}(H_{P}(\delta m) - T_{P}) \left(\sum_{j=1}^{t} k_{f_{Q,s,y}^{\eta}, P \cap M_{Q}, \mathbf{0}_{j}}^{M}(\delta m) \right) \eta(\mathrm{Nrd}(m)) |\mathrm{Nrd}(m_{1})|_{\mathbb{A}}^{s} \\ & \quad \cdot e^{(2\rho_{Q}, + -2\rho_{Q_{H}})(H_{Q_{H}}(m))} dm. \end{split}$$

Then we finish the proof.

7. An infinitesimal trace formula for $Mat_{p \times q,D} \oplus Mat_{q \times p,D} / / GL_{p,D} \times GL_{q,D}$

THEOREM 7.1. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $s \in \mathbb{R}$,

$$\sum_{\mathfrak{o}\in\mathcal{O}}J^G_{\mathfrak{o}}(\eta,s,f)=\sum_{\mathfrak{o}\in\mathcal{O}}J^G_{\mathfrak{o}}(\eta,s,\hat{f}),$$

where \hat{f} is the Fourier transform of f defined by (3.5.2), and $J^G_{\mathfrak{o}}(\eta, s, \cdot)$ denotes the constant term of $J^{G,T}_{\mathfrak{o}}(\eta, s, \cdot)$.

PROOF. From the Poisson summation formula, we know that for any $x \in H(\mathbb{A})$,

$$\sum_{X \in \mathfrak{s}(F)} f(x^{-1}Xx) = \sum_{X \in \mathfrak{s}(F)} \hat{f}(x^{-1}Xx),$$

i.e.,

$$k_{f,G}(x) = k_{\hat{f},G}(x).$$

Using Corollary 4.13, for all sufficiently regular T satisfying $\alpha(T) \ge \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}$, we have

$$\left| J^{G,T}(\eta, s, f) - \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} F^G(x, T) k_{f,G}(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|^s_{\mathbb{A}} dx \right| \le C_1 e^{-N ||T||}$$

and

$$\left| J^{G,T}(\eta, s, \hat{f}) - \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} F^G(x, T) k_{\hat{f}, G}(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx \right| \le C_2 e^{-N ||T||}.$$

Thus

$$|J^{G,T}(\eta, s, f) - J^{G,T}(\eta, s, \hat{f})| \le (C_1 + C_2)e^{-N||T||}.$$

By Corollary 5.6, we know that both of $J^{G,T}(\eta, s, f)$ and $J^{G,T}(\eta, s, \hat{f})$ are exponential polynomials in T. Because we can choose N to be large enough, we deduce that

$$J^{G,T}(\eta,s,f) = J^{G,T}(\eta,s,\hat{f})$$

Since

$$J^{G,T}(\eta,s,f) = \sum_{\mathfrak{o}\in\mathcal{O}} J^{G,T}_{\mathfrak{o}}(\eta,s,f)$$

and

$$J^{G,T}(\eta, s, \hat{f}) = \sum_{\mathfrak{o} \in \mathcal{O}} J^{G,T}_{\mathfrak{o}}(\eta, s, \hat{f}),$$

we obtain

$$\sum_{\mathfrak{o}\in\mathcal{O}}J^{G,T}_{\mathfrak{o}}(\eta,s,f)=\sum_{\mathfrak{o}\in\mathcal{O}}J^{G,T}_{\mathfrak{o}}(\eta,s,\hat{f}).$$

We may conclude by taking the constant terms of both sides.

8. The second modified kernel

In this section and the next, we shall focus on the case where p = q = n in order to get better description for distributions associated to regular semi-simple orbits. We shall change our notation by denoting $G := GL_{2n,D}$ and $H := GL_{n,D} \times GL_{n,D}$ without further mention.

Let $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, P be a relatively standard parabolic subgroup of G and $\mathfrak{o} \in \mathcal{O}_{rs}$ (see Section 3.3). For $x \in P_H(F) \setminus H(\mathbb{A})$, define

$$j_{f,P,\mathfrak{o}}(x) := \sum_{X \in \mathfrak{m}_P(F) \cap \mathfrak{o}} \sum_{n \in N_{P_H}(F)} f((nx)^{-1}Xnx).$$

Let $T \in \mathfrak{a}_0$. For $x \in H(F) \setminus H(\mathbb{A})$, define

$$j_{f,\mathfrak{o}}^{T}(x) := \sum_{\{P: \tilde{P}_{0} \subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta \in P_{H}(F) \setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \cdot j_{f,P,\mathfrak{o}}(\delta x).$$

From [3, Lemma 5.1], we know that the sum over $\delta \in P_H(F) \setminus H(F)$ is finite. Recall that since $\mathfrak{o} \in \mathcal{O}_{rs} \subseteq \mathcal{O}^{\times}$, if $\mathfrak{m}_P(F) \cap \mathfrak{o} \neq \emptyset$, then P is ω -stable by Proposition 5.1. Thus the above definitions only involve the relatively standard parabolic subgroups that are ω -stable.

LEMMA 8.1. Let P be a relatively standard parabolic subgroup of G and $\mathfrak{o} \in \mathcal{O}_{rs}$. For $X \in \mathfrak{m}_P(F) \cap \mathfrak{o}$, the map

$$N_{P_H} \to \mathfrak{n}_P \cap \mathfrak{s}, n \mapsto n^{-1}Xn - X$$

is an F-isomorphism of algebraic varieties and preserves the Haar measures on \mathbb{A} -points.

PROOF. Since P is relatively standard and ω -stable, we can suppose

$$P = \begin{pmatrix} \mathfrak{p}_{n,D} & \mathfrak{p}_{n,D} \\ \mathfrak{p}_{n,D} & \mathfrak{p}_{n,D} \end{pmatrix}^{\times},$$

where

$$P_{n,D} = \begin{pmatrix} GL_{n_1,D} & \operatorname{Mat}_{n_1 \times n_2,D} & \cdots & \operatorname{Mat}_{n_1 \times n_l,D} \\ & GL_{n_2,D} & \cdots & \operatorname{Mat}_{n_2 \times n_l,D} \\ & & \ddots & \vdots \\ & & & GL_{n_l,D} \end{pmatrix}$$

Then we have

$$\mathfrak{m}_P \cap \mathfrak{s} = \begin{pmatrix} \mathfrak{m}_{P_{n,D}} \\ \mathfrak{m}_{P_{n,D}} \end{pmatrix}, N_{P_H} = \begin{pmatrix} N_{P_{n,D}} \\ N_{P_{n,D}} \end{pmatrix}, \mathfrak{n}_P \cap \mathfrak{s} = \begin{pmatrix} \mathfrak{n}_{P_{n,D}} \\ \mathfrak{n}_{P_{n,D}} \end{pmatrix}.$$

Let

$$X = \begin{pmatrix} A_1 & & \\ & & \ddots & \\ B_1 & & & A_l \\ & \ddots & & \\ & & B_l \end{pmatrix} \in \mathfrak{m}_P(F) \cap \mathfrak{o},$$

where $A_i, B_i \in GL_{n_i}(D)$ for $1 \le i \le l$, and

$$n = \begin{pmatrix} 1 & C_{12} & \cdots & C_{1l} & & & \\ & 1 & \cdots & C_{2l} & & & \\ & & \ddots & \vdots & & & \\ & & & 1 & & & \\ & & & & 1 & D_{12} & \cdots & D_{1l} \\ & & & & & 1 & D_{12} & \cdots & D_{2l} \\ & & & & & & \ddots & \vdots \\ & & & & & & & 1 \end{pmatrix} \in N_{P_H},$$

where $C_{ij}, D_{ij} \in \operatorname{Mat}_{n_i \times n_j, D}$ for $1 \le i < j \le l$. Then

$$Xn - nX = \begin{pmatrix} 0 & A_1D_{12} - C_{12}A_2 & \cdots & A_1D_{1l} - C_{1l}A_l \\ 0 & \cdots & A_2D_{2l} - C_{2l}A_l \\ & \ddots & \vdots \\ 0 & B_1C_{12} - D_{12}B_2 & \cdots & B_1C_{1l} - D_{1l}B_l \\ 0 & \cdots & B_2C_{2l} - D_{2l}B_l \\ & \ddots & \vdots \\ 0 & & 0 \end{pmatrix} \in \mathfrak{n}_P \cap \mathfrak{s}$$

We claim that the morphism of F-affine spaces

$$\begin{aligned} \operatorname{Mat}_{n_i \times n_j, D} \oplus \operatorname{Mat}_{n_i \times n_j, D} &\to \operatorname{Mat}_{n_i \times n_j, D} \oplus \operatorname{Mat}_{n_i \times n_j, D} \\ (C_{ij}, D_{ij}) &\mapsto (A_i D_{ij} - C_{ij} A_j, B_i C_{ij} - D_{ij} B_j) \end{aligned}$$

induces an *F*-linear isomorphism on *F*-points. In fact, since it gives an *F*-linear map between finite dimensional linear spaces of the same dimension, we only need to prove that this map is injective under base change to an algebraic closure of *F*. Then without loss of generality, it suffices to consider the case where D = F. If $A_i D_{ij} - C_{ij} A_j = B_i C_{ij} - D_{ij} B_j = 0$, then $C_{ij} A_j B_j = A_i D_{ij} B_j = A_i B_i C_{ij}$ and $D_{ij} B_j A_j = B_i C_{ij} A_j = B_i A_i D_{ij}$. Since X is regular semi-simple, $A_i B_i$ and $A_j B_j$ (resp. $B_i A_i$ and $B_j A_j$) have no common eigenvalue. By the classical theory of Sylvester equation, we know that $C_{ij} = D_{ij} = 0$ and conclude.

From this claim, we know that the map

$$N_{P_H} \to \mathfrak{n}_P \cap \mathfrak{s}, n \mapsto Xn - nX$$

is an *F*-isomorphism of algebraic varieties and preserves the Haar measures on A-points. Notice that $n^{-1}Xn - X = n^{-1}(Xn - nX)$. It is not hard to check that here n^{-1} functions as some translation $A_iD_{ij} - C_{ij}A_j \mapsto A_iD_{ij} - C_{ij}A_j + (a \text{ polynomial of } C_{i'j'} \text{ and } D_{i'j'}, i' > i, j' \leq j \text{ or } i' \geq i, j' < j)$, so an analogous assertion still holds for the map $n \mapsto n^{-1}Xn - X$.

THEOREM 8.2. For all sufficiently regular T, all $s \in \mathbb{R}$ and $\mathfrak{o} \in \mathcal{O}_{rs}$,

$$\int_{H(F)\setminus H(\mathbb{A})\cap G(\mathbb{A})^1} |j_{f,\mathfrak{o}}^T(x)| |\mathrm{Nrd}(x_1)|_{\mathbb{A}}^s dx < \infty,$$

where we write $x = (x_1, x_2) \in GL_{n,D}(\mathbb{A}) \times GL_{n,D}(\mathbb{A})$. Moreover, for $s \in \mathbb{C}$,

$$J_{\mathfrak{o}}^{G,T}(\eta,s,f) = \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} j_{f,\mathfrak{o}}^T(x)\eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx.$$

PROOF. As in the proof of Theorem 4.11, using the left invariance of $j_{f,P,\mathfrak{o}}$ by $P_H(F)$, we reduce ourselves to proving

$$\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \chi^T_{P_1,P_2}(x) |j_{P_1,P_2,\mathfrak{o}}(x)| |\mathrm{Nrd}(x_1)|^s_{\mathbb{A}} dx < \infty,$$

where $P_1 \subsetneq P_2$ are a pair of relatively standard parabolic subgroups of G and for $x \in P_{1,H}(F) \setminus H(\mathbb{A})$, we put

$$j_{P_1,P_2,\mathfrak{o}}(x) := \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} j_{f,P,\mathfrak{o}}(x).$$

In addition,

$$j_{f,P,\mathfrak{o}}(x) = \sum_{\{R: P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{X \in (\mathfrak{n}_R^P \cap \mathfrak{s})(F)} \sum_{n \in N_{P_H}(F)} f((nx)^{-1}(\xi + X)nx).$$

Applying Lemma 8.1, we get

$$j_{f,P,\mathfrak{o}}(x) = \sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{X \in (\mathfrak{n}_R^P \cap \mathfrak{s})(F)} \sum_{u \in (\mathfrak{n}_P \cap \mathfrak{s})(F)} f(x^{-1}(\xi + X + u)x)$$
$$= \sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{P_1}^R(F) \cap \mathfrak{o}} \sum_{X \in (\mathfrak{n}_R \cap \mathfrak{s})(F)} f(x^{-1}(\xi + X)x).$$

Hence

$$j_{P_{1},P_{2},\mathfrak{o}}(x) = \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{G})} \left(\sum_{\{R:P_{1}\subseteq R\subseteq P\}} \sum_{\xi\in \tilde{m}_{P_{1}}^{R}(F)\cap\mathfrak{o}} \sum_{X\in(\mathfrak{n}_{R}\cap\mathfrak{s})(F)} f(x^{-1}(\xi+X)x) \right)$$
$$= \sum_{\{R:P_{1}\subseteq R\subseteq P_{2}\}} \sum_{\xi\in \tilde{m}_{P_{1}}^{R}(F)\cap\mathfrak{o}} \left(\sum_{\{P:R\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{G})} \right) \sum_{X\in(\mathfrak{n}_{R}\cap\mathfrak{s})(F)} f(x^{-1}(\xi+X)x).$$

By [3, Proposition 1.1], we have

$$j_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_G)} \sum_{\xi \in \widetilde{m}_{P_1}^{P_2}(F) \cap \mathfrak{o}} \sum_{X \in (\mathfrak{n}_{P_2} \cap \mathfrak{s})(F)} f(x^{-1}(\xi + X)x).$$

Applying Lemma 8.1 again, we obtain

$$j_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_G)} \sum_{\xi \in \widetilde{m}_{P_1}^{P_2}(F) \cap \mathfrak{o}} \sum_{n_2 \in N_{P_2,H}(F)} f((n_2 x)^{-1} \xi n_2 x),$$

where we denote $P_{2,H} := P_2 \cap H$.

Decomposing the integral over $x \in P_{1,H}(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$ into double integrals $n_1 \in N_{P_{1,H}}(F) \setminus N_{P_{1,H}}(\mathbb{A})$ and $y \in M_{P_{1,H}}(F) N_{P_{1,H}}(\mathbb{A}) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$, and using the fact that $\chi^T_{P_1,P_2}(x)$ is left invariant under $N_{P_{1,H}}(\mathbb{A})$, we have

$$\begin{split} &\int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)|j_{P_{1},P_{2},\mathfrak{o}}(x)||\mathrm{Nrd}(x_{1})|^{s}_{\mathbb{A}}dx \\ &=\int_{M_{P_{1,H}}(F)N_{P_{1,H}}(\mathbb{A})\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\int_{N_{P_{1,H}}(F)\backslash N_{P_{1,H}}(\mathbb{A})}\chi^{T}_{P_{1},P_{2}}(n_{1}y) \\ &\cdot \left|\sum_{\xi\in \widetilde{m}^{P_{2}}_{P_{1}}(F)\cap\mathfrak{o}}\sum_{n_{2}\in N_{P_{2,H}}(F)}f((n_{2}n_{1}y)^{-1}\xi n_{2}n_{1}y)\right| \left|\mathrm{Nrd}(y_{1})|^{s}_{\mathbb{A}}dn_{1}dy \right| \\ &\leq \int_{M_{P_{1,H}}(F)N_{P_{1,H}}(\mathbb{A})\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(y)\sum_{\xi\in \widetilde{m}^{P_{2}}_{P_{1}}(F)\cap\mathfrak{o}} \\ &\left(\int_{N_{P_{1,H}}(F)\backslash N_{P_{1,H}}(\mathbb{A})}\sum_{n_{2}\in N_{P_{2,H}}(F)}|f((n_{2}n_{1}y)^{-1}\xi n_{2}n_{1}y)|dn_{1}\right)|\mathrm{Nrd}(y_{1})|^{s}_{\mathbb{A}}dy. \end{split}$$

Since $P_{1,H} \subseteq P_{2,H}$ and $\operatorname{vol}(N_{P_{2,H}}(F) \setminus N_{P_{2,H}}(\mathbb{A})) = 1$, we see that

$$\begin{split} &\int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \sum_{n_{2}\in N_{P_{2,H}}(F)} |f((n_{2}n_{1}y)^{-1}\xi n_{2}n_{1}y)|dn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{N_{P_{2,H}}(F)\setminus N_{P_{2,H}}(\mathbb{A})} \sum_{n_{2}\in N_{P_{2,H}}(F)} |f((n_{2}nn_{1}y)^{-1}\xi n_{2}nn_{1}y)|dndn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{N_{P_{2,H}}(\mathbb{A})} |f((nn_{1}y)^{-1}\xi nn_{1}y)|dndn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{(\mathfrak{n}_{P_{2}}\cap\mathfrak{s})(\mathbb{A})} |f((n_{1}y)^{-1}(\xi+U)n_{1}y)|dUdn_{1}, \end{split}$$

where we have applied Lemma 8.1 in the last equality. Therefore

$$\begin{split} & \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) |j_{P_{1},P_{2},\mathfrak{o}}(x)| |\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx \\ & \leq \int_{M_{P_{1,H}}(F)N_{P_{1,H}}(\mathbb{A})\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(y) \sum_{\xi\in \widetilde{m}_{P_{1}}^{P_{2}}(F)\cap\mathfrak{o}} \\ & \left(\int_{N_{P_{1,H}}(F)\backslash N_{P_{1,H}}(\mathbb{A})} \int_{(\mathfrak{n}_{P_{2}}\cap\mathfrak{s})(\mathbb{A})} |f((n_{1}y)^{-1}(\xi+U)n_{1}y)| dU dn_{1}\right) |\mathrm{Nrd}(y_{1})|_{\mathbb{A}}^{s} dy \\ & = \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\xi\in \widetilde{m}_{P_{1}}^{P_{2}}(F)\cap\mathfrak{o}} \left(\int_{(\mathfrak{n}_{P_{2}}\cap\mathfrak{s})(\mathbb{A})} |f(x^{-1}(\xi+U)x)| dU\right) |\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s} dx, \end{split}$$

whose convergence results from that of the formula (4.2.2) when $R = P_2$.

Now we begin to prove the second statement. From the first statement, now we have the right to write

$$\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} j_{f,\mathfrak{o}}^T(x)\eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx$$
$$= \sum_{\{P_1,P_2:\widetilde{P}_0\subseteq P_1\subseteq P_2\}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^1} \chi_{P_1,P_2}^T(x)j_{P_1,P_2,\mathfrak{o}}(x)\eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_1)|_{\mathbb{A}}^s dx,$$

where

$$j_{P_1,P_2,\mathfrak{o}}(x) = \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} j_{f,P,\mathfrak{o}}(x)$$
$$= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_G)} \left(\sum_{X \in \mathfrak{m}_P(F) \cap \mathfrak{o}} \sum_{n \in N_{P_H}(F)} f((nx)^{-1}Xnx) \right).$$

Decompose the integral over $x \in P_{1,H}(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$ into double integrals over $n_1 \in N_{P_{1,H}}(F) \setminus N_{P_{1,H}}(\mathbb{A})$ and $y \in M_{P_{1,H}}(F) N_{P_{1,H}}(\mathbb{A}) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1$. Since $N_{P_{1,H}}(F) \setminus N_{P_{1,H}}(\mathbb{A})$ is compact, by Lemma 8.1 and [53, §41],

$$\sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \sum_{n \in N_{P_{H}}(F)} |f((nn_{1}y)^{-1}Xnn_{1}y)| = \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \sum_{u \in (\mathfrak{n}_{P} \cap \mathfrak{s})(F)} |f((n_{1}y)^{-1}(X+u)n_{1}y)|$$

is bounded on $n_1 \in N_{P_{1,H}}(F) \setminus N_{P_{1,H}}(\mathbb{A})$. Then using the fact that $\chi^T_{P_1,P_2}(x)$ is left invariant under $N_{P_{1,H}}(\mathbb{A})$, we have

$$\begin{split} &\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} j_{f,\mathfrak{o}}^{T}(x)\eta(\operatorname{Nrd}(x))|\operatorname{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx \\ &= \sum_{\{P_{1},P_{2}:\tilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{M_{P_{1,H}}(F)N_{P_{1,H}}(\mathbb{A})\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(y) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{G})} \\ &\sum_{X\in\mathfrak{m}_{P}(F)\cap\mathfrak{o}} \left(\int_{N_{P_{1,H}}(F)\backslash N_{P_{1,H}}(\mathbb{A})} \sum_{n\in N_{P_{H}}(F)} f((nn_{1}y)^{-1}Xnn_{1}y)dn_{1} \right) \eta(\operatorname{Nrd}(y))|\operatorname{Nrd}(y_{1})|_{\mathbb{A}}^{s}dy. \end{split}$$

Since $P_{1,H} \subseteq P_H$ and $\operatorname{vol}(N_{P_H}(F) \setminus N_{P_H}(\mathbb{A})) = 1$, we see that

$$\begin{split} &\int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \sum_{n\in N_{P_{H}}(F)} f((nn_{1}y)^{-1}Xnn_{1}y)dn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{N_{P_{H}}(F)\setminus N_{P_{H}}(\mathbb{A})} \sum_{n\in N_{P_{H}}(F)} f((nn_{2}n_{1}y)^{-1}Xnn_{2}n_{1}y)dn_{2}dn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{N_{P_{H}}(\mathbb{A})} f((nn_{1}y)^{-1}Xnn_{1}y)dndn_{1} \\ &= \int_{N_{P_{1,H}}(F)\setminus N_{P_{1,H}}(\mathbb{A})} \int_{(\mathfrak{n}_{P}\cap\mathfrak{s})(\mathbb{A})} f((n_{1}y)^{-1}(X+U)n_{1}y)dUdn_{1}, \end{split}$$

where we have applied Lemma 8.1 in the last equality. Therefore

$$\begin{split} &\int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} j_{f,\mathfrak{o}}^{T}(x)\eta(\mathrm{Nrd}(x))|\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx \\ &= \sum_{\{P_{1},P_{2}:\tilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{M_{P_{1},H}(F)N_{P_{1},H}(\mathbb{A})\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(y) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{G})} \\ &\sum_{X\in\mathfrak{m}_{P}(F)\cap\mathfrak{o}} \left(\int_{N_{P_{1},H}(F)\backslash N_{P_{1},H}(\mathbb{A})} \int_{(\mathfrak{n}_{P}\cap\mathfrak{s})(\mathbb{A})} f((n_{1}y)^{-1}(X+U)n_{1}y)dUdn_{1} \right) \eta(\mathrm{Nrd}(y))|\mathrm{Nrd}(y_{1})|_{\mathbb{A}}^{s}dy \\ &= \sum_{\{P_{1},P_{2}:\tilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{G})} \\ &\cdot \left(\sum_{X\in\mathfrak{m}_{P}(F)\cap\mathfrak{o}} \int_{(\mathfrak{n}_{P}\cap\mathfrak{s})(\mathbb{A})} f(x^{-1}(X+U)x)dU\right) \eta(\mathrm{Nrd}(x))|\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx \\ &= \sum_{\{P_{1},P_{2}:\tilde{P}_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{P_{1,H}(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x)k_{P_{1},P_{2},\mathfrak{o}}(x)\eta(\mathrm{Nrd}(x))|\mathrm{Nrd}(x_{1})|_{\mathbb{A}}^{s}dx. \end{split}$$

From Theorem 4.11, we are authorised to write

$$\begin{aligned} J^{G,T}_{\mathfrak{o}}(\eta, s, f) &= \int_{H(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} k^T_{f, \mathfrak{o}}(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|^s_{\mathbb{A}} dx \\ &= \sum_{\{P_1, P_2: \widetilde{P}_0 \subseteq P_1 \subseteq P_2\}} \int_{P_{1,H}(F) \setminus H(\mathbb{A}) \cap G(\mathbb{A})^1} \chi^T_{P_1, P_2}(x) k_{P_1, P_2, \mathfrak{o}}(x) \eta(\operatorname{Nrd}(x)) |\operatorname{Nrd}(x_1)|^s_{\mathbb{A}} dx, \end{aligned}$$
ch completes the proof.

which completes the proof.

9. Weighted orbital integrals

As in the last section, we shall assume that p = q = n in the following discussion. Moreover, we shall suppose that s = 0 in the orbital integral for convenience, since $|\operatorname{Nrd}(x_1)|^s_{\mathbb{A}}$ is not invariant under the translation by A_G^{∞} . Recall that for $\mathfrak{o} \in \mathcal{O}$ and $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we denote by $J_{\mathfrak{o}}^G(\eta, f)$ the constant term of $J^{G,T}_{\mathfrak{o}}(\eta, 0, f)$.

9.1. Weyl groups. From Section 5.5, we may choose P_0 to be the stabiliser in G of the flag

$$0 \subsetneq \langle e_1 \rangle_D \subsetneq \langle e_1, f_1 \rangle_D \subsetneq \langle e_1, f_1, e_2 \rangle_D \subsetneq \langle e_1, f_1, e_2, f_2 \rangle_D \subsetneq \cdots \subsetneq \langle e_1, f_1 \cdots, e_n, f_n \rangle_D = V \oplus W$$

by the notation in Section 3.4. Then all ω -stable relatively standard parabolic subgroups of G contain P_0 . Denote by \mathscr{P}_0 the stabiliser in G of the flag

 $0 \subsetneq \langle e_1, f_1 \rangle_D \subsetneq \langle e_1, f_1, e_2, f_2 \rangle_D \subsetneq \cdots \subsetneq \langle e_1, f_1 \cdots, e_n, f_n \rangle_D = V \oplus W.$

It is the minimal ω -stable relatively standard parabolic subgroup of G. A parabolic subgroup P of G is relatively standard and ω -stable if and only if $\mathscr{P}_0 \subseteq P$. Let $\mathscr{P}_{0,n}$ be the group of upper triangular matrices in $GL_{n,D}$. We can talk about positive roots for G, H and $GL_{n,D}$ with respect to P_0, P_0 and $\mathcal{P}_{0,n}$ respectively.

LEMMA 9.1. Let $P_1 = \begin{pmatrix} \mathfrak{p}_{1,n} & \mathfrak{p}_{1,n} \\ \mathfrak{p}_{1,n} & \mathfrak{p}_{1,n} \end{pmatrix}^{\times}$ and $P_2 = \begin{pmatrix} \mathfrak{p}_{2,n} & \mathfrak{p}_{2,n} \\ \mathfrak{p}_{2,n} & \mathfrak{p}_{2,n} \end{pmatrix}^{\times}$ be a pair of ω -stable relatively standard parabolic subgroups of G, where $P_{1,n}$ and $P_{2,n}$ are standard parabolic subgroups of $GL_{n,D}$.

- 1) The map $s_n \mapsto s = \begin{pmatrix} s_n \\ s_n \end{pmatrix}$ induces a bijection from
- a) the set of representatives s_n of $\Omega^{GL_{n,D}}(\mathfrak{a}_{P_{1,n}},\mathfrak{a}_{P_{2,n}})$ in $\Omega^{GL_{n,D}}$ such that $s_n^{-1}\alpha > 0$ for all $\alpha \in$ $\Delta^{P_{2,n}}_{\mathscr{P}_{0,n}}$

to

- b) the set of representatives s of $\Omega^G(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2})$ in Ω^G such that $s^{-1}\alpha > 0$ for all $\alpha \in \Delta_{P_0}^{P_2}$.
- 2) The map $s_n \mapsto s = \begin{pmatrix} s_n \\ s_n \end{pmatrix}$ induces a bijection from a) the set of representatives s_n of $\Omega^{GL_{n,D}}(\mathfrak{a}_{P_{1,n}}; P_{2,n})$ in $\Omega^{GL_{n,D}}$ such that $s_n(\mathfrak{a}_{P_{1,n}}) \supseteq \mathfrak{a}_{P_{2,n}}$ and
- $s_n^{-1}\alpha > 0$ for all $\alpha \in \Delta_{\mathscr{P}_{0,n}}^{P_{2,n}}$

to

b) the set of representatives s of $\Omega^G(\mathfrak{a}_{P_1}; P_2)$ in Ω^G such that $s(\mathfrak{a}_{P_1}) \supseteq \mathfrak{a}_{P_2}$ and $s^{-1}\alpha > 0$ for all $\alpha \in \Delta_{P_0}^{P_2}$.

PROOF. Suppose that $P_{1,n}$ and $P_{2,n}$ correspond to the partitions (n_1, \dots, n_l) and $(n'_1, \dots, n'_{l'})$ respectively of n. Then P_1 and P_2 correspond to the partitions $(2n_1, \dots, 2n_l)$ and $(2n'_1, \dots, 2n'_{l'})$ respectively of 2n. For an integer m > 0, denote by S_m the symmetric group of degree m.

1) From [9, p. 33], the set $\Omega^{GL_{n,D}}(\mathfrak{a}_{P_{1,n}},\mathfrak{a}_{P_{2,n}})$ is empty unless l = l', in which case

(9.1.1)
$$\Omega^{GL_{n,D}}(\mathfrak{a}_{P_{1,n}},\mathfrak{a}_{P_{2,n}}) \simeq \{s_n \in S_l : n'_i = n_{s_n(i)}, 1 \le i \le l\}.$$

Similarly, the set $\Omega^G(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2})$ is empty unless l = l', in which case

(9.1.2)
$$\Omega^G(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2}) \simeq \{s \in S_l : 2n'_i = 2n_{s(i)}, 1 \le i \le l\}.$$

The map in the lemma is induced by the obvious bijection between the right hand sides of (9.1.1) and (9.1.2).

2) From [9, p. 59], the set a) is identified with the set of $s_n \in S_l \subseteq S_n$ such that $(n_{s_n(1)}, \dots, n_{s_n(l)})$ is finer than $(n'_1, \dots, n'_{l'})$, and such that $s_n^{-1}(i) < s_n^{-1}(i+1)$ for any $1 \le i \le n-1$ that is not of the form $n'_1 + \cdots + n'_k$ for some $1 \le k \le l'$. Similarly, the set b) is identified with the set of $s \in S_l \subseteq S_{2n}$ such that $(2n_{s(1)}, \dots, 2n_{s(l)})$ is finer than $(2n'_1, \dots, 2n'_{l'})$, and such that $s^{-1}(i) < s^{-1}(i+1)$ for any $1 \le i \le 2n-1$ that is not of the form $2n'_1 + \cdots + 2n'_k$ for some $1 \le k \le l'$. The map in the lemma is induced by the obvious bijection between these two sets.

For P_1 and P_2 a pair of ω -stable relatively standard parabolic subgroups of G, denote by $\Omega^H(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2})$ the set (perhaps empty) of distinct isomorphisms from \mathfrak{a}_{P_1} to \mathfrak{a}_{P_2} obtained by restriction of elements in Ω^{H} . It is a subset of $\Omega^{G}(\mathfrak{a}_{P_{1}},\mathfrak{a}_{P_{2}})$ a priori. However, since the image of the map in Lemma 9.1.1) is contained in Ω^H , we actually have $\Omega^H(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2}) = \Omega^G(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2})$ (cf. [39, Lemme 2.8.1]). Denote by $\Omega^H(\mathfrak{a}_{P_1};P_2)$ the set of $s \in \bigcup \Omega^H(\mathfrak{a}_{P_1},\mathfrak{a}_Q)$ such that $s(\mathfrak{a}_{P_1}) \supseteq \mathfrak{a}_{P_2}$ and $s^{-1}\alpha > 0$ for each $\alpha \in \Delta_{Q_H}^{P_2 \cap H}$, where the union takes over all \mathfrak{a}_Q associated to some ω -stable relatively standard parabolic subgroup Qof G. Then $\Omega^H(\mathfrak{a}_{P_1}; P_2) = \Omega^G(\mathfrak{a}_{P_1}; P_2)$ by Lemma 9.1.2).

9.2. Regular semi-simple terms. Let $\mathfrak{o} \in \mathcal{O}_{rs}$ (see Section 3.3). It is possible to choose an element $X_1 \in \mathfrak{o}$ and a relatively standard parabolic subgroup P_1 of G such that $X_1 \in \mathfrak{m}_{P_1}(F)$ (thus P_1 is ω -stable by Proposition 5.1) but X_1 can not be H(F)-conjugate to an element in the Lie algebra of any relatively standard parabolic subgroup $R \subsetneq P_1$. We call such X_1 an elliptic element in $(\mathfrak{m}_{P_1} \cap \mathfrak{s})(F)$.

Let $P_1 = \begin{pmatrix} \mathfrak{p}_{1,n} & \mathfrak{p}_{1,n} \\ \mathfrak{p}_{1,n} & \mathfrak{p}_{1,n} \end{pmatrix}^{\times}$ be an ω -stable relatively standard parabolic subgroup of G, where $P_{1,n}$ is a standard parabolic subgroup of $GL_{n,D}$. Let $X_1 = \begin{pmatrix} 0 & A_1 \\ B_1 & 0 \end{pmatrix} \in (\mathfrak{m}_{P_1} \cap \mathfrak{s})(F)$ be a regular semi-simple element in \mathfrak{s} . Then X_1 is elliptic in $(\mathfrak{m}_{P_1} \cap \mathfrak{s})(F)$ if and only if A_1B_1 is elliptic in $\mathfrak{m}_{P_{1,n}}(F)$ in the usual sense, i.e., $\operatorname{Prd}_{A_1B_1}$ is irreducible (see [56, Proposition 5] for example). Let H_{X_1} be the centraliser of X_1 in *H*. Then X_1 is elliptic in $(\mathfrak{m}_{P_1} \cap \mathfrak{s})(F)$ if and only if the maximal *F*-split torus in H_{X_1} is A_{P_1} .

THEOREM 9.2. Let $\mathfrak{o} \in \mathcal{O}_{rs}, P_1$ be a relatively standard parabolic subgroup of G and $X_1 \in \mathfrak{o}$ be an elliptic element in $(\mathfrak{m}_{P_1} \cap \mathfrak{s})(F)$. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we have

$$J_{\mathbf{0}}^{G}(\eta, f) = \operatorname{vol}(A_{P_{1}}^{\infty}H_{X_{1}}(F) \setminus H_{X_{1}}(\mathbb{A})) \cdot \int_{H_{X_{1}}(\mathbb{A}) \setminus H(\mathbb{A})} f(x^{-1}X_{1}x)v_{P_{1}}(x)\eta(\operatorname{Nrd}(x))dx,$$

where $v_{P_1}(x)$ is left-invariant under $H_{X_1}(\mathbb{A})$ and equals the volume of the projection onto $\mathfrak{a}_{P_1}^G$ of the convex hull of $\{-H_Q(x)\}$, where Q takes over all semi-standard parabolic subgroups of G with $M_Q = M_{P_1}$.

PROOF. Consider a relatively standard parabolic subgroup P of G and $X \in \mathfrak{m}_P(F) \cap \mathfrak{o}$ (thus P is ω -stable by Proposition 5.1). There exists an ω -stable relatively standard parabolic subgroup $P_2 \subseteq P$ and $X_2 \in (\mathfrak{m}_{P_2} \cap \mathfrak{s})(F)$ in the form of $\binom{1}{*}$ such that X_2 is conjugate to X via an element in $M_{P_H}(F)$ and the maximal F-split torus in H_{X_2} is A_{P_2} . Then any element in H(F) which conjugates X_1 and X_2 will conjugate A_{P_1} and A_{P_2} . It follows that there exists $s \in \Omega^H(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_2})$ and $m \in M_{P_H}(F)$ such that

$$X = m\omega_s X_1 \omega_s^{-1} m^{-1}$$

Suppose that $P_3 \subseteq P$ is another relatively standard parabolic subgroup, $s' \in \Omega^H(\mathfrak{a}_{P_1},\mathfrak{a}_{P_3})$ and $m' \in$ $M_{P_H}(F)$ such that

$$X = m'\omega_{s'}X_1\omega_{s'}^{-1}{m'}^{-1}.$$

Then there is $\zeta \in H_X(F)$ such that

$$m'\omega_{s'} = \zeta m\omega_s.$$

Since $H_X \subseteq M_{P_H}$, we see that

$$\omega_{s'} = \xi \omega_s$$

for some $\xi \in M_{P_H}(F)$. In sum, for any given P a relatively standard parabolic subgroup of G and $X \in \mathfrak{m}_P(F) \cap \mathfrak{o}$, there is a unique $s \in \Omega^H(\mathfrak{a}_{P_1}; P)$ such that $X = m\omega_s X_1 \omega_s^{-1} m^{-1}$ for some $m \in M_{P_H}(F)$.

For $x \in P_H(F) \setminus H(\mathbb{A})$, we obtain

$$j_{f,P,\mathfrak{o}}(x) = \sum_{X \in \mathfrak{m}_{P}(F) \cap \mathfrak{o}} \sum_{n \in N_{P_{H}}(F)} f((nx)^{-1}Xnx)$$

$$= \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{m \in M_{P_{H},\omega_{s}X_{1}\omega_{s}^{-1}}(F) \setminus M_{P_{H}}(F)} \sum_{n \in N_{P_{H}}(F)} f((mnx)^{-1}\omega_{s}X_{1}\omega_{s}^{-1}mnx)$$

$$= \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{m \in M_{P_{H},\omega_{s}X_{1}\omega_{s}^{-1}}(F) \setminus P_{H}(F)} f((mx)^{-1}\omega_{s}X_{1}\omega_{s}^{-1}mx),$$

where $M_{P_H,\omega_s X_1 \omega_s^{-1}}$ denotes the centraliser of $\omega_s X_1 \omega_s^{-1}$ in M_{P_H} . For $T \in \mathfrak{a}_0$ and $x \in H(F) \setminus H(\mathbb{A})$, we have

$$\begin{split} j_{f,\mathfrak{o}}^{T}(x) &= \sum_{\{P:\mathscr{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta \in P_{H}(F) \setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \cdot j_{f,P,\mathfrak{o}}(\delta x) \\ &= \sum_{\{P:\mathscr{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{\delta \in P_{H}(F) \setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \\ &\quad \cdot \left(\sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{m \in M_{P_{H},\omega_{s}X_{1}\omega_{s}^{-1}}(F) \setminus P_{H}(F)} f((m\delta x)^{-1}\omega_{s}X_{1}\omega_{s}^{-1}m\delta x) \right) \\ &= \sum_{\{P:\mathscr{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{\delta \in M_{P_{H},\omega_{s}X_{1}\omega_{s}^{-1}}(F) \setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \\ &\quad \cdot f((\delta x)^{-1}\omega_{s}X_{1}\omega_{s}^{-1}\delta x). \end{split}$$

Notice that the centraliser of $\omega_s X_1 \omega_s^{-1}$ in H is actually contained in M_{P_H} . We deduce that

$$\begin{aligned} j_{f,\mathfrak{o}}^{T}(x) &= \sum_{\{P:\mathscr{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{s\in\Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{\delta\in H_{\omega_{s}X_{1}\omega_{s}^{-1}}(F)\setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\delta x) - T_{P}) \cdot f((\delta x)^{-1}\omega_{s}X_{1}\omega_{s}^{-1}\delta x) \\ &= \sum_{\{P:\mathscr{P}_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{G})} \sum_{s\in\Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{\delta\in H_{X_{1}}(F)\setminus H(F)} \widehat{\tau}_{P}^{G}(H_{P}(\omega_{s}\delta x) - T_{P}) \cdot f((\delta x)^{-1}X_{1}\delta x). \end{aligned}$$

For $y \in H(\mathbb{A})$, write

$$\chi_T(y) := \sum_{\{P:\mathscr{P}_0 \subseteq P\}} (-1)^{\dim(A_P/A_G)} \sum_{s \in \Omega^H(\mathfrak{a}_{P_1};P)} \widehat{\tau}_P^G(H_P(\omega_s y) - T_P).$$

Then

$$j_{f,\mathfrak{o}}^T(x) = \sum_{\delta \in H_{X_1}(F) \setminus H(F)} f((\delta x)^{-1} X_1 \delta x) \cdot \chi_T(\delta x).$$

For sufficiently regular T, using Theorem 8.2 and the fact that $j_{f,\mathfrak{o}}^T(x)\eta(\operatorname{Nrd}(x))$ is left invariant by A_G^{∞} , we have

$$\begin{aligned} J^{G,T}_{\mathfrak{o}}(\eta,0,f) &= \int_{H(F)\backslash H(\mathbb{A})\cap G(\mathbb{A})^{1}} j^{T}_{f,\mathfrak{o}}(x)\eta(\operatorname{Nrd}(x))dx \\ &= \int_{A^{\infty}_{G}H(F)\backslash H(\mathbb{A})} \left(\sum_{\delta\in H_{X_{1}}(F)\backslash H(F)} f((\delta x)^{-1}X_{1}\delta x) \cdot \chi_{T}(\delta x)\right)\eta(\operatorname{Nrd}(x))dx. \end{aligned}$$

Hence,

(9.2.1)
$$J_{\mathbf{o}}^{G,T}(\eta,0,f) = \operatorname{vol}(A_{P_{1}}^{\infty}H_{X_{1}}(F)\backslash H_{X_{1}}(\mathbb{A})) \cdot \int_{H_{X_{1}}(\mathbb{A})\backslash H(\mathbb{A})} f(x^{-1}X_{1}x)v_{P_{1}}(x,T)\eta(\operatorname{Nrd}(x))dx.$$

where

$$v_{P_1}(x,T) := \int_{A_G^{\infty} \setminus A_{P_1}^{\infty}} \chi_T(ax) da.$$

Here we have cheated by assuming that $v_{P_1}(x, T)$ is well-defined and left-invariant under $H_{X_1}(\mathbb{A})$ in the last equality, which is explained below along with its geometric interpretation.

Let Q be a parabolic subgroup of G containing P_0 . Since $\mathscr{P}_0 \subseteq P_1$, by the characterisation in [9, p. 59], $\Omega^G(\mathfrak{a}_{P_1}; Q)$ is empty unless $\mathscr{P}_0 \subseteq Q$, in which case we have $\Omega^G(\mathfrak{a}_{P_1}; Q) = \Omega^H(\mathfrak{a}_{P_1}; Q)$ by Lemma 9.1.2). Therefore, we have

$$\chi_T(y) = \sum_{\{Q: P_0 \subseteq Q\}} (-1)^{\dim(A_Q/A_G)} \sum_{s \in \Omega^G(\mathfrak{a}_{P_1}; Q)} \widehat{\tau}_Q^G(H_Q(\omega_s y) - T_Q).$$

Compared to [3, p. 951], $v_{P_1}(x,T)$ is nothing but the restriction to $H(\mathbb{A})$ of Arthur's weight for $G(\mathbb{A})$. It showed in [2, Corollary 3.3] that the integral over a can be taken over a compact subset. From [2, Corollary 3.5], $v_{P_1}(x,T)$ equals the volume of the projection onto $\mathfrak{a}_{P_1}^G$ of the convex hull of $\{T_Q - H_Q(x)\}$, where Q takes over all semi-standard parabolic subgroups of G with $M_Q = M_{P_1}$. For $y \in H_{X_1}(\mathbb{A}) \subseteq$ $M_{P_1 \cap H}(\mathbb{A})$, the convex hull associated to $v_{P_1}(yx,T)$ is a translation of that associated to $v_{P_1}(x,T)$, so they have the same volume, i.e., $v_{P_1}(yx,T) = v_{P_1}(x,T)$. By taking constant terms of both sides of (9.2.1), we obtain the theorem.

REMARK 9.3. As mentioned in the proof of Theorem 9.2, the weights we get for regular semi-simple orbits are the restriction to $H(\mathbb{A})$ of Arthur's weights (see [3, p. 951]) for $G(\mathbb{A})$. They are also the same as those (see [39, p. 131]) appearing in the twisted trace formula for $(GL_{n,D} \times GL_{n,D}) \rtimes \sigma$, where σ acts on $GL_{n,D} \times GL_{n,D}$ by $\sigma(x,y) := (y,x)$. For P_n a standard parabolic subgroup of $GL_{n,D}$ and $P = \begin{pmatrix} \mathfrak{p}_n & \mathfrak{p}_n \\ \mathfrak{p}_n & \mathfrak{p}_n \end{pmatrix}^{\times}$ an ω -stable relatively standard parabolic subgroup of G, we may identify \mathfrak{a}_P with the σ -invariant subspace of $\mathfrak{a}_{P_n \times P_n}$. The ω -stable relatively standard parabolic subgroups of G here play the role of the σ -stable standard parabolic subgroups of $GL_{n,D} \times GL_{n,D}$, which correspond to the standard parabolic subsets of $(GL_{n,D} \times GL_{n,D}) \rtimes \sigma$ in the sense of [39, §2.7]. However, we need more (namely relatively standard) parabolic subgroups in our truncation to deal with $\mathfrak{o} \notin \mathcal{O}^{\times}$.

CHAPTER 3

An infinitesimal variant of Guo-Jacquet trace formula: the case of a central simple algebra containing a quadratic extension

We establish an infinitesimal variant of Guo-Jacquet trace formula for the case of a central simple algebra over a number field F containing a quadratic field extension E/F. It is an equality between a sum of geometric distributions on the tangent space of some symmetric space and its Fourier transform. To prove this, we need to define an analogue of Arthur's truncation and then use the Poisson summation formula. We describe the terms attached to regular semi-simple orbits as explicit weighted orbital integrals. To compare them to those for another case studied in our previous work, we state and prove the weighted fundamental lemma at the infinitesimal level by using Labesse's work on the base change for GL_n .

1. Introduction

Guo and Jacquet have proposed a conjecture [23] in order to generalise Waldspurger's famous result [50], which relates toric periods and central values of automorphic L-functions for GL_2 , to higher ranks. The approach of relative trace formulae makes it possible to reduce the conjectural comparison of periods (related to the spectral side) to the comparison of (weighted) orbital integrals (related to the geometric side) on different symmetric spaces. This approach was first adopted by Jacquet [29] to reprove Waldspurger's theorem. For higher ranks, Feigon-Martin-Whitehouse [21] obtained some partial results using a simple form of relative trace formulae. For the comparison of local orbital integrals, Guo reduced the fundamental lemma [23] to that of the base change for GL_n and Zhang proved the smooth transfer [58] by global methods.

However, an obstruction in the approach is the divergence of sums of integrals in both sides of relative trace formulae. Such a problem has already existed in the classical Arthur-Selberg trace formula and Arthur introduced a truncation process [3][4] to tackle it (see also [13] for its Lie algebra variant). We start working at the infinitesimal level (namely the tangent space of a symmetric space) for a couple of reasons. Firstly, our truncation for the tangent space is expected to be adapted to a truncation for the symmetric space. Secondly, infinitesimal trace formulae should be useful for the proof of results on the transfer (see Zhang's work [58] on the ordinary orbital integrals).

Guo-Jacquet trace formulae concern two symmetric pairs. The first one is (G', H'), where $G' := GL_{2n}$ and $H' := GL_n \times GL_n$ are reductive groups over a number field F and H' embeds diagonally in G'. Let $\mathfrak{s}' \simeq \mathfrak{gl}_n \oplus \mathfrak{gl}_n$ be the tangent space at the neutral element of the symmetric space G'/H'. We have established an infinitesimal trace formula in Chapter 2 for the action of H' on \mathfrak{s}' by conjugation. The second one denoted by (G, H) is the main object in this paper. Before introducing it, we remark that we shall work in a more general setting than the original one. The reason is that the converse direction of Guo-Jacquet conjecture was originally proposed only for n odd. In our searching for an analogue for n even, the related local conjecture of Prasad and Takloo-Bighash [44, Conjecture 1] suggests that we should consider more inner forms of G'. Some recent progress on this local conjecture has been made by Xue [55] with the help of a simple form of global relative trace formulae.

Let E/F be a quadratic extension of number fields. Suppose that $E = F(\alpha)$, where $\alpha \in E$ and $\alpha^2 \in F$. Let \mathfrak{g} be a central simple algebra over F containing E. Write \mathfrak{h} to be the centralizer of α in \mathfrak{g} . Denote by G and H the groups of invertible elements in \mathfrak{g} and \mathfrak{h} respectively. Both of them are viewed as reductive groups over F. Let $\mathfrak{s} := \{X \in \mathfrak{g} : \operatorname{Ad}(\alpha)(X) = -X\}$, where Ad denotes the adjoint action of G on \mathfrak{g} . It is the tangent space at the neutral element of the symmetric space G/H. The main global result in this paper is an infinitesimal trace formula for the action of H on \mathfrak{s} by conjugation.

Denote by \mathbb{A} the ring of adèles of F and by $H(\mathbb{A})^1$ the subset of elements in $H(\mathbb{A})$ with absolutevalue-1 reduced norm. We define a relation of equivalence on $\mathfrak{s}(F)$: two elements of $\mathfrak{s}(F)$ are equivalent if and only if they lie in the same fibre of the categorical quotient $\mathfrak{s}//H$. Denote by \mathcal{O} the set of classes of equivalence. Let f be a Bruhat-Schwartz function on $\mathfrak{s}(\mathbb{A})$. For each $\mathfrak{o} \in \mathcal{O}$ and $x \in H(F) \setminus H(\mathbb{A})$, define

$$k_{f,\mathfrak{o}}(x) := \sum_{Y \in \mathfrak{o}} f(\operatorname{Ad}(x^{-1})(Y)).$$

As mentioned, we are facing the problem that

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\setminus H(\mathbb{A})^1}k_{f,\mathfrak{o}}(x)dx$$

is divergent. We define the truncation $k_{f,\mathfrak{o}}^T(x)$ (see (4.0.1)) which is an analogue of Arthur's truncation in [3], where T is a truncation parameter in some cone $T_+ + \mathfrak{a}_{P_0}^+$ of the coroot space of H, such that the following theorem holds.

THEOREM 1.1 (see Theorem 4.2). For all $T \in T_+ + \mathfrak{a}_{P_0}^+$,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\setminus H(\mathbb{A})^1}k_{f,\mathfrak{o}}^T(x)dx$$

is absolutely convergent.

We also know the behaviour of each term (viewed as a distribution) with respect to the truncation parameter. It is even simpler than that in the case of (G', H') (cf. Theorem 1.2 in Chapter 2).

THEOREM 1.2 (see Corollary 5.3). For all $T \in T_+ + \mathfrak{a}_{P_0}^+$ and $\mathfrak{o} \in \mathcal{O}$,

$$J^T_{\mathfrak{o}}(f) := \int_{H(F) \setminus H(\mathbb{A})^1} k^T_{f,\mathfrak{o}}(x) dx$$

is a polynomial in T.

Now we can take the constant term of each term to eliminate the truncation parameter T. Denote by $J_{\mathfrak{o}}(f)$ the constant term of $J_{\mathfrak{o}}^{T}(f)$. These distributions are not invariant by $H(\mathbb{A})^{1}$ (see Proposition 6.1), but we can write the regular semi-simple terms as explicit weighted orbital integrals with the same weights as Arthur's in [3].

THEOREM 1.3 (see Theorem 9.2). Let $\mathfrak{o} \in \mathcal{O}$ be a class associated to regular semi-simple orbits, P_1 a standard parabolic subgroup of H and $Y_1 \in \mathfrak{o}$ an elliptic element with respect to P_1 (see the precise definition in Section 9). Denote by H_{Y_1} the centralizer of Y_1 in H. We have

$$J_{\mathfrak{o}}(f) = \operatorname{vol}([H_{Y_1}]) \cdot \int_{H_{Y_1}(\mathbb{A}) \setminus H(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(Y_1)) v_{P_1}(x) dx$$

where vol($[H_{Y_1}]$) is the volume associated to H_{Y_1} and $v_{P_1}(x)$ is the volume of some convex hull.

Thanks to the truncation, we solve the divergence issue in the following infinitesimal trace formula. It is a consequence of the Poisson summation formula.

THEOREM 1.4 (see Theorem 7.1). We have the equality

$$\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(f)=\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}(\hat{f}),$$

where \hat{f} (see (3.3.1)) is the Fourier transform of f.

Notice that the symmetric pairs (G, H) and (G', H') are the same after the base change to an algebraic closure of F containing E. In fact, the truncation and most proofs of the global results above are simpler than those in Chapter 2 in some sense. The simplicity results from the equality $H(\mathbb{A})^1 = H(\mathbb{A}) \cap G(\mathbb{A})^1$ here, where $G(\mathbb{A})^1$ denotes the subset of elements in $G(\mathbb{A})$ with absolute-value-1 reduced norm. Moreover, there is a bijection between the set of standard parabolic subgroups in H and the set of semi-standard parabolic subgroups in G whose intersection with H is a standard parabolic subgroup in H. One may consult Section 3.4 for more details. However, there are still some rationality issues. We shall give sufficient details and self-contained proofs here for completeness.

At the end of this paper, we hope to provide some new evidence of noninvariant comparison of Guo-Jacquet trace formulae. We shall turn to the local setting with F denoting a local field. In the comparison of geometric sides, an important case is the so-called fundamental lemma. It roughly says that some basic functions for two symmetric pairs should have associated local orbital integrals on matching orbits at almost all unramified places. Guo [23] proved it for the units of spherical Hecke algebras for Guo-Jacquet trace formulae with the help of the base change fundamental lemma for the full spherical Hecke algebras for GL_n known by Kottwitz [36, Lemma 8.8] and Arthur-Clozel [10, Theorem 4.5 in Chapter 1]. An infinitesimal version [58, Lemma 5.18] was used by Zhang to prove the smooth transfer for Guo-Jacquet trace formulae following the same philosophy of Waldspurger's work [52] on the endoscopic transfer. We would like to generalise [58, Lemma 5.18] in the weighted context.

For almost all unramified places, (G, H) is isomorphic to $(GL_{2n}, \operatorname{Res}_{E/F}GL_{n,E})$ and $\mathfrak{s}(F) \simeq \mathfrak{gl}_n(E)$. Denote by \mathcal{O}_F (resp. \mathcal{O}_E) the ring of integers in F (resp. E). For f and f' a pair of locally constant and compactly supported complex functions on $\mathfrak{s}(F)$ and $\mathfrak{s}'(F)$ respectively, we define the notion of being "strongly associated" (see the precise definition in Definition 10.4) inspired by [37, Definition III.3.2]. Roughly speaking, f and f' are said to be strongly associated if their local weighted orbital integrals are equal at matching orbits. Let f_0 and f'_0 be the characteristic functions of $\mathfrak{s}(\mathcal{O}_F) \simeq \mathfrak{gl}_n(\mathcal{O}_E)$ and $\mathfrak{s}'(\mathcal{O}_F) \simeq (\mathfrak{gl}_n \oplus \mathfrak{gl}_n)(\mathcal{O}_F)$ respectively. Because the weighted orbital integrals that we got share the same weights with those in twisted trace formulae (see Remark 9.3 and Remark 9.3 in Chapter 2), we are able to show the following result by using Labesse's work on the base change weighted fundamental lemma for the full spherical Hecke algebras for GL_n .

THEOREM 1.5 (see Theorem 8.1). For almost all unramified places, f_0 and f'_0 are strongly associated.

Acknowledgement. I would like to thank my PhD advisor Pierre-Henri Chaudouard for suggesting considering a more general case than Guo-Jacquet's original one. I have also benefited a lot from his comments on an earlier draft of this article. This work was supported by grants from Région Ile-de-France.

2. Notation

We shall use F to denote a number field in this article except for the last section where F denotes a non-archimedean local field of characteristic 0.

2.1. Roots and weights. Let F be a number field or a non-archimedean local field of characteristic 0. Suppose that H is a reductive group defined over F. Fix a minimal Levi F-subgroup M_0 of H. All the following groups are assumed to be defined over F without further mention. We call a parabolic subgroup or a Levi subgroup of H semi-standard if it contains M_0 . Fix a minimal semi-standard parabolic subgroup P_0 of H. We call a parabolic subgroup P of H standard if $P_0 \subseteq P$. For any semi-standard parabolic subgroup P of H, we usually write M_P for the Levi factor containing M_0 and N_P the unipotent radical. Denote by A_P the maximal F-split torus in the centre of M_P . Let $X(M_P)_F$ be the group of characters of M_P defined over F. Then define

$$\mathfrak{a}_P := \operatorname{Hom}_{\mathbb{Z}}(X(M_P)_F, \mathbb{R})$$

and its dual space

$$\mathfrak{a}_P^* := X(M_P)_F \otimes_{\mathbb{Z}} \mathbb{R},$$

which are both \mathbb{R} -linear spaces of dimension dim (A_P) . Notice that the restriction $X(M_P)_F \hookrightarrow X(A_P)_F$ induces an isomorphism

$$\mathfrak{a}_P^* \simeq X(A_P)_F \otimes_\mathbb{Z} \mathbb{R}.$$

Suppose that $P_1 \subseteq P_2$ are a pair of standard parabolic subgroups of H. The restriction $X(M_{P_2})_F \hookrightarrow X(M_{P_1})_F$ induces $\mathfrak{a}_{P_2}^* \hookrightarrow \mathfrak{a}_{P_1}^*$ and its dual map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. Denote by $\mathfrak{a}_{P_1}^{P_2}$ the kernel of the latter map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. The restriction $X(A_{P_1})_F \twoheadrightarrow X(A_{P_2})_F$ induces $\mathfrak{a}_{P_1}^* \twoheadrightarrow \mathfrak{a}_{P_2}^*$ and its dual map $\mathfrak{a}_{P_2} \hookrightarrow \mathfrak{a}_{P_1}$. The latter map $\mathfrak{a}_{P_2} \hookrightarrow \mathfrak{a}_{P_1}$ provides a section of the previous map $\mathfrak{a}_{P_1} \twoheadrightarrow \mathfrak{a}_{P_2}$. Thus we have decompositions

$$\mathfrak{a}_{P_1} = \mathfrak{a}_{P_2} \oplus \mathfrak{a}_{P_1}^{P_2}$$

and

$$\mathfrak{a}_{P_1}^* = \mathfrak{a}_{P_2}^* \oplus (\mathfrak{a}_{P_1}^{P_2})^*.$$

When $P_1 = P_0$, we write \mathfrak{a}_{P_1} , A_{P_1} and $\mathfrak{a}_{P_1}^{P_2}$ as \mathfrak{a}_0 , A_0 and $\mathfrak{a}_0^{P_2}$ respectively.

For a pair of standard parabolic subgroups $P_1 \subseteq P_2$ of H, write $\Delta_{P_1}^{P_2}$ for the set of simple roots for the action of A_{P_1} on $N_{P_1}^{P_2} := N_{P_1} \cap M_{P_2}$. Notice that $\Delta_{P_1}^{P_2}$ is a basis of $(\mathfrak{a}_{P_1}^{P_2})^*$. Let

$$(\widehat{\Delta}_{P_1}^{P_2})^{\vee} := \{ \varpi_{\alpha}^{\vee} : \alpha \in \Delta_{P_1}^{P_2} \}$$

be the basis of $\mathfrak{a}_{P_1}^{P_2}$ dual to $\Delta_{P_1}^{P_2}$. One has the coroot β^{\vee} associated to any $\beta \in \Delta_{P_0}^{P_2}$. For every $\alpha \in \Delta_{P_1}^{P_2}$, let α^{\vee} be the projection of β^{\vee} to $\mathfrak{a}_{P_1}^{P_2}$, where $\beta \in \Delta_{P_0}^{P_2}$ whose restriction to $\mathfrak{a}_{P_1}^{P_2}$ is α . Define

$$(\Delta_{P_1}^{P_2})^{\vee} := \{ \alpha^{\vee} : \alpha \in \Delta_{P_1}^{P_2} \}$$

which is a basis of $\mathfrak{a}_{P_1}^{P_2}$. Denote by

$$\widehat{\Delta}_{P_1}^{P_2} := \{ \varpi_\alpha : \alpha \in \Delta_{P_1}^{P_2} \}$$

the basis of $(\mathfrak{a}_{P_1}^{P_2})^*$ dual to $(\Delta_{P_1}^{P_2})^{\vee}$.

For a standard parabolic subgroup P of H, set

$$\mathfrak{a}_P^+ := \{ T \in \mathfrak{a}_P : \alpha(T) > 0, \alpha \in \Delta_P^H \}.$$

For $P_1 \subseteq P_2$ as above, define $\tau_{P_1}^{P_2}$ and $\hat{\tau}_{P_1}^{P_2}$ as the characteristic functions of

$$\{T \in \mathfrak{a}_0 : \alpha(T) > 0, \alpha \in \Delta_{P_1}^{P_2}\}$$

and

$$\{T \in \mathfrak{a}_0 : \varpi(T) > 0, \varpi \in \widehat{\Delta}_{P_1}^{P_2}\}$$

respectively.

2.2. The functions H_P and F^P . Let F be a number field. Let \mathbb{A} be the ring of adèles of F and let $|\cdot|_{\mathbb{A}}$ be the product of normalised local absolute values on the group of idèles \mathbb{A}^* . Fix a maximal compact subgroup K of $H(\mathbb{A})$ that is admissible relative to M_0 in the sense of [5, p. 9]. In this paper, we choose the standard maximal compact subgroup when $G(F) = GL_n(D)$, where D is a central division algebra over a finite field extension E of F. That is to say, $K := \prod_v K_v$ where at every non-archimedean place v of E, K_v is the group of automorphism of some lattice (see [54, p. 191]) and at every archimedean place, K_v is the unitary group with respect to some hermitian form (see [54, p. 199]). Suppose that P is a standard parabolic subgroup of H. Let H_P be the homomorphism $M_P(\mathbb{A}) \to \mathfrak{a}_P$ given by

$$\forall m \in M_P(\mathbb{A}), \langle H_P(m), \chi \rangle = \log(|\chi(m)|_{\mathbb{A}}), \chi \in X(M_P)_F.$$

Write $M_P(\mathbb{A})^1$ for the kernel of H_P and A_P^{∞} for the neutral component for the topology of \mathbb{R} -manifolds of the group of \mathbb{R} -points of the maximal \mathbb{Q} -split torus in $\operatorname{Res}_{F/\mathbb{Q}}A_P$. Then any element $x \in H(\mathbb{A})$ can be written as x = nmak, where $n \in N_P(\mathbb{A})$, $m \in M_P(\mathbb{A})^1$, $a \in A_P^{\infty}$ and $k \in K$. We can define a continuous map $H_P : H(\mathbb{A}) \to \mathfrak{a}_P$ by setting $H_P(x) := H_P(a)$ with respect to this decomposition. Notice that H_P induces an isomorphism from A_P^{∞} to \mathfrak{a}_P . If $P \subseteq Q$ are a pair of semi-standard parabolic subgroups, write

$$A_P^{Q,\infty} := A_P^{\infty} \cap M_Q(\mathbb{A})^1.$$

Then H_P also induces an isomorphism from $A_P^{Q,\infty}$ to \mathfrak{a}_P^Q .

Denote by Ω^H the Weyl group of (H, A_0) . In the cases to be considered in this paper, for every $s \in \Omega^H$, we can always choose one representative $\omega_s \in H(F) \cap K$ such that ω_s normalises A_0 . In fact, we are dealing with the restriction of scalars of inner forms of GL_n , thus we can choose Ω^H to be permutation matrices.

From the reduction theory (see [3, p. 941]), we know that there exists a real number $t_0 < 0$ and a compact subset $\omega_{P_0} \subseteq N_{P_0}(\mathbb{A})M_0(\mathbb{A})^1$ such that for any standard parabolic subgroup P of H, we have

$$H(\mathbb{A}) = P(F)\mathfrak{S}_{P_0}^P(\omega_{P_0}, t_0)$$

Here the Siegel set $\mathfrak{S}_{P_0}^P(\omega_{P_0}, t_0)$ is defined by

$$\mathfrak{S}_{P_0}^P(\omega_{P_0}, t_0) := \omega_{P_0} A_{P_0}^\infty(P, t_0) K,$$

where

$$A_{P_0}^{\infty}(P, t_0) := \{ a \in A_{P_0}^{\infty} : \alpha(H_{P_0}(a)) > t_0, \alpha \in \Delta_{P_0}^P \}$$

We shall fix such t_0 and ω_{P_0} . Moreover, we require that $(M_P(\mathbb{A}) \cap \omega_{P_0}, M_P(\mathbb{A}) \cap K, P_0 \cap M_P, t_0)$ will play the role of $(\omega_{P_0}, K, P_0, t_0)$ for any standard parabolic subgroup P of H.

Let t_0 be as above. For $T \in \mathfrak{a}_0$, define the truncated Siegel set

$$\mathfrak{S}_{P_0}^P(\omega_{P_0}, t_0, T) := \omega_{P_0} A_{P_0}^\infty(P, t_0, T) K_{P_0}^\infty(P, t_0, T) K_{P_0}^$$

where

$$A_{P_0}^{\infty}(P, t_0, T) := \{ a \in A_{P_0}^{\infty}(P, t_0) : \varpi(H_{P_0}(a) - T) \le 0, \varpi \in \widehat{\Delta}_{P_0}^P \}$$

Denote by $F_{P_0}^P(\cdot,T)$ the characteristic function of the projection of $\mathfrak{S}_{P_0}^P(\omega_{P_0},t_0,T)$ to $P(F)\setminus H(\mathbb{A})$.

2.3. Bruhat-Schwartz functions and Haar measures. Let F be a number field. Write \mathfrak{h} for the Lie algebra of H. For an F-linear subspace \mathfrak{s} of \mathfrak{h} , denote by $\mathcal{S}(\mathfrak{s}(\mathbb{A}))$ the Bruhat-Schwartz space of $\mathfrak{s}(\mathbb{A})$, namely the \mathbb{C} -linear space of functions on $\mathfrak{s}(\mathbb{A})$ generated by $f_{\infty} \otimes \chi^{\infty}$, where f_{∞} is a Schwartz function on $\mathfrak{s}(F \otimes_{\mathbb{Q}} \mathbb{R})$ and χ^{∞} is the characteristic function of an open compact subgroup of $\mathfrak{s}(\mathbb{A}^{\infty})$, where \mathbb{A}^{∞} denotes the ring of finite adèles of F.

Let P be a standard parabolic subgroup of H. For every connected subgroup V of N_P (resp. every subspace \mathfrak{v} of \mathfrak{h}), choose the unique Haar measure on $V(\mathbb{A})$ (resp. on $\mathfrak{v}(\mathbb{A})$) such that $\operatorname{vol}(V(F)\setminus V(\mathbb{A})) = 1$ (resp. $\operatorname{vol}(\mathfrak{v}(F)\setminus \mathfrak{v}(\mathbb{A})) = 1$). We also take the Haar measure on K such that $\operatorname{vol}(K) = 1$.

Fix a Euclidean norm $\|\cdot\|$ on \mathfrak{a}_0 invariant by the group Ω^H and Haar measures on subspaces of \mathfrak{a}_0 compatible with this norm. If $P \subseteq Q$ are a pair of standard parabolic subgroups, we obtain the Haar measures on A_P^{∞} and $A_P^{Q,\infty}$ via the isomorphism H_P .

Denote by $\rho_P \in (\mathfrak{a}_P^H)^*$ the half of the sum of weights (with multiplicities) for the action of A_P on \mathfrak{n}_P . We choose compatible Haar measures on $H(\mathbb{A})$ and its subgroups by requiring that for any $f \in L^1(H(\mathbb{A}))$,

$$\int_{H(\mathbb{A})} f(x)dx = \int_{N_P(\mathbb{A})} \int_{M_P(\mathbb{A})} \int_K f(nmk)e^{-2\rho_P(H_P(m))}dndmdk$$
$$= \int_{N_P(\mathbb{A})} \int_{M_P(\mathbb{A})^1} \int_{A_P^{\infty}} \int_K f(nmak)e^{-2\rho_P(H_P(a))}dndmdadk.$$

3. The symmetric pair

3.1. Groups and linear spaces. Let F be a number field and E a quadratic extension of F. Let \mathfrak{g} be a central simple algebra over F with a fixed embedding $E \to \mathfrak{g}$ as F-algebras. Denote by $\mathfrak{h} := \operatorname{Cent}_{\mathfrak{g}}(E)$ the centralizer of E in \mathfrak{g} . Then by the the double centralizer theorem (see [42, Theorem 3.1 in Chapter IV] for example), $\mathfrak{h}(F)$ is a central simple algebra over E. Write $G := \mathfrak{g}^{\times}$ and $H := \mathfrak{h}^{\times}$ for the group of invertible elements. They are considered as algebraic groups over F with Lie algebra \mathfrak{g} and \mathfrak{h} respectively.

Let $\alpha \in E$ such that $\alpha^2 \in F$ and that $E = F(\alpha)$. Denote by Ad the adjoint action of G on \mathfrak{g} . Define an involution θ on \mathfrak{g} by $\theta(X) := \operatorname{Ad}(\alpha)(X)$. Then $H = G^{\theta}$, where G^{θ} denotes the θ -invariant subgroup of G. Thus S := G/H is a symmetric space. Define an anti-involution on G by $\iota(g) := \theta(g^{-1})$. Denote by G^{ι} the ι -invariant subvariety of G. Then there is a symmetrization map $s : G \to G^{\iota}$ defined by $s(g) := \mathfrak{gl}(g)$.

LEMMA 3.1. The symmetrization map s induces a bijection $S(F) \simeq G^{\iota}(F)$.

REMARK 3.2. For the special case $(G, H) = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,E})$, where D is a quaternion algebra over F containing E, this result is included in [24, p. 282].

PROOF OF LEMMA 3.1. Since $H^1(F, H) = 1$, we have S(F) = G(F)/H(F). For $g \in G(F)$, let $s_0(g) := s(g)\alpha = \operatorname{Ad}(g)(\alpha)$. Let $G_0 := G^t\alpha = \{g \in G : g^2 = \alpha^2\}$. It suffices to prove that the map $s_0 : G(F) \to G_0(F)$ is surjective. Let $g \in G_0(F)$. Its minimal polynomial in $\mathfrak{g}(F)$ is $\lambda^2 - \alpha^2$, which is irreducible over F. Therefore, its reduced characteristic polynomial in $\mathfrak{g}(F)$ must be $(\lambda^2 - \alpha^2)^m$, where $\dim_F(\mathfrak{g}(F)) = (2m)^2$. We deduce that all elements in $G_0(F)$ are conjugate by G(F) (see [56, Theorem 9] for example). Since $\alpha \in G_0(F)$, we draw our conclusion.

One may consider the left and right translation of $H \times H$ on G and the conjugation of H on S. Denote by \mathfrak{s} the tangent space of S at the neutral element. We shall always view \mathfrak{s} as a subspace in \mathfrak{g} . Then $\mathfrak{s} = \{X \in \mathfrak{g} : \theta(X) = -X\}$ and H acts on \mathfrak{s} by conjugation.

3.2. Semi-simple elements. We say that an element Y of \mathfrak{s} is semi-simple if the orbit $\operatorname{Ad}(H)(Y)$ is Zariski closed in \mathfrak{s} . By a regular element Y of \mathfrak{s} , we mean that the centralizer H_Y of Y in H has minimal dimension.

PROPOSITION 3.3. The map $Y \mapsto Y^2$ from $\mathfrak{s}(F)$ to $\mathfrak{h}(F)$ induces an injection from the set of H(F)conjugacy classes of semi-simple elements in $\mathfrak{s}(F)$ to the set of conjugacy classes of semi-simple elements in $\mathfrak{h}(F)$.

REMARK 3.4. In the special case $(G, H) = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,E})$, where D is a quaternion algebra over F containing E, this map plays the role of the norm map (see [10, §1 in Chapter 1]).

PROOF OF PROPOSITION 3.3. Let $\chi_{\mathfrak{g},F}(X)$ be the reduced characteristic polynomial of $X \in \mathfrak{g}(F)$ and $\chi_{\mathfrak{h},E}(X^*)$ the reduced characteristic polynomial of $X^* \in \mathfrak{h}(F)$ (viewed as a central simple algebra over E). After the base change to an algebraic closure of F containing E, the embedding $\mathfrak{h} \subseteq \mathfrak{g}$ is identical to the diagonal embedding $\mathfrak{h}' := \mathfrak{gl}_m \oplus \mathfrak{gl}_m \subseteq \mathfrak{g}' := \mathfrak{gl}_{2m}$ and $\mathfrak{s} \subseteq \mathfrak{g}$ becomes $\mathfrak{s}' := \left\{ \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} : A, B \in \mathfrak{gl}_m \right\} \subseteq \mathfrak{g}'$, where m denotes the degree of $\mathfrak{h}(F)$ (viewed as a central simple algebra over E). Let H' be

the group of invertible elements in \mathfrak{h}' , which is viewed as an algebraic group $GL_m \times GL_m$ over F. Since

$$\det\left(\lambda I_{2m} - \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix}\right) = \det(\lambda^2 I_m - AB),$$

we see that for $Y \in \mathfrak{s}(F) \subseteq \mathfrak{g}(F)$,

 $\chi_{\mathfrak{g},F}(Y)(\lambda) = \chi_{\mathfrak{h},E}(Y^2)(\lambda^2),$

which implies that $\chi_{\mathfrak{h},E}(Y^2)$ is actually defined over F.

It is known that the semi-simple conjugacy classes in $\mathfrak{h}(F)$ are uniquely determined by $\chi_{\mathfrak{h},E}$ (see [56, Theorem 9] for example). Thus it suffices to prove that the semi-simple H(F)-conjugacy classes in $\mathfrak{s}(F)$ are uniquely determined by $\chi_{\mathfrak{g},F}$. From [31, Proposition 2.1], we know that the semi-simple H-conjugacy classes in $\mathfrak{s}(F)$ are uniquely determined by $\chi_{\mathfrak{g},F}$. Therefore, we reduce ourselves to proving that each semi-simple H-conjugacy class in $\mathfrak{s}(F)$ contains a unique H(F)-conjugacy class.

For a semi-simple element $Y \in \mathfrak{s}(F)$, the H(F)-orbits in Ad(H)(Y) are parametrized by

$$\ker[H^1(F, H_Y) \to H^1(F, H)] = H^1(F, H_Y),$$

where H_Y is the centralizer of Y in H. By [47, Exercice 2 in p. 160], we obtain

$$H^1(F, H_Y) = 1,$$

which completes our proof.

3.3. Invariants. Denote by \mathfrak{c} the affine space \mathbf{A}^m , where m denotes the degree of $\mathfrak{h}(F)$ (viewed as a central simple algebra over E). Define a morphism $\pi : \mathfrak{s} \to \mathfrak{c}$ which is contant on H-orbits by mapping $Y \in \mathfrak{s}$ to the coefficients of the reduced polynomial of $Y \in \mathfrak{g}$. In fact, we see that the coefficients in odd degrees vanish for $Y \in \mathfrak{s}$ from the proof of Proposition 3.3. On F-points, alternatively, π is given by mapping $Y \in \mathfrak{s}(F)$ to the coefficients of the reduced polynomial of $Y^2 \in \mathfrak{h}(F)$ (viewed as a central simple algebra over E).

PROPOSITION 3.5. The pair (\mathfrak{c}, π) defines a categorical quotient of \mathfrak{s} by H over F.

PROOF. By the proof of Proposition 3.3 in Chapter 2, after the base change to an algebraic closure \overline{F} of F containing E, the pair $(\mathfrak{c}_{\overline{F}}, \pi_{\overline{F}})$ defines a categorical quotient of $\mathfrak{s}_{\overline{F}}$ by $H_{\overline{F}}$. That is to say, we have an isomorphism of \overline{F} -algebras $\overline{F}[\mathfrak{c}] \simeq \overline{F}[\mathfrak{s}]^H$ dual to $\pi_{\overline{F}}$. But this isomorphism is obtained from the base change of a morphism of F-algebras $F[\mathfrak{c}] \rightarrow F[\mathfrak{s}]^H$ dual to π . By Galois descent, the latter morphism is necessarily an isomorphism of F-algebras. Then the pair (\mathfrak{c}, π) defines a categorical quotient of \mathfrak{s} by H over F.

REMARK 3.6. The morphism π is surjective as a morphism of algebraic varieties (see the proof of Proposition 3.3 in Chapter 2) but not surjective on the level of *F*-points.

We define a relation of equivalence on $\mathfrak{s}(F)$ using the categorical quotient (\mathfrak{c}, π) , where two elements are in the same class if and only if they have the same image under π . We denote by \mathcal{O} the set of equivalent classes for this relation. From the proof of Proposition 3.3, we see that two semi-simple elements of $\mathfrak{s}(F)$ belong to the same class of \mathcal{O} if and only if they are conjugate by H(F). Denote by \mathcal{O}_{rs} the subset of \mathcal{O} formed by $Y \in \mathfrak{s}(F)$ such that $\chi_{\mathfrak{h},E}(Y^2)$ is separable and $\chi_{\mathfrak{h},E}(Y^2)(0) \neq 0$, where $\chi_{\mathfrak{h},E}$ denotes the reduced polynomial of an element in $\mathfrak{h}(F)$ (viewed as a central simple algebra over E). By Proposition 3.2 in Chapter 2 and the base change to an algebraic closure of F containing E, we see that each class in \mathcal{O}_{rs} is a regular semi-simple H(F)-orbit in $\mathfrak{s}(F)$.

3.4. Explicit description of $H \hookrightarrow G$. First of all, we would like to describe the symmetric pair (G, H) in a more explicit way. By the Noether-Skolem theorem (see [42, Theorem 2.10 of Chapter IV] for example), the embedding $E \to \mathfrak{g}(F)$ is unique up to conjugation by an element of G(F). From the Wedderburn-Artin theorem, we know that G is isomorphic to $GL_{n,D}$, which denotes the reductive group over F whose F-points are $GL_n(D)$, for some central division algebra D over F. We recall that n is called the capacity of $\mathfrak{g}(F)$ and we denote it by $\operatorname{capa}(\mathfrak{g}(F))$. Let d be the degree of D, i.e., $\dim_F(D) = d^2$. Since there is an embedding $E \to \mathfrak{g}(F)$ as F-algebras, we know that nd is even.

PROPOSITION 3.7. Up to conjugation by G(F), the embedding $H \hookrightarrow G$ is reduced to one of the two cases below.

Case I: if there is an embedding $E \to D$ as F-algebras, then the embedding $H \hookrightarrow G$ is isomorphic to $\operatorname{Res}_{E/F}GL_{n,D'} \hookrightarrow GL_{n,D}$ up to conjugation by G(F). Here $D' := \operatorname{Cent}_D(E)$ denotes the centralizer of E in D and is a central division algebra over E.

Case II: if there is no embedding $E \to D$ as F-algebras, then the embedding $H \hookrightarrow G$ is isomorphic to $\operatorname{Res}_{E/F}GL_{\frac{n}{2},D\otimes_F E} \hookrightarrow GL_{n,D}$ up to conjugation by G(F). Here $D \otimes_F E$ is a central division algebra over E.

PROOF. Case I: there is an embedding $E \to D$ as *F*-algebras. This case is a direct consequence of the Noether-Skolem theorem. By the double centralizer theorem, we know that D' is a central division algebra over *E*.

Case II: there is no embedding $E \to D$ as *F*-algebras. By [48, Theorem 1.1.2], when *nd* is even, there is an embedding $E \to \mathfrak{g}(F)$ as *F*-algebras if and only if $n \cdot \operatorname{capa}(D \otimes_F E)$ is even, where $\operatorname{capa}(D \otimes_F E)$ denotes the capacity of the central simple algebra $D \otimes_F E$ over *E* (see [42, Proposition 2.15 in Chapter IV] for example). Additionally, from [48, Theorem 1.1.3], we show that $\operatorname{capa}(D \otimes_F E) \leq [E : F] = 2$. In this case, there are two possibilities.

- (1) d is even. By [48, Theorem 1.1.2], $\operatorname{capa}(D \otimes_F E)$ is odd, so $\operatorname{capa}(D \otimes_F E) = 1$. Since $n \cdot \operatorname{capa}(D \otimes_F E)$ is even, we know that n is even.
- (2) d is odd. Since nd is even, we see that n is even. Besides, from [48, Theorem 1.1.3], we also deduce that capa $(D \otimes_F E) = 1$.

In sum, we have shown that n is even and that $D \otimes_F E$ is a central division algebra over E. The tensor of $\mathfrak{gl}_{\frac{n}{2},D}$ and a fixed embedding $\operatorname{Res}_{E/F}\mathfrak{gl}_{1,E} \to \mathfrak{gl}_2$ gives the indicated way to embed \mathfrak{h} to \mathfrak{g} . By the Noether-Skolem theorem, such an embedding is unique up to conjugation by G(F).

Next, we describe the correspondence of some parabolic subgroups in H and G in both cases.

Case I: $(G, H) = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,D'})$, where $D' := \operatorname{Cent}_D(E)$. We denote by $M_0 \simeq (\operatorname{Res}_{E/F}\mathbb{G}_{m,D'})^n$ the subgroup of diagonal elements in H, which is a minimal Levi F-subgroup of H, and by $M_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^n$ the subgroup of diagonal elements in G, which is a minimal Levi F-subgroup of G. We also fix P_0 the subgroup of upper triangular elements in H, which is a minimal parabolic F-subgroup of H. There is a bijection $P \mapsto \widetilde{P}$ between the set of standard parabolic subgroups P (namely $P_0 \subseteq P$) in H and the set of semi-standard parabolic subgroups \widetilde{P} (namely $M_{\widetilde{0}} \subseteq \widetilde{P}$) in G which contain P_0 . In this case, the latter is exactly the set of standard parabolic subgroups (namely containing $\widetilde{P_0}$ the subgroup of upper triangular elements in G) of G. We shall always write \widetilde{P} for the image of P under this bijection. Notice that $P = \widetilde{P} \cap H$ and that we can identify A_P with $A_{\widetilde{P}}$.

Case II: $(G, H) = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{\frac{n}{2},D\otimes_FE})$. We denote by $M_0 \simeq (\operatorname{Res}_{E/F}\mathbb{G}_{m,D\otimes_FE})^{\frac{n}{2}}$ the subgroup of diagonal elements in H, which is a minimal Levi F-subgroup of H, and by $M_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^n$ the subgroup of diagonal elements in G, which is a minimal Levi F-subgroup of G. We also fix P_0 the subgroup of upper triangular elements in H, which is a minimal parabolic F-subgroup of H. There is a bijection $P \mapsto \widetilde{P}$ between the set of standard parabolic subgroups P (namely $P_0 \subseteq P$) in H and the set of semi-standard parabolic subgroups \widetilde{P} (namely $M_{\widetilde{0}} \subseteq \widetilde{P}$) in G which contain P_0 . In this case, the latter is a subset of the set of standard parabolic subgroups (namely containing the subgroup of upper triangular elements in G) of G. We shall always write \widetilde{P} for the image of P under this bijection. Notice that $P = \widetilde{P} \cap H$ and that we can identify A_P with $A_{\widetilde{p}}$.

PROPOSITION 3.8. Let P be a standard parabolic subgroup of H. For all $Y \in (\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$ and $U \in (\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(F)$, we have

$$\pi(Y) = \pi(Y + U).$$

PROOF. It is obvious, since the reduced characteristic polynomial of $Y + U \in \mathfrak{g}$ is equal to that of $Y \in \mathfrak{g}$.

COROLLARY 3.9. Let P be a standard parabolic subgroup of H and $\mathfrak{o} \in \mathcal{O}$. For all subsets $S_1 \subseteq (\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$ and $S_2 \subseteq (\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(F)$, we have $\mathfrak{o} \cap (S_1 \oplus S_2) = (\mathfrak{o} \cap S_1) \oplus S_2$.

Let P be a standard parabolic subgroup of H. We denote by $\Phi(A_0, \mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})$ (resp. $\Phi(A_0, \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})$) the set of weights of A_0 in $\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s}$ (resp. $\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}$). We also denote by $\Phi(A_0, \mathfrak{m}_P)$ (resp. $\Phi(A_0, \mathfrak{n}_P)$) the set of weights of A_0 in \mathfrak{m}_P (resp. \mathfrak{n}_P).

PROPOSITION 3.10. For any standard parabolic subgroup P of H, we have

 $\Phi(A_0,\mathfrak{m}_{\widetilde{P}}\cap\mathfrak{s})=\Phi(A_0,\mathfrak{m}_P)$

and

$$\Phi(A_0,\mathfrak{n}_{\widetilde{P}}\cap\mathfrak{s})=\Phi(A_0,\mathfrak{n}_P)$$

Moreover, each weight of A_0 has the same multiplicity in $\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s}$ (resp. $\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}$) and \mathfrak{m}_P (resp. \mathfrak{n}_P).

PROOF. It is obvious for both of Case I and Case II described above.

For P a standard parabolic subgroup of H, let $\rho_{P,\mathfrak{s}}$ (resp. ρ_P) denote the half of the sum of weights (with multiplicities) of A_0 in $\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}$ (resp. \mathfrak{n}_P).

COROLLARY 3.11. For any standard parabolic subgroup P of H, we have

$$\rho_{P,\mathfrak{s}} = \rho_P$$

At the end of this subsection, we point out a non-canonical *F*-linear isomorphism between \mathfrak{h} and \mathfrak{s} which will be useful for some technical problems. We have chosen an element $\alpha \in E$ in Section 3.1. Let $\tau \in D^{\times}$ in **Case I** (resp. $\tau \in GL_2(D)$ in **Case II**) be an element such that $\operatorname{Ad}(\alpha)(\tau) = -\tau$.

PROPOSITION 3.12. There is a non-canonical isomorphism induced by multiplication by τ between \mathfrak{h} and \mathfrak{s} as free D'-modules (resp. $D \otimes_F E$ -modules), i.e.,

$$\mathfrak{s} = \mathfrak{h}\tau = \tau\mathfrak{h}$$

Moreover, for any standard parabolic subgroup P of H, we have

$$\mathfrak{m}_{\widetilde{P}}\cap\mathfrak{s}=\mathfrak{m}_{P}\tau=\tau\mathfrak{m}_{P}$$

and

$$\mathfrak{n}_{\widetilde{P}}\cap\mathfrak{s}=\mathfrak{n}_{P}\tau=\tau\mathfrak{n}_{P}.$$

PROOF. It is obvious for both of **Case I** and **Case II** described above.

3.5. Fourier transform. Fix a nontrivial unitary character Ψ of \mathbb{A}/F . Let $\langle \cdot, \cdot \rangle$ be the $H(\mathbb{A})$ -invariant bilinear form on $\mathfrak{s}(\mathbb{A})$ defined by

(3.5.1)
$$\forall Y_1, Y_2 \in \mathfrak{s}(\mathbb{A}), \langle Y_1, Y_2 \rangle := \operatorname{Trd}_{\mathfrak{g},F}(Y_1Y_2),$$

where $\operatorname{Trd}_{\mathfrak{g},F}(Y_1Y_2)$ denotes the reduced trace of $Y_1Y_2 \in \mathfrak{g}(\mathbb{A})$. It is non-degenerate, which can be seen after the base change to an algebraic closure of F. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, its Fourier transform \hat{f} is defined by

(3.5.2)
$$\forall \widehat{Y} \in \mathfrak{s}(\mathbb{A}), \widehat{f}(\widehat{Y}) := \int_{\mathfrak{s}(\mathbb{A})} f(Y) \Psi(\langle Y, \widehat{Y} \rangle) dY.$$

4. Integrability of the modified kernel

Let $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, P be a standard parabolic subgroup of H and $\mathfrak{o} \in \mathcal{O}$. For $x \in M_P(F)N_P(\mathbb{A}) \setminus H(\mathbb{A})$, define

$$k_{f,P,\mathfrak{o}}(x) := \sum_{Y \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(Y+U)) dU.$$

For $T \in \mathfrak{a}_0$ and $x \in H(F) \setminus H(\mathbb{A})$, define

$$(4.0.1) k_{f,\mathfrak{o}}^T(x) := \sum_{\{P:P_0 \subseteq P\}} (-1)^{\dim(A_P/A_H)} \sum_{\delta \in P(F) \setminus H(F)} \widehat{\tau}_P^H(H_P(\delta x) - T) \cdot k_{f,P,\mathfrak{o}}(\delta x).$$

By [3, Lemma 5.1], we know that the sum over $\delta \in P(F) \setminus H(F)$ is finite.

LEMMA 4.1. There is a $T_+ \in \mathfrak{a}_{P_0}^+$ such that for all standard parabolic subgroup P of $H, T \in T_+ + \mathfrak{a}_{P_0}^+$ and $x \in H(\mathbb{A})$, we have

$$\sum_{P_1:P_0 \subseteq P_1 \subseteq P\}} \sum_{\delta_1 \in P_1(F) \setminus P(F)} F^{P_1}(\delta_1 x, T) \tau^P_{P_1}(H_{P_1}(\delta_1 x) - T) = 1$$

PROOF. This is [3, Lemma 6.4].

{

We shall fix such a T_+ .

THEOREM 4.2. For all $T \in T_+ + \mathfrak{a}_{P_0}^+$,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\backslash H(\mathbb{A})^1}|k_{f,\mathfrak{o}}^T(x)|dx<\infty$$

PROOF. Let $P_1 \subseteq P_2$ be a pair of standard parabolic subgroups of H. As in [3, §6], for $T_1 \in \mathfrak{a}_{P_1}$, define the characteristic function

$$\tau_{P_1}^{P_2}(T_1) := \sum_{\{Q: P_2 \subseteq Q\}} (-1)^{\dim(A_{P_2}/A_Q)} \tau_{P_1}^Q(T_1) \hat{\tau}_Q^H(T_1).$$

Recall that for $P \supseteq P_1$ a standard parabolic subgroup of H, we have

$$\tau_{P_1}^P(T_1)\widehat{\tau}_P^H(T_1) = \sum_{\{P_2: P \subseteq P_2\}} \sigma_{P_1}^{P_2}(T_1)$$

For $x \in P_1(F) \setminus H(\mathbb{A})$, we write

$$\chi^T_{P_1,P_2}(x) := F^{P_1}(x,T)\sigma^{P_2}_{P_1}(H_{P_1}(x)-T)$$

and

$$k_{P_1,P_2,\mathfrak{o}}(x) := \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} k_{f,P,\mathfrak{o}}(x).$$

By Lemma 4.1 and the left invariance of H_P and $k_{f,P,\mathfrak{o}}$ by P(F), we obtain

$$k_{f,\mathfrak{o}}^{T}(x) = \sum_{\{P_{1},P_{2}:P_{0}\subseteq P_{1}\subseteq P_{2}\}} \sum_{\delta\in P_{1}(F)\backslash H(F)} \chi_{P_{1},P_{2}}^{T}(\delta x) k_{P_{1},P_{2},\mathfrak{o}}(\delta x).$$

Thus

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{H(F)\backslash H(\mathbb{A})^1}|k_{f,\mathfrak{o}}^T(x)|dx\leq \sum_{\mathfrak{o}\in\mathcal{O}}\sum_{\{P_1,P_2:P_0\subseteq P_1\subseteq P_2\}}\int_{P_1(F)\backslash H(\mathbb{A})^1}\chi_{P_1,P_2}^T(x)|k_{P_1,P_2,\mathfrak{o}}(x)|dx.$$

Then we only need to show that for any pair of standard parabolic subgroups $P_1 \subseteq P_2$ of H,

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_1(F)\setminus H(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)|k_{P_1,P_2,\mathfrak{o}}(x)|dx<\infty.$$

If $P_1 = P_2 \neq H$, by [3, Lemma 6.1], we have $\sigma_{P_1}^{P_2} = 0$ and then $\chi_{P_1,P_2}^T = 0$, so the integration vanishes. If $P_1 = P_2 = H$, since $F^H(\cdot, T)$ is a characteristic function with compact support in $H(F) \setminus H(\mathbb{A})^1$, the integration is convergent. Hence, we reduce ourselves to proving the following proposition.

PROPOSITION 4.3. Let $f \in S(\mathfrak{s}(\mathbb{A}))$ and $P_1 \subsetneq P_2$ be a pair of standard parabolic subgroups of H. Suppose that ϵ_0 and N are two arbitrary but fixed positive real numbers. Then there exists a constant C such that

$$\sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_1(F)\setminus H(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)|k_{P_1,P_2,\mathfrak{o}}(x)|dx\leq Ce^{-N\|T\|}$$

for all $T \in T_+ + \mathfrak{a}_{P_0}^+$ satisfying $\alpha(T) \geq \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}^H$.

For $x \in H(F) \setminus H(\mathbb{A})$, define

$$k_{f,H}(x) := \sum_{\mathfrak{o} \in \mathcal{O}} k_{f,H,\mathfrak{o}}(x) = \sum_{Y \in \mathfrak{s}(F)} f(\operatorname{Ad}(x^{-1})(Y))$$

and

(4.0.2)
$$k_f^T(x) := \sum_{\mathbf{o} \in \mathcal{O}} k_{f,\mathbf{o}}^T(x)$$

_	_	_
-	-	

COROLLARY 4.4. Let $f \in S(\mathfrak{s}(\mathbb{A}))$. For two arbitrary but fixed positive real number ϵ_0 and N, there exists a constant C such that

$$\int_{H(F)\setminus H(\mathbb{A})^1} |k_f^T(x) - F^H(x,T)k_{f,H}(x)| dx \le Ce^{-N||T|}$$

for all $T \in T_+ + \mathfrak{a}_{P_0}^+$ satisfying $\alpha(T) \ge \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}^H$.

PROOF OF PROPOSITION 4.3. Let *P* be any standard parabolic subgroup of *H* such that $P_1 \subseteq P \subseteq P_2$. For any $Y \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o}$, there is a unique standard parabolic subgroup *R* of *H* such that $P_1 \subseteq R \subseteq P$ and $Y \in (\mathfrak{m}_{\widetilde{P}}(F) \cap \widetilde{\mathfrak{r}}(F) \cap \mathfrak{o}) - \left(\bigcup_{P_1 \subseteq O \subseteq R} \mathfrak{m}_{\widetilde{P}}(F) \cap \widetilde{\mathfrak{q}}(F) \cap \mathfrak{o}\right)$. We denote

$$\widetilde{\mathfrak{m}}_{\widetilde{P_{1}}}^{\widetilde{R}} := \mathfrak{m}_{\widetilde{R}} - \left(\bigcup_{\{Q:P_{1} \subseteq Q \subsetneq R\}} \mathfrak{m}_{\widetilde{R}} \cap \widetilde{\mathfrak{q}}\right)$$

and

$$\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} := \mathfrak{n}_{\widetilde{R}} \cap \mathfrak{m}_{\widetilde{P}}$$

From Corollary 3.9, we get

$$(\mathfrak{m}_{\widetilde{P}}(F) \cap \widetilde{\mathfrak{r}}(F) \cap \mathfrak{o}) - \left(\bigcup_{P_1 \subseteq Q \subsetneq R} \mathfrak{m}_{\widetilde{P}}(F) \cap \widetilde{\mathfrak{q}}(F) \cap \mathfrak{o}\right) = (\widetilde{\mathfrak{m}}_{\widetilde{P}_1}^{\widetilde{R}}(F) \cap \mathfrak{o}) \oplus ((\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)).$$

Thus

$$\begin{split} k_{f,P,\mathfrak{o}}(x) &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(Y+U)) dU \\ &= \sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{\tilde{P}_1}^{\tilde{R}}(F) \cap \mathfrak{o}} \sum_{Y \in (\mathfrak{n}_{\tilde{R}}^{\tilde{P}} \cap \mathfrak{s})(F)} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(\xi+Y+U)) dU. \end{split}$$

We write $\overline{\widetilde{P}}$ for the parabolic subgroup of G opposite to \widetilde{P} and

$$\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} := \mathfrak{n}_{\overline{\widetilde{R}}} \cap \mathfrak{m}_{\widetilde{P}}.$$

Notice that the restriction of $\langle \cdot, \cdot \rangle$ (see (3.5.1)) to $((\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})) \times ((\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A}))$ is also non-degenerate. For any $\xi \in (\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A})$, applying the Poisson summation formula to the Bruhat-Schwartz function $\int_{(\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(\xi + \cdot + U)) dU$, we have

$$\sum_{Y \in (\mathfrak{n}_{\tilde{R}}^{\tilde{P}} \cap \mathfrak{s})(F)} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(\xi + Y + U)) dU = \sum_{\widehat{Y} \in (\overline{\mathfrak{n}}_{\tilde{R}}^{\tilde{P}} \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}),$$

where the partial Fourier transform $\Phi_{\xi}^{x,R}$ of $\int_{(\mathfrak{n}_{\tilde{\rho}}\cap\mathfrak{s})(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(\xi+\cdot+U)) dU$ is defined by

$$\forall \widehat{Y} \in (\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A}), \Phi_{\xi}^{x,R}(\widehat{Y}) := \int_{(\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})} \left(\int_{(\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(\xi + Y + U)) dU \right) \Psi(\langle Y, \widehat{Y} \rangle) dY.$$

Since $\langle U, \widehat{Y} \rangle = 0$ for $U \in (\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})$ and $\widehat{Y} \in (\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A})$, as well as $\mathfrak{n}_{\widetilde{R}} = \mathfrak{n}_{\widetilde{P}} \oplus \mathfrak{n}_{\widetilde{R}}^{\widetilde{P}}$, we have

$$\forall \widehat{Y} \in (\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(\mathbb{A}), \Phi_{\xi}^{x,R}(\widehat{Y}) = \int_{(\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(\xi + U)) \Psi(\langle U, \widehat{Y} \rangle) dU,$$

which is actually independent of P.

In sum,

$$k_{f,P,\mathfrak{o}}(x) = \sum_{\{R: P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{\overline{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{\widehat{Y} \in (\overline{\mathfrak{n}}_{\overline{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}).$$

Then we have

$$\begin{split} k_{P_1,P_2,\mathfrak{o}}(x) &= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} k_{f,P,\mathfrak{o}}(x) \\ &= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} \left(\sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{\widehat{Y} \in (\widetilde{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}) \right) \\ &= \sum_{\{R:P_1 \subseteq R \subseteq P_2\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \left(\sum_{\{P:R \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} \sum_{\widehat{Y} \in (\widetilde{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}) \right). \end{split}$$

For P_3 a standard parabolic subgroup of H containing R, denote

$$(\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P_3}})' := \overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P_3}} - \left(\bigcup_{\{Q:R\subseteq Q\subsetneq P_3\}} \overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{Q}}\right).$$

We write

$$\begin{split} &\sum_{\{P:R\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_H)} \sum_{\widehat{Y}\in(\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P}}\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}) \\ &= \sum_{\{P:R\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_H)} \sum_{\{P_3:R\subseteq P_3\subseteq P\}} \sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_3}})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}) \\ &= (-1)^{\dim(A_{P_2}/A_H)} \sum_{\{P_3:R\subseteq P_3\subseteq P_2\}} \left(\sum_{\{P:P_3\subseteq P\subseteq P_2\}} (-1)^{\dim(A_P/A_{P_2})} \right) \sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_3}})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}) \\ &= (-1)^{\dim(A_{P_2}/A_H)} \sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_2}})'\cap\mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}), \end{split}$$

where we have used [3, Proposition 1.1] in the last equality. Then

$$k_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_H)} \sum_{\{R: P_1 \subseteq R \subseteq P_2\}} \sum_{\xi \in \widetilde{\mathfrak{m}}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{\widehat{Y} \in ((\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P_2}})' \cap \mathfrak{s})(F)} \Phi_{\xi}^{x,R}(\widehat{Y}).$$

Now we get

$$\begin{split} &\sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_1(F)\backslash H(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)|k_{P_1,P_2,\mathfrak{o}}(x)|dx\\ &\leq \sum_{\mathfrak{o}\in\mathcal{O}}\int_{P_1(F)\backslash H(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)\left(\sum_{\{R:P_1\subseteq R\subseteq P_2\}}\sum_{\xi\in\widetilde{\mathfrak{m}}_{\widehat{P_1}}^{\widehat{R}}(F)\cap\mathfrak{o}}\sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_2}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|\right)dx\\ &=\sum_{\{R:P_1\subseteq R\subseteq P_2\}}\int_{P_1(F)\backslash H(\mathbb{A})^1}\chi^T_{P_1,P_2}(x)\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\widehat{P_1}}^{\widehat{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_2}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|dx.\end{split}$$

Thus it suffices to bound

(4.0.3)
$$\int_{P_1(F)\backslash H(\mathbb{A})^1} \chi^T_{P_1,P_2}(x) \sum_{\xi \in (\widetilde{\mathfrak{m}}_{\widetilde{P_1}}^{\widetilde{R}} \cap \mathfrak{s})(F)} \sum_{\widehat{Y} \in ((\overline{\mathfrak{m}}_{\widetilde{R}}^{\widetilde{P_2}})' \cap \mathfrak{s})(F)} |\Phi_{\xi}^{x,R}(\widehat{Y})| dx$$

for any fixed standard parabolic subgroup R of H such that $P_1 \subseteq R \subseteq P_2$.
Denote $A_{P_1}^{H,\infty} := A_{P_1}^{\infty} \cap H(\mathbb{A})^1$. By Iwasawa decomposition and our choice of measures, we have

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}}\chi_{P_{1},P_{2}}^{T}(x)\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\widehat{P_{1}}}^{\widehat{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|dx\\ &=\int_{K}\int_{A_{P_{1}}^{H,\infty}}\int_{M_{P_{1}}(F)\backslash M_{P_{1}}(\mathbb{A})^{1}}\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})}F^{P_{1}}(m_{1},T)\sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a_{1})-T)\\ &\cdot\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\widehat{P_{1}}}^{\widehat{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\overline{\mathfrak{n}}_{\widehat{R}}^{\widehat{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{n_{1}m_{1}a_{1}k,R}(\widehat{Y})|e^{-2\rho_{P_{1}}(H_{P_{0}}(a_{1}))}dn_{1}dm_{1}da_{1}dk \end{split}$$

Because only those m_1 satisfying $F^{P_1}(m_1,T) \neq 0$ contribute to the integration, we can restrict the integration over those having representatives in $(N_{P_0}(\mathbb{A})M_{P_0}(\mathbb{A})^1A_{P_0}^{P_1,\infty}(t_0,T)K) \cap M_{P_1}(\mathbb{A})^1$, where $A_{P_0}^{P_1,\infty}(t_0,T) := A_{P_0}^{\infty}(P_1,t_0,T) \cap M_{P_1}(\mathbb{A})^1$. Then

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\overline{P_{1}}}^{\overline{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\widetilde{\mathfrak{m}}_{\overline{R}}^{\overline{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|dx\\ \leq &c_{1}\int_{K}\int_{[cpt\subseteq M_{P_{0}}(\mathbb{A})^{1}]}\int_{A_{P_{0}}^{P_{1},\infty}(t_{0},T)}\int_{A_{P_{1}}^{H,\infty}}\int_{[cpt\subseteq N_{P_{0}}(\mathbb{A})]}\int_{[cpt\subseteq N_{P_{2}}(\mathbb{A})]}\sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a_{1})-T)\\ &\cdot\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\overline{P_{1}}}^{\overline{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\widetilde{\mathfrak{m}}_{\overline{R}}^{\overline{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{n_{2}na_{1}amk,R}(\widehat{Y})|e^{-2\rho_{P_{0}}(H_{P_{0}}(a_{1}a))}dn_{2}dnda_{1}dadmdk, \end{split}$$

where $c_1 := \operatorname{vol}(K \cap M_{P_1}(\mathbb{A})^1)$ is a constant independent of T. Here we use the notation $[cpt \subseteq *]$ for denoting a compact subset in * independent of T.

We claim that for $n_2 \in N_{P_2}(\mathbb{A})$,

$$\Phi_{\xi}^{n_2 x, R}(\widehat{Y}) = \Phi_{\xi}^{x, R}(\widehat{Y}).$$

In fact, let $U_2 := \operatorname{Ad}(n_2^{-1})(\xi) - \xi$. Then

$$\begin{split} \Phi_{\xi}^{n_{2}x,R}(\widehat{Y}) &= \int_{(\mathfrak{n}_{\widehat{R}}\cap\mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(n_{2}x)^{-1}(\xi+U))\Psi(\langle U,\widehat{Y}\rangle)dU \\ &= \int_{(\mathfrak{n}_{\widehat{R}}\cap\mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(\xi+U_{2}+\mathrm{Ad}(n_{2}^{-1})(U)))\Psi(\langle U,\widehat{Y}\rangle)dU \end{split}$$

As both of U_2 and $\operatorname{Ad}(n_2^{-1})(U) - U$ belong to $(\mathfrak{n}_{\widetilde{P_2}} \cap \mathfrak{s})(\mathbb{A})$, we get

$$\langle U_2 + \operatorname{Ad}(n_2^{-1})(U) - U, \widehat{Y} \rangle = 0.$$

Hence

$$\Phi_{\xi}^{n_2x,R}(\widehat{Y}) = \int_{(\mathfrak{n}_{\widetilde{R}}\cap\mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(\xi + U_2 + \mathrm{Ad}(n_2^{-1})(U)))\Psi(\langle U_2 + \mathrm{Ad}(n_2^{-1})(U), \widehat{Y}\rangle) dU.$$

Since the change of variables $U_2 + \operatorname{Ad}(n_2^{-1})(U) \mapsto U$ does not change the Haar measure, we proved our claim.

By this claim, we have

$$\Phi_{\xi}^{n_2na_1amk,R}(\widehat{Y}) = \Phi_{\xi}^{na_1amk,R}(\widehat{Y}) = \Phi_{\xi}^{(a_1a)(a_1a)^{-1}n(a_1a)mk,R}(\widehat{Y}).$$

Applying change of variables $\operatorname{Ad}(a_1a)^{-1}(U) \mapsto U$ and the fact that

$$\langle U, \widehat{Y} \rangle = \langle \operatorname{Ad}(a_1 a)^{-1}(U), \operatorname{Ad}(a_1 a)^{-1}(\widehat{Y}) \rangle,$$

we deduce that

$$\Phi_{\xi}^{n_2na_1amk,R}(\widehat{Y}) = e^{2\rho_{R,\mathfrak{s}}(H_{P_0}(a_1a))} \Phi_{\mathrm{Ad}(a_1a)^{-1}(\xi)}^{(a_1a)^{-1}n(a_1a)mk,R}(\mathrm{Ad}(a_1a)^{-1}(\widehat{Y}))$$

Recall that $\rho_{R,\mathfrak{s}} = \rho_R$ by Corollary 3.11. From the reduction theory (see [3, p. 944]), for a_1 satisfying $\sigma_{P_1}^{P_2}(H_{P_1}(a_1) - T) \neq 0$, we know that $\operatorname{Ad}(a_1a)^{-1}(n)$ belongs to a compact subset independent of T. To

sum up,

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)\sum_{\xi\in(\widetilde{\mathfrak{m}}_{P_{1}}^{\overline{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\overline{\mathfrak{m}}_{\overline{R}}^{\overline{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|dx\\ \leq &c_{2}\sup_{y\in\Gamma}\int_{A_{P_{0}}^{P_{1},\infty}(t_{0},T)}\int_{A_{P_{1}}^{H,\infty}}e^{(2\rho_{R}-2\rho_{P_{0}})(H_{P_{0}}(a_{1}a))}\sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a_{1})-T)\\ &\cdot\sum_{\xi\in(\widetilde{\mathfrak{m}}_{P_{1}}^{\overline{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\overline{\mathfrak{m}}_{\overline{R}}^{\overline{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\mathrm{Ad}(a_{1}a)^{-1}(\xi)}^{y,R}(\mathrm{Ad}(a_{1}a)^{-1}(\widehat{Y}))|da_{1}da, \end{split}$$

where c_2 is a constant independent of T, and Γ is a compact subset independent of T.

Let \mathcal{O}_F denote the ring of integers of F. We fix an F-basis for each weight space for the action of A_0 on $\mathfrak{s}(F)$. Then \mathcal{O}_F -points of such a weight space make sense. Since the $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ is compactly supported on finite places, there exists a positive integer N_1 independent of T such that the sums over $\xi \in (\widetilde{\mathfrak{m}}_{\widetilde{P}_1}^{\widetilde{R}} \cap \mathfrak{s})(F)$ and $\widehat{Y} \in ((\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}_2})' \cap \mathfrak{s})(F)$ can be restricted to lattices $\frac{1}{N_1}(\widetilde{\mathfrak{m}}_{\widetilde{P}_1}^{\widetilde{R}} \cap \mathfrak{s})(\mathcal{O}_F)$ and $\frac{1}{N_1}((\overline{\mathfrak{n}}_{\widetilde{R}}^{\widetilde{P}_2})' \cap \mathfrak{s})(\mathcal{O}_F)$ respectively, which can be explicit as in [13, §1.9] (we need to replace \mathfrak{m}_R and \mathfrak{n}_R in *loc. cit.* by $\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s}$ and $\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s}$ respectively).

Fix a Euclidean norm $\|\cdot\|$ on the \mathbb{R} -linear space $\mathfrak{s}(F \otimes_{\mathbb{Q}} \mathbb{R})$. Consider a sufficiently large positive integer k to be precise. Thanks to Proposition 3.10, as in [13, (4.10) in p. 372], there exists an integer $m \geq 0$, a real number $k_{\alpha} \geq 0$ for each $\alpha \in \Delta_{P_0}^{P_2}$, and a real number $c_3 > 0$ satisfying the following conditions:

- (1) if $R = P_2, m = 0;$
- (2) for all $\alpha \in \Delta_{P_0}^{P_2} \Delta_{P_0}^R, k_\alpha \ge k;$
- (3) for all $a \in A_{P_0}^{\infty}$,

(4.0.4)
$$\sum_{\widehat{Y} \in \frac{1}{N_1}((\overline{\mathfrak{n}_{\widehat{R}}^{\widetilde{P_2}}})' \cap \mathfrak{s})(\mathcal{O}_F)} \|\mathrm{Ad}(a^{-1})(\widehat{Y})\|^{-m} \le c_3 \prod_{\alpha \in \Delta_{P_0}^{P_2}} e^{-k_\alpha \alpha(H_{P_0}(a))}$$

We fix such data.

For a multi-index \vec{i} , denote by $\partial \vec{i}$ the corresponding differential operator on $\mathfrak{s}(F \otimes \mathbb{R})$. It can be extended to $\mathfrak{s}(\mathbb{A})$ by $\partial \vec{i} (f_{\infty} \otimes \chi^{\infty}) := (\partial \vec{i} f_{\infty}) \otimes \chi^{\infty}$, where we use the notation in Section 2.3. Choose a multi-index \vec{i} whose sum of components is m. Denote

$$\Phi_{\xi}^{x,R,\overrightarrow{i}}(\widehat{Y}) := \int_{(\mathfrak{n}_{\widehat{R}}\cap\mathfrak{s})(\mathbb{A})} (\partial^{\overrightarrow{i}}f)(\mathrm{Ad}(x^{-1})(\xi+U))\Psi(\langle U,\widehat{Y}\rangle)dU.$$

Using integration by parts, for $\hat{Y} \neq 0$, we get

$$|\Phi_{\mathrm{Ad}(a_1a)^{-1}(\xi)}^{y,R}(\mathrm{Ad}(a_1a)^{-1}(\widehat{Y}))| = c_4(y) \|\mathrm{Ad}(a_1a)^{-1}(\widehat{Y})\|^{-m} |\Phi_{\mathrm{Ad}(a_1a)^{-1}(\xi)}^{y,R,\,\overrightarrow{i}}(\mathrm{Ad}(a_1a)^{-1}(\widehat{Y}))|,$$

where $c_4(y)$ is a continuous function of y.

For $\mu \in \Phi(A_0, \mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})$ (refer to Section 3.4 for the notation), denote by $(\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})_{\mu}$ the corresponding weight space. From [53, §41], there exists a function $\phi_{\mu} \in \mathcal{S}((\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})_{\mu}(\mathbb{A}))$ for each $\mu \in \Phi(A_0, \mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})$ and a function $\phi_{\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s}} \in \mathcal{S}((\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A}))$ such that for all $\xi + U \in (\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A}) \oplus (\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A})$ and $y \in \Gamma$,

$$|(\partial^{\overrightarrow{i}}f)(\mathrm{Ad}(y^{-1})(\xi+U))| \leq \left(\prod_{\mu\in\Phi(A_0,\mathfrak{m}_{\overline{R}}\cap\mathfrak{s})}\phi_{\mu}(\xi_{\mu})\right)\phi_{\mathfrak{n}_{\overline{R}}\cap\mathfrak{s}}(U),$$

where ξ_{μ} denotes the projection to $(\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})_{\mu}(\mathbb{A})$ of ξ .

In sum, we deduce that

$$\begin{split} &\sum_{\xi \in (\tilde{\mathfrak{m}}_{\tilde{P}_{1}}^{\tilde{R}} \cap \mathfrak{s})(F)} \sum_{\hat{Y} \in ((\tilde{\mathfrak{m}}_{\tilde{R}}^{\tilde{P}_{2}})' \cap \mathfrak{s})(F)} |\Phi_{\mathrm{Ad}(a_{1}a)^{-1}(\xi)}^{y,R} (\mathrm{Ad}(a_{1}a)^{-1}(\hat{Y}))| \\ &= \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{\tilde{P}_{1}}^{\tilde{R}} \cap \mathfrak{s})(\mathcal{O}_{F})} \sum_{\hat{Y} \in \frac{1}{N_{1}} ((\tilde{\mathfrak{m}}_{\tilde{R}}^{\tilde{P}_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} |\Phi_{\mathrm{Ad}(a_{1}a)^{-1}(\xi)}^{y,R} (\mathrm{Ad}(a_{1}a)^{-1}(\hat{Y}))| \\ &= \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{\tilde{P}_{1}}^{\tilde{R}} \cap \mathfrak{s})(\mathcal{O}_{F})} \sum_{\hat{Y} \in \frac{1}{N_{1}} ((\tilde{\mathfrak{m}}_{\tilde{R}}^{\tilde{P}_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} c_{4}(y) ||\mathrm{Ad}(a_{1}a)^{-1}(\hat{Y})||^{-m} |\Phi_{\mathrm{Ad}(a_{1}a)^{-1}(\xi)}^{y,R,\frac{1}{i}} (\mathrm{Ad}(a_{1}a)^{-1}(\hat{Y}))| \\ &\leq c_{5} \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{\tilde{P}_{1}}^{\tilde{R}} \cap \mathfrak{s})(\mathcal{O}_{F})} \left(\prod_{\mu \in \Phi(A_{0},\mathfrak{m}_{\tilde{R}} \cap \mathfrak{s})} \phi_{\mu}(\mu(a_{1}a)^{-1}\xi_{\mu}) \right) \cdot \sum_{\hat{Y} \in \frac{1}{N_{1}} ((\tilde{\mathfrak{m}}_{\tilde{R}}^{\tilde{P}_{2}})' \cap \mathfrak{s})(\mathcal{O}_{F})} ||\mathrm{Ad}(a_{1}a)^{-1}(\hat{Y})||^{-m} \\ &\leq c_{5} c_{3} \sum_{\xi \in \frac{1}{N_{1}} (\tilde{\mathfrak{m}}_{\tilde{P}_{1}}^{\tilde{R}} \cap \mathfrak{s})(\mathcal{O}_{F})} \left(\prod_{\mu \in \Phi(A_{0},\mathfrak{m}_{\tilde{R}} \cap \mathfrak{s})} \phi_{\mu}(\mu(a_{1}a)^{-1}\xi_{\mu}) \right) \cdot \prod_{\alpha \in \Delta_{P_{0}}^{P_{2}}} e^{-k_{\alpha}\alpha(H_{P_{0}}(a_{1}a))}, \end{split}$$

where $c_5 := \sup_{y \in \Gamma} c_4(y) \int_{(\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(\mathbb{A})} \phi_{\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s}}(U) dU$; in the last inequality, we have used (4.0.4). Thus

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)\sum_{\xi\in(\widetilde{\mathfrak{m}}_{\overline{P_{1}}}^{\overline{R}}\cap\mathfrak{s})(F)}\sum_{\widehat{Y}\in((\widetilde{\mathfrak{m}}_{\overline{R}}^{\overline{P_{2}}})'\cap\mathfrak{s})(F)}|\Phi_{\xi}^{x,R}(\widehat{Y})|dx\\ \leq &c_{2}c_{5}c_{3}\int_{A_{P_{0}}^{P_{1},\infty}(t_{0},T)}\int_{A_{P_{1}}^{H,\infty}}e^{(2\rho_{R}-2\rho_{P_{0}})(H_{P_{0}}(a_{1}a))}\sigma_{P_{1}}^{P_{2}}(H_{P_{1}}(a_{1})-T)\\ &\cdot\sum_{\xi\in\frac{1}{N_{1}}(\widetilde{\mathfrak{m}}_{\overline{P_{1}}}^{\overline{R}}\cap\mathfrak{s})(\mathcal{O}_{F})}\left(\prod_{\mu\in\Phi(A_{0},\mathfrak{m}_{\overline{R}}\cap\mathfrak{s})}\phi_{\mu}(\mu(a_{1}a)^{-1}\xi_{\mu})\right)\cdot\prod_{\alpha\in\Delta_{P_{0}}^{P_{2}}}e^{-k_{\alpha}\alpha(H_{P_{0}}(a_{1}a))}da_{1}da. \end{split}$$

From [13, p. 375], when $\sigma_{P_1}^{P_2}(H_{P_1}(a_1) - T) \neq 0$, we have $\alpha(H_{P_0}(a_1a)) > t_0$ for all $\alpha \in \Delta_{P_0}^{P_2}$. Denote by $\Sigma_{P_0}^{\mathfrak{m}_{\tilde{R}} \cap \mathfrak{s}}$ the positive weights of $\mathfrak{m}_{\tilde{R}} \cap \mathfrak{s}$ under the action of A_0 . Consider the subsets S of $\Sigma_{P_0}^{\mathfrak{m}_{\tilde{R}} \cap \mathfrak{s}}$ with the following property: for all $\alpha \in \Delta_{P_0}^R - \Delta_{P_0}^{P_1}$, there exists $\mu \in S$ such that its α -coordinate is > 0. Then

$$\begin{split} & \sum_{\xi \in \frac{1}{N_1} (\tilde{\mathfrak{m}}_{\widetilde{P_1}}^{\widetilde{R}} \cap \mathfrak{s})(\mathcal{O}_F)} \left(\prod_{\mu \in \Phi(A_0, \mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})} \phi_{\mu}(\mu(a_1 a)^{-1} \xi_{\mu}) \right) \\ & \leq \sum_{S} \left[\prod_{\mu \in S} \left(\sum_{\xi_- \in \frac{1}{N_1} \mathfrak{m}_{-\mu}(\mathcal{O}_F) - \{0\}} \phi_{-\mu}(\mu(a_1 a) \xi_-) \right) \right] \left[\prod_{\mu \in \Sigma_{P_0}^{\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s}}} \left(\sum_{\xi_+ \in \frac{1}{N_1} \mathfrak{m}_{\mu}(\mathcal{O}_F)} \phi_{\mu}(\mu(a_1 a)^{-1} \xi_+) \right) \right] \\ & \cdot \left[\sum_{\xi_0 \in \frac{1}{N_1} \mathfrak{m}_0(\mathcal{O}_F)} \phi_0(\xi_0) \right]. \end{split}$$

Denote by $\Sigma_{P_0}^{\mathfrak{m}_R}$ the positive weights of \mathfrak{m}_R under the action of A_0 . From Proposition 3.10, we know that $\Sigma_{P_0}^{\mathfrak{m}_{\widetilde{R}}\cap\mathfrak{s}} = \Sigma_{P_0}^{\mathfrak{m}_R}$ and that each weight has the same multiplicity in $\mathfrak{m}_{\widetilde{R}}\cap\mathfrak{s}$ and \mathfrak{m}_R . From now on, we are in exactly the same situation as in [13, p. 373] and able to borrow the rest of its proof to conclude. \Box

5. Polynomial distributions

Let $T \in T_+ + \mathfrak{a}_{P_0}^+$ and $\mathfrak{o} \in \mathcal{O}$. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, define

(5.0.1)
$$J_{\mathfrak{o}}^{H,T}(f) := \int_{H(F)\setminus H(\mathbb{A})^1} k_{f,\mathfrak{o}}^T(x) dx$$

and

(5.0.2)
$$J^{H,T}(f) := \int_{H(F)\backslash H(\mathbb{A})^1} k_f^T(x) dx,$$

where $k_{f,\mathfrak{o}}^T(x)$ and $k_f^T(x)$ are defined by (4.0.1) and (4.0.2) respectively. From Theorem 4.2, we know that $J_{\mathfrak{o}}^{H,T}$ and $J^{H,T}$ are well-defined distributions on $\mathcal{S}(\mathfrak{s}(\mathbb{A}))$. We also have

$$J^{H,T}(f) = \sum_{\mathfrak{o} \in \mathcal{O}} J^{H,T}_{\mathfrak{o}}(f),$$

which is an analogue of the geometric side of Arthur's trace formula. In this section, we shall prove that $J_{\mathfrak{o}}^{H,T}(f)$ and $J^{H,T}(f)$ can be extended to polynomials in $T \in \mathfrak{a}_{P_0}$ (see Corollary 5.3 below), whose constant terms will be denoted by $J_{\mathfrak{o}}^H(f)$ and $J^H(f)$ respectively.

Let us begin with a generalisation of our results in last section. Let Q be a standard parabolic subgroup of H. Recall the two cases studied in Section 3.4. In **Case I**, we have

$$M_Q \simeq \operatorname{Res}_{E/F} GL_{n_1,D'} \times \cdots \times \operatorname{Res}_{E/F} GL_{n_l,D}$$

and

$$M_{\widetilde{O}} \simeq GL_{n_1,D} \times \cdots \times GL_{n_l,D},$$

where $\sum_{i=1}^{l} n_i = n$. In **Case II**, we have

$$M_Q \simeq \operatorname{Res}_{E/F} GL_{\frac{n_1}{2}, D \otimes_F E} \times \cdots \times \operatorname{Res}_{E/F} GL_{\frac{n_l}{2}, D \otimes_F E}$$

and

$$M_{\widetilde{Q}} \simeq GL_{n_1,D} \times \cdots \times GL_{n_l,D},$$

where n_i is even for all $1 \leq i \leq l$ and $\sum_{i=1}^{l} n_i = n$. In either case of the two, the tangent space of $M_{\widetilde{Q}}/M_Q$ is $\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s}$, on which M_Q acts by conjugation. We remark that our results in last section can be generalised to the product setting here, whose proofs are similar and will be omitted. Define a relation of equivalence on $(\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(F)$ which is similar to that on $\mathfrak{s}(F)$ on each component. We denote by $\mathcal{O}^{\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s}}$ the set of equivalent classes for this relation. For $\mathfrak{o} \in \mathcal{O}$, the intersection $\mathfrak{o} \cap \mathfrak{m}_{\widetilde{Q}}(F)$ is a finite (perhaps empty) union of classes $\mathfrak{o}_1, \dots, \mathfrak{o}_t \in \mathcal{O}^{\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s}}$. Notice that there exists a bijection between the set of standard parabolic subgroups P of H contained in Q and the set of standard parabolic subgroups P^* of M_Q (namely $P_0 \cap M_Q \subseteq P^*$) given by $P \mapsto P \cap M_Q$, whose inverse is given by $P^* \mapsto P^*N_Q$. Let $f^* \in \mathcal{S}((\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}))$, P^* be a standard parabolic subgroup of M_Q and $1 \leq j \leq t$. For $x \in M_{P^*}(F)N_{P^*}(\mathbb{A})\backslash M_Q(\mathbb{A})$, define

(5.0.3)
$$k_{f^*,P^*,\mathfrak{o}_j}^{M_Q}(x) := \sum_{Y \in \mathfrak{m}_{\widehat{P^*}}(F) \cap \mathfrak{o}_j} \int_{(\mathfrak{n}_{\widehat{P^*}} \cap \mathfrak{s})(\mathbb{A})} f^*(\mathrm{Ad}(x^{-1})(Y+U)) dU.$$

For $T \in \mathfrak{a}_0$ and $x \in M_Q(F) \setminus M_Q(\mathbb{A})$, define

$$k_{f^*,\mathfrak{o}_j}^{M_Q,T}(x) := \sum_{\{P^*: P_0 \cap M_Q \subseteq P^*\}} (-1)^{\dim(A_{P^*}/A_{M_Q})} \sum_{\delta \in P^*(F) \setminus M_Q(F)} \widehat{\tau}_{P^*}^{M_Q}(H_{P^*}(\delta x) - T) \cdot k_{f^*,P^*,\mathfrak{o}_j}^{M_Q}(\delta x).$$

For $T \in T_+ + \mathfrak{a}_{P_0}^+$, define

$$J^{M_Q,T}_{\mathfrak{o}_j}(f^*) := \int_{\substack{M_Q(F) \setminus M_Q(\mathbb{A})^1 \\ M_Q,T}} k^{M_Q,T}_{f^*,\mathfrak{o}_j}(x) dx.$$

Then we obtain a well-defined distribution $J_{\mathfrak{o}_j}^{M_Q,T}$ on $\mathcal{S}((\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}))$. Now we define

(5.0.4)
$$J_{\mathfrak{o}}^{M_Q,T} := \sum_{j=1}^t J_{\mathfrak{o}_j}^{M_Q,T}$$

and

(5.0.5)
$$J^{M_Q,T} := \sum_{\mathfrak{o} \in \mathcal{O}} J^{M_Q,T}_{\mathfrak{o}}.$$

For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, define $f_Q \in \mathcal{S}((\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}))$ by

(5.0.6)
$$\forall Y \in (\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}), f_Q(Y) := \int_K \int_{(\mathfrak{n}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(k^{-1})(Y+V)) dV dk.$$

Let $T_1, T_2 \in \mathfrak{a}_{P_0}$. As in [5, §2], we define the function $\Gamma_P(T_1, T_2)$ inductively on dim (A_P/A_H) by setting

(5.0.7)
$$\hat{\tau}_P^H(T_1 - T_2) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_Q/A_H)} \hat{\tau}_P^Q(T_1) \Gamma_Q(T_1, T_2)$$

for any standard parabolic subgroup P of H. This definition can be explicitly given by [5, (2.1) in p. 13] and only depends on the projections of T_1, T_2 onto \mathfrak{a}_P^H .

LEMMA 5.1. Let $T_2 \in \mathfrak{a}_{P_0}$ and Q be a standard parabolic subgroup of H. The function $T_1 \mapsto \Gamma_Q(T_1, T_2)$ is compactly supported on \mathfrak{a}_Q^H . Moreover, the function $T_2 \mapsto \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1, T_2) dT_1$ is a homogeneous polynomial in T_2 of degree dim (A_Q/A_H) .

PROOF. This is [5, Lemmas 2.1 and 2.2].

THEOREM 5.2. Let $T' \in T_+ + \mathfrak{a}_{P_0}^+$, $\mathfrak{o} \in \mathcal{O}$ and $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$. Then for all $T \in T_+ + \mathfrak{a}_{P_0}^+$,

$$J^{H,T}_{\mathfrak{o}}(f) = \sum_{\{Q: P_0 \subseteq Q\}} J^{M_Q,T'}_{\mathfrak{o}}(f_Q) \int_{\mathfrak{a}^H_Q} \Gamma_Q(T_1, T - T') dT_1$$

where $J_{\mathfrak{o}}^{H,T}$, $J_{\mathfrak{o}}^{M_Q,T'}$ and f_Q are defined by the formulae (5.0.1), (5.0.4) and (5.0.6) respectively.

COROLLARY 5.3. Let $\mathfrak{o} \in \mathcal{O}$ and $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$. Then $J^{H,T}_{\mathfrak{o}}(f)$ and $J^{H,T}(f)$ (defined by (5.0.2)) are polynomials in T of degree $\leq n-1$ for $T \in T_+ + \mathfrak{a}^+_{P_0}$. Thus we can extend them to all $T \in \mathfrak{a}_{P_0}$.

REMARK 5.4. We fix M_0 and $M_{\tilde{0}}$ which are minimal Levi subgroups of H and G respectively. The distributions $J_o^H(f)$ and $J^H(f)$ (defined as constant terms of $J_o^{H,T}(f)$ and $J^{H,T}(f)$ respectively) are independent of the choice of the minimal parabolic subgroup $P_0 \supseteq M_0$ of H. In fact, the argument of [13, Proposition 4.6] after some minor modifications applies here because elements in Ω^H have representatives in $H(F) \cap K$ in our cases.

PROOF OF THEOREM 5.2. Let P be any standard parabolic subgroup of H, $\delta \in P(F) \setminus H(F)$ and $x \in H(\mathbb{A})^1$. By substituting $T_1 = H_P(\delta x) - T'$ and $T_2 = T - T'$ in (5.0.7), we have

$$\widehat{\tau}_P^H(H_P(\delta x) - T) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_Q/A_H)} \widehat{\tau}_P^Q(H_P(\delta x) - T') \Gamma_Q(H_P(\delta x) - T', T - T').$$

Then

$$\begin{split} J_{\mathbf{0}}^{H,T}(f) &= \int_{H(F)\backslash H(\mathbb{A})^1} \left(\sum_{\{P:P_0 \subseteq P\}} (-1)^{\dim(A_P/A_H)} \sum_{\delta \in P(F)\backslash H(F)} \widehat{\tau}_P^H(H_P(\delta x) - T) \cdot k_{f,P,\mathbf{0}}(\delta x) \right) dx \\ &= \int_{H(F)\backslash H(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P\}} (-1)^{\dim(A_P/A_H)} \sum_{\delta \in P(F)\backslash H(F)} \\ &\left(\sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_Q/A_H)} \widehat{\tau}_P^Q(H_P(\delta x) - T') \Gamma_Q(H_P(\delta x) - T', T - T') \right) k_{f,P,\mathbf{0}}(\delta x) dx. \end{split}$$

By exchanging the order of two sums over P and Q and decomposing the sum over $P(F)\setminus H(F)$ into two sums over $P(F)\setminus Q(F)$ and $Q(F)\setminus H(F)$, we obtain

$$J_{\mathfrak{o}}^{H,T}(f) = \sum_{\{Q:P_0 \subseteq Q\}} \int_{H(F) \setminus H(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in P(F) \setminus Q(F)} \sum_{\delta' \in Q(F) \setminus H(F)} \widehat{\tau}_P^Q(H_P(\delta\delta'x) - T')\Gamma_Q(H_P(\delta\delta'x) - T', T - T')k_{f,P,\mathfrak{o}}(\delta\delta'x)dx.$$

Combining the integral over $H(F) \setminus H(\mathbb{A})^1$ and the sum over $Q(F) \setminus H(F)$ into the integral over $Q(F) \setminus H(\mathbb{A})^1$, and noticing that

$$P(F) \setminus Q(F) \simeq (P(F) \cap M_Q(F)) \setminus M_Q(F),$$

we have

$$J_{\mathfrak{o}}^{H,T}(f) = \sum_{\{Q:P_0 \subseteq Q\}} \int_{Q(F) \setminus H(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P(F) \cap M_Q(F)) \setminus M_Q(F)} \widehat{\tau}_P^Q(H_P(\delta x) - T') \Gamma_Q(H_P(\delta x) - T', T - T') k_{f,P,\mathfrak{o}}(\delta x) dx.$$

By the Iwasawa decomposition and our choice of measures, we get

We notice that

$$\hat{\tau}_P^Q(H_P(\delta namk) - T') = \hat{\tau}_P^Q(H_P(\delta m) + H_P(a) - T') = \hat{\tau}_P^Q(H_P(\delta m) - T')$$

and that

$$\Gamma_Q(H_P(\delta namk) - T', T - T') = \Gamma_Q(H_Q(\delta namk) - T', T - T') = \Gamma_Q(H_Q(a) - T', T - T').$$

Additionally, using $A_Q=A_{\widetilde{Q}}$ and change of variables, we see that

$$\begin{split} k_{f,P,\mathfrak{o}}(\delta namk) &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(\delta namk)^{-1}(Y+U)) dU \\ &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(\delta a^{-1}namk)^{-1}(Y+a^{-1}Ua)) dU \\ &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(\delta a^{-1}namk)^{-1}(Y+U)) e^{2\rho_{Q,\mathfrak{s}}(H_{P_0}(a))} dU \\ &= e^{2\rho_{Q,\mathfrak{s}}(H_{P_0}(a))} k_{f,P,\mathfrak{o}}(\delta a^{-1}namk), \end{split}$$

where $\rho_{Q,\mathfrak{s}}$ is defined in Section 3.4. Since $\delta a^{-1}na\delta^{-1} \in N_Q(\mathbb{A}) \subseteq N_P(\mathbb{A})$ and $k_{f,P,\mathfrak{o}}$ is left invariant by $N_P(\mathbb{A})$, we deduce that

$$k_{f,P,\mathfrak{o}}(\delta namk) = e^{2\rho_{Q,\mathfrak{o}}(H_{P_0}(a))} k_{f,P,\mathfrak{o}}(\delta mk).$$

To sum up, the integrand in $J_{\mathfrak{o}}^{H,T}(f)$ is independent of $n \in N_Q(F) \setminus N_Q(\mathbb{A})$. Recall that we choose the Haar measure such that $\operatorname{vol}(N_Q(F) \setminus N_Q(\mathbb{A})) = 1$. By Corollary 3.11, the factors $e^{-2\rho_Q(H_{P_0}(a))}$ and $e^{2\rho_{Q,\mathfrak{s}}(H_{P_0}(a))}$ cancel, and then

$$\begin{split} J_{\mathfrak{o}}^{H,T}(f) &= \sum_{\{Q:P_0 \subseteq Q\}} \left(\int_{A_Q^{H,\infty}} \Gamma_Q(H_Q(a) - T', T - T') da \right) \int_{M_Q(F) \setminus M_Q(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} \\ & (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P(F) \cap M_Q(F)) \setminus M_Q(F)} \widehat{\tau}_P^Q(H_P(\delta m) - T') \left(\int_K k_{f,P,\mathfrak{o}}(\delta mk) dk \right) dm. \end{split}$$

From the definition of the Haar measure on $A_Q^{H,\infty},$ we have

$$\int_{A_Q^{H,\infty}} \Gamma_Q(H_Q(a) - T', T - T') da := \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1 - T', T - T') dT_1$$
$$= \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1, T - T') dT_1.$$

Since $\mathfrak{n}_{\widetilde{P}} = \mathfrak{n}_{\widetilde{P}}^{\widetilde{Q}} \oplus \mathfrak{n}_{\widetilde{Q}}$, by change of variables, we deduce that

$$\begin{split} k_{f,P,\mathfrak{o}}(\delta mk) &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}}^{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(\delta mk)^{-1}(Y+U+V)) dV \\ &= \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}}^{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(k^{-1})(\mathrm{Ad}(\delta m)^{-1}(Y+U)+V)) dV, \end{split}$$

where we need to verify that the change of variables $V \mapsto \operatorname{Ad}(\delta m)(V)$ does not change the Haar measure. This can be shown by Proposition 3.12 in two steps: firstly, $\mathfrak{n}_{\widetilde{O}} \cap \mathfrak{s} = \mathfrak{n}_Q \tau$ shows that $V \mapsto V' := (\delta m) V$ does not change any Haar measure; secondly, $\mathfrak{n}_{\widetilde{Q}} \cap \mathfrak{s} = \tau \mathfrak{n}_Q$ shows that $V' \mapsto V'(\delta m)^{-1}$ does not change any Haar measure. Then we can write

$$\int_{K} k_{f,P,\mathfrak{o}}(\delta m k) dk = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}}^{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f_{Q}(\mathrm{Ad}(\delta m)^{-1}(Y+U)) dU$$
$$= \sum_{j=1}^{t} k_{f_{Q},P \cap M_{Q},\mathfrak{o}_{j}}^{M_{Q}}(\delta m)$$

by (5.0.3). Now we can conclude by noting that

$$\begin{split} J_{\mathbf{0}}^{M_Q,T'}(f_Q) &= \sum_{j=1}^t \int_{M_Q(F) \setminus M_Q(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_{P \cap M_Q}/A_{M_Q})} \sum_{\delta \in (P(F) \cap M_Q(F)) \setminus M_Q(F)} \\ &\quad \widehat{\tau}_{P \cap M_Q}^{M_Q}(H_{P \cap M_Q}(\delta m) - T') k_{f_Q,P \cap M_Q,\mathbf{0}_j}^{M_Q}(\delta m) dm \\ &= \int_{M_Q(F) \setminus M_Q(\mathbb{A})^1} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P(F) \cap M_Q(F)) \setminus M_Q(F)} \\ &\quad \widehat{\tau}_{P}^Q(H_P(\delta m) - T') \left(\sum_{j=1}^t k_{f_Q,P \cap M_Q,\mathbf{0}_j}^{M_Q}(\delta m) \right) dm. \end{split}$$

6. Noninvariance

Let Q be a standard parabolic subgroup of H and $y \in H(\mathbb{A})^1$. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, define $f_{Q,y} \in \mathcal{S}(\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A})$ by

(6.0.1)
$$\forall Y \in (\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}), f_{Q,y}(Y) := \int_{K} \int_{(\mathfrak{n}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(k^{-1})(Y+V)) p_{Q}(-H_{Q}(ky)) dV dk,$$

where for $T \in \mathfrak{a}_{P_0}$, we write

$$p_Q(T) := \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1, T) dT_1.$$

We can also extend our results in last section to the product setting by the same argument. Let $\mathfrak{o} \in \mathcal{O}$ and $f^* \in \mathcal{S}((\mathfrak{m}_{\widetilde{Q}} \cap \mathfrak{s})(\mathbb{A}))$. For $T \in T_+ + \mathfrak{a}_{P_0}^+$, $J_{\mathfrak{o}}^{M_Q,T}(f^*)$ and $J^{M_Q,T}(f^*)$ (defined by (5.0.4) and (5.0.5) respectively) are polynomials in T. Then we can extend them to all $T \in \mathfrak{a}_{P_0}$. Denote by $J_{\mathfrak{o}}^{M_Q}(f^*)$ the constant term of $J_{\mathfrak{o}}^{M_Q,T}(f^*)$.

PROPOSITION 6.1. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $y \in H(\mathbb{A})^1$, we denote $f^y(x) := f(\operatorname{Ad}(y)(x))$. For $\mathfrak{o} \in \mathcal{O}$, we have

$$J^H_{\mathfrak{o}}(f^y) = \sum_{\{Q: P_0 \subseteq Q\}} J^{M_Q}_{\mathfrak{o}}(f_{Q,y})$$

PROOF. Let $T \in T_+ + \mathfrak{a}_{P_0}^+$. By definition,

$$J_{\mathfrak{o}}^{H,T}(f^{y}) = \int_{H(F)\backslash H(\mathbb{A})^{1}} \left(\sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{\delta \in P(F)\backslash H(F)} \widehat{\tau}_{P}^{H}(H_{P}(\delta x) - T) \cdot k_{f^{y},P,\mathfrak{o}}(\delta x) \right) dx,$$

where

$$k_{f^y,P,\mathfrak{o}}(\delta x) = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(y) \mathrm{Ad}(\delta x)^{-1} (Y+U)) dU = k_{f,P,\mathfrak{o}}(\delta x y^{-1}).$$

Invoking change of variables, we get

$$J_{\mathfrak{o}}^{H,T}(f^y) = \int_{H(F)\backslash H(\mathbb{A})^1} \left(\sum_{\{P:P_0 \subseteq P\}} (-1)^{\dim(A_P/A_H)} \sum_{\delta \in P(F)\backslash H(F)} \widehat{\tau}_P^H(H_P(\delta xy) - T) \cdot k_{f,P,\mathfrak{o}}(\delta x) \right) dx.$$

For $x \in H(\mathbb{A})$ and P a standard parabolic subgroup of H, let $k_P(x)$ be an element in K satisfying $xk_P(x)^{-1} \in P(\mathbb{A})$. Then

$$\widehat{\tau}_P^H(H_P(\delta xy) - T) = \widehat{\tau}_P^H(H_P(\delta x) - T + H_P(k_P(\delta x)y))$$

By substituting $T_1 = H_P(\delta x) - T$ and $T_2 = -H_P(k_P(\delta x)y)$ in (5.0.7), we get

$$\widehat{\tau}_P^H(H_P(\delta xy) - T) = \sum_{\{Q:P \subseteq Q\}} (-1)^{\dim(A_Q/A_H)} \widehat{\tau}_P^Q(H_P(\delta x) - T) \Gamma_Q(H_P(\delta x) - T, -H_P(k_P(\delta x)y)) = 0$$

Then

$$\begin{aligned} J_{\mathbf{o}}^{H,T}(f^{y}) &= \int_{H(F)\backslash H(\mathbb{A})^{1}} \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{\delta \in P(F)\backslash H(F)} \\ & \left(\sum_{\{Q:P\subseteq Q\}} (-1)^{\dim(A_{Q}/A_{H})} \widehat{\tau}_{P}^{Q} (H_{P}(\delta x) - T) \Gamma_{Q} (H_{P}(\delta x) - T, -H_{P}(k_{P}(\delta x)y)) \right) \cdot k_{f,P,\mathbf{o}}(\delta x) dx, \end{aligned}$$

By exchanging the order of two sums over P and Q, and decomposing the sum over $P(F) \setminus H(F)$ into two sums over $P(F) \setminus Q(F)$ and $Q(F) \setminus H(F)$, we deduce that

$$J_{\mathbf{o}}^{H,T}(f^{y}) = \sum_{\{Q:P_{0}\subseteq Q\}} \int_{H(F)\backslash H(\mathbb{A})^{1}} \sum_{\{P:P_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta\in P(F)\backslash Q(F)} \sum_{\delta'\in Q(F)\backslash H(F)} \widehat{\tau}_{P}^{Q}(H_{P}(\delta\delta'x) - T)\Gamma_{Q}(H_{P}(\delta\delta'x) - T, -H_{P}(k_{P}(\delta\delta'x)y))k_{f,P,\mathbf{o}}(\delta\delta'x)dx.$$

By combining the integral over $H(F)\setminus H(\mathbb{A})^1$ and the sum over $Q(F)\setminus H(F)$ into the integral over $Q(F)\setminus H(\mathbb{A})^1$ and using the fact that

$$P(F)\backslash Q(F) \simeq (P(F) \cap M_Q(F))\backslash M_Q(F),$$

we have

$$J_{\mathbf{o}}^{H,T}(f^{y}) = \sum_{\{Q:P_{0}\subseteq Q\}} \int_{Q(F)\backslash H(\mathbb{A})^{1}} \sum_{\{P:P_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta\in(P(F)\cap M_{Q}(F))\backslash M_{Q}(F)} \hat{\tau}_{P}^{Q}(H_{P}(\delta x) - T)\Gamma_{Q}(H_{P}(\delta x) - T, -H_{P}(k_{P}(\delta x)y))k_{f,P,\mathbf{o}}(\delta x)dx.$$

By the Iwasawa decomposition and our choice of measures, we get

$$J_{\mathbf{0}}^{H,T}(f^{y}) = \sum_{\{Q:P_{0}\subseteq Q\}} \int_{K} \int_{M_{Q}(F)\backslash M_{Q}(\mathbb{A})^{1}} \int_{A_{Q}^{H,\infty}} \int_{N_{Q}(F)\backslash N_{Q}(\mathbb{A})} \sum_{\{P:P_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})}$$
$$\sum_{\delta \in (P(F)\cap M_{Q}(F))\backslash M_{Q}(F)} \widehat{\tau}_{P}^{Q}(H_{P}(\delta namk) - T)\Gamma_{Q}(H_{P}(\delta namk) - T, -H_{P}(k_{P}(\delta namk)y))$$
$$\cdot k_{f,P,\mathbf{0}}(\delta namk)e^{-2\rho_{Q}(H_{P_{0}}(a))}dndadmdk.$$

As in the proof of Theorem 5.2, we see that

$$\hat{\tau}_P^Q(H_P(\delta namk) - T) = \hat{\tau}_P^Q(H_P(\delta m) - T)$$

and that

$$k_{f,P,\mathfrak{o}}(\delta namk) = e^{2\rho_Q(H_{P_0}(a))} k_{f,P,\mathfrak{o}}(\delta mk).$$

Additionally,

$$\begin{split} \Gamma_Q(H_P(\delta namk) - T, -H_P(k_P(\delta namk)y)) &= \Gamma_Q(H_Q(\delta namk) - T, -H_Q(k_P(\delta namk)y)) \\ &= \Gamma_Q(H_Q(a) - T, -H_Q(k_Q(\delta namk)y)) \\ &= \Gamma_Q(H_Q(a) - T, -H_Q(ky)). \end{split}$$

In sum, the integrand in $J_{\mathfrak{o}}^{H,T}(f^y)$ is independent of $n \in N_Q(F) \setminus N_Q(\mathbb{A})$. Recall that we choose the Haar measure such that $\operatorname{vol}(N_Q(F) \setminus N_Q(\mathbb{A})) = 1$. Then

$$\begin{split} J^{H,T}_{\mathbf{o}}(f^y) &= \sum_{\{Q:P_0 \subseteq Q\}} \int_K \int_{M_Q(F) \backslash M_Q(\mathbb{A})^1} \int_{A^{H,\infty}_Q} \sum_{\{P:P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P(F) \cap M_Q(F)) \backslash M_Q(F)} \\ \widehat{\tau}^Q_P(H_P(\delta m) - T) \Gamma_Q(H_Q(a) - T, -H_Q(ky)) k_{f,P,\mathbf{o}}(\delta mk) dadmdk. \end{split}$$

First, let us consider the integral on $A_Q^{H,\infty},$ which is

$$\begin{split} \int_{A_Q^{H,\infty}} \Gamma_Q(H_Q(a) - T, -H_Q(ky)) da &:= \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1 - T, -H_Q(ky)) dT_1 \\ &= \int_{\mathfrak{a}_Q^H} \Gamma_Q(T_1, -H_Q(ky)) dT_1 \\ &= p_Q(-H_Q(ky)). \end{split}$$

Next, we compute the integral on K, which is

$$\int_{K} k_{f,P,\mathbf{o}}(\delta m k) p_Q(-H_Q(ky)) dk.$$

As in the proof of Theorem 5.2, we see that

$$k_{f,P,\mathfrak{o}}(\delta mk) = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}}^{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} dU \int_{(\mathfrak{n}_{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(k^{-1})(\mathrm{Ad}(\delta m)^{-1}(Y+U)+V))dV,$$

so we can write

$$\int_{K} k_{f,P,\mathfrak{o}}(\delta m k) p_{Q}(-H_{Q}(ky)) dk = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \int_{(\mathfrak{n}_{\tilde{P}}^{\tilde{Q}} \cap \mathfrak{s})(\mathbb{A})} f_{Q,y}(\mathrm{Ad}(\delta m)^{-1}(Y+U)) dU$$
$$= \sum_{j=1}^{t} k_{f_{Q,y},P \cap M_{Q},\mathfrak{o}_{j}}^{M_{Q}}(\delta m)$$

by (5.0.3). Therefore, we obtain

$$J_{\mathbf{o}}^{H,T}(f^{y}) = \sum_{\{Q:P_{0}\subseteq Q\}} \int_{M_{Q}(F)\backslash M_{Q}(\mathbb{A})^{1}} \sum_{\{P:P_{0}\subseteq P\subseteq Q\}} (-1)^{\dim(A_{P}/A_{Q})} \sum_{\delta\in(P(F)\cap M_{Q}(F))\backslash M_{Q}(F)} \hat{\tau}_{P}^{Q}(H_{P}(\delta m) - T) \left(\sum_{j=1}^{t} k_{f_{Q,y},P\cap M_{Q},\mathbf{o}_{j}}^{M}(\delta m)\right) dm.$$

As in the proof of Theorem 5.2, we notice that

$$\begin{split} J^{M_Q,T}_{\mathfrak{o}}(f_{Q,y}) = & \int_{M_Q(F) \backslash M_Q(\mathbb{A})^1} \sum_{\{P: P_0 \subseteq P \subseteq Q\}} (-1)^{\dim(A_P/A_Q)} \sum_{\delta \in (P(F) \cap M_Q(F)) \backslash M_Q(F)} \\ & \widehat{\tau}^Q_P(H_P(\delta m) - T) \left(\sum_{j=1}^t k^{M_Q}_{f_{Q,y}, P \cap M_Q, \mathfrak{o}_j}(\delta m) \right) dm. \end{split}$$
 due that

Thus we deduce that

$$J^{H,T}_{\mathfrak{o}}(f^y) = \sum_{\{Q: P_0 \subseteq Q\}} J^{M_Q,T}_{\mathfrak{o}}(f_{Q,y})$$

We may conclude by taking the constant terms of both sides.

7. An infinitesimal trace formula for $\mathfrak{s}//H$

Recall that for $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we have defined its Fourier transform $\hat{f} \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ by (3.3.1) and denoted the constant term of $J^{H,T}_{\mathfrak{o}}(f)$ by $J^{H}_{\mathfrak{o}}(f)$.

THEOREM 7.1. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we have the equality,

$$\sum_{\mathfrak{o}\in\mathcal{O}}J^{H}_{\mathfrak{o}}(f)=\sum_{\mathfrak{o}\in\mathcal{O}}J^{H}_{\mathfrak{o}}(\hat{f}).$$

PROOF. Applying the Poisson summation formula, for any $x \in H(\mathbb{A})$, we have

$$\sum_{Y \in \mathfrak{s}(F)} f(\mathrm{Ad}(x^{-1})(Y)) = \sum_{Y \in \mathfrak{s}(F)} \widehat{f}(\mathrm{Ad}(x^{-1})(Y)),$$

i.e.,

$$k_{f,H}(x) = k_{\hat{f},H}(x).$$

By Corollary 4.4, for all $T \in T_+ + \mathfrak{a}_{P_0}^+$ satisfying $\alpha(T) \ge \epsilon_0 \parallel T \parallel$ for any $\alpha \in \Delta_{P_0}$, we get

$$|J^{H,T}(f) - \int_{H(F)\setminus H(\mathbb{A})^1} F^H(x,T)k_{f,H}(x)dx| \le C_1 e^{-N||T||}$$

and

$$|J^{H,T}(\hat{f}) - \int_{H(F)\setminus H(\mathbb{A})^1} F^H(x,T) k_{\hat{f},H}(x) dx| \le C_2 e^{-N||T||}.$$

Hence

$$|J^{H,T}(f) - J^{H,T}(\hat{f})| \le (C_1 + C_2)e^{-N||T||}.$$

By Corollary 5.3, we know that both of $J^{H,T}(f)$ and $J^{H,T}(\hat{f})$ are polynomials in T, so we deduce that $J^{H,T}(f) = J^{H,T}(\hat{f}).$

From

$$J^{H,T}(f) = \sum_{\mathfrak{o} \in \mathcal{O}} J^{H,T}_{\mathfrak{o}}(f)$$

and

$$J^{H,T}(\hat{f}) = \sum_{\mathfrak{o} \in \mathcal{O}} J^{H,T}_{\mathfrak{o}}(\hat{f}),$$

we obtain

$$\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}^{H,T}(f)=\sum_{\mathfrak{o}\in\mathcal{O}}J_{\mathfrak{o}}^{H,T}(\widehat{f}).$$

We can draw the conclusion by taking the constant terms of both sides.

8. The second modified kernel

Let $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, P be a standard parabolic subgroup of H and $\mathfrak{o} \in \mathcal{O}_{rs}$ (see Section 3.3). For $x \in P(F) \setminus H(\mathbb{A})$, define

$$j_{f,P,\mathfrak{o}}(x):=\sum_{Y\in\mathfrak{m}_{\tilde{P}}(F)\cap\mathfrak{o}}\sum_{n\in N_{P}(F)}f(\mathrm{Ad}(nx)^{-1}(Y)).$$

For $T \in \mathfrak{a}_0$ and $x \in H(F) \setminus H(\mathbb{A})$, define

$$j_{f,\mathfrak{o}}^{T}(x) := \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{\delta \in P(F) \setminus H(F)} \widehat{\tau}_{P}^{H}(H_{P}(\delta x) - T) \cdot j_{f,P,\mathfrak{o}}(\delta x).$$

By [3, Lemma 5.1], we know that the sum over $\delta \in P(F) \setminus H(F)$ is finite.

LEMMA 8.1. Let P be a standard parabolic subgroup of H and $\mathfrak{o} \in \mathcal{O}_{rs}$. For $Y \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o}$, the map

$$N_P \to \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}, n \mapsto \operatorname{Ad}(n^{-1})(Y) - Y$$

is an F-isomorphism of algebraic varieties and preserves the Haar measures on A-points.

PROOF. Recall that there are two cases considered in Section 3.4. First let us focus on Case I. In this case, we can suppose

$$P = \begin{pmatrix} \operatorname{Res}_{E/F}GL_{n_1,D'} & \operatorname{Res}_{E/F}Mat_{n_1 \times n_2,D'} & \cdots & \operatorname{Res}_{E/F}Mat_{n_1 \times n_l,D'} \\ & \operatorname{Res}_{E/F}GL_{n_2,D'} & \cdots & \operatorname{Res}_{E/F}Mat_{n_2 \times n_l,D'} \\ & & \ddots & \vdots \\ & & & \operatorname{Res}_{E/F}GL_{n_l,D'} \end{pmatrix}$$

Then

$$\widetilde{P} = \begin{pmatrix} GL_{n_1,D} & Mat_{n_1 \times n_2,D} & \cdots & Mat_{n_1 \times n_l,D} \\ GL_{n_2,D} & \cdots & Mat_{n_2 \times n_l,D} \\ & & \ddots & \vdots \\ & & & GL_{n_l,D} \end{pmatrix}$$

.

We have chosen an element $\tau \in D^{\times}$ in Section 3.4. Recall also Proposition 3.12. Let

$$Y = \left(\begin{array}{cc} Y_1 & & \\ & \ddots & \\ & & Y_l \end{array}\right) \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o},$$

where $Y_i \in GL_{n_i}(D')\tau$ for $1 \leq i \leq l$, and

$$n = \begin{pmatrix} 1 & n_{12} & \cdots & n_{1l} \\ 1 & \cdots & n_{2l} \\ & \ddots & \vdots \\ & & & 1 \end{pmatrix} \in N_P,$$

where $n_{ij} \in \text{Res}_{E/F} Mat_{n_i \times n_j, D'}$ for $1 \le i < j \le l$. Then

$$Yn - nY = \begin{pmatrix} 0 & Y_1n_{12} - n_{12}Y_2 & \cdots & Y_1n_{1l} - n_{1l}Y_l \\ 0 & \cdots & Y_2n_{2l} - n_{2l}Y_l \\ & \ddots & \vdots \\ 0 & & 0 \end{pmatrix} \in \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}.$$

Now we claim that the morphism of F-affine spaces

$$\operatorname{Res}_{E/F} Mat_{n_i \times n_j, D'} \to (\operatorname{Res}_{E/F} Mat_{n_i \times n_j, D'})\tau$$
$$n_{ij} \mapsto Y_i n_{ij} - n_{ij} Y_j$$

induces an *F*-linear isomorphism on *F*-points. In fact, since it gives an *F*-linear map between finite dimensional linear spaces of the same dimension, we only need to prove that this map is injective. If $Y_i n_{ij} - n_{ij}Y_j = 0$, then $Y_i^2 n_{ij} = Y_i n_{ij}Y_j = n_{ij}Y_j^2$. We view this as an equation of matrices with entries in *D'* or its base change to an algebraic closure of *E*. Since *Y* is regular semi-simple, Y^2 is regular semi-simple in $\mathfrak{h}(F)$ (viewed as a central simple algebra over *E*), so Y_i^2 and Y_j^2 have no common eigenvalue. By the classical theory of Sylvester equation, we know that $n_{ij} = 0$ and conclude.

Using this claim, we see that the map

$$N_P \to \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}, n \mapsto Yn - nY$$

is an *F*-isomorphism of algebraic varieties and preserves the Haar measure on A-points. Notice that $\operatorname{Ad}(n^{-1})(Y) - Y = n^{-1}(Yn - nY)$. It is easy to see that here n^{-1} functions as some translation $Y_i n_{ij} - n_{ij} Y_j \mapsto Y_i n_{ij} - n_{ij} Y_j + \sum_{k>j} (a \text{ polynomial of } n_{i'j'}, i' > i, j' \leq j \text{ or } i' \geq i, j' < j) \cdot (Y_k n_{kj} - n_{kj} Y_j),$

so an analogous assertion still holds for the map $n \mapsto \operatorname{Ad}(n^{-1})(Y) - Y$.

Next let us turn to **Case II** whose proof is close to the first one. In this case, we may suppose

$$P = \begin{pmatrix} \operatorname{Res}_{E/F}GL_{\frac{n_1}{2}, D\otimes_F E} & \operatorname{Res}_{E/F}Mat_{\frac{n_1}{2} \times \frac{n_2}{2}, D\otimes_F E} & \cdots & \operatorname{Res}_{E/F}Mat_{\frac{n_1}{2} \times \frac{n_l}{2}, D\otimes_F E} \\ & \operatorname{Res}_{E/F}GL_{\frac{n_2}{2}, D\otimes_F E} & \cdots & \operatorname{Res}_{E/F}Mat_{\frac{n_2}{2} \times \frac{n_l}{2}, D\otimes_F E} \\ & & \ddots & & \vdots \\ & & & \operatorname{Res}_{E/F}GL_{\frac{n_l}{2}, D\otimes_F E} \end{pmatrix}.$$

Then

$$\widetilde{P} = \begin{pmatrix} GL_{n_1,D} & Mat_{n_1 \times n_2,D} & \cdots & Mat_{n_1 \times n_l,D} \\ & GL_{n_2,D} & \cdots & Mat_{n_2 \times n_l,D} \\ & & \ddots & \vdots \\ & & & GL_{n_l,D} \end{pmatrix}$$

We have chosen an element $\tau \in GL_2(D)$ in Section 3.4. Recall again Proposition 3.12. Let

$$Y = \left(\begin{array}{cc} Y_1 & & \\ & \ddots & \\ & & Y_l \end{array}\right) \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o},$$

where $Y_i \in GL_{\frac{n_i}{2}}(D \otimes_F E)\tau$ for $1 \leq i \leq l$, and

$$n = \begin{pmatrix} 1 & n_{12} & \cdots & n_{1l} \\ 1 & \cdots & n_{2l} \\ & & \ddots & \vdots \\ & & & 1 \end{pmatrix} \in N_P,$$

where $n_{ij} \in \text{Res}_{E/F} Mat_{\frac{n_i}{2} \times \frac{n_j}{2}, D \otimes_F E}$ for $1 \leq i < j \leq l$. Then

$$Yn - nY = \begin{pmatrix} 0 & Y_1n_{12} - n_{12}Y_2 & \cdots & Y_1n_{1l} - n_{1l}Y_l \\ 0 & \cdots & Y_2n_{2l} - n_{2l}Y_l \\ & \ddots & \vdots \\ 0 & & 0 \end{pmatrix} \in \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}.$$

As in the proof of the first case, we show that the morphism of F-affine spaces

$$\begin{aligned} \operatorname{Res}_{E/F} Mat_{\frac{n_i}{2} \times \frac{n_j}{2}, D \otimes_F E} &\to (\operatorname{Res}_{E/F} Mat_{\frac{n_i}{2} \times \frac{n_j}{2}, D \otimes_F E}) \tau \\ n_{ij} &\mapsto Y_i n_{ij} - n_{ij} Y_j \end{aligned}$$

induces an F-linear isomorphism on F-points. This implies that the map

$$N_P \to \mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s}, n \mapsto Yn - nY$$

is an *F*-isomorphism of algebraic varieties and preserves the Haar measure on A-points. By an argument similar to that in the first case, we deduce that an analogous assertion is still true for the map $n \mapsto \operatorname{Ad}(n^{-1})(Y) - Y$.

THEOREM 8.2. For all $T \in T_+ + \mathfrak{a}_{P_0}^+$ and $\mathfrak{o} \in \mathcal{O}_{rs}$, we have

$$\int_{H(F)\backslash H(\mathbb{A})^1} |j_{f,\mathfrak{o}}^T(x)| dx < \infty$$

and

$$J_{\mathfrak{o}}^{H,T}(f) = \int_{H(F) \setminus H(\mathbb{A})^1} j_{f,\mathfrak{o}}^T(x) dx.$$

PROOF. As in the proof of Theorem 4.2, by the left invariance of $j_{f,P,\mathfrak{o}}$ by P(F), we reduce the first statement to

$$\int_{P_1(F)\backslash H(\mathbb{A})^1} \chi^T_{P_1,P_2}(x) |j_{P_1,P_2,\mathfrak{o}}(x)| dx < \infty,$$

where $P_1 \subsetneq P_2$ are a pair of standard parabolic subgroups of H and for $x \in P_1(F) \setminus H(\mathbb{A})$, we put

$$j_{P_1,P_2,\mathfrak{o}}(x) := \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} j_{f,P,\mathfrak{o}}(x)$$

Additionally,

$$j_{f,P,\mathfrak{o}}(x) = \sum_{\{R: P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{Y \in (\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)} \sum_{n \in N_P(F)} f(\mathrm{Ad}(nx)^{-1}(\xi + Y)).$$

From Lemma 8.1, we have

$$\begin{split} j_{f,P,\mathfrak{o}}(x) &= \sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{Y \in (\mathfrak{n}_{\widetilde{R}}^{\widetilde{P}} \cap \mathfrak{s})(F)} \sum_{u \in (\mathfrak{n}_{\widetilde{P}} \cap \mathfrak{s})(F)} f(\mathrm{Ad}(x^{-1})(\xi + Y + u)) \\ &= \sum_{\{R:P_1 \subseteq R \subseteq P\}} \sum_{\xi \in \widetilde{m}_{\widetilde{P_1}}^{\widetilde{R}}(F) \cap \mathfrak{o}} \sum_{Y \in (\mathfrak{n}_{\widetilde{R}} \cap \mathfrak{s})(F)} f(\mathrm{Ad}(x^{-1})(\xi + Y)). \end{split}$$

Hence

$$j_{P_{1},P_{2},\mathfrak{o}}(x) = \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{H})} \left(\sum_{\{R:P_{1}\subseteq R\subseteq P\}} \sum_{\xi\in \tilde{m}_{P_{1}}^{\tilde{R}}(F)\cap\mathfrak{o}} \sum_{Y\in(\mathfrak{n}_{\tilde{R}}\cap\mathfrak{s})(F)} f(\mathrm{Ad}(x^{-1})(\xi+Y)) \right)$$
$$= \sum_{\{R:P_{1}\subseteq R\subseteq P_{2}\}} \sum_{\xi\in \tilde{m}_{P_{1}}^{\tilde{R}}(F)\cap\mathfrak{o}} \left(\sum_{\{P:R\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{H})} \right) \sum_{Y\in(\mathfrak{n}_{\tilde{R}}\cap\mathfrak{s})(F)} f(\mathrm{Ad}(x^{-1})(\xi+Y)).$$

Using [3, Proposition 1.1], we get

$$j_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_H)} \sum_{\xi \in \widetilde{m}_{\widetilde{P_1}}^{\widetilde{P_2}}(F) \cap \mathfrak{o}} \sum_{Y \in (\mathfrak{n}_{\widetilde{P_2}} \cap \mathfrak{s})(F)} f(\mathrm{Ad}(x^{-1})(\xi + Y)).$$

Applying Lemma 8.1 again, we obtain

$$j_{P_1,P_2,\mathfrak{o}}(x) = (-1)^{\dim(A_{P_2}/A_H)} \sum_{\xi \in \widetilde{m}_{\widetilde{P_1}}^{\widetilde{P_2}}(F) \cap \mathfrak{o}} \sum_{n_2 \in N_{P_2}(F)} f(\mathrm{Ad}(n_2 x)^{-1}(\xi)).$$

Now we decompose the integral over $x \in P_1(F) \setminus H(\mathbb{A})^1$ into double integrals $n_1 \in N_{P_1}(F) \setminus N_{P_1}(\mathbb{A})$ and $y \in M_{P_1}(F)N_{P_1}(\mathbb{A}) \setminus H(\mathbb{A})^1$ and use the fact that $\chi^T_{P_1,P_2}(x)$ is left invariant under $N_{P_1}(\mathbb{A})$. We have

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(x)|j_{P_{1},P_{2},\mathfrak{o}}(x)|dx\\ &=\int_{M_{P_{1}}(F)N_{P_{1}}(\mathbb{A})\backslash H(\mathbb{A})^{1}}\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})}\chi^{T}_{P_{1},P_{2}}(n_{1}y)\left|\sum_{\xi\in\widetilde{m}^{\widetilde{P_{2}}}_{P_{1}}(F)\cap\mathfrak{o}}\sum_{n_{2}\in N_{P_{2}}(F)}f(\mathrm{Ad}(n_{2}n_{1}y)^{-1}(\xi))\right|dn_{1}dy\\ &\leq\int_{M_{P_{1}}(F)N_{P_{1}}(\mathbb{A})\backslash H(\mathbb{A})^{1}}\chi^{T}_{P_{1},P_{2}}(y)\sum_{\xi\in\widetilde{m}^{\widetilde{P_{2}}}_{P_{2}}(F)\cap\mathfrak{o}}\left(\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})}\sum_{n_{2}\in N_{P_{2}}(F)}|f(\mathrm{Ad}(n_{2}n_{1}y)^{-1}(\xi))|dn_{1}\right)dy. \end{split}$$

Since $P_1 \subseteq P_2$ and $\operatorname{vol}(N_{P_2}(F) \setminus N_{P_2}(\mathbb{A})) = 1$, we see that

$$\begin{split} & \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \sum_{n_2 \in N_{P_2}(F)} |f(\operatorname{Ad}(n_2n_1y)^{-1}(\xi))| dn_1 \\ & = \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{N_{P_2}(F)\setminus N_{P_2}(\mathbb{A})} \sum_{n_2 \in N_{P_2}(F)} |f(\operatorname{Ad}(n_2nn_1y)^{-1}(\xi))| dn dn_1 \\ & = \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{N_{P_2}(\mathbb{A})} |f(\operatorname{Ad}(nn_1y)^{-1}(\xi))| dn dn_1 \\ & = \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{(\mathfrak{n}_{\widetilde{P_2}}\cap\mathfrak{s})(\mathbb{A})} |f(\operatorname{Ad}(n_1y)^{-1}(\xi+U))| dU dn_1, \end{split}$$

where we have applied Lemma 8.1 in the last equality. Hence

$$\begin{split} &\int_{P_{1}(F)\backslash H(\mathbb{A})^{1}} \chi^{T}_{P_{1},P_{2}}(x) |j_{P_{1},P_{2},\mathfrak{o}}(x)| dx \\ &\leq \int_{M_{P_{1}}(F)N_{P_{1}}(\mathbb{A})\backslash H(\mathbb{A})^{1}} \chi^{T}_{P_{1},P_{2}}(y) \sum_{\xi \in \widetilde{m}_{\overline{P_{1}}}^{\overline{P_{2}}}(F)\cap\mathfrak{o}} \left(\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})} \int_{(\mathfrak{n}_{\overline{P_{2}}}\cap\mathfrak{s})(\mathbb{A})} |f(\operatorname{Ad}(n_{1}y)^{-1}(\xi+U))| dU dn_{1} \right) dy \\ &= \int_{P_{1}(F)\backslash H(\mathbb{A})^{1}} \chi^{T}_{P_{1},P_{2}}(x) \sum_{\xi \in \widetilde{m}_{\overline{P_{1}}}^{\overline{P_{2}}}(F)\cap\mathfrak{o}} \int_{(\mathfrak{n}_{\overline{P_{2}}}\cap\mathfrak{s})(\mathbb{A})} |f(\operatorname{Ad}(x^{-1})(\xi+U))| dU dx, \end{split}$$

whose convergence comes from that of the formula (4.0.3) when $R = P_2$.

Next we begin to prove the second statement. From the first statement, now we are authorised to write

$$\int_{H(F)\backslash H(\mathbb{A})^1} j_{f,\mathfrak{o}}^T(x) dx = \sum_{\{P_1, P_2: P_0 \subseteq P_1 \subseteq P_2\}} \int_{P_1(F)\backslash H(\mathbb{A})^1} \chi_{P_1, P_2}^T(x) j_{P_1, P_2, \mathfrak{o}}(x) dx,$$

where

$$j_{P_1,P_2,\mathfrak{o}}(x) = \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} j_{f,P,\mathfrak{o}}(x)$$
$$= \sum_{\{P:P_1 \subseteq P \subseteq P_2\}} (-1)^{\dim(A_P/A_H)} \left(\sum_{Y \in \mathfrak{m}_{\bar{P}}(F) \cap \mathfrak{o}} \sum_{n \in N_P(F)} f(\operatorname{Ad}(nx)^{-1}(Y)) \right).$$

Decompose the integral over $x \in P_1(F) \setminus H(\mathbb{A})^1$ into double integrals over $n_1 \in N_{P_1}(F) \setminus N_{P_1}(\mathbb{A})$ and $y \in M_{P_1}(F) \setminus N_{P_1}(\mathbb{A}) \setminus H(\mathbb{A})^1$. Since $N_{P_1}(F) \setminus N_{P_1}(\mathbb{A})$ is compact, from Lemma 8.1 and [53, §41], we know

that

$$\sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \sum_{n \in N_{P}(F)} |f(\operatorname{Ad}(nn_{1}y)^{-1}(Y))| = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \sum_{u \in (\mathfrak{n}_{\tilde{P}} \cap \mathfrak{s})(F)} |f(\operatorname{Ad}(n_{1}y)^{-1}(Y+u))|$$

is bounded on $n_1 \in N_{P_1}(F) \setminus N_{P_1}(\mathbb{A})$. Then using the fact that $\chi^T_{P_1,P_2}(x)$ is left invariant under $N_{P_1}(\mathbb{A})$, we have

$$\int_{H(F)\backslash H(\mathbb{A})^{1}} j_{f,\mathfrak{o}}^{T}(x)dx = \sum_{\{P_{1},P_{2}:P_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{M_{P_{1}}(F)N_{P_{1}}(\mathbb{A})\backslash H(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(y) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{Y\in\mathfrak{m}_{\bar{P}}(F)\cap\mathfrak{o}} \left(\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})} \sum_{n\in N_{P}(F)} f(\operatorname{Ad}(nn_{1}y)^{-1}(Y))dn_{1} \right) dy.$$

Since $P_1 \subseteq P$ and $\operatorname{vol}(N_P(F) \setminus N_P(\mathbb{A})) = 1$, we see that

$$\int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \sum_{n\in N_P(F)} f(\operatorname{Ad}(nn_1y)^{-1}(Y))dn_1$$

$$= \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{N_P(F)\setminus N_P(\mathbb{A})} \sum_{n\in N_P(F)} f(\operatorname{Ad}(nn_2n_1y)^{-1}(Y))dn_2dn_1$$

$$= \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{N_P(\mathbb{A})} f(\operatorname{Ad}(nn_1y)^{-1}(Y))dndn_1$$

$$= \int_{N_{P_1}(F)\setminus N_{P_1}(\mathbb{A})} \int_{(\mathfrak{n}_{\tilde{F}}\cap\mathfrak{s})(\mathbb{A})} f(\operatorname{Ad}(n_1y)^{-1}(Y+U))dUdn_1,$$

where we have applied Lemma 8.1 in the last equality. Therefore, we have

$$\begin{split} \int_{H(F)\backslash H(\mathbb{A})^{1}} j_{f,\mathfrak{o}}^{T}(x)dx &= \sum_{\{P_{1},P_{2}:P_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{M_{P_{1}}(F)N_{P_{1}}(\mathbb{A})\backslash H(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(y) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{H})} \\ &= \sum_{Y\in\mathfrak{m}_{\bar{P}}(F)\cap\mathfrak{o}} \left(\int_{N_{P_{1}}(F)\backslash N_{P_{1}}(\mathbb{A})} \int_{(\mathfrak{n}_{\bar{P}}\cap\mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(n_{1}y)^{-1}(Y+U))dUdn_{1} \right) dy \\ &= \sum_{\{P_{1},P_{2}:P_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{P_{1}(F)\backslash H(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x) \sum_{\{P:P_{1}\subseteq P\subseteq P_{2}\}} (-1)^{\dim(A_{P}/A_{H})} \\ &\quad \cdot \left(\sum_{Y\in\mathfrak{m}_{\bar{P}}(F)\cap\mathfrak{o}} \int_{(\mathfrak{n}_{\bar{P}}\cap\mathfrak{s})(\mathbb{A})} f(\mathrm{Ad}(x^{-1})(Y+U))dU \right) dx \\ &= \sum_{\{P_{1},P_{2}:P_{0}\subseteq P_{1}\subseteq P_{2}\}} \int_{P_{1}(F)\backslash H'(\mathbb{A})^{1}} \chi_{P_{1},P_{2}}^{T}(x)k_{P_{1},P_{2},\mathfrak{o}}(x)dx. \end{split}$$

Thanks to Theorem 4.2, we are able to write

$$J_{\mathfrak{o}}^{H,T}(f) = \int_{H(F)\setminus H(\mathbb{A})^1} k_{f,\mathfrak{o}}^T(x) dx = \sum_{\{P_1, P_2: P_0 \subseteq P_1 \subseteq P_2\}} \int_{P_1(F)\setminus H(\mathbb{A})^1} \chi_{P_1, P_2}^T(x) k_{P_1, P_2, \mathfrak{o}}(x) dx,$$

completes our proof.

which completes our proof.

9. Weighted orbital integrals

Let $\mathfrak{o} \in \mathcal{O}_{rs}$ (see Section 3.3). There is an element $Y_1 \in \mathfrak{o}$ and a standard parabolic subgroup P_1 of H such that $Y_1 \in \mathfrak{m}_{\widetilde{P_1}}(F)$ but Y_1 can not be $M_{P_1}(F)$ -conjugate to an element in \widetilde{R} (or equivalently in $M_{\widetilde{R}}$ by Proposition 3.3) for any standard parabolic subgroup $R \subsetneq P_1$. We call such Y_1 an elliptic element in $(\mathfrak{m}_{\widetilde{P_1}} \cap \mathfrak{s})(F)$. For P_1 and P_2 a pair of standard parabolic subgroups of H, denote by $\Omega^H(\mathfrak{a}_{P_1},\mathfrak{a}_{P_2})$ the set (perhaps empty) of distinct isomorphisms from \mathfrak{a}_{P_1} to \mathfrak{a}_{P_2} obtained by restriction of elements in Ω^H .

LEMMA 9.1. Let P be a standard parabolic subgroup of H. Let $Y \in (\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$ be a regular semisimple element in \mathfrak{s} . Then Y is an elliptic element in $(\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$ if and only if the maximal F-split torus in H_Y is A_P .

PROOF. It is evident that $H_Y \subseteq H_{Y^2}$. From $Y \in (\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$, one knows that $A_P \subseteq H_Y$. Since Y is regular semi-simple in \mathfrak{s} , one deduces that $Y^2 \in \mathfrak{m}_P(F)$ is regular semi-simple in $\mathfrak{h}(F)$ (viewed as a central simple algebra over E). Thus $H_{Y^2} \subseteq M_P$.

On the one hand, suppose that Y is an elliptic element in $(\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$. If the maximal F-split torus in H_Y is not A_P , then there exists a F-split torus A_* such that $A_P \subsetneq A_* \subseteq H_Y$. For $A_* \subseteq H_Y \subseteq H_{Y^2} \subseteq M_P$, the centralizer $\operatorname{Cent}_{M_P}(A_*)$ of A_* in M_P is a Levi subgroup of M_P . There exists $m \in M_P(F)$ such that $\operatorname{Ad}(m)(\operatorname{Cent}_{M_P}(A_*)) = M_{R_*}$ for some standard parabolic subgroup R_* of M_P . Then $\operatorname{Ad}(m)(A_*) \subseteq A_{R_*}$ and the centralizer of $\operatorname{Ad}(m)(A_*)$ in M_P is M_{R_*} . Let R be the unique standard parabolic subgroup of H such that $R \subseteq P$ and that $R \cap M_P = R_*$. Then $A_R = A_{R_*}$ and $M_R = M_{R_*}$. Since τ (see Section 3.4) commutes with A_0 , by Proposition 3.12, the centralizer of $\operatorname{Ad}(m)(A_*)$ in $\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s}$ is $\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s}$. From $A_* \subseteq H_Y$, one obtains $\operatorname{Ad}(m)(A_*) \subseteq H_{\operatorname{Ad}(m)(Y)}$. Since $\operatorname{Ad}(m)(Y) \in (\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$, we deduce that $\operatorname{Ad}(m)(Y) \in (\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})(F)$. Because $A_P \subsetneq A_*$ and $\operatorname{Ad}(m)(A_*) \subseteq A_R$, we have $A_P \subsetneq A_R$ and thus $R \subsetneq P$. That is to say, Y is not an elliptic element in $(\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$. It is a contradiction. This proves one direction.

On the other hand, suppose that the maximal F-split torus in H_Y is A_P . If Y is not an elliptic element in $(\mathfrak{m}_{\widetilde{P}} \cap \mathfrak{s})(F)$, there exists $m \in M_P(F)$ such that $\operatorname{Ad}(m)(Y) \in (\mathfrak{m}_{\widetilde{R}} \cap \mathfrak{s})(F)$ for some standard parabolic subgroup $R \subsetneq P$. Then $A_R \subseteq H_{\operatorname{Ad}(m)(Y)}$, i.e., $\operatorname{Ad}(m^{-1})(A_R) \subseteq H_Y$. For $R \subsetneq P$, one sees that $A_P \subsetneq \operatorname{Ad}(m^{-1})(A_R)$. That is to say, $\operatorname{Ad}(m^{-1})(A_R)$ is a strictly larger split torus than A_P in H_Y . It contradicts our hypothesis. This proves the other direction. \Box

THEOREM 9.2. Let $\mathfrak{o} \in \mathcal{O}_{rs}$, P_1 be a standard parabolic subgroup of H and $Y_1 \in \mathfrak{o}$ be an elliptic element in $(\mathfrak{m}_{\widetilde{P_1}} \cap \mathfrak{s})(F)$. For $f \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we have

$$J^{H}_{\mathbf{o}}(f) = \operatorname{vol}(A^{\infty}_{P_{1}}H_{Y_{1}}(F) \setminus H_{Y_{1}}(\mathbb{A})) \cdot \int_{H_{Y_{1}}(\mathbb{A}) \setminus H(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(Y_{1})) v_{P_{1}}(x) dx$$

where $v_{P_1}(x)$ is left-invariant under $H_{Y_1}(\mathbb{A})$ and is equal to the volume of the projection onto $\mathfrak{a}_{P_1}^H$ of the convex hull of $\{-H_Q(x)\}$, where Q takes over all semi-standard parabolic subgroups of H with $M_Q = M_{P_1}$.

REMARK 9.3. The weights that we obtain for regular semi-simple orbits are the same as Arthur's in [3, p. 951]. These weights are also the same as those (see [39, p. 131]) appearing in the twisted trace formula for $H \rtimes \sigma$, where σ acts on H by $\operatorname{Ad}(\tau)$ (see Section 3.4 for the choice of τ). Notice that the action σ stabilises P_0 and M_0 . All standard parabolic subgroups P of H are σ -stable and σ fixes $\mathfrak{a}_P = \mathfrak{a}_{\widetilde{P}}$.

PROOF OF THEOREM 9.2. Let P be any standard parabolic subgroup of H and $Y \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o}$. There exists a standard parabolic subgroup $P_2 \subseteq P$ and Y_2 an elliptic element in $(\mathfrak{m}_{\widetilde{P}_2} \cap \mathfrak{s})(F)$ such that Y_2 is $M_P(F)$ -conjugate to Y. By Lemma 9.1, the maximal F-split torus in H_{Y_2} is A_{P_2} . Any element in H(F) which conjugates Y_1 and Y_2 will conjugate A_{P_1} and A_{P_2} . It follows that there exists $s \in \Omega^H(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_2})$ and $m \in M_P(F)$ such that

$$Y = \operatorname{Ad}(m\omega_s)(Y_1).$$

Suppose that $P_3 \subseteq P$ is another standard parabolic subgroup, $s' \in \Omega^H(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_3})$ and $m' \in M_P(F)$ such that

$$Y = \operatorname{Ad}(m'\omega_{s'})(Y_1)$$

Then there is $\zeta \in H_Y(F)$ such that

$$m'\omega_{s'} = \zeta m\omega_s.$$

From $H_Y \subseteq M_P$, we see that

 $\omega_{s'} = \xi \omega_s$

for some $\xi \in M_P(F)$. Denote by $\Omega^H(\mathfrak{a}_{P_1}; P)$ the set of $s \in \bigcup_{\mathfrak{a}_{P_2}} \Omega^H(\mathfrak{a}_{P_1}, \mathfrak{a}_{P_2})$ satisfying $\mathfrak{a}_P \subseteq s\mathfrak{a}_{P_1}$ and $s^{-1}\alpha > 0$ for each $\alpha \in \Delta_{P_2}^P$. In sum, for any given P a standard parabolic subgroup of H and $Y \in \mathfrak{m}_{\widetilde{P}}(F) \cap \mathfrak{o}$, there is a unique $s \in \Omega^H(\mathfrak{a}_{P_1}; P)$ such that $Y = \mathrm{Ad}(m\omega_s)(Y_1)$ for some $m \in M_P(F)$. For $x \in P(F) \setminus H(\mathbb{A})$, we have

$$j_{f,P,\mathfrak{o}}(x) = \sum_{Y \in \mathfrak{m}_{\tilde{P}}(F) \cap \mathfrak{o}} \sum_{n \in N_{P}(F)} f(\operatorname{Ad}(nx)^{-1}(Y))$$

$$= \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}}; P)} \sum_{m \in M_{P,\operatorname{Ad}(\omega_{s})(Y_{1})}(F) \setminus M_{P}(F)} \sum_{n \in N_{P}(F)} f(\operatorname{Ad}(mnx)^{-1}\operatorname{Ad}(\omega_{s})(Y_{1}))$$

$$= \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}}; P)} \sum_{m \in M_{P,\operatorname{Ad}(\omega_{s})(Y_{1})}(F) \setminus P(F)} f(\operatorname{Ad}(mx)^{-1}\operatorname{Ad}(\omega_{s})(Y_{1})),$$

where $M_{P,\operatorname{Ad}(\omega_s)(Y_1)}$ denotes the centralizer of $\operatorname{Ad}(\omega_s)(Y_1)$ in M_P . Then for $T \in \mathfrak{a}_0$ and $x \in H(F) \setminus H(\mathbb{A})$, we deduce that

$$j_{f,\mathfrak{o}}^{T}(x) = \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{\delta \in P(F) \setminus H(F)} \widehat{\tau}_{P}^{H}(H_{P}(\delta x) - T) \cdot j_{f,P,\mathfrak{o}}(\delta x)$$

$$= \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{\delta \in P(F) \setminus H(F)} \widehat{\tau}_{P}^{H}(H_{P}(\delta x) - T)$$

$$\cdot \left(\sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}}; P)} \sum_{m \in M_{P,\mathrm{Ad}(\omega_{s})(Y_{1})}(F) \setminus P(F)} f(\mathrm{Ad}(m\delta x)^{-1}\mathrm{Ad}(\omega_{s})(Y_{1}))\right)$$

$$= \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{s \in \Omega^{H}(\mathfrak{a}_{P_{1}}; P)} \sum_{\delta \in M_{P,\mathrm{Ad}(\omega_{s})(Y_{1})}(F) \setminus H(F)} \widehat{\tau}_{P}^{H}(H_{P}(\delta x) - T)$$

$$\cdot f(\mathrm{Ad}(\delta x)^{-1}\mathrm{Ad}(\omega_{s})(Y_{1})).$$

Notice that the centralizer of $Ad(\omega_s)(Y_1)$ in H is actually contained in M_P , so

$$j_{f,\mathfrak{o}}^{T}(x) = \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{s\in\Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{\delta\in H_{\omega_{s}Y_{1}\omega_{s}^{-1}}(F)\setminus H(F)} \hat{\tau}_{P}^{H}(H_{P}(\delta x) - T) \cdot f(\mathrm{Ad}(\delta x)^{-1}\mathrm{Ad}(\omega_{s})(Y_{1}))$$

$$= \sum_{\{P:P_{0}\subseteq P\}} (-1)^{\dim(A_{P}/A_{H})} \sum_{s\in\Omega^{H}(\mathfrak{a}_{P_{1}};P)} \sum_{\delta\in H_{Y_{1}}(F)\setminus H(F)} \hat{\tau}_{P}^{H}(H_{P}(\omega_{s}\delta x) - T) \cdot f(\mathrm{Ad}(\delta x)^{-1}(Y_{1})).$$

For $y \in H(\mathbb{A})$, we write

$$\chi_T(y) := \sum_{\{P: P_0 \subseteq P\}} (-1)^{\dim(A_P/A_H)} \sum_{s \in \Omega^H(\mathfrak{a}_{P_1}; P)} \widehat{\tau}_P^H(H_P(\omega_s y) - T).$$

Then

$$j_{f,\mathbf{o}}^T(x) = \sum_{\delta \in H_{Y_1}(F) \setminus H(F)} f(\mathrm{Ad}(\delta x)^{-1}(Y_1)) \cdot \chi_T(\delta x)$$

For $T \in T_+ + \mathfrak{a}_{P_0}^+$, applying Theorem 8.2 and the fact that $j_{f,\mathfrak{o}}^T(x)$ is left invariant by A_H^∞ , we have

$$\begin{aligned} J_{\mathbf{o}}^{H,T}(f) &= \int_{H(F)\backslash H(\mathbb{A})^1} j_{f,\mathbf{o}}^T(x) dx \\ &= \int_{A_H^{\infty} H(F)\backslash H(\mathbb{A})} \left(\sum_{\delta \in H_{Y_1}(F)\backslash H(F)} f(\operatorname{Ad}(\delta x)^{-1}(Y_1)) \cdot \chi_T(\delta x) \right) dx \end{aligned}$$

Then we obtain

(9.0.1)
$$J_{\mathfrak{o}}^{H,T}(f) = \operatorname{vol}(A_{P_1}^{\infty}H_{Y_1}(F) \setminus H_{Y_1}(\mathbb{A})) \cdot \int_{H_{Y_1}(\mathbb{A}) \setminus H(\mathbb{A})} f(\operatorname{Ad}(x^{-1})(Y_1)) v_{P_1}(x,T) dx,$$

where

$$v_{P_1}(x,T) := \int_{A_H^\infty \setminus A_{P_1}^\infty} \chi_T(ax) da.$$

Here we have used the fact that $v_{P_1}(x,T)$ is well-defined and left-invariant under $H_{Y_1}(\mathbb{A}) \subseteq M_P(\mathbb{A})$. Moreover, $v_{P_1}(x,T)$ is equal to the volume of the projection onto $\mathfrak{a}_{P_1}^H$ of the convex hull of $\{T_Q - H_Q(x)\}$, where T_Q denotes the projection of sT in \mathfrak{a}_Q for any $s \in \Omega^H$ satisfying $sP_0 \subseteq Q$, and Q takes over all semi-standard parabolic subgroups of H with $M_Q = M_{P_1}$. These properties follow from [3, p. 951]. We have also assumed the finiteness of $\operatorname{vol}(A_{P_{P}}^{\infty}H_{Y_{1}}(F)\setminus H_{Y_{1}}(\mathbb{A}))$, which results from Lemma 9.1.

In the end, we may conclude by taking contant terms of both sides of (9.0.1).

10. The weighted fundamental lemma

In this section, we turn to the local setting and change the notation by letting F be a non-archimedean local field of characteristic 0.

10.1. (H, M)-families associated to local weighted orbital integrals. Suppose that H is a reductive group defined over F. Fix a maximal compact subgroup K of H(F) which is admissible relative to M_0 in the sense of [5, p. 9]. Again we choose the standard one for an inner of GL_n . For a parabolic subgroup P of H and $x \in H(F)$, we have H(F) = P(F)K by Iwasawa decomposition and define $H_P(x)$ as in Section 2.2 by replacing $|\cdot|_{\mathbb{A}}$ with the normalised absolute value $|\cdot|_F$ on F. Suppose that M is a Levi subgroup of H containing M_0 . Let $\mathcal{P}(M)$ be the set of parabolic subgroups of H with Levi component M. According to [5, p. 40-41],

$$v_P(\lambda, x) := e^{-\lambda(H_P(x))}, \forall \lambda \in i\mathfrak{a}_M^*, P \in \mathcal{P}(M),$$

is an (H, M)-family in the sense of [5, p. 36]. Let $P \in \mathcal{P}(M)$ and Q be a parabolic subgroup of H containing P. Define

$$\theta_P^Q(\lambda) := \operatorname{vol}(\mathfrak{a}_P^Q/\mathbb{Z}(\Delta_P^Q)^{\vee})^{-1} \prod_{\alpha^{\vee} \in (\Delta_P^Q)^{\vee}} \lambda(\alpha^{\vee})$$

where $\mathbb{Z}(\Delta_P^Q)^{\vee}$ denotes the lattice in \mathfrak{a}_P^Q generated by $(\Delta_P^Q)^{\vee}$. Then we obtain a function

$$v_M^Q(x) := \lim_{\lambda \to 0} \sum_{\{P \in \mathcal{P}(M): P \subseteq Q\}} v_P(\lambda, x) \theta_P^Q(\lambda)^{-1}, \forall x \in H(F).$$

10.2. Matching of orbits. Assume that F has odd residue characteristic and that E is an unramified quadratic extension over F. Let $G := GL_{2n}$ and $H := \operatorname{Res}_{E/F}GL_{n,E}$ be the centralizer of E^{\times} in G. Let $H' := GL_n \times GL_n$ be the subgroup of G by diagonal embedding. Denote by \mathcal{O}_F the ring of integers of F. All of G, H and H' are regarded as group schemes over \mathcal{O}_F . Let \mathfrak{s} (resp. \mathfrak{s}') be the tangent space at the neutral element of G/H (resp. G/H'), which is viewed as a subspace of \mathfrak{g} . Here we can and shall identify $\mathfrak{s}(F) \simeq \mathfrak{h}(F)$, on which H(F) acts by twisted conjugation, i.e., $h \cdot Y = hY\overline{h}^{-1}$, where \overline{h} denotes the nontrivial Galois conjugate of $h \in \mathfrak{h}(F)$. In fact, we can write concretely $G(F) \simeq \left\{ \left(\begin{array}{c} A \\ \overline{B} \end{array} \right) : A, B \in GL_n(E) \right\}, H(F) \simeq \left\{ \left(\begin{array}{c} A \\ \overline{A} \end{array} \right) : A \in GL_n(E) \right\} \text{ and } \mathfrak{s}(F) \simeq \left\{ \left(\begin{array}{c} 0 \\ \overline{B} \end{array} \right) : B \in \mathfrak{gl}_n(E) \right\}, \text{ and choose } \tau = \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix} \text{ in this form. Additionally, if we write } G(F)$

in the usual form of invertible $2n \times 2n$ -matrices over F and $H'(F) = \left\{ \begin{pmatrix} A \\ B \end{pmatrix} : A, B \in GL_n(F) \right\}$, then $\mathfrak{s}'(F) = \left\{ \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} : A, B \in \mathfrak{gl}_n(F) \right\} \simeq \mathfrak{gl}_n(F) \oplus \mathfrak{gl}_n(F)$. We see that H'(F) acts on $\mathfrak{s}'(F)$ by

conjugation, i.e., $(x_1, x_2) \cdot (A, B) = (x_1 A x_2^{-1}, x_2 B x_1^{-1}).$

Recall [10, Lemma 1.1 of Chapter 1] that the norm map $Y \mapsto Y\overline{Y}$ induces an injection from the set of twisted conjugacy classes in $GL_n(E)$ to the set of conjugacy classes in $GL_n(F)$, whose image is denoted by $N(GL_n(E))$; in particular, we write NE^{\times} for $N(GL_1(E))$. We have the notions of regular semisimple elements in $\mathfrak{s}(F)$ and $\mathfrak{s}'(F)$ (whose sets are denoted by $\mathfrak{s}_{rs}(F)$ and $\mathfrak{s}'_{rs}(F)$ respectively) as before, which are explicitly described as follows.

PROPOSITION 10.1. 1) An element Y of $\mathfrak{s}(F)$ is regular semi-simple if and only if \overline{YY} belongs to $GL_n(E)$ and is regular semi-simple. The map $Y \mapsto Y\overline{Y}$ from $\mathfrak{s}(F)$ to $GL_n(E)$ induces an injection from the set of H(F)-conjugacy classes of regular semi-simple elements in $\mathfrak{s}(F)$ into the set of regular semi-simple conjugacy classes in $GL_n(F)$.

2) An element X of $\mathfrak{s}'(F)$ is regular semi-simple if and only if it is H'(F)-conjugate to an element of the form

$$X(A) := \left(\begin{array}{cc} 0 & 1_n \\ A & 0 \end{array}\right)$$

with $A \in GL_n(F)$ being regular semi-simple. The map $\begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \mapsto AB$ from $\mathfrak{s}'(F)$ to $GL_n(F)$ induces a bijection between the set of H'(F)-conjugacy classes of regular semi-simple elements in $\mathfrak{s}'(F)$ and the set of regular semi-simple conjugacy classes in $GL_n(F)$.

PROOF. 1) is contained in [24, Lemma 2.1], while 2) is proved in [31, Proposition 2.1 and Lemma 2.1]. \Box

To sum up, the composition of the map in 1) and the inverse of the map in 2) above induces an injection from the set of H(F)-orbits in $\mathfrak{s}_{rs}(F)$ into the set of H'(F)-orbits in $\mathfrak{s}'_{rs}(F)$. We shall say that $Y \in \mathfrak{s}_{rs}(F)$ and $X \in \mathfrak{s}'_{rs}(F)$ have matching orbits if their orbits are matched under this injection. Alternatively, this can be canonically characterized by an identification of categorical quotients $\mathfrak{s}//H \simeq \mathfrak{s}'//H'$ (see Proposition 3.5 and Proposition 3.3 in Chapter 2). With our identification $\mathfrak{s}(F) \simeq \mathfrak{h}(F)$, we see that $Y \in \mathfrak{s}_{rs}(F)$ and $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}'_{rs}(F)$ have matching orbits if and only if the characteristic polynomial of $Y\overline{Y} \in GL_n(E)$ equals that of $AB \in GL_n(F)$.

10.3. Matching of Levi subgroups involved. We recall some terminology in §3.4 and §5.2 in Chapter 2. The subgroup of diagonal matrices in G is a common minimal Levi subgroup of G and H'. We also fix a minimal semi-standard parabolic subgroup of H' to be the group of products of upper triangular matrices. We say that a semi-standard parabolic subgroup of G is "relatively standard" if its intersection with H' is a standard parabolic subgroup of H'. Let $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. We say that a semi-standard parabolic subgroup of H'. Let $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. We say that a semi-standard parabolic subgroup of H'. Let $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. We say that a semi-standard parabolic subgroup of G is " ω -stable" if $Ad(\omega)(P) = P$. Recall that if the Lie algebra of a relatively standard parabolic subgroup P of G has non-empty intersection with \mathfrak{s}'_{rs} , then P must be ω -stable (see Proposition 5.1 in Chapter 2).

We shall say that a semi-standard Levi subgroup M' of G is " ω -stable" if $M' = M_{P'}$ for some ω -stable parabolic subgroup P' of G. We should remark that this condition is stronger than $\operatorname{Ad}(\omega)(M') = M'$: for example, the minimal Levi subgroup of diagonal matrices in G is not considered to be ω -stable in our sense. Here ω -stable Levi subgroups of G play the role of semi-standard Levi subsets of $(GL_n \times GL_n) \rtimes \sigma'$ in the sense of [37, §I.1], where σ' exchanges two copies of GL_n . For any linear subspace \mathfrak{v} of \mathfrak{g} , we denote by \mathfrak{v}^{\times} the intersection of \mathfrak{v} and G in \mathfrak{g} . Notice that there is a bijection between the set of semi-standard Levi subgroups of GL_n and the set of semi-standard Levi subgroups of H (resp. the set of ω -stable

Levi subgroups of G) induced by $M_n \mapsto M = \operatorname{Res}_{E/F} M_{n,E}$ (resp. $M_n \mapsto M' = \begin{pmatrix} \mathfrak{m}_n & \mathfrak{m}_n \\ \mathfrak{m}_n & \mathfrak{m}_n \end{pmatrix}^{\times}$); here \mathfrak{m}_n denotes the Lie algebra of M_n . We shall use the notations M_n, M, M' to denote the corresponding semi-standard or ω -stable Levi subgroups of different groups under these bijections after fixing one of the three. We also have bijections among semi-standard or ω -stable parabolic subgroups (denoted by Q_n, Q, Q') of different groups containing these Levi subgroups.

Let M' be an ω -stable Levi subgroup of G. We shall say that $Y \in \mathfrak{m}(F) \cap \mathfrak{s}_{rs}(F)$ and $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$ have M'-matching orbits if in each pair of blocks of \mathfrak{m} and \mathfrak{m}' , their components have matching orbits.

10.4. Transfer factor. Let η be the quadratic character of F^{\times}/NE^{\times} attached to the quadratic field extension E/F. For $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}'_{rs}(F)$, define a transfer factor (see [58, Definition 5.8])

$$\kappa(X) := \eta(\det(A)),$$

which satisfies $\kappa(\operatorname{Ad}(x^{-1})(X)) = \eta(\det(x))\kappa(X)$ for any $x \in H'(F)$.

10.5. Transfer of weighted orbital integrals. Fix the Haar measures on H(F) and H'(F) such that $\operatorname{vol}(H(\mathcal{O}_F)) = \operatorname{vol}(H'(\mathcal{O}_F)) = 1$. For a locally compact and totally disconnected space X, denote by $\mathcal{C}_c^{\infty}(X)$ the \mathbb{C} -linear space of locally constant and compactly supported functions on X.

DEFINITION 10.2. 1) Let M be a semi-standard Levi subgroup of H and Q a parabolic subgroup of H containing M. For $Y \in \mathfrak{m}(F) \cap \mathfrak{s}_{rs}(F)$ and $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we define the weighted orbital integral of f at Y by

$$J_M^Q(Y,f) := \int_{H_Y(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(Y)) v_M^Q(x) dx.$$

2) Let M' be an ω -stable Levi subgroup of G and Q' a parabolic subgroup of G containing M' (thus Q' is ω -stable). For $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we define the weighted η -orbital integral of f' at X by

$$J_{M'}^{\eta,Q'}(X,f') := \int_{H'_X(F) \setminus H'(F)} f'(\mathrm{Ad}(x^{-1})(X))\eta(\det(x))v_{M'}^{Q'}(x)dx.$$

REMARK 10.3. 1) v_M^Q is a local analogue of the weight that we got in Theorem 9.2. By Remark 9.3, it is the same as $v_{(\text{Res}_{E/F}Q_{n,E})\rtimes\sigma}^{(\text{Res}_{E/F}Q_{n,E})\rtimes\sigma}$ in [37, §I.3], where σ is the nontrivial Galois conjugation.

2) $v_{M'}^{Q'}$ is a local analogue of the weight that we got in Theorem 9.2 in Chapter 2. By Remark 9.3 in Chapter 2, it is the same as $v_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}$ in [37, §I.3], where σ' exchanges two copies.

If $Y \in \mathfrak{s}_{rs}(F)$ and $X \in \mathfrak{s}'_{rs}(F)$ have matching orbits, their centralizers H_Y and H'_X are canonically isomorphic. We shall fix compatible Haar measures on them.

DEFINITION 10.4. For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we say that f and f' are strongly associated if for all ω -stable Levi subgroup M' of G and all parabolic subgroup Q' of G containing M' (thus Q' is ω -stable), we have

(1) if $Y \in \mathfrak{m}(F) \cap \mathfrak{s}_{rs}(F)$ and $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$ have M'-matching orbits, then

$$\kappa(X)J_{M'}^{\eta,Q'}(X,f') = J_M^Q(Y,f);$$

(2) if
$$X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$$
 satisfies $\xi(AB) \notin NE^{\times}$ for some $\xi \in X(M_{Q_n})_F$, then $J_{M'}^{\eta,Q'}(X,f') = 0.$

We remark that this definition is inspired by [37, Definition III.3.2] on the base change for GL_n . The following result (cf. [37, Remark III.3.2.(i)]) shows that to check the vanishing statement (2) in the above definition, it suffices to check it for all ω -stable Levi subgroup M' of G such that X is an elliptic element in $\mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$ (i.e. $A_{M'}$ is the maximal F-split torus in H'_X).

PROPOSITION 10.5. Let $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$. The following two conditions are equivalent:

1) for all ω -stable Levi subgroup M' of G and all parabolic subgroup Q' of G containing M', if $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F) \text{ satisfies } \xi(AB) \notin NE^{\times} \text{ for some } \xi \in X(M_{Q_n})_F, \text{ then}$

$$J_{M'}^{\eta,Q'}(X,f') = 0;$$

2) for all ω -stable Levi subgroup M' of G and all parabolic subgroup Q' of G containing M', if $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \text{ is an elliptic element in } \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F) \text{ and satisfies } \xi(AB) \notin NE^{\times} \text{ for some}$ $\xi \in X(M_{Q_n})_F$, then

$$J_{M'}^{\eta,Q'}(X,f') = 0.$$

PROOF. The direction $1 \ge 2$) is trivial. Now we assume 2) and prove 1).

Let $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$ satisfy $\xi(AB) \notin NE^{\times}$ for some $\xi \in X(M_{Q_n})_F$. There is an ω -stable Levi subgroup M'_* of G contained in M' and an element $y \in M'(F) \cap H'(F)$ such that $X_* := \operatorname{Ad}(y)(X)$ is an elliptic element in $\mathfrak{m}'_*(F) \cap \mathfrak{s}'_{rs}(F)$. We have

(10.5.1)
$$J_{M'}^{\eta,Q'}(X_*,f') = \eta(\det(y))J_{M'}^{\eta,Q'}(X,f').$$

Suppose that $X_* = \begin{pmatrix} 0 & A_* \\ B_* & 0 \end{pmatrix}$. Then $\xi(A_*B_*) \notin NE^{\times}$ for the above $\xi \in X(M_{Q_n})_F$. By the descent formula for (G, M')-families (see [37, Lemma I.1.2]), we have

$$v_{M'}^{Q'} = \sum_{L' \in \mathscr{L}^{Q'}(M'_*)} d_{M'_*}^{Q'}(M',L') v_{M'_*}^{Q'_{L'}}$$

where $\mathscr{L}^{Q'}(M'_*)$ denotes the set of Levi subgroups of G contained in Q' and containing M'_* (thus L'is ω -stable), $Q'_{L'}$ is some parabolic subgroup of G with Levi factor L' (thus $Q'_{L'}$ is ω -stable), and $d_{M'_*}^{Q'}(M',L')\in\mathbb{R}_{\geq 0}$ is defined in [6, p. 356]. Thus

(10.5.2)
$$J_{M'}^{\eta,Q'}(X_*,f') = \sum_{L' \in \mathscr{L}^{Q'}(M'_*)} d_{M'_*}^{Q'}(M',L') J_{M'_*}^{\eta,Q'_{L'}}(X_*,f').$$

For all $L' \in \mathscr{L}^{Q'}(M'_*)$, let $\xi_L \in X(L_n)_F$ be the image of ξ under the restriction $X(M_{Q_n})_F \hookrightarrow X(L_n)_F$. Then $\xi_L(A_*B_*) \notin NE^{\times}$. By our assumption 2), we have

$$J_{M'_*}^{\eta,Q'_{L'}}(X_*,f') = 0.$$

Then by (10.5.1) and (10.5.2), we obtain

$$J_{M'}^{\eta,Q'}(X,f') = \eta(\det(y))^{-1} J_{M'}^{\eta,Q'}(X_*,f') = 0,$$

which shows 1).

The proposition below (cf. [37, Lemma III.3.3]) shows that strongly associated functions are smooth transfers of each other in the sense of [58, Definition 5.10.(ii)].

PROPOSITION 10.6. If $f' \in C_c^{\infty}(\mathfrak{s}'(F))$ satisfies the conditions in Proposition 10.5, then for $X \in \mathfrak{s}'_{rs}(F)$ with no matching orbit in $\mathfrak{s}_{rs}(F)$, we have

$$J_{G'}^{\eta,G'}(X,f') = 0.$$

To prove this proposition, we recall two basic facts.

LEMMA 10.7. Suppose that $\sum_{j=1}^{l} n_j = n$. Let $A = (A_1, ..., A_l) \in GL_{n_1}(F) \times \cdots \times GL_{n_l}(F)$ be a regular semi-simple element in $GL_n(F)$. Then $A \in N(GL_n(E))$ if and only if $A_j \in N(GL_{n_j}(E))$ for all $1 \leq j \leq l$.

PROOF. This is known, but we include its proof here for completeness (cf. [36, Lemma 8.8]). For $A \in N(GL_n(E))$, there exists $B \in GL_n(E)$ such that $A = B\overline{B}$. Since $A \in GL_n(F)$, we have $B\overline{B} = \overline{B}B$, which implies that AB = BA. But A is regular semi-simple in $GL_n(E)$. Thus $B \in GL_{n_1}(E) \times \cdots \times GL_{n_l}(E)$. We write $B = (B_1, ..., B_l)$ with $B_j \in GL_{n_j}(E)$ for all $1 \leq j \leq l$. Then we obtain $A_j = B_j \overline{B_j} \in N(GL_{n_j}(E))$ for all $1 \leq j \leq l$. This shows one direction. The other direction is trivial.

LEMMA 10.8. Let $A \in GL_n(F)$ be an elliptic regular element. Then $A \in N(GL_n(E))$ if and only if $det(A) \in NE^{\times}$.

PROOF. This is a special case of [10, Lemma 1.4 in Chapter 1].

PROOF OF PROPOSITION 10.6. Up to conjugation by H(F), it suffices to consider $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix}$ with A an elliptic regular element in $M_n(F)$ for some semi-standard Levi subgroup M_n of GL_n . Then by Lemmas 10.7 and 10.8, X has no matching orbit in $\mathfrak{s}_{rs}(F)$ if and only if $\xi(A) \notin NE^{\times}$ for some $\xi \in X(M_n)_F$. Let Q' be a parabolic subgroup of G with Levi factor M'. We have

$$J_{G'}^{\eta,G'}(X,f') = J_{M'}^{\eta,Q'}(X,f').$$

But $J_{M'}^{\eta,Q'}(X,f')$ vanishes for $f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$ satisfying the conditions in Proposition 10.5. Then we finish the proof.

10.6. The weighted fundamental lemma. Let $f_0 \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ (resp. $f'_0 \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$) be the characteristic function of $\mathfrak{s}(\mathcal{O}_F) \simeq \mathfrak{gl}_n(\mathcal{O}_E)$ (resp. of $\mathfrak{s}'(\mathcal{O}_F) \simeq (\mathfrak{gl}_n \oplus \mathfrak{gl}_n)(\mathcal{O}_F)$).

THEOREM 10.9. The functions f_0 and f'_0 are strongly associated.

The rest of the section is devoted to the proof of this theorem with the help of (split and unramified) base changes for GL_n . Suppose that M' is an ω -stable Levi subgroup of G and that Q' is a parabolic subgroup of G containing M' (thus Q' is ω -stable). For $x = (x_{i,j}) \in \mathfrak{gl}_n(E)$, let $|x| := \max_{i,j} |x_{i,j}\overline{x_{i,j}}|_F^{1/2}$.

10.6.1. Split base change. Let $A \in GL_n(F)$ be regular semi-simple and denote $v := |\det(A)|_F$. We shall define $\Phi_v \in \mathcal{C}_c^{\infty}(GL_n(F) \times GL_n(F))$ and $\Psi_v \in \mathcal{C}_c^{\infty}(GL_n(F))$ as in the proof of [58, Lemma 5.18]. Let Φ_v be the characteristic function of the subset of $(x_1, x_2) \in GL_n(F) \times GL_n(F)$ satisfying $|x_1| \leq 1$, $|x_2| \leq 1$ and $|\det(x_1x_2)|_F = v$. Let Ψ_v be the function on $GL_n(F)$ defined by

$$\Psi_v(g) := \int_{GL_n(F)} \Phi_v(x_2, x_2^{-1}g) \eta(\det(x_2)) dx_2.$$

We also define $\Theta_v(x_1, x_2) := \Phi_v(x_1, x_2)\eta(\det(x_1)) \in \mathcal{C}_c^{\infty}(GL_n(F) \times GL_n(F)).$

We shall consider the action σ' on $GL_n \times GL_n$ which exchanges two copies. Denote by $(GL_n \times GL_n)_{(1_n,A),\sigma'}$ the twisted (by σ') centralizer of $GL_n \times GL_n$ at $(1_n, A)$ and by $GL_n(F)_A$ the centralizer of $GL_n(F)$ at A. Recall the (split) base change homomorphism (see [10, §5 of Chapter 1] for example)

$$bc_{F \times F/F} : \mathcal{H}(GL_n(F) \times GL_n(F), GL_n(\mathcal{O}_F) \times GL_n(\mathcal{O}_F)) \to \mathcal{H}(GL_n(F), GL_n(\mathcal{O}_F))$$

defined by the convolution product, where $\mathcal{H}(GL_n(F) \times GL_n(F), GL_n(\mathcal{O}_F) \times GL_n(\mathcal{O}_F))$ and $\mathcal{H}(GL_n(F), GL_n(\mathcal{O}_F))$ denote the corresponding spherical Hecke algebras. Notice that $\Phi_v, \Theta_v \in \mathcal{H}(GL_n(F) \times GL_n(F), GL_n(\mathcal{O}_F) \times GL_n(\mathcal{O}_F))$ and that $\Psi_v \in \mathcal{H}(GL_n(F), GL_n(\mathcal{O}_F))$.

LEMMA 10.10. We have

=

$$\Psi_v = bc_{F \times F/F}(\Theta_v).$$

PROOF. Let $\psi_n := bc_{F \times F/F}(\Theta_v)$. Via the Satake isomorphism, it suffices to prove that ψ_n and Ψ_v have the same orbital integrals at any regular element in the diagonal torus $A_n(F)$ of $GL_n(F)$. Let $a \in A_n(F)$ be a regular element in $GL_n(F)$. From [10, §5 in Chapter 1], we know that the orbital integral of ψ_n at a is equal to the twisted (by σ') orbital integral of Θ_v at $(1_n, a)$. By change of variables (cf. the proof of [58, Lemma 5.18]), the latter is

$$\int_{(GL_n \times GL_n)_{(1_n,a),\sigma'}(F) \setminus GL_n(F) \times GL_n(F)} \Phi_v(x_1^{-1}x_2, x_2^{-1}ax_1)\eta(\det(x_1^{-1}x_2))dx_1dx_2$$

=
$$\int_{(GL_n(F)_a \setminus GL_n(F)) \times GL_n(F)} \Phi_v(x_2, x_2^{-1}x_1^{-1}ax_1)\eta(\det(x_2))dx_2dx_1$$

=
$$\int_{GL_n(F)_a \setminus GL_n(F)} \Psi_v(\operatorname{Ad}(x_1^{-1})(a))dx_1,$$

which is the orbital integral of Ψ_v at a. This completes the proof.

Suppose additionally that A belongs to the Levi subgroup $M_n(F)$. The twisted (by σ') weighted orbital integral of $\Theta_v \in \mathcal{C}^{\infty}_c(GL_n(F) \times GL_n(F))$ at $(1_n, A)$ is defined by

$$J_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}((1_n, A), \Theta_v) := \int_{(GL_n \times GL_n)_{(1,A),\sigma'}(F) \setminus GL_n(F) \times GL_n(F)} \Theta_v(x^{-1}(1_n, A)\sigma'(x)) v_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}(x) dx.$$

The weighted orbital integral of $\Psi_v \in \mathcal{C}^{\infty}_c(GL_n(F))$ at A is defined by

$$J_{M_n}^{Q_n}(A,\Psi_v) := \int_{GL_n(F)_A \backslash GL_n(F)} \Psi_v(\operatorname{Ad}(x^{-1})(A)) v_{M_n}^{Q_n}(x) dx.$$

COROLLARY 10.11. For $A \in M_n(F)$ which is regular semi-simple in $GL_n(F)$, we have

$$J_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}((1_n, A), \Theta_v) = J_{M_n}^{Q_n}(A, \Psi_v).$$

PROOF. It results from Lemma 10.10 and [37, Theorem IV.5.2] for the (split) base change $F \times F/F$.

Let $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix} \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$. Then $\kappa(X) = 1$. By Remark 10.3.1), since $\eta(\det(x_1x_2)) = \eta(\det(x_1^{-1}x_2))$ for $(x_1, x_2) \in GL_n(F) \times GL_n(F)$, we see that

(10.6.1)
$$\kappa(X)J_{M'}^{\eta,Q'}(X,f_0') = J_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}((1_n,A),\Theta_v).$$

10.6.2. Unramified base change. Let $B \in GL_n(E)$ be such that $B\overline{B}$ is regular semi-simple and denote $w := |\det(B\overline{B})|_F$. We shall define $\Xi_w \in \mathcal{C}^{\infty}_c(GL_n(E))$ as in the proof of [58, Lemma 5.18]. Let Ξ_w be the characteristic function of the subset of $x \in GL_n(E)$ satisfying $|x| \leq 1$ and $|\det(x\overline{x})|_F = w$.

We shall consider the nontrivial Galois conjugation σ on $\operatorname{Res}_{E/F}GL_{n,E}$. Denote by $GL_n(E)_{B,\sigma}$ the twisted (by σ) centralizer of $GL_n(E)$ at B. Recall the (unramified) base change homomorphism (see [10, §4.2 of Chapter 1] for example)

$$bc_{E/F} : \mathcal{H}(GL_n(E), GL_n(\mathcal{O}_E)) \to \mathcal{H}(GL_n(F), GL_n(\mathcal{O}_F))$$

described via the Satake transform by $f(z) \mapsto f(z^2)$, where $\mathcal{H}(GL_n(E), GL_n(\mathcal{O}_E))$ denotes the corresponding spherical Hecke algebra. Note that $\Xi_w \in \mathcal{H}(GL_n(E), GL_n(\mathcal{O}_E))$.

LEMMA 10.12. We have

$$\Psi_w = bc_{E/F}(\Xi_w).$$

PROOF. This is essentially included in [23, Corollary 3.7]. Via the Satake isomorphism, it suffices to prove that $bc_{E/F}(\Xi_w)$ and Ψ_v have the same orbital integrals at any regular element in the diagonal torus $A_n(F)$ of $GL_n(F)$. From [10, Theorem 4.5 in Chapter 1], we reduce ourselves to comparing the twisted (by σ) orbital integral of Ξ_w at $\beta \in A_n(E)$ such that $\beta\overline{\beta}$ belongs to $A_n(F)$ and is regular with the orbital integral of Ψ_v at regular elements in $A_n(F)$. The former is computed in [23, the first case in p. 139], while the latter is computed in [23, the first case in p. 137].

Suppose additionally that B belongs to the Levi subgroup $M_n(E)$. The twisted (by σ) weighted orbital integral of $\Xi_w \in \mathcal{C}^{\infty}_c(GL_n(E))$ at B is defined by

$$J_{(\operatorname{Res}_{E/F}Q_{n,E})\rtimes\sigma}^{(\operatorname{Res}_{E/F}Q_{n,E})\rtimes\sigma}(B,\Xi_w) := \int_{GL_n(E)_{B,\sigma}\backslash GL_n(E)} \Xi_w(x^{-1}B\sigma(x)) v_{(\operatorname{Res}_{E/F}M_{n,E})\rtimes\sigma}^{(\operatorname{Res}_{E/F}Q_{n,E})\rtimes\sigma}(x) dx.$$

COROLLARY 10.13. For $B \in M_n(E)$ such that $A = B\overline{B}$ belongs to $M_n(F)$ and is regular semi-simple in $GL_n(F)$, we have

$$J_{(\operatorname{Res}_{E/F}M_{n,E})\rtimes\sigma}^{(\operatorname{Res}_{E/F}Q_{n,E})\rtimes\sigma}(B,\Xi_w) = J_{M_n}^{Q_n}(A,\Psi_v).$$

PROOF. It results from Lemma 10.12 and [37, Theorem IV.5.2] for the (unramified) base change E/F.

Let $Y = B \in \mathfrak{m}(F) \cap \mathfrak{s}_{rs}(F)$. By Remark 10.3.2), we have

(10.6.2)
$$J_M^Q(Y, f_0) = J_{(\operatorname{Res}_{E/F} M_{n,E}) \rtimes \sigma}^{(\operatorname{Res}_{E/F} Q_{n,E}) \rtimes \sigma}(B, \Xi_w).$$

10.6.3. A reduction formula. We fix Haar measures on $M_{Q'}(F) \cap H'(F)$ and $N_{Q'}(F) \cap H'(F)$ such that $\operatorname{vol}(M_{Q'}(F) \cap H'(\mathcal{O}_F)) = \operatorname{vol}(N_{Q'}(F) \cap H'(\mathcal{O}_F)) = 1$. Then for $f^{H'} \in \mathcal{C}^{\infty}_{c}(H'(F))$, we have (see [12, §4.1])

$$\int_{H'(F)} f^{H'}(x) dx = \int_{M_{Q'}(F) \cap H'(F)} \int_{N_{Q'}(F) \cap H'(F)} \int_{H'(\mathcal{O}_F)} f^{H'}(mnk) dk dn dm.$$

We choose the Haar measure on $\mathfrak{n}_{Q'}(F) \cap \mathfrak{h}'(F)$ compatible with that on $N_{Q'}(F) \cap H'(F)$ under the exponential map. We choose the same Haar measure on four copies of $\mathfrak{n}_{Q_n}(F)$ in $\mathfrak{n}_{Q'}(F) = \begin{pmatrix} \mathfrak{n}_{Q_n}(F) & \mathfrak{n}_{Q_n}(F) \\ \mathfrak{n}_{Q_n}(F) & \mathfrak{n}_{Q_n}(F) \end{pmatrix}$. Then $\operatorname{vol}(\mathfrak{n}_{Q'}(F) \cap \mathfrak{s}'(\mathcal{O}_F)) = 1$.

Let $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$. We may define a distribution $J_{M'}^{\eta,M_{Q'}}(X,\cdot)$ on $\mathcal{C}_c^{\infty}(\mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'(F))$ as in Definition 10.2.2). It appears as a product of distributions in the form of $J_{M'}^{\eta,G'}(X,\cdot)$ in lower ranks. As in [58, §3.2], we define the Weyl discriminant factor by

$$|D^{\mathfrak{m}_{Q'}\cap\mathfrak{s}'}(X)|_F := |\det(\mathrm{ad}(X)|_{\mathfrak{m}_{Q'}/\mathfrak{m}_{Q',X}})|_F^{1/2} > 0,$$

where $\mathfrak{m}_{Q',X}$ denotes the centralizer of X in $\mathfrak{m}_{Q'}$. For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ which is invariant under $\mathrm{Ad}(H'(\mathcal{O}_{F}))$, we define its constant term $f'_{Q'} \in \mathcal{C}^{\infty}_{c}(\mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'(F))$ by

$$f'_{Q'}(Z) := \int_{\mathfrak{n}_{Q'}(F) \cap \mathfrak{s}'(F)} f(Z+U) dU, \forall Z \in \mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'(F).$$

Let $f_0^{M_{Q'}} \in \mathcal{C}_c^{\infty}(\mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'(F))$ be the characteristic function of $\mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'(\mathcal{O}_F)$. Then $(f_0')_{Q'} = f_0^{M_{Q'}}$.

PROPOSITION 10.14. Let $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$. For all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ which is invariant under $\operatorname{Ad}(H'(\mathcal{O}_{F}))$, we have

$$J_{M'}^{\eta,Q'}(X,f') = |D^{\mathfrak{s}'}(X)|_F^{-1/2} |D^{\mathfrak{m}_{Q'}\cap\mathfrak{s}'}(X)|_F^{1/2} J_{M'}^{\eta,M_{Q'}}(X,f_{Q'}')$$

PROOF. We apply the change of variables x = mnk to $x \in H'(F)$, where $m \in M_{Q'}(F) \cap H'(F)$, $n \in N_{Q'}(F) \cap H'(F)$ and $k \in H'(\mathcal{O}_F)$. Notice that $v_{M'}^{Q'}(x) = v_{M'}^{M_{Q'}}(m)$. Since E/F is an unramified extension, the restriction of $\eta(\det(\cdot))$ on $H'(\mathcal{O}_F)$ is trivial. Recall that $\operatorname{vol}(H'(\mathcal{O}_F)) = 1$ and that $H'_X \subseteq M' \cap H'$ for $X \in \mathfrak{m}'(F) \cap \mathfrak{s}'_{rs}(F)$. We deduce that

$$J_{M'}^{\eta,Q'}(X,f') = \int_{H'_X(F) \setminus M_{Q'}(F) \cap H'(F)} \int_{N_{Q'}(F) \cap H'(F)} f'(\mathrm{Ad}(mn)^{-1}(X)) \eta(\det(m)) v_{M'}^{M_{Q'}}(m) dn dm.$$

By Lemma 8.1 in Chapter 2, for $Z := \operatorname{Ad}(m^{-1})(X) \in \mathfrak{m}_{Q'}(F) \cap \mathfrak{s}'_{rs}(F)$, the map

$$N_{Q'}(F) \cap H'(F) \to \mathfrak{n}_{Q'}(F) \cap \mathfrak{s}'(F), n \mapsto \operatorname{Ad}(n^{-1})(Z) - Z$$

is an isomorphism of F-analytic varieties. From the proof of [58, Proposition 6.3.(ii)], its Jacobian is

$$c(X) := |D^{\mathfrak{s}'}(X)|_F^{1/2} |D^{\mathfrak{m}_{Q'} \cap \mathfrak{s}'}(X)|_F^{-1/2} > 0.$$

Then

$$\begin{split} J_{M'}^{\eta,Q'}(X,f') =& c(X)^{-1} \int_{H'_X(F) \backslash M_{Q'}(F) \cap H'(F)} \int_{\mathfrak{n}_{Q'}(F) \cap \mathfrak{s}'(F)} f'(\operatorname{Ad}(m^{-1})(X) + U) \eta(\det(m)) v_{M'}^{M_{Q'}}(m) dU dm \\ =& c(X)^{-1} \int_{H'_X(F) \backslash M_{Q'}(F) \cap H'(F)} f'_Q(\operatorname{Ad}(m^{-1})(X)) \eta(\det(m)) v_{M'}^{M_{Q'}}(m) dm \\ =& c(X)^{-1} J_{M'}^{\eta,M_{Q'}}(X,f'_{Q'}). \end{split}$$

10.6.4. End of the proof.

LEMMA 10.15. For $v \notin |NE^{\times}|_F$, we have

 $\Psi_v = 0.$

PROOF. This is essentially included the proof of [23, Proposition 3.7 and Corollary 3.7]. In fact, our assertion is equivalent to [23, the first line in p. 138] since E/F is unramified. But we shall also give a direct proof as follows.

Let $g \in GL_n(F)$. By the change of variables $x_2 = gx^{-1}$, we obtain

$$\Psi_{v}(g) = \int_{GL_{n}(F)} \Phi_{v}(x_{2}, x_{2}^{-1}g)\eta(\det(x_{2}))dx_{2} = \eta(\det(g)) \int_{GL_{n}(F)} \Phi_{v}(gx^{-1}, x)\eta(\det(x))dx_{2}$$

For all $x_1, x_2 \in GL_n(F)$, we notice that

$$\Phi_v(x_1, x_2) = \Phi_v(x_2^t, x_1^t),$$

where the transpose of $x \in GL_n(F)$ is denoted by x^t . Therefore, we have

$$\int_{GL_n(F)} \Phi_v(gx^{-1}, x) \eta(\det(x)) dx = \int_{GL_n(F)} \Phi_v(x^t, (x^t)^{-1}g^t) \eta(\det(x)) dx.$$

By the change of variables $x^t \mapsto x$, we see that the last integral is equal to $\Psi_v(g^t)$. Thus

$$\Psi_v(g) = \eta(\det(g))\Psi_v(g^t).$$

Because $\Psi_v \in \mathcal{H}(GL_n(F), GL_n(\mathcal{O}_F))$, by Cartan decomposition, we have

$$\Psi_v(g^t) = \Psi_v(g).$$

Then

(10.6.3)
$$\Psi_v(g) = \eta(\det(g))\Psi_v(g).$$

Suppose that $v \notin |NE^{\times}|_F$. We see from the definition that $\Psi_v(g) = 0$ unless $|\det(g)|_F = v$, in which case we have $\det(g) \notin NE^{\times}$ since E/F is unramified. Thus $\eta(\det(g)) = -1$ in this case, which implies that $\Psi_v(g) = 0$ by (10.6.3).



PROOF OF THEOREM 8.1. For (1) in Definition 10.4, it suffices to consider $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix}$ and Y = B, where $B \in M_n(E)$ is such that $A = B\overline{B}$ belongs to $M_n(F)$ and is regular semi-simple in $GL_n(F)$. By Corollaries 10.11 and 10.13, we obtain

(10.6.4)
$$J_{(M_n \times M_n) \rtimes \sigma'}^{(Q_n \times Q_n) \rtimes \sigma'}((1_n, A), \Theta_v) = J_{(\operatorname{Res}_{E/F} M_{n,E}) \rtimes \sigma}^{(\operatorname{Res}_{E/F} Q_{n,E}) \rtimes \sigma}(B, \Xi_w).$$

Combining the formulas (10.6.1), (10.6.2) and (10.6.4), we obtain

$$\kappa(X)J_{M'}^{\eta,Q}(X,f_0') = J_M^Q(Y,f_0).$$

For (2) in Definition 10.4, it suffices to consider $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix}$ with $A \in M_n(F)$ being regular semisimple in $GL_n(F)$ such that $\xi(A) \notin NE^{\times}$ for some $\xi \in X(M_{Q_n})_F$. We still have Corollary 10.11. For the case Q' = G, we conclude by Lemma 10.15. We now consider a general Q'. Applying the reduction formula (Proposition 10.14) to f'_0 , we may write

(10.6.5)
$$J_{M'}^{\eta,Q'}(X,f_0') = |D^{\mathfrak{s}'}(X)|_F^{-1/2} |D^{\mathfrak{m}_{Q'}\cap\mathfrak{s}'}(X)|_F^{1/2} J_{M'}^{\eta,M_{Q'}}(X,f_0^{M_{Q'}}).$$

Suppose that

$$M_{Q'} \simeq GL_{2n_1} \times \cdots \times GL_{2n_j}$$

and that

$$M' \simeq M'_1 \times \cdots \times M'_l$$

where $\sum_{i=1}^{l} n_i = n$ and M'_i is an ω -stable Levi subgroup of GL_{2n_i} for $1 \le i \le l$. We have

$$f_0^{M_{Q'}} = f'_{0,1} \otimes \cdots \otimes f'_{0,l}$$

and

$$X = (X_1, \cdot \cdot \cdot, X_l),$$

where $f'_{0,i}$ (resp. X_i) is an analogue of f'_0 (resp. X) when n is replaced by n_i for $1 \le i \le l$. Then

$$J_{M'}^{\eta, M_{Q'}}(X, f_0^{M_{Q'}}) = \prod_{i=1}^l J_{M'_i}^{\eta, GL_{2n_i}}(X_i, f'_{0,i}).$$

Our condition on A and the special case Q' = G above tell us that at least one factor $J_{M'_i}^{\eta,GL_{2n_i}}(X_i, f'_{0,i})$ in the above product vanishes. Thus $J_{M'}^{\eta,Q'}(X, f'_0) = 0$ by (10.6.5).

CHAPTER 4

A local trace formula for *p*-adic infinitesimal symmetric spaces: the case of Guo-Jacquet

We establish an invariant local trace formula for the tangent space of some symmetric spaces over a non-archimedean local field of characteristic zero. These symmetric spaces are studied in Guo-Jacquet trace formulae and our methods are inspired by works of Waldspurger and Arthur. Some other results are given during the proof including a noninvariant local trace formula, Howe's finiteness for weighted orbital integrals and the representability of the Fourier transform of weighted orbital integrals. These local results are prepared for the comparison of regular semi-simple terms, which are weighted orbital integrals, of an infinitesimal variant of Guo-Jacquet trace formulae.

1. Introduction

The Guo-Jacquet trace formula [23] is a promising tool to generalise Waldspurger's result [50] on the relation between toric periods and central values of automorphic L-functions for GL_2 to higher ranks. It is inspired by Jacquet's new proof [29] of Waldspurger's theorem. Although such a formula has not been established in full generality, its simple form was used by Feigon-Martin-Whitehouse [21] to obtain some evidence for the conjecture of Guo-Jacquet. For applications, one needs to compare geometric sides of Guo-Jacquet trace formulae for different symmetric pairs. Some local results on the comparison of relative orbital integrals include Guo's fundamental lemma [23] and Zhang's smooth transfer [58].

In order to study the Guo-Jacquet trace formula and its comparison, one may begin with an infinitesimal variant. That is to say, we replace a symmetric space by its tangent space (called an infinitesimal symmetric space). Such a variant should share some similarities with the geometric side of Guo-Jacquet trace formula. It is simpler than the original formula because spectral objects are replaced by the Fourier transform of geometric objects (cf. [51] and [13]). Moreover, by the method of descent dating back to Harish-Chandra's works, the comparison at the infinitesimal level should imply the comparison of geometric sides of original formulae (see [58] on the transfer of orbital integrals).

An infinitesimal variant of Guo-Jacquet trace formulae has been established in Chapter 2 and Chapter 3 via an analogue of Arthur's truncation process in [3] (see also [13] for its Lie algebra variant). We actually consider more general cases suggested by [57] and [44]. Most (namely regular semi-simple) terms appearing in these formulae can be written as explicit weighted orbital integrals on infinitesimal symmetric spaces over a number field (see Theorem 9.2 in Chapter 2 and Theorem 9.2 in Chapter 3). They are noninvariant analogues of ordinary orbital integrals (which can be compared locally thanks to [23] and [58]) and should be the next objects to be compared. As the first evidence, the weighted fundamental lemma has been proved in Theorem 10.9 in Chapter 3 thanks to Labesse's work [37] on the base change for GL_n .

The same philosophy of Waldspurger's work [52] on the endoscopic transfer has been followed by Zhang [58] to prove the transfer of local orbital integrals on infinitesimal symmetric spaces of Guo-Jacquet. A simple form of the local trace formula [58, Lemma 6.5], Howe's finiteness for orbital integrals [45, Theorem 6.1] and representability of the Fourier transform of orbital integrals [58, Theorem 6.1] apart from the fundamental lemma [58, Lemma 5.18] at the infinitesimal level have been used in Zhang's proof. It is expected that such a strategy should be extended to the weighted context. In fact, some successful attempts have been made in [14] and [15] on the stable base change. We would like to follow these ideas in the comparison of local weighted orbital integrals on infinitesimal symmetric spaces of Guo-Jacquet. However, further study in noninvariant local harmonic analysis on infinitesimal symmetric spaces is needed to achieve our goal. This paper aims to prepare some essential ingredients such as a noninvariant local trace formula, Howe's finiteness for weighted orbital integrals, representability of the Fourier transform of weighted orbital integrals and an invariant local trace formula. Our methods are mainly inspired by the works of Waldspurger's [51] and Arthur's [8]. Let E/F be a quadratic field extension of non-archimedean local fields of characteristic zero. Denote by D a central division algebra over F and by $GL_{n,D}$ the reductive group over F whose F-points are $GL_n(D)$. We study two generalised cases of Guo-Jacquet trace formulae. The first case is (G, H), where $G := GL_{2n,D}$ and $H := GL_{n,D} \times GL_{n,D}$ denotes its diagonal subgroup by diagonal embedding. Denote by \mathfrak{s} the tangent space of the symmetric space G/H at the neutral element, on which H acts by conjugation. The second case is (G', H'), where G' is the group of invertible elements in a central simple algebra \mathfrak{g}' over F containing E, and H' is the centraliser of E^{\times} in G'. Denote by \mathfrak{s}' the corresponding infinitesimal symmetric space. Notice that (G, H) and (G', H') are the same symmetric pair after a base change to an algebraic closure of F containing E. In the rest of the introduction and this paper, we shall focus on results in the first case and provide complete proofs. The second case is similar in statements and proofs, so we shall only state main results, point out additional ingredients and sketch necessary steps for later use.

To explain the main theorems of this paper, we first introduce some notations. Denote by η the quadratic character of F^{\times}/NE^{\times} attached to E/F, where NE^{\times} denotes the norm of E^{\times} . Let M be an ω -stable Levi subgroup of G (see Section 3.2). Let Q be a parabolic subgroup of G containing M. Suppose that $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, where \mathfrak{m} is the Lie algebra of M and \mathfrak{s}_{rs} denotes the subset of regular semi-simple elements in \mathfrak{s} (see Section 3.1). This paper is organised in the following way.

In Section 2, we fix some notations of local harmonic analysis and recall some facts of Arthur's (G, M)-families, most of which can be found in [51, §I-II].

In Section 3, we prepare some properties of infinitesimal symmetric spaces. Some of them are stated for a general symmetric pair and most of them are relative avatars of classical works of Harish-Chandra [26]. Preliminaries on symmetric pairs can be found in [45] and [1].

In Section 4, we define local weighted orbital integrals $J_M^Q(\eta, X, \cdot)$ by (4.1.1) for the action of H on \mathfrak{s} and study their properties. They are distributions on $\mathfrak{s}(F)$ and local analogues of the global weighted orbital integrals obtained in Theorem 9.2 in Chapter 2.

In Section 5, we establish the noninvariant local trace formula which results from the Plancherel formula and an analogue of Arthur's truncation process in [8]. Let $C_c^{\infty}(\mathfrak{s}(F))$ be the space of locally constant, compactly supported, complex-valued functions on $\mathfrak{s}(F)$. For $f \in C_c^{\infty}(\mathfrak{s}(F))$, we define its Fourier transform \hat{f} by (3.2.1). For $f, f' \in C_c^{\infty}(\mathfrak{s}(F))$, we define $J^G(\eta, f, f')$ by (5.1.2).

THEOREM 1.1 (see Theorem 5.3). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$J^G(\eta, f, \hat{f}') = J^G(\eta, \hat{f}, f').$$

We can not deduce it via the exponential map as in $[51, \S V]$ for lack of a local trace formula for symmetric spaces. One needs to return to the proof of [8] instead.

In Section 6, we show Howe's finiteness for weighted orbital integrals on $\mathfrak{s}(F)$ (see Proposition 6.1). The proof originates from Howe's seminal work [27] which is extended to weighted orbital integrals on Lie algebras by [51]. We modify the argument in [51, §IV] to make it apply to our case.

In Section 7, we show that the distribution on $\mathfrak{s}(F)$ defined by $f \mapsto J_M^G(\eta, X, f)$ is represented by a locally integrable function on $\mathfrak{s}(F)$ (see Proposition 7.2). Its proof is similar to that in [51, \S V] and makes use of the noninvariant trace formula and Howe's finiteness for weighted orbital integrals.

In Section 8, we modify weighted orbital integrals to obtain invariant distributions $I_M^G(\eta, X, \cdot)$ on $\mathfrak{s}(F)$ by (8.1.1) and (8.1.7). The method is close to Arthur's standard one, but it is simpler here since there is no spectral object involved, which is also a feature of [51].

In Section 9, we establish the invariant local trace formula which is deduced from the noninvariant one. For $f, f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we define $I^G(\eta, f, f')$ by (9.1.1).

THEOREM 1.2 (see Theorem 9.1). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$I^G(\eta, f, f') = I^G(\eta, f', f).$$

For its proof, we mainly consult [51, §VII].

In Section 10, we prove a vanishing property at "infinity" of the function on $\mathfrak{s}(F)$ representing the Fourier transform of $I_M^G(\eta, X, \cdot)$ with $M \neq G$ (see Proposition 10.1). It is an analogue of [14] and serves as a complement of the limit formula in [58, §7.1].

In the end, we remark that although we concentrate on the case of Guo-Jacquet here, many results in this paper might be extended to other symmetric pairs, which can be seen from their proofs.

2. Notation and preliminaries

2.1. Fields. Let F be a non-archimedean local field of characteristic zero. Denote by $|\cdot|_F$ (resp. $v_F(\cdot)$) the normalised absolute value (resp. the valuation) on F and by \mathcal{O}_F the ring of integers of F. Fix a uniformiser ϖ of \mathcal{O}_F . Let q be the cardinality of the residue field of \mathcal{O}_F .

2.2. Groups and the map H_P . Let G be a reductive group defined over F. All algebraic groups and algebraic varieties are assumed to be defined over F in this article. Fix a Levi subgroup M_0 of a minimal parabolic subgroup of G.

Denote by A_G the maximal F-split torus in the centre of G. Define

$$\mathfrak{a}_G := \operatorname{Hom}_{\mathbb{Z}}(X(G)_F, \mathbb{R}),$$

where $X(G)_F$ is the group of F-rational characters of G. Define the homomorphism $H_G: G(F) \to \mathfrak{a}_G$ by

$$\langle H_G(x), \chi \rangle = \log(|\chi(x)|_F)$$

for all $x \in G(F)$ and $\chi \in X(G)_F$. Set $\mathfrak{a}_{G,F} := H_G(G(F))$, which is a lattice in \mathfrak{a}_G .

Fix a maximal compact subgroup $K = K_G$ of G(F) which is admissible relative to M_0 in the sense of [5, p. 9]. In this paper, when $G(F) = GL_n(D)$ with D being a central division algebra over F, we choose the standard maximal compact subgroup $K = GL_n(\mathcal{O}_D)$ with \mathcal{O}_D being the ring of integers of D (see [54, p. 191] for example). Set $W_0^G := \operatorname{Norm}_{G(F)}(M_0)/M_0(F)$ to be the Weyl group of (G, M_0) , where $\operatorname{Norm}_{G(F)}(M_0)$ denotes the normaliser of M_0 in G(F). It is known that any element in W_0^G admits a representative in K.

By a Levi subgroup of G, we mean a group M containing M_0 which is the Levi component of some parabolic subgroup of G. For such a group M, set $K_M := M(F) \cap K$. Then the triplet (M, K_M, M_0) satisfies the same hypotheses as (G, K, M_0) . Denote by $\mathscr{F}^G(M)$, $\mathscr{P}^G(M)$ and $\mathscr{L}^G(M)$ the set of parabolic subgroups of G containing M, parabolic subgroups of G with Levi factor M and Levi subgroups of G containing M respectively.

For $P \in \mathscr{F}^G(M_0)$, let M_P be the Levi component containing M_0 and N_P the unipotent radical. Denote $A_P := A_{M_P}$ and $\mathfrak{a}_P := \mathfrak{a}_{M_P}$ whose dual \mathbb{R} -linear space is denoted by \mathfrak{a}_P^* . Define a map $H_P : G(F) \to \mathfrak{a}_{M_P}$ by

$$H_P(mnk) = H_{M_P}(m)$$

for all $m \in M_P(F)$, $n \in N_P(F)$ and $k \in K$. Let $\overline{P} \in \mathscr{P}^G(M_P)$ be the parabolic subgroup opposite to P. For $P \subseteq Q$ a pair of parabolic subgroups in $\mathscr{F}^G(M_0)$, the restriction $X(M_Q)_F \hookrightarrow X(M_P)_F$ induces

a pair of dual maps $\mathfrak{a}_P \to \mathfrak{a}_Q$ and $\mathfrak{a}_Q^* \to \mathfrak{a}_P^*$. Let \mathfrak{a}_P^Q be the kernel of the former map $\mathfrak{a}_P \to \mathfrak{a}_Q$. Set Δ_P^Q to be the set of simple roots for the action of A_P on $P \cap M_Q$. Denote by $(\Delta_P^Q)^{\vee}$ the set of "coroots" as in [9, p. 26]. Then $(\Delta_P^Q)^{\vee}$ is a basis of the \mathbb{R} -linear space \mathfrak{a}_P^Q .

2.3. Heights. We fix a height function $\|\cdot\| : G(F) \to \mathbb{R}$ as in [8, §4]. It satisfies the following properties:

(1) $||x|| \ge 1, \forall x \in G(F);$

(2) $||xy|| \le ||x|| ||y||, \forall x, y \in G(F);$

(3) there exists c > 0 and $N \in \mathbb{N}$ such that $||x^{-1}|| \le c ||x||^N, \forall x \in G(F)$.

If $P \in \mathscr{F}^G(M_0)$, for any $x \in G(F)$, we can choose $m_P(x) \in M_P(F)$, $n_P(x) \in N_P(F)$ and $k_P(x) \in K$ such that $x = m_P(x)n_P(x)k_P(x)$. Then

(4) there exists c > 0 and $N \in \mathbb{N}$ such that $||m_P(x)|| + ||n_P(x)|| \le c||x||^N$.

We also fix a Euclidean norm (still denoted by $\|\cdot\|$) on the \mathbb{R} -linear space \mathfrak{a}_{M_0} which is invariant under the action of W_0^G on \mathfrak{a}_{M_0} . Then

(5) there exist $c_1, c_2 > 0$ such that

$$c_1(1 + \log \|y\|) \le 1 + \|H_{M_0}(y)\| \le c_2(1 + \log \|y\|), \forall y \in M_0(F).$$

In addition, we require that $\|\cdot\|$ is a norm on G(F) in the sense of [35, §18.2]. This is possible. For example, for $G = GL_n$, by writing $(g, g^{-1}) = (g_{ij}, h_{ij})_{1 \le i,j \le n}$, one may define $\|g\| := \sup\{|g_{ij}|_F, |h_{ij}|_F\}$

for $g \in G(F)$. Since $\{g_{ij}, h_{ij}\}_{1 \le i,j \le n}$ is a set of generators for the ring of regular functions of G (viewed as an affine variety over F), this defines a norm in the sense of [35, §18.2] on $GL_n(F)$. For general G, one can choose an closed embedding $G \to GL_n$ over F and define the norm on G(F) by the pull-back of the norm on $GL_n(F)$. By [35, Proposition 18.1.(2)], this defines a norm in the sense of [35, §18.2] on G(F). **2.4. Functions and distributions.** Let $\mathfrak{g} := \operatorname{Lie}(G)$. More generally, we shall use a minuscule Fraktur letter to denote the Lie algebra of its corresponding algebraic group. Denote by Ad the adjoint action of G on itself or \mathfrak{g} . The adjoint action of \mathfrak{g} on itself is denoted by ad.

For a locally compact and totally disconnected topological space X (e.g. G(F) or $\mathfrak{g}(F)$), denote by $\mathcal{C}_c^{\infty}(X)$ the space of locally constant, compactly supported, complex-valued functions on X. For $f \in \mathcal{C}_c^{\infty}(X)$, denote by $\operatorname{Supp}(f)$ its support. Denote by $\mathcal{C}_c^{\infty}(X)^*$ the space of distributions on X, i.e., the linear dual of $\mathcal{C}_c^{\infty}(X)$.

Suppose that G(F) acts on such an X. Then G(F) acts on $\mathcal{C}_c^{\infty}(X)$ (or more generally the space of complex functions on X) by

$$g \cdot f(x) := f(g^{-1} \cdot x), \forall g \in G(F), f \in \mathcal{C}^{\infty}_{c}(X), x \in X.$$

Moreover, G(F) acts on $\mathcal{C}^{\infty}_{c}(X)^{*}$ by

$$g \cdot d(f) := d(g^{-1} \cdot f), \forall g \in G(F), d \in \mathcal{C}^{\infty}_{c}(X)^{*}, f \in \mathcal{C}^{\infty}_{c}(X)$$

Let $\eta: G(F) \to \mathbb{C}^{\times}$ be a locally constant character. We say a function $f \in \mathcal{C}^{\infty}_{c}(X)$ (resp. a distribution $d \in \mathcal{C}^{\infty}_{c}(X)^{*}$) is η -invariant if $g \cdot f = \eta(g)f$ (resp. $g \cdot d = \eta(g)d$) for all $g \in G(F)$. For trivial η , we simply say that such a function (resp. distribution) is invariant.

2.5. Haar measures. Fix the Haar measure on K such that vol(K) = 1. Following [51, §I.4], for all $P \in \mathscr{F}^G(M_0)$, we fix a Haar measure on $N_P(F)$ such that

$$\int_{N_P(F)} \exp(2\rho_{\overline{P}}(H_{\overline{P}}(n))) dn = 1,$$

where $\rho_{\overline{P}}$ is the half of the sum of roots (with multiplicity) associated to the parabolic subgroup \overline{P} opposite to P. From [8, p. 12], for all $M \in \mathscr{L}^G(M_0)$, there are compatible Haar measures on G(F) and M(F) such that for all $P \in \mathscr{P}^G(M)$ and $f \in \mathcal{C}^\infty_c(G(F))$, we have

$$\int_{G(F)} f(x)dx = \int_{M(F) \times N_P(F) \times K} f(mnk)dkdndm.$$

We shall fix such measures.

For a F-split torus T, we choose the Haar measure on T(F) such that the maximal compact subgroup of T(F) is of volume 1. For a general torus T, we choose the Haar measure on T such that the induced measure on $T(F)/A_T(F)$ satisfies $vol(T(F)/A_T(F)) = 1$.

Notice that if M_0 is a torus, we have associated to it two measures. However, it will be clear which one should be used according to the context.

Fix open neighbourhoods $V_{\mathfrak{g}}$ of 0 in \mathfrak{g} and V_G of 1 in G such that the exponential map induces a homeomorphism between them. Choose the unique Haar measure on \mathfrak{g} such that the exponential map $V_{\mathfrak{g}} \to V_G$ preserves the measures. Similarly, we obtain Haar measures on Lie algebras of subgroups of G.

From the fixed Euclidean norm $\|\cdot\|$ on \mathfrak{a}_{M_0} , we deduce measures on \mathfrak{a}_{M_0} and its subspaces.

2.6. (G, M)-families. Following [5, p. 15], we define

$$\theta_P^Q(\lambda) := \operatorname{vol}(\mathfrak{a}_P^Q/\mathbb{Z}(\Delta_P^Q)^{\vee})^{-1} \prod_{\alpha^{\vee} \in (\Delta_P^Q)^{\vee}} \lambda(\alpha^{\vee}), \forall \lambda \in i\mathfrak{a}_P^*,$$

where $\mathbb{Z}(\Delta_P^Q)^{\vee}$ denotes the lattice in \mathfrak{a}_P^Q generated by $(\Delta_P^Q)^{\vee}$.

Suppose that $M \in \mathscr{L}^G(M_0)$ and that $Q \in \mathscr{F}^G(M)$. Let $(c_P)_{P \in \mathscr{P}^G(M)}$ be a (G, M)-family in the sense of [5, p. 36]. By [5, Lemma 6.2], we can define

$$c_M^Q := \lim_{\lambda \to 0} \sum_{\{P \in \mathscr{P}^G(M) : P \subseteq Q\}} c_P(\lambda) \theta_P^Q(\lambda)^{-1}.$$

We sometimes write $c_M := c_M^G$ if Q = G.

An important example is following. According to [5, p. 40-41], for $x \in G(F)$,

$$v_P(\lambda, x) := e^{-\lambda(H_P(x))}, \forall \lambda \in i\mathfrak{a}_M^*, P \in \mathscr{P}^G(M),$$

is a (G, M)-family (denoted by $(v_P(x))_{P \in \mathscr{P}^G(M)}$). Then we obtain a function

$$v_M^Q(x) := \lim_{\lambda \to 0} \sum_{\{P \in \mathscr{P}^G(M) : P \subseteq Q\}} v_P(\lambda, x) \theta_P^Q(\lambda)^{-1}, \forall x \in G(F).$$

For a smooth function $c_P(\lambda)$ on $i\mathfrak{a}_P^*$, we can associate to it a smooth function $c'_P(\lambda)$ on $i\mathfrak{a}_P^*$ as in [5, (6.3) in §6]. Denote by c'_P the value of $c'_P(\lambda)$ at $\lambda = 0$. Let $(c_P)_{P \in \mathscr{P}^G(M)}$ and $(d_P)_{P \in \mathscr{P}^G(M)}$ be two (G, M)-families, we define their product $((cd)_P)_{P \in \mathscr{P}^G(M)}$ in the obvious way and have the following product formula (see [5, Lemma 6.3])

(2.6.1)
$$(cd)_M = \sum_{Q \in \mathscr{F}^G(M)} c'_Q d^Q_M.$$

2.7. The maps d_M^G and s. Suppose that $M \in \mathscr{L}^G(M_0)$. As in [6, p. 356], we define a map

$$d_M^G: \mathscr{L}^G(M) \times \mathscr{L}^G(M) \to \mathbb{R}_{\geq 0}$$

such that for all $(L_1, L_2) \in \mathscr{L}^G(M) \times \mathscr{L}^G(M)$, (1) $d_M^G(G, M) = d_M^G(M, G) = 1$; (2) $d_M^G(L_1, L_2) = d_M^G(L_2, L_1)$; (3) $d_M^G(L_1, L_2) \neq 0$ if and only if $\mathfrak{a}_M^G = \mathfrak{a}_M^{L_1} \oplus \mathfrak{a}_M^{L_2}$.

Following $[51, \S II.4]$, we also choose a map (not unique)

$$s: \mathscr{L}^G(M) \times \mathscr{L}^G(M) \to \mathscr{F}^G(M) \times \mathscr{F}^G(M)$$

such that for all $(L_1, L_2) \in \mathscr{L}^G(M) \times \mathscr{L}^G(M)$, (4) $s(L_1, L_2) \in \mathscr{P}^G(L_1) \times \mathscr{P}^G(L_2)$;

- (5) if $s(L_1, L_2) = (Q_1, Q_2)$, then $s(L_2, L_1) = (\overline{Q_2}, \overline{Q_1})$;
- (6) (splitting formula) if $(c_P)_{P \in \mathscr{P}^G(M)}$ and $(d_P)_{P \in \mathscr{P}^G(M)}$ are (G, M)-families, we have the equality

$$(cd)_M = \sum_{L_1, L_2 \in \mathscr{L}^G(M)} d_M^G(L_1, L_2) c_M^{Q_1} c_M^{Q_2},$$

where $(Q_1, Q_2) := s(L_1, L_2);$

(7) (descent formula) if $(c_P)_{P \in \mathscr{P}^G(M)}$ is a (G, M)-family and $L \in \mathscr{L}^G(M)$, we have the equality

$$c_L = \sum_{L' \in \mathscr{L}^G(M)} d_M^G(L, L') c_M^{Q'}$$

where Q' denotes the second component of s(L, L').

3. Symmetric pairs

3.1. General cases. Following [1, Definition 7.1.1], by a symmetric pair, we mean a triple (G, H, θ) where $H \subseteq G$ are a pair of reductive groups, and θ is an involution on G such that H is the subgroup of fixed points of θ .

Suppose that (G, H, θ) is a symmetric pair. Let $\mathfrak{g} := \operatorname{Lie}(G)$ and $\mathfrak{h} := \operatorname{Lie}(H)$. Write $d\theta$ for the differential of θ . Then $\mathfrak{h} = \{X \in \mathfrak{g} : (d\theta)(X) = X\}$. Let \mathfrak{s} be the tangent space at the neutral element of the symmetric space S := G/H. We shall always view \mathfrak{s} as a subspace of \mathfrak{g} . Then $\mathfrak{s} = \{X \in \mathfrak{g} :$ $(d\theta)(X) = -X$ and H acts on \mathfrak{s} by restriction of the adjoint action.

We say an element $X \in \mathfrak{s}$ is semi-simple if the orbit $\operatorname{Ad}(H)(X)$ is Zariski closed in \mathfrak{s} . From [45, Fact A, p. 108-109, we know that $X \in \mathfrak{s}(F)$ is semi-simple if and only if Ad(H(F))(X) is closed in $\mathfrak{s}(F)$ in the analytic topology. By a regular element $X \in \mathfrak{s}$, we mean that the centraliser H_X of X in H has minimal dimension. Denote by \mathfrak{s}_{rs} the principal Zariski open subset (see [45, end of p. 107]) of \mathfrak{s} consisting of regular semi-simple elements in $\mathfrak{s}.$

By a Cartan subspace of \mathfrak{s} , we mean a maximal abelian subspace for the Lie bracket $\mathfrak{c} \subseteq \mathfrak{s}$ defined over F consisting of semi-simple elements. For such \mathfrak{c} , denote by \mathfrak{c}_{reg} the subset of regular elements in \mathfrak{c} . Denote by $T_{\mathfrak{c}}$ the centraliser of \mathfrak{c} in H, which is a torus. Set $\mathfrak{t}_{\mathfrak{c}} := \operatorname{Lie}(T_{\mathfrak{c}})$.

Following [58, p. 1828], for $X \in \mathfrak{c}_{reg}(F)$, where \mathfrak{c} is a Cartan subspace of \mathfrak{s} , we define the Weyl discriminant factor

$$(3.1.1) |D^{\mathfrak{s}}(X)|_F := |\det(\mathrm{ad}(X)|_{\mathfrak{h}/\mathfrak{t}_{\mathfrak{c}}\oplus\mathfrak{s}/\mathfrak{c}})|_F^{1/2}.$$

For a Cartan subspace $\mathfrak{c} \subseteq \mathfrak{s}$, set $W(H,\mathfrak{c}) := \operatorname{Norm}_{H(F)}(\mathfrak{c})/T_{\mathfrak{c}}(F)$ to be its Weyl group, where Norm_{H(F)}(\mathfrak{c}) denotes the normaliser of \mathfrak{c} in H(F). Fix a set $\mathscr{T}(\mathfrak{s})$ of representatives for H(F)-conjugacy

classes of Cartan subspaces in \mathfrak{s} , which is a finite set by [45, p. 105]. Then we have the Weyl integration formula (see [45, p. 106])

(3.1.2)
$$\int_{\mathfrak{s}(F)} f(X)dX = \sum_{\mathfrak{c}\in\mathscr{T}(\mathfrak{s})} |W(H,\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_F \int_{T_{\mathfrak{c}}(F)\setminus H(F)} f(\mathrm{Ad}(x^{-1})(X))dxdX$$

for all $f \in C_c^{\infty}(\mathfrak{s}(F))$. Recall that the adjoint action induces a local isomorphism $\beta : (T_{\mathfrak{c}}(F) \setminus H(F)) \times \mathfrak{c}_{\mathrm{reg}}(F) \to \mathfrak{s}_{\mathrm{rs}}(F)$ of *F*-analytic manifolds, whose image is open in $\mathfrak{s}_{\mathrm{rs}}(F)$. Here we should use compatible Haar measures on $\mathfrak{s}(F)$ and $\mathfrak{c}_{\mathrm{reg}}(F)$, i.e., we require that β should preserve the measures. For particular cases to be considered, we shall fix Haar measures on $\mathfrak{s}(F)$ in the following sections. Notice that we shall not use the Haar measure on $\mathfrak{c}_{\mathrm{reg}}(F)$ obtained via the exponential map.

The lemma below makes the definition of Fourier transform on $\mathfrak{s}(F)$ possible.

LEMMA 3.1. Let (G, H, θ) be a symmetric pair. Then there exists a G-invariant θ -invariant nondegenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$ on \mathfrak{g} . In particular, $\mathfrak{g} = \mathfrak{h} \oplus \mathfrak{s}$ is an orthogonal direct sum with respect to $\langle \cdot, \cdot \rangle$, and the restriction of $\langle \cdot, \cdot \rangle$ to \mathfrak{h} or \mathfrak{s} is non-degenerate.

PROOF. This is [1, Lemma 7.1.9].

The following lemma is a special case of [59, Lemma 3.10], which is an analogue of Harish-Chandra's compactness lemma [26, Lemma 25].

LEMMA 3.2. Let $\sigma_{\mathfrak{s}}$ be a compact subset of $\mathfrak{s}(F)$. Suppose that \mathfrak{c} is a Cartan subspace of \mathfrak{s} . Let $\sigma_{\mathfrak{c}}$ be a compact subset of $\mathfrak{c}_{reg}(F)$. Then

$$\{x \in T_{\mathfrak{c}}(F) \setminus H(F) : \operatorname{Ad}(x^{-1})(\sigma_{\mathfrak{c}}) \cap \sigma_{\mathfrak{s}} \neq \emptyset\}$$

is relatively compact in $T_{\mathfrak{c}}(F) \setminus H(F)$.

PROOF. Choose an arbitrary $X \in \sigma_{\mathfrak{c}}(F)$. We have $H_X = T_{\mathfrak{c}}$. Let $N^{\mathfrak{s}}_{\operatorname{Ad}(H)(X),X}$ be the normal space (see [1, Notation 2.3.3]) to $\operatorname{Ad}(H)(X)$ in \mathfrak{s} at the point X. Let \mathfrak{s}_X be the centraliser of X in \mathfrak{s} . By [1, Proposition 7.2.1], since $X \in \mathfrak{s}$ is semi-simple, one has $N^{\mathfrak{s}}_{\operatorname{Ad}(H)(X),X} \simeq \mathfrak{s}_X$ as H_X -spaces. Note that since $X \in \mathfrak{c}_{\operatorname{reg}}, \mathfrak{c}_{\operatorname{reg}} \subseteq \mathfrak{s}_X$ is an étale Luna slice at X in the sense of [1, Theorem A.2.3]. Thus we can apply [59, Lemma 3.10].

The next lemma is an analogue of [26, Lemma 28].

LEMMA 3.3. Let $\sigma \subseteq \mathfrak{s}(F)$ be a compact subset. Let \mathfrak{c} be a Cartan subspace of \mathfrak{s} . Then $\mathfrak{c}(F) \cap \operatorname{Cl}(\operatorname{Ad}(H(F))(\sigma))$ is relatively compact in $\mathfrak{c}(F)$, where Cl denotes the closure of a subset in $\mathfrak{s}(F)$.

PROOF. This is [58, Lemma 6.12], whose proof relying on the Chevalley restriction theorem for symmetric spaces [49, Theorem 7 in $\S4.4$] applies to an arbitrary symmetric pair.

The following lemma is an analogue of [51, Lemme III.4].

LEMMA 3.4. Let $\sigma \subseteq \mathfrak{s}(F)$ be a compact subset. Let \mathfrak{c} be a Cartan subspace of \mathfrak{s} and $T_{\mathfrak{c}}$ the centraliser of \mathfrak{c} in H. Then there exists $c_{\sigma} > 0$ such that for all $x \in H(F)$ and $X \in \mathfrak{c}_{reg}(F)$ satisfying $Ad(x^{-1})(X) \in \sigma$, we have

$$\inf_{\tau \in T_{\mathfrak{s}}(F)} \log \|\tau x\| \le c_{\sigma} \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}.$$

PROOF. Let $\|\cdot\|_{T_{\mathfrak{c}}\setminus H}$ be any norm on $(T_{\mathfrak{c}}\setminus H)(F)$ in the sense of [35, §18.2]. Applying the argument of [35, Lemma 20.3] to the finite morphism

$$\beta: (T_{\mathfrak{c}} \setminus H) \times \mathfrak{c}_{\mathrm{reg}} \to \mathfrak{s}_{\mathrm{rs}}$$

of affine algebraic varieties defined by $\beta(x, X) := \operatorname{Ad}(x^{-1})(X)$, we show the inequality

$$\log \|x\|_{T_{\mathfrak{c}} \setminus H} \le c_{\sigma} \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}.$$

By [35, Proposition 18.3], the quotient $H \to T_{\mathfrak{c}} \backslash H$ has the norm descent property in the sense of [35, §18.6]. That is to say, the restriction of $\|\cdot\|_{T_{\mathfrak{c}} \backslash H}$ to $T_{\mathfrak{c}}(F) \backslash H(F)$ is equivalent to the abstract norm $\inf_{\tau \in T_{\mathfrak{c}}(F)} \|\tau \cdot \|$ on $T_{\mathfrak{c}}(F) \backslash H(F)$.

The lemma below is an analogue of [26, Lemma 44].

LEMMA 3.5. There exists $\varepsilon > 0$ such that the function $|D^{\mathfrak{s}}(X)|_{F}^{-\varepsilon}$ is locally integrable on $\mathfrak{c}(F)$ for any Cartan subspace \mathfrak{c} of \mathfrak{s} .

PROOF. See [59, Lemma 4.3].

COROLLARY 3.6. For any $r \ge 0$, the function $\sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^r$ is locally integrable on $\mathfrak{c}(F)$ for any Cartan subspace \mathfrak{c} of \mathfrak{s} .

PROOF. We have the elementary fact (cf. the proof of [35, Corollary 20.2]): for $\varepsilon > 0$ and $r \ge 0$, there exist c > 0 such that

$$\sup\{1, \log y\}^r \le cy^\varepsilon + 1, \forall y > 0.$$

Then it suffices to apply Lemma 3.5.

We say an element $X \in \mathfrak{s}(F)$ is nilpotent if $0 \in \operatorname{Cl}(\operatorname{Ad}(H(F))(X))$, where Cl denotes the closure of a subset in $\mathfrak{s}(F)$. From [1, Lemmas 2.3.12 and 7.3.8], we know that $X \in \mathfrak{s}(F)$ is nilpotent if and only if it is a nilpotent element in \mathfrak{g} . Denote by $\mathcal{N}^{\mathfrak{s}}$ the set of nilpotent elements in $\mathfrak{s}(F)$, which is a cone. The following lemma is an analogue of Jacobson-Morozov theorem.

LEMMA 3.7. Let (G, H, θ) be a symmetric pair and $X \in \mathcal{N}^{\mathfrak{s}}$. Then there exists a group homomorphism $\varphi : SL_2(F) \to G(F)$ such that

$$d\varphi \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} = X, d\varphi \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} \in \mathfrak{s}(F) \text{ and } \varphi \begin{pmatrix} t \\ t^{-1} \end{pmatrix} \in H(F), \forall t \in F^{\times}$$

PROOF. This is [1, Lemma 7.1.11].

Let $X \in \mathfrak{s}_{rs}(F)$. The orbital integral of X is the distribution I_X on $\mathfrak{s}(F)$ defined by

(3.1.3)
$$\forall f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F)), I_X(f) := |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(X)) dx.$$

The next lemma is an analogue of Harish-Chandra's submersion principle [26, Theorem 11].

LEMMA 3.8. Let $I : \mathfrak{s}_{rs}(F) \to \mathbb{C}$ be a function. The following conditions are equivalent:

(1) I is locally constant, invariant by the adjoint action of H(F) and of support included in $Ad(H(F))(\sigma)$ with $\sigma \subseteq \mathfrak{s}_{rs}(F)$ a compact subset;

(2) there exists $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}_{rs}(F))$ such that

$$\forall X \in \mathfrak{s}_{\rm rs}(F), I(X) = I_X(f)$$

PROOF. For $\mathfrak{c} \in \mathscr{T}(\mathfrak{s})$, apply the argument of [16, Lemme 6.1] to the morphism

$$\Phi_{\mathfrak{c}}: (T_{\mathfrak{c}}(F) \setminus H(F)) \times \mathfrak{c}_{\mathrm{reg}}(F) \to \mathfrak{s}_{\mathrm{rs}}(F)$$

defined by $\Phi_{\mathfrak{c}}(x,X) := \operatorname{Ad}(x^{-1})(X)$. Then glue the results for all $\mathfrak{c} \in \mathscr{T}(\mathfrak{s})$ together.

3.2. The case of (G, H). Let D be a central division algebra over F. Denote by $GL_{n,D}$ the reductive group over F whose F-points are $GL_n(D)$. Let $G := GL_{2n,D}$ and $H := GL_{n,D} \times GL_{n,D}$ denotes its subgroup by diagonal embedding. Then H is the subgroup of fixed points of the involution $\operatorname{Ad}(\epsilon)$ on G, where $\epsilon := \begin{pmatrix} 1_n \\ -1_n \end{pmatrix}$. Here we can embed G into \mathfrak{g} in the standard way. For a linear subspace $\mathfrak{v} \subseteq \mathfrak{g}$, we write $\mathfrak{v}^{\times} := \mathfrak{v} \cap G$. Recall that $\mathfrak{s}_{rs} \subseteq \mathfrak{s}^{\times}$ in our case.

LEMMA 3.9. Let P be a parabolic subgroup of G. Then $P \cap H$ is a parabolic subgroup of H if and only if $\epsilon \in P$. Moreover, if ϵ belongs to a Levi factor M of P, then $M \cap H$ is a Levi factor of $P \cap H$.

PROOF. One may consider all the groups over an algebraic closure of F. We first suppose that $P \cap H$ is a parabolic subgroup of H. Then $\epsilon \in Cent(H) \subseteq P \cap H$, where Cent(H) denotes the centre of H. This establishes one direction.

We now suppose that $\epsilon \in P$. Denote by N the unipotent radical of P and let M be a Levi factor of P. By the argument in the last paragraph of the proof of Lemma 4.1 in Chapter 2, we show that ϵ is N-conjugate to an element in M with the help of [3, Lemma 2.1] (actually we need its variant over a local field for the characteristic function of a singleton here, whose proof is similar). Then replacing M by its N-conjugate if necessary, we may assume that $\epsilon \in M$.

Let G = GL(V) for a vector space $V = \bigoplus_{1 \le i \le r} V_i$. Suppose that

$$P = \{g \in G : g(V_1 \oplus \ldots \oplus V_i) \subseteq V_1 \oplus \ldots \oplus V_i, \forall 1 \le i \le r\}$$

and that

$$M = \{g \in G : g(V_i) \subseteq V_i, \forall 1 \le i \le r\}$$

Since $\epsilon \in M$, we have $\epsilon(V_i) \subseteq V_i$ for all $1 \leq i \leq r$. Let V_i^+ (resp. V_i^-) be the +1(resp. -1)-eigenspace of V_i under the action of ϵ . For $\epsilon^2 = 1$, we have $V_i = V_i^+ \oplus V_i^-$. Let $V^+ := \bigoplus_{1 \leq i \leq r} V_i^+$ and $V^- := \bigoplus_{1 \leq i \leq r} V_i^-$. Then

$$H = \{g \in G : g(V^+) \subseteq V^+, g(V^-) \subseteq V^-\}.$$

Hence,

 $P \cap H = \{g \in G : g(V_1^+ \oplus \ldots \oplus V_i^+) \subseteq V_1^+ \oplus \ldots \oplus V_i^+, g(V_1^- \oplus \ldots \oplus V_i^-) \subseteq V_1^- \oplus \ldots \oplus V_i^-, \forall 1 \le i \le r\}.$

It means exactly that $P \cap H$ is a parabolic subgroup of H and proves the other direction. Moreover,

$$M \cap H = \{g \in G : g(V_i^+) \subseteq V_i^+, g(V_i^-) \subseteq V_i^-, \forall 1 \le i \le r\}.$$

That is to say, $M \cap H$ is a Levi factor of $P \cap H$.

Let M_0 be the group of diagonal matrices in G. Set $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. For $P \in \mathscr{F}^G(M_0)$, we say that P is " ω -stable" if $\omega \in P$. Denote by $\mathscr{F}^{G,\omega}(M_0)$ the subset of ω -stable parabolic subgroups in $\mathscr{F}^G(M_0)$. For $M \in \mathscr{L}^G(M_0)$, we say that M is " ω -stable" if $M = M_P$ for some $P \in \mathscr{F}^{G,\omega}(M_0)$. This condition is stronger than $\operatorname{Ad}(\omega)(M) = M$; for example, M_0 is not considered to be ω -stable in our sense. Denote by $\mathscr{L}^{G,\omega}(M_0)$ the subset of ω -stable Levi subgroups in $\mathscr{L}^G(M_0)$. Let A_n be the group of diagonal matrices in GL_n . Recall that there is a bijection between $\mathscr{L}^{GL_n}(A_n)$ and $\mathscr{L}^{G,\omega}(M_0)$ induced by $M_n \mapsto M = \begin{pmatrix} \mathfrak{m}_{n,D} & \mathfrak{m}_{n,D} \\ \mathfrak{m}_{n,D} & \mathfrak{m}_{n,D} \end{pmatrix}^{\times}$. We shall always use the notation M_n to denote the preimage of Munder this bijection. Notice that if $M \in \mathscr{L}^{G,\omega}(M_0)$ and $Q \in \mathscr{F}^G(M)$, then $Q \in \mathscr{F}^{G,\omega}(M_0)$.

Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$. We say an element $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ is *M*-elliptic if A_M is the maximal *F*-split torus in H_X . Denote by $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$ the set of *M*-elliptic elements in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Write $M_H := M \cap H$. Denote by $\Gamma_{ell}((\mathfrak{m} \cap \mathfrak{s}_{rs})(F))$ the set of $M_H(F)$ -conjugacy classes in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$. We say a Cartan subspace $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ is *M*-elliptic if $A_{T_c} = A_M$. Since $(M_H, \mathfrak{m} \cap \mathfrak{s})$ appears as the product of some copies of the form (H, \mathfrak{s}) in lower dimensions, we define $W(M_H, \mathfrak{c})$ and $\mathscr{T}(\mathfrak{m} \cap \mathfrak{s})$ as in Section 3.1. Denote by $\mathscr{T}_{ell}(\mathfrak{m} \cap \mathfrak{s})$ the subset of *M*-elliptic Cartan subspaces in $\mathscr{T}(\mathfrak{m} \cap \mathfrak{s})$.

LEMMA 3.10. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\{X\} \in \Gamma_{\mathrm{ell}}((\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)).$

1) Let $M' \in \mathscr{L}^{G,\omega}(M_0)$ and $\{X'\} \in \Gamma_{\text{ell}}((\mathfrak{m}' \cap \mathfrak{s}_{rs})(F))$ be such that X' is H(F)-conjugate to X. Then there exists

$$w \in \left\{ \begin{pmatrix} \omega_n & \\ & \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\},$$

where $W_0^{GL_{n,D}}$ denotes the Weyl group of $(GL_{n,D}, A_{n,D})$, such that

$$\operatorname{Ad}(w)(M), \{\operatorname{Ad}(w)(X)\}) = (M', \{X'\}).$$

2) The cardinality of

$$\{(M', \{X'\}) : M' \in \mathscr{L}^{G,\omega}(M_0), \{X'\} \in \Gamma_{\text{ell}}((\mathfrak{m}' \cap \mathfrak{s}_{\text{rs}})(F)), X' \text{ is } H(F) \text{-conjugate to } X\}$$

is

$$|W_0^{GL_n}||W_0^{M_n}|^{-1}$$

where $W_0^{GL_n}$ (resp. $W_0^{M_n}$) denotes the Weyl group of (GL_n, A_n) (resp. (M_n, A_n)).

PROOF. 1) Let $x \in H(F)$ be such that $\operatorname{Ad}(x)(X) = X'$. Then $\operatorname{Ad}(x)(H_X) = H_{X'}$. Since $X \in (\mathfrak{m} \cap \mathfrak{s}_{\operatorname{rs}})(F)_{\operatorname{ell}}$ and $X' \in (\mathfrak{m}' \cap \mathfrak{s}_{\operatorname{rs}})(F)_{\operatorname{ell}}$, we have $\operatorname{Ad}(x)(A_M) = A_{M'}$ and thus $\operatorname{Ad}(x)(M) = M'$. As $x \in H(F)$, we have $\operatorname{Ad}(x)(M_H) = M'_H$. We see that $\operatorname{Ad}(x)(A_{M_0}) \subseteq M'_H$ is a maximal F-split torus, so there exists $m' \in M'_H(F)$ such that $\operatorname{Ad}(m'^{-1}x)(A_{M_0}) = A_{M_0}$. That is to say, $w' := m'^{-1}x \in \operatorname{Norm}_{H(F)}(A_{M_0}) = \operatorname{Norm}_{H(F)}(M_0)$, where $\operatorname{Norm}_{H(F)}(A_{M_0})$ denotes the normaliser of A_{M_0} in H(F). Now $\operatorname{Ad}(x)(A_M) = A_{M'}$ implies that $\operatorname{Ad}(w')(A_M) = A_{M'}$. Because $M, M' \in \mathscr{L}^{G,\omega}(M_0)$, it is shown in §9.1 in Chapter 2 that any isomorphism $A_M \to A_{M'}$ induced by W_0^H can be given by $\left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\}$. Hence, there exists $w \in \left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\}$ such that $w^{-1}w' \in \operatorname{Cent}_{W_0^H}(A_M) = W_0^{M_H}$, where $\operatorname{Cent}_{W_0^H}(A_M)$ denotes the centraliser of A_M in W_0^H . We can check that such a w satisfies the condition in the lemma.

2) By 1), the group $\left\{ \begin{pmatrix} \omega_n \\ & \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\}$ acts transitively on this set. Let $w \in \left\{ \begin{pmatrix} \omega_n \\ & \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\}$

be such that

$$(\mathrm{Ad}(w)(M), \{\mathrm{Ad}(w)(X)\}) = (M, \{X\}).$$

Then $w \in M_H(F)$. Thus the condition on w is equivalent to

$$w \in \left\{ \begin{pmatrix} \omega_n \\ & \omega_n \end{pmatrix} : \omega_n \in W_0^{M_{n,D}} \right\},\$$

where $W_0^{M_{n,D}}$ denotes the Weyl group of $(M_{n,D}, A_{n,D})$. We see that the cardinality of the set in the lemma is $|W_0^{GL_{n,D}}||W_0^{M_{n,D}}|^{-1}$ or $|W_0^{GL_n}||W_0^{M_n}|^{-1}$.

PROPOSITION 3.11. For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$\int_{\mathfrak{s}(F)} f(X)dX = \sum_{\substack{M \in \mathscr{L}^{G,\omega}(M_0) \\ f(\operatorname{Ad}(x^{-1})(X))dxdX.}} |W_0^{GL_n}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\operatorname{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_H,\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\operatorname{reg}}(F)} |D^{\mathfrak{s}}(X)|_F \int_{A_M(F) \setminus H(F)} |W(M_H,\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\operatorname{reg}}(F)} |D^{\mathfrak{s}}(X)|_F \int_{\mathfrak{c}_{\operatorname{reg}}(F)} |W(M_H,\mathfrak{c})|^{-1} |W(M_H,\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\operatorname{reg}}(F)} |W(M_H,\mathfrak{c})|^{-1} |W(M_H,\mathfrak{c})|^{-1} |W(M_$$

PROOF. Recall that any H(F)-conjugacy class in $\mathfrak{s}_{rs}(F)$ is the image of a class $\{X\} \in \Gamma_{ell}((\mathfrak{m} \cap \mathfrak{s}_{rs})(F))$ for some $M \in \mathscr{L}^{G,\omega}(M_0)$ in our case. By Lemma 3.10, the Weyl integration formula (3.1.2) can be written as the above equality (cf. [8, p. 16-17] and [51, (3) in §I.3]).

Recall that there is a bijection between $\mathscr{F}^{GL_n}(A_n)$ and $\mathscr{F}^{G,\omega}(M_0)$ induced by $P_n \mapsto P = \begin{pmatrix} \mathfrak{p}_{n,D} & \mathfrak{p}_{n,D} \\ \mathfrak{p}_{n,D} & \mathfrak{p}_{n,D} \end{pmatrix}^{\times}$. We shall always use the notation P_n to denote the preimage of P under this bijection. Following [58, p. 1846], we shall fix the Haar measures on some subspaces of $\mathfrak{s}(F)$ as follows. Let $P \in \mathscr{F}^{G,\omega}(M_0)$. Then we have $\mathfrak{m}_P = \begin{pmatrix} \mathfrak{m}_{n,D} & \mathfrak{m}_{n,D} \\ \mathfrak{m}_{n,D} & \mathfrak{m}_{n,D} \end{pmatrix}$ and $\mathfrak{n}_P = \begin{pmatrix} \mathfrak{n}_{n,D} & \mathfrak{n}_{n,D} \\ \mathfrak{n}_{n,D} & \mathfrak{n}_{n,D} \end{pmatrix}$, where we denote $M_n := M_{P_n}$ and $N_n := N_{P_n}$. We have fixed the Haar measures on $\mathfrak{m}_n(D)$ and $\mathfrak{n}_n(D)$ in Section 2.5. We shall choose the same Haar measure for any of the four copies in $\mathfrak{m}_P(F)$ or $\mathfrak{n}_P(F)$ under these identifications. In particular, we obtain the Haar measures on $(\mathfrak{m}_P \cap \mathfrak{s})(F)$.

LEMMA 3.12. Let $Q \in \mathscr{F}^{G,\omega}(M_0)$. For $Y \in (\mathfrak{m}_Q \cap \mathfrak{s}_{rs})(F)$, the map

$$N_{Q_H}(F) \to (\mathfrak{n}_Q \cap \mathfrak{s})(F), n \mapsto \mathrm{Ad}(n^{-1})(Y) - Y$$

is an isomorphism of F-analytic manifolds whose Jacobian is $|D^{\mathfrak{s}}(Y)|_{F}^{1/2}|D^{\mathfrak{m}_{Q}\cap\mathfrak{s}}(Y)|_{F}^{-1/2}$.

PROOF. See Lemma 8.1 in Chapter 2 for the isomorphism and the proof of [58, Proposition 6.3.(ii)] for the Jacobian. $\hfill \Box$

Fix a continuous and nontrivial unitary character $\Psi : F \to \mathbb{C}^{\times}$. Let $\langle \cdot, \cdot \rangle$ be the non-degenerate symmetric bilinear form on $\mathfrak{g}(F)$ defined by

$$\langle X, Y \rangle := \operatorname{Trd}(XY), \forall X, Y \in \mathfrak{g}(F),$$

where Trd denotes the reduced trace on $\mathfrak{g}(F)$. It is invariant by the adjoint action of G(F) and $\mathrm{Ad}(\epsilon)$. For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, define its normalised Fourier transform $\hat{f} \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ by

(3.2.1)
$$\forall X \in \mathfrak{s}(F), \hat{f}(X) := c_{\Psi}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} f(Y) \Psi(\langle X, Y \rangle) dY,$$

where $c_{\Psi}(\mathfrak{s}(F))$ is the unique constant such that $\hat{f}(X) = f(-X)$ for all $f \in \mathcal{C}_{c}^{\infty}(\mathfrak{s}(F))$ and all $X \in \mathfrak{s}(F)$. For any $M \in \mathscr{L}^{G,\omega}(M_{0})$, the restriction of $\langle \cdot, \cdot \rangle$ on $\mathfrak{m} \cap \mathfrak{s}$ is non-degenerate. Then we can define similarly the normalised Fourier transform of $f \in \mathcal{C}_{c}^{\infty}((\mathfrak{m} \cap \mathfrak{s})(F))$.

Suppose that $P \in \mathscr{F}^{G,\omega}(M_0)$. Let η be the quadratic character of F^{\times} attached to a quadratic extension E/F. Denote by Nrd the reduced norm on G(F). For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we define a function (parabolic descent) $f_{P}^{\eta} \in \mathcal{C}^{\infty}_{c}((\mathfrak{m}_{P} \cap \mathfrak{s})(F))$ by

(3.2.2)
$$f_P^{\eta}(Z) := \int_{K_H \times (\mathfrak{n}_P \cap \mathfrak{s})(F)} f(\mathrm{Ad}(k^{-1})(Z+U))\eta(\mathrm{Nrd}(k))dUdk$$

for all $Z \in (\mathfrak{m}_P \cap \mathfrak{s})(F)$. We show that $(\hat{f})_P^\eta = (f_P^\eta)$, so we shall denote it by \hat{f}_P^η without confusion. In fact, the integral on $(\mathfrak{n}_P \cap \mathfrak{s})(F)$ and the Fourier transform commute by our choices of Haar measures (see [51, §I.7]); the commutativity of the integral on K_H and the Fourier transform results from the H(F)-invariance of $\langle \cdot, \cdot \rangle$.

The following result is an analogue of [26, Theorem 13].

PROPOSITION 3.13. Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$. Then

$$\sup_{X \in \mathfrak{s}_{\mathrm{rs}}(F)} |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{H_{X}(F) \setminus H(F)} |f(\mathrm{Ad}(x^{-1})(X))| dx < +\infty.$$

PROOF. It is proved in [58, Theorem 6.11] (see also [57, p. 77]) that for any fixed Cartan subspace \mathfrak{c} of $\mathfrak{s},$

$$\sup_{X \in \mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{H_{X}(F) \setminus H(F)} |f(\mathrm{Ad}(x^{-1})(X))| dx < +\infty.$$

Since $\mathscr{T}(\mathfrak{s})$ is a finite set and the orbital integral is constant on any H(F)-orbit, we obtain a uniform bound for all $X \in \mathfrak{s}_{rs}(F)$.

The lemma below is an analogue of [26, Theorem 15].

LEMMA 3.14. There exists $\varepsilon > 0$ such that the function $|D^{\mathfrak{s}}(X)|_{F}^{-\frac{1}{2}-\varepsilon}$ is locally integrable on $\mathfrak{s}(F)$.

PROOF. Choose $\varepsilon > 0$ verifying the condition of Lemma 3.5. Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ with $f \ge 0$. By the Weyl integration formula (3.1.2), we have

$$\int_{\mathfrak{s}(F)} |D^{\mathfrak{s}}(X)|_{F}^{-\frac{1}{2}-\varepsilon} f(X) dX = \sum_{\mathfrak{c}\in\mathscr{T}(\mathfrak{s})} |W(H,\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_{F}^{\frac{1}{2}-\varepsilon} \int_{A_{T_{\mathfrak{c}}}(F)\setminus H(F)} f(\mathrm{Ad}(x^{-1})(X)) dx dX.$$

The convergence of the right hand side results from Proposition 3.13 and Lemmas 3.3 and 3.5.

COROLLARY 3.15. For any $r \ge 0$, the function $|D^{\mathfrak{s}}(X)|_F^{-\frac{1}{2}} \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^r$ is locally integrable on $\mathfrak{s}(F)$.

PROOF. It is the same as the proof of Corollary 3.6.

3.3. The case of
$$(G', H')$$
. Let E be a quadratic extension of F . Let \mathfrak{g}' be a central simple algebra over F with a fixed embedding of F -algebras $E \hookrightarrow \mathfrak{g}'$. Let $\mathfrak{h}' := \operatorname{Cent}_{\mathfrak{g}'}(E)$ be the centraliser of E in \mathfrak{g}' . By the the double centraliser theorem (see [42, Theorem 3.1 in Chapter IV] for example), $\mathfrak{h}'(F)$ is a central simple algebra over E . Denote by $G' := \mathfrak{g}'^{\times}$ (resp. $H' := \mathfrak{h}'^{\times}$) the group of invertible elements in \mathfrak{g}' (resp. $\mathfrak{h}')$, which is considered as an algebraic group over F with Lie algebra \mathfrak{g}' (resp. $\mathfrak{h}')$). Let $\alpha \in E \setminus F$ such that $\alpha^2 \in F$. Then $E = F(\alpha)$ and H' is the subgroup of fixed points of the involution $\operatorname{Ad}(\alpha)$ on G' . Denote by \mathfrak{s}' the corresponding tangent space of G'/H' . For a linear subspace $\mathfrak{v}' \subseteq \mathfrak{g}'$, we write $\mathfrak{v}'^{\times} := \mathfrak{v}' \cap G'$. Then we see that $\mathfrak{s}'_{rs} \subseteq \mathfrak{s}'^{\times}$ via a base change to an algebraic closure of F containing E .

LEMMA 3.16 (cf. Lemma 3.9). Let $\widetilde{P'}$ be a parabolic subgroup of G'. Then $\widetilde{P'} \cap H'$ is a parabolic subgroup of H' if and only if $\alpha \in \widetilde{P'}$. Moreover, if α belongs to a Levi factor $\widetilde{M'}$ of $\widetilde{P'}$, then $\widetilde{M'} \cap H'$ is a Levi factor of $\widetilde{P'} \cap H'$.

By the Wedderburn-Artin theorem, G' is isomorphic to $GL_{n,D}$ for some positive integer n and some central division algebra D over F. Since E embeds into $\mathfrak{g}'(F)$, we see that $n \deg(D)$ is even, where $\deg(D)$ denotes the degree of D. From the Noether-Skolem theorem (see [42, Theorem 2.10 of Chapter IV] for example), up to conjugation by G'(F), the emdedding $H' \hookrightarrow G'$ is reduced to one of the two cases below (see [18, §2.1 and §3.1] and §3.4 in Chapter 3).

Case I: if deg(D) is even, then $(G', H') = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{n,D'})$, where $D' := \operatorname{Cent}_D(E)$ denoting the centraliser of E in D is a central division algebra over E of degree $\frac{\deg(D)}{2}$. Let $M'_0 \simeq (\operatorname{Res}_{E/F}\mathbb{G}_{m,D'})^n$ (resp. $M'_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^n$) be the subgroup of diagonal elements in H' (resp. G'). Recall that there is a bijection $M' \mapsto \widetilde{M'}$ between $\mathscr{L}^{H'}(M'_0)$ and $\mathscr{L}^{G'}(M'_{\widetilde{0}})$. We shall always write $\widetilde{M'}$ for the image of M'under this bijection. Notice that $M' = \widetilde{M'} \cap H'$ and that we can identify $A_{M'}$ with $A_{\widetilde{M'}}$.

Case II: if deg(D) is odd, then $(G', H') = (GL_{n,D}, \operatorname{Res}_{E/F}GL_{\frac{n}{2}, D\otimes_F E})$, where $D \otimes_F E$ is a central division algebra over E of degree deg(D). Let $M'_0 \simeq (\operatorname{Res}_{E/F}\mathbb{G}_{m,D\otimes_F E})^{\frac{n}{2}}$ (resp. $M'_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^n$) be

the subgroup of diagonal elements in H' (resp. G'). Denote by $\mathscr{L}^{G'}(M'_{\widetilde{0}}, M'_{0})$ the subset of elements in $\mathscr{L}^{G'}(M'_{\widetilde{0}})$ containing M'_{0} . Recall that there is a bijection $M' \mapsto \widetilde{M'}$ between $\mathscr{L}^{H'}(M'_{0})$ and $\mathscr{L}^{G'}(M'_{\widetilde{0}}, M'_{0})$. We shall always write $\widetilde{M'}$ for the image of M' under this bijection. Notice that $M' = \widetilde{M'} \cap H'$ and that we can identify $A_{M'}$ with $A_{\widetilde{M'}}$.

Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$. We say an element $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ is M'-elliptic if $A_{M'}$ is the maximal F-split torus in H'_Y . Denote by $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$ the set of M'-elliptic elements in $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Denote by $\Gamma_{ell}((\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F))$ the set of M'(F)-conjugacy classes in $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$. We say a Cartan subspace $\mathfrak{c}' \subseteq \widetilde{\mathfrak{m}'} \cap \mathfrak{s}'$ is M'-elliptic if $A_{T_{\mathfrak{c}'}} = A_{M'}$. Since $(M', \widetilde{\mathfrak{m}'} \cap \mathfrak{s}')$ appears as the product of some copies of the form (H', \mathfrak{s}') in lower dimensions, we define $W(M', \mathfrak{c}')$ and $\mathscr{T}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')$ as in Section 3.1. Denote by $\mathscr{T}_{ell}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')$ the subset of M'-elliptic Cartan subspaces in $\mathscr{T}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')$.

LEMMA 3.17 (cf. Lemma 3.10). Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $\{Y'\} \in \Gamma_{\mathrm{ell}}((\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)).$

1) Let $M \in \mathscr{L}^{H'}(M'_0)$ and $\{Y\} \in \Gamma_{\text{ell}}((\widetilde{\mathfrak{m}} \cap \mathfrak{s}'_{\text{rs}})(F))$ be such that Y is H'(F)-conjugate to Y'. Then there exists $w \in W_0^{H'}$ such that

$$(\mathrm{Ad}(w)(M'), \{\mathrm{Ad}(w)(Y')\}) = (M, \{Y\})$$

2) The cardinality of

$$\{(M, \{Y\}): M \in \mathscr{L}^{H'}(M'_0), \{Y\} \in \Gamma_{\mathrm{ell}}((\widetilde{\mathfrak{m}} \cap \mathfrak{s}'_{\mathrm{rs}})(F)), Y \text{ is } H'(F) \text{-conjugate to } Y'\}$$

is

$$|W_0^{H'}||W_0^{M'}|^{-1}$$

PROPOSITION 3.18. For $f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$, we have the equality

$$\begin{split} \int_{\mathfrak{s}'(F)} f'(Y) dY &= \sum_{\substack{M' \in \mathscr{L}^{H'}(M'_0) \\ f'(\mathrm{Ad}(x^{-1})(Y)) dx dY.}} |W_0^{M'}| |W_0^{H'}|^{-1} \sum_{\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |D^{\mathfrak{s}'}(Y)|_F \int_{A_{M'}(F) \setminus H'(F)} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |D^{\mathfrak{s}'}(Y)|_F \int_{A_{M'}(F) \setminus H'(F)} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M', \mathfrak{c}')|^{-1} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M', \mathfrak{c}')|^{-1} |W$$

PROOF. Recall that any H'(F)-conjugacy class in $\mathfrak{s}'_{rs}(F)$ is the image of a class $\{Y\} \in \Gamma_{ell}((\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F))$ for some $M' \in \mathscr{L}^{H'}(M'_0)$ in our case. By Lemma 3.17, the Weyl integration formula (3.1.2) can be written as the above equality (cf. [8, p. 16-17] and [51, (3) in §I.3]).

Recall there is a bijection $P' \mapsto \widetilde{P'}$ between $\mathscr{F}^{H'}(M'_0)$ and $\mathscr{F}^{G'}(\widetilde{M'_0})$ in both of **Case I** and **Case II**. We shall always write $\widetilde{P'}$ for the image of P' under this bijection. Let $\tau \in D^{\times}$ in **Case I** (resp. $\tau \in GL_2(D)$ in **Case II**) be an element such that $\operatorname{Ad}(\alpha)(\tau) = -\tau$. Let $P' \in \mathscr{F}^{H'}(M'_0)$. Then we have $\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}' = \mathfrak{m}_{P'}\tau = \tau\mathfrak{m}_{P'}$ and $\mathfrak{n}_{\widetilde{P'}} \cap \mathfrak{s}' = \mathfrak{n}_{P'}\tau = \tau\mathfrak{n}_{P'}$ by Proposition 3.12 in Chapter 3. We have fixed the Haar measures on $\mathfrak{m}_{P'}(F)$ and $\mathfrak{n}_{P'}(F)$ in Section 2.5. We shall choose the same Haar measures on $(\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F)$ using above identifications induced by τ . Such Haar measures depend on the choice of τ .

LEMMA 3.19. Let $Q' \in \mathscr{F}^{H'}(M'_0)$. For $X \in (\mathfrak{m}_{\widetilde{O'}} \cap \mathfrak{s}'_{rs})(F)$, the map

$$N_{Q'}(F) \to (\mathfrak{n}_{\widetilde{Q'}} \cap \mathfrak{s}')(F), n \mapsto \operatorname{Ad}(n^{-1})(X) - X$$

is an isomorphism of F-analytic manifolds whose Jacobian is $|D^{\mathfrak{s}'}(X)|_F^{1/2}|D^{\mathfrak{m}_{\widetilde{Q'}}\cap\mathfrak{s}'}(X)|_F^{-1/2}$.

PROOF. See Lemma 8.1 in Chapter 3 for the isomorphism. The computation of its Jacobian is close to the proof of [58, Proposition 6.3.(ii)].

Fix a continuous and nontrivial unitary character $\Psi: F \to \mathbb{C}^{\times}$. Let $\langle \cdot, \cdot \rangle$ be the symmetric bilinear form on $\mathfrak{g}'(F)$ defined by

$$\langle Y, X \rangle := \operatorname{Trd}(YX), \forall Y, X \in \mathfrak{g}'(F),$$

where Trd denotes the reduced trace on $\mathfrak{g}'(F)$. It is non-degenerate, which can be seen after the base change to an algebraic closure of F. It is also invariant by the adjoint action of G'(F) and $\operatorname{Ad}(\alpha)$. For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, define its normalised Fourier transform $\hat{f}' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ by

(3.3.1)
$$\forall Y \in \mathfrak{s}'(F), \hat{f}'(Y) := c_{\Psi}(\mathfrak{s}'(F)) \int_{\mathfrak{s}'(F)} f'(X) \Psi(\langle Y, X \rangle) dX,$$
where $c_{\Psi}(\mathfrak{s}'(F))$ is the unique constant such that $\hat{f}'(Y) = f'(-Y)$ for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ and all $Y \in \mathfrak{s}'(F)$. For any $M' \in \mathscr{L}^{H'}(M'_{0})$, the restriction of $\langle \cdot, \cdot \rangle$ on $\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'$ is non-degenerate. Then we can define similarly the normalised Fourier transform of $f' \in \mathcal{C}^{\infty}_{c}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(F)$).

Suppose that $P' \in \mathscr{F}^{H'}(M'_0)$. For $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$, we define a function (parabolic descent) $f'_{P'} \in \mathcal{C}^{\infty}_c((\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F))$ by

(3.3.2)
$$f'_{P'}(Z) := \int_{K_{H'} \times (\mathfrak{n}_{\widetilde{P'}} \cap \mathfrak{s}')(F)} f'(\mathrm{Ad}(k^{-1})(Z+U)) dU dk$$

for all $Z \in (\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F)$. By our choices of Haar measures (see [51, §I.7]) and the H'(F)-invariance of $\langle \cdot, \cdot \rangle$, we show that $(\hat{f'})_{P'} = (f'_{P'})^{\hat{}}$, which will be denoted by $\hat{f'}_{P'}$ without confusion.

PROPOSITION 3.20. Let $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$. Then

$$\sup_{Y \in \mathfrak{s}'_{\mathrm{rs}}(F)} |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} |f'(\mathrm{Ad}(x^{-1})(Y))| dx < +\infty.$$

COROLLARY 3.21 (cf. Corollary 3.15). For any $r \ge 0$, the function $|D^{\mathfrak{s}'}(Y)|_F^{-\frac{1}{2}} \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^r$ is locally integrable on $\mathfrak{s}'(F)$.

The rest of this section is devoted to the proof of Proposition 3.20. We shall follow the main steps in [58, §6.3], which is similar to the proof of [26, Theorem 13], and only point out some additional ingredients. Let n is the F-rank of G'. Denote $G'_n := G', H'_n := H'$ and $\mathfrak{s}'_n := \mathfrak{s}'$. Recall that the F-rank of H'_n is n in **Case I** (resp. $\frac{n}{2}$ in **Case II**). We shall use induction on n. For n = 1 in **Case I** (resp. n = 2 in **Case II**), the proposition is evident since $H'_Y(F) \setminus H'(F)$ is compact in our case.

The following description of semi-simple elements and descendants (see [1, Definition 7.2.2]) is a generalisation of [24, Lemma 2.1] (see also [58, Proposition 4.7]).

PROPOSITION 3.22. 1) An element Y of $\mathfrak{s}'(F)$ is semi-simple if and only if it is H'(F)-conjugate to an element of the form

$$Y(B) := \begin{pmatrix} B & 0 \\ 0 & 0 \end{pmatrix},$$

with $B \in \mathfrak{s}'_m^{\times}(F)$ being semi-simple with respect to the H'_m -action. More precisely, the set of H'(F)conjugacy classes of semi-simple elements in $\mathfrak{s}'(F)$ is bijective to the set of pairs $(m, \{B\})$ where $0 \leq m \leq n$ is an integer in **Case I** (resp. an even number in **Case II**) and $\{B\}$ is a semi-simple $H'_m(F)$ conjugacy class in $\mathfrak{s}'_m^{\times}(F)$. Moreover, Y(B) is regular semi-simple if and only if m = n and B is regular semi-simple in $\mathfrak{s}'^{\times}(F)$.

2) Let $Y = Y(B) \in \mathfrak{s}'(F)$ be semi-simple. Then the descendant (H'_Y, \mathfrak{s}'_Y) (as a representation) is isomorphic to

$$(H'_{m,B},\mathfrak{s}'_{m,B})\times(H'_{n-m},\mathfrak{s}'_{n-m}),$$

where $H'_{m,B}$ (resp. $\mathfrak{s}'_{m,B}$) denotes the centraliser of B in H'_m (resp. \mathfrak{s}'_m).

PROOF. 1) By the base change to an algebraic closure of F containing E, we see from [31, Proposition 2.1] that an element $Y \in \mathfrak{s}'(F)$ which is H'(F)-conjugate to Y(B) in the proposition is semi-simple. Now we suppose that $Y \in \mathfrak{s}'(F)$ is semi-simple. Since $Y^2 \in \mathfrak{h}'(F)$, up to H'(F) conjugation, we may suppose that $Y^2 = \begin{pmatrix} A & 0 \\ 0 & 0 \end{pmatrix}$ with $A \in \mathfrak{h}'_m{}^{\times}(F)$ being semi-simple in the usual sense. From $\begin{pmatrix} A & 0 \\ 0 & 0 \end{pmatrix} Y = Y\begin{pmatrix} A & 0 \\ 0 & 0 \end{pmatrix}$, we deduce that $Y = \begin{pmatrix} B & 0 \\ 0 & C \end{pmatrix}$ for some $B \in \mathfrak{s}'_m(F)$ such that AB = BA and some $C \in \mathfrak{s}'_{n-m}(F)$. As $Y^2 = \begin{pmatrix} A & 0 \\ 0 & 0 \end{pmatrix}$, we have $B \in \mathfrak{s}'_m{}^{\times}(F)$. Because Y is semi-simple, it is shown in [31, p. 71] that Y and Y^2 have the same rank over an algebraic closure of F containing E. Then C = 0. We can also see from [31, Proposition 2.1] that B is semi-simple with respect to the H'_m -action after the base change. We have established the first statement.

For the second statement, it suffices to notice that two such elements $Y(B_1)$ with $B_1 \in \mathfrak{s}'_{m_1}^{\times}(F)$ and $Y(B_2)$ with $B_2 \in \mathfrak{s}'_{m_2}^{\times}(F)$ in the proposition are H'(F)-conjugate if and only if $m_1 = m_2$ (denoted by m) and B_1 and B_2 are $H'_m(F)$ -conjugate.

The third statement follows from the base change or 2).

2) It can be shown by direct calculation.

Fix a Cartan subspace \mathfrak{c}' of \mathfrak{s}' . The lemma below is an analogue of [26, Lemma 29].

LEMMA 3.23. Let $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F) - \mathcal{N}^{\mathfrak{s}'})$. Then

$$\sup_{Y \in \mathfrak{c}'_{\rm reg}(F)} |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} |f'({\rm Ad}(x^{-1})(Y))| dx < +\infty.$$

PROOF. We may apply the argument of [58, Lemma 6.14] relying on Lemma 3.3 and [59, Proposition 3.11], which is an analogue of Harish-Chandra's semi-simple descent for orbital integrals [35, Lemma 16.1]. By Proposition 3.22.2), it suffices to prove the boundedness of orbital integrals for

$$(H'_{m,B},\mathfrak{s}'_{m,B})\times(H'_{n-m},\mathfrak{s}'_{n-m})$$

with $B \in \mathfrak{s}'_m^{\times}(F)$ being semi-simple with respect to the H'_m -action and $0 < m \leq n$. Since there exists an $H'_{m,B}$ -equivariant linear isomorphism $\mathfrak{s}'_{m,B} \to \mathfrak{h}'_{m,B}$ induced by $Z \mapsto ZB$, the first factor $(H'_{m,B},\mathfrak{s}'_{m,B})$ is covered by Harish-Chandra's work [26, Theorem 13] on classical orbital integrals on Lie algebras. Then we conclude by applying the induction hypothesis to the second factor $(H'_{n-m},\mathfrak{s}'_{n-m})$.

Consider $X_0 \in \mathcal{N}^{\mathfrak{s}'}$. By the Jacobson-Morozov theorem for symmetric spaces (Lemma 3.7), there exists a group homomorphism $\varphi : SL_2(F) \to G'(F)$ such that

$$X_0 = d\varphi \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}, Y_0 := d\varphi \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} \in \mathfrak{s}'(F) \text{ and } d := d\varphi \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} \in \mathfrak{h}'(F).$$

Write $r' := \dim \mathfrak{s}'_{Y_0}$ and $m' := \frac{1}{2} \operatorname{Tr}(\operatorname{ad}(-d)|_{\mathfrak{s}'_{Y_0}})$, where \mathfrak{s}'_{Y_0} denotes the centraliser of Y_0 in \mathfrak{s}' .

LEMMA 3.24. We have 1) $r' \ge \frac{1}{2}\sqrt{\dim \mathfrak{g}'};$ 2) $r' + m' > \frac{1}{4}\dim \mathfrak{g}' + \frac{1}{4}\sqrt{\dim \mathfrak{g}'}.$

PROOF. It suffices to check these relations after a base change to an algebraic closure of F containing E. Then the lemma is exactly [58, Proposition 4.4].

Let \mathfrak{s}'_{bdd} be the set of $X \in \mathfrak{s}'(F)$ such that there exists an open neighbourhood σ' of X in $\mathfrak{s}'(F)$ satisfying

$$\sup_{Y \in \mathfrak{c}'_{\mathrm{reg}}(F)} |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} |f'(\mathrm{Ad}(x^{-1})(Y))| dx < +\infty$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ with $\operatorname{Supp}(f') \subseteq \sigma'$. The next lemma is an analogue of [26, Lemma 38].

LEMMA 3.25. We have $\mathcal{N}^{\mathfrak{s}'} - \{0\} \subseteq \mathfrak{s}'_{bdd}$.

PROOF. We may apply the argument of [58, Lemma 6.16] thanks to Lemma 3.24.

PROOF OF PROPOSITION 3.20. We may use the argument in [26, \S VI.7] to show that $0 \in \mathfrak{s}'_{bdd}$. Then the proposition follows from Lemmas 3.23 and 3.25.

4. Weighted orbital integrals

4.1. The case of (G, H). Let E/F be a quadratic field extension and η the quadratic character of F^{\times}/NE^{\times} attached to it, where NE^{\times} denotes the norm of E^{\times} . For $x \in H(F)$, which is viewed as an element in G(F), we denote by Nrd(x) its reduced norm. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$ and that $Q \in \mathscr{F}^G(M)$. For all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we define the weighted orbital integral

(4.1.1)
$$J_M^Q(\eta, X, f) := |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(X))\eta(\mathrm{Nrd}(x))v_M^Q(x)dx$$

Since $v_M^Q(x)$ is left-invariant by M(F) and we have $H_X \subseteq M_H$ for $X \in \mathfrak{m} \cap \mathfrak{s}_{rs}$, we see that $v_M^Q(x)$ is left-invariant by $H_X(F)$. This integral is absolutely convergent since the orbit $\operatorname{Ad}(H(F))(X)$ is closed in $\mathfrak{s}(F)$, which ensures that the integrand is a compactly supported (and locally constant) function on the homogeneous space.

Notice that for $x \in M_H(F)$, we have $J_M^Q(\eta, \operatorname{Ad}(x^{-1})(X), f) = \eta(\operatorname{Nrd}(x))J_M^Q(\eta, X, f)$. Sometimes it is convenient to introduce a transfer factor as in [58, Definition 5.7]: for $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$, define

(4.1.2)
$$\kappa(X) := \eta(\operatorname{Nrd}(A))$$

where $\operatorname{Nrd}(A)$ denotes the reduced norm of $A \in GL_n(D)$. Then we have $\kappa(\operatorname{Ad}(x^{-1})(X)) = \eta(\operatorname{Nrd}(x))\kappa(X)$, and thus the function $\kappa(\cdot)J_M^Q(\eta, \cdot, f)$ is constant on $\operatorname{Ad}(M_H(F))(X)$.

Though we mainly consider $M \in \mathscr{L}^{G,\omega}(M_0)$, it is unharmful to extend our definition by (4.1.1) to all Levi subgroups of the form $M = \operatorname{Ad}(w)(L)$, where $L \in \mathscr{L}^{G,\omega}(M_0)$ and $w \in W_0^H$.

One may also extend in the obvious way the definition (4.1.1) of weighted orbital integrals to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G, \omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \mathrm{Ad}(\epsilon))$ in lower dimensions.

PROPOSITION 4.1. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$ and that $Q \in \mathscr{F}^G(M)$.

1) For $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ fixed, the support of the distribution $J^Q_M(\eta, X, \cdot)$ is contained in the closed orbit $\operatorname{Ad}(H(F))(X)$.

2) For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ fixed, the function $J^{Q}_{M}(\eta, \cdot, f)$ is locally constant on $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. If $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ is a Cartan subspace, the restriction of this function to $\mathfrak{c}_{reg}(F)$ vanishes outside a compact subset of $\mathfrak{c}(F)$.

3) If $w \in \operatorname{Norm}_{H(F)}(M_0)$, $x \in M_H(F)$ and $k \in K_H$, we have the equality

$$J^{G}_{\operatorname{Ad}(w)(M)}(\eta, \operatorname{Ad}(wx)(X), \operatorname{Ad}(k)(f)) = \eta(\operatorname{Nrd}(wxk))J^{G}_{M}(\eta, X, f)$$

for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$.

4) For $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$J_M^Q(\eta, X, f) = J_M^{M_Q}(\eta, X, f_Q^{\eta}),$$

where $f_Q^{\eta} \in \mathcal{C}_c^{\infty}((\mathfrak{m}_Q \cap \mathfrak{s})(F))$ is defined by (3.2.2). 5) (Descent formula) If $L \in \mathscr{L}^{G,\omega}(M_0)$, $L \subseteq M$ and $X \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$, we have

$$J^G_M(\eta,X,f) = \sum_{L' \in \mathscr{L}^G(L)} d^G_L(M,L') J^{L'}_L(\eta,X,f^\eta_{Q'})$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, where Q' denotes the second component of $\mathfrak{s}(M, L')$ (see Section 2.7).

6) (Non-equivariance) For $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, $y \in H(F)$ and $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$J_M^G(\eta, X, \operatorname{Ad}(y^{-1})(f)) = \eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^G(M)} J_M^{M_Q}(\eta, X, f_{Q,y}^{\eta})$$

where $f_{Q,y}^{\eta} \in \mathcal{C}_c^{\infty}((\mathfrak{m}_Q \cap \mathfrak{s})(F))$ is defined by

$$(4.1.3) \qquad f_{Q,y}^{\eta}(Z) := \int_{K_H \times (\mathfrak{n}_Q \cap \mathfrak{s})(F)} f(\mathrm{Ad}(k^{-1})(Z+U))\eta(\mathrm{Nrd}(k))v_Q'(ky)dUdk, \forall Z \in (\mathfrak{m}_Q \cap \mathfrak{s})(F).$$

PROOF. 1) This is obvious from the definition.

2) Let $Y \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Let \mathfrak{c} be the centraliser of Y in \mathfrak{s} . Then $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ is a Cartan subspace and $Y \in \mathfrak{c}_{reg}(F)$. Since $Ad(M_H(F))(\mathfrak{c}_{reg}(F))$ is an open subset of $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ (see [45, p. 105]), in order to prove the first statement, it suffices to find a neighbourhood U of Y in $\mathfrak{c}_{\mathrm{reg}}(F)$ on which the function $\kappa(\cdot)J_M^Q(\eta,\cdot,f)$ is constant. We shall follow the proof of [35, Theorem 17.11]. Consider the function ϕ on $\mathfrak{c}_{\mathrm{reg}}(F) \times (T_{\mathfrak{c}}(F) \setminus H(F))$ defined by $\phi(X, x) := (\kappa f)(\mathrm{Ad}(x^{-1})(X))$. Then ϕ is locally constant but usually not compactly supported. However, now choosing a compact neighbourhood $\sigma_{\mathfrak{c}}$ of Y in $\mathfrak{c}_{reg}(F)$, we see from Harish-Chandra's compactness lemma for symmetric spaces (Lemma 3.2) applied to $\sigma_{\mathfrak{s}} := \operatorname{Supp}(f)$ that the restriction of ϕ to $\sigma_{c} \times (T_{c}(F) \setminus H(F))$ is compactly supported. By [35, Lemma 2.1], there exists an open neighbourhood U of Y in $\sigma_{\mathfrak{c}}$ such that $\phi(X, x) = \phi(Y, x)$ for all $X \in U$ and $x \in T_{\mathfrak{c}}(F) \setminus H(F)$. It follows that the function $\kappa(\cdot)J^Q_M(\eta,\cdot,f)$ is constant on U.

The second statement is a corollary of Lemma 3.3.

3) The effect of Ad(w) is a consequence of our choice of Haar measures. The effect of Ad(x) results from the left-invariance of $v_M^G(x)$ by $M_H(F)$. The effect of Ad(k) comes from the right-invariance of

 $v_M^G(x)$ by K_H . One should keep in mind the effect of $\eta(\operatorname{Nrd}(x))$ in every step. 4) Write $Q_H := Q \cap H \in \mathscr{F}^H(M_0)$. One sees that $M_{Q_H} = M_Q \cap H$ and that $N_{Q_H} = N_Q \cap H$. Applying the change of variables x = mnk with $m \in M_{Q_H}(F)$, $n \in N_{Q_H}(F)$ and $k \in K_H$ in (4.1.1), since $v_M^Q(x) = v_M^{M_Q}(m)$, we have

$$J_{M}^{Q}(\eta, X, f) = |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{(M_{Q_{H}, X}(F) \setminus M_{Q_{H}}(F)) \times N_{Q_{H}}(F) \times K_{H}} f(\mathrm{Ad}(mnk)^{-1}(X))\eta(\mathrm{Nrd}(mk))v_{M}^{M_{Q}}(m)dkdndm.$$

Applying Lemma 3.12 to $Y = \operatorname{Ad}(m^{-1})(X)$, we deduce

$$\begin{split} J_{M}^{Q}(\eta, X, f) = & |D^{\mathfrak{m}_{Q}\cap\mathfrak{s}}(X)|_{F}^{1/2} \int_{(M_{Q_{H}, X}(F)\setminus M_{Q_{H}}(F))\times K_{H}\times(\mathfrak{n}_{Q}\cap\mathfrak{s})(F)} f(\mathrm{Ad}(k^{-1})(\mathrm{Ad}(m^{-1})(X) + U)) \\ & \eta(\mathrm{Nrd}(mk))v_{M}^{M_{Q}}(m)dUdkdm \\ = & |D^{\mathfrak{m}_{Q}\cap\mathfrak{s}}(X)|_{F}^{1/2} \int_{M_{Q_{H}, X}(F)\setminus M_{Q_{H}}(F)} f_{Q}^{\eta}(\mathrm{Ad}(m^{-1})(X))\eta(\mathrm{Nrd}(m))v_{M}^{M_{Q}}(m)dm \\ = & J_{M}^{M_{Q}}(\eta, X, f_{Q}^{\eta}). \end{split}$$

- 5) It follows from (7) in Section 2.7 and 4).
- 6) By the change of variables, we see that

$$J_M^G(\eta, X, \operatorname{Ad}(y^{-1})(f)) = |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X))\eta(\operatorname{Nrd}(xy))v_M(xy)dx.$$

For $x \in H(F)$ and $Q \in \mathscr{F}^{G,\omega}(M_0)$, let $k_Q(x)$ be an element in K_H such that $xk_Q(x)^{-1} \in Q_H(F)$. It follows from the product formula (2.6.1) that (see the proof of [5, Lemma 8.2])

$$v_M(xy) = \sum_{Q \in \mathscr{F}^G(M)} v_M^Q(x) v_Q'(k_Q(x)y).$$

As in 4), we write

$$\begin{split} J_M^G(\eta, X, \operatorname{Ad}(y^{-1})(f)) = & \eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^G(M)} |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) \eta(\operatorname{Nrd}(x)) v_M^Q(x) \\ & v_Q'(k_Q(x)y) dx \\ = & \eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^G(M)} |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{(M_{Q_H,X}(F) \setminus M_{Q_H}(F)) \times N_{Q_H}(F) \times K_H} \\ & f(\operatorname{Ad}(mnk)^{-1}(X)) \eta(\operatorname{Nrd}(mk)) v_M^{M_Q}(m) v_Q'(ky) dk dn dm. \end{split}$$

Applying again Lemma 3.12 to $Y = \operatorname{Ad}(m^{-1})(X)$, we obtain

$$\begin{split} J_{M}^{G}(\eta, X, \operatorname{Ad}(y^{-1})(f)) = &\eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} |D^{\mathfrak{m}_{Q} \cap \mathfrak{s}}(X)|_{F}^{1/2} \int_{(M_{Q_{H}, X}(F) \setminus M_{Q_{H}}(F)) \times K_{H} \times (\mathfrak{n}_{Q} \cap \mathfrak{s})(F)} \\ & f(\operatorname{Ad}(k^{-1})(\operatorname{Ad}(m^{-1})(X) + U))\eta(\operatorname{Nrd}(mk))v_{M}^{M_{Q}}(m)v_{Q}'(ky)dUdkdm \\ = &\eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} |D^{\mathfrak{m}_{Q} \cap \mathfrak{s}}(X)|_{F}^{1/2} \int_{M_{Q_{H}, X}(F) \setminus M_{Q_{H}}(F)} f_{Q, y}^{\eta}(\operatorname{Ad}(m^{-1})(X)) \\ & \eta(\operatorname{Nrd}(m))v_{M}^{M_{Q}}(m)dm \\ = &\eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} J_{M}^{M_{Q}}(\eta, X, f_{Q, y}^{\eta}). \end{split}$$

LEMMA 4.2. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$ and that $Q \in \mathscr{F}^G(M)$. Let $\sigma \subseteq \mathfrak{s}(F)$ be a compact subset. There exists c > 0 and $N \in \mathbb{N}$ such that if $x \in H(F)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ satisfy $\operatorname{Ad}(x^{-1})(X) \in \sigma$, then

$$|v_M^Q(x)| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

PROOF. It is shown in the proof of [51, Lemme III.5] that there exists $c_1 > 0$ and $N \in \mathbb{N}$ such that for all $x \in G(F)$,

$$|v_M^Q(x)| \le c_1 (1 + \log ||x||)^N.$$

Suppose that $x \in H(F)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ satisfy $\operatorname{Ad}(x^{-1})(X) \in \sigma$. If we replace x by yx and X by $\operatorname{Ad}(y)(X)$, where $y \in M_H(F)$, the two sides in the inequality to be proved remain unchanged. Since $\mathscr{T}(\mathfrak{m} \cap \mathfrak{s})$ is a finite set, we may fix a Cartan subspace $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ and suppose that $X \in \mathfrak{c}_{reg}(F)$. Let $\tau \in T_{\mathfrak{c}}(F)$ be such that

$$\|\tau x\| = \inf_{\tau' \in T_{\mathfrak{c}}(F)} \|\tau' x\|.$$

Then

$$|v_M^Q(x)| = |v_M^Q(\tau x)| \le c_1 (1 + \log \|\tau x\|)^N = c_1 (1 + \inf_{\tau' \in T_{\mathfrak{c}}(F)} \log \|\tau' x\|)^N.$$

Now it suffices to apply Lemma 3.4.

COROLLARY 4.3. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$ and that $Q \in \mathscr{F}^G(M)$. Let $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$. There exists c > 0 and $N \in \mathbb{N}$ such that for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we have

$$|J_M^{M_Q}(\eta, X, f_Q^{\eta})| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N.$$

PROOF. By Proposition 4.1.4) and Lemma 4.2 applied to $\sigma = \text{Supp}(f)$, we see that

$$\begin{split} |J_M^{M_Q}(\eta, X, f_Q^{\eta})| \leq & |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} |f(\operatorname{Ad}(x^{-1})(X)) v_M^Q(x)| dx \\ \leq & c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} |f(\operatorname{Ad}(x^{-1})(X))| dx. \end{split}$$

Now we draw our conclusion by Proposition 3.13.

4.2. The case of (G', H'). Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$ and that $Q' \in \mathscr{F}^{H'}(M')$. For all $f' \in \mathcal{T}^{H'}(M')$. $\mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, we define the weighted orbital integral

(4.2.1)
$$J_{M'}^{Q'}(Y,f') := |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} f'(\mathrm{Ad}(x^{-1})(Y)) v_{M'}^{Q'}(x) dx.$$

By the base change to an algebraic closure of F containing E, we see that $H'_Y \subseteq H'_{Y^2} \subseteq M'$ for $Y \in \widetilde{\mathfrak{m}}' \cap \mathfrak{s}'_{rs}$. Then $v_{M'}^{Q'}(x)$ is left-invariant by $H'_Y(F)$. This integral is absolutely convergent since the orbit $\operatorname{Ad}(H'(F))(Y)$ is closed in $\mathfrak{s}'(F)$. Notice that for $x \in M'(F)$, we have $J_{M'}^{Q'}(\operatorname{Ad}(x^{-1})(Y), f') = J_{M'}^{Q'}(Y, f')$, i.e., the function $J_{M'}^{Q'}(\cdot, f')$ is constant on $\mathrm{Ad}(M'(F))(Y)$. One may extend in the obvious way the definition (4.2.1) to the symmetric pair $(\widetilde{M}', M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$, since it appears as the product of some copies of the form $(G', H', Ad(\alpha))$ in lower dimensions.

PROPOSITION 4.4. Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$ and that $Q' \in \mathscr{F}^{H'}(M')$.

1) For $Y \in (\widetilde{\mathfrak{m}}' \cap \mathfrak{s}'_{rs})(F)$ fixed, the support of the distribution $J^{Q'}_{M'}(Y, \cdot)$ is contained in the closed orbit $\operatorname{Ad}(H'(F))(Y)$.

2) For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ fixed, the function $J^{Q'}_{M'}(\cdot, f')$ is locally constant on $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. If $\mathfrak{c}' \subseteq \widetilde{\mathfrak{m}'} \cap \mathfrak{s}'$ is a Cartan subspace, the restriction of this function to $\mathfrak{c}'_{reg}(F)$ vanishes outside a compact subset of $\mathfrak{c}'(F)$.

3) If $w \in \operatorname{Norm}_{H'(F)}(M'_0)$, $x \in M'(F)$ and $k \in K_{H'}$, we have the equality

$$J_{\operatorname{Ad}(w)(M')}^{H'}(\operatorname{Ad}(wx)(Y), \operatorname{Ad}(k)(f')) = J_{M'}^{H'}(Y, f')$$

for all $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$. 4) For $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we have the equality

$$J_{M'}^{Q'}(Y,f') = J_{M'}^{M_{Q'}}(Y,f'_{Q'}),$$

where $f'_{Q'} \in \mathcal{C}^{\infty}_{c}((\mathfrak{m}_{\widetilde{Q'}} \cap \mathfrak{s}')(F))$ is defined by (3.3.2).

5) (Descent formula) If $L' \in \mathscr{L}^{H'}(M'_0)$, $L' \subseteq M'$ and $Y \in (\widetilde{\mathfrak{l}} \cap \mathfrak{s}'_{rs})(F)$, we have

$$J_{M'}^{H'}(Y,f') = \sum_{L \in \mathscr{L}^{H'}(L')} d_{L'}^{H'}(M',L) J_{L'}^{L}(Y,f'_Q)$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, where Q denotes the second component of $\mathfrak{s}(M', L)$ (see Section 2.7).

6) (Noninvariance) For $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, $y \in H'(F)$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we have the equality

$$J_{M'}^{H'}(Y, \operatorname{Ad}(y^{-1})(f')) = \sum_{Q' \in \mathscr{F}^{H'}(M')} J_{M'}^{M_{Q'}}(Y, f'_{Q', y})$$

where $f'_{Q',y} \in \mathcal{C}^{\infty}_{c}((\mathfrak{m}_{\widetilde{Q'}} \cap \mathfrak{s}')(F))$ is defined by

$$(4.2.2) f'_{Q',y}(Z) := \int_{K_{H'} \times (\mathfrak{n}_{\widetilde{Q'}} \cap \mathfrak{s}')(F)} f'(\mathrm{Ad}(k^{-1})(Z+U))v'_{Q'}(ky)dUdk, \forall Z \in (\mathfrak{m}_{\widetilde{Q'}} \cap \mathfrak{s}')(F).$$

PROOF. It is almost the same as the proof of Proposition 4.1, except that one needs to use Lemma 3.19 to show 4) and 6).

LEMMA 4.5 (cf. Lemma 4.2). Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$ and that $Q' \in \mathscr{F}^{H'}(M')$. Let $\sigma' \subseteq \mathfrak{s}'(F)$ be a compact subset. There exists c > 0 and $N \in \mathbb{N}$ such that if $x \in H'(F)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ satisfy $\operatorname{Ad}(x^{-1})(Y) \in \sigma'$, then

$$|v_{M'}^{Q'}(x)| \le c \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^N.$$

COROLLARY 4.6. Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$ and that $Q' \in \mathscr{F}^{H'}(M')$. Let $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$. There exists c > 0 and $N \in \mathbb{N}$ such that for all $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, we have

$$|J_{M'}^{M_{Q'}}(Y, f_{Q'}')| \le c \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^N.$$

PROOF. We may apply the argument of Corollary 4.3 with the help of Proposition 4.4.4), Lemma 4.5 and Proposition 3.20.

5. The noninvariant trace formula

5.1. The case of (G, H). Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$. For $x, y \in G(F)$, we define a (G, M)-family $(v_P(x, y))_{P \in \mathscr{P}^G(M)}$ as in [8, (12.1) in §12] by

$$v_P(\lambda, x, y) := e^{-\lambda(H_P(y) - H_{\overline{P}}(x))}, \forall \lambda \in i\mathfrak{a}_M^*, P \in \mathscr{P}^G(M),$$

where $\overline{P} \in \mathscr{P}^G(M)$ is the parabolic subgroup opposite to P. Let E/F be a quadratic field extension and η the quadratic character of F^{\times}/NE^{\times} attached to it. For $f, f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$, we define

$$J_M^G(\eta, X, f, f') := |D^{\mathfrak{s}}(X)|_F \int_{(A_M(F)\setminus H(F))^2} f(\mathrm{Ad}(x^{-1})(X)) f'(\mathrm{Ad}(y^{-1})(X)) \eta(\mathrm{Nrd}(x^{-1}y)) v_M(x, y) dx dy.$$

PROPOSITION 5.1. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$ and that $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$.

1) The integral (5.1.1) is absolutely convergent.

2) The function $J_M^G(\eta, \cdot, f, f')$ is locally constant on $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$.

3) If $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ is an *M*-elliptic Cartan subspace, the restriction of $J_M^G(\eta, \cdot, f, f')$ to $\mathfrak{c}_{reg}(F)$ vanishes outside a compact subset of $\mathfrak{c}(F)$.

4) If $w \in \operatorname{Norm}_{H(F)}(M_0)$, $x \in M_H(F)$ and $k, k' \in K_H$, we have the equality

$$J^G_{\operatorname{Ad}(w)(M)}(\eta, \operatorname{Ad}(wx)(X), \operatorname{Ad}(k)(f), \operatorname{Ad}(k')(f')) = \eta(\operatorname{Nrd}(kk'))J^G_M(\eta, X, f, f')$$

for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$.

5) There exists c > 0 and $N \in \mathbb{N}$ such that for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$, we have

$$|J_M^G(\eta, X, f, f')| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

6) For all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$, we have

$$J_M^G(\eta, X, f, f') = \sum_{L_1, L_2 \in \mathscr{L}^G(M)} d_M^G(L_1, L_2) J_M^{L_1}(\eta, X, f_{\overline{Q_1}}^{\eta}) J_M^{L_2}(\eta, X, f'_{\overline{Q_2}}),$$

where $(Q_1, Q_2) := s(L_1, L_2)$ (see Section 2.7).

PROOF. The statements 1)-4) can be proved in the same way as the proof of analogous properties for (4.1.1) in Section 4. Notice that the $\eta(Nrd(\cdot))$ -invariant effects coming from x and y may sometimes cancel.

For $x \in G(F)$, we define a (G, M)-family $(\overline{v}_P(x))_{P \in \mathscr{P}^G(M)}$ by

$$\overline{v}_P(\lambda, x) := e^{\lambda(H_{\overline{P}}(x))}, \forall \lambda \in i\mathfrak{a}_M^*, P \in \mathscr{P}^G(M).$$

Then $v_P(x,y) = \overline{v}_P(x)v_P(y)$ as the product of (G, M)-families. Notice that for all $Q \in \mathscr{P}^G(M)$ and $x \in G(F)$, we have

$$\overline{v}_M^Q(x) = v_M^Q(x).$$

The statement 6) is a consequence of the splitting formula of (G, M)-families ((6) in Section 2.7) and Proposition 4.1.4). It together with Corollary 4.3 implies the statement 5).

For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we define

(5.1.2)
$$J^{G}(\eta, f, f') := \sum_{\substack{M \in \mathscr{L}^{G, \omega}(M_{0}) \\ J^{G}_{M}(\eta, X, f, f') dX.}} |W^{GL_{n}}|^{-1} (-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W(M_{H}, \mathfrak{c})|^{-1} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |W(M_{H}, \mathfrak{c})|^{-1} |W(M_{H}, \mathfrak{c})|$$

This expression is absolutely convergent by Proposition 5.1.5) and Corollary 3.6.

REMARK 5.2. We have the equality

$$J^G(\eta, f, f') = J^G(\eta, f', f).$$

It results from the fact that for all $M \in \mathscr{L}^{G,\omega}(M_0)$ and all $x, y \in G(F)$, we have $v_M(x,y) = v_M(y,x)$.

Again, one may extend in the obvious way the definitions (5.1.1) and (5.1.2) to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \operatorname{Ad}(\epsilon))$ in lower dimensions.

THEOREM 5.3 (Noninvariant trace formula). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$J^G(\eta, f, \hat{f}') = J^G(\eta, \hat{f}, f').$$

The rest of this section is devoted to the proof of Theorem 5.3. Fix $P_0 \in \mathscr{P}^G(M_0)$. Denote

$$\overline{\mathfrak{a}_{P_0}^+} := \{ T \in \mathfrak{a}_{M_0} : \alpha(T) \ge 0, \forall \alpha \in \Delta_{P_0}^G \}.$$

For $T \in \overline{\mathfrak{a}_{P_0}^+}$, write

$$d(T) := \inf_{\alpha \in \Delta_{P_0}^G} \alpha(T),$$

which is invariant under the translation by \mathfrak{a}_G . Set $R_0 := (\mathfrak{a}_{M_0,F} + \mathfrak{a}_G)/\mathfrak{a}_G$, which is a lattice in $\mathfrak{a}_{M_0}/\mathfrak{a}_G$. For $T \in R_0 \cap (\overline{\mathfrak{a}_{P_0}^+}/\mathfrak{a}_G)$, we define a function $u(\cdot,T)$ on $A_G(F) \setminus G(F)$ as in [8, p. 21], which is the characteristic function of certain compact subset. To be precise, let $C_{M_0}(T)$ be the convex hull in $\mathfrak{a}_{M_0}/\mathfrak{a}_G$ of

$$\{T_B: B \in \mathscr{P}^G(M_0)\},\$$

where T_B denotes the unique W_0^G -translate of T which lies in $\overline{\mathfrak{a}_B^+}$. Then u(x,T) is defined as the characteristic function of the set of points

$$x = k_1 m k_2, m \in A_G(F) \backslash M_0(F), k_1, k_2 \in K_G,$$

in $A_G(F) \setminus G(F)$ such that $H_{M_0}(m)$ lies in $C_{M_0}(T)$. Let $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$. For $x \in H(F)$, we define

$$k(x, f, f') := \int_{\mathfrak{s}(F)} f(X) f'(\operatorname{Ad}(x^{-1})(X)) dX.$$

For $T \in R_0 \cap (\overline{\mathfrak{a}_{P_0}^+}/\mathfrak{a}_G)$, we define

$$K^{T}(\eta, f, f') := \int_{A_{G}(F) \setminus H(F)} k(x, f, f') \eta(\operatorname{Nrd}(x)) u(x, T) dx$$

Since $A_G(F) \setminus H(F)$ is a closed subgroup of $A_G(F) \setminus G(F)$, the restriction of u(x,T) to $A_G(F) \setminus H(F)$ is also compactly supported, and the above integral is absolutely convergent.

By the Weyl integration formula (Proposition 3.11), we obtain the geometric expansion

$$K^{T}(\eta, f, f') = \sum_{M \in \mathscr{L}^{G, \omega}(M_{0})} |W_{0}^{M_{n}}| |W_{0}^{GL_{n}}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{ell}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{reg}(F)} K^{T}(\eta, X, f, f') dX,$$

where

$$\begin{split} K^{T}(\eta, X, f, f') &:= |D^{\mathfrak{s}}(X)|_{F} \int_{A_{G}(F) \setminus H(F)} \int_{A_{M}(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) f'(\operatorname{Ad}(xy)^{-1}(X)) \eta(\operatorname{Nrd}(y)) \\ & u(y, T) dx dy. \end{split}$$

By the change of variables $xy \mapsto y$, we can write (5.1.3)

$$K^{T}(\eta, X, f, f') = |D^{\mathfrak{s}}(X)|_{F} \int_{(A_{M}(F)\backslash H(F))^{2}} f(\operatorname{Ad}(x^{-1})(X))f'(\operatorname{Ad}(y^{-1})(X))\eta(\operatorname{Nrd}(x^{-1}y))u_{M}(x, y, T)dxdy$$

where

$$u_M(x,y,T) := \int_{A_G(F) \setminus A_M(F)} u(x^{-1}ay,T) da$$

is defined as in [8, p. 21].

For $x, y \in G(F)$ and $T \in R_0 \cap (\overline{\mathfrak{a}_{P_0}^+}/\mathfrak{a}_G)$, we define the second weight function $v_M(x, y, T)$ as in [8, p. 30], which is left-invariant under the multiplication of $A_M(F)$ on x or y. To be precise, let $\lambda \in \mathfrak{a}_M^* \otimes_{\mathbb{R}} \mathbb{C}$ be a point whose real part $\lambda_{\mathbb{R}} \in \mathfrak{a}_M^*$ is in general position. For $P \in \mathscr{P}^G(M)$, set

$$\Delta_P^{\lambda} := \{ \alpha \in \Delta_P^G : \lambda_{\mathbb{R}}(\alpha^{\vee}) < 0 \},\$$

where α^{\vee} is the "coroot" associated to $\alpha \in \Delta_P^G$ (see [9, p. 26]). Denote by φ_P^{λ} the characteristic function of the set of $T' \in \mathfrak{a}_M$ such that $\varpi_{\alpha}(T') > 0$ for each $\alpha \in \Delta_P^{\lambda}$ and that $\varpi_{\alpha}(T') \leq 0$ for each $\alpha \in \Delta_P^G - \Delta_P^{\lambda}$, where $\{\varpi_{\alpha} : \alpha \in \Delta_P^G\}$ is the basis of $(\mathfrak{a}_P^G)^*$ which is dual to $\{\alpha^{\vee} : \alpha \in \Delta_P^G\}$. Let

$$Y_P(x, y, T) := T_P + H_P(x) - H_{\overline{P}}(y), \forall P \in \mathscr{P}^G(M)$$

The set $\mathscr{Y}_M(x, y, T) = \{Y_P(x, y, T) : P \in \mathscr{P}^G(M)\}$ is a (G, M)-orthogonal set in the sense of [8, p. 19]. Define

$$\sigma_M(T',\mathscr{Y}_M(x,y,T)) := \sum_{P \in \mathscr{P}^G(M)} (-1)^{|\Delta_P^{\lambda}|} \varphi_P^{\lambda}(T' - Y_P(x,y,T)), \forall T' \in \mathfrak{a}_M/\mathfrak{a}_G.$$

The function $\sigma_M(\cdot, \mathscr{Y}_M(x, y, T))$ is known to be compactly supported (see [8, p. 22]). Then $v_M(x, y, T)$ is defined as the integral

$$v_M(x,y,T) := \int_{A_G(F) \setminus A_M(F)} \sigma_M(H_M(a), \mathscr{Y}_M(x,y,T)) da.$$

Now, we define the corresponding weighted orbital integral (5.1.4)

$$J^{T}(\eta, X, f, f') := |D^{\mathfrak{s}}(X)|_{F} \int_{(A_{M}(F)\setminus H(F))^{2}} f(\operatorname{Ad}(x^{-1})(X))f'(\operatorname{Ad}(y^{-1})(X))\eta(\operatorname{Nrd}(x^{-1}y))v_{M}(x, y, T)dxdy$$

Let $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ be an *M*-elliptic Cartan subspace. For $\varepsilon > 0$ and $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with large ||T||, consider the domain near the singular set

$$\mathfrak{c}(\varepsilon, T) := \{ X \in \mathfrak{c}_{\mathrm{reg}}(F) : |D^{\mathfrak{s}}(X)| \le e^{-\varepsilon ||T||} \}.$$

LEMMA 5.4. Fix an arbitrary constant $\varepsilon_0 > 0$. Fix a constant $\varepsilon' > 0$ satisfying the condition of Lemma 3.5. Let $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ be an M-elliptic Cartan subspace. Given $\varepsilon > 0$, there exists c > 0 such that for any $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with $||T|| \ge \varepsilon_0$,

$$\int_{\mathfrak{c}(\varepsilon,T)} (|K^T(\eta, X, f, f')| + |J^T(\eta, X, f, f')|) dX \le c e^{-\frac{\varepsilon' \varepsilon ||T||}{2}}$$

PROOF. It is shown in [8, (4.8) in p. 31] that there exist $c_1, d_1 > 0$ such that for all $x, y \in G(F)$ and $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with $||T|| \ge \varepsilon_0$,

$$u_M(x, y, T) \le c_1(||T|| + \log ||x|| + \log ||y||)^{d_1}$$

For any $a_1, a_2 \in A_M(F)$, we deduce that

$$u_M(x, y, T) = u_M(a_1 x, a_2 y, T) \le c_1(||T|| + \log ||a_1 x|| + \log ||a_2 y||)^{d_1}.$$

Since $T_{\mathfrak{c}}(F)/A_M(F)$ is compact, there exists $c_2 > 0$ such that

$$u_M(x, y, T) \le c_2(||T|| + \inf_{\tau_1 \in T_{\mathfrak{c}}(F)} \log ||\tau_1 x|| + \inf_{\tau_2 \in T_{\mathfrak{c}}(F)} \log ||\tau_2 y||)^{d_1}.$$

Now let $x, y \in H(F)$ and $X \in \mathfrak{c}(\varepsilon, T)$, and assume that

$$f(\mathrm{Ad}(x^{-1})(X))f'(\mathrm{Ad}(y^{-1})(X)) \neq 0.$$

Let $\sigma \subseteq \mathfrak{s}(F)$ be a compact subset containing $\operatorname{Supp}(f) \cup \operatorname{Supp}(f')$. From Lemma 3.4, there exists $c_{\sigma} > 0$ such that

$$\inf_{\tau_1 \in T_{\mathfrak{c}}(F)} \log \|\tau_1 x\|, \inf_{\tau_2 \in T_{\mathfrak{c}}(F)} \log \|\tau_2 y\| \le c_\sigma \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}$$

Therefore, there exists $c'_{\sigma} > 0$ such that

(5.1.5)
$$u_M(x, y, T) \le c'_{\sigma} (\|T\| - \log |D^{\mathfrak{s}}(X)|_F)^{d_1}$$

By Proposition 3.13, there exists $c_3 > 0$ such that

(5.1.6)
$$|D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{A_{M}(F)\setminus H(F)} |f(x^{-1}Xx)| dx \le c_{\mathfrak{Z}}$$

and

(5.1.7)
$$|D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{A_{M}(F)\setminus H(F)} |f'(y^{-1}Xy)| dy \le c_{3}$$

for all $X \in \mathfrak{c}_{\mathrm{reg}}(F)$.

Putting the estimates (5.1.5), (5.1.6) and (5.1.7) into the definition (5.1.3) of $K^T(\eta, X, f, f')$, we obtain the inequality

$$|K^T(\eta, X, f, f')| \le c'_{\sigma} c_3^2(||T|| - \log |D^{\mathfrak{s}}(X)|_F)^{d_1}.$$

By Lemma 3.5, for any subset B of $\mathfrak{c}_{reg}(F)$ which is relatively compact in $\mathfrak{c}(F)$, there exists $c_B > 0$ such that

$$\int_{B} |D^{\mathfrak{s}}(X)|_{F}^{-\varepsilon'} dX \le c_{B}.$$

We deduce that for $m \in \mathbb{Z}$,

(5.1.8)
$$\operatorname{vol}(B \cap \{X \in \mathfrak{c}_{\operatorname{reg}}(F) : |D^{\mathfrak{s}}(X)|_{F} = q^{-\frac{m}{2}}\}) \leq c_{B}q^{-\frac{\varepsilon'm}{2}}.$$

We claim that for any B as above, there exists $c'_B > 0$ such that

$$\int_{B\cap\mathfrak{c}(\varepsilon,T)} (\|T\| - \log |D^{\mathfrak{s}}(X)|_F)^{d_1} dX \le c'_B e^{-\frac{\varepsilon'\varepsilon\|T\|}{2}}.$$

This is an analogue of the exercise in [8, p. 32] and we include here a proof for completeness. For $X \in \mathfrak{c}(\varepsilon, T)$, we have

$$||T|| \le -\frac{1}{\varepsilon} \log |D^{\mathfrak{s}}(X)|_F.$$

Therefore,

$$\int_{B\cap\mathfrak{c}(\varepsilon,T)} (\|T\| - \log|D^{\mathfrak{s}}(X)|_F)^{d_1} dX \le \left(1 + \frac{1}{\varepsilon}\right)^{d_1} \int_{B\cap\mathfrak{c}(\varepsilon,T)} (-\log|D^{\mathfrak{s}}(X)|_F)^{d_1} dX.$$

Since

$$B \cap \mathfrak{c}(\varepsilon, T) = \coprod_{m \ge \frac{2\varepsilon \|T\|}{\log q}} (B \cap \{X \in \mathfrak{c}_{\mathrm{reg}}(F) : |D^{\mathfrak{s}}(X)|_{F} = q^{-\frac{m}{2}}\}),$$

we have

$$\int_{B\cap\mathfrak{c}(\varepsilon,T)} (-\log|D^{\mathfrak{s}}(X)|_F)^{d_1} dX = \sum_{\substack{m \ge \frac{2\varepsilon ||T||}{\log q}}} \left(\frac{m\log q}{2}\right)^{d_1} \operatorname{vol}(B \cap \{X \in \mathfrak{c}_{\operatorname{reg}}(F) : |D^{\mathfrak{s}}(X)|_F = q^{-\frac{m}{2}}\}).$$

Applying (5.1.8), we obtain

$$\int_{B\cap\mathfrak{c}(\varepsilon,T)} (-\log|D^{\mathfrak{s}}(X)|_F)^{d_1} dX \le \sum_{m\ge \frac{2\varepsilon\|T\|}{\log q}} \left(\frac{m\log q}{2}\right)^{d_1} c_B q^{-\frac{\varepsilon'm}{2}}.$$

Now we can confirm our claim by noting the basic fact: for d > 0 and a > 1, there exists $c_{d,a} > 0$ such that

$$\sum_{m \ge x} \frac{m^d}{a^m} \le c_{d,a} a^{-\frac{x}{2}}, \forall x \ge 0.$$

Taking

$$B = \{ X \in \mathfrak{c}_{\mathrm{reg}}(F) : K^T(\eta, X, f, f') \neq 0 \},\$$

we see that

$$\int_{\mathfrak{c}(\varepsilon,T)} |K^T(\eta, X, f, f')| dX \le c'_{\sigma} c_3^2 c'_B e^{-\frac{\varepsilon' \varepsilon ||T||}{2}}.$$

This is half of the lemma.

It is proved in [8, p. 32] that there exist $c_4, d_2 > 0$ such that for all $x, y \in G(F)$ and $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with $||T|| \ge \varepsilon_0$,

 $v_M(x, y, T) \le c_4(||T|| + \log ||x|| + \log ||y||)^{d_2}.$

By the same argument as before, we obtain

$$\int_{\mathfrak{c}(\varepsilon,T)} |J^T(\eta, X, f, f')| dX \le c_5 e^{-\frac{\varepsilon'\varepsilon ||T||}{2}}$$

for some $c_5 > 0$. This establishes the other half of the lemma.

LEMMA 5.5. Suppose that $\delta > 0$. Then there exist $c, \varepsilon_1, \varepsilon_2 > 0$ such that

$$|u_M(x, y, T) - v_M(x, y, T)| \le c e^{-\varepsilon_1 ||T|}$$

for all $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with $d(T) \ge \delta \|T\|$, and all $x, y \in \{x \in G(F) : \|x\| \le e^{\varepsilon_2 \|T\|}\}.$

PROOF. This is Arthur's main geometric lemma [8, Lemma 4.4].

LEMMA 5.6. Suppose that $\delta > 0$. Let $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ be an M-elliptic Cartan subspace. Then there exist $c, \varepsilon > 0$ such that

$$\int_{\mathfrak{c}_{\mathrm{reg}}(F)} |K^T(\eta, X, f, f') - J^T(\eta, X, f, f')| dX \le c e^{-\varepsilon \|T\|}$$

for all $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with sufficiently large ||T|| and $d(T) \ge \delta ||T||$.

PROOF. Fix $\varepsilon_2 > 0$ to be the constant given by Lemma 5.5. Let $x, y \in H(F)$ and $X \in \mathfrak{c}_{reg}(F) - \mathfrak{c}(\frac{\varepsilon_2}{2}, T)$, and assume that

$$f(\operatorname{Ad}(x^{-1})(X))f'(\operatorname{Ad}(y^{-1})(X)) \neq 0.$$

Let $\sigma \subseteq \mathfrak{s}(F)$ be a compact subset containing $\operatorname{Supp}(f) \cup \operatorname{Supp}(f')$. From Lemma 3.4, there exists $c_{\sigma} > 0$ such that

$$\inf_{1 \in T_{\mathfrak{c}}(F)} \|\tau_1 x\|, \inf_{\tau_2 \in T_{\mathfrak{c}}(F)} \|\tau_2 y\| \le c_{\sigma} \sup\{1, |D^{\mathfrak{s}}(X)|_F^{-1}\}.$$

Since $X \in \mathfrak{c}_{reg}(F) - \mathfrak{c}(\frac{\varepsilon_2}{2}, T)$, we have

$$\sup\{1, |D^{\mathfrak{s}}(X)|_{F}^{-1}\} \le \sup\{1, e^{\frac{\varepsilon_{2} ||T||}{2}}\} = e^{\frac{\varepsilon_{2} ||T||}{2}}$$

Then, multiplying x and y by elements in $T_{\mathfrak{c}}(F)$ if necessary, and taking $||T|| \geq \frac{2 \log c_{\sigma}}{\varepsilon_2}$, we can assume that

$$||x||, ||y|| \le e^{\varepsilon_2 ||T||}$$

It follows from Lemma 5.5 that

$$|u_M(x, y, T) - v_M(x, y, T)| \le c e^{-\varepsilon_1 ||T||}.$$

By the definitions (see (5.1.3) and (5.1.4)) of $K^T(\eta, X, f, f')$ and $J^T(\eta, X, f, f')$, we obtain that

$$\int_{\mathfrak{c}_{\mathrm{reg}}(F)-\mathfrak{c}(\frac{\varepsilon_2}{2},T)} |K^T(\eta, X, f, f') - J^T(\eta, X, f, f')| dX \le c_1 e^{-\varepsilon_1 ||T||},$$

where

$$c_1 := c \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_F \int_{(A_M(F)\backslash H(F))^2} |f(\mathrm{Ad}(x^{-1})(X))f'(\mathrm{Ad}(y^{-1})(X))| dxdydX$$

Dependention 2.12 and Lemma 2.2

is finite by Proposition 3.13 and Lemma 3.3.

One can draw the conclusion by combining this with Lemma 5.4.

Define

$$J^{T}(\eta, f, f') := \sum_{M \in \mathscr{L}^{G, \omega}(M_{0})} |W_{0}^{M_{n}}| |W_{0}^{GL_{n}}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J^{T}(\eta, X, f, f') dX,$$

where $J^T(\eta, X, f, f')$ is defined by (5.1.4).

PROPOSITION 5.7. Suppose that $\delta > 0$. Then there exist $c, \varepsilon > 0$ such that

$$|K^T(\eta, f, f') - J^T(\eta, f, f')| \le ce^{-\varepsilon ||T||}$$

for all $T \in \mathfrak{a}_{M_0,F} \cap \overline{\mathfrak{a}_{P_0}^+}$ with sufficiently large ||T|| and $d(T) \ge \delta ||T||$.

PROOF. Apply Lemma 5.6.

It is proved in [8, (6.5) in p. 46] that $v_M(x, y, T)$ is an exponential polynomial in $T \in R_0 \cap (\mathfrak{a}_{P_0}^+/\mathfrak{a}_G)$. Denote by $\tilde{v}_M(x, y)$ the constant term of $v_M(x, y, T)$ as in [8, (6.6) in p. 46]. Then for $f, f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}, J^T(\eta, X, f, f')$ is also an exponential polynomial in $T \in R_0 \cap (\mathfrak{a}_{P_0}^+/\mathfrak{a}_G)$ whose constant term is given by (5.1.9)

$$\tilde{J}_M(\eta, X, f, f') := |D^{\mathfrak{s}}(X)|_F \int_{(A_M(F)\setminus H(F))^2} f(\mathrm{Ad}(x^{-1})(X)) f'(\mathrm{Ad}(y^{-1})(X)) \eta(\mathrm{Nrd}(x^{-1}y)) \tilde{v}_M(x, y) dx dy.$$

Thus for $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), J^{T}(\eta, f, f')$ is still an exponential polynomial in $T \in R_{0} \cap (\overline{\mathfrak{a}_{P_{0}}^{+}}/\mathfrak{a}_{G})$ whose constant term is given by

$$(5.1.10) \ \tilde{J}(\eta, f, f') := \sum_{M \in \mathscr{L}^{G, \omega}(M_0)} |W_0^{M_n}| |W_0^{GL_n}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\text{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_H, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\text{reg}}(F)} \tilde{J}_M(\eta, X, f, f') dX.$$

COROLLARY 5.8. For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$\tilde{I}(\eta, f, \hat{f}') = \tilde{J}(\eta, \hat{f}, f').$$

PROOF. By the Plancherel formula, for $x \in H(F)$, we have

$$k(x, f, \tilde{f'}) = k(x, \tilde{f}, f').$$

Then for all $T \in R_0 \cap (\overline{\mathfrak{a}_{P_0}^+}/\mathfrak{a}_G)$,

$$K^T(\eta, f, \hat{f}') = K^T(\eta, \hat{f}, f')$$

Finally, apply Proposition 5.7 to conclude.

LEMMA 5.9. For all $Q \in \mathscr{F}^{G,\omega}(M_0)$, there exists a constant c'_Q such that for all $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$, we have the equality

$$\tilde{J}(\eta, f, f') = \sum_{Q \in \mathscr{F}^{G, \omega}(M_0)} |W_0^{M_{Q_n}}| |W_0^{GL_n}|^{-1} (-1)^{\dim(A_Q/A_G)} J^{M_Q}(\eta, f_{\overline{Q}}^{\eta}, f'_Q^{\eta}) c'_Q,$$

where $J^{M_Q}(\eta, f^{\eta}_{\overline{Q}}, f'^{\eta}_{Q})$ is defined by (5.1.2).

PROOF. Suppose that $M \in \mathscr{L}^{G,\omega}(M_0)$. It is shown in [8, p. 92] that

$$\tilde{v}_M(x,y) = (-1)^{\dim(A_M/A_G)} \sum_{Q \in \mathscr{F}^G(M)} v_M^Q(x,y) c'_Q,$$

where c'_Q is a constant for each $Q \in \mathscr{F}^{G,\omega}(M_0)$.

Now substitude this in the definition (5.1.9) of $\tilde{J}_M(\eta, X, f, f')$. Note that

$$v_M^Q(m_1n_1k_1, m_2n_2k_2) = v_M^{M_Q}(m_1, m_2)$$

for $m_1 \in M_{\overline{Q}_H}(F), n_1 \in N_{\overline{Q}_H}(F), m_2 \in M_{Q_H}(F), n_2 \in N_{Q_H}(F), k_1, k_2 \in K_H$. By the same argument as the proof of Proposition 4.1.4), one shows that

$$|D^{\mathfrak{s}}(X)|_{F} \int_{(A_{M}(F)\setminus H(F))^{2}} f(\mathrm{Ad}(x^{-1})(X))f'(\mathrm{Ad}(y^{-1})(X))\eta(\mathrm{Nrd}(x^{-1}y))v_{M}^{Q}(x,y)dxdy = J_{M}^{M_{Q}}(\eta, X, f_{\overline{Q}}^{\eta}, f'_{Q}^{\eta}),$$

where $J_M^{M_Q}(\eta, X, f_{\overline{Q}}^{\eta}, f'_Q^{\eta})$ is defined by (5.1.1). Therefore, we have

$$\tilde{J}_M(\eta,X,f,f') = (-1)^{\dim(A_M/A_G)} \sum_{Q \in \mathscr{F}^G(M)} J_M^{M_Q}(\eta,X,f_{\overline{Q}}^{\eta},f'_Q^{\eta}) c'_Q.$$

Then the lemma follows from the definition (5.1.10) of $\tilde{J}(\eta, f, f')$.

PROOF OF THEOREM 5.3. Using Lemma 5.9 and Corollary 5.8, we can prove the theorem by induction on the dimension of G.

 \Box

5.2. The case of (G', H'). Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$. For $x, y \in H'(F)$, we define an (H', M')-family $(v_{P'}(x, y))_{P' \in \mathscr{P}^{H'}(M')}$ by

$$v_{P'}(\lambda, x, y) := e^{-\lambda(H_{P'}(y) - H_{\overline{P'}}(x))}, \forall \lambda \in i\mathfrak{a}_{M'}^*, P' \in \mathscr{P}^{H'}(M'), \forall \lambda \in i\mathfrak{a}_{M'}^*, P' \in \mathscr{P}^{H'}(M'), \forall \lambda \in i\mathfrak{a}_{M'}^*, \forall \lambda \in i\mathfrak{a}_{M'}^*, P' \in \mathscr{P}^{H'}(M'), \forall \lambda \in i\mathfrak{a}_{M'}^*, P' \in \mathscr{P}^{H'}(M'), \forall \lambda \in i\mathfrak{a}_{M'}^*, \forall \lambda \in i\mathfrak{a}$$

where $\overline{P'} \in \mathscr{P}^{H'}(M')$ is the parabolic subgroup opposite to P'. For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$, we define

(5.2.1)
$$J_{M'}^{H'}(Y,f,f') := |D^{\mathfrak{s}'}(Y)|_F \int_{(A_{M'}(F)\setminus H'(F))^2} f(\operatorname{Ad}(x^{-1})(Y))f'(\operatorname{Ad}(y^{-1})(Y))v_{M'}(x,y)dxdy.$$

PROPOSITION 5.10. Suppose that $M' \in \mathscr{L}^{H'}(M'_0)$ and that $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$.

1) The integral (5.2.1) is absolutely convergent.

2) The function $J_{M'}^{H'}(\cdot, f, f')$ is locally constant on $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$.

3) If $\mathfrak{c}' \subseteq \widetilde{\mathfrak{m}'} \cap \mathfrak{s}'$ is an M'-elliptic Cartan subspace, the restriction of $J_{M'}^{H'}(\cdot, f, f')$ to $\mathfrak{c}'_{reg}(F)$ vanishes outside a compact subset of $\mathfrak{c}'(F)$.

4) If $w \in \operatorname{Norm}_{H'(F)}(M'_0)$, $x \in M'(F)$ and $k, k' \in K_{H'}$, we have the equality

$$J_{\operatorname{Ad}(w)(M')}^{H'}(\operatorname{Ad}(wx)(Y), \operatorname{Ad}(k)(f), \operatorname{Ad}(k')(f')) = J_{M'}^{H'}(Y, f, f')$$

for all $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$.

5) There exists c > 0 and $N \in \mathbb{N}$ such that for all $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$, we have

$$|J_{M'}^{H'}(Y, f, f')| \le c \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^N.$$

6) For all $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$, we have

$$J_{M'}^{H'}(Y,f,f') = \sum_{L'_1,L'_2 \in \mathscr{L}^{H'}(M')} d_{M'}^{H'}(L'_1,L'_2) J_{M'}^{L'_1}(Y,f_{\overline{Q'_1}}) J_{M'}^{L'_2}(Y,f'_{Q'_2}),$$

where $(Q'_1, Q'_2) := s(L'_1, L'_2)$ (see Section 2.7).

PROOF. It is almost the same as the proof of Proposition 5.1, except that one needs to use Proposition 4.4.4) and Corollary 4.6 to show 6) and 5). \Box

For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we define

(5.2.2)
$$J^{H'}(f,f') := \sum_{\substack{M' \in \mathscr{L}^{H'}(M'_0) \\ J^{H'}_{M'}(Y,f,f') dY.}} |W_0^{H'}| |W_0^{M'}|^{-1} (-1)^{\dim(A_{M'}/A_{H'})} \sum_{\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')} |W(M',\mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M',\mathfrak{c}')|^{-1} |W(M',\mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M',\mathfrak{c}')|^{-1} |W(M$$

This expression is absolutely convergent by Proposition 5.10.5) and Corollary 3.6. One may extend in the obvious way the definitions (5.2.1) and (5.2.2) to the symmetric pair $(\widetilde{M}', M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$.

REMARK 5.11. We have the equality

$$J^{H'}(f, f') = J^{H'}(f', f).$$

It results from the fact that for all $M' \in \mathscr{L}^{H'}(M'_0)$ and all $x, y \in H'(F)$, we have $v_{M'}(x, y) = v_{M'}(y, x)$.

THEOREM 5.12 (Noninvariant trace formula). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we have the equality

$$J^{H'}(f, \hat{f}') = J^{H'}(\hat{f}, f').$$

PROOF. We may simply copy the proof of Theorem 5.3 here with obvious modifications. Especially, one needs to use Proposition 3.20 to show analogues of Lemmas 5.4 and 5.6 for the case of (G', H'). \Box

6. Howe's finiteness for weighted orbital integrals

6.1. The case of (G, H). For an open compact subgroup r of $\mathfrak{s}(F)$, denote by $\mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r)$ the subspace of $\mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ consisting of the functions invariant by translation of r. Let $i_r^{\mathfrak{s}^*} : \mathcal{C}_c^{\infty}(\mathfrak{s}(F))^* \to \mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r)^*$ be the dual map of the natural injection $\mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r) \hookrightarrow \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$.

For any set σ , denote by $\mathbb{C}[\sigma]$ the \mathbb{C} -linear space of maps from σ to \mathbb{C} with finite support. For $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\sigma \subseteq (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we define the linear map

$$\begin{split} \delta_M^G &: \mathbb{C}[\sigma] \to \mathcal{C}_c^\infty(\mathfrak{s}(F))^*, \\ & (z_X)_{X \in \sigma} \mapsto \sum_{X \in \sigma} z_X J_M^G(\eta, X, \cdot) \end{split}$$

where $z_X \in \mathbb{C}$ is the coordinate at $X \in \sigma$.

Once again, one may extend in the obvious way the definitions of $i_r^{\mathfrak{s}*}$ and δ_M^G to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \operatorname{Ad}(\epsilon))$ in lower dimensions.

PROPOSITION 6.1 (Howe's finiteness). Let r be an open compact subgroup of $\mathfrak{s}(F)$, $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\sigma \subseteq (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Suppose that there exists a compact subset $\sigma_0 \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ such that $\sigma \subseteq \mathrm{Ad}(M_H(F))(\sigma_0)$. Then the image of the linear map

$$i_r^{\mathfrak{s}^*} \circ \delta_M^G : \mathbb{C}[\sigma] \to \mathcal{C}_c^\infty(\mathfrak{s}(F)/r)^*$$

is of finite dimension.

REMARK 6.2. For M = G, Proposition 6.1 is essentially included in a more general result [45, Theorem 6.1] in the context of θ -groups (in the sense of [49, p. 467]).

The rest of this section is devoted to the proof of Proposition 6.1. We shall follow the main steps in [51, §IV.2-6]. We may and shall suppose that σ is relatively compact in $(\mathfrak{m} \cap \mathfrak{s})(F)$. The proposition will be proved by induction on the dimension of G.

Recall that we have chosen the standard maximal compact subgroup $K = GL_{2n}(\mathcal{O}_D)$ of $G(F) = GL_{2n}(D)$. Let $k = \mathfrak{g}(\mathcal{O}_F) := \mathfrak{gl}_{2n}(\mathcal{O}_D)$, which is an \mathcal{O}_F -lattice in $\mathfrak{g}(F) = \mathfrak{gl}_{2n}(D)$ and is stable under the adjoint action of K. Since $H \in \mathscr{L}^G(M_0)$, we have set $K_H = K \cap H(F) = GL_n(\mathcal{O}_D) \times GL_n(\mathcal{O}_D)$. Let $\mathfrak{h}(\mathcal{O}_F) := k \cap \mathfrak{h}(F)$ and $\mathfrak{s}(\mathcal{O}_F) := k \cap \mathfrak{s}(F)$. Then we see that $k = \mathfrak{h}(\mathcal{O}_F) \oplus \mathfrak{s}(\mathcal{O}_F)$ and that $\mathfrak{s}(\mathcal{O}_F)$ is stable under the adjoint action of K_H . For all $P \in \mathscr{F}^G(M_0)$, we fix $a_P \in A_P(F)$ such that $|\alpha(a_P)|_F < 1, \forall \alpha \in \Delta_F^{\Omega}$.

Recall that we denote by $\mathcal{N}^{\mathfrak{s}}$ the set of nilpotent elements in $\mathfrak{s}(F)$ and fix a uniformiser ϖ of \mathcal{O}_F . Let $X \in \mathcal{N}^{\mathfrak{s}} \cap (k - \varpi k)$. By the Jacobson-Morozov theorem for symmetric spaces (Lemma 3.7), there exists a group homomorphism $\varphi : SL_2(F) \to G(F)$ such that

$$d\varphi \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} = X \text{ and } a^{\varphi} := \varphi \begin{pmatrix} \varpi & \\ & \varpi^{-1} \end{pmatrix} \in H(F).$$

We define the parabolic subgroup P^X of G as in [51, §IV.3]. More concretely, set

$$\mathfrak{g}[i] := \{ Y \in \mathfrak{g} : \mathrm{Ad}(a^{\varphi})(Y) = \varpi^i Y, \forall i \in \mathbb{Z} \}$$

and

$$\mathfrak{p}^X := \bigoplus_{i \ge 0} \mathfrak{g}[i];$$

then let

$$(6.1.1) P^X := \{ x \in G : \operatorname{Ad}(x)(\mathfrak{p}^X) = \mathfrak{p}^X \}.$$

Note that P^X is independent of the choice of φ by [11, Proposition 5.7.1]. Since a^{φ} commutes with ϵ , one has $\epsilon \in P^X$. By Lemma 3.9, $P^X \cap H$ is a parabolic subgroup of H. Then there exists an element $x \in K_H$ such that $P' := \operatorname{Ad}(x)(P^X) \in \mathscr{F}^G(M_0)$. We shall fix such an x. Let $a^X := \operatorname{Ad}(x^{-1})(a_{P'}) \in H(F)$. Note that a^X depends on the choice of x, but this is unimportant. By [51, (3) in §IV.3], we have

(6.1.2)
$$\operatorname{Ad}(a^X)(X) \in (\varpi k) \cap \mathfrak{s}(F) = \varpi \mathfrak{s}(\mathcal{O}_F)$$

LEMMA 6.3. There exists an integer $h \in \mathbb{N}$ such that for all $Y \in \varpi^h \mathfrak{s}(\mathcal{O}_F)$, all integer $l \geq h$ and all $Z \in \operatorname{Ad}(a^X)^{-1}(\varpi^l \mathfrak{s}(\mathcal{O}_F))$, there exists $\gamma \in K_H$ with $\eta(\operatorname{Nrd}(\gamma)) = 1$ such that

$$\operatorname{Ad}(\gamma)(X+Y+Z) \in X+Y+\varpi^{l}\mathfrak{s}(\mathcal{O}_{F})$$

PROOF. We may choose h following the proof of [51, Lemme IV.3]. The point is that we require $\gamma \in H(F)$ and $\eta(\operatorname{Nrd}(\gamma)) = 1$ here.

Set

$$\mathfrak{n}^X:=\bigoplus_{i\geq 1}\mathfrak{g}[i]$$

Since a^{φ} commutes with ϵ , one has $\mathfrak{g}[i] = (\mathfrak{g}[i] \cap \mathfrak{h}) \oplus (\mathfrak{g}[i] \cap \mathfrak{s})$.

By the intersection of $\mathfrak{s}(\mathcal{O}_F)$ with [51, (4) in §IV.3], we have

(6.1.3)
$$\operatorname{Ad}(a^X)^{-1}(\mathfrak{s}(\mathcal{O}_F)) \subseteq \mathfrak{s}(\mathcal{O}_F) + \operatorname{Ad}(a^X)^{-1}(\mathfrak{s}(\mathcal{O}_F)) \cap \mathfrak{n}^X.$$

Let $c, c', c'', h \in \mathbb{N}$ be as in the proof of [51, Lemme IV.3], except that we require additionally $\exp(\varpi^l k) \subseteq \{x \in K : \eta(\operatorname{Nrd}(x)) = 1\}, \forall l \geq c'$. Then

(6.1.4)
$$\mathfrak{n}^X \cap \operatorname{Ad}(a^X)^{-1}(\mathfrak{s}(\mathcal{O}_F)) \subseteq \operatorname{ad}(X)(\varpi^{-c}\mathfrak{h}(\mathcal{O}_F))$$

Let Y, l and Z be as in the statement. Thanks to (6.1.3), we can write $Z = Z_1 + Z_2$, where $Z_1 \in \varpi^l \mathfrak{s}(\mathcal{O}_F)$ and $Z_2 \in \operatorname{Ad}(a^X)^{-1}(\varpi^l \mathfrak{s}(\mathcal{O}_F)) \cap \mathfrak{n}^X$. Because of (6.1.4), we can choose $Z' \in \varpi^{l-c}\mathfrak{h}(\mathcal{O}_F)$ such that $Z_2 = \operatorname{ad}(X)(Z')$. Since $Z' \in \varpi^{c'}\mathfrak{s}(\mathcal{O}_F)$ from the hypothesis $l \ge h \ge c + c'$, we can define $\gamma := \exp(Z')$ as in the proof of [51, Lemme IV.3]. Then we have $\gamma \in K_H$ and $\eta(\operatorname{Nrd}(\gamma)) = 1$. Such a γ verifies [51, Lemme IV.3] and thus our statement.

For $X \in \mathcal{N}^{\mathfrak{s}} \cap (k - \varpi k)$, we shall fix an integer h^X such that

- (1) h^X verifies the condition of Lemma 6.3;
- $(2) h^X \ge 1;$
- (3) $\operatorname{Ad}(a^X)(\varpi^{h^X}\mathfrak{s}(\mathcal{O}_F)) \subseteq \varpi\mathfrak{s}(\mathcal{O}_F).$

Denote by \mathcal{N}^G the set of nilpotent elements in $\mathfrak{g}(F)$. Let $\mathbb{P}\mathfrak{g}(F)$ (resp. $\mathbb{P}\mathfrak{s}(F)$) be the projective space associated to $\mathfrak{g}(F)$ (resp. $\mathfrak{s}(F)$) and $\pi : \mathfrak{g}(F) - \{0\} \to \mathbb{P}\mathfrak{g}(F)$ the natural projection. Since $\pi(\mathcal{N}^G - \{0\})$ is compact and $\mathcal{N}^{\mathfrak{s}} - \{0\}$ is a closed subset of $\mathcal{N}^G - \{0\}$, we know that $\pi(\mathcal{N}^{\mathfrak{s}} - \{0\})$ is compact. One also sees that

$$\pi(\mathcal{N}^{\mathfrak{s}} \cap (k - \varpi k)) = \pi(\mathcal{N}^{\mathfrak{s}} - \{0\}).$$

We can and shall choose a finite set $\mathcal{N}_0 \subseteq \mathcal{N}^{\mathfrak{s}} \cap (k - \varpi k)$ such that

$$\bigcup_{X\in\mathcal{N}_0}\pi(X+\varpi^{h^X}\mathfrak{s}(\mathcal{O}_F))$$

is an open neighbourhood of $\pi(\mathcal{N}^{\mathfrak{s}} - \{0\}) \subseteq \mathbb{P}\mathfrak{s}(F)$.

LEMMA 6.4. There exists an integer $c \in \mathbb{N}$ such that for all $d \in \delta_M^G(\mathbb{C}[\sigma])$ and all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ satisfying $d(f) \neq 0$, we have

$$\operatorname{Supp}(f) \cap [\varpi^{-c}\mathfrak{s}(\mathcal{O}_F) \cup \bigcup_{X \in \mathcal{N}_0} F^{\times}(X + \varpi^{h^X}\mathfrak{s}(\mathcal{O}_F))] \neq \emptyset.$$

PROOF. Recall that σ is assumed to be relatively compact in $(\mathfrak{m} \cap \mathfrak{s})(F)$. Fix an open compact neighbourhood σ' of σ in $\mathfrak{s}(F)$. Fix $P_0 \in \mathscr{P}^H(M_0)$ and set

$$A_{P_0}^+ := \{ a \in A_{M_0}(F) : |\alpha(a)|_F \ge 1, \forall \alpha \in \Delta_{P_0}^H \}.$$

Similarly, for all $B \in \mathscr{P}^G(M_0)$, set

$$A_B^+ := \{ a \in A_{M_0}(F) : |\alpha(a)|_F \ge 1, \forall \alpha \in \Delta_B^G \}$$

We see from the argument of Lemma 4.8 in Chapter 2 that

$$A_{P_0}^+ \subseteq \bigcup_{\{B \in \mathscr{P}^G(M_0): P_0 \subseteq B\}} A_B^+.$$

By the Cartan decomposition, there exists a compact subset $\Gamma \subseteq H(F)$ such that $H(F) = K_H A_{P_0}^+ \Gamma$. Fix such a Γ . Then

$$H(F) \subseteq \bigcup_{\{B \in \mathscr{P}^G(M_0): P_0 \subseteq B\}} K_H A_B^+ \Gamma.$$

Fix $c' \in \mathbb{N}$ such that $\operatorname{Ad}(\Gamma)(\sigma') \subseteq \varpi^{-c'}\mathfrak{s}(\mathcal{O}_F)$. Since $\operatorname{Ad}(A_B^+)(\mathfrak{s}(\mathcal{O}_F)) \subseteq \mathfrak{s}(\mathcal{O}_F) + (\mathfrak{n}_B \cap \mathfrak{s})(F)$ and $(\mathfrak{n}_B \cap \mathfrak{s})(F) \subseteq \mathcal{N}^{\mathfrak{s}}$, we obtain

$$\operatorname{Ad}(H(F))(\sigma') \subseteq \varpi^{-c'}\mathfrak{s}(\mathcal{O}_F) + \mathcal{N}^{\mathfrak{s}}$$

Choose an integer c such that

$$c \ge c' + \sup_{X \in \mathcal{N}_0} h^X.$$

Let d and f be as in the statement. It is evident that

 $\operatorname{Supp}(f) \cap \operatorname{Ad}(H(F))(\sigma') \neq \emptyset.$

Suppose that $X_1 \in \varpi^{-c'} \mathfrak{s}(\mathcal{O}_F), X_2 \in \mathcal{N}^{\mathfrak{s}}$ satisfy $X_1 + X_2 \in \operatorname{Supp}(f)$. If $X_2 \in \varpi^{-c} \mathfrak{s}(\mathcal{O}_F)$, we are done. If not, let $X \in \mathcal{N}_0$ be such that $\pi(X_2) \in \pi(X + \varpi^{h^X} \mathfrak{s}(\mathcal{O}_F))$. Then there exists $\lambda \in F^{\times}$ such that $X_2 \in \lambda(X + \varpi^{h^X}\mathfrak{s}(\mathcal{O}_F))$. Since $X_2 \notin \varpi^{-c}\mathfrak{s}(\mathcal{O}_F)$, we have $v_F(\lambda) < -c$ and thus $v_F(\lambda) + h^X < -c'$. Therefore, $X_1 \in \lambda \varpi^{h^X}\mathfrak{s}(\mathcal{O}_F)$ and then $X_1 + X_2 \in \lambda(X + \varpi^{h^X}\mathfrak{s}(\mathcal{O}_F)) \subseteq F^{\times}(X + \varpi^{h^X}\mathfrak{s}(\mathcal{O}_F))$. This is exactly what we want to prove. \square

We shall fix an integer c verifying the condition in Lemma 6.4. We shall also fix another integer hsuch that

$$(6.1.5) h \ge \sup_{X \in \mathcal{N}_0} h^X$$

Denote

$$C := \{ f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})) : \operatorname{Supp}(f) \subseteq \varpi^{-c}\mathfrak{s}(\mathcal{O}_{F}) \},$$

which is a \mathbb{C} -linear space of finite dimension. For all $L \in \mathscr{L}^{G}(M), L \neq G$, let $r(L) := \varpi^{h}(\mathfrak{l} \cap \mathfrak{s})(\mathcal{O}_{F}).$

LEMMA 6.5. Let $\mathfrak{z} = (z_X)_{X \in \sigma} \in \mathbb{C}[\sigma]$. Suppose that 1) $i_{r(L)}^{\mathfrak{l}\cap\mathfrak{s}} \circ \delta_{M}^{L}(\mathfrak{z}) = 0$ for all $L \in \mathscr{L}^{G}(M), L \neq G;$ 2) $\delta^{G}_{M}(\mathfrak{z})(C) = 0.$ Then $i^{\mathfrak{s}}_{\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})}^{*} \circ \delta^{G}_{M}(\mathfrak{z}) = 0.$

PROOF. Write $d := \delta_M^G(\mathfrak{z})$. It suffices to prove by induction on the integer $e \ge c$ the assertion $(A)_e$: for all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F)/\varpi^h \mathfrak{s}(\mathcal{O}_F))$ with $\operatorname{Supp}(f) \subseteq \varpi^{-e} \mathfrak{s}(\mathcal{O}_F)$, we have d(f) = 0.

If e = c, this is true by the hypothesis 2). Fix an e > c and suppose that $(A)_{e-1}$ is true. For all open compact subset $s \subseteq \mathfrak{g}(F)$, denote by 1_s its characteristic function. It suffices to prove that for all $Y \in \overline{\omega}^{-e} \mathfrak{s}(\mathcal{O}_F) - \overline{\omega}^{-e+1} \mathfrak{s}(\mathcal{O}_F), \text{ we have } d(1_{Y+\overline{\omega}^h \mathfrak{s}(\mathcal{O}_F)}) = 0.$

Suppose that $Y \in \varpi^{-e}\mathfrak{s}(\mathcal{O}_F) - \varpi^{-e+1}\mathfrak{s}(\mathcal{O}_F)$. The hypothesis e > c implies $\varpi^{-c}\mathfrak{s}(\mathcal{O}_F) \subseteq \varpi^{-e+1}\mathfrak{s}(\mathcal{O}_F)$ and thus $Y \notin \varpi^{-c} \mathfrak{s}(\mathcal{O}_F)$. Suppose on the contary that $d(1_{Y+\varpi^h \mathfrak{s}(\mathcal{O}_F)}) \neq 0$. By Lemma 6.4, there exists $X \in \mathcal{N}_0$ and $\lambda \in F^{\times}$ such that $(Y + \varpi^h \mathfrak{s}(\mathcal{O}_F)) \cap \lambda(X + \varpi^{h^X} \mathfrak{s}(\mathcal{O}_F)) \neq \emptyset$. Fix such X and λ . Since $v_F(Y) = -e$ and $v_F(X) = 0$, we have $v_F(\lambda) = -e$. As $h \ge h^X \ge h^X - e$ (see (6.1.5)), we obtain $\varpi^{h}\mathfrak{s}(\mathcal{O}_{F}) \subseteq \lambda \varpi^{h^{X}}\mathfrak{s}(\mathcal{O}_{F}) \text{ and then } Y \in \lambda(X + \varpi^{h^{X}}\mathfrak{s}(\mathcal{O}_{F})). \text{ Let } Y' \in \varpi^{h^{X}}\mathfrak{s}(\mathcal{O}_{F}) \text{ such that } Y = \lambda(X + Y').$ Let $Z \in \operatorname{Ad}(a^{\hat{X}})^{-1}(\varpi^h \mathfrak{s}(\mathcal{O}_F))$. Since $h + e \ge h^{\hat{X}}$, we can apply Lemma 6.3 to $X, Y', \lambda^{-1}Z$ and l := h + e. Then there exists $\gamma \in K_H$ with $\eta(\operatorname{Nrd}(\gamma)) = 1$ such that

Ad(

d

$$\operatorname{Ad}(\gamma)(X+Y'+\lambda^{-1}Z) \in X+Y'+\varpi^{h+e}\mathfrak{s}(\mathcal{O}_F).$$

From $v_F(\lambda) = -e$, we deduce that

$$\gamma)(\lambda(X+Y')+Z) \in \lambda(X+Y') + \varpi^h \mathfrak{s}(\mathcal{O}_F)$$

i.e.,

$$\operatorname{Ad}(\gamma)(Y+Z) \in Y + \varpi^h \mathfrak{s}(\mathcal{O}_F).$$

Since $\gamma \in K_H$, this is equivalent to

$$\operatorname{Ad}(\gamma)(Y + Z + \varpi^h \mathfrak{s}(\mathcal{O}_F)) = Y + \varpi^h \mathfrak{s}(\mathcal{O}_F)$$

or

$$\operatorname{Ad}(\gamma)(1_{Y+Z+\varpi^h\mathfrak{s}(\mathcal{O}_F)})=1_{Y+\varpi^h\mathfrak{s}(\mathcal{O}_F)}.$$

By Proposition 4.1.3), we obtain

$$(1_{Y+Z+\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})})=\eta(\mathrm{Nrd}(\gamma))d(1_{Y+\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})}).$$

Because $\eta(\operatorname{Nrd}(\gamma)) = 1$, we have

$$d(1_{Y+Z+\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})}) = d(1_{Y+\varpi^{h}\mathfrak{s}(\mathcal{O}_{F})}).$$

Now, by the sum over $Z \in (\operatorname{Ad}(a^X)^{-1}(\varpi^h \mathfrak{s}(\mathcal{O}_F)) + \varpi^h \mathfrak{s}(\mathcal{O}_F)) / \varpi^h \mathfrak{s}(\mathcal{O}_F)$ (a finite set), we get $d(1_{Y+\varpi^h\mathfrak{s}(\mathcal{O}_F)}) = [k':\mathfrak{s}(\mathcal{O}_F)]^{-1}d(1_{Y+\varpi^hk'}),$ (6.1.6)

where

(6.1.7)
$$k' := \operatorname{Ad}(a^X)^{-1}(\mathfrak{s}(\mathcal{O}_F)) + \mathfrak{s}(\mathcal{O}_F).$$

By Proposition 4.1.6), we have the equality

$$d\left(\operatorname{Ad}(a^X)(1_{Y+\varpi^hk'}) - \eta(\operatorname{Nrd}(a^X))1_{Y+\varpi^hk'}\right) = \eta(\operatorname{Nrd}(a^X))\sum_{Q\in\mathscr{F}^G(M), Q\neq G}\delta_M^{M_Q}(\mathfrak{z})\left((1_{Y+\varpi^hk'})_{Q,(a^X)^{-1}}^\eta\right).$$

From (4.1.3), it is clear that $(1_{Y+\varpi^h k'})^{\eta}_{Q,(a^X)^{-1}}$ is invariant by translation of $r(M_Q)$. By the hypothesis 1), we have $\delta_M^{M_Q}(\mathfrak{z})\left((1_{Y+\varpi^h k'})^{\eta}_{Q,(a^X)^{-1}}\right) = 0$ for all $Q \in \mathscr{F}^G(M), Q \neq G$, so

(6.1.8)
$$\eta(\operatorname{Nrd}(a^X))d(1_{Y+\varpi^h k'}) = d\left(\operatorname{Ad}(a^X)(1_{Y+\varpi^h k'})\right).$$

We see easily that

$$\operatorname{Ad}(a^X)(1_{Y+\varpi^h k'}) = 1_s,$$

where

(6.1.9)
$$s := \operatorname{Ad}(a^X)(Y) + \varpi^h \operatorname{Ad}(a^X)(k')$$

Recall $Y = \lambda(X + Y')$ above. As $X \in \mathcal{N}_0$, by (6.1.2), we have $\operatorname{Ad}(a^X)(X) \in \varpi \mathfrak{s}(\mathcal{O}_F)$. Since $Y' \in \varpi^{h^X} \mathfrak{s}(\mathcal{O}_F)$, by the hypothesis (3) on h^X , we have $\operatorname{Ad}(a^X)(Y') \in \varpi \mathfrak{s}(\mathcal{O}_F)$. For $v_F(\lambda) = -e$, we obtain

(6.1.10)
$$\operatorname{Ad}(a^X)(Y) \in \varpi^{-e+1}\mathfrak{s}(\mathcal{O}_F).$$

We see from (6.1.7) that

$$\operatorname{Ad}(a^X)(k') = \mathfrak{s}(\mathcal{O}_F) + \operatorname{Ad}(a^X)(\mathfrak{s}(\mathcal{O}_F)).$$

By (6.1.5) and the hypothesis (3) on h^X , we have $\varpi^h \operatorname{Ad}(a^X)(\mathfrak{s}(\mathcal{O}_F)) \subseteq \varpi^{h^X} \operatorname{Ad}(a^X)(\mathfrak{s}(\mathcal{O}_F)) \subseteq \varpi \mathfrak{s}(\mathcal{O}_F)$. Then by (6.1.5) and the hypothesis (2) on h^X , we have

(6.1.11)
$$\varpi^h \operatorname{Ad}(a^X)(k') = \varpi^h \mathfrak{s}(\mathcal{O}_F) + \varpi^h \operatorname{Ad}(a^X)(\mathfrak{s}(\mathcal{O}_F)) \subseteq \varpi^{h^X} \mathfrak{s}(\mathcal{O}_F) + \varpi \mathfrak{s}(\mathcal{O}_F) = \varpi \mathfrak{s}(\mathcal{O}_F).$$

From (6.1.9), (6.1.10) and (6.1.11), we see that

$$\operatorname{Supp}(1_s) \subseteq \varpi^{-e+1} \mathfrak{s}(\mathcal{O}_F).$$

Since $\mathfrak{s}(\mathcal{O}_F) \subseteq \operatorname{Ad}(a^X)(k')$, we know that 1_s is invariant by translation of $\varpi^h \mathfrak{s}(\mathcal{O}_F)$. Using the induction hypothesis $(A)_{e-1}$, we have

$$d(1_s) = 0$$

Thanks to (6.1.6) and (6.1.8), we obtain

$$d(1_{Y+\varpi^h\mathfrak{s}(\mathcal{O}_F)})=0.$$

This proves $(A)_e$ and thus the lemma.

PROOF OF PROPOSITION 6.1. We use induction on the dimension of G. Suppose that for all $L \in \mathscr{L}^G(M), L \neq G$, and all open compact subgroup r_L of $(\mathfrak{l} \cap \mathfrak{s})(F)$, the image of the linear map

$$i_{r_L}^{\mathfrak{l}\cap\mathfrak{s}^*}\circ\delta_M^L:\mathbb{C}[\sigma]\to\mathcal{C}_c^\infty((\mathfrak{l}\cap\mathfrak{s})(F)/r_L)^*$$

is of finite dimension. This is actually a product form of the proposition in lower dimensions. Now we would like to prove the proposition. The argument below is also valid for the case G = M.

Enlarge h in (6.1.5) if necessary such that $r \supseteq \varpi^h \mathfrak{s}(\mathcal{O}_F)$. We shall prove that the image of $i_{\varpi^h \mathfrak{s}(\mathcal{O}_F)}^{\mathfrak{s}} \circ \delta_M^G$ is of finite dimension. Admit this for the moment. Since $i_r^{\mathfrak{s}^*}$ factorises by $i_{\varpi^h \mathfrak{s}(\mathcal{O}_F)}^{\mathfrak{s}_h}^{\mathfrak{s}}$, the image of $i_r^{\mathfrak{s}^*} \circ \delta_M^G$ is also of finite dimension. Then we finish the proof.

Let \mathcal{K}_1 be the kernel of the linear map

I

$$\bigoplus_{\boldsymbol{\nu}\in\mathscr{L}^G(M), L\neq G} i_{r(L)}^{\mathfrak{l}\cap\mathfrak{s}}{}^*\circ\delta_M^L:\mathbb{C}[\sigma]\rightarrow \bigoplus_{L\in\mathscr{L}^G(M), L\neq G} \mathcal{C}_c^\infty((\mathfrak{l}\cap\mathfrak{s})(F)/r(L))^*$$

whose image is of finite dimension by our induction hypothesis applied to $r_L := r(L)$ for all $L \in \mathscr{L}^G(M), L \neq G$. Hence, to prove that $i_{\varpi^h \mathfrak{s}(\mathcal{O}_F)}^{\mathfrak{s}} \circ \delta_M^G(\mathbb{C}[\sigma])$ is of finite dimension, it suffices to prove that $i_{\varpi^h \mathfrak{s}(\mathcal{O}_F)}^{\mathfrak{s}} * \circ \delta_M^G(\mathcal{K}_1)$ is of finite dimension.

Consider the composition of the linear maps

$$d_1 := i_{\varpi^h \mathfrak{s}(\mathcal{O}_F)}^{\mathfrak{s}} \circ \delta_M^G \big|_{\mathcal{K}_1} : \mathcal{K}_1 \to \mathcal{C}_c^{\infty}(\mathfrak{s}(F)/\varpi^h \mathfrak{s}(\mathcal{O}_F))^*$$

and

Res:
$$\mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)/\varpi^{h}\mathfrak{s}(\mathcal{O}_{F}))^{*} \to C^{*}$$

The latter map is the natural restriction. Lemma 6.5 says that

$$\ker(\operatorname{Res} \circ d_1) = \ker(d_1)$$

which is denoted by \mathcal{K}_2 . Then

$$d_1(\mathcal{K}_1) \simeq \mathcal{K}_1/\mathcal{K}_2 \simeq \operatorname{Res} \circ d_1(\mathcal{K}_1) \subseteq C^*$$

Since C is of finite dimension, we see that $d_1(\mathcal{K}_1)$ is of finite dimension.

COROLLARY 6.6. Let r be an open compact subgroup of $\mathfrak{s}(F)$, $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\sigma \subseteq (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Suppose that there exists a compact subset $\sigma_0 \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ such that $\sigma \subseteq \operatorname{Ad}(M_H(F))(\sigma_0)$. Then there exists a finite subset $\{X_i : i \in I\} \subseteq \sigma$ and a finite subset $\{f_i : i \in I\} \subseteq \mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r)$ such that for all $X \in \sigma$ and all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r)$, we have the equality

$$J_M^G(\eta, X, f) = \sum_{i \in I} J_M^G(\eta, X_i, f) J_M^G(\eta, X, f_i)$$

PROOF. By Proposition 6.1, there exists a finite subset $\{X_i : i \in I\} \subseteq \sigma$ such that $\{i_r^{\mathfrak{s}^*} \circ \delta_M^G(X_i) : i \in I\}$ is a basis of $i_r^{\mathfrak{s}^*} \circ \delta_M^G(\mathbb{C}[\sigma])$. By linear algebra, there exists a finite subset $\{f_i : i \in I\} \subseteq \mathcal{C}_c^{\infty}(\mathfrak{s}(F)/r)$ such that $i_r^{\mathfrak{s}^*} \circ \delta_M^G(X_i)(f_j) = \delta_{ij}, \forall i, j \in I$, where δ_{ij} denotes the Kronecker delta function. Choose such $\{X_i : i \in I\}$ and $\{f_i : i \in I\}$.

Then, for all $X \in \sigma$, there exists $\lambda_i, \forall i \in I$ such that

$$J_M^G(\eta, X, \cdot) = \sum_{i \in I} \lambda_i J_M^G(\eta, X_i, \cdot) \in \mathcal{C}_c^\infty(\mathfrak{s}(F)/r)^*.$$

Hence, for all $i \in I$,

$$J_M^G(\eta, X, f_i) = \sum_{j \in I} \lambda_j J_M^G(\eta, X_j, f_i) = \sum_{j \in I} \lambda_j \delta_{ji} = \lambda_i.$$
of.

We have finished the proof.

6.2. The case of (G', H'). For an open compact subgroup r' of $\mathfrak{s}'(F)$, denote by $\mathcal{C}_c^{\infty}(\mathfrak{s}'(F)/r')$ the subspace of $\mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$ consisting of the functions invariant by translation of r'. Let $i_{r'}^{\mathfrak{s}'*} : \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))^* \to \mathcal{C}_c^{\infty}(\mathfrak{s}'(F)/r')^*$ be the dual map of the natural injection $\mathcal{C}_c^{\infty}(\mathfrak{s}'(F)/r') \hookrightarrow \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$.

For any set σ' , denote by $\mathbb{C}[\sigma']$ the \mathbb{C} -linear space with a basis σ' . For $M' \in \mathscr{L}^{H'}(M'_0)$ and $\sigma' \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, we define the linear map

$$\delta_{M'}^{H'}: \mathbb{C}[\sigma'] \to \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))^*,$$
$$(z_Y)_{Y \in \sigma'} \mapsto \sum_{Y \in \sigma'} z_Y J_{M'}^{H'}(Y, \cdot)$$

where $z_Y \in \mathbb{C}$ is the coordinate at $Y \in \sigma'$.

One may extend in the obvious way the definitions of $i_{r'}^{\mathfrak{s}'^*}$ and $\delta_{M'}^{H'}$ to the symmetric pair $(\widetilde{M'}, M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$.

PROPOSITION 6.7 (Howe's finiteness). Let r' be an open compact subgroup of $\mathfrak{s}'(F)$, $M' \in \mathscr{L}^{H'}(M'_0)$ and $\sigma' \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Suppose that there exists a compact subset $\sigma'_0 \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(F)$ such that $\sigma' \subseteq \operatorname{Ad}(M'(F))(\sigma'_0)$. Then the image of the linear map

$$i_{r'}^{\mathfrak{s}'^*} \circ \delta_{M'}^{H'} : \mathbb{C}[\sigma'] \to \mathcal{C}_c^{\infty}(\mathfrak{s}'(F)/r')^*$$

is of finite dimension.

The rest of this section is devoted to the proof of Proposition 6.7. It is similar to the proof of Proposition 6.1 and we only point out some additional argument. Recall that we have chosen the standard maximal compact subgroup $K_{H'} = GL_n(\mathcal{O}_{D'})$ of $H'(F) = GL_n(D')$ in **Case I** (resp. $K_{H'} = GL_{\frac{n}{2}}(\mathcal{O}_{D\otimes_F E})$ of $H'(F) = GL_{\frac{n}{2}}(D\otimes_F E)$ in **Case II**). Snice $\operatorname{Ad}(\tau)(D') = D'$ in **Case I** (resp. $\operatorname{Ad}(\tau)(D\otimes_F E) = D\otimes_F E$ in **Case II**), we deduce that $\operatorname{Ad}(\tau)(\mathcal{O}_{D'}) = \mathcal{O}_{D'}$ in **Case I** (resp. $\operatorname{Ad}(\tau)(\mathcal{O}_{D\otimes_F E}) = \mathcal{O}_{D\otimes_F E}$ in **Case II**). Thus $\operatorname{Ad}(\tau)(K_{H'}) = K_{H'}$. Let $\mathfrak{h}'(\mathcal{O}_F) := \mathfrak{gl}_n(\mathcal{O}_{D'})$ in **Case I** (resp. $\mathfrak{h}'(\mathcal{O}_F) := \mathfrak{gl}_{\frac{n}{2}}(\mathcal{O}_{D\otimes_F E})$ in **Case II**). Let $\mathfrak{s}'(\mathcal{O}_F) := \mathfrak{h}'(\mathcal{O}_F)\tau = \tau \mathfrak{h}'(\mathcal{O}_F)$ be an \mathcal{O}_F -lattice in $\mathfrak{s}'(F)$ (see Section 3.3 for the choice of

au). Let $k' := \mathfrak{h}'(\mathcal{O}_F) \oplus \mathfrak{s}'(\mathcal{O}_F)$, whose decomposition is stable under the adjoint action of $K_{H'}$ because $\operatorname{Ad}(\tau)(K_{H'}) = K_{H'}$. For all $P' \in \mathscr{F}^{H'}(M'_0)$, we fix $a_{P'} \in A_{P'}(F)$ such that $|\alpha(a_{P'})|_F < 1, \forall \alpha \in \Delta_{P'}^{H'}$. Starting from $X \in \mathcal{N}^{\mathfrak{s}'} \cap (\mathfrak{s}'(\mathcal{O}_F) - \varpi \mathfrak{s}'(\mathcal{O}_F))$, we obtain a group homomorphism $\varphi : SL_2(F) \to G'(F)$

by the Jacobson-Morozov theorem for symmetric spaces (Lemma 3.7). Denote $a^{\varphi} := \varphi \begin{pmatrix} \varpi \\ \varpi^{-1} \end{pmatrix} \in$

H'(F). Define P^X as in (6.1.1), which contains the centraliser $\operatorname{Cent}_{G'}(a^{\varphi})$ of a^{φ} in G' as a Levi factor. Since a^{φ} commutes with α , by Lemma 3.16, $P^X \cap H'$ is a parabolic subgroup of H', which contains the centraliser $\operatorname{Cent}_{H'}(a^{\varphi})$ of a^{φ} in H' as a Levi factor. We want to show that there exists $x \in K_{H'}$ such that $\operatorname{Ad}(x)(P^X \cap H') \in \mathscr{F}^{H'}(M'_0)$ and that $\operatorname{Ad}(x)(P^X) \in \mathscr{F}^{G'}(M'_{\widetilde{0}})$.

LEMMA 6.8. For $Y \in \mathcal{N}^{\mathfrak{s}'}$, there exists $x \in H'(F)$ such that $\operatorname{Ad}(x)(Y)$ is in the Jordan normal form, i.e., diagonal block matrices with entries in $D'\tau$ in **Case I** (resp. $(D \otimes_F E)\tau$ in **Case II**) whose blocks are of the form



PROOF. It can be proved in the same way as [24, Lemmas 2.2 and 2.3] by linear algebra over a division ring. $\hfill \Box$

Thanks to Lemma 6.8, we can construct explicitly the above morphism φ (see [19, p. 184]). If X is in the Jordan normal form, by *loc. cit.*, we may choose $a^{\varphi} \in A_{L'}(F)$ for some $L' \in \mathscr{L}^{H'}(M'_0)$ such that $\operatorname{Cent}_{H'}(a^{\varphi}) = L'$ and that $\operatorname{Cent}_{G'}(a^{\varphi}) = \widetilde{L'}$. For a general X as above, by Lemma 6.8, there exists $y \in H'(F)$ such that $\operatorname{Ad}(y)(a^{\varphi}) \in A_{L'}(F)$ for some $L' \in \mathscr{L}^{H'}(M'_0)$ satisfying $\operatorname{Cent}_{H'}(\operatorname{Ad}(y)(a^{\varphi})) = L'$ and $\operatorname{Cent}_{G'}(\operatorname{Ad}(y)(a^{\varphi})) = \widetilde{L'}$. Let $x \in K_{H'}$ be such that $x^{-1}y \in (P^X \cap H')(F)$. Then $\operatorname{Ad}(x)(P^X \cap H') = \operatorname{Ad}(y)(P^X \cap H')$ contains L' as a Levi factor and $\operatorname{Ad}(x)(P^X) = \operatorname{Ad}(y)(P^X \cap H')$ contains $\widetilde{L'}$ as a Levi factor. Furthermore, since $\operatorname{Ad}(x)(P^X) \cap H' = \operatorname{Ad}(x)(P^X \cap H')$, we see that $\operatorname{Ad}(x)(P^X \cap H')$ and $\operatorname{Ad}(x)(P^X)$ are associated under the bijection $P' \mapsto \widetilde{P'}$ between $\mathscr{F}^{H'}(M'_0)$ and $\mathscr{F}^{G'}(M'_0)$.

Fix $x \in K_{H'}$ as above and denote $P' := \operatorname{Ad}(x)(P^X \cap H') \in \mathscr{F}^{H'}(M'_0)$. Then $\widetilde{P'} = \operatorname{Ad}(x)(P^X)$. Put $a^X := \operatorname{Ad}(x^{-1})(a_{P'}) \in H'(F)$. By the argument of [51, (3) in §IV.3], we show that

(6.2.1)
$$\operatorname{Ad}(a^X)(X) \in \varpi \mathfrak{s}'(\mathcal{O}_F).$$

PROOF OF PROPOSITION 6.7. We may apply the argument of Proposition 6.7 with obvious modifications. Especially, one needs to use Proposition 4.4 and (6.2.1) to show an analogue of Lemma 6.5 for the case of (G', H'). Additionally, to prove an analogue of Lemma 6.3 for this case, one may resort to the argument rather than the consequence of some steps in the proof of [51, Lemme IV.3] since our definition of k' is different from $\mathfrak{g}'(\mathcal{O}_F) := \mathfrak{gl}_n(\mathcal{O}_D)$. However, there is no essential difficulty with our preparation above and we omit details here.

COROLLARY 6.9 (cf. Corollary 6.6). Let r' be an open compact subgroup of $\mathfrak{s}'(F)$, $M' \in \mathscr{L}^{H'}(M'_0)$ and $\sigma' \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Suppose that there exists a compact subset $\sigma'_0 \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(F)$ such that $\sigma' \subseteq \mathrm{Ad}(M'(F))(\sigma'_0)$. Then there exists a finite subset $\{Y_i : i \in I\} \subseteq \sigma'$ and a finite subset $\{f_i : i \in I\} \subseteq \mathcal{C}^{\infty}_c(\mathfrak{s}'(F)/r')$ such that for all $Y \in \sigma'$ and all $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F)/r')$, we have the equality

$$J_{M'}^{H'}(Y,f') = \sum_{i \in I} J_{M'}^{H'}(Y_i,f') J_{M'}^{H'}(Y,f'_i)$$

7. Representability of the Fourier transform of weighted orbital integrals

7.1. The case of (G, H). Following [51, \S V.6], we denote by $\mathscr{E}^{\mathfrak{s}}$ the space of functions $e : \mathfrak{s}_{rs}(F) \to \mathbb{C}$ such that

(1) e is locally constant;

(2) for all open compact subset r of $\mathfrak{s}(F)$, there exists c > 0 and $N \in \mathbb{N}$ such that for all $X \in r \cap \mathfrak{s}_{rs}(F)$, one has the inequality

$$|e(X)| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

If $e \in \mathscr{E}^{\mathfrak{s}}$, the function $X \mapsto |D^{\mathfrak{s}}(X)|_{F}^{-1/2} e(X)$ is locally integrable on $\mathfrak{s}(F)$ thanks to Corollary 3.15. It defines then a distribution on $\mathfrak{s}(F)$:

(7.1.1)
$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{s}(F)} f(X)e(X)|D^{\mathfrak{s}}(X)|_{F}^{-1/2}dX.$$

Denote by $\mathscr{D}^{\mathfrak{s}}$ the space of distributions obtained in this way. The map $\mathscr{E}^{\mathfrak{s}} \to \mathscr{D}^{\mathfrak{s}}$ defined above is an isomorphism. For $d \in \mathscr{D}^{\mathfrak{s}}$, we shall always denote by e_d its preimage in $\mathscr{E}^{\mathfrak{s}}$.

Notice that the notion $\mathscr{E}^{\mathfrak{s}}$ can be defined for any symmetric pair, and that the definition $\mathscr{D}^{\mathfrak{s}}$ can at least be extended to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \operatorname{Ad}(\epsilon))$ in lower dimensions.

If $d \in \mathscr{D}^{\mathfrak{s}}$ is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of H(F), then so is $e_d \in \mathscr{E}^{\mathfrak{s}}$ and by the Weyl integration formula (Proposition 3.11), we have the equality

(7.1.2)
$$d(f) = \sum_{M \in \mathscr{L}^{G,\omega}(M_0)} |W_0^{M_n}| |W_0^{GL_n}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_H, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_G^G(\eta, X, f) e_d(X) dX$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, where $J^{G}_{G}(\eta, X, f)$ is defined by (4.1.1).

REMARK 7.1 (Glueing). Let $d \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))^*$ and $(r_i)_{i \in I}$ be a family of open compact subsets of $\mathfrak{s}(F)$ such that $\bigcup_{i \in I} r_i = \mathfrak{s}(F)$. Suppose that for all $i \in I$, there exists $d_i \in \mathscr{D}^{\mathfrak{s}}$ such that $d(f) = d_i(f)$ for all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ with $\operatorname{Supp}(f) \subseteq r_i$. Then $d \in \mathscr{D}^{\mathfrak{s}}$. Refer to [51, Remarque V.6] for the details.

Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Denote by $\hat{J}_M^G(\eta, X, \cdot)$ the distribution on $\mathfrak{s}(F)$ defined by

$$\hat{J}_M^G(\eta, X, f) := J_M^G(\eta, X, \hat{f})$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, where the right hand side is defined by (4.1.1). We also have a similar definition for the symmetric pair $(M, M_{H}, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_{0})$. The main result of this section is the following.

PROPOSITION 7.2 (Representability). Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Then the distribution $\hat{J}_M^G(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$.

REMARK 7.3. For M = G, Proposition 7.2 is essentially [58, Theorem 6.1.(i)] (see also [57, Theorem 6.2]).

The rest of this section is devoted to the proof of Proposition 7.2. We shall follow the main steps in [51, §V.7-10].

Let \mathfrak{c} be a Cartan subspace of \mathfrak{s} . Recall that $T_{\mathfrak{c}}$ denotes the centraliser of \mathfrak{c} in H. Suppose that $e_0: (T_{\mathfrak{c}}(F) \setminus H(F)) \times \mathfrak{c}_{reg}(F) \to \mathbb{C}$ is a function such that

(1) e_0 is locally constant;

(2) for all open compact subset r of $\mathfrak{s}(F)$, there exists c > 0 and $N \in \mathbb{N}$ such that for all $x \in T_{\mathfrak{c}}(F) \setminus H(F)$ and $X \in \mathfrak{c}_{reg}(F)$ satisfying $\operatorname{Ad}(x^{-1})(X) \in r$, one has the inequality

$$e_0(x,X) \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N.$$

Following [51, §V.7], for $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we define

(7.1.3)
$$d_0(f) := \int_{\mathfrak{c}_{\rm reg}(F)} |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{T_{\mathfrak{c}}(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) e_0(x, X) dx dX$$

LEMMA 7.4. Let \mathfrak{c} be a Cartan subspace of \mathfrak{s} . Suppose that e_0 satisfies the above hypotheses. Then the integral (7.1.3) is absolutely convergent. Moreover, the distribution $d_0 \in \mathscr{D}^{\mathfrak{s}}$.

PROOF. We define a function $e' : \mathfrak{s}_{rs}(F) \to \mathbb{C}$ by

(7.1.4)
$$e'(X) := \sum_{\{x \in T_{\mathfrak{c}}(F) \setminus H(F) : \operatorname{Ad}(x)(X) \in \mathfrak{c}(F)\}} e_0(x, \operatorname{Ad}(x)(X))$$

for all $X \in \mathfrak{s}_{rs}(F)$. If $X \notin Ad(H(F))(\mathfrak{c}_{reg}(F))$, then e'(X) = 0. If $X \in Ad(H(F))(\mathfrak{c}_{reg}(F))$, then the sum in (7.1.4) is actually over the finite set $W(H, \mathfrak{c})y$, where $y \in H(F)$ is any element such that $Ad(y)(X) \in \mathfrak{c}(F)$. Hence, e' is well-defined. Additionally, one may check that $e' \in \mathscr{E}^{\mathfrak{s}}$ from the hypotheses on e_0 . Let $d' \in \mathscr{D}^{\mathfrak{s}}$ be the distribution associated to e' by (7.1.1). For all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$, by the Weyl integration formula (3.1.2), we have

(7.1.5)
$$d'(f) = |W(H, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{reg}(F)} |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{T_{\mathfrak{c}}(F)\setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) e'(\operatorname{Ad}(x^{-1})(X)) dx dX.$$

Since

$$e'(\operatorname{Ad}(x^{-1})(X)) = \sum_{w \in W(H, \mathfrak{c})} e_0(wx, \operatorname{Ad}(w)(X))$$

for all $x \in T_{\mathfrak{c}}(F) \setminus H(F)$ and all $X \in \mathfrak{c}_{reg}(F)$, we deduce that

$$d'(f) = |W(H, \mathfrak{c})|^{-1} \sum_{w \in W(H, \mathfrak{c})} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{T_{\mathfrak{c}}(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(X)) e_{0}(wx, \mathrm{Ad}(w)(X)) dx dX.$$

Applying the change of variables $X \mapsto \operatorname{Ad}(w^{-1})(X)$ and $x \mapsto w^{-1}x$, which does not modify the Haar measures, we obtain

$$d'(f) = d_0(f)$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$. That is to say, $d_{0} = d' \in \mathscr{D}^{\mathfrak{s}}$.

Note that in the argument above, we have used the convergence of an analogue of (7.1.5) with e_0 and f replaced by their absolute values. It also results in the absolute convergence of (7.1.3).

COROLLARY 7.5 (Parabolic induction). Let $M \in \mathscr{L}^{G,\omega}(M_0)$, $P \in \mathscr{P}^G(M)$ and $d \in \mathscr{D}^{\mathfrak{m} \cap \mathfrak{s}}$. Then the distribution on $\mathfrak{s}(F)$ defined by $\forall f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F)), f \mapsto d(f_P^{\eta})$ belongs to $\mathscr{D}^{\mathfrak{s}}$, where f_P^{η} is defined by (3.2.2).

PROOF. Applying the Weyl integration formula (3.1.2) to $d(f_P^{\eta})$, we see that it suffices to fix a Cartan subspace $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ and prove that the distribution on $\mathfrak{s}(F)$ defined by (7.1.6)

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{m} \cap \mathfrak{s}}(X)|_{F}^{1/2} \int_{T_{\mathfrak{c}}(F) \setminus M_{H}(F)} f_{P}^{\eta}(\mathrm{Ad}(x^{-1})(X)) e_{d}(\mathrm{Ad}(x^{-1})(X)) dx dX$$

belongs to $\mathscr{D}^{\mathfrak{s}}$. Recall that $e_d \in \mathscr{E}^{\mathfrak{m} \cap \mathfrak{s}}$ is associated to d by (7.1.1).

Define a function $e_1: H(F) \times \mathfrak{c}_{reg}(F) \to \mathbb{C}$ by

(7.1.7)
$$e_1(x,X) := \eta(\operatorname{Nrd}(x)) \int_{M_H(F)\cap K} e_d(\operatorname{Ad}(m_P(x)k)^{-1}(X))\eta(\operatorname{Nrd}(m_P(x)k))dk$$

for all $x \in H(F)$ and all $X \in \mathfrak{c}_{\mathrm{reg}}(F)$, where $m_P(x) \in M_H(F)$ is any element such that $m_P(x)^{-1}x \in N_{P_H}(F)K_H$. Since $P_H(F) \cap K = (M_H(F) \cap K)(N_{P_H}(F) \cap K)$ (see [5, Property (iii), p. 9] for example), the integral (7.1.7) is independent of the choice of $m_P(x)$. The function e_1 is left $T_{\mathfrak{c}}(F)$ -invariant on the first variable, so it induces a function (still denoted by e_1) : $T_{\mathfrak{c}}(F) \setminus H(F) \times \mathfrak{c}_{\mathrm{reg}}(F) \to \mathbb{C}$.

We shall check that e_1 verifies the hypotheses of Lemma 7.4. Firstly, e_1 is locally constant because e_d is locally constant and e_1 is right K_H -invariant on the first variable. Secondly, suppose that r is an open compact subset of $\mathfrak{s}(F)$. We fix an open compact subset $r_M \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ such that if $X \in (\mathfrak{m} \cap \mathfrak{s})(F), U \in (\mathfrak{n}_P \cap \mathfrak{s})(F), k \in K_H$ satisfy $\operatorname{Ad}(k)(X+U) \in r$, then $X \in r_M$; this is possible for it suffices to let r_M contain the projection of $\operatorname{Ad}(K_H)(r)$ to $(\mathfrak{m} \cap \mathfrak{s})(F)$. Replacing r_M with $\operatorname{Ad}(M_H(F) \cap K)(r_M)$ if necessary, we may additionally assume that

(7.1.8)
$$\operatorname{Ad}(M_H(F) \cap K)(r_M) = r_M.$$

Since $e_d \in \mathscr{E}^{\mathfrak{m} \cap \mathfrak{s}}$, there exists c > 0 and $N \in \mathbb{N}$ such that

$$|e_d(X)| \le c \sup\{1, -\log |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_F\}^N$$

for all $X \in r_M \cap \mathfrak{s}_{rs}(F)$. One sees from (3.1.1) that for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$,

$$|D^{\mathfrak{s}}(X)|_{F}|D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_{F}^{-1} = |\det(\mathrm{ad}(X)|_{\mathfrak{g/m}})|_{F}^{1/2}$$

Hence, $|D^{\mathfrak{s}}(X)|_F |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_F^{-1}$ is bounded for $X \in r_M \cap \mathfrak{s}_{rs}(F)$. We deduce that there exists c' > 0 such that

(7.1.9)
$$|e_d(X)| \le c' \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

for all $X \in r_M \cap \mathfrak{s}_{rs}(F)$. Now, suppose that $x \in T_{\mathfrak{c}}(F) \setminus H(F)$ and $X \in \mathfrak{c}_{reg}(F)$ satisfy $\operatorname{Ad}(x^{-1})(X) \in r$. Write x = mnk with $m \in M_H(F), n \in N_{P_H}(F)$ and $k \in K_H$. Then

$$Ad(x^{-1})(X) = Ad(k^{-1})(Ad(m^{-1})(X) + U)$$

where $U := \operatorname{Ad}(n^{-1}m^{-1})(X) - \operatorname{Ad}(m^{-1})(X) \in (\mathfrak{n}_P \cap \mathfrak{s})(F)$. Thus $\operatorname{Ad}(m^{-1})(X) \in r_M$ by our assumption on r_M . Thanks to (7.1.8) and (7.1.9), we obtain

$$|e_1(x,X)| \le c' \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

To sum up, e_1 verifies the hypotheses of Lemma 7.4.

Applying Lemma 7.4 to \mathfrak{c} and e_1 , we know that the distribution d_1 on $\mathfrak{s}(F)$ defined by

$$\forall f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F)), d_1(f) := \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{T_{\mathfrak{c}}(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(X)) e_1(x, X) dx dX$$

belongs to $\mathscr{D}^{\mathfrak{s}}$. Note that $e_1(mnk, X) = \eta(\operatorname{Nrd}(k))e_1(m, X)$ for $m \in M_H(F), n \in N_{P_H}(F), k \in K_H$. By the same argument as the proof of Proposition 4.1.4), one shows that

$$\begin{aligned} d_{1}(f) &= \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_{F}^{1/2} \int_{T_{\mathfrak{c}}(F)\backslash M_{H}(F)} f_{P}^{\eta}(\mathrm{Ad}(m^{-1})(X))e_{1}(m,X)dmdX \\ &= \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_{F}^{1/2} \int_{T_{\mathfrak{c}}(F)\backslash M_{H}(F)} f_{P}^{\eta}(\mathrm{Ad}(m^{-1})(X)) \int_{M_{H}(F)\cap K} e_{d}(\mathrm{Ad}(mk)^{-1}(X))\eta(\mathrm{Nrd}(k)) \\ &\quad dkdmdX. \end{aligned}$$

Note that for $k \in M_H(F) \cap K$, we have $\operatorname{Ad}(k^{-1})f_P^{\eta} = \eta(\operatorname{Nrd}(k))f_P^{\eta}$. By the change of variables $mk \mapsto m$, one can eliminate the integral over $M_H(F) \cap K$ and see that d_1 is the same as (7.1.6).

Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $d \in \mathscr{D}^{\mathfrak{m}\cap\mathfrak{s}}$. Suppose that d is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $M_H(F)$. Following [51, §V.9], we define a distribution $\operatorname{Ind}_M^{G,w}(d)$ on $\mathfrak{s}(F)$ by (7.1.10)

$$\mathrm{Ind}_{M}^{G,w}(d)(f) := \sum_{\{L \in \mathscr{L}^{G,\omega}(M_{0}): L \subseteq M\}} |W_{0}^{L_{n}}| |W_{0}^{M_{n}}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{l} \cap \mathfrak{s})} |W(L_{H},\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{M}^{G}(\eta, X, f) e_{d}(X) dX$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, where $J^{G}_{M}(\eta, X, f)$ is defined by (4.1.1). In particular, if M = G and $d \in \mathscr{D}^{\mathfrak{s}}$ is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of H(F), we have $\operatorname{Ind}^{G,w}_{G}(d) = d$ by (7.1.2).

COROLLARY 7.6. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $d \in \mathscr{D}^{\mathfrak{m}\cap\mathfrak{s}}$. Suppose that d is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $M_H(F)$. Then the integral (7.1.10) is absolutely convergent. Moreover, the distribution $\operatorname{Ind}_M^{G,w}(d) \in \mathscr{D}^{\mathfrak{s}}$.

REMARK 7.7. This corollary is unnecessary for the proof of Proposition 7.2 but useful in Section 8.1.

PROOF OF COROLLARY 7.6. It suffices to fix a Cartan subspace $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ and prove the same assertion for the distribution on $\mathfrak{s}(F)$ defined by

(7.1.11)
$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J^{G}_{M}(\eta, X, f) e_{d}(X) dX.$$

Define a function $e_2: (T_{\mathfrak{c}}(F) \setminus H(F)) \times \mathfrak{c}_{reg}(F) \to \mathbb{C}$ by

(7.1.12)
$$e_2(x,X) := \eta(\operatorname{Nrd}(x))v_M^G(x)e_d(X)$$

for all $x \in T_{\mathfrak{c}}(F) \setminus H(F)$ and all $X \in \mathfrak{c}_{reg}(F)$. It is locally constant. Note that $e_d(\operatorname{Ad}(m^{-1})(X)) = \eta(\operatorname{Nrd}(m))e_d(X)$ for $x \in M_H(F)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ by our assumption on d. Thus we may use the same argument as in the proof of Corollary 7.5 to show the inequality

$$|e_d(X)| = |e_d(\mathrm{Ad}(m^{-1})(X))| \le c' \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N.$$

Thanks to Lemma 4.2, one has a similar bound for $v_M^G(x)$. In sum, e_2 verifies the hypotheses of Lemma 7.4.

Applying Lemma 7.4 to \mathfrak{c} and e_2 , we know that the integral

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{c}_{\mathrm{reg}}(F)} |D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{T_{\mathfrak{c}}(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(X)) e_{2}(x, X) dx dX$$

is absolutely convergent and defines a distribution in $\mathscr{D}^{\mathfrak{s}}$. This distribution is the same as (7.1.11).

REMARK 7.8. Instead of the hypothesis $e_d \in \mathscr{E}^{\mathfrak{m} \cap \mathfrak{s}}$, an analogue of Corollary 7.6 holds if one assumes that d is defined via (7.1.1) by a function $e_d : (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \to \mathbb{C}$ locally constant, $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $M_H(F)$, and such that for all open compact subset r of $(\mathfrak{m} \cap \mathfrak{s})(F)$, there exists c > 0 and $N \in \mathbb{N}$ such that for all $X \in r \cap \mathfrak{s}_{rs}(F)$, one has the inequality

$$|e_d(X)| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N.$$

The proof is the same, except that we need not use the boundedness of $|D^{\mathfrak{s}}(X)|_F |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_F^{-1}$ for $X \in r_M \cap \mathfrak{s}_{rs}(F)$ (see the proof of Corollary 7.5).

LEMMA 7.9. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ be an *M*-elliptic Cartan subspace.

1) Let $M' \in \mathscr{L}^{G,\omega}(M_0), \mathfrak{c}' \in \mathscr{T}_{ell}(\mathfrak{m}' \cap \mathfrak{s})$ and $x \in H(F)$ be such that $\operatorname{Ad}(x)(\mathfrak{c}) = \mathfrak{c}'$. Then there exists $m' \in M'_H(F)$ and $w \in \operatorname{Norm}_{H(F)}(M_0)$ such that x = m'w.

2) The cardinality of

 $\{(M', \mathfrak{c}') : M' \in \mathscr{L}^{G, \omega}(M_0), \mathfrak{c}' \in \mathscr{T}_{ell}(\mathfrak{m}' \cap \mathfrak{s}), \mathfrak{c}' \text{ is } H(F) \text{-conjugate to } \mathfrak{c}\}$

is

$$|W_0^{GL_n}||W_0^{M_n}|^{-1}|W(M_H,\mathfrak{c})||W(H,\mathfrak{c})|^{-1}.$$

PROOF. 1) Since $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ (resp. $\mathfrak{c}' \subseteq \mathfrak{m}' \cap \mathfrak{s}$) is *M*-elliptic (resp. *M'*-elliptic), we have $A_{T_{\mathfrak{c}}} = A_M$ (resp. $A_{T_{\mathfrak{c}'}} = A_{M'}$). From $\operatorname{Ad}(x)(\mathfrak{c}) = \mathfrak{c}'$, we obtain $\operatorname{Ad}(x)(A_{T_{\mathfrak{c}}}) = A_{T_{\mathfrak{c}'}}$, $\operatorname{Ad}(x)(A_M) = A_{M'}$ and Ad(x)(M) = M', which implies that $Ad(x)(M_H) = M'_H$. Then 1) can be shown by the same argument as in the proof of Lemma 3.10.1).

2) We can and shall identify an *M*-elliptic (resp. *M'*-elliptic) Cartan subspace in $\mathfrak{m} \cap \mathfrak{s}$ (resp. $\mathfrak{m}' \cap \mathfrak{s}$) with its $M_H(F)$ (resp. $M'_H(F)$)-conjugacy class. Then $\mathscr{T}_{ell}(\mathfrak{m}' \cap \mathfrak{s})$ is identified to the set of $M'_H(F)$ conjugacy classes of M'-elliptic Cantan subspaces in $\mathfrak{m}' \cap \mathfrak{s}$. As in the proof of Lemma 3.10.1), we also

see that the group $W_0^{H,\omega} := \left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{GL_{n,D}} \right\}$ acts transitively on the set of pairs in 2). Firstly, let us count M' appearing in the pairs (cf. [35, p. 426]). Since $M \in \mathscr{L}^{G,\omega}(M_0)$, for $w \in W_0^{H,\omega}$, we see that $w = \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} \in \operatorname{Norm}_{W_0^{GL_{n,D}}}(M)$ if and only if $\omega_n \in \operatorname{Norm}_{W_0^{GL_{n,D}}}(M_{n,D})$, where $\operatorname{Norm}_{W_0^{H,\omega}}(M)$ (resp. $\operatorname{Norm}_{W_0^{GL_{n,D}}}(M_{n,D})$) denotes the normaliser of M (resp. $M_{n,D}$) in $W_0^{H,\omega}$ (resp. $W_0^{GL_{n,D}}$). Hence, the number of M' is

$$|W_0^{GL_{n,D}}||\operatorname{Norm}_{W_0^{GL_{n,D}}}(M_{n,D})|^{-1}$$

Secondly, for such an M' fixed, we count \mathfrak{c}' such that (M', \mathfrak{c}') belongs to the set of pairs in 2) (cf. [**35**, Lemma 7.1]).

For $x \in H(F)$, we claim that $\operatorname{Ad}(x)(\mathfrak{c}') \subseteq \mathfrak{m}' \cap \mathfrak{s}$ if and only if $x \in \operatorname{Norm}_{H(F)}(M')$, which denotes the normaliser of M' in H(F). On the one hand, suppose that $\operatorname{Ad}(x)(\mathfrak{c}') \subseteq \mathfrak{m}' \cap \mathfrak{s}$. Then $A_{M'} \subseteq \operatorname{Cent}_H(\mathfrak{m}' \cap \mathfrak{s})$ $\mathfrak{s} \subseteq \mathrm{Ad}(x)(T_{\mathfrak{c}'})$, where $\mathrm{Cent}_H(\mathfrak{m}' \cap \mathfrak{s})$ denotes the centraliser of $\mathfrak{m}' \cap \mathfrak{s}$ in H. Since $\mathrm{Ad}(x)(A_{M'}) =$ $\operatorname{Ad}(x)(A_{T_{\mathfrak{c}'}})$ is the maximal F-split torus in $\operatorname{Ad}(x)(T_{\mathfrak{c}'})$, we have $A_{M'} \subseteq \operatorname{Ad}(x)(A_{M'})$. By comparison of dimensions, we deduce that $\operatorname{Ad}(x)(A_{M'}) = A_{M'}$, so $x \in \operatorname{Norm}_{H(F)}(M')$. On the other hand, suppose that $x \in \operatorname{Norm}_{H(F)}(M')$. Since $x \in H(F)$, we have $\operatorname{Ad}(x)(\mathfrak{m}' \cap \mathfrak{s}) = (\mathfrak{m}' \cap \mathfrak{s})$. But $\mathfrak{c}' \subseteq \mathfrak{m}' \cap \mathfrak{s}$, so we obtain $\operatorname{Ad}(x)(\mathfrak{c}') \subseteq \mathfrak{m}' \cap \mathfrak{s}$. In sum, we have proved our claim.

From this claim, the number of \mathfrak{c}' is

$$|M'_H(F) \setminus \operatorname{Norm}_{H(F)}(M') / \operatorname{Norm}_{H(F)}(\mathfrak{c}')|$$

Since $M'_{H}(F)$ is a normal subgroup of Norm_{H(F)}(M'), we know that the number of the double cosets is equal to

$$|\operatorname{Norm}_{H(F)}(M')/M'_{H}(F)||\operatorname{Norm}_{H(F)}(\mathfrak{c}')/(\operatorname{Norm}_{H(F)}(\mathfrak{c}') \cap M'_{H}(F))|^{-1}$$

=|Norm_{H(F)}(M')/M'_{H}(F)||Norm_{H(F)}(\mathfrak{c}')/Norm_{M'_{H}(F)}(\mathfrak{c}')|^{-1}.

For $x \in \operatorname{Norm}_{H(F)}(M')$, we have $\operatorname{Ad}(x)(A_{M'}) = A_{M'}$. Because $M' \in \mathscr{L}^{G,\omega}(M_0)$, there exists $w \in W_0^{H,\omega}$ such that $w^{-1}x \in \operatorname{Cent}_{H(F)}(A_{M'}) = M'_H(F)$, where $\operatorname{Cent}_{H(F)}(A_{M'})$ denotes the centraliser of $A_{M'}$ in H(F). Since $x \in \operatorname{Norm}_{H(F)}(M')$, we have $w \in \operatorname{Norm}_{W^{H,\omega}}(M')$. That is to say,

$$\operatorname{Norm}_{H(F)}(M') = \operatorname{Norm}_{W_0^{H,\omega}}(M')M'_H(F).$$

Therefore,

$$\begin{aligned} |\mathrm{Norm}_{H(F)}(M')/M'_{H}(F)| = |\mathrm{Norm}_{W_{0}^{H,\omega}}(M')||(\mathrm{Norm}_{W_{0}^{H,\omega}}(M') \cap M'_{H}(F))|^{-1} \\ = |\mathrm{Norm}_{W_{0}^{GL_{n,D}}}(M'_{n,D})||(W_{0}^{H,\omega} \cap M'_{H}(F))|^{-1} \\ = |\mathrm{Norm}_{W_{0}^{GL_{n,D}}}(M'_{n,D})||W_{0}^{M'_{n,D}}|^{-1}. \end{aligned}$$

Since M' and M are $W_0^{H,\omega}$ -conjugate, $M'_{n,D}$ and $M_{n,D}$ are $W_0^{GL_{n,D}}$ -conjugate. Hence,

$$|\operatorname{Norm}_{H(F)}(M')/M'_{H}(F)| = |\operatorname{Norm}_{W_{\circ}^{GL_{n,D}}}(M_{n,D})||W_{0}^{M_{n,D}}|^{-1}.$$

We also have

$$|\operatorname{Norm}_{H(F)}(\mathfrak{c}')/\operatorname{Norm}_{M'_{H}(F)}(\mathfrak{c}')| = |\operatorname{Norm}_{H(F)}(\mathfrak{c}')/T_{\mathfrak{c}'}(F)||\operatorname{Norm}_{M'_{H}(F)}(\mathfrak{c}')/T_{\mathfrak{c}'}(F)|^{-1}$$
$$= |W(H,\mathfrak{c}')||W(M'_{H},\mathfrak{c}')|^{-1}.$$

Since (M', \mathfrak{c}') and (M, \mathfrak{c}) are $W_0^{H, \omega}$ -conjugate, we obtain

$$|\operatorname{Norm}_{H(F)}(\mathfrak{c}')/\operatorname{Norm}_{M'_{H}(F)}(\mathfrak{c}')| = |W(H,\mathfrak{c})||W(M_{H},\mathfrak{c})|^{-1}.$$

To sum up, the number of \mathfrak{c}' is

$$|\operatorname{Norm}_{W_0^{GL_{n,D}}}(M_{n,D})||W_0^{M_{n,D}}|^{-1}|W(M_H,\mathfrak{c})||W(H,\mathfrak{c})|^{-1}.$$

Finally, combining the numbers of M' and \mathfrak{c}' , we obtain the number of pairs (M', \mathfrak{c}') in 2).

PROOF OF PROPOSITION 7.2. First of all, suppose that $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$. Let \mathfrak{c} be the centraliser of X in \mathfrak{s} . Then $\mathfrak{c} \subseteq \mathfrak{m} \cap \mathfrak{s}$ is an *M*-elliptic Cartan subspace and $X \in \mathfrak{c}_{reg}(F)$. Fix an open compact subgroup $r \subseteq \mathfrak{s}(F)$ and set $r^* := \{Y \in \mathfrak{s}(F) : \forall Z \in r, \Psi(\langle Y, Z \rangle) = 1\}$, which is also an open compact subgroup of $\mathfrak{s}(F)$. For all $L \in \mathscr{L}^G(M)$, fix an open compact subgroup $r_L \subseteq (\mathfrak{l} \cap \mathfrak{s})(F)$ such that if $Q \in \mathscr{P}^G(L)$ and if $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ satisfies $\operatorname{Supp}(f) \subseteq r$, then $\operatorname{Supp}(f_Q^\eta) \subseteq r_L$, where f_Q^η is defined by (3.2.2). Define r_L^* in the same way as r^* .

There exists a neighbourhood σ of X in $\mathfrak{c}_{\mathrm{reg}}(F)$ such that for all $L \in \mathscr{L}^G(M)$ and all $f \in \mathcal{C}^{\infty}_c((\mathfrak{l} \cap \mathfrak{s})(F)/r_L^*)$, the function $J^L_M(\eta, \cdot, f)$ is constant on σ . In fact, for L and f fixed, this results from Proposition 4.1.2) (actually its product form is needed). It suffices to apply Howe's finiteness (the product form of Corollary 6.6) to each symmetric pair $(L, L_H, \mathrm{Ad}(\epsilon))$ and an arbitrary compact neighbourhood of X in $\mathfrak{c}_{\mathrm{reg}}(F)$, and then take the intersection of a finite number of neighbourhoods involved.

We shall fix a σ satisfying the above condition and such that if two elements of σ are H(F)-conjugate (or equivalently $W(H, \mathfrak{c})$ -conjugate), then they are the same. The latter condition is achievable since the $W(H, \mathfrak{c})$ -conjugates of an element in $\mathfrak{c}_{reg}(F)$ form a finite subset, which is discrete. Consider the local isomorphism $\beta : (T_{\mathfrak{c}}(F) \setminus H(F)) \times \mathfrak{c}_{reg}(F) \to \mathfrak{s}_{rs}(F)$ of F-analytic manifolds induced by the adjoint action. Its restriction to $(T_{\mathfrak{c}}(F) \setminus H(F)) \times \sigma$ is injective. Choose a neighbourhood ε of 1 in $T_{\mathfrak{c}}(F) \setminus H(F)$ such that $\eta(\operatorname{Nrd}(\varepsilon)) = 1$. The set $\beta(\varepsilon, \sigma)$ is a neighbourhood of X in $\mathfrak{s}(F)$. Fix a function $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ such that $\operatorname{Supp}(f') \subseteq \beta(\varepsilon, \sigma), f' \geq 0$ and $f'(X) \neq 0$.

Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ with $\operatorname{Supp}(f) \subseteq r$. We shall calculate $J^G(\eta, \hat{f}, f')$, which is defined by (5.1.2).

Consider $M' \in \mathscr{L}^{G,\omega}(M_0)$ and $\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m}' \cap \mathfrak{s})$. If \mathfrak{c}' and \mathfrak{c} are not H(F)-conjugate, by our choice of f', the function $J_{M'}^G(\eta, \cdot, \hat{f}, f')$ vanishes on $\mathfrak{c}'_{\mathrm{reg}}(F)$. Now suppose that \mathfrak{c}' and \mathfrak{c} are H(F)-conjugate. Let $x \in H(F)$ be such that $\mathrm{Ad}(x)(\mathfrak{c}) = \mathfrak{c}'$. By Lemma 7.9.1), there exists $m' \in M'_H(F)$ and $w \in \mathrm{Norm}_{H(F)}(M_0)$ such that x = m'w. By Proposition 5.1.4), for $X' \in \mathfrak{c}'_{\mathrm{reg}}(F)$, we have

$$\begin{split} J^G_{M'}(\eta, X', f, f') &= J^G_{M'}(\eta, \operatorname{Ad}(m'^{-1})(X'), f, f') \\ &= J^G_{\operatorname{Ad}(w)(M)}(\eta, \operatorname{Ad}(wx^{-1})(X'), \hat{f}, f') \\ &= J^G_M(\eta, \operatorname{Ad}(x^{-1})(X'), \hat{f}, f'). \end{split}$$

From our choices of Haar measures, we obtain

$$\int_{\mathfrak{c}'_{\mathrm{reg}}(F)} J^G_{M'}(\eta, X', \hat{f}, f') = \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J^G_M(\eta, Y, \hat{f}, f') dY.$$

By Lemma 7.9.2), the number of pairs (M', \mathfrak{c}') with \mathfrak{c}' being H(F)-conjugate to \mathfrak{c} is $|W_0^{GL_n}||W_0^{M_n}|^{-1}|W(M_H, \mathfrak{c})||W(H, \mathfrak{c})|^{-1}.$ We deduce that

$$J^{G}(\eta, \hat{f}, f') = (-1)^{\dim(A_{M}/A_{G})} |W(H, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J^{G}_{M}(\eta, Y, \hat{f}, f') dY.$$

It follows from our choice of f' that the support of the restriction to $\mathfrak{c}_{\mathrm{reg}}(F)$ of the function $J_M^G(\eta, \cdot, \hat{f}, f')$ is contained in $\coprod_{w \in W(H,\mathfrak{c})} \mathrm{Ad}(w)(\sigma)$. Then

$$\int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_M^G(\eta, Y, \hat{f}, f') dY = \sum_{w \in W(H, \mathfrak{c})} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_M^G(\eta, Y, \hat{f}, f') \mathbf{1}_{\mathrm{Ad}(w)(\sigma)}(Y) dY,$$

where $1_{\operatorname{Ad}(w)(\sigma)}$ denotes the characteristic function of $\operatorname{Ad}(w)(\sigma)$. By the change of variables $Y \mapsto \operatorname{Ad}(w)(Y)$, which does not modify the Haar measure, we have

$$\int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_M^G(\eta, Y, \hat{f}, f') \mathbf{1}_{\mathrm{Ad}(w)(\sigma)}(Y) dY = \int_{\sigma} J_M^G(\eta, \mathrm{Ad}(w)(Y), \hat{f}, f') dY$$

Since $w \in W(H, \mathfrak{c})$, we have shown above that

$$J_M^G(\eta, \operatorname{Ad}(w)(Y), \hat{f}, f') = J_M^G(\eta, Y, \hat{f}, f'),$$

which is independent of w. Therefore,

$$J^{G}(\eta, \hat{f}, f') = (-1)^{\dim(A_{M}/A_{G})} \int_{\sigma} J^{G}_{M}(\eta, Y, \hat{f}, f') dY$$

Let $Y \in \sigma$. Applying the splitting formula for $J_M^G(\eta, Y, \hat{f}, f')$ (Proposition 5.1.6)), we have

$$J_M^G(\eta, Y, \hat{f}, f') = \sum_{L_1, L_2 \in \mathscr{L}^G(M)} d_M^G(L_1, L_2) J_M^{L_1}(\eta, Y, \hat{f}_{\overline{Q}_1}^{\eta}) J_M^{L_2}(\eta, Y, f'_{\overline{Q}_2}).$$

For all $Q \in \mathscr{F}^G(M)$, since $\operatorname{Supp}(f_Q^{\eta}) \subseteq r_{M_Q}$, \hat{f}_Q^{η} is invariant by translation of $r_{M_Q}^*$. In particular, $\hat{f}_{Q_1}^{\eta}$ is invariant by $r_{L_1}^*$. Then by our assumption on σ , $J_M^{L_1}(\eta, \cdot, \hat{f}_{Q_1}^{\eta})$ is constant on σ and thus equal to $J_M^{L_1}(\eta, X, \hat{f}_{Q_1}^{\eta})$. Therefore,

$$J^{G}(\eta, \hat{f}, f') = \sum_{L_{1}, L_{2} \in \mathscr{L}^{G}(M)} c(L_{1}, L_{2}) J^{L_{1}}_{M}(\eta, X, \hat{f}^{\eta}_{Q_{1}}),$$

where

$$c(L_1, L_2) := d_M^G(L_1, L_2)(-1)^{\dim(A_M/A_G)} \int_{\sigma} J_M^{L_2}(\eta, Y, f'_{Q_2}) dY.$$

We claim that $c(G, M) \neq 0$. In fact, from (1) and (4) in Section 2.7, we have

$$c(G,M) = (-1)^{\dim(A_M/A_G)} \int_{\sigma} J_M^M(\eta, Y, f'_{Q_2}) dY,$$

where $M_{Q_2} = M$. By Proposition 4.1.4), we have

$$J_M^M(\eta, Y, {f'}_{Q_2}^{\eta}) = J_M^{Q_2}(\eta, Y, f').$$

Since $v_M^{Q_2} = 1$, we obtain

$$J_M^{Q_2}(\eta, Y, f') = J_G^G(\eta, Y, f').$$

Hence,

$$c(G,M) = (-1)^{\dim(A_M/A_G)} \int_{\sigma} |D^{\mathfrak{s}}(Y)|_F^{1/2} \int_{H_Y(F)\setminus H(F)} f'(\mathrm{Ad}(x^{-1})(Y))\eta(\mathrm{Nrd}(x))dxdY.$$

If $\operatorname{Ad}(x^{-1})(Y) \in \operatorname{Supp}(f') \subseteq \beta(\varepsilon, \sigma)$, since the restriction of β to $(H_Y(F) \setminus H(F)) \times \sigma$ is injective, we have $x \in \varepsilon$ and then $\eta(\operatorname{Nrd}(x)) = 1$. Since $f' \geq 0$ and $f'(X) \neq 0$, we deduce our claim. Now, because of (3) in Section 2.7, we have

$$(7.1.13) \qquad \hat{J}_{M}^{G}(\eta, X, f) = J^{G}(\eta, \hat{f}, f') - \sum_{L_{1}, L_{2} \in \mathscr{L}^{G}(M), L_{1} \neq G} c(L_{1}, L_{2}) c(G, M)^{-1} \hat{J}_{M}^{L_{1}}(\eta, X, f_{\overline{Q_{1}}}^{\eta})$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ with $\operatorname{Supp}(f) \subseteq r$.

By induction on the dimension of G and parabolic induction (Corollary 7.5), one can suppose that for all $L_1 \in \mathscr{L}^G(M), L_1 \neq G$, the distribution on $\mathfrak{s}(F)$ defined by $\forall f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F)), f \mapsto \hat{J}_M^{L_1}(\eta, X, f_{\overline{Q_1}}^{\eta})$ belongs to $\mathscr{D}^{\mathfrak{s}}$. This is actually a product form of the induction hypothesis in lower dimensions.

We claim that the distribution on $\mathfrak{s}(F)$ defined by $\forall f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F)), f \mapsto J^G(\eta, \hat{f}, f')$ belongs to $\mathscr{D}^{\mathfrak{s}}$ (cf. [51, (3) in §V.10]). In fact, thanks to the noninvariant trace formula (Theorem 5.3), one can replace $J^{G}(\eta, \hat{f}, f')$ with $J^{G}(\eta, f, \hat{f}')$. By its definition (5.1.2), it suffices to fix $M' \in \mathscr{L}^{G, \omega}(M_0), \mathfrak{c}' \in \mathscr{T}_{\text{ell}}(\mathfrak{m}' \cap \mathfrak{s})$ and prove that the distribution on $\mathfrak{s}(F)$ defined by

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} J^{G}_{M'}(\eta, Y, f, \hat{f'}) dY$$

belongs to $\mathscr{D}^{\mathfrak{s}}$. By the splitting formula (Proposition 5.1.6)), it suffices to fix $L'_1, L'_2 \in \mathscr{L}^G(M')$ and prove that the distribution on $\mathfrak{s}(F)$ defined by

$$\forall f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F)), f \mapsto \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} J^{L'_{1}}_{M'}(\eta, Y, f^{\eta}_{\overline{Q'_{1}}}) J^{L'_{2}}_{M'}(\eta, Y, \hat{f'}^{\eta}_{Q'_{2}}) dY_{M'}(\eta, Y, f'_{Q'_{2}}) dY_{M'}(\eta, Y, f'$$

where $(Q'_1, Q'_2) := s(L'_1, L'_2)$, belongs to $\mathscr{D}^{\mathfrak{s}}$. By Proposition 4.1.4) and the definition (4.1.1), we have

$$J_{M'}^{L'_1}(\eta, Y, f_{\overline{Q'_1}}^{\eta}) = J_{M'}^{\overline{Q'_1}}(\eta, Y, f) = |D^{\mathfrak{s}}(Y)|_F^{1/2} \int_{H_Y(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(Y))\eta(\operatorname{Nrd}(x))v_{M'}^{\overline{Q'_1}}(x)dx.$$

Then

$$\int_{\mathfrak{c}_{\mathrm{reg}}'(F)} J_{M'}^{L_{1}'}(\eta, Y, f_{\overline{Q}_{1}'}^{\eta}) J_{M'}^{L_{2}'}(\eta, Y, \hat{f}_{Q_{2}'}^{\eta}) dY$$

$$= \int_{\mathfrak{c}_{\mathrm{reg}}'(F)} |D^{\mathfrak{s}}(Y)|_{F}^{1/2} \int_{T_{\mathfrak{c}'}(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(Y)) \eta(\mathrm{Nrd}(x)) v_{M'}^{\overline{Q}_{1}'}(x) J_{M'}^{L_{2}'}(\eta, Y, \hat{f}_{Q_{2}'}^{\eta}) dx dY.$$

Define a function $e_3: (T_{\mathfrak{c}'}(F) \setminus H(F)) \times \mathfrak{c}'_{\operatorname{reg}}(F) \to \mathbb{C}$ by

$$e_{3}(x,Y) := \eta(\operatorname{Nrd}(x))v_{M'}^{\overline{Q'_{1}}}(x)J_{M'}^{L'_{2}}(\eta,Y,\hat{f'}_{Q'_{2}}).$$

It is locally constant by the product form of Proposition 4.1.2). Using Lemma 4.2 to dominate $v_{M'}^{\overline{Q'_1}}(x)$ and Corollary 4.3 to dominate $J_{M'}^{L'_2}(\eta, Y, \hat{f'}_{Q'_2}^{\eta})$, we check that e_3 verifies the hypotheses of Lemma 7.4, which implies our claim.

Now (7.1.13) shows that the distribution $\hat{J}_M^G(\eta, X, \cdot)$ conincides with some element in $\mathscr{D}^{\mathfrak{s}}$ for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ with $\operatorname{Supp}(f) \subseteq r$. By glueing (Remark 7.1), the distribution $\hat{J}^{G}_{M}(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$.

Finally, consider a general $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. There exists $x \in M_H(F)$ such that $Y := \operatorname{Ad}(x^{-1})(X) \in (\mathfrak{l} \cap \mathfrak{s})(F)_{\text{ell}}$ for some $L \in \mathscr{L}^{G,\omega}(M_0), L \subseteq M$. Then $\hat{J}_M^G(\eta, X, \cdot) = \eta(\operatorname{Nrd}(x))\hat{J}_M^G(\eta, Y, \cdot)$. Applying the descent formula (Proposition 4.1.5)), the product form of the elliptic case that we have just proved (applied to $Y \in (\mathfrak{l} \cap \mathfrak{s})(F)_{ell}$) and parabolic induction (Corollary 7.5), we deduce that the distribution $\hat{J}_M^G(\eta, Y, \cdot) \in \mathscr{D}^{\mathfrak{s}}$. Thus the distribution $\hat{J}_M^G(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$. \square

7.2. The case of (G', H'). We define $\mathscr{E}^{\mathfrak{s}'}$ in the same way as the previous case. For $e \in \mathscr{E}^{\mathfrak{s}'}$, thanks to Corollary 3.21, it defines a distribution on $\mathfrak{s}'(F)$:

(7.2.1)
$$\forall f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F)), f' \mapsto \int_{\mathfrak{s}'(F)} f'(Y) e(Y) |D^{\mathfrak{s}'}(Y)|_{F}^{-1/2} dY.$$

Denote by $\mathscr{D}^{\mathfrak{s}'}$ the space of distributions obtained in this way. For $d \in \mathscr{D}^{\mathfrak{s}'}$, we shall always denote by $e_d \in \mathscr{E}^{\mathfrak{s}'}$ its preimage under the isomorphism $\mathscr{E}^{\mathfrak{s}'} \to \mathscr{D}^{\mathfrak{s}'}$ defined above. One may extend these definitions to the symmetric pair $(\widetilde{M}', M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$. If $d \in \mathscr{D}^{\mathfrak{s}'}$ is invariant with respect to the adjoint action of H'(F), then so is $e_d \in \mathscr{E}^{\mathfrak{s}'}$ and by the Weyl integration formula (Proposition 3.18), we have the equality

(7.2.2)
$$d(f') = \sum_{M' \in \mathscr{L}^{H'}(M'_0)} |W_0^{M'}| |W_0^{H'}|^{-1} \sum_{\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')} |W(M', \mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} J_{H'}^{H'}(Y, f') e_d(Y) dY$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, where $J^{H'}_{H'}(Y, f')$ is defined by (4.2.1).

Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Denote by $\hat{J}_{M'}^{H'}(Y, \cdot)$ the distribution on $\mathfrak{s}'(F)$ defined by $(V f') := J_{M'}^{H'}(Y, \hat{f}')$ $\hat{J}_{M'}^{H'}$

$$J_{M'}^{H}(Y, f') := J_{M'}^{H}(Y, f)$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, where the right hand side is defined by (4.2.1). One also has a similar definition for the symmetric pair $(\widetilde{M}', M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$.

PROPOSITION 7.10 (Representability). Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Then the distribution $\hat{J}_{M'}^{H'}(Y, \cdot) \in \mathscr{D}^{\mathfrak{s}'}$.

The rest of this section is devoted to the proof of Proposition 7.10. Although it is similar to the proof of Proposition 7.2, we shall sketch some steps for later use.

Let \mathfrak{c}' be a Cartan subspace of \mathfrak{s}' . Recall that $T_{\mathfrak{c}'}$ denotes the centraliser of \mathfrak{c}' in H'. Suppose that $e_0: (T_{\mathfrak{c}'}(F) \setminus H'(F)) \times \mathfrak{c}'_{\mathrm{reg}}(F) \to \mathbb{C}$ is a function such that

(1) e_0 is locally constant;

(2) for all open compact subset r' of $\mathfrak{s}'(F)$, there exists c > 0 and $N \in \mathbb{N}$ such that for all $x \in T_{\mathfrak{c}'}(F) \setminus H'(F)$ and $Y \in \mathfrak{c}'_{reg}(F)$ satisfying $\operatorname{Ad}(x^{-1})(Y) \in r'$, one has the inequality

$$e_0(x,Y) \le c \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^N$$

For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we define

(7.2.3)
$$d_0(f') := \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{T_{\mathfrak{c}'}(F)\setminus H'(F)} f'(\mathrm{Ad}(x^{-1})(Y)) e_0(x,Y) dx dY.$$

LEMMA 7.11 (cf. Lemma 7.4). Let \mathfrak{c}' be a Cartan subspace of \mathfrak{s}' . Suppose that e_0 satisfies the above hypotheses. Then the integral (7.2.3) is absolutely convergent. Moreover, the distribution $d_0 \in \mathscr{D}^{\mathfrak{s}'}$.

COROLLARY 7.12 (Parabolic induction). Let $M' \in \mathscr{L}^{H'}(M'_0)$, $P' \in \mathscr{P}^{H'}(M')$ and $d \in \mathscr{D}^{\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'}$. Then the distribution on $\mathfrak{s}'(F)$ defined by $\forall f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F)), f' \mapsto d(f'_{P'})$ belongs to $\mathscr{D}^{\mathfrak{s}'}$, where $f'_{P'}$ is defined by (3.3.2).

PROOF. We may apply the argument of Corollary 7.5 with the aid of Lemma 3.19. \Box

Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $d \in \mathscr{D}^{\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'}$. Suppose that d is invariant with respect to the adjoint action of M'(F). We define a distribution $\operatorname{Ind}_{M'}^{H',w}(d)$ on $\mathfrak{s}'(F)$ by (7.2.4)

$$\operatorname{Ind}_{M'}^{H',w}(d)(f') := \sum_{\{L' \in \mathscr{L}^{H'}(M'_0): L' \subseteq M'\}} |W_0^{L'}| |W_0^{M'}|^{-1} \sum_{\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{l}'} \cap \mathfrak{s}')} |W(L',\mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} J_{M'}^{H'}(Y,f') e_d(Y) dY$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, where $J^{H'}_{M'}(Y, f')$ is defined by (4.2.1). In particular, if M' = H' and $d \in \mathscr{D}^{\mathfrak{s}'}$ is invariant with respect to the adjoint action of H'(F), we have $\operatorname{Ind}^{H',w}_{H'}(d) = d$ by (7.2.2).

COROLLARY 7.13. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $d \in \mathscr{D}^{\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'}$. Suppose that d is invariant with respect to the adjoint action of M'(F). Then the integral (7.2.4) is absolutely convergent. Moreover, the distribution $\operatorname{Ind}_{M'}^{H',w}(d) \in \mathscr{D}^{\mathfrak{s}'}$.

REMARK 7.14. This corollary is unnecessary for the proof of Proposition 7.10 but useful in Section 8.2.

PROOF OF COROLLARY 7.13. We may apply the argument of Corollary 7.6 thanks to Lemmas 4.5 and 7.11. $\hfill \Box$

LEMMA 7.15 (cf. Lemma 7.9). Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $\mathfrak{c}' \subseteq \widetilde{\mathfrak{m}'} \cap \mathfrak{s}'$ be an M'-elliptic Cartan subspace.

1) Let $M \in \mathscr{L}^{H'}(M'_0)$, $\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}} \cap \mathfrak{s}')$ and $x \in H'(F)$ be such that $\mathrm{Ad}(x)(\mathfrak{c}') = \mathfrak{c}$. Then there exists $m \in M(F)$ and $w \in \mathrm{Norm}_{H'(F)}(M'_0)$ such that x = mw.

2) The cardinality of

$$\{(M, \mathfrak{c}): M \in \mathscr{L}^{H'}(M'_0), \mathfrak{c} \in \mathscr{T}_{ell}(\widetilde{\mathfrak{m}} \cap \mathfrak{s}'), \mathfrak{c} \text{ is } H'(F) \text{-conjugate to } \mathfrak{c}'\}$$

is

$$|W_0^{H'}||W_0^{M'}|^{-1}|W(M',\mathfrak{c}')||W(H',\mathfrak{c}')|^{-1}.$$

PROOF OF PROPOSITION 7.10. We may apply the argument of Proposition 7.2 with obvious modifications. One needs almost all results that we have prepared in this and previous sections, notably Howe's finiteness (Corollary 6.9) and the noninvariant trace formula (Theorem 5.12). \Box

8. Invariant weighted orbital integrals

8.1. The case of (G, H). Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$. We shall define a distribution $\hat{I}_M^G(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$ which is $\eta(\mathrm{Nrd}(\cdot))$ -invariant with respect to the adjoint action of H(F) by induction on $\dim(G)$. Suppose that we have defined a distribution $\hat{I}_M^L(\eta, X, \cdot) \in \mathscr{D}^{\Gamma \cap \mathfrak{s}}$ which is $\eta(\mathrm{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $L_H(F)$ for all $L \in \mathscr{L}^G(M), L \neq G$. This is actually a product form of the induction hypothesis in lower dimensions. Denote by $\hat{I}_M^{L,G,w}(\eta, X, \cdot)$ its image under $\mathrm{Ind}_L^{G,w}$ (see (7.1.10)). As in [51, (1) in §VI.1], for $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we set

(8.1.1)
$$\hat{I}_{M}^{G}(\eta, X, f) := \hat{J}_{M}^{G}(\eta, X, f) - \sum_{L \in \mathscr{L}^{G}(M), L \neq G} \hat{I}_{M}^{L,G,w}(\eta, X, f).$$

PROPOSITION 8.1. The distribution $\hat{I}_M^G(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$ and is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of H(F).

PROOF. The first statement results from the representability of $\hat{J}_M^G(\eta, X, \cdot)$ (Proposition 7.2), the induction hypothesis and Corollary 7.6. Now let us consider the second one.

Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ and $y \in H(F)$. By the H(F)-invariance of $\langle \cdot, \cdot \rangle$, we see that $(\operatorname{Ad}(y^{-1})(f))^{\hat{}} = \operatorname{Ad}(y^{-1})(\hat{f})$. Applying Proposition 4.1.6), we have

$$\hat{J}_{M}^{G}(\eta, X, \mathrm{Ad}(y^{-1})(f)) = J_{M}^{G}(\eta, X, \mathrm{Ad}(y^{-1})(\hat{f})) = \eta(\mathrm{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} J_{M}^{M_{Q}}(\eta, X, (\hat{f})_{Q, y}^{\eta}).$$

For all $Q \in \mathscr{F}^G(M)$, we show that

$$(\widehat{f})_{Q,y}^\eta = (f_{Q,y}^\eta)^{\hat{}}$$

by the same argument of an analogous property of (3.2.2). Then

(8.1.2)
$$\hat{J}_{M}^{G}(\eta, X, \operatorname{Ad}(y^{-1})(f)) = \eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} \hat{J}_{M}^{M_{Q}}(\eta, X, f_{Q,y}^{\eta}).$$

Let $L \in \mathscr{L}^G(M), L \neq G$. Applying Proposition 4.1.6) again to $J_L^G(\eta, Y, \operatorname{Ad}(y^{-1})(f))$ in the integrand of the definition (7.1.10) of $\operatorname{Ind}_L^{G,w}$, we obtain

(8.1.3)
$$\hat{I}_{M}^{L,G,w}(\eta, X, \operatorname{Ad}(y^{-1})(f)) = \eta(\operatorname{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(L)} \hat{I}_{M}^{L,M_{Q},w}(\eta, X, f_{Q,y}^{\eta}).$$

where $\hat{I}_M^{L,M_Q,w}(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{m}_Q \cap \mathfrak{s}}$ denotes the image of $\hat{I}_M^L(\eta, X, \cdot)$ under $\mathrm{Ind}_L^{M_Q,w}$, which is defined by a product form of (7.1.10).

From (8.1.1), (8.1.2) and (8.1.3), we deduce that

$$\begin{split} \hat{I}_{M}^{G}(\eta, X, \mathrm{Ad}(y^{-1})(f)) = & \hat{J}_{M}^{G}(\eta, X, \mathrm{Ad}(y^{-1})(f)) - \sum_{L \in \mathscr{L}^{G}(M), L \neq G} \hat{I}_{M}^{L,G,w}(\eta, X, \mathrm{Ad}(y^{-1})(f)) \\ = & \eta(\mathrm{Nrd}(y)) \sum_{Q \in \mathscr{F}^{G}(M)} \left(\hat{J}_{M}^{M_{Q}}(\eta, X, f_{Q,y}^{\eta}) - \sum_{L \in \mathscr{L}^{M_{Q}}(M), L \neq G} \hat{I}_{M}^{L,M_{Q},w}(\eta, X, f_{Q,y}^{\eta}) \right). \end{split}$$

Consider $Q \neq G$ first. By the induction hypothesis, $\hat{I}_M^{M_Q}(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{m}_Q \cap \mathfrak{s}}$ is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $M_{Q_H}(F)$, so $\hat{I}_M^{M_Q,M_Q,w}(\eta, X, \cdot) = \hat{I}_M^{M_Q}(\eta, X, \cdot)$. By the definition of $\hat{I}_M^{M_Q}(\eta, X, \cdot)$ (a product form of (8.1.1)), the term in brackets is zero. Thus it remains the term for Q = G. Note that $f_{G,y}^{\eta} = f_G^{\eta}$. By Proposition 4.1.4) applied to Q = G, we see that the term in brackets is exactly $\hat{I}_M^G(\eta, X, f)$ defined by (8.1.1). Therefore, we show that

$$\hat{I}_M^G(\eta, X, \operatorname{Ad}(y^{-1})(f)) = \eta(\operatorname{Nrd}(y))\hat{I}_M^G(\eta, X, f),$$

which is the second statement.

Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$. Denote by $\hat{i}^G_M(\eta, X, \cdot)$ (resp. $\hat{j}^G_M(\eta, X, \cdot)$) the element of $\mathscr{E}^{\mathfrak{s}}$ associated to $\hat{I}^G_M(\eta, X, \cdot)$ (resp. $\hat{J}^G_M(\eta, X, \cdot)) \in \mathscr{D}^{\mathfrak{s}}$ by (7.1.1). That is to say, for all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$,

$$\hat{I}_{M}^{G}(\eta, X, f) = \int_{\mathfrak{s}(F)} f(Y)\hat{i}_{M}^{G}(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY$$

and

$$\hat{J}_M^G(\eta, X, f) = \int_{\mathfrak{s}(F)} f(Y) \hat{j}_M^G(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_F^{-1/2} dY.$$

We also have a similar definition for the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G, \omega}(M_0)$.

LEMMA 8.2. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Let $L \in \mathscr{L}^{G,\omega}(M_0)$ and $Y \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$. Then $\hat{i}_M^G(\eta, X, Y) = \hat{j}_M^G(\eta, X, Y)$.

PROOF. Let $L_2 \in \mathscr{L}^G(M), L_2 \neq G, L_1 \in \mathscr{L}^{G,\omega}(M_0), L_1 \subseteq L_2$ and $\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{l}_1 \cap \mathfrak{s})$. We define a distribution $d_{L_1,L_2,\mathfrak{c}}^{G,w}(\eta, X, \cdot)$ on $\mathfrak{s}(F)$ by

(8.1.4)
$$d_{L_1,L_2,\mathfrak{c}}^{G,w}(\eta, X, f) := \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{L_2}^G(\eta, Z, f) \hat{\imath}_M^{L_2}(\eta, X, Z) dZ$$

for all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$. By a product form of Proposition 8.1, the distribution $\hat{I}_M^{L_2}(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{l}_2 \cap \mathfrak{s}}$ and is $\eta(\operatorname{Nrd}(\cdot))$ -invariant with respect to the adjoint action of $(L_2 \cap H)(F)$. We see from the proof of Corollary 7.6 that the integral (8.1.4) is absolutely convergent and the distribution $d_{L_1,L_2,\mathfrak{c}}^{G,w}(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$. We denote by $e_{L_1,L_2,\mathfrak{c}}^{G,w}(\eta, X, \cdot)$ its associated element in $\mathscr{E}^{\mathfrak{s}}$ by (7.1.1). From the definitions (8.1.1) and (7.1.10), we have

$$\begin{split} \hat{i}_{M}^{G}(\eta, X, \cdot) = & \hat{j}_{M}^{G}(\eta, X, \cdot) - \sum_{L_{2} \in \mathscr{L}^{G}(M), L_{2} \neq G} \sum_{\{L_{1} \in \mathscr{L}^{G, \omega}(M_{0}): L_{1} \subseteq L_{2}\}} |W_{0}^{L_{1, n}}| |W_{0}^{L_{2, n}}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{l}_{1} \cap \mathfrak{s})} |W(L_{1} \cap H, \mathfrak{c})|^{-1} e_{L_{1}, L_{2}, \mathfrak{c}}^{G, w}(\eta, X, \cdot). \end{split}$$

To prove the lemma, it suffices to fix such a triple (L_2, L_1, \mathfrak{c}) and prove that $e_{L_1, L_2, \mathfrak{c}}^{G, w}(\eta, X, Y) = 0$. But (7.1.12) and (7.1.4) in the proofs of Corollary 7.6 and Lemma 7.4 respectively allow us to calculate $e_{L_1, L_2, \mathfrak{c}}^{G, w}(\eta, X, Y)$; explicitly, we have

$$(8.1.5) e^{G,w}_{L_1,L_2,\mathfrak{c}}(\eta,X,Y) = \sum_{\{x \in T_{\mathfrak{c}}(F) \setminus H(F): \operatorname{Ad}(x)(Y) \in \mathfrak{c}(F)\}} \eta(\operatorname{Nrd}(x)) v^G_{L_2}(x) \hat{i}^{L_2}_M(\eta,X,\operatorname{Ad}(x)(Y)).$$

Let $x \in T_{\mathfrak{c}}(F) \setminus H(F)$ such that $\operatorname{Ad}(x)(Y) \in \mathfrak{c}(F)$. As $Y \in (\mathfrak{l} \cap \mathfrak{s}_{\operatorname{rs}})(F)_{\operatorname{ell}}$ and $\mathfrak{c} \in \mathscr{T}_{\operatorname{ell}}(\mathfrak{l}_{1} \cap \mathfrak{s})$, from the proof of Lemma 3.10.1), there exists $l_{1} \in (L_{1} \cap H)(F)$ and $w \in \operatorname{Norm}_{H(F)}(M_{0})$ such that $x = l_{1}w$. Since any element in W_{0}^{H} admits a representative in K_{H} , we can suppose that $w \in K_{H}$. Then $v_{L_{2}}^{G}(x) = v_{L_{2}}^{G}(1)$ since $L_{1} \subseteq L_{2}$. But $v_{L_{2}}^{G}(1) = 0$ for $L_{2} \neq G$. Thus $e_{L_{1},L_{2},\mathfrak{c}}^{G}(\eta, X, Y) = 0$ by (8.1.5).

LEMMA 8.3. Let $M \in \mathscr{L}^{G,\omega}(M_0)$. 1) The function $(X, Y) \mapsto \hat{i}_M^G(\eta, X, Y)$ is locally constant on $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$. 2) If $w \in \operatorname{Norm}_{H(F)}(M_0)$, $x \in M_H(F)$ and $y \in H(F)$, we have the equality

$$\hat{i}_{\operatorname{Ad}(w)(M)}^{G}(\eta, \operatorname{Ad}(wx)(X), \operatorname{Ad}(y)(Y)) = \eta(\operatorname{Nrd}(wxy))\hat{i}_{M}^{G}(\eta, X, Y)$$

for all $(X, Y) \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$. 3) If $\lambda \in F^{\times}$, we have the equality

$$\hat{i}_M^G(\eta, \lambda X, Y) = \hat{i}_M^G(\eta, X, \lambda Y)$$

for all $(X, Y) \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$.

4) Let $r_M \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ and $r \subseteq \mathfrak{s}(F)$ be two compact subsets. Then there exists c > 0 and $N \in \mathbb{N}$ such that

$$|\hat{i}_M^G(\eta, X, Y)| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N \sup\{1, -\log |D^{\mathfrak{s}}(Y)|_F\}^N$$

for all $X \in r_M \cap \mathfrak{s}_{rs}$ and $Y \in r \cap \mathfrak{s}_{rs}$.

5) Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and $r_{M} \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ be a compact subset. Then there exists c > 0 and $N \in \mathbb{N}$ such that

$$|I_M^G(\eta, X, f)| \le c \sup\{1, -\log |D^{\mathfrak{s}}(X)|_F\}^N$$

for all $X \in r_M \cap \mathfrak{s}_{rs}$.

PROOF. Let $r_M \subseteq (\mathfrak{m} \cap \mathfrak{s})(F)$ and $r \subseteq \mathfrak{s}(F)$ be two open compact subgroups. Set $r^* := \{Y \in \mathfrak{s}(F) : \forall Z \in r, \Psi(\langle Y, Z \rangle) = 1\}$, which is an open compact subgroup of $\mathfrak{s}(F)$. Notice that if $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ satisfies $\operatorname{Supp}(f) \subseteq r$, then $\hat{f} \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F)/r^*)$. Applying Howe's finiteness (Corollary 6.6) to r^* and $r_M \cap \mathfrak{s}_{rs}$, we

know that there exists a finite subset $\{X_i : i \in I\} \subseteq r_M \cap \mathfrak{s}_{rs}$ and a finite subset $\{f_i : i \in I\} \subseteq \mathcal{C}^{\infty}_c(\mathfrak{s}(F)/r^*)$ such that for all $X \in r_M \cap \mathfrak{s}_{rs}$ and all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ with $\operatorname{Supp}(f) \subseteq r$, we have

$$J_M^G(\eta, X, \hat{f}) = \sum_{i \in I} J_M^G(\eta, X_i, \hat{f}) J_M^G(\eta, X, f_i)$$

We deduce that

(8.1.6)
$$\hat{j}_{M}^{G}(\eta, X, Y) = \sum_{i \in I} \hat{j}_{M}^{G}(\eta, X_{i}, Y) J_{M}^{G}(\eta, X, f_{i})$$

for all $X \in r_M \cap \mathfrak{s}_{rs}$ and $Y \in r \cap \mathfrak{s}_{rs}$.

1) The local constancy of $(X, Y) \mapsto \hat{j}_M^G(\eta, X, Y)$ on $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$ results from (8.1.6), Proposition 4.1.2) and $\hat{j}_M^G(\eta, X_i, \cdot) \in \mathscr{E}^{\mathfrak{s}}$ for $i \in I$. For $L \in \mathscr{L}^{G,\omega}(M_0)$, we deduce from Lemma 8.2 the local constancy of $(X, Y) \mapsto \hat{i}_M^G(\eta, X, Y)$ on $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$. Let $(X, Y) \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$. Choose $L \in \mathscr{L}^{G,\omega}(M_0)$ and $Y' \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$ such that Y' is H(F)-conjugate to Y. Fix a neighbourhood $V_1 \times V_2$ of (X, Y') in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$ such that $(X, Y) \mapsto \kappa(Y)\hat{i}_M^G(\eta, X, Y)$ is constant on $V_1 \times V_2$. Thanks to the $\eta(\operatorname{Nrd}(\cdot))$ -invariance of $\hat{i}_M^G(\eta, X, \cdot)$ with respect to the adjoint action of H(F)(Proposition 8.1), we know that $(X, Y) \mapsto \kappa(Y)\hat{i}_M^G(\eta, X, Y)$ is constant on $V_1 \times \operatorname{Ad}(H(F))(V_2)$ which is a neighbourhood of (X, Y) in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$. Since $\kappa(\cdot)$ is locally constant on $\mathfrak{s}_{rs}(F)$, we show that $(X, Y) \mapsto \hat{i}_M^G(\eta, X, Y)$ is constant on a neighbourhood of (X, Y) in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$.

2) The effect of $\operatorname{Ad}(y)$ comes from Propostion 8.1. Then when considering the effects of $\operatorname{Ad}(w)$ and $\operatorname{Ad}(x)$, up to H(F)-conjugation, we may and shall suppose that $Y \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$ for some $L \in \mathscr{L}^{G,\omega}(M_0)$. That is to say, it suffices to prove the equality

$$\hat{i}_{\mathrm{Ad}(w)(M)}^G(\eta, \mathrm{Ad}(wx)(X), Y) = \eta(\mathrm{Nrd}(wx))\hat{i}_M^G(\eta, X, Y)$$

for all $(X,Y) \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$. By Lemma 8.2, we may replace $\hat{i}_{Ad(w)(M)}^G$ and \hat{i}_M^G by $\hat{j}_{Ad(w)(M)}^G$ and \hat{j}_M^G respectively in the equality to be proved. Now the equality results from Proposition 4.1.3).

3) Let $\lambda \in F^{\times}, X \in (\mathfrak{m} \cap \mathfrak{s}_{reg})(F)$ and $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$. From the definition (3.1.1), we have

$$|D^{\mathfrak{s}}(\lambda X)|_{F} = |\lambda|_{F}^{(\dim(\mathfrak{g}) - \operatorname{rank}(\mathfrak{g}))/2} |D^{\mathfrak{s}}(X)|_{F},$$

where $\dim(\mathfrak{g})$ and $\operatorname{rank}(\mathfrak{g})$ denote the dimension and rank (over an algebraic closure of F) of \mathfrak{g} respectively. Then we have

$$\begin{split} &\int_{\mathfrak{s}(F)} f(Y)\hat{j}_{M}^{G}(\eta,\lambda X,Y)|D^{\mathfrak{s}}(Y)|_{F}^{-1/2}dY = \hat{J}_{M}^{G}(\eta,\lambda X,f) = J_{M}^{G}(\eta,\lambda X,\hat{f}) \\ = &|D^{\mathfrak{s}}(\lambda X)|_{F}^{1/2} \int_{H_{\lambda X}(F) \setminus H(F)} \hat{f}(\mathrm{Ad}(x^{-1})(\lambda X))\eta(\mathrm{Nrd}(x))v_{M}^{Q}(x)dx \\ = &|\lambda|_{F}^{(\dim(\mathfrak{g})-\mathrm{rank}(\mathfrak{g}))/4} J_{M}^{G}(\eta,X,\hat{f}(\lambda \cdot)). \end{split}$$

But

$$\begin{split} \hat{f}(\lambda \cdot) = & c_{\Psi}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} f(Z) \Psi(\langle \lambda \cdot, Z \rangle) dZ = c_{\Psi}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} f(Z) \Psi(\langle \cdot, \lambda Z \rangle) dZ \\ = & |\lambda|_{F}^{-\dim(\mathfrak{s})} c_{\Psi}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} f(\lambda^{-1}Z) \Psi(\langle \cdot, Z \rangle) dZ \\ = & |\lambda|_{F}^{-\dim(\mathfrak{g})/2} (f(\lambda^{-1} \cdot))^{\widehat{}}. \end{split}$$

Thus we have

$$\begin{split} J_M^G(\eta, X, \widehat{f}(\lambda \cdot)) = &|\lambda|_F^{-\dim(\mathfrak{g})/2} \widehat{J}_M^G(\eta, X, f(\lambda^{-1} \cdot)) = |\lambda|_F^{-\dim(\mathfrak{g})/2} \int_{\mathfrak{s}(F)} f(\lambda^{-1}Y) \widehat{j}_M^G(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_F^{-1/2} dY \\ = &\int_{\mathfrak{s}(F)} f(Y) \widehat{j}_M^G(\eta, X, \lambda Y) |D^{\mathfrak{s}}(\lambda Y)|_F^{-1/2} dY \\ = &|\lambda|_F^{(\operatorname{rank}(\mathfrak{g}) - \dim(\mathfrak{g}))/4} \int_{\mathfrak{s}(F)} f(Y) \widehat{j}_M^G(\eta, X, \lambda Y) |D^{\mathfrak{s}}(Y)|_F^{-1/2} dY. \end{split}$$

Therefore, we deduce the equality

$$\hat{j}_M^G(\eta, \lambda X, Y) = \hat{j}_M^G(\eta, X, \lambda Y)$$

for all $\lambda \in F^{\times}$ and all $(X, Y) \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F) \times \mathfrak{s}_{rs}(F)$. We obtain a similar equality for \hat{i}_M^G thanks to Lemma 8.2 and the $\eta(\operatorname{Nrd}(\cdot))$ -invariance of $\hat{i}_M^G(\eta, X, \cdot)$ with respect to the adjoint action of H(F) (Proposition 8.1).

4) A similar bound for $(X, Y) \mapsto \hat{j}_M^G(\eta, X, Y)$ on $(r_M \cap \mathfrak{s}_{rs}) \times (r \cap \mathfrak{s}_{rs})$ results from (8.1.6), Corollary 4.3 (together with Proposition 4.1.4)) and $\hat{j}_M^G(\eta, X_i, \cdot) \in \mathscr{E}^{\mathfrak{s}}$ for $i \in I$. For $L \in \mathscr{L}^{G,\omega}(M_0)$, we deduce from Lemma 8.2 a similar bound of $(X, Y) \mapsto \hat{i}_M^G(\eta, X, Y)$ on $(r_M \cap \mathfrak{s}_{rs}) \times (r \cap (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell})$. Let $(X, Y) \in (r_M \cap \mathfrak{s}_{rs}) \times (r \cap \mathfrak{s}_{rs})$. Thanks to the $\eta(\operatorname{Nrd}(\cdot))$ -invariance of $\hat{i}_M^G(\eta, X, \cdot)$ with respect to the adjoint action of H(F) (Proposition 8.1), if we replace Y by $\operatorname{Ad}(y)(Y)$, where $y \in H(F)$, the two sides in the inequality to be proved remain unchanged. Since any Cartan subspace in \mathfrak{s} is H(F)-conjugate to an element in $\mathscr{T}_{ell}(\mathfrak{l} \cap \mathfrak{s})$ for some $L \in \mathscr{L}^{G,\omega}(M_0)$, with the help of Lemma 3.3, it suffices to fix $L \in \mathscr{L}^{G,\omega}(M_0), \mathfrak{c} \in \mathscr{T}_{ell}(\mathfrak{l} \cap \mathfrak{s})$ and $r_{\mathfrak{c}} \subseteq \mathfrak{c}(F)$ a compact subset, prove a similar bound for $(X, Y) \in$ $(r_M \cap \mathfrak{s}_{rs}) \times (r_{\mathfrak{c}} \cap \mathfrak{c}_{reg})$, and then obtain a uniform bound for $(X, Y) \in (r_M \cap \mathfrak{s}_{rs}) \times (r \cap \mathfrak{s}_{rs})$ by the finiteness of $\mathscr{T}_{ell}(\mathfrak{l} \cap \mathfrak{s})$. But this is what we have established.

5) It is a consequence of 4) applied to r := Supp(f) and Corollary 3.15.

For
$$M \in \mathscr{L}^{G,\omega}(M_0)$$
 and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we define a distribution $I_M^G(\eta, X, \cdot)$ on $\mathfrak{s}(F)$ by
 $I_M^G(m, X, \hat{f}) = I_M^G(m, X, f)$

(8.1.7) $I_M^G(\eta, X, f) := I_M^G(\eta, X, f)$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$.

REMARK 8.4. For M = G, it is evident that $I_G^G(\eta, X, f) = J_G^G(\eta, X, f)$ for all $X \in \mathfrak{s}_{rs}(F)$ and $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$.

One may easily extend the definitions (8.1.1) and (8.1.7) to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \operatorname{Ad}(\epsilon))$ in lower dimensions.

LEMMA 8.5. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. The distribution $I_M^G(\eta, X, \cdot)$ on $\mathfrak{s}(F)$ is independent of the choice of the H(F)-invariant non-degenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$ on $\mathfrak{s}(F)$ or the continuous nontrivial unitary character Ψ of F.

PROOF. Suppose that $\langle \cdot, \cdot \rangle'$ is another bilinear form and that Ψ' is another character. Denote by $f \mapsto \tilde{f}$ the associated Fourier transform and by $\tilde{I}_M^G(\eta, X, \cdot)$ (resp. $\tilde{J}_M^G(\eta, X, \cdot)$) the associated analogue of $\hat{I}_M^G(\eta, X, \cdot)$ (resp. $\hat{J}_M^G(\eta, X, \cdot)$). Since $\tilde{f}(\cdot) = \hat{f}(\cdot) = f(-\cdot)$ for all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, it suffices to prove the equality

$$\tilde{I}_M^G(\eta, X, \tilde{f}) = \hat{I}_M^G(\eta, X, \hat{f})$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$.

Let τ' be the linear automorphism of $\mathfrak{s}(F)$ such that

$$\forall Y, Z \in \mathfrak{s}(F), \langle Y, Z \rangle' = \langle \tau'(Y), Z \rangle.$$

Let $a \in F^{\times}$ such that $\Psi'(\cdot) = \Psi(a \cdot)$. Set $\tau := a\tau'$. Then

$$\tilde{f}(\cdot) = \frac{c_{\Psi'}(\mathfrak{s}(F))}{c_{\Psi}(\mathfrak{s}(F))} \hat{f}(\tau(\cdot))$$

for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$. One may check that τ is an H(F)-equivariant linear automorphism of $\mathfrak{s}(F)$ thanks to H(F)-invariance of two bilinear forms. One also deduces that

$$\forall Y,Z\in\mathfrak{s}(F),\langle\tau(Y),Z\rangle=\langle Y,\tau(Z)\rangle$$

from the symmetry of two bilinear forms. Now for all $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and all $Y \in \mathfrak{s}(F)$, we have

$$\begin{split} f(-Y) &= \tilde{\tilde{f}}(Y) = c_{\Psi'}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} \tilde{f}(Z) \Psi'(\langle Y, Z \rangle') dZ = c_{\Psi'}(\mathfrak{s}(F)) \int_{\mathfrak{s}(F)} \frac{c_{\Psi'}(\mathfrak{s}(F))}{c_{\Psi}(\mathfrak{s}(F))} \hat{f}(\tau(Z)) \Psi(\langle \tau(Y), Z \rangle) dZ \\ &= \frac{c_{\Psi'}(\mathfrak{s}(F))^2}{c_{\Psi}(\mathfrak{s}(F))} \int_{\mathfrak{s}(F)} \hat{f}(\tau(Z)) \Psi(\langle Y, \tau(Z) \rangle) dZ = \frac{c_{\Psi'}(\mathfrak{s}(F))^2}{c_{\Psi}(\mathfrak{s}(F))|\det_{\mathfrak{s}(F)}(\tau)|_F} \int_{\mathfrak{s}(F)} \hat{f}(Z') \Psi(\langle Y, Z' \rangle) dZ' \\ &= \frac{c_{\Psi'}(\mathfrak{s}(F))^2}{c_{\Psi}(\mathfrak{s}(F))^2|\det_{\mathfrak{s}(F)}(\tau)|_F} \hat{f}(Y) = \frac{c_{\Psi'}(\mathfrak{s}(F))^2}{c_{\Psi}(\mathfrak{s}(F))^2|\det_{\mathfrak{s}(F)}(\tau)|_F} f(-Y). \end{split}$$

Therefore, we obtain $\frac{c_{\Psi'}(\mathfrak{s}(F))}{c_{\Psi}(\mathfrak{s}(F))} = |\det_{\mathfrak{s}(F)}(\tau)|_F^{1/2}$. Then for all $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$, we have

$$\tilde{f}(\cdot) = |\det_{\mathfrak{s}(F)}(\tau)|_F^{1/2} \hat{f}(\tau(\cdot))$$

Denote by $\tilde{i}_M^G(\eta, X, \cdot)$ (resp. $\tilde{j}_M^G(\eta, X, \cdot)$) the element of $\mathscr{E}^{\mathfrak{s}}$ associated to $\tilde{I}_M^G(\eta, X, \cdot)$ (resp. $\tilde{J}_M^G(\eta, X, \cdot)$) $\in \mathscr{D}^{\mathfrak{s}}$ by (7.1.1). For $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have

$$\begin{split} \tilde{I}_{M}^{G}(\eta, X, \tilde{f}) &= \int_{\mathfrak{s}(F)} \tilde{f}(Y) \tilde{i}_{M}^{G}(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY \\ &= |\det_{\mathfrak{s}(F)}(\tau)|_{F}^{1/2} \int_{\mathfrak{s}(F)} \hat{f}(\tau(Y)) \tilde{i}_{M}^{G}(\eta, X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY \\ &= |\det_{\mathfrak{s}(F)}(\tau)|_{F}^{-1/2} \int_{\mathfrak{s}(F)} \hat{f}(Y) \tilde{i}_{M}^{G}(\eta, X, \tau^{-1}(Y)) |D^{\mathfrak{s}}(\tau^{-1}(Y))|_{F}^{-1/2} dY. \end{split}$$

We reduce ourselves to proving the equality

 $|\det_{\mathfrak{s}(F)}(\tau)|_{F}^{-1/2}\tilde{i}_{M}^{G}(\eta, X, \tau^{-1}(Y))|D^{\mathfrak{s}}(\tau^{-1}(Y))|_{F}^{-1/2} = \hat{i}_{M}^{G}(\eta, X, Y)|D^{\mathfrak{s}}(Y)|_{F}^{-1/2}$ (8.1.8)for all $Y \in \mathfrak{s}_{rs}(F)$. But we have the equality

$$\tilde{J}_M^G(\eta,X,\tilde{f})=\hat{J}_M^G(\eta,X,\hat{f})$$

since both sides equal $J_M^G(\eta, X, f(-\cdot))$, which is defined by (4.1.1). The same computation as above shows that the equality (8.1.8) is true when one replaces \tilde{i}_M^G and \tilde{i}_M^G with \tilde{j}_M^G and \tilde{j}_M^G respectively. Recall that τ is H(F)-equivariant, so $H_Y = H_{\tau^{-1}(Y)}$ for $Y \in \mathfrak{s}_{\rm rs}(F)$. As a consequence, for $L \in \mathscr{L}^{G,\omega}(M_0)$, $Y \in (\mathfrak{l} \cap \mathfrak{s}_{\rm rs})(F)_{\rm ell}$ if and only if $\tau^{-1}(Y) \in (\mathfrak{l} \cap \mathfrak{s}_{\rm rs})(F)_{\rm ell}$. One may conclude by Lemma 8.2 together with the $\eta(\operatorname{Nrd}(\cdot))$ -invariance of $\hat{i}_M^G(\eta, X, \cdot)$ with respect to the adjoint action of H(F) (Proposition 8.1).

8.2. The case of (G', H'). Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. We shall define a distribution $\hat{I}_{M'}^{H'}(Y,\cdot) \in \mathscr{D}^{\mathfrak{s}'}$ which is invariant with respect to the adjoint action of H'(F) by induction on $\dim(H')$. Suppose that we have defined a distribution $\hat{I}_{M'}^{L'}(Y, \cdot) \in \mathscr{D}^{\widetilde{l'} \cap \mathfrak{s}'}$ which is invariant with respect to the adjoint action of L'(F) for all $L' \in \mathscr{L}^{H'}(M'), L' \neq H'$. This is actually a product form of the induction hypothesis in lower dimensions. Denote by $\hat{I}_{M'}^{L',H',w}(Y,\cdot)$ its image under $\operatorname{Ind}_{L'}^{H',w}$ (see (7.2.4)). For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we set

(8.2.1)
$$\hat{I}_{M'}^{H'}(Y,f') := \hat{J}_{M'}^{H'}(Y,f') - \sum_{L' \in \mathscr{L}^{H'}(M'), L' \neq H'} \hat{I}_{M'}^{L',H',w}(Y,f').$$

PROPOSITION 8.6. The distribution $\hat{I}_{M'}^{H'}(Y, \cdot) \in \mathscr{D}^{\mathfrak{s}'}$ and is invariant with respect to the adjoint action of H'(F).

PROOF. We may apply the argument of Proposition 8.1 thanks to the representability of $\hat{J}^{H'}_{M'}(Y, \cdot)$ (Proposition 7.10), Corollary 7.13 and Proposition 4.4.6).

Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$. Denote by $\hat{i}_{M'}^{H'}(Y, \cdot)$ (resp. $\hat{j}_{M'}^{H'}(Y, \cdot)$) the element of $\mathscr{E}^{\mathfrak{s}'}$ associated to $\hat{I}_{M'}^{H'}(Y, \cdot)$ (resp. $\hat{J}_{M'}^{H'}(Y, \cdot)) \in \mathscr{D}^{\mathfrak{s}'}$ by (7.2.1). That is to say, for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$,

$$\hat{I}_{M'}^{H'}(Y,f') = \int_{\mathfrak{s}'(F)} f'(X)\hat{i}_{M'}^{H'}(Y,X) |D^{\mathfrak{s}'}(X)|_F^{-1/2} dX$$

and

$$\hat{J}_{M'}^{H'}(Y,f') = \int_{\mathfrak{s}'(F)} f'(X)\hat{j}_{M'}^{H'}(Y,X) |D^{\mathfrak{s}'}(X)|_F^{-1/2} dX$$

One has a similar definition for the symmetric pair $(\widetilde{M}', M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$.

LEMMA 8.7. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$. Let $L' \in \mathscr{L}^{H'}(M'_0)$ and $X \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)_{\mathrm{ell}}$. Then $\hat{i}_{M'}^{H'}(Y, X) = \hat{j}_{M'}^{H'}(Y, X)$.

PROOF. We may apply the argument of Lemma 8.2 by using Proposition 8.6 and consulting the proofs of Corollary 7.13, Lemmas 7.11 and 3.17.1).

LEMMA 8.8. Let $M' \in \mathscr{L}^{H'}(M'_0)$.

1) The function $(Y, X) \mapsto \hat{i}_{M'}^{H'}(Y, X)$ is locally constant on $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F) \times \mathfrak{s}'_{rs}(F)$. 2) If $w \in \operatorname{Norm}_{H'(F)}(M'_0)$, $x \in M'(F)$ and $y \in H'(F)$, we have the equality

$$\operatorname{Ad}_{\operatorname{Ad}(w)(M')}(\operatorname{Ad}(wx)(Y), \operatorname{Ad}(y)(X)) = i_{M'}^{H'}(Y, X)$$

for all $(Y, X) \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F) \times \mathfrak{s}'_{rs}(F)$.

3) If $\lambda \in F^{\times}$, we have the equality

$$\hat{i}_{M'}^{H'}(\lambda Y, X) = \hat{i}_{M'}^{H'}(Y, \lambda X)$$

for all $(Y, X) \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F) \times \mathfrak{s}'_{\mathrm{rs}}(F).$

4) Let $r'_{M'} \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(F)$ and $r' \subseteq \mathfrak{s}'(F)$ be two compact subsets. Then there exists c > 0 and $N \in \mathbb{N}$ such that

$$|\hat{i}_{M'}^{H'}(Y,X)| \le c \sup\{1, -\log|D^{\mathfrak{s}'}(Y)|_F\}^N \sup\{1, -\log|D^{\mathfrak{s}'}(X)|_F\}^N$$

for all $Y \in r'_{M'} \cap \mathfrak{s}'_{rs}$ and $X \in r' \cap \mathfrak{s}'_{rs}$.

5) Let $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ and $r'_{M'} \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(F)$ be a compact subset. Then there exists c > 0 and $N \in \mathbb{N}$ such that

$$|\hat{I}_{M'}^{H'}(Y, f')| \le c \sup\{1, -\log |D^{\mathfrak{s}'}(Y)|_F\}^{\Lambda}$$

for all $Y \in r'_{M'} \cap \mathfrak{s}'_{rs}$.

PROOF. It is almost the same as the proof of Lemma 8.3, except that one needs to use Howe's finiteness (Corollary 6.9), Proposition 8.6 and Lemma 8.7. We also need Proposition 4.4.2) for 1), Proposition 4.4.3) for 2), Corollary 4.6 for 4) and Corollary 3.21 for 5). \Box

For
$$M' \in \mathscr{L}^{H'}(M'_0)$$
 and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$, we define a distribution $I_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}'(F)$ by

(8.2.2)
$$I_{M'}^{H'}(Y, \hat{f}') := \hat{I}_{M'}^{H'}(Y, f')$$

for all $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$. One may easily extend the definitions (8.2.1) and (8.2.2) to the symmetric pair $(\widetilde{M'}, M', \operatorname{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_{0})$.

REMARK 8.9. For M' = H', it is evident that $I_{H'}^{H'}(Y, f') = J_{H'}^{H'}(Y, f')$ for all $Y \in \mathfrak{s}'_{rs}(F)$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$.

LEMMA 8.10. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. The distribution $I_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}'(F)$ is independent of the choice of the H'(F)-invariant non-degenerate symmetric bilinear form $\langle \cdot, \cdot \rangle$ on $\mathfrak{s}'(F)$ or the continuous nontrivial unitary character Ψ of F.

PROOF. We may apply the argument of Lemma 8.5 thanks to Proposition 8.6 and Lemma 8.7. \Box

9. The invariant trace formula

9.1. The case of (G, H). For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we define

(9.1.1)
$$I^{G}(\eta, f, f') := \sum_{\substack{M \in \mathscr{L}^{G, \omega}(M_{0}) \\ \hat{I}^{G}_{M}(\eta, X, f) I^{G}_{G}(\eta, X, f') dX.}} |W^{GL_{n}|^{-1}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\text{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\text{reg}}(F)} \hat{I}^{G}_{M}(\eta, X, f) I^{G}_{G}(\eta, X, f') dX.$$

From Proposition 4.1.2), for any $\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})$, $I_G^G(\eta, \cdot, f')$ vanishes outside a compact subset of $\mathfrak{c}(F)$, so one may apply Lemma 8.3.5) to show that this expression is absolutely convergent with the help of Proposition 3.13 and Corollary 3.6.

THEOREM 9.1 (Invariant trace formula). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$, we have the equality

$$I^G(\eta, f, f') = I^G(\eta, f', f).$$

The rest of this section is devoted to the proof of Theorem 9.1. We shall follow the main steps in [51, §VII.2-3]. The theorem will be proved by induction on the dimension of G.

Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}_{\mathrm{rs}}(F))$ and $M \in \mathscr{L}^{G,\omega}(M_0)$. By Proposition 4.1.2) and 3), the function $\kappa(\cdot)J_M^G(\eta, \cdot, f)$: $(\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F) \to \mathbb{C}$ is locally constant and invariant by the adjoint action of $M_H(F)$, where κ is defined by (4.1.2). Moreover, the support of its restriction to $\mathfrak{c}(F)$ for any $\mathfrak{c} \in \mathscr{T}(\mathfrak{m} \cap \mathfrak{s})$ is included in the compact subset $\mathfrak{c}_{\mathrm{reg}}(F) \cap \mathrm{Ad}(H(F))(\mathrm{Supp}(f))$. From Harish-Chandra's submersion principle (Lemma 3.8), there exists $f' \in \mathcal{C}_c^{\infty}((\mathfrak{m} \cap \mathfrak{s})_{\mathrm{rs}}(F))$ such that

$$\kappa(X)J_M^G(\eta, X, f) = |D^{\mathfrak{m}\cap\mathfrak{s}}(X)|_F^{1/2} \int_{M_{H,X}(F)\backslash M_H(F)} f'(\mathrm{Ad}(x^{-1})(X))dx$$

for all $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Let $f'' := \kappa f' \in \mathcal{C}_c^{\infty}((\mathfrak{m} \cap \mathfrak{s})_{rs}(F))$, where we extend the definition of κ to the product form. Then we have

(9.1.2)
$$J_M^G(\eta, X, f) = J_M^M(\eta, X, f''), \forall X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$$

We have shown that for $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}_{\mathrm{rs}}(F))$ and $M \in \mathscr{L}^{G,\omega}(M_{0})$, there exists a function $f'' \in \mathcal{C}^{\infty}_{c}((\mathfrak{m} \cap \mathfrak{s})_{\mathrm{rs}}(F))$ such that (9.1.2) holds. We shall fix such an f'' and denote it by $\phi_{M}^{G}(f)$.

As before, one may extend in the obvious way the definition (9.1.1) and the notation $\phi_M^G(f)$ to the symmetric pair $(M, M_H, \operatorname{Ad}(\epsilon))$, where $M \in \mathscr{L}^{G,\omega}(M_0)$, since it appears as the product of some copies of the form $(G, H, \operatorname{Ad}(\epsilon))$ in lower dimensions.

LEMMA 9.2. Let $M \in \mathscr{L}^{G,\omega}(M_0), X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$ and $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}_{rs}(F))$. Then we have the equality

$$(9.1.3) J_M^G(\eta, X, \hat{f}, f') = \sum_{L \in \mathscr{L}^G(M)} \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) I_L^L(\eta, X, \phi_L^{L_2}(f'_{Q_2})),$$

where $J_M^G(\eta, X, \hat{f}, f')$ is defined by (5.1.1), and $(Q_1, Q_2) := s(L_1, L_2)$ (see Section 2.7).

PROOF. By definition,

$$J_M^G(\eta, X, \hat{f}, f') = |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{A_M(F) \setminus H(F)} f'(\operatorname{Ad}(y^{-1})(X))\eta(\operatorname{Nrd}(y))\varphi_1(y)dy,$$

where

$$\varphi_1(y) := |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{A_M(F)\backslash H(F)} \hat{f}(\operatorname{Ad}(x^{-1})(X))\eta(\operatorname{Nrd}(x^{-1}))v_M(x,y)dx.$$

For $L \in \mathscr{L}^{G}(M)$ and $L_{2} \in \mathscr{L}^{G}(L)$, since $Q_{2} \in \mathscr{P}^{G}(L_{2})$, by Proposition 4.1.4), we have (9.1.4) $I_{L}^{L}(\eta, X, \phi_{L}^{L_{2}}(f'_{Q_{2}}^{\eta})) = J_{L}^{L_{2}}(\eta, X, f'_{Q_{2}}) = J_{L}^{Q_{2}}(\eta, X, f').$

Since $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$, the right hand side of (9.1.3) is

$$|D^{\mathfrak{s}}(X)|_{F}^{1/2} \int_{A_{M}(F)\setminus H(F)} f'(\mathrm{Ad}(y^{-1})(X))\eta(\mathrm{Nrd}(y))\varphi_{2}(y)dy$$

where

$$\varphi_2(y) := \sum_{L \in \mathscr{L}^G(M)} \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^L(\eta, X, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y) + \sum_{L \in \mathscr{L}^G(M)} d_L^G(L_1, L_2) \hat{I}_M^{Q_2}(y) + \sum_{L \in$$

It suffices to fix $y \in H(F)$ and prove that $\varphi_1(y) = \varphi_2(y)$.

Let $L \in \mathscr{L}^G(M)$ and

$$h_L := \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta}) v_L^{Q_2}(y).$$

Then

$$\varphi_2(y) = \sum_{L \in \mathscr{L}^G(M)} \hat{I}_M^L(\eta, X, h_L).$$

For $Y \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$, we have

$$J_L^L(\eta, Y, h_L) = \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) J_L^L(\eta, Y, \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})) v_L^{Q_2}(y).$$

For $L_1 \in \mathscr{L}^G(L)$, as in (9.1.4), we have

$$J_{L}^{L}(\eta, Y, \phi_{L}^{L_{1}}(f_{\overline{Q_{1}}}^{\eta})) = J_{L}^{L_{1}}(\eta, Y, f_{\overline{Q_{1}}}^{\eta}) = J_{L}^{\overline{Q_{1}}}(\eta, Y, f).$$

Then

$$J_{L}^{L}(\eta, Y, h_{L}) = |D^{\mathfrak{s}}(Y)|_{F}^{1/2} \int_{H_{Y}(F) \setminus H(F)} f(\mathrm{Ad}(x^{-1})(Y))\eta(\mathrm{Nrd}(x))h(x, y)dx$$

where

$$h(x,y) := \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) v_L^{\overline{Q_1}}(x) v_L^{Q_2}(y).$$

It is shown in the proof of [51, Lemme VII.2] that

$$h(x,y) = v_L(x,y) = \sum_{Q \in \mathscr{F}^G(L)} v'_Q(y) v_L^{\overline{Q}}(x).$$

Thus

$$J_L^L(\eta, Y, h_L) = \sum_{Q \in \mathscr{F}^G(L)} v'_Q(y) J_L^{\overline{Q}}(\eta, Y, f).$$

As in (9.1.4), we have

$$J^Q_L(\eta,Y,f) = J^{M_Q}_L(\eta,Y,f^\eta_{\overline{Q}}) = J^L_L(\eta,Y,\phi^{M_Q}_L(f^\eta_{\overline{Q}})).$$

Let

$$h'_L := \sum_{Q \in \mathscr{F}^G(L)} v'_Q(y) \phi_L^{M_Q}(f_{\overline{Q}}^{\eta}).$$

Then we obtain

(9.1.5)
$$J_L^L(\eta, Y, h_L) = J_L^L(\eta, Y, h'_L)$$

for all $Y \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$.

By a product form of Proposition 8.1, the distribution $d := \hat{I}_M^L(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{l} \cap \mathfrak{s}}$ and is $\eta(\operatorname{Nrd}(\cdot))$ invariant with respect to the adjoint action of $L_H(F)$. By a product form of (7.1.2), we deduce from
(9.1.5) that

$$\hat{I}_M^L(\eta, X, h_L) = \hat{I}_M^L(\eta, X, h'_L)$$

Therefore,

$$\begin{split} \varphi_2(y) &= \sum_{L \in \mathscr{L}^G(M)} \hat{I}^L_M(\eta, X, h_L) = \sum_{L \in \mathscr{L}^G(M)} \hat{I}^L_M(\eta, X, h'_L) \\ &= \sum_{Q \in \mathscr{F}^G(M)} v'_Q(y) \sum_{L \in \mathscr{L}^{M_Q}(M)} \hat{I}^L_M(\eta, X, \phi_L^{M_Q}(f\frac{\eta}{Q})). \end{split}$$

By (7.1.10) and (7.1.2) (actually their product forms are needed), we have

$$\begin{split} \hat{I}_{M}^{L,M_{Q},w}(\eta,X,f_{\overline{Q}}^{\eta}) &= \mathrm{Ind}_{L}^{M_{Q},w}(d)(f_{\overline{Q}}^{\eta}) \\ &= \sum_{\{L'\in\mathscr{L}^{G,\omega}(M_{0}):L'\subseteq L\}} |W_{0}^{L'_{n}}||W_{0}^{L_{n}}|^{-1} \sum_{\mathfrak{c}\in\mathscr{T}_{\mathrm{ell}}(l'\cap\mathfrak{s})} |W(L'_{H},\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{L}^{M_{Q}}(\eta,Z,f_{\overline{Q}}^{\eta})e_{d}(Z)dZ \\ &= \sum_{\{L'\in\mathscr{L}^{G,\omega}(M_{0}):L'\subseteq L\}} |W_{0}^{L'_{n}}||W_{0}^{L_{n}}|^{-1} \sum_{\mathfrak{c}\in\mathscr{T}_{\mathrm{ell}}(l'\cap\mathfrak{s})} |W(L'_{H},\mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{L}^{L}(\eta,Z,\phi_{L}^{M_{Q}}(f_{\overline{Q}}^{\eta}))e_{d}(Z)dZ \\ &= d(\phi_{L}^{M_{Q}}(f_{\overline{Q}}^{\eta})) = \hat{I}_{M}^{L}(\eta,X,\phi_{L}^{M_{Q}}(f_{\overline{Q}}^{\eta})). \end{split}$$

Then by (8.1.1), we get

$$\sum_{L \in \mathscr{L}^{M_Q}(M)} \hat{I}_M^L(\eta, X, \phi_L^{M_Q}(f_{\overline{Q}}^{\eta})) = \sum_{L \in \mathscr{L}^{M_Q}(M)} \hat{I}_M^{L,M_Q,w}(\eta, X, f_{\overline{Q}}^{\eta})$$
$$= \hat{J}_M^{M_Q}(\eta, X, f_{\overline{Q}}^{\eta}) = J_M^{M_Q}(\eta, X, \hat{f}_{\overline{Q}}^{\eta}) = J_M^{\overline{Q}}(\eta, X, \hat{f}).$$

Hence,

$$\begin{split} \varphi_2(y) &= \sum_{Q \in \mathscr{F}^G(M)} v'_Q(y) J_M^{\overline{Q}}(\eta, X, \hat{f}) \\ &= |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{A_M(F) \setminus H(F)} \hat{f}(\operatorname{Ad}(x^{-1})(X)) \eta(\operatorname{Nrd}(x)) \sum_{Q \in \mathscr{F}^G(M)} v'_Q(y) v_M^{\overline{Q}}(x) dx. \end{split}$$

But

$$\sum_{Q\in\mathscr{F}^G(M)} v'_Q(y) v^{\overline{Q}}_M(x) = v_M(x,y),$$

which implies that $\varphi_1(y) = \varphi_2(y)$.

PROOF OF THEOREM 9.1. We use induction on the dimension of G. Suppose that the equality is true for $L \in \mathscr{L}^{G,\omega}(M_0), L \neq G$, which is actually a product form in lower dimensions. Now we would like to prove the equality for G. The argument below is also valid for the case $\mathscr{L}^{G,\omega}(M_0) = \{G\}$, i.e., n = 1.

First of all, suppose that $f, f' \in C_c^{\infty}(\mathfrak{s}_{rs}(F))$. Applying Lemma 9.2 to the definition (5.1.2) of $J^G(\eta, \hat{f}, f')$, we obtain

$$\begin{split} J^{G}(\eta, \hat{f}, f') &= \sum_{M \in \mathscr{L}^{G, \omega}(M_{0})} |W_{0}^{M_{n}}| |W_{0}^{GL_{n}}|^{-1}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} \\ J^{G}_{M}(\eta, X, \hat{f}, f') dX \\ &= \sum_{M \in \mathscr{L}^{G, \omega}(M_{0})} |W_{0}^{M_{n}}| |W_{0}^{GL_{n}}|^{-1}(-1)^{\dim(A_{M}/A_{G})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} \\ &\sum_{L \in \mathscr{L}^{G}(M)} \sum_{L_{1}, L_{2} \in \mathscr{L}^{G}(L)} d^{G}_{L}(L_{1}, L_{2}) \hat{I}^{L}_{M}(\eta, X, \phi^{L_{1}}_{L}(f^{\eta}_{Q_{1}})) I^{L}_{L}(\eta, X, \phi^{L_{2}}_{L}(f'^{\eta}_{Q_{2}})) dX \\ &= \sum_{L \in \mathscr{L}^{G, \omega}(M_{0})} |W_{0}^{GL_{n}}|^{-1}(-1)^{\dim(A_{L}/A_{G})} B_{L}(\eta, f, f'), \end{split}$$

where

$$\begin{split} B_{L}(\eta, f, f') &\coloneqq \sum_{L_{1}, L_{2} \in \mathscr{L}^{G}(L)} d_{L}^{G}(L_{1}, L_{2}) \sum_{\{M \in \mathscr{L}^{G, \omega}(M_{0}): M \subseteq L\}} |W_{0}^{M_{n}}| |W_{0}^{L_{n}}|^{-1}(-1)^{\dim(A_{M}/A_{L})} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})} |W(M_{H}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} \hat{I}_{M}^{L}(\eta, X, \phi_{L}^{L_{1}}(f_{\overline{Q_{1}}}^{\eta})) I_{L}^{L}(\eta, X, \phi_{L}^{L_{2}}(f'_{Q_{2}})) dX \\ &= \sum_{L_{1}, L_{2} \in \mathscr{L}^{G}(L)} d_{L}^{G}(L_{1}, L_{2}) I^{L}(\eta, \phi_{L}^{L_{1}}(f_{\overline{Q_{1}}}^{\eta}), \phi_{L}^{L_{2}}(f'_{Q_{2}})). \end{split}$$

Here we have used the absolute convergence of the expressions above to exchange the order of sums, and $I^{L}(\eta, \phi_{L}^{L_{1}}(f_{\overline{Q_{1}}}^{\eta}), \phi_{L}^{L_{2}}(f'_{Q_{2}}^{\eta}))$ is defined by a product form of (9.1.1).

By the noninvariant trace formula (Theorem 5.3) and Remark 5.2, we have the equality $J^G(\eta, \hat{f}, f') =$ $J^G(\eta, \widehat{f}', f).$ Therefore,

(9.1.6)
$$\sum_{L \in \mathscr{L}^{G,\omega}(M_0)} |W_0^{L_n}| |W_0^{GL_n}|^{-1} (-1)^{\dim(A_L/A_G)} (B_L(\eta, f, f') - B_L(\eta, f', f)) = 0$$

Let $L \in \mathscr{L}^{G,\omega}(M_0), L \neq G$. Applying the induction hypothesis, we have

$$B_L(\eta, f, f') = \sum_{L_1, L_2 \in \mathscr{L}^G(L)} d_L^G(L_1, L_2) I^L(\eta, \phi_L^{L_2}(f'_{Q_2}), \phi_L^{L_1}(f_{\overline{Q_1}}^{\eta})).$$

By exchanging L_1 and L_2 and by using (2) and (5) in Section 2.7, we obtain $B_L(\eta, f, f') = B_L(\eta, f', f)$. We deduce from (9.1.6) that $B_G(\eta, f, f') = B_G(\eta, f', f)$. But

$$B_G(\eta, f, f') = I^G(\eta, f, f'),$$

which implies $I^G(\eta, f, f') = I^G(\eta, f', f)$. Now consider $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ in general. Let $\{\Omega_i\}_{i\geq 1}$ be a sequence of increasing open compact subsets of $\mathfrak{s}_{rs}(F)$ such that $\bigcup_{i\geq 1}^{\infty} \Omega_i = \mathfrak{s}_{rs}(F)$. Such a consequence exists. For example, one may take $\Omega_i := \{X \in \mathfrak{s}_{rs}(F) : ||X|| \leq i\}$ for all $i \geq 1$, where $||\cdot||$ denotes the abstract norm on $\mathfrak{s}_{rs}(F)$ defined by [35, (18.2.1) in §18.2]. From [35, Proposition 18.1.(3)], since $\|\cdot\|$ is continuous, we deduce that Ω_i is compact for all $i \ge 1$. It is obvious that Ω_i is open for all $i \ge 1$ and that $\bigcup_{i\ge 1}^{\infty} \Omega_i = \mathfrak{s}_{rs}(F)$. For all $i \ge 1$,

denote by 1_{Ω_i} the characteristic function of Ω_i . Let $f_i := f 1_{\Omega_i}$ and $f'_i := f' 1_{\Omega_i}$. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $\mathfrak{c} \in \mathscr{L}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})$. For all $X \in \mathfrak{c}_{\mathrm{reg}}(F)$, by Lebesgue's dorminated convergence theorem, we have $\lim_{i \to \infty} I^G_G(\eta, X, f'_i) = I^G_G(\eta, X, f')$. For $X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$, because $\hat{I}^G_M(\eta, X, \cdot) \in \mathscr{D}^{\mathfrak{s}}$ (see Proposition 8.1), again by Lebesgue's dorminated convergence theorem, we have $\lim_{i \to \infty} \hat{I}_M^G(\eta, X, f_i) =$ $\hat{I}_M^G(\eta, X, f)$. Because of Lemma 3.3 applied to $\operatorname{Supp}(f')$, there exists a compact subset $r \subseteq \mathfrak{c}(F)$ such that for all $X \in \mathfrak{c}_{\operatorname{reg}}(F) - r, I_G^G(\eta, X, f'_i) = 0$ for all $i \geq 1$. By Lemma 8.3.4) applied to r and $\operatorname{Supp}(f')$,

there exists c > 0 and $N \in \mathbb{N}$ such that

$$\begin{aligned} |\hat{I}_{M}^{G}(\eta, X, f_{i})| &= \int_{\mathfrak{s}(F)} |f_{i}(Y)\hat{i}_{M}^{G}(\eta, X, Y)| |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY \\ &\leq c \sup\{1, -\log|D^{\mathfrak{s}}(X)|_{F}\}^{N} \int_{\mathfrak{s}(F)} |f(Y)| \sup\{1, -\log|D^{\mathfrak{s}}(Y)|_{F}\}^{N} |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dY \end{aligned}$$

for all $i \geq 1$ and $X \in r \cap \mathfrak{c}_{reg}$. For all $X \in \mathfrak{s}_{rs}(F)$, we also have $|I_G^G(\eta, X, f'_i)| \leq I_X(|f'|)$, where I_X is defined by (3.1.3). Combining Corollary 3.15, Corollary 3.6 and Proposition 3.13, we deduce that $\{\hat{I}_M^G(\eta, X, f_i) | I_G^G(\eta, X, f'_i)\}_{i\geq 1}$ is bounded by an integrable function on $\mathfrak{c}_{reg}(F)$. Using Lebesgue's dorminated convergence theorem once again, we obtain

$$\int_{\mathfrak{c}_{\mathrm{reg}}(F)} \hat{I}^G_M(\eta, X, f) I^G_G(\eta, X, f') dX = \lim_{i \to \infty} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} \hat{I}^G_M(\eta, X, f_i) I^G_G(\eta, X, f'_i) dX$$

Therefore,

$$I^G(\eta, f, f') = \lim_{i \to \infty} I^G(\eta, f_i, f'_i).$$

By exchanging f and f' and using the regular semi-simple support case that we have proved, we draw the conclusion.

COROLLARY 9.3. Let $M, L \in \mathscr{L}^{G,\omega}(M_0), X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$ and $Y \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$. Then we have the equality

$$(-1)^{\dim(A_M/A_G)}\hat{i}^G_M(\eta, X, Y) = (-1)^{\dim(A_L/A_G)}\hat{i}^G_L(\eta, Y, X).$$

PROOF. By Lemma 8.3.2), up to $M_H(F)$ -conjugation on X and $L_H(F)$ -conjugation on Y, we may and shall suppose that there exists $\mathfrak{c}_1 \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{m} \cap \mathfrak{s})$ and $\mathfrak{c}_2 \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{l} \cap \mathfrak{s})$ such that $X \in \mathfrak{c}_{1,\mathrm{reg}}(F)$ and $Y \in \mathfrak{c}_{2,\mathrm{reg}}(F)$. As in the proof of Proposition 7.2, we can choose an open compact neighbourhood V_1 of Xin $\mathfrak{c}_{1,\mathrm{reg}}$ (resp. V_2 of Y in $\mathfrak{c}_{2,\mathrm{reg}}$) such that if two elements in V_1 (resp. V_2) are H(F)-conjugate, then they are the same. Let $f, f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ with $\mathrm{Supp}(f) \subseteq \mathrm{Ad}(H(F))(V_2)$ and $\mathrm{Supp}(f') \subseteq \mathrm{Ad}(H(F))(V_1)$. By an analogous calculation to that of $J^G(\eta, \hat{f}, f')$ in the proof of Proposition 7.2, with the help of Lemma 8.3.2) and Proposition 4.1.3), we show the equalities

$$I^{G}(\eta, f, f') = (-1)^{\dim(A_{M}/A_{G})} \int_{V_{1}} \hat{I}^{G}_{M}(\eta, X_{1}, f) I^{G}_{G}(\eta, X_{1}, f') dX_{1}$$

and

$$\hat{I}_{M}^{G}(\eta, X_{1}, f) = \int_{V_{2}} \hat{i}_{M}^{G}(\eta, X_{1}, Y_{2}) I_{G}^{G}(\eta, Y_{2}, f) dY_{2}$$

for all $X_1 \in V_1$ by (7.1.2). Then

$$I^{G}(\eta, f, f') = (-1)^{\dim(A_{M}/A_{G})} \int_{V_{1} \times V_{2}} \hat{i}^{G}_{M}(\eta, X_{1}, Y_{2}) I^{G}_{G}(\eta, Y_{2}, f) I^{G}_{G}(\eta, X_{1}, f') dY_{2} dX_{1}.$$

Similarly, we have

$$I^{G}(\eta, f', f) = (-1)^{\dim(A_{L}/A_{G})} \int_{V_{2} \times V_{1}} \hat{i}_{L}^{G}(\eta, Y_{2}, X_{1}) I_{G}^{G}(\eta, X_{1}, f') I_{G}^{G}(\eta, Y_{2}, f) dX_{1} dY_{2}.$$

By Harish-Chandra's submersion principle (Lemma 3.8), when f' varies, the function $X_1 \mapsto I_{X_1}(f') = \kappa(X_1)I_G^G(\eta, X_1, \kappa f')$ on V_1 runs over all $\mathcal{C}_c^{\infty}(V_1)$, so the function $I_G^G(\eta, \cdot, f')$ on V_1 also runs over all $\mathcal{C}_c^{\infty}(V_1)$. Similarly, when f varies, the function $I_G^G(\eta, \cdot, f)$ on V_2 runs over all $\mathcal{C}_c^{\infty}(V_2)$. Then from the invariant trace formula (Theorem 9.1), we deduce that

$$(-1)^{\dim(A_M/A_G)}\hat{i}_M^G(\eta, X_1, Y_2) = (-1)^{\dim(A_L/A_G)}\hat{i}_L^G(\eta, Y_2, X_1)$$

for all $(X_1, Y_2) \in V_1 \times V_2$. We conclude by $(X, Y) \in V_1 \times V_2$.
9.2. The case of (G', H'). For $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we define

(9.2.1)
$$I^{H'}(f,f') := \sum_{\substack{M' \in \mathscr{L}^{H'}(M'_0) \\ \hat{I}^{H'}_{M'}(Y,f) I^{H'}_{H'}(Y,f') dY.}} |W^{M'}_0|^{-1} (-1)^{\dim(A_{M'}/A_{H'})} \sum_{\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')} |W(M',\mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M',\mathfrak{c}')|^{-1} |W(M',\mathfrak{c}')|^{-1} \int_{\mathfrak{c}'_{\mathrm{reg}}(F)} |W(M',\mathfrak{c}')|^{-1} |W(M',\mathfrak{c}')|^{-1}$$

From Proposition 4.4.2), for any $\mathfrak{c}' \in \mathscr{T}_{\mathrm{ell}}(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')$, $I_{H'}^{H'}(\cdot, f')$ vanishes outside a compact subset of $\mathfrak{c}'(F)$, so one may apply Lemma 8.8.5) to show that this expression is absolutely convergent with the help of Proposition 3.20 and Corollary 3.6. One may extend in the obious way the definition (9.2.1) to the symmetric pair $(\widetilde{M}', M', \mathrm{Ad}(\alpha))$, where $M' \in \mathscr{L}^{H'}(M'_0)$.

THEOREM 9.4 (Invariant trace formula). For all $f, f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we have the equality

$$I^{H'}(f, f') = I^{H'}(f', f).$$

PROOF. We may apply the argument of Theorem 9.1 with obvious modifications. It is deduced from the noninvariant trace formula (Theorem 5.12) and other results that we have prepared in previous sections. \Box

COROLLARY 9.5. Let $M', L' \in \mathscr{L}^{H'}(M'_0), Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$ and $X \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$. Then we have the equality

$$(-1)^{\dim(A_{M'}/A_{H'})}\hat{i}_{M'}^{H'}(Y,X) = (-1)^{\dim(A_{L'}/A_{H'})}\hat{i}_{L'}^{H'}(Y,X).$$

PROOF. We may apply the argument of Corollary 9.3 by using the invariant trace formula (Theorem 9.4) and consulting the proof of Proposition 7.10. \Box

10. A vanishing property at infinity

10.1. The case of (G, H). The following proposition is an analogue of [14, Proposition 2.2].

PROPOSITION 10.1. Let $M \in \mathscr{L}^{G,\omega}(M_0), M \neq G$. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in \mathfrak{s}_{rs}(F)$. Then there exists $N \in \mathbb{N}$ such that if $\lambda \in F^{\times}$ satisfies $v_F(\lambda) < -N$, we have

$$\hat{i}_M^G(\eta, \lambda X, Y) = 0.$$

REMARK 10.2. A limit formula at infinity for $\hat{i}_G^G(\eta, \lambda X, Y)$ in the spirit of Laplace transform is given in [58, Proposition 7.1] (see also [57, Proposition 6.4]), which is an analogue of [51, Proposition VIII.1].

PROOF OF PROPOSITION 10.1. We shall imitate the proof of [14, Proposition 2.2].

By Lemma 8.3.2), up to H(F)-conjugation on Y, we may and shall suppose that there exists $L \in \mathscr{L}^{G,\omega}(M_0)$ and an L-elliptic Cartan subspace $\mathfrak{c} \subseteq \mathfrak{l} \cap \mathfrak{s}$ such that $Y \in \mathfrak{c}_{\mathrm{reg}}(F)$. By Lemma 8.2, we have the equality

$$\hat{i}_M^G(\eta, \lambda X, Y) = \hat{j}_M^G(\eta, \lambda X, Y).$$

Thus it suffices to prove that there exists $N \in \mathbb{N}$ such that if $\lambda \in F^{\times}$ satisfies $v_F(\lambda) < -N$, we have

$$\hat{j}_M^G(\eta, \lambda X, Y) = 0$$

Fix an \mathcal{O}_F -lattice $k_{\mathfrak{h}}$ (resp. $k_{\mathfrak{s}}$) of $\mathfrak{h}(F)$ (resp. $\mathfrak{s}(F)$). Denote by $\widetilde{k_{\mathfrak{s}}}$ the dual \mathcal{O}_F -lattice of $k_{\mathfrak{s}}$ in $\mathfrak{s}(F)$, i.e.,

$$\widetilde{k_{\mathfrak{s}}} := \{ Z \in \mathfrak{s}(F) : \forall Z' \in k_{\mathfrak{s}}, \Psi(\langle Z, Z' \rangle) = 1 \}.$$

Set

$$\mathfrak{c}(X) := \{ X' \in \mathfrak{c}(F) : \exists x \in H(F), \operatorname{Ad}(x)(X') = X \},\$$

which is a finite (perhaps empty) set. For $\lambda \in F^{\times}$, choose $h_{\lambda} \in \mathbb{N}$ such that both of the functions $\hat{j}_{M}^{G}(\eta, \lambda X, \cdot)$ and $|D^{\mathfrak{s}}(\cdot)|_{F}$ are constant on $Y + \varpi^{h_{\lambda}} k_{\mathfrak{s}}$.

Let f (resp. f') $\in C_c^{\infty}(\mathfrak{s}(F))$ be the characteristic function of $Y + \varpi^{h_{\lambda}} k_{\mathfrak{s}}$ (resp. $\varpi^{-h_{\lambda}} \widetilde{k_{\mathfrak{s}}}$). Then for $Z \in \mathfrak{s}(F)$, we see that

$$\begin{split} \hat{f}(Z) = & c_{\Psi}(\mathfrak{s}(F)) \int_{Y + \varpi^{h_{\lambda}} k_{\mathfrak{s}}} \Psi(\langle Z, Z' \rangle) dZ' = c_{\Psi}(\mathfrak{s}(F)) \Psi(\langle Z, Y \rangle) \int_{\varpi^{h_{\lambda}} k_{\mathfrak{s}}} \Psi(\langle Z, Z' \rangle) dZ' \\ = & c_{\Psi}(\mathfrak{s}(F)) \mathrm{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \Psi(\langle Z, Y \rangle) f'(Z). \end{split}$$

Now there are two expressions for $J_M^G(\eta, \lambda X, \hat{f})$. One the one hand, (10.1.1)

$$J_{M}^{G}(\eta, \lambda X, \hat{f}) = \int_{\mathfrak{s}(F)} f(Z) \hat{j}_{M}^{G}(\eta, \lambda X, Z) |D^{\mathfrak{s}}(Z)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{vol}(\varpi^{h_{\lambda}} k_{\mathfrak{s}}) \hat{j}_{M}^{G}(\eta, \lambda X, Y) |D^{\mathfrak{s}}(Y)|_{F}^{-1/2} dZ = \operatorname{v$$

On the other hand,

$$(10.1.2)$$

$$J_{M}^{G}(\eta, \lambda X, \hat{f}) = |D^{\mathfrak{s}}(\lambda X)|_{F}^{1/2} \int_{H_{\lambda X}(F) \setminus H(F)} \hat{f}(\operatorname{Ad}(x^{-1})(\lambda X))\eta(\operatorname{Nrd}(x))v_{M}^{G}(x)dx$$

$$= c_{\Psi}(\mathfrak{s}(F))\operatorname{vol}(\varpi^{h_{\lambda}}k_{\mathfrak{s}})|D^{\mathfrak{s}}(\lambda X)|_{F}^{1/2} \int_{H_{X}(F) \setminus H(F)} \Psi(\langle \operatorname{Ad}(x^{-1})(\lambda X), Y \rangle)f'(\operatorname{Ad}(x^{-1})(\lambda X))\eta(\operatorname{Nrd}(x))v_{M}^{G}(x)dx$$

Fix an open neighbourhood $\mathscr{V}_{\mathfrak{h}}$ of 0 in $\mathfrak{h}(F)$ which is invariant by the adjoint action of H(F) such that a homeomorphic exponential map is defined on $\mathscr{V}_{\mathfrak{h}}$. Choose $a \in \mathbb{N}$ verifying the following conditions:

- (1) $\varpi^a k_{\mathfrak{h}} \subseteq \mathscr{V}_{\mathfrak{h}};$ (2) $K_a := \exp(\varpi^a k_{\mathfrak{h}})$ is a subgroup of $K_H;$
- (3) $\eta(\operatorname{Nrd}(K_a)) = 1;$
- (4) the adjoint action of K_a stabilises $k_{\mathfrak{s}}$ (and thus $\widetilde{k_{\mathfrak{s}}}$).

Fix a set Γ of representatives in H(F) of double cosets $H_X(F) \setminus H(F)/K_a$. We may and shall suppose that if $x \in \Gamma$ and $y \in H_X(F)xK_a$ verify $\operatorname{Ad}(y^{-1})(X) \in \mathfrak{c}(F)$, then $\operatorname{Ad}(x^{-1})(X) \in \mathfrak{c}(F)$.

The integral in (10.1.2) can be decomposed as

$$\sum_{x\in\Gamma} \int_{H_X(F)\backslash H_X(F) \times K_a} \Psi(\langle \operatorname{Ad}(y^{-1})(\lambda X), Y \rangle) f'(\operatorname{Ad}(y^{-1})(\lambda X)) \eta(\operatorname{Nrd}(y)) v_M^G(y) dy$$

By the conditions (2), (3) and (4) on a respectively, the factors v_M^G , η and f' can be extracted from the integral. By comparing (10.1.1) and (10.1.2), since $\langle \cdot, \cdot \rangle$ is invariant by the adjoint action of K_a , we obtain

ſ

$$\begin{split} \hat{j}_{M}^{G}(\eta, \lambda X, Y) = & c_{\Psi}(\mathfrak{s}(F)) |D^{\mathfrak{s}}(\lambda X) D^{\mathfrak{s}}(Y)|_{F}^{1/2} \sum_{x \in \Gamma} f'(\operatorname{Ad}(x^{-1})(\lambda X)) \eta(\operatorname{Nrd}(x)) v_{M}^{G}(x) \int_{H_{X}(F) \setminus H_{X}(F) \times K_{a}} \\ & \Psi(\langle \operatorname{Ad}(y^{-1})(\lambda X), Y \rangle) dy \\ = & c_{\Psi}(\mathfrak{s}(F)) |D^{\mathfrak{s}}(\lambda X) D^{\mathfrak{s}}(Y)|_{F}^{1/2} \operatorname{vol}(K_{a})^{-1} \sum_{x \in \Gamma} \operatorname{vol}(H_{X}(F) \setminus H_{X}(F) x K_{a}) f'(\operatorname{Ad}(x^{-1})(\lambda X)) \\ & \eta(\operatorname{Nrd}(x)) v_{M}^{G}(x) i(x), \end{split}$$

where

(10.1.4)

$$i(x) := \int_{K_a} \Psi(\langle \operatorname{Ad}(x^{-1})(\lambda X), \operatorname{Ad}(y)(Y) \rangle) dy.$$

For $x \in \Gamma$, consider the map $K_a \to F$ defined by

$$\forall y \in K_a, y \mapsto \langle \operatorname{Ad}(x^{-1})(X), \operatorname{Ad}(y)(Y) \rangle.$$

Its differential at the point $y_0 \in K_a$ is the map $\mathfrak{h}(F) \to F$ defined by

(10.1.5)
$$\forall Z \in \mathfrak{h}(F), Z \mapsto \langle \operatorname{Ad}(x^{-1})(X), \operatorname{Ad}(y_0)([Z, Y]) \rangle$$

Since $\langle \cdot, \cdot \rangle$ is invariant by the adjoint action of G(F), we see that

$$\langle \operatorname{Ad}(x^{-1})(X), \operatorname{Ad}(y_0)([Z,Y]) \rangle = \langle [Y, \operatorname{Ad}(xy_0)^{-1}(X)], Z \rangle.$$

Because the restriction of $\langle \cdot, \cdot \rangle$ to $\mathfrak{h}(F)$ is non-degenerate, the map (10.1.5) is not surjective if and only if

$$[Y, \mathrm{Ad}(xy_0)^{-1}(X)] = 0$$

Since $Y \in \mathfrak{c}_{reg}(F)$, this condition is equivalent to

$$\operatorname{Ad}(xy_0)^{-1}(X) \in \mathfrak{c}(F).$$

From our choice of Γ , as $y_0 \in K_a$, it implies that

$$\operatorname{Ad}(x^{-1})(X) \in \mathfrak{c}(F).$$

Let

$$\Gamma' := \{ x \in \Gamma : \operatorname{Ad}(x^{-1})(X) \in \mathfrak{c}(F) \}$$

which is a finite (perhaps empty) subset of Γ . Then for $x \in \Gamma - \Gamma'$, the map (10.1.4) is a submersion. Define

$$\Omega := \bigcup_{x \in \Gamma - \Gamma'} H_X(F) x K_a,$$

which is an open and closed subset of H(F). Fix a basis of the *F*-linear space $\mathfrak{s}(F)$. For $Z \in \mathfrak{s}(F)$, define its norm $||Z|| \in \mathbb{R}_{\geq 0}$ as the maximum of normalised absolute values of coefficients of *Z* with respect to the fixed basis. For $Z \in \mathfrak{s}(F) - \{0\}$, define $\nu(Z) \in \mathbb{Z}$ by $||Z|| = |\varpi^{\nu(Z)}|_F$. Let S_X be the closure of

$$S_X^0 := \{ \varpi^{-\nu(\operatorname{Ad}(y^{-1})(X))} \operatorname{Ad}(y^{-1})(X) : y \in \Omega \}$$

in the unit sphere $S_{\mathfrak{s}} := \{Z \in \mathfrak{s}(F) : ||Z|| = 1\}$. Then S_X is compact. Recall that we denote by $\mathcal{N}^{\mathfrak{s}}$ the set of nilpotent elements in $\mathfrak{s}(F)$.

LEMMA 10.3. We have

$$S_X - S_X^0 \subseteq \mathcal{N}^{\mathfrak{s}} - \{0\}$$

PROOF OF LEMMA 10.3. Since $S_X \subseteq S_s$, it is obvious that $\{0\} \notin S_X$. Let $Z \in S_X$. There exists a sequence $\{y_i\}$ in Ω such that when $i \to \infty$,

$$\varpi^{-\nu(\operatorname{Ad}(y_i^{-1})(X))}\operatorname{Ad}(y_i^{-1})(X) \to Z.$$

We distinguish two cases.

i) Suppose that the sequence $\{\|\operatorname{Ad}(y_i^{-1})(X)\|\}$ remains bounded. By Harish-Chandra's compactness lemma for symmetric spaces (Lemma 3.2), the projection of the sequence $\{y_i\}$ to $H_X(F)\setminus H(F)$ is contained in a compact subset. By taking a subsequence, since the projection of Ω to $H_X(F)\setminus H(F)$ is closed, we may assume that when $i \to \infty$, $\operatorname{Ad}(y_i^{-1})(X) \to \operatorname{Ad}(y^{-1})(X)$ with $y \in \Omega$. Thus $Z \in S_X^0$ in this case.

ii) Suppose that the sequence $\{\|\operatorname{Ad}(y_i^{-1})(X)\|\}$ is unbounded. By taking a subsequence, we may assume that when $i \to \infty$, $\|\operatorname{Ad}(y_i^{-1})(X)\| \to +\infty$. The eigenvalues of $\operatorname{ad}(\operatorname{Ad}(y_i^{-1})(X))$ are the same as those of $\operatorname{ad}(X)$; here $\operatorname{ad}(\operatorname{Ad}(y_i^{-1})(X))$ and $\operatorname{ad}(X)$ are viewed as linear endomorphisms of \mathfrak{g} . Thus the eigenvalues of $\operatorname{ad}(\varpi^{-\nu(\operatorname{Ad}(y_i^{-1})(X))}\operatorname{Ad}(y_i^{-1})(X))$ tend to zero when $i \to \infty$. Hence $\operatorname{ad}(Z)$ is nilpotent. We shall prove that $Z \in \mathcal{N}^{\mathfrak{s}}$ in this case.

Since \mathfrak{g} is reductive, one has $\mathfrak{g} = \mathfrak{z} \oplus \mathfrak{g}_{der}$, where \mathfrak{z} denotes the centre of \mathfrak{g} and \mathfrak{g}_{der} denotes the derived algebra of \mathfrak{g} , and \mathfrak{g}_{der} is semisimple. Let $Z = Z_1 + Z_2$ with $Z_1 \in \mathfrak{z}(F)$ and $Z_2 \in \mathfrak{g}_{der}(F)$. Since $\mathrm{ad}(Z)$ is nilpotent as a linear endomorphism of \mathfrak{g} , we deduce that $\mathrm{ad}(Z_2)$ is nilpotent as a linear endomorphism of \mathfrak{g}_{der} . As \mathfrak{g}_{der} is semisimple, we obtain that Z_2 is a nilpotent element in \mathfrak{g} . Let $X = X_1 + X_2$ with $X_1 \in \mathfrak{z}(F)$ and $X_2 \in \mathfrak{g}_{der}(F)$. The projection of $\varpi^{-\nu(\mathrm{Ad}(y_i^{-1})(X))}\mathrm{Ad}(y_i^{-1})(X)$ to $\mathfrak{z}(F)$ is equal to $\varpi^{-\nu(\mathrm{Ad}(y_i^{-1})(X))}X_1$, which tends to zero when $i \to \infty$. Thus $Z_1 = 0$, and $Z = Z_2$ is a nilpotent element in \mathfrak{g} . Hence $Z \in \mathcal{N}^{\mathfrak{s}}$.

For $U \in \mathcal{N}^{\mathfrak{s}} - \{0\}$, consider the map $K_a \to F$ defined by

(10.1.6)
$$\forall y \in K_a, y \mapsto \langle U, \operatorname{Ad}(y)(Y) \rangle$$

Its differential at the point $y_0 \in K_a$ is the map $\mathfrak{h}(F) \to F$ defined by

$$\forall Z \in \mathfrak{h}(F), Z \mapsto \langle U, \mathrm{Ad}(y_0)([Z, Y]) \rangle = \langle [Y, \mathrm{Ad}(y_0^{-1})(U)], Z \rangle$$

by the G(F)-invariance of $\langle \cdot, \cdot \rangle$. Since $Y \in \mathfrak{c}_{\mathrm{reg}}(F)$ and $\mathrm{Ad}(y_0^{-1})(U) \in \mathcal{N}^{\mathfrak{s}} - \{0\}$, we have

 $[Y, \operatorname{Ad}(y_0^{-1})(U)] \neq 0.$

Then the map (10.1.6) is a submersion by the non-degeneration of $\langle \cdot, \cdot \rangle$ on $\mathfrak{h}(F)$.

Using Lemma 10.3 and combining our discussion on the maps (10.1.4) and (10.1.6), we deduce that there exists an open compact neighbourhood \widetilde{S}_X of S_X in $S_{\mathfrak{s}}$ such that the map $\varphi: K_a \times \widetilde{S}_X \to F \times \widetilde{S}_X$ defined by

$$\forall (y, Z) \in K_a \times S_X, (y, Z) \mapsto (\langle Z, \operatorname{Ad}(y)(Y) \rangle, Z)$$

is a submersion. Since any submersion is open, the image of φ (denoted by $\operatorname{Im}(\varphi)$) is an open compact subset of $F \times \widetilde{S}_X$. Then the map φ induces a surjective submersion $\varphi' : K_a \times \widetilde{S}_X \to \operatorname{Im}(\varphi)$. Applying Harish-Chandra's submersion principle [26, Theorem 11] to φ' , there exists a function $\phi \in \mathcal{C}_c^{\infty}(\operatorname{Im}(\varphi))$ such that for all $\Phi' \in \mathcal{C}_c^{\infty}(\operatorname{Im}(\varphi))$,

$$\int_{K_a \times \widetilde{S_X}} \Phi'(\langle Z, \operatorname{Ad}(y)(Y) \rangle, Z) dZ dy = \int_{\operatorname{Im}(\varphi)} \phi(t, Z) \Phi'(t, Z) dZ dt.$$

Fix such a ϕ . Denote by $\mathcal{C}^{\infty}(F \times \widetilde{S_X})$ the space of locally constant, complexed-valued functions on $F \times \widetilde{S_X}$. For $\Phi \in \mathcal{C}^{\infty}(F \times \widetilde{S_X})$, the restriction of Φ to $\operatorname{Im}(\varphi)$ belongs to $\mathcal{C}^{\infty}_c(\operatorname{Im}(\varphi))$, so we obtain

$$\int_{K_a \times \widetilde{S_X}} \Phi(\langle Z, \operatorname{Ad}(y)(Y) \rangle, Z) dZ dy = \int_{F \times \widetilde{S_X}} \phi(t, Z) \Phi(t, Z) dZ dt.$$

By taking $\Phi(t, Z) := \Psi(\mu t)\beta(Z)$ with $\mu \in F$ and $\beta \in \mathcal{C}^{\infty}_{c}(\widetilde{S}_{X})$, we deduce that for all $Z \in \widetilde{S}_{X}$,

$$\int_{K_a} \Psi(\mu \langle Z, \mathrm{Ad}(y)(Y) \rangle) dy = \int_F \phi(t, Z) \Psi(\mu t) dt$$

Since $\operatorname{Im}(\phi)$ is an open compact subset of $F \times \widetilde{S}_X$, we see that $\phi \in \mathcal{C}_c^{\infty}(\operatorname{Im}(\varphi)) \subseteq \mathcal{C}_c^{\infty}(F \times \widetilde{S}_X) = \mathcal{C}_c^{\infty}(F) \otimes \mathcal{C}_c^{\infty}(\widetilde{S}_X)$. Suppose that $\phi = \sum_{1 \leq j \leq m} c_j \cdot \xi_j \otimes \chi_j$ with $c_j \in \mathbb{C}, \xi_j \in \mathcal{C}_c^{\infty}(F)$ and $\chi_j \in \mathcal{C}_c^{\infty}(\widetilde{S}_X)$. Then

$$\int_{F} \phi(t, Z) \Psi(\mu t) dt = \sum_{1 \le j \le m} c_j \hat{\xi}_j(\mu) \chi_j(Z),$$

where $\hat{\xi}_j \in \mathcal{C}_c^{\infty}(F)$ is the Fourier transform of ξ_j . We see that there exists $N_0 \in \mathbb{N}$ such that for all $\mu \in F^{\times}$ satisfying $v_F(\mu) < -N_0$ and all $Z \in S_X$, we have

$$\int_{K_a} \Psi(\mu \langle Z, \operatorname{Ad}(y)(Y) \rangle) dy = 0.$$

Fix such an N_0 .

Recall that for $x \in \Gamma$,

$$i(x) = \int_{K_a} \Psi(\mu \langle Z, \operatorname{Ad}(y)(Y) \rangle) dy,$$

where $\mu := \lambda \varpi^{\nu(\operatorname{Ad}(x^{-1})(X))}$ and $Z := \varpi^{-\nu(\operatorname{Ad}(x^{-1})(X))} \operatorname{Ad}(x^{-1})(X)$. For $x \in \Gamma - \Gamma'$, we have $Z \in S_X$, so i(x) = 0 if $v_F(\lambda) + \nu(\operatorname{Ad}(x^{-1})(X)) < -N_0.$

Set

$$\nu_0 := \sup \nu(\operatorname{Ad}(x^{-1})(X)),$$

which is finite thanks to Harish-Chandra's compactness lemma for symmetric spaces (Lemma 3.2). Now let

$$N := N_0 + \nu_0$$

Suppose that $v_F(\lambda) < -N$. From (10.1.3), to show $\hat{j}_M^G(\eta, \lambda X, Y) = 0$, it suffices to prove $v_M^G(x) = 0$ for all $x \in \Gamma'$.

For $x \in \Gamma'$, we have $\operatorname{Ad}(x^{-1})(X) \in \mathfrak{c}_{\operatorname{reg}}(F)$. Then $\operatorname{Ad}(x^{-1})(H_X) = T_{\mathfrak{c}}$. Since $X \in (\mathfrak{m} \cap \mathfrak{s}_{\operatorname{rs}})(F)$, we see that $\operatorname{Ad}(x^{-1})(A_M)$ is an *F*-split torus in $T_{\mathfrak{c}}$. As $\mathfrak{c} \subseteq \mathfrak{l} \cap \mathfrak{s}$ is *L*-elliptic, A_L is the maximal *F*-split torus in $T_{\mathfrak{c}}$. Thus $\operatorname{Ad}(x^{-1})(A_M) \subseteq A_L$. Then $A_M \subseteq \operatorname{Ad}(x)(A_L) \subseteq \operatorname{Ad}(x)(A_{M_0})$. We deduce that $\operatorname{Ad}(x)(A_{M_0})$ is a maximal *F*-split torus in M_H , so it is $M_H(F)$ -conjugate to A_{M_0} . Therefore, $x \in M_H(F)\operatorname{Norm}_{H(F)}(M_0) \subseteq M_H(F)K_H$. Consequently, we have $v_M^G(x) = 0$ and conclude. \Box

10.2. The case of (G', H').

PROPOSITION 10.4. Let $M' \in \mathscr{L}^{H'}(M'_0), M' \neq H'$. Let $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ and $X \in \mathfrak{s}'_{rs}(F)$. Then there exists $N \in \mathbb{N}$ such that if $\lambda \in F^{\times}$ satisfies $v_F(\lambda) < -N$, we have

$$\hat{i}_{M'}^{H'}(\lambda Y, X) = 0.$$

PROOF. It is almost the same as the proof of Proposition 10.1, except that one needs to use Lemma 8.8.2) and Lemma 8.7. $\hfill \Box$

CHAPTER 5

On certain identities between Fourier transforms of weighted orbital integrals on infinitesimal symmetric spaces of Guo-Jacquet

In an infinitesimal variant of Guo-Jacquet trace formulae, the regular semi-simple terms are expressed as noninvariant weighted orbital integrals on two global infinitesimal symmetric spaces. We prove some relations between the Fourier transforms of invariant weighted orbital integrals on the corresponding local infinitesimal symmetric spaces. These relations should be useful in the noninvariant comparison of the infinitesimal variant of Guo-Jacquet trace formulae.

1. Introduction

Inspired by Jacquet's new proof [29] of Waldspurger's well-known result [50] on the central values of automorphic L-functions for GL_2 , Guo-Jacquet have suggested comparison of two relative trace formulae in [23] in order to generalise this theorem to higher ranks. This approach has also been followed by Feigon-Martin-Whitehouse [21] via a simple trace formula. However, if one wants to remove the restrictive conditions in [21], some additional terms in the Guo-Jacquet trace formula other than relative orbital integrals can not be neglected.

Our starting point is an infinitesimal analogue of Guo-Jacquet trace formulae and their comparison. It means that we first work on the tangent space of a symmetric space (called an infinitesimal symmetric space). A reason for this is that at the infinitesimal level, the spectral side of the relative trace formula is replaced by the Fourier transform of the geometric side where the harmonic analysis is simpler. Another reason is that the comparison of trace formulae for infinitesimal symmetric spaces should imply the comparison of the original relative trace formulae for symmetric spaces. For example, one may consult Zhang's proof of the transfer of relative local orbital integrals [58].

We have established an infinitesimal variant of Guo-Jacquet trace formulae in Chapter 2 and Chapter 3, where the main (namely regular semisimple) terms are explicit weighted orbital integrals. These distributions should be the first ones to be studied and compared after orbital integrals. However, some new difficulties arise since these distributions are noninvariant. Instead of making the trace formula invariant as Arthur did (see [5] and others), we would like to follow Labesse's proposal [37] of noninvariant comparison which seems more direct. For example, we have established the weighted fundamental lemma for infinitesimal Guo-Jacquet trace formulae in Chapter 3 as a noninvariant and infinitesimal avatar of Guo's fundamental lemma [23]. The strategy of noninvariant comparison has been also adopted in [14] and [15] on the stable base change. These works provide some indications to our work.

Let us recall some basic objects in the local setting. Let E/F be a quadratic extension of local fields of characteristic zero. Let η be the quadratic character of F/NE^{\times} attached to E/F, where NE^{\times} denotes the norm of E^{\times} . The first symmetric pair is $(G, H) = (GL_{2n}, GL_n \times GL_n)$. Let $\mathfrak{s} \simeq \mathfrak{gl}_n \oplus \mathfrak{gl}_n$ be the corresponding infinitesimal symmetric space. Denote by \mathfrak{s}_{rs} the set of regular semi-simple elements in \mathfrak{s} (see Section 2.2). Let M be an ω -stable (see Section 2.3) Levi subgroup of G, and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Let f be a locally constant and compactly supported function on $\mathfrak{s}(F)$. We define the weighted orbital integral $J_M^G(\eta, X, f)$ by (2.3.2). We have proved in Chapter 4 that its Fourier transform is represented by a locally constant function $\hat{j}_M^G(\eta, X, \cdot)$ on $\mathfrak{s}_{rs}(F)$. We have also defined the (H, η) -invariant weighted orbital integrals $I_M^G(\eta, X, f)$ in *loc. cit.* by Arthur's standard method, whose Fourier transform is represented by a locally constant function $\hat{i}_M^G(\eta, X, \cdot)$ on $\mathfrak{s}_{rs}(F)$. The second symmetric pair is (G', H'), where G' is the group of invertible elements in a central simple algebra over F containing E, and H' is the centraliser of E^{\times} in G'. It is inspired by the related local conjecture of Prasad and Takloo-Bighash [44] and more general than Guo-Jacquet's original setting. Denote by \mathfrak{s}' the corresponding infinitesimal symmetric space. For a Levi subgroup M' of H' and $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ (see Section 2.4), we similarly obtain local constant functions $\hat{j}_{M'}^{H'}(Y, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}_{rs}(F)$.

The functions $\hat{j}_M^G(\eta, X, \cdot)$ is decomposed as their invariant analogues $\hat{i}_M^G(\eta, X, \cdot)$ and weight functions v_M^G . The decomposition for the functions $\hat{j}_{M'}^{H'}(Y, \cdot)$ is similar. In order to obtain relations between $\hat{j}_M^G(\eta, X, \cdot)$ and $\hat{j}_{M'}^{H'}(Y, \cdot)$, which is part of the noninvariant comparison of the infinitesimal variant of Guo-Jacquet trace formulae, we shall focus on the relations between $\hat{i}_M^G(\eta, X, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ in this paper.

There is an injection $M' \mapsto M$ from the set of Levi subgroups of H' into the set of ω -stable Levi subgroups of G (see Section 4). We fix such a matching pair of Levi subgroups. We define the notion of matching orbits between $\mathfrak{s}_{\rm rs}(F)$ and $\mathfrak{s}'_{\rm rs}(F)$ by Definition 4.1. For $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{\rm rs}(F)$, we denote $\eta(X) := \eta(\det(AB))$. Our main result is as follows.

THEOREM 1.1 (see Corollary 5.6 and Proposition 5.9). 1) Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ have *M*-matching orbits. Let $U \in \mathfrak{s}_{rs}(F)$ and $V \in \mathfrak{s}'_{rs}(F)$ have matching orbits. Then we have the equality

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X)\kappa(U)\hat{i}_{M}^{G}(\eta, X, U) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{i}_{M'}^{H'}(Y, V),$$

where $\gamma_{\psi}(\mathfrak{h}(F))$ and $\gamma_{\psi}(\mathfrak{h}'(F))$ are Weil constants (see Section 2.2).

2) Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in \mathfrak{s}_{rs}(F)$. If $\eta(X) \neq \eta(U)$, then

$$\hat{i}_M^G(\eta, X, U) = 0$$

This theorem generalises some of the main results in [58] to the weighted context. As in *loc. cit.*, we use Waldspurger's global method on the endoscopic transfer [52] to show 1) and a local method to show 2). To show 1), we define a notion of matching weighted orbital integrals (see Definition 5.2) and prove that this property commutes with Fourier transform under some restriction (see Theorem 5.3). Then we may extract the relations between $\hat{i}_M^G(\eta, X, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ with the help of Labesse's lemma [37, Lemma 1.7.1]. These steps are close to those in [14]. However, there is an important distinction. While the weighted fundamental lemma for inner forms is tautological in *loc. cit.*, the vanishing condition of Lemma 8.1 here is more subtle. It makes the comparison of global trace formulae by Waldspurger's method, which is a simple case of the noninvariant comparison, even trickier. We translate our definition of matching orbits into the language of cohomology (see Sections 4.3 and 4.4) and use abelian Galois cohomology (see [38]) to go through some technical difficulties.

This paper is organised as follows. We introduce some notations and recall some preliminaries in Section 2. Then we recall Labesse's lemma in Section 3. We define the notion of matching orbits and give a cohomological criterion in Section 4. Our main results are stated in Section 5. The rest of the paper is devoted to the proof Proposition 5.5 by Waldspurger's global method. We recall limit formulae of $\hat{i}_M^G(\eta, X, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$, the weighted fundamental lemma and an infinitesimal variant of Guo-Jacquet trace formulae in Sections 6, 8 and 10 respectively. We explain the construction of test functions and the globalisation of local data in Sections 7 and 9. These results are prepared for our final proof in Section 11.

2. Notation and preliminaries

2.1. Groups. Let F be a local field of characteristic zero or a number field. Denote by \mathcal{O}_F the ring of integers of F. Let E be a quadratic extension of F. If F is a local (resp. global) field, denote by η the quadratic character of F^{\times}/NE^{\times} (resp. $\mathbb{A}^{\times}/F^{\times}$) attached to E/F, where $NE^{\times} = N_{E/F}E^{\times}$ denotes the norm of E^{\times} in F^{\times} (resp. $\mathbb{A} = \mathbb{A}_F$ denotes the ring of adèles of F).

Let G be a reductive group over F. Denote by $\operatorname{rk}_F(G)$ the F-rank of G. Let \widehat{G} be the Langlands dual of G which is a complex reductive group. All algebraic groups (except \widehat{G}) and varieties are assumed to be defined over F until further notice. Denote by G_{ad} the adjoint group of G, by G_{der} the derived subgroup of G and by G_{sc} the simple connected cover of G_{der} . Denote by Z_G the centre of G and by $C_G := G/G_{\operatorname{der}}$ the cocentre of G. Fix an algebraic closure \overline{F} of F. Let $\Gamma := \operatorname{Gal}(\overline{F}/F)$. For an F-variety V, we sometimes abuse notation and also write V for $V(\overline{F})$ when there is no confusion.

We use a minuscule Fraktur letter to denote the Lie algebra of its corresponding algebraic group. For example, we write $\mathfrak{g} := \text{Lie}(G)$. Denote by Ad the adjoint action of G on itself or \mathfrak{g} . If G acts on an F-variety V and $X \in V(F)$, denote by G_X the centraliser of X in G. If \mathfrak{v} is an F-subvariety of \mathfrak{g} , denote by \mathfrak{v}_X the centraliser of $X \in \mathfrak{g}(F)$ in \mathfrak{v} . If θ is an automorphism on G, denote by G^{θ} the subgroup of fix points of G under θ .

Fix a Levi *F*-factor M_0 of a minimal parabolic *F*-subgroup of *G*. By a Levi subgroup of *G*, we mean a Levi *F*-factor of some parabolic *F*-subgroup of *G*. For a semi-standard (namely containing M_0) Levi subgroup *M* of *G*, denote by $\mathscr{F}^G(M)$, $\mathscr{P}^G(M)$ and $\mathscr{L}^G(M)$ the sets of parabolic *F*-subgroups of *G* containing *M*, parabolic *F*-subgroups of *G* with Levi factor *M* and Levi subgroups of *G* containing *M* respectively. For $P \in \mathscr{F}^G(M_0)$, denote by M_P the unique Levi factor containing M_0 and by N_P the unipotent radical. Let \overline{P} be the parabolic subgroup opposite to *P*.

For $M \in \mathscr{L}^G(M_0)$, define the Weyl group of (G, M) by

$$W^G(M) := \operatorname{Norm}_{G(F)}(M)/M(F).$$

In particular, we also write $W_0^G := W^G(M_0)$. For $M, L \in \mathscr{L}^G(M_0)$, denote

$$\operatorname{Tran}_G(M,L) := \{ w \in W_0^L \setminus W_0^G : \operatorname{Ad}(w)(M) \subseteq L \}.$$

Denote by A_G the maximal *F*-split central torus of *G*. Let $X(G)_F$ be the group of *F* rational characters of *G*. Define the \mathbb{R} -linear space

$$\mathfrak{a}_G := \operatorname{Hom}_{\mathbb{Z}}(X(G)_F, \mathbb{R}),$$

whose dual space is denoted by \mathfrak{a}_G^* . Fix a scalar product on \mathfrak{a}_{M_0} which is invariant under the action of W_0^G , from which we deduce Haar measures on all subspaces of \mathfrak{a}_{M_0} . Denote by \mathfrak{a}_M^G the orthogonal complement of \mathfrak{a}_G in \mathfrak{a}_M .

Let D a central division algebra over F. Denote by $\deg(D)$ the degree of D, i.e., $\dim_F(D) = \deg(D)^2$. Denote by $GL_{n,D}$ the reductive group over F whose F-points are $GL_n(D)$. For $x \in \mathfrak{gl}_n(D)$, we write $\operatorname{Nrd}(x)$, $\operatorname{Trd}(x)$ and Prd_x for its reduced norm, reduced trace and reduced characteristic polynomial respectively. If D = F, we also write them as $\det(x)$, $\operatorname{Tr}(x)$ and χ_x respectively.

Now suppose that F is a local field of characteristic zero. Denote by $|\cdot|_F$ the normalised absolute value on F. Define a homomorphism $H_G: G(F) \to \mathfrak{a}_G$ by

$$\langle H_G(x), \chi \rangle = \log(|\chi(x)|_F)$$

for all $x \in G(F)$ and $\chi \in X(G)_F$. Fix a maximal compact subgroup K of G(F) which is admissible relative to M_0 in the sense of [5, p. 9]. In this paper, we choose the standard maximal compact subgroup when $G(F) = GL_n(D)$, where D is a central division algebra over a finite field extension of F. That is to say, if F is non-archimedean, $K = GL_n(\mathcal{O}_D)$ with \mathcal{O}_D being the ring of integers of D (see [54, p. 191]), while if F is archimedean, K is the unitary group with respect to some hermitian form (see [54, p. 199]). We may extend the function H_M to a map $H_P : G(F) \to \mathfrak{a}_{M_P}$ using the decomposition $G(F) = M_P(F)N_P(F)K$.

Fix the Haar measure on K such that vol(K) = 1. For $P \in \mathscr{F}^G(M_0)$, fix a Haar measure on $N_P(F)$ such that

$$\int_{N_P(F)} \exp(2\rho_{\overline{P}}(H_{\overline{P}}(n))) dn = 1,$$

where $\rho_{\overline{P}}$ is the half of the sum of roots (with multiplicity) associated to \overline{P} . For $M \in \mathscr{L}^G(M_0)$, there are compatible Haar measures on G(F) and M(F) in the sense of [8, (1.1), p. 12] such that for all $P \in \mathscr{P}^G(M)$ and all continuous and compactly supported function f on G(F), we have the equality

$$\int_{G(F)} f(x)dx = \int_{M(F) \times N_P(F) \times K} f(mnk)dkdndm$$

We shall choose such measures.

Let V be an F-linear space of finite dimension. If F is non-archimedean, denote by $\mathcal{C}_c^{\infty}(V) = \mathcal{S}(V)$ the space of locally constant, compactly supported and complex-valued functions on V. If F is archimedean, denote by $\mathcal{S}(V)$ the space of Schwartz functions on V. For $f \in \mathcal{S}(V)$, denote by $\operatorname{Supp}(f)$ its support.

Fix a continuous and nontrivial unitary character $\psi : F \to \mathbb{C}^{\times}$. Let $\langle \cdot, \cdot \rangle$ be a non-degenerate symmetric bilinear form on $\mathfrak{g}(F)$ which is invariant under conjugation. Let V be an F-linear subspace of $\mathfrak{g}(F)$, on which the restriction of $\langle \cdot, \cdot \rangle$ is non-degenerate. It is equipped with the unique self-dual Haar measure with respect to $\psi(\langle \cdot, \cdot \rangle)$. For $f \in \mathcal{S}(V)$, define its Fourier transform $\hat{f} \in \mathcal{S}(V)$ by

$$\forall X \in V, \hat{f}(X) := \int_{V} f(Y)\psi(\langle X, Y \rangle)dY.$$

Let
$$M \in \mathscr{L}^G(M_0)$$
 and $Q \in \mathscr{F}^G(M)$. By [5, p. 40-41], for $x \in G(F)$,
 $v_P(\lambda, x) := e^{-\lambda(H_P(x))}, \forall \lambda \in \mathfrak{ia}_M^*, P \in \mathscr{P}^G(M)$,

is a (G, M)-family in the sense of [5, p. 36]. Define the weight function

(2.1.1)
$$v_M^Q(x) := \lim_{\lambda \to 0} \sum_{\{P \in \mathscr{P}^G(M) : P \subseteq Q\}} v_P(\lambda, x) \theta_P^Q(\lambda)^{-1}, \forall x \in G(F).$$

Let V be an F-linear space of finite dimension equipped with a non-degenerate symmetric bilinear form $q(\cdot, \cdot)$ and a Haar measure. Denote by $\gamma_{\psi}(q)$ the Weil constant given in [53, Théorème 2].

2.2. Symmetric pairs. Let F be a local field of characteristic zero or a number field. A symmetric pair in the sense of [1, Definition 7.1.1] is a triple (G, H, θ) where $H \subseteq G$ are a pair of reductive groups, and θ is an involution of G such that $H = G^{\theta}$. Let \mathfrak{s} be the tangent space at the neutral element of the symmetric space S := G/H. We shall always view \mathfrak{s} as a subspace of \mathfrak{g} . Thus

$$\mathfrak{s} = \{ X \in \mathfrak{g} : (d\theta)(X) = -X \}$$

on which H acts by the restriction of Ad. By [1, Lemma 7.1.9], there exists a G-invariant θ -invariant non-degenerate symmetric bilinear form on \mathfrak{g} .

An element $X \in \mathfrak{s}$ is said to be semi-simple if $\operatorname{Ad}(H)(X)$ is Zariski closed in \mathfrak{s} . If F is a local field of characteristic zero, $X \in \mathfrak{s}(F)$ is semi-simple if and only if $\operatorname{Ad}(H(F))(X)$ is closed in $\mathfrak{s}(F)$ in the analytic topology by [45, Fact A, p. 108-109]. We say an element $X \in \mathfrak{s}$ is regular if H_X has minimal dimension. Denote by \mathfrak{s}_{rs} the subset of \mathfrak{s} consisting of regular semi-simple elements in \mathfrak{s} .

Now suppose that F is a local field of characteristic zero. A Cartan subspace of \mathfrak{s} is defined as a maximal abelian subspace $\mathfrak{c} \subseteq \mathfrak{s}$ defined over F consisting of semi-simple elements. Denote by $\mathscr{T}^{\mathfrak{s}}$ the set of Cartan subspaces of \mathfrak{s} . Fix a (finite) set of representatives $\mathscr{T}_0^{\mathfrak{s}}$ for H(F)-conjugacy classes in $\mathscr{T}^{\mathfrak{s}}$. Let $\mathfrak{c} \in \mathscr{T}^{\mathfrak{s}}$. Denote by $T_{\mathfrak{c}}$ the centraliser of \mathfrak{c} in H, which is a torus. Define the Weyl group

$$W(H, \mathfrak{c}) := \operatorname{Norm}_{H(F)}(\mathfrak{c})/T_{\mathfrak{c}}(F)$$

For $\mathfrak{c}_1, \mathfrak{c}_2 \in \mathscr{T}^{\mathfrak{s}}$, denote by $W(H, \mathfrak{c}_1, \mathfrak{c}_2)$ the set of isomorphisms from \mathfrak{c}_1 onto \mathfrak{c}_2 induced by $\operatorname{Ad}(x)$ for some $x \in H(F)$. If $\mathfrak{c}_1 = \mathfrak{c}_2$, we see that $W(H, \mathfrak{c}_1, \mathfrak{c}_1)$ is nothing but $W(H, \mathfrak{c}_1)$ (viewed as a set).

For $\mathfrak{c} \in \mathscr{T}^{\mathfrak{s}}$ and $X \in (\mathfrak{c} \cap \mathfrak{s}_{rs})(F)$, define the Weyl discriminant factor

$$|D^{\mathfrak{s}}(X)|_F := |\det(\mathrm{ad}(X)|_{\mathfrak{h}/\mathfrak{t}_{\mathfrak{c}}\oplus\mathfrak{s}/\mathfrak{c}})|_F^{1/2}$$

Let $\langle \cdot, \cdot \rangle$ be a *G*-invariant θ -invariant non-degenerate symmetric bilinear form on \mathfrak{g} . For any *F*-linear subspace \mathfrak{v} of $\mathfrak{g}(F)$ such that the restriction of $\langle \cdot, \cdot \rangle$ on \mathfrak{v} is non-degenerate, denote by $\gamma_{\psi}(\mathfrak{v})$ the Weil constant associated to \mathfrak{v} . Let $\mathfrak{c} \in \mathscr{T}^{\mathfrak{s}}$. For $X, Y \in (\mathfrak{c} \cap \mathfrak{s}_{rs})(F)$, define a bilinear form $q_{X,Y}$ on $\mathfrak{h}(F)/\mathfrak{t}_{\mathfrak{c}}(F)$ by

$$q_{X,Y}(Z,Z') := \langle [Z,X], [Y,Z'] \rangle$$

It is non-degenerate and symmetric and we have $q_{X,Y} = q_{Y,X}$. Write

(2.2.1)
$$\gamma_{\psi}(X,Y) := \gamma_{\psi}(q_{X,Y}).$$

2.3. The case of (G, H). Let F be a local field of characteristic zero or a number field. Let $G := GL_{2n}$ and denote by $H := GL_n \times GL_n$ its subgroup via diagonal embedding. In fact, H is the subgroup of fixed points of the involution $\operatorname{Ad}(\epsilon)$ on G, where $\epsilon := \begin{pmatrix} 1_n \\ -1_n \end{pmatrix}$. We shall embed G into \mathfrak{g} in the standard way. For an F-subvariety \mathfrak{v} of \mathfrak{g} , we write $\mathfrak{v}^{\times} := \mathfrak{v} \cap G$. Recall that $\mathfrak{s}_{rs} \subseteq \mathfrak{s}^{\times}$ in our case. Let $\langle \cdot, \cdot \rangle$ be the non-degenerate symmetric bilinear form on $\mathfrak{g}(F)$ defined by

(2.3.1)
$$\langle X, Y \rangle := \operatorname{Tr}(XY), \forall X, Y \in \mathfrak{g}(F),$$

which is invariant by the adjoint action of G(F) and $Ad(\epsilon)$.

Let M_0 be the group of diagonal matrices in G. Set $\omega := \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix} \in G(F)$. For $P \in \mathscr{F}^G(M_0)$, we say that P is " ω -stable" if $\omega \in P$. Denote by $\mathscr{F}^{G,\omega}(M_0)$ the subset in $\mathscr{F}^G(M_0)$ consisting of ω -stable parabolic subgroups. For $M \in \mathscr{L}^G(M_0)$, we say that M is " ω -stable" if $M = M_P$ for some $P \in \mathscr{F}^{G,\omega}(M_0)$. Denote by $\mathscr{L}^{G,\omega}(M_0)$ the subset in $\mathscr{L}^G(M_0)$ consisting of ω -stable Levi subgroups. Let A_n be the group of diagonal matrices in GL_n . Recall that there is a bijection between $\mathscr{L}^{GL_n}(A_n)$ and $\mathscr{L}^{G,\omega}(M_0)$ induced by $M_n \mapsto M = \begin{pmatrix} \mathfrak{m}_n & \mathfrak{m}_n \\ \mathfrak{m}_n & \mathfrak{m}_n \end{pmatrix}^{\times}$. We shall always write M_n for the preimage of M under this bijection. Notice that if $M \in \mathscr{L}^{G,\omega}(M_0)$ and $Q \in \mathscr{F}^G(M)$, then $Q \in \mathscr{F}^{G,\omega}(M_0)$. There is also a bijection between $\mathscr{F}^{GL_n}(A_n)$ and $\mathscr{F}^{G,\omega}(M_0)$ induced by $P_n \mapsto P = \begin{pmatrix} \mathfrak{p}_n & \mathfrak{p}_n \\ \mathfrak{p}_n & \mathfrak{p}_n \end{pmatrix}^{\times}$. We shall always write P_n for the preimage of P under this bijection. For $\mathfrak{c} \in \mathscr{T}^{\mathfrak{s}}$ and $L \in \mathscr{L}^{G,\omega}(M_0)$, define

$$W(H,\mathfrak{c},\mathfrak{l}\cap\mathfrak{s}):=\bigsqcup_{\mathfrak{c}_{2}\in\mathscr{T}_{0}^{\mathfrak{l}\cap\mathfrak{s}}}W(H,\mathfrak{c},\mathfrak{c}_{2}).$$

Let $M \in \mathscr{L}^{G,\omega}(M_0)$. We say an element $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ is elliptic if A_M is the maximal F-split torus in H_X . Denote by $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{\text{ell}}$ the set of elliptic elements in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Write $M_H := M \cap H$. Denote by $\Gamma_{\text{ell}}((\mathfrak{m} \cap \mathfrak{s}_{rs})(F))$ the set of $M_H(F)$ -conjugacy classes in $(\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{\text{ell}}$. For $X \in \mathfrak{s}_{rs}(F)$, $X \in \mathfrak{s}_{rs}(F)_{\text{ell}}$ if and only if $\chi_X(\lambda) = p(\lambda^2)$ for some irreducible polynomial $p(\lambda) \in F(\lambda)$ of degree n.

Denote by $\Gamma_{\text{eff}}(\mathfrak{m} + \mathfrak{s}_{\text{rs}})(\Gamma)$ the set of $M_H(\Gamma)$ -conjugacy classes in $(\mathfrak{m} + \mathfrak{s}_{\text{rs}})(\Gamma)_{\text{eff}}$. For $X \in \mathfrak{s}_{\text{rs}}(\Gamma)$, $X \in \mathfrak{s}_{\text{rs}}(F)_{\text{eff}}$ if and only if $\chi_X(\lambda) = p(\lambda^2)$ for some irreducible polynomial $p(\lambda) \in F[\lambda]$ of degree n. Now suppose that F is a local field of characteristic zero. Let $P \in \mathscr{F}^{G,\omega}(M_0)$. Then $\mathfrak{m}_P = \begin{pmatrix} \mathfrak{m}_n & \mathfrak{m}_n \\ \mathfrak{m}_n & \mathfrak{m}_n \end{pmatrix}$ and $\mathfrak{n}_P = \begin{pmatrix} \mathfrak{n}_n & \mathfrak{n}_n \\ \mathfrak{n}_n & \mathfrak{n}_n \end{pmatrix}$, where we denote $M_n := M_{P_n}$ and $N_n := N_{P_n}$. We shall choose the same Haar measure for any of the four copies in $\mathfrak{m}_P(F)$ or $\mathfrak{n}_P(F)$ under these identifications. For $f \in \mathcal{S}(\mathfrak{s}(F))$, we define a function $f_P^{\eta} \in \mathcal{S}((\mathfrak{m}_P \cap \mathfrak{s})(F))$ by

$$f_P^{\eta}(Z) := \int_{K_H \times (\mathfrak{n}_P \cap \mathfrak{s})(F)} f(\mathrm{Ad}(k^{-1})(Z+U))\eta(\det(k)) dU dk$$

for all $Z \in (\mathfrak{m}_P \cap \mathfrak{s})(F)$. Recall that $(\hat{f})_P^{\eta} = (f_P^{\eta})$, and we shall denote it by \hat{f}_P^{η} without confusion. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $Q \in \mathscr{F}^G(M)$. For $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$, define the weighted

Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $Q \in \mathscr{F}^G(M)$. For $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, define the weighted orbital integral

(2.3.2)
$$J_M^Q(\eta, X, f) := |D^{\mathfrak{s}}(X)|_F^{1/2} \int_{H_X(F) \setminus H(F)} f(\operatorname{Ad}(x^{-1})(X)) \eta(\det(x)) v_M^Q(x) dx$$

For $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$, define a transfer factor $\kappa(X) := \eta(\det(A))$ (see [58, Definition 5.7]). Then $\kappa(\operatorname{Ad}(x^{-1})(X)) = \eta(\det(x))\kappa(X)$, and the function $\kappa(\cdot)J_M^Q(\eta, \cdot, f)$ is constant on $\operatorname{Ad}(M_H(F))(X)$.

Now suppose additionally that F is non-archimedean. Let $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. In §8.1 in Chapter 4, we deduce from $J_M^G(\eta, X, \cdot)$ an (H, η) -invariant distribution $I_M^G(\eta, X, \cdot)$ on $\mathfrak{s}(F)$. By Propositions 7.2 and 8.1 in Chapter 4, there are unique locally constant functions $\hat{j}_M^G(\eta, X, \cdot)$ and $\hat{i}_M^G(\eta, X, \cdot)$ on $\mathfrak{s}_{rs}(F)$ representing the distributions $\hat{J}_M^G(\eta, X, \cdot)$ and $\hat{I}_M^G(\eta, X, \cdot)$ respectively. That is to say, for all $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we have

$$\hat{J}_{M}^{G}(\eta, X, f) := J_{M}^{G}(\eta, X, \hat{f}) = \int_{\mathfrak{s}(F)} f(U) \hat{j}_{M}^{G}(\eta, X, U) |D^{\mathfrak{s}}(U)|_{F}^{-1/2} dU$$

and

$$\hat{I}_{M}^{G}(\eta, X, f) := I_{M}^{G}(\eta, X, \hat{f}) = \int_{\mathfrak{s}(F)} f(U) \hat{i}_{M}^{G}(\eta, X, U) |D^{\mathfrak{s}}(U)|_{F}^{-1/2} dU$$

2.4. The case of (G', H'). Let F be a local field of characteristic zero or a number field. Let E be a quadratic extension of F. Let \mathfrak{g}' be a central simple algebra over F with a fixed embedding $E \hookrightarrow \mathfrak{g}'(F)$ as F-algebras. Let $\mathfrak{h}' := \operatorname{Cent}_{\mathfrak{g}'}(E)$ be the centraliser of E in \mathfrak{g}' . Then $\mathfrak{h}'(F)$ is a central simple algebra over E by the double centraliser theorem. Denote by $G' := \mathfrak{g}'^{\times}$ (resp. $H' := \mathfrak{h}'^{\times}$) the group of invertible elements in \mathfrak{g}' (resp. \mathfrak{h}'). Let $\alpha \in E \setminus F$ be such that $\alpha^2 \in F$, so $E = F(\alpha)$. In fact, H' is the subgroup of fixed points of the involution $\operatorname{Ad}(\alpha)$ on G'. Denote by \mathfrak{s}' the corresponding tangent space of G'/H' at the neutral element. For a linear subspace $\mathfrak{v}' \subseteq \mathfrak{g}'$, we write $\mathfrak{v}'^{\times} := \mathfrak{v}' \cap G'$. Then $\mathfrak{s}'_{rs} \subseteq \mathfrak{s}'^{\times}$ in our case. Let $\langle \cdot, \cdot \rangle$ be the non-degenerate symmetric bilinear form on $\mathfrak{g}'(F)$ defined by

(2.4.1)
$$\langle X, Y \rangle := \operatorname{Trd}(XY), \forall X, Y \in \mathfrak{g}'(F),$$

which is invariant by the adjoint action of G'(F) and $Ad(\alpha)$.

By the Wedderburn-Artin theorem, G' is isomorphic to $GL_{r,D}$ for some integer $r \geq 1$ and some central division algebra D over F such that $r \deg(D)$ is even. By the Noether-Skolem theorem, up to conjugation by G'(F), the emdedding $H' \hookrightarrow G'$ is isomorphic to one of the two cases below (see [18, §2.1 and §3.1] and §3.4 in Chapter 3).

Case I: if there is an embedding $E \to D$ as *F*-algebras, then $(G', H') \simeq (GL_{r,D}, \operatorname{Res}_{E/F}GL_{r,D'})$, where $D' := \operatorname{Cent}_D(E)$ is a central division algebra over *E* of degree $\frac{\deg(D)}{2}$. Let $M'_0 \simeq (\operatorname{Res}_{E/F}\mathbb{G}_{m,D'})^r$ (resp. $M'_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^r$) be the subgroup of diagonal elements in H' (resp. G'). Recall that there is a bijection $M' \mapsto \widetilde{M'}$ between $\mathscr{L}^{H'}(M'_0)$ and $\mathscr{L}^{G'}(M'_{\widetilde{0}})$. We shall always denote by $\widetilde{M'}$ the image of M' under this bijection. Notice that $M' = \widetilde{M'} \cap H'$ and that we can identify $A_{M'}$ with $A_{\widetilde{M'}}$.

Case II: if there is no embedding $E \to D$ as *F*-algebras, then $(G', H') \simeq (GL_{r,D}, \operatorname{Res}_{E/F}GL_{\frac{r}{2}, D\otimes_F E})$, where $D \otimes_F E$ is a central division algebra over *E* of degree deg(D). Let $M'_0 \simeq (\operatorname{Res}_{E/F} \mathbb{G}_{m,D\otimes_F E})^{\frac{r}{2}}$ (resp. $M'_{\widetilde{0}} \simeq (\mathbb{G}_{m,D})^r$) be the subgroup of diagonal elements in H' (resp. G'). Denote by $\mathscr{L}^{G'}(M'_{\widetilde{0}}, M'_0)$ the subset of elements in $\mathscr{L}^{G'}(M'_{\widetilde{0}})$ containing M'_0 . Recall that there is a bijection $M' \mapsto \widetilde{M'}$ between $\mathscr{L}^{H'}(M'_0)$ and $\mathscr{L}^{G'}(M'_{\widetilde{0}}, M'_0)$. We shall always denote by $\widetilde{M'}$ the image of M' under this bijection. Notice that $M' = \widetilde{M'} \cap H'$ and that we can identify $A_{M'}$ with $A_{\widetilde{M'}}$.

that $M' = \widetilde{M'} \cap H'$ and that we can identify $A_{M'}$ with $A_{\widetilde{M'}}$. If $\operatorname{rk}_F(G') = r$, we also write $G'_r := G'$, $H'_r := H'$ and $\mathfrak{s}'_r := \mathfrak{s}'$. Notice that $\operatorname{rk}_F(H'_r) = r$ in **Case I** (resp. $= \frac{r}{2}$) in **Case II**. There is also a bijection $P' \mapsto \widetilde{P'}$ between $\mathscr{F}^{H'}(M'_0)$ and $\mathscr{F}^{G'}(\widetilde{M'_0})$ in both of **Case I** and **Case II**. We shall always denote by $\widetilde{P'}$ the image of P' under this bijection. Let $\tau \in D^{\times}$ in **Case I** (resp. $\tau \in GL_2(D)$ in **Case II**) be an element such that $\operatorname{Ad}(\alpha)(\tau) = -\tau$. Let $P' \in \mathscr{F}^{H'}(M'_0)$. By Proposition 3.12 in Chapter 3, we have $\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}' = \mathfrak{m}_{P'}\tau = \tau\mathfrak{m}_{P'}$ and $\mathfrak{n}_{\widetilde{P'}} \cap \mathfrak{s}' = \mathfrak{n}_{P'}\tau = \tau\mathfrak{n}_{P'}$. For $\mathfrak{c}' \in \mathscr{T}^{\mathfrak{s}'}$ and $L' \in \mathscr{L}^{H'}(M'_0)$, define

$$W(H',\mathfrak{c}',\widetilde{\mathfrak{l}'}\cap\mathfrak{s}'):=\bigsqcup_{\mathfrak{c}_2'\in\mathscr{T}_0^{\widetilde{\mathfrak{l}'}\cap\mathfrak{s}'}}W(H',\mathfrak{c}',\mathfrak{c}_2').$$

Let $M' \in \mathscr{L}^{H'}(M'_0)$. We say an element $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ is elliptic if $A_{M'}$ is the maximal F-split torus in H'_Y . Denote by $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$ the set of elliptic elements in $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. Denote by $\Gamma_{ell}((\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F))$ the set of M'(F)-conjugacy classes in $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)_{ell}$. For $Y \in \mathfrak{s}'_{rs}(F), Y \in \mathfrak{s}'_{rs}(F)_{ell}$ if and only if $\operatorname{Prd}_Y(\lambda) = p(\lambda^2)$ for some irreducible polynomial $p(\lambda) \in F[\lambda]$ of degree $\frac{r \deg(D)}{2}$.

Now suppose that F is a local field of characteristic zero. Let $P' \in \mathscr{F}^{H'}(M'_0)$. We shall choose the same Haar measures on $(\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F)$ and $(\mathfrak{n}_{\widetilde{P'}} \cap \mathfrak{s}')(F)$ using above identifications induced by τ . Such Haar measures are independent of the choice of τ . For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we define a function $f'_{P'} \in \mathcal{C}^{\infty}_{c}((\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F))$ by

$$f_{P'}'(Z) := \int_{K_{H'} \times (\mathfrak{n}_{\widetilde{P'}} \cap \mathfrak{s}')(F)} f'(\mathrm{Ad}(k^{-1})(Z+U)) dU dk$$

for all $Z \in (\mathfrak{m}_{\widetilde{P'}} \cap \mathfrak{s}')(F)$. Recall that $(\hat{f'})_{P'} = (f'_{P'})$, and we shall denote it by $\hat{f'}_{P'}$ without confusion.

Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Q' \in \mathscr{F}^{H'}(M')$. For $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$, define the weighted orbital integral

$$J_{M'}^{Q'}(Y,f') := |D^{\mathfrak{s}'}(Y)|_F^{1/2} \int_{H'_Y(F) \setminus H'(F)} f'(\operatorname{Ad}(x^{-1})(Y)) v_{M'}^{Q'}(x) dx.$$

Now suppose additionally that F is non-archimedean. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}}' \cap \mathfrak{s}'_{rs})(F)$. In §8.2 in Chapter 4, we deduce from $J_{M'}^{H'}(Y, \cdot)$ an H'-invariant distribution $I_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}'(F)$. By Propositions 7.10 and 8.6 in Chapter 4, there are unique locally constant functions $\hat{j}_{M'}^{H'}(Y, \cdot)$ and $\hat{i}_{M'}^{H'}(Y, \cdot)$ on $\mathfrak{s}'_{rs}(F)$ representing the distributions $\hat{J}_{M'}^{H'}(Y, \cdot)$ and $\hat{I}_{M'}^{H'}(Y, \cdot)$ respectively. That is to say, for all $f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$, we have

$$\hat{J}_{M'}^{H'}(Y,f') := J_{M'}^{H'}(Y,\hat{f}') = \int_{\mathfrak{s}'(F)} f'(V)\hat{j}_{M'}^{H'}(Y,V) |D^{\mathfrak{s}'}(V)|_F^{-1/2} dV$$

and

$$\hat{I}_{M'}^{H'}(Y,f') := I_{M'}^{H'}(Y,\hat{f}') = \int_{\mathfrak{s}'(F)} f'(V)\hat{i}_{M'}^{H'}(Y,V) |D^{\mathfrak{s}'}(V)|_F^{-1/2} dV$$

3. Labesse's lemma

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F. Let η the quadratic character of F^{\times}/NE^{\times} attached to E/F.

For $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, we say that the weighted orbital integrals of f vanish for nontrivial weights if for all $M \in \mathscr{L}^{G,\omega}(M_0), Q \in \mathscr{F}^G(M) - \mathscr{P}^G(M)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we have

$$J_M^Q(\eta, X, f) = 0$$

Suppose that f satisfies this condition. By definition (8.1.1) in Chapter 4, for all $M \in \mathscr{L}^{G,\omega}(M_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we have the equality

$$J_M^G(\eta, X, \hat{f}) = I_M^G(\eta, X, \hat{f}).$$

For $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$, we say that the weighted orbital integrals of f' vanish for nontrivial weights if for all $M' \in \mathscr{L}^{H'}(M'_0), Q' \in \mathscr{F}^{H'}(M') - \mathscr{P}^{H'}(M')$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, we have

$$J_{M'}^{Q'}(Y,f') = 0$$

Suppose that f' satisfies this condition. By definition (8.2.1) in Chapter 4, for all $M' \in \mathscr{L}^{H'}(M'_0)$ and $Y \in (\widetilde{\mathfrak{m}} \cap \mathfrak{s}'_{rs})(F)$, we have the equality

$$J_{M'}^{H'}(Y,\hat{f}') = I_{M'}^{H'}(Y,\hat{f}').$$

LEMMA 3.1 (see [37, Lemma I.7.1]). 1) Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ be such that $\operatorname{Supp}(f) \subseteq \mathfrak{s}_{rs}(F)$. Then there exists $\phi \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ such that

- (a) $\operatorname{Supp}(\phi) \subseteq \operatorname{Ad}(H(F))(\operatorname{Supp}(f));$
- (b) for all $X \in \mathfrak{s}_{\mathrm{rs}}(F)$, $J_G^G(\eta, X, \phi) = J_G^G(\eta, X, f)$;
- (c) the weighted orbital integrals of f vanish for nontrivial weights.

2) Let $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ be such that $\operatorname{Supp}(f') \subseteq \mathfrak{s}'_{rs}(F)$. Then there exists $\phi' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ such that (a) $\operatorname{Supp}(\phi') \subseteq \operatorname{Ad}(H'(F))(\operatorname{Supp}(f'));$

- (b) for all $Y \in \mathfrak{s}'_{rs}(F)$, $J_{H'}^{H'}(Y, \phi') = J_{H'}^{H'}(Y, f');$
- (c) the weighted orbital integrals of f' vanish for nontrivial weights.

4. Matching of orbits

Let F be a local field of characteristic zero or a number field. Let E be a quadratic extension of F. Assume that $\dim_{\overline{F}}(G) = \dim_{\overline{F}}(G')$, i.e., $2n = r \deg(D)$.

4.1. Definition by invariants. There is an injection $M' \mapsto M$ from $\mathscr{L}^{H'}(M'_0)$ into $\mathscr{L}^{G,\omega}(M_0)$ induced by the injection from the set of partitions of r in **Case I** (resp. $\frac{r}{2}$ in **Case II**) into the set of partitions of n. We shall always denote by M the image of M' under this injection.

Denote by \mathbf{A}^n the affine space over F of dimension n. By Proposition 3.3 in Chapter 2, the map $\mathfrak{s} \to \mathbf{A}^n, X \mapsto \chi_X$ defines a categorical quotient $\mathfrak{s}//H$ over F, where χ_X denotes the characteristic polynomial of $X \in \mathfrak{g}$. By Proposition 3.5 in Chapter 3, the map $\mathfrak{s}' \to \mathbf{A}^n, Y \mapsto \operatorname{Prd}_Y$ defines a categorical quotient $\mathfrak{s}'//H'$ over F, where Prd_Y denotes the reduced characteristic polynomial of $Y \in \mathfrak{g}'$. Therefore, we can identify $\mathfrak{s}//H \simeq \mathbf{A}^n \simeq \mathfrak{s}'//H'$. By Proposition 3.3 in Chapter 3, it induces an injection from the set of H'(F)-orbits in $\mathfrak{s}'_{rs}(F)$ into the set of H(F)-orbits in $\mathfrak{s}(F)$.

DEFINITION 4.1. Let $X \in \mathfrak{s}_{rs}(F)$ and $Y \in \mathfrak{s}'_{rs}(F)$. If $\chi_X = \operatorname{Prd}_Y$, we say that X and Y have matching orbits and write $X \leftrightarrow Y$. For $X \in \mathfrak{s}_{rs}(F)$, if there is an element $Y \in \mathfrak{s}'_{rs}(F)$ such that $X \leftrightarrow Y$, we also say that X comes from $\mathfrak{s}'_{rs}(F)$.

REMARK 4.2. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Then X comes from $\mathfrak{s}'_{rs}(F)$ if and only if there is an element $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ such that $X \leftrightarrow Y$, in which case we also say that X comes from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$.

Now suppose that F is a local field of characteristic zero. For $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$, we denote $\eta(X) := \eta(\det(AB))$.

LEMMA 4.3. Let $X \in \mathfrak{s}_{rs}(F)_{ell}$. Then the following conditions are equivalent:

- (1) X comes from $\mathfrak{s}'_{rs}(F)$;
- (2) $\eta(X) = (-1)^r$, where $r := \operatorname{rk}_F(G')$;
- (3) $\eta(X) = e(G')$, where e(G') denotes the Kottwitz sign of G' in the sense of [33].

PROOF. See the proof of [55, Lemma 2.7] for $(1) \Leftrightarrow (2)$. Now we show $(2) \Leftrightarrow (3)$. Let $\frac{i}{2n} = \frac{i_0}{\deg(D)} \in \mathbb{Q} \cap [0, 1)$ be the invariant of $\mathfrak{g}'(F)$, where i_0 and $\deg(D)$ are coprime. Since $2n = r \deg(D)$ is even, by [33, Corollary (7)], e(G') = -1 if and only if i is odd. We have $i = \frac{2ni_0}{\deg(D)} = ri_0$. If i is odd, then r is odd. Conversely, if r is odd, then $\deg(D)$ is even, so i_0 is odd, which implies that i is odd. We have shown that $e(G') = (-1)^r$ when $\deg(\mathfrak{g}'(F))$ is even and thus proved $(2) \Leftrightarrow (3)$.

Let $M' \in \mathscr{L}^{H'}(M'_0)$. Assume that

$$(M, M_H) \simeq (GL_{2n_1}, GL_{n_1} \times GL_{n_1}) \times \cdots \times (GL_{2n_\ell}, GL_{n_\ell} \times GL_{n_\ell})$$

and that

$$(\widetilde{M'}, M') \simeq (G'_{r_1}, H'_{r_1}) \times \cdots \times (G'_{r_\ell}, H'_{r_\ell}).$$

DEFINITION 4.4. Let $M' \in \mathscr{L}^{H'}(M'_0)$, $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$. If each pair of factors of X and Y have matching orbits, we say that X and Y have M-matching orbits.

Let M' be as above and $X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, where

$$A := \operatorname{diag}(A_1, \cdots, A_\ell), B := \operatorname{diag}(B_1, \cdots, B_\ell) \in \mathfrak{gl}_{n_1}(F) \oplus \cdots \oplus \mathfrak{gl}_{n_\ell}(F).$$

Denote $\eta_M(X) := (\eta(\det(A_iB_i)))_{1 \le i \le \ell} \in \{\pm 1\}^{\ell}$. Denote $e_{M'} := ((-1)^{r_i})_{1 \le i \le \ell} \in \{\pm 1\}^{\ell}$.

DEFINITION 4.5. Let $M' \in \mathscr{L}^{H'}(M'_0)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. If $\eta_M(X) = e_{M'}$, we say that X comes potentially from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$.

COROLLARY 4.6. Let $M' \in \mathscr{L}^{H'}(M'_0)$. Then we have

- (1) if $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ comes from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, then X comes potentially from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$;
- (2) if $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)_{ell}$ comes potentially from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$, then X comes from $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$.

4.2. Centralisers. Recall that $E = F(\alpha)$ with $\alpha^2 \in F$. Set $\alpha_0 := \begin{pmatrix} 0 & \alpha^2 1_n \\ 1_n & 0 \end{pmatrix} \in G(F)$. Then $E \simeq F(\alpha_0) \subseteq \mathfrak{g}(F)$. Denote $H_0 := \operatorname{Cent}_G(\alpha) \simeq \operatorname{Res}_{E/F} GL_{n,E}$. Then

$$\mathfrak{h}_0 = \left\{ \begin{pmatrix} A & \alpha^2 C \\ C & A \end{pmatrix} : A, C \in \mathfrak{gl}_n \right\}.$$

Denote by \mathfrak{s}_0 the corresponding tangent space of G/H_0 at the neutral element. Then

$$\mathfrak{s}_0 = \left\{ \begin{pmatrix} A & -\alpha^2 C \\ C & -A \end{pmatrix} : A, C \in \mathfrak{gl}_n
ight\}.$$

Set $\omega_0 := \begin{pmatrix} 1_n & 0\\ 0 & -1_n \end{pmatrix} \in \mathfrak{s}_0(F)$. Then $\omega_0^2 = 1_{2n}$ and $\mathfrak{s}_0 = \mathfrak{h}_0 \omega_0 = \omega_0 \mathfrak{h}_0$. The action $\operatorname{Ad}(\omega_0)$ induces an involution on H_0 and \mathfrak{h}_0 .

Recall that \mathfrak{g}' is a central simple algebra over F with a fixed embedding $E \hookrightarrow \mathfrak{g}'(F)$ as F-algebras. Let $\varphi : \mathfrak{g}' \to \mathfrak{g}$ be an isomorphism over \overline{F} such that for all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, we have

$$\varphi \circ \sigma \circ \varphi^{-1} \circ \sigma^{-1} = \operatorname{Ad}(u_{\sigma}),$$

where u_{σ} is a Galois 1-cocycle with values in G_{ad} . Let α' be the image of α in $\mathfrak{g}'(F)$. Then $\operatorname{Prd}_{\alpha'}$ is defined over F and ${\alpha'}^2 - \alpha^2 = 0$. We deduce that $\operatorname{Prd}_{\alpha'}(\lambda) = (\lambda^2 - \alpha^2)^n = \chi_{\alpha_0}(\lambda) \in F[\lambda]$. Because both of α_0 and α' are semi-simple in the classical sense, there exists $x \in G$ such that $\operatorname{Ad}(x) \circ \varphi(\alpha') = \alpha_0$. Since $\alpha' \in \mathfrak{g}'(F)$, we have $\varphi(\alpha') = \phi \circ \sigma(\alpha') = \operatorname{Ad}(u_{\sigma}) \circ \sigma \circ \varphi(\alpha')$. As $\alpha_0 \in G(F)$, we obtain $\operatorname{Ad}(x^{-1})(\alpha_0) = \operatorname{Ad}(u_{\sigma}) \circ \sigma \circ \operatorname{Ad}(x^{-1})(\alpha_0) = \operatorname{Ad}(u_{\sigma}) \circ \sigma \circ \operatorname{Ad}(\pi^{-1})(\alpha_0) = \operatorname{Ad}(u_{\sigma}) \circ \pi \circ \operatorname{Ad}(\sigma(x)^{-1})(\alpha_0)$. It implies that $xu_{\sigma}\sigma(x)^{-1} \in H_0$. It turns out that by changing u_{σ} in its class in $H^1(F, G_{ad})$, we may and shall suppose that $\varphi(\alpha') = \alpha$ and that u_{σ} is a Galois 1-cocycle with values in H_0/Z_G .

Set
$$\alpha_1 := \begin{pmatrix} \alpha 1_n \\ -\alpha 1_n \end{pmatrix} \in G(E)$$
. Then $H = \operatorname{Cent}_G(\alpha_1)$. Recall that $\omega = \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. Thus $= \mathfrak{h}\omega = \omega\mathfrak{h}$. The action $\operatorname{Ad}(\omega)$ induces an involution on H and \mathfrak{h} .

Let $y \in G$ such that

5

(4.2.1)
$$\operatorname{Ad}(y) \circ \varphi(\alpha') = \alpha_1$$

i.e., $\operatorname{Ad}(y)(\alpha_0) = \alpha_1$. Then the morphism $\operatorname{Ad}(y) \circ \varphi$ induces an isomorphism over \overline{F} from \mathfrak{s}' to \mathfrak{s} . For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, we have $\operatorname{Ad}(\sigma(y)) \circ \sigma \circ \varphi(\alpha') = \sigma(\alpha_1) = \varepsilon_{\sigma}\alpha_1$, where ε_{σ} denotes the quadratic character of $\operatorname{Gal}(\overline{F}/F)$ associated to E/F. Since $\sigma \circ \varphi(\alpha') = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi \circ \sigma(\alpha') = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi(\alpha') = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi(\alpha') = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi(\alpha')$

$$\operatorname{Ad}(yu_{\sigma}\sigma(y)^{-1})(\alpha_1) = \varepsilon_{\sigma}\alpha_1.$$

For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, define $w_{\sigma} := 1_{2n}$ if $\sigma \in \operatorname{Gal}(\overline{F}/E)$ and $w_{\sigma} := \omega$ otherwise. Then $\operatorname{Ad}(w_{\sigma})(\alpha_1) = \varepsilon_{\sigma}\alpha_1$. We deduce that

$$(4.2.2) yu_{\sigma}\sigma(y)^{-1}w_{\sigma} \in H.$$

The elements y verifying (4.2.1) form an H-torsor of the form Hv where

(4.2.3)
$$v := \begin{pmatrix} 1_n & \alpha 1_n \\ 1_n & -\alpha 1_n \end{pmatrix} \in G(E).$$

We easily check that $\sigma(v) = w_{\sigma}v$ for all $\sigma \in \operatorname{Gal}(\overline{F}/F)$ and that $\omega v = v\omega_0$.

Let $Y \in \mathfrak{s}'(F)$ be a semi-simple element. There exists $y \in Hv$ and a semi-simple element $X \in \mathfrak{s}(F)$ such that

$$\mathrm{Ad}(y) \circ \varphi(Y) = X.$$

Then $\operatorname{Ad}(y) \circ \varphi$ induces an isomorphism over \overline{F} from H'_Y to H_X . For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, we have $\operatorname{Ad}(\sigma(y)) \circ \sigma \circ \varphi(Y) = \sigma(X) = X$. Since $\sigma \circ \varphi(Y) = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi \circ \sigma(Y) = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi(Y) = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \operatorname{Ad}(y^{-1})(X)$, we obtain

$$yu_{\sigma}\sigma(y)^{-1} \in G_X.$$

Combined with (4.2.2), we have $yu_{\sigma}\sigma(y)^{-1} \in G_X \cap Hw_{\sigma}$. Since w_{σ} normalises H (resp. \mathfrak{s}), we see that $yu_{\sigma}\sigma(y)^{-1}$ normalises H_X (resp. \mathfrak{s}_X).

LEMMA 4.7. Let $X \in \mathfrak{s}_{rs}(F)$ and $Y \in \mathfrak{s}'_{rs}(F)$ be such that $X \leftrightarrow Y$. There exists $y \in Hv$ such that $\operatorname{Ad}(y) \circ \varphi(Y) = X$ and that $\operatorname{Ad}(y) \circ \varphi$ induces isomorphisms $H'_Y \to H_X$ and $\mathfrak{s}'_Y \to \mathfrak{s}_X$ over F.

PROOF. This is a generalisation of [58, Lemma 7.4]. For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, we see that $(\operatorname{Ad}(y) \circ \varphi) \circ \sigma \circ (\operatorname{Ad}(y) \circ \varphi)^{-1} \circ \sigma^{-1} = \operatorname{Ad}(y) \circ (\varphi \circ \sigma \circ \varphi^{-1} \circ \sigma^{-1}) \circ (\sigma \circ \operatorname{Ad}(y^{-1}) \circ \sigma^{-1}) = \operatorname{Ad}(yu_{\sigma}\sigma(y)^{-1})$. For $X \in \mathfrak{s}_{rs}$, we know that X is also regular semi-simple in G in the classical sense. Since $yu_{\sigma}\sigma(y)^{-1} \in G_X$, the action $\operatorname{Ad}(yu_{\sigma}\sigma(y)^{-1})$ on H_X (resp. \mathfrak{s}_X) is trivial, which implies that $\operatorname{Ad}(y) \circ \varphi$ induces an isomorphism $H'_Y \to H_X$ (resp. $\mathfrak{s}'_Y \to \mathfrak{s}_X$) over F.

4.3. Cohomological criterion. Let $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$. Then $GL_{n,A}$ is a maximal F-torus in GL_n . Denote $X_0 := \operatorname{Ad}(v^{-1})(X) \in \mathfrak{s}_0(E)$, where v is defined by (4.2.3). Notice that $H_{0,X_0} = \operatorname{Ad}(v^{-1})(H_X) = H_X \simeq GL_{n,A}$. Let $H_A := \operatorname{Cent}_H(\operatorname{diag}(A, A)) = GL_{n,A} \times GL_{n,A}$, which is a maximal F-torus in H. Denote $T_X := \operatorname{Cent}_{H_0}(\operatorname{diag}(A, A)) = \operatorname{Ad}(v^{-1})(H_A)$. For all closed subvariety V of H defined over F and stable by ω_0 . Thus T_X is a maximal F-torus in H_0 . Notice that $H_X = H_A^{\omega}$ and that $H_{0,X_0} = T_X^{\omega_0}$. We see that $T_X \simeq \operatorname{Res}_{E/F}(GL_{n,A})_E$ and that the inclusion $H_{0,X_0} \subseteq T_X$ is isomorphic to the inclusion $GL_{n,A} \subseteq \operatorname{Res}_{E/F}(GL_{n,A})_E$. For simplicity, we also write $T := T_X$ and $R := H_{0,X_0}$.

LEMMA 4.8. There exists a unique Galois 1-cocycle t_{σ} with values in T/R such that for all $\sigma \in \text{Gal}(\overline{F}/F)$, we have

$$\operatorname{Ad}(t_{\sigma}) \circ \sigma(X_0) = X_0.$$

PROOF. The uniqueness is obvious by definition. It suffices to consider the existence. The cocycle condition is also automatic. We only need to check the equality. If $\sigma \in \operatorname{Gal}(\overline{F}/E)$, then $\sigma(X_0) = X_0$, so it suffices to take $t_{\sigma} = 1$. Now suppose that $\sigma \notin \operatorname{Gal}(\overline{F}/E)$. Since $\sigma(X_0) = \operatorname{Ad}(\sigma(v)^{-1}) \circ \sigma(X) = \operatorname{Ad}(v^{-1}w_{\sigma}^{-1})(X)$, it suffices to find an element $t_{\sigma} \in T$ such that $vt_{\sigma}v^{-1}w_{\sigma}^{-1} \in G_X$. In fact, we can take $t_{\sigma} = \operatorname{Ad}(v^{-1}) \begin{pmatrix} 1 \\ & A \end{pmatrix}$.

By the inflation-restriction exact sequence and Hilbert's Theorem 90, the cohomology group $H^1(F, T/R)$ is identified with $H^1(\operatorname{Gal}(E/F), T(E)/R(E))$. For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$ and $h' = \operatorname{Ad}(v^{-1})(h) \in T$ where $h \in H_A$, we see that $\sigma(h') = \operatorname{Ad}(\sigma(v)^{-1}) \circ \sigma(h) = \operatorname{Ad}(v^{-1}) \circ \operatorname{Ad}(w_{\sigma}) \circ \sigma(h)$. Therefore, we can regard T as the subgroup H_A of G equipped with the Galois action $\operatorname{Ad}(w_{\sigma}) \circ \sigma$. Then the inclusion $R(E) \subseteq T(E)$ is isomorphic to the inclusion $H_X(E) \subseteq H_A(E)$. Let σ be the nontrivial element in $\operatorname{Gal}(E/F)$. We see that $u_{\sigma} = (B, C) \in H_A(E)/H_X(E)$ is a 1-cocycle if and only if $u_{\sigma}\operatorname{Ad}(w_{\sigma}) \circ \sigma(u_{\sigma}) \in H_X(E)$, i.e., $BC^{\sigma} = CB^{\sigma}$. We also see that $u_{\sigma} \in H_A(E)/H_X(E)$ is a 1coboundary if and only if $u_{\sigma} = (B, C)^{-1}\operatorname{Ad}(w_{\sigma}) \circ \sigma(B, C)$ for some $(B, C) \in H_A(E)/H_X(E)$, i.e., $u_{\sigma} = (C^{\sigma}B^{-1}, B^{\sigma}C^{-1}) \in H_{A}(E)/H_{X}(E). \text{ Then } H^{1}(\text{Gal}(E/F), T(E)/R(E)) \text{ is the quotient of} \\ \{(B, C) \in H_{A}(E)/H_{X}(E) : BC^{\sigma} = CB^{\sigma}\} = \{(1, CB^{-1}) \in H_{A}(E)/H_{X}(E) : (CB^{-1})^{\sigma} = CB^{-1}\} \\ = \{(1, B) \in H_{A}(E)/H_{X}(E) : B^{\sigma} = B\} = \{(1, B) \in H_{A}(E)/H_{X}(E) : B \in GL_{n,A}(F)\}$

by

$$\{(C^{\sigma}B^{-1}, B^{\sigma}C^{-1}) \in H_A(E)/H_X(E) : B, C \in GL_{n,A}(E)\} = \{(B^{-\sigma}, B) \in H_A(E)/H_X(E) : B \in GL_{n,A}(E)\} = \{(1, BB^{\sigma}) \in H_A(E)/H_X(E) : B \in GL_{n,A}(E)\}.$$

That is to say, it is the quotient of $G_A(F)$ by the group of norms of $GL_{n,A}(E)$. Under the identification of T and H_A with the twisted Galois action, the Galois 1-cocycle t_{σ} in Lemma 4.8 corresponds exactly to the class of A.

Since A is regular semi-simple in $GL_n(F)$ in the classical sense, its characteristic polynomial χ_A is separable. Let $\chi_A = \prod_{i \in I} \chi_i$ be the factorisation of χ_A into a product of monic irreducible polynomials over F. We see that

$$GL_{n,A} \simeq \prod_{i \in I} \operatorname{Res}_{F_i/F} \mathbb{G}_{m,F_i},$$

where $F_i = F[\lambda]/(\chi_i(\lambda))$. Denote $E_i := F_i \otimes_F E$. Let I_0 be the subset of I consisting of i such that E_i is a field. We have

$$H^{1}(F, T/R) = \prod_{i \in I} F_{i}^{\times} / N_{E_{i}/F_{i}}(E_{i}^{\times}) = \prod_{i \in I_{0}} F_{i}^{\times} / N_{E_{i}/F_{i}}(E_{i}^{\times}).$$

If F is a local field of characteristic zero, then $H^1(F, T/R) = (\mathbb{Z}/2\mathbb{Z})^{I_0}$.

LEMMA 4.9. Let $t \in H^1(F, T/R)$ be the class of the Galois 1-cocycle t_σ in Lemma 4.8. Let $u \in H^1(F, H_0/Z_G)$ be the class of the Galois 1-cocycle u_σ associated to \mathfrak{g}' . There exists $Y \in \mathfrak{s}'(F)$ and $h \in H_0$ such that $\operatorname{Ad}(h) \circ \varphi(Y) = X_0$ if and only if there exists an element of $H^1(F, T/Z_G)$ which has images $t \in H^1(F, T/R)$ and $u \in H^1(F, H_0/Z_G)$ under the natural maps:

$$H^{1}(F, T/Z_{G}) \longrightarrow H^{1}(F, T/R)$$

$$\downarrow$$

$$H^{1}(F, H_{0}/Z_{G})$$

REMARK 4.10. The condition in the above lemma says exactly that X comes from $\mathfrak{s}'_{rs}(F)$.

PROOF OF LEMMA 4.9. For all $h \in H_0$, let $Y \in \mathfrak{s}'$ be the unique element such that $\operatorname{Ad}(h) \circ \varphi(Y) = X_0$. For all $\sigma \in \operatorname{Gal}(\overline{F}/F)$, we have $\operatorname{Ad}(\sigma(h)) \circ \sigma \circ \varphi(Y) = \sigma(X_0) = \operatorname{Ad}(t_{\sigma}^{-1})(X_0)$ and $\sigma \circ \varphi(Y) = \operatorname{Ad}(u_{\sigma}^{-1}) \circ \varphi \circ \sigma(Y)$. But $Y \in \mathfrak{s}'(F)$ if and only if $\varphi \circ \sigma(Y) = \varphi(Y) = \operatorname{Ad}(h^{-1})(X_0)$. We have shown that there exists $Y \in \mathfrak{s}'(F)$ and $h \in H_0$ such that $\operatorname{Ad}(h) \circ \varphi(Y) = X_0$ if and only if there exists $h \in H_0$ such that

$$\operatorname{Ad}(\sigma(h)u_{\sigma}^{-1}h^{-1})(X_0) = \operatorname{Ad}(t_{\sigma}^{-1})(X_0)$$

for all $\sigma \in \operatorname{Gal}(\overline{F}/F)$. If this equality is satisfied, then $hu_{\sigma}\sigma(h)^{-1} \in Rt_{\sigma} \subseteq T$ defines a Galois 1-cocycle with values in T/Z_G which has desired images. Conversely, any Galois 1-cocycle with values in T/Z_G having image u is of the form $hu_{\sigma}\sigma(h)^{-1}$ where $h \in H_0$. If it also has image t, it means that by replacing h with t'h where $t' \in T$, we may suppose that $hu_{\sigma}\sigma(h)^{-1} \in Rt_{\sigma}$. Then the above equality is satisfied for such an h.

Recall that $H_0^{\omega_0} = H^{\omega} \simeq GL_n$. We shall abuse notation and denote by R_{der} (resp. T_{der}) the preimage of R (resp. T) in $H_{0,\text{der}}^{\omega_0} = H_{0,\text{sc}}^{\omega_0}$ (resp. $H_{0,\text{der}} = H_{0,\text{sc}}$). We shall the index "ab" to denote the abelianised cohomology defined in [38, §1.6 and 1.8]. Recall that the abelianisation maps $H^1(F, H_0^{\omega_0}/Z_G) \rightarrow H_{ab}^1(F, H_0^{\omega_0}/Z_G)$ and $H^1(F, H_0/Z_G) \rightarrow H_{ab}^1(F, H_0/Z_G)$ are surjective (see [38, Proposition 1.6.7]). If F is a non-archimedean local field of characteristic zero, they are also injective by Kneser's theorem (see *loc. cit.*). We have the following commutative diagram with exact columns and rows.

$$\begin{array}{cccc} H^{1}(F, R/Z_{G}) & \longrightarrow & H^{1}(F, T/Z_{G}) & \longrightarrow & H^{1}(F, T/R) & \longrightarrow & H^{2}(F, R/Z_{G}) \\ & & & \downarrow & & \downarrow & & \downarrow \\ H^{1}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G}) & \longrightarrow & H^{1}_{ab}(F, H_{0}/Z_{G}) & \longrightarrow & H^{2}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G}) & \longrightarrow & H^{2}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G}) \\ & & \downarrow & & \downarrow & & \downarrow \\ & & & \downarrow & & \downarrow \\ H^{2}(F, R_{der}) & \longrightarrow & H^{2}(F, T_{der}) & \longrightarrow & H^{2}(F, T_{der}/R_{der}) \end{array}$$

PROPOSITION 4.11. There is a canonical identification

 $H^{1}_{\rm ab}(F, H^{\omega_0}_0/Z_G \to H_0/Z_G) \simeq H^{1}(F, C_{H_0}/C_{H^{\omega_0}}).$

PROOF. The group on the left is isomorphic to $H^1(F, T_{\text{der}}/R_{\text{der}} \to T/R)$. Since $[T_{\text{der}}/R_{\text{der}} \to T/R]$ is injective, it is quasi-isomorphic to $[1 \to (T/R)/(T_{\text{der}}/R_{\text{der}})]$. But $(T/R)/(T_{\text{der}}/R_{\text{der}}) \simeq (T/T_{\text{der}})/(R/R_{\text{der}})$, where $T/T_{\text{der}} = C_{H_0}$ and $R/R_{\text{der}} = C_{H_0^{\omega_0}}$.

COROLLARY 4.12. Suppose that F is a local field of characteristic zero. Then

$$H^1_{\rm ab}(F, H^{\omega_0}_0/Z_G \to H_0/Z_G) = \mathbb{Z}/2\mathbb{Z}.$$

The morphism

$$H^{1}(F, T/R) = (\mathbb{Z}/2\mathbb{Z})^{I_{0}} \to H^{1}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G} \to H_{0}/Z_{G}) = \mathbb{Z}/2\mathbb{Z}$$

is the sum of components.

Denote by ε_X the image of t (defined in Lemma 4.8) under the morphism

$$H^{1}(F, T/R) \to H^{1}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G} \to H_{0}/Z_{G}).$$

COROLLARY 4.13. Suppose that F is a local field of characteristic zero. The map

$$\mathfrak{s}_{\rm rs}(F) \to H^1_{\rm ab}(F, H^{\omega_0}_0/Z_G \to H_0/Z_G) = \mathbb{Z}/2\mathbb{Z} : X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \mapsto \varepsilon_X$$

where $X' := \begin{pmatrix} 0 & 1_n \\ AB & 0 \end{pmatrix} \in \mathfrak{s}_{rs}(F)$ is understood as the map $\mathfrak{s}_{rs}(F) \to \{\pm 1\} : X \mapsto \eta(X).$

PROOF. We may reduce ourselves to the case where $X \in \mathfrak{s}_{rs}(F)_{ell}$. In the elliptic case, it results from [10, Lemmas 1.3 and 1.4]

Denote by ε' the image of u (associated to \mathfrak{g}') under the morphism

$$H^{1}(F, H_{0}/Z_{G}) \simeq H^{1}_{ab}(F, H_{0}/Z_{G}) \to H^{1}_{ab}(F, H^{\omega_{0}}_{0}/Z_{G} \to H_{0}/Z_{G}).$$

LEMMA 4.14. If X comes from $\mathfrak{s}'_{rs}(F)$, then $\varepsilon_X = \varepsilon'$.

PROOF. It results from Lemma 4.9 and the above commutative diagram.

PROPOSITION 4.15. Suppose that F is a non-archimedean local field of characteristic zero and that A is elliptic in $GL_n(F)$. Then X comes from $\mathfrak{s}'_{rs}(F)$ if and only if $\varepsilon_X = \varepsilon'$.

PROOF. By Lemma 4.14, it suffices to prove the reverse direction. Since F is non-archimedean, we have $H^2_{\rm ab}(F, H^{\omega_0}_0/Z_G) = 0$ by [38, Lemme 1.5.1]. Since F is local and the tori $R/Z_G, R_{\rm der}$ and $T_{\rm der}$ are F-anisotropic, the groups $H^2(F, R/Z_G), H^2(F, R_{\rm der}), H^2(F, T_{\rm der})$ and $H^2(F, T_{\rm der}/R_{\rm der})$ vanish (see *loc. cit.*). The above commutative diagram is simplified as follows.

Choose an arbitrary preimage $t' \in H^1(F, T/Z_G)$ of t. The image of t' in $H^1_{ab}(F, H_0/Z_G)$ is of the form $ug(u_1)$ where $u_1 \in H^1_{ab}(F, H_0^{\omega_0}/Z_G)$ because both of u and this image map to $\varepsilon' = \varepsilon_X \in H^1_{ab}(F, H_0^{\omega_0}/Z_G) \to H_0/Z_G)$ by our assumption. Let $t_1 \in H^1(F, R/Z_G)$ be a preimage of u_1 . Then $t'f(t_1)^{-1} \in H^1(F, T/Z_G)$ has images t and u. We may conclude by the bijectivity of the abelianisation map $H^1(F, H_0/Z_G) \to H^1_{ab}(F, H_0/Z_G)$ and Lemma 4.9.

COROLLARY 4.16. Suppose that F is a non-archimedean local field of characteristic zero. The map

$$H^1(F, H_0/Z_G) \simeq H^1_{\rm ab}(F, H_0/Z_G) \to H^1_{\rm ab}(F, H_0^{\omega_0}/Z_G \to H_0/Z_G) = \mathbb{Z}/2\mathbb{Z} : u \mapsto \varepsilon'$$

is understood as the Kottwitz sign

$$H^{1}(F, H_{0}/Z_{G}) \to H^{1}(F, G_{ad}) \to \{\pm 1\}$$

PROOF. It can be computed directly. However, it is also a consequence of Lemma 4.3, Corollary 4.13 and Proposition 4.15. $\hfill \Box$

4.4. Levi subgroups. Let $M \in \mathscr{L}^{G,\omega}(M_0)$. Denote $M_{H_0} := \operatorname{Ad}(v^{-1})(M) \cap H_0$, where v is defined by (4.2.3). Then M_{H_0} is a Levi subgroup of H_0 defined over F. Let $X = \begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix} \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Then $T \subseteq M_{H_0}$ and $R \subseteq M_{H_0}^{\omega_0}$. We have an obvious generalisation to the product form of some results in the previous section.

PROPOSITION 4.17 (cf. Proposition 4.11). There is a canonical identification

$$H^1_{\rm ab}(F, M^{\omega_0}_{H_0}/Z_M \to M_{H_0}/Z_M) \simeq H^1(F, C_{M_{H_0}}/C_{M^{\omega_0}_{H_0}})$$

COROLLARY 4.18 (cf. Corollary 4.12). Suppose that F is a local field of characteristic zero. Then

$$H^{1}_{\rm ab}(F, M^{\omega_0}_{H_0}/Z_M \to M_{H_0}/Z_M) = (\mathbb{Z}/2\mathbb{Z})^{\ell},$$

where ℓ is the number of blocks of M. The morphism

$$H^{1}(F, T/R) = (\mathbb{Z}/2\mathbb{Z})^{I_{0}} \to H^{1}_{ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}) = (\mathbb{Z}/2\mathbb{Z})^{\ell}$$

is the sum of components in each block of M. If $L \in \mathscr{L}^G(M)$, then the morphism

$$H^{1}_{\rm ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}) = (\mathbb{Z}/2\mathbb{Z})^{\ell} \to H^{1}_{\rm ab}(F, L^{\omega_{0}}_{H_{0}}/Z_{L} \to L_{H_{0}}/Z_{L}) = (\mathbb{Z}/2\mathbb{Z})^{\ell'}$$

is the sum of components in each block of L, where ℓ' is the number of blocks of L.

PROPOSITION 4.19. Suppose that F is a non-archimedean local field of characteristic zero. The group $H^1_{ab}(F, M^{\omega_0}_{H_0}/Z_M \to M_{H_0}/Z_M)$ is canonically isomorphic to the Pontryagin dual of the finite group $Z_{\widehat{M}}[2]$ of elements $z \in Z_{\widehat{M}}$ such that $z^2 = 1$. If $L \in \mathscr{L}^G(M)$, then the morphism

$$H^{1}_{\rm ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}) \to H^{1}_{\rm ab}(F, L^{\omega_{0}}_{H_{0}}/Z_{L} \to L_{H_{0}}/Z_{L})$$

is the dual of the canonical embedding $Z_{\widehat{L}}[2] \hookrightarrow Z_{\widehat{M}}[2]$.

PROOF. By [34, Proposition 6.4] and Proposition 4.17, the group $H^1_{ab}(F, M^{\omega_0}_{H_0}/Z_M \to M_{H_0}/Z_M)$ is canonically isomorphic to the Pontryagin dual of $\pi_0((Z_{\widehat{M_{H_0}}}/Z_{\widehat{M_{H_0}}})^{\Gamma})$. Notice that $Z_{\widehat{M_{H_0}}} = Z_{\widehat{M_{H_0}}} \times Z_{\widehat{M_{H_0}}}^{\omega_0}$, on which the nontrivial element $\sigma \in \text{Gal}(E/F)$ acts by exchanging two components. Via the morphism $(z, z') \mapsto zz'^{-1}$ and the diagonal embedding, we obtain

$$Z_{\widehat{M}_{H_0}}/Z_{\widehat{M}_{H_0}^{\omega_0}} \simeq Z_{\widehat{M}_{H_0}^{\omega_0}} \simeq Z_{\widehat{M}}.$$

These isomorphisms are Γ -equivariant if we define the action of Γ on $Z_{\widehat{M_{H_0}^{\omega_0}}}$ and $Z_{\widehat{M}}$ by the lift of the action of $\operatorname{Gal}(E/F)$ with $\sigma(z) = z^{-1}$. Then

$$\pi_0((Z_{\widehat{M_{H_0}}}/Z_{\widehat{M_{H_0}}})^{\Gamma}) \simeq Z_{\widehat{M}}[2].$$

Denote by ε_X^M the image of t (defined in Lemma 4.8) under the morphism

$$H^{1}(F, T/R) \to H^{1}_{ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}).$$

PROPOSITION 4.20 (cf. Corollary 4.13). Suppose that F is a local field of characteristic zero. The map

$$(\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F) \to H^1_{\mathrm{ab}}(F, M^{\omega_0}_{H_0}/Z_M \to M_{H_0}/Z_M) = (\mathbb{Z}/2\mathbb{Z})^{\ell} : X = \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} \mapsto \varepsilon^M_X$$

where $X' := \begin{pmatrix} 0 & 1_n \\ AB & 0 \end{pmatrix} \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ is understood as the map

$$(\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F) \to \{\pm 1\}^{\ell} : X \mapsto \eta_M(X)$$

For our purpose in this paper, we shall fix $M' \in \mathscr{L}^{H'}(M'_0)$. It means that we may and shall start from some $M \in \mathscr{L}^{G,\omega}(M_0)$ such that u (associated to \mathfrak{g}') belongs to the image of the injective map $H^1(F, M_{H_0}/Z_G) \to H^1(F, H_0/Z_G)$. Denote by $u_M \in H^1(F, M_{H_0}/Z_G)$ the preimage of u. Denote by ε'^M the image of u_M under the morphism

 $H^{1}(F, M_{H_{0}}/Z_{G}) \to H^{1}(F, M_{H_{0}}/Z_{M}) \to H^{1}_{ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}).$

PROPOSITION 4.21 (cf. Corollary 4.16). Suppose that F is a non-archimedean local field of characteristic zero. The map

$$H^{1}(F, M_{H_{0}}/Z_{G}) \to H^{1}(F, M_{H_{0}}/Z_{M}) \to H^{1}_{ab}(F, M^{\omega_{0}}_{H_{0}}/Z_{M} \to M_{H_{0}}/Z_{M}) = (\mathbb{Z}/2\mathbb{Z})^{\ell} : u_{M} \mapsto \varepsilon'^{M}$$

is understood as the map

$$H^{1}(F, M_{H_{0}}/Z_{G}) \to H^{1}(F, M_{H_{0}}/Z_{M}) \to H^{1}(F, M_{\mathrm{ad}}) \to \{\pm 1\}^{\ell} : u_{M} \mapsto e_{M'}$$

5. Statement of results

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F. Fix $M' \in \mathscr{L}^{H'}(M'_0)$. Recall that its image in $\mathscr{L}^{G,\omega}(M_0)$ is denoted by M.

DEFINITION 5.1. Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$. We say that f and f' are partially M-associated if they satisfy the following condition: for all $L \in \mathscr{L}^{G}(M)$ and all $Q \in \mathscr{F}^{G}(L)$, if $X \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{rs})(F)$ have L-matching orbits, then

$$\kappa(X)J_L^Q(\eta, X, f) = J_{L'}^{Q'}(Y, f').$$

DEFINITION 5.2. Let $f \in C_c^{\infty}(\mathfrak{s}(F))$ and $f' \in C_c^{\infty}(\mathfrak{s}'(F))$. We say that f and f' are M-associated if they are partially M-associated and satisfy the additional condition: for all $L \in \mathscr{L}^G(M)$, $Q \in \mathscr{F}^G(L)$ and $X \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$, we have

$$J_L^Q(\eta, X, f) = 0$$

unless X comes potentially from $(\mathfrak{m}_{\widetilde{O'}} \cap \mathfrak{s}'_{rs})(F)$ (see Definition 4.5).

THEOREM 5.3. Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ be partially M-associated and satisfy the following conditions.

(a) The weighted orbital integrals of f and f' vanish for nontrivial weights.

(b) If $X \in \mathfrak{s}_{rs}(F)$ does not come from $\mathfrak{s}'_{rs}(F)$, then

$$J_G^G(\eta, X, f) = 0.$$

Then $\gamma_{\psi}(\mathfrak{h}(F))^{-1}\hat{f}$ and $\gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{f}'$ are *M*-associated.

PROOF. Combine Proposition 5.5 and Corollary 5.11 below.

REMARK 5.4. The two conditions in the above theorem imply the additional condition in Definition 5.2. They may be weakened, but they are enough for our purpose.

Recall that we denote by $\gamma_{\psi}(\mathfrak{h}(F))$ (resp. $\gamma_{\psi}(\mathfrak{h}'(F))$) the Weil constants associated to $\mathfrak{h}(F)$ (resp. $\mathfrak{h}'(F)$).

PROPOSITION 5.5. Let $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ and $f' \in \mathcal{C}_c^{\infty}(\mathfrak{s}'(F))$ be partially *M*-associated and satisfy the two conditions in Theorem 5.3. Then $\gamma_{\psi}(\mathfrak{h}(F))^{-1}\hat{f}$ and $\gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{f}'$ are also partially *M*-associated.

The rest of this paper will be denoted to the proof of Proposition 5.5. Its corollary below may be more useful for applications.

COROLLARY 5.6. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(F)$ have *M*-matching orbits. Let $U \in \mathfrak{s}_{rs}(F)$ and $V \in \mathfrak{s}'_{rs}(F)$ be such that $U \leftrightarrow V$. Then we have the equality

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X)\kappa(U)\hat{i}_{M}^{G}(\eta,X,U) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}\hat{i}_{M'}^{H'}(Y,V).$$

PROOF. By Lemma 4.7, we may and shall fix an isomorphism $\varphi : \mathfrak{s}'_V(F) \to \mathfrak{s}_U(F)$ such that $\varphi(V) = U$. Recall that $W(H, \mathfrak{s}_U)$ (resp. $W(H', \mathfrak{s}'_V)$) denotes the Weyl group associated to $\mathfrak{s}_U \in \mathscr{T}^{\mathfrak{s}}$ (resp. $\mathfrak{s}'_V \in \mathscr{T}^{\mathfrak{s}'}$). Choose open compact neighbourhoods ω of U in $(\mathfrak{s}_U \cap \mathfrak{s}_{rs})(F)$ and ω' of V in $(\mathfrak{s}'_V \cap \mathfrak{s}'_{rs})(F)$ which are small enough such that

- (i) the sets $i(\omega)$ where $i \in W(H, \mathfrak{s}_U)$ are mutually disjoint;
- (ii) the sets $i'(\omega')$ where $i' \in W(H', \mathfrak{s}'_V)$ are mutually disjoint;
- (iii) $\varphi(\omega') = \omega;$
- (iv) $\kappa(\cdot)\hat{i}_M^G(\eta, X, \cdot)$ is constant on ω ;
- (v) $\hat{i}_{M'}^{H'}(Y, \cdot)$ is constant on ω' .

Notice that the conditions (iv) and (v) are assured by Lemmas 8.3.1) and 8.8.1) in Chapter 4. By the condition (i) on ω and Lemma 3.1.1), we can construct a function $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ such that

- (i) $\operatorname{Supp}(f) \subseteq \operatorname{Ad}(H(F))(\omega);$
- (ii) for all $Z \in \omega$, $\kappa(Z)J_G^G(\eta, Z, f) = 1$;

(iii) the weighted orbital integrals of f vanish for nontrivial weights.

By the condition (ii) on ω' and Lemma 3.1.2), we can construct a function $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ such that

- (i) $\operatorname{Supp}(f') \subseteq \operatorname{Ad}(H'(F))(\omega');$
- (ii) for all $Z' \in \omega', J_{H'}^{H'}(Z', f') = 1;$
- (iii) the weighted orbital integrals of f' vanish for nontrivial weights.

We see that f and f' are partially *M*-associated and satisfy the two conditions in Theorem 5.3. By Proposition 5.5, we have the equality

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X)J_{M}^{G}(\eta, X, \hat{f}) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}J_{M'}^{H'}(Y, \hat{f}')$$

By the condition (iii) on f and the Weyl integration formula (7.1.2) in Chapter 4, we have

$$J_M^G(\eta, X, \hat{f}) = I_M^G(\eta, X, \hat{f}) = \sum_{\mathfrak{c} \in \mathscr{T}_0^{\mathfrak{s}}} |W(H, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_G^G(\eta, Z, f) \hat{i}_M^G(\eta, X, Z) dZ.$$

By the conditions (i) and (ii) on f and the condition (iv) on ω , the last expression equals

$$\int_{\omega} J_G^G(\eta, Z, f) \hat{i}_M^G(\eta, X, Z) dZ = \operatorname{vol}(\omega) \kappa(U) \hat{i}_M^G(\eta, X, U).$$

Similarly, with the help of the conditions on f', (7.2.2) in Chapter 4 and the condition (iv) on ω , we obtain

$$J_M^G(\eta, X, \hat{f}) = I_M^G(\eta, X, \hat{f}) = \operatorname{vol}(\omega')\hat{i}_{M'}^{H'}(Y, V).$$

Since $vol(\omega) = vol(\omega')$, we deduce the equality in the corollary.

Recall that $\omega = \begin{pmatrix} 0 & 1_n \\ 1_n & 0 \end{pmatrix}$. For $X \in \mathfrak{s}(F)$, denote $X^{\omega} := \operatorname{Ad}(\omega)(X)$. For $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$, define $f^{\omega}(X) := f(X^{\omega})$ for all $X \in \mathfrak{s}(F)$.

LEMMA 5.7. Let
$$f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$$
 and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$. Then we have
(1) $(\hat{f})^{\omega} = (f^{\omega})^{\hat{}};$
(2) $J^{G}_{M}(\eta, X, f^{\omega}) = \eta(X)J^{G}_{M}(\eta, X, f).$

PROOF. Similar properties are used in the proof of [58, Lemma 8.3] though our involutions are slightly different. It suffices to notice additionally that $v_M^G(\operatorname{Ad}(w)(x)) = v_M^G(x)$ for $x \in H(F)$. \Box

LEMMA 5.8. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in \mathfrak{s}_{rs}(F)$. Then we have the equality

$$\hat{i}_M^G(\eta, X, U^\omega) = \eta(X)\hat{i}_M^G(\eta, X, U).$$

PROOF. From Lemma 5.7, we deduce that

(5.0.1)
$$\hat{j}_M^G(\eta, X, U^\omega) = \eta(X)\hat{j}_M^G(\eta, X, U)$$

There exists $x \in H(F)$, $L \in \mathscr{L}^{G,\omega}(M_0)$ and $Z \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$ such that $U = \operatorname{Ad}(x)(Z)$. By Lemmas (8.3.2) and (8.2) in Chapter 4, we have

$$\eta(X)\hat{i}_M^G(\eta, X, U) = \eta(\det(x))\eta(X)\hat{i}_M^G(\eta, X, Z) = \eta(\det(x))\eta(X)\hat{j}_M^G(\eta, X, Z).$$

Applying (5.0.1) to X and Z, we have

$$\eta(X)\hat{j}_M^G(\eta, X, Z) = \hat{j}_M^G(\eta, X, Z^{\omega})$$

Since $Z^{\omega} \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)_{ell}$, by Lemma 8.2 in Chapter 4 again, we obtain

$$\hat{j}_M^G(\eta, X, Z^\omega) = \hat{i}_M^G(\eta, X, Z^\omega).$$

Thus

$$\eta(X)\hat{i}_M^G(\eta, X, U) = \eta(\det(x))\hat{i}_M^G(\eta, X, Z^{\omega}).$$

We see that $U^{\omega} = \operatorname{Ad}(\omega x \omega^{-1})(Z^{\omega})$, where $\omega x \omega^{-1} \in H(F)$. By Lemma 8.3.2) in Chapter 4 again, we have

$$\hat{i}_M^G(\eta, X, U^\omega) = \eta(\det(\omega x \omega^{-1}))\hat{i}_M^G(\eta, X, Z^\omega) = \eta(\det(x))\hat{i}_M^G(\eta, X, Z^\omega).$$

Then the lemma follows.

PROPOSITION 5.9. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in \mathfrak{s}_{rs}(F)$. If $\eta(X) \neq \eta(U)$, then

$$\hat{i}_M^G(\eta, X, U) = 0$$

PROOF. We see that $U^{\omega} = \operatorname{Ad}(\omega U)(U)$, where $\omega U \in H(F)$. By Lemma 8.3.2) in Chapter 4, we have

$$\hat{i}_M^G(\eta, X, U^\omega) = \eta(\det(\omega U))\hat{i}_M^G(\eta, X, U) = \eta(U)\hat{i}_M^G(\eta, X, U)$$

One may conclude by comparing this equality with Lemma 5.8.

REMARK 5.10. By the same argument, we can generalise the above proposition to the following form. Let $L \in \mathscr{L}^G(M)$. Let $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in (\mathfrak{l} \cap \mathfrak{s})_{rs}(F)$. If $\eta_L(X) \neq \eta_L(U)$, then

$$\hat{i}_M^L(\eta, X, U) = 0$$

COROLLARY 5.11. Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ satisfies the additional condition in Definition 5.2. Then \hat{f} also satisfies this condition.

PROOF. By induction, it suffices to show that for all $Q \in \mathscr{F}^G(M)$ and $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we have

$$J_M^Q(\eta, X, \hat{f}) = 0$$

unless X comes potentially from $(\mathfrak{m}_{\widetilde{Q'}} \cap \mathfrak{s}'_{rs})(F)$.

By Proposition 4.1.4) and (8.1.1) in Chapter 4, we obtain

$$J^Q_M(\eta, X, \hat{f}) = J^{M_Q}_M(\eta, X, \hat{f}^{\eta}_Q) = \sum_{L \in \mathscr{L}^{M_Q}(M)} \hat{I}^{L, M_Q, w}_M(\eta, X, f^{\eta}_Q),$$

where

$$\begin{split} \hat{I}_{M}^{L,M_{Q},w}(\eta,X,f_{Q}^{\eta}) &= \sum_{\{R \in \mathscr{L}^{G,\omega}(M_{0}): R \subseteq L\}} |W_{0}^{R_{n}}| |W_{0}^{L_{n}}|^{-1} \sum_{\mathfrak{c} \in \mathscr{T}_{\mathrm{ell}}(\mathfrak{r} \cap \mathfrak{s})} |W(R_{H},\mathfrak{c})|^{-1} \\ &\int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{L}^{M_{Q}}(\eta,Z,f_{Q}^{\eta}) \hat{i}_{M}^{L}(\eta,X,Z) dZ. \end{split}$$

By Proposition 4.1.4) in Chapter 4 again, we have

$$J_L^{M_Q}(\eta, Z, f_Q^{\eta}) = J_L^Q(\eta, Z, f)$$

If $J_M^Q(\eta, X, \hat{f}) \neq 0$, then $J_L^Q(\eta, Z, f)\hat{i}_M^L(\eta, X, Z) \neq 0$ for some Z. Since $J_L^Q(\eta, Z, f) \neq 0$, by our assumption on f, we see that Z comes potentially from $(\mathfrak{m}_{\widetilde{Q'}} \cap \mathfrak{s}'_{rs})(F)$. Since $\hat{i}_M^L(\eta, X, Z) \neq 0$, by Remark 5.10, we have $\eta_L(X) = \eta_L(Z)$. As $L \subseteq M_Q$, it implies that $\eta_{M_Q}(X) = \eta_{M_Q}(Z)$. Thus X also comes potentially from $(\mathfrak{m}_{\widetilde{O'}} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$.

 \square

6. Limit formulae

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F. Recall that for $\mathfrak{c} \in \mathfrak{s}$ and $X, U \in (\mathfrak{c} \cap \mathfrak{s}_{rs})(F)$, we define $\gamma_{\psi}(X, U)$ by (2.2.1). A similar notation is used in the case of (G', H').

Let $M \in \mathscr{L}^{G,\omega}(M_0)$. For all $L \in \mathscr{L}^G(M), X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(F)$ and $U \in (\mathfrak{l} \cap \mathfrak{s}_{\mathrm{rs}})(F)$, we define

$$\hat{a}_{M}^{M}(\eta, X, U) := \sum_{x \in (M_{H})_{U}(F) \setminus M_{H}(F), \operatorname{Ad}(x)(X) \in \mathfrak{s}_{U}(F)} \eta(\operatorname{Nrd}(x))\gamma_{\psi}(\operatorname{Ad}(x)(X), U)\psi(\langle \operatorname{Ad}(x)(X), U \rangle)$$

if L = M and

$$\hat{a}_M^L(\eta, X, U) := 0$$

if $L \neq M$.

Let
$$M' \in \mathscr{L}^{H'}(M'_0)$$
. For all $L' \in \mathscr{L}^{H'}(M'), Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$ and $V \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$, we define

$$\hat{a}_{M'}^{M'}(Y,V) := \sum_{x \in M'_V(F) \setminus M'(F), \operatorname{Ad}(x)(Y) \in \mathfrak{s}'_V(F)} \gamma_{\psi}(\operatorname{Ad}(x)(Y), V)\psi(\langle \operatorname{Ad}(x)(Y), V \rangle)$$

if L' = M' and

$$\hat{a}_{M'}^{L'}(Y,V) := 0$$

if $L' \neq M'$.

PROPOSITION 6.1. 1) Let $M \in \mathscr{L}^{G,\omega}(M_0)$, $L \in \mathscr{L}^G(M)$, $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $U \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(F)$. Then there exists $N \in \mathbb{N}$ such that if $\mu \in F^{\times}$ satisfies $v_F(\mu) < -N$, we have the equality

$$\hat{i}_M^L(\eta, \mu X, U) = \hat{a}_M^L(\eta, \mu X, U).$$

2) Let $M' \in \mathscr{L}^{H'}(M'_0)$, $L' \in \mathscr{L}^{H'}(M')$, $Y \in (\widetilde{\mathfrak{w}'} \cap \mathfrak{s}'_{rs})(F)$ and $V \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{rs})(F)$. Then there exists $N \in \mathbb{N}$ such that if $\mu \in F^{\times}$ satisfies $v_F(\mu) < -N$, we have the equality

$$\hat{i}_{M'}^{L'}(\mu Y, V) = \hat{a}_{M'}^{L'}(\mu Y, V).$$

PROOF. This is a generalisation of [58, Proposition 7.1] and Propositions 10.1 and 10.4 in Chapter 4. $\hfill \Box$

7. Construction of test functions

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F.

LEMMA 7.1. Let X, Y, y be as in Lemma 4.7. Let $V \in (\mathfrak{s}'_Y \cap \mathfrak{s}'_{rs})(F)$ and $U := \operatorname{Ad}(y) \circ \varphi(V)$. Then we have

(1)
$$\langle X, U \rangle = \langle Y, V \rangle;$$

(2)
$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\gamma_{\psi}(X,U) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}\gamma_{\psi}(Y,V).$$

PROOF. This is a generalisation of [58, Lemma 7.5].

Fix $M' \in \mathscr{L}^{H'}(M'_0)$. Recall that its image in $\mathscr{L}^{G,\omega}(M_0)$ is denoted by M.

PROPOSITION 7.2. Let $X_0 \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y_0 \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ be such that $X_0 \leftrightarrow Y_0$. Then there exists $f \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ and $f' \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$ satisfying the following conditions.

(a) If $X \in \text{Supp}(f)$, there exists $Y \in (\mathfrak{s}'_{Y_0} \cap \mathfrak{s}'_{rs})(F)$ such that $X \leftrightarrow Y$.

(b) If $Y \in \text{Supp}(f')$, then Y is H'(F)-conjugate to an element in $(\mathfrak{s}'_{Y_0} \cap \mathfrak{s}'_{rs})(F)$.

(c) The weighted orbital integrals of f and f' vanish for nontrivial weights.

(d) The functions f and f' are partially G-associated and satisfy the condition: if $X \in \mathfrak{s}_{rs}(F)$ does not come from $\mathfrak{s}'_{rs}(F)$, then

$$J_G^G(\eta, X, f) = 0$$

 $(e) \ \ {\it For} \ Q \in \mathscr{F}^G(M) - \mathscr{P}^G(M),$

$$J_M^Q(\eta, X_0, \hat{f}) = J_{M'}^{Q'}(Y_0, \hat{f}') = 0.$$

(f) We have the equality

$$\gamma_{\psi}(\mathfrak{h}(F))^{-1}\kappa(X_0)J_G^G(\eta, X_0, \hat{f}) = \gamma_{\psi}(\mathfrak{h}'(F))^{-1}J_{H'}^{H'}(Y_0, \hat{f}') \neq 0.$$

PROOF. This is a generalisation of [58, Proposition 7.6] whose additional ingredients are Lemma 3.1 and Proposition 6.1 (cf. $[14, \S6]$).

Fix an isomorphism $\varphi : \mathfrak{s}'_{Y_0}(F) \to \mathfrak{s}_{X_0}(F)$ such that $\varphi(Y_0) = X_0$ as in Lemma 4.7. Choose $V_0 \in (\mathfrak{s}'_{Y_0} \cap \mathfrak{s}'_{rs})(F)$ and denote $U_0 := \varphi(V_0) \in (\mathfrak{s}_{X_0} \cap \mathfrak{s}_{rs})(F)$ such that

- (i) for all $i \in W(H, \mathfrak{s}_{X_0}), i \neq 1$, we have $\langle i(X_0) X_0, U_0 \rangle \neq 0$;
- (ii) for all $i' \in W(H', \mathfrak{s}'_{Y_0}), i' \neq 1$, we have $\langle i'(Y_0) Y_0, V_0 \rangle \neq 0$;
- (iii) $\kappa(U_0) = \kappa(X_0).$

Fix $r \in \mathbb{N}^*$ such that

- (i) $1 + \varpi^r \mathcal{O}_F \subseteq F^{\times 2}$;
- (ii) the sets $i((1 + \varpi^r \mathcal{O}_F)U_0)$ where $i \in W(H, \mathfrak{s}_{X_0})$ are mutually disjoint;
- (iii) the sets $i'((1 + \varpi^r \mathcal{O}_F)V_0)$ where $i' \in W(H', \mathfrak{s}'_{Y_0})$ are mutually disjoint.

By Propositions 6.1, there exists $N \in \mathbb{N}$ such that if $\mu \in F^{\times}$ satisfies $v_F(\mu) < -N$, then

(i) for $L_1, L_2 \in \mathscr{L}^G(M)$ with $L_1 \subseteq L_2, i_1 \in W(L_{1,H}, \mathfrak{s}_{X_0}, \mathfrak{l}_1 \cap \mathfrak{s})$ and $i_2 \in W(H, \mathfrak{s}_{X_0}, \mathfrak{l}_2 \cap \mathfrak{s})$, we have

$$\hat{i}_{L_1}^{L_2}(\eta, i_1(X_0), i_2(\mu U_0)) = \hat{a}_{L_1}^{L_2}(\eta, i_1(X_0), i_2(\mu U_0));$$

(ii) for $L'_1, L'_2 \in \mathscr{L}^{H'}(M')$ with $L'_1 \subseteq L'_2, i'_1 \in W(L'_1, \mathfrak{s}'_{Y_0}, \widetilde{\mathfrak{l}'_1} \cap \mathfrak{s}')$ and $i'_2 \in W(H', \mathfrak{s}'_{Y_0}, \widetilde{\mathfrak{l}'_2} \cap \mathfrak{s}')$, we have

$$\hat{i}_{L'_1}^{L'_2}(i'_1(Y_0), i'_2(\mu V_0)) = \hat{a}_{L'_1}^{L'_2}(i'_1(Y_0), i'_2(\mu V_0)).$$

Fix such an integer N.

Fix $\mu \in F^{\times}$ with $v_F(\mu) < -N$ such that

- (i) $\eta(\mu) = 1;$
- (ii) for all $i \in W(H, \mathfrak{s}_{X_0}), i \neq 1$, the character $\lambda \mapsto \psi(\varpi^r \mu \lambda \langle i(X_0) X_0, U_0 \rangle)$ is nontrivial on \mathcal{O}_F ;
- (iii) for all $i' \in W(H', \mathfrak{s}'_{Y_0}), i' \neq 1$, the character $\lambda \mapsto \psi(\varpi^r \mu \lambda \langle i'(Y_0) Y_0, V_0 \rangle)$ is nontrivial on \mathcal{O}_F .

Notice that the conditions (ii) and (iii) are possible because of the conditions (i) and (ii) on U_0 and V_0 . Set $\omega'_0 := \mu(1 + \varpi^r \mathcal{O}_F)V_0$. Denote by \mathfrak{d}' the *F*-vector space generated by V_0 . Fix a complement \mathfrak{e}' of \mathfrak{d}' in $\mathfrak{s}'_{Y_0}(F)$. For $V \in \mathfrak{s}'_{Y_0}(F)$, denote by $V_{\mathfrak{d}'}$ its projection to \mathfrak{d}' with respect to the decomposition $\mathfrak{s}'_{Y_0}(F) = \mathfrak{d}' \oplus \mathfrak{e}'$.

Set $\omega_0 := \mu(1 + \varpi^r \mathcal{O}_F) U_0$. Denote by $\mathfrak{d} := \varphi(\mathfrak{d}')$ the *F*-vector space generated by U_0 . Let $\mathfrak{e} := \varphi(\mathfrak{e}')$ be the complement of \mathfrak{d} in $\mathfrak{s}_{X_0}(F)$. For $U \in \mathfrak{s}_{X_0}(F)$, denote by $U_\mathfrak{d}$ its projection to \mathfrak{d} with respect to the decomposition $\mathfrak{s}_{X_0}(F) = \mathfrak{d} \oplus \mathfrak{e}$.

Choose open compact neighbourhoods $\omega_{\mathfrak{e}}$ of 0 in \mathfrak{e} and $\omega_{\mathfrak{e}'}$ of 0 in \mathfrak{e}' which are small enough such that $\omega := \omega_0 \oplus \omega_{\mathfrak{e}}$ and $\omega' := \omega'_0 \oplus \omega'_{\mathfrak{e}'}$ satisfy

- (i) the sets $i(\omega)$ where $i \in W(H, \mathfrak{s}_{X_0})$ are mutually disjoint;
- (ii) the sets $i'(\omega')$ where $i' \in W(H', \mathfrak{s}'_{Y_0})$ are mutually disjoint;
- (iii) $\omega \subseteq (\mathfrak{s}_{X_0} \cap \mathfrak{s}_{rs})(F), \, \omega' \subseteq (\mathfrak{s}'_{Y_0} \cap \mathfrak{s}'_{rs})(F) \text{ and } \varphi(\omega') = \omega;$
- (iv) for $L_1, L_2 \in \mathscr{L}^G(M)$ with $L_1 \subseteq L_2, i_1 \in W(L_{1,H}, \mathfrak{s}_{X_0}, \mathfrak{l}_1 \cap \mathfrak{s}), i_2 \in W(H, \mathfrak{s}_{X_0}, \mathfrak{l}_2 \cap \mathfrak{s})$ and $U \in \omega$, we have

$$\hat{i}_{L_1}^{L_2}(\eta, i_1(X_0), i_2(U)) = \hat{a}_{L_1}^{L_2}(\eta, i_1(X_0), i_2(U)) = \hat{a}_{L_1}^{L_2}(\eta, i_1(X_0), i_2(U_{\mathfrak{d}}));$$

(v) for $L'_1, L'_2 \in \mathscr{L}^{H'}(M')$ with $L'_1 \subseteq L'_2, i'_1 \in W(L'_1, \mathfrak{s}'_{Y_0}, \widetilde{\mathfrak{l}'_1} \cap \mathfrak{s}'), i'_2 \in W(H', \mathfrak{s}'_{Y_0}, \widetilde{\mathfrak{l}'_2} \cap \mathfrak{s}')$ and $V \in \omega'$, we have $\hat{i}^{L'_2}(i'(Y_2), i'_2(V_1)) - \hat{a}^{L'_2}(i'(Y_2), i'_2(V_1)) - \hat{a}^{L'_2}(i'(Y_2), i'_2(V_2)))$.

$$\hat{i}_{L_{1}'}^{L_{2}}(i_{1}'(Y_{0}), i_{2}'(V)) = \hat{a}_{L_{1}'}^{L_{2}}(i_{1}'(Y_{0}), i_{2}'(V)) = \hat{a}_{L_{1}'}^{L_{2}}(i_{1}'(Y_{0}), i_{2}'(V_{\mathfrak{d}'}))$$

is constant on (

(vi) the function κ is constant on ω .

Notice that the conditions (i) and (ii) follow from the conditions (ii) and (iii) on r. Besides, the conditions (iv) and (v) are assured by $v_F(\mu) < -N$ and $r \ge 1$. Moreover, the condition (vi) results from the condition (i) on μ and the condition (i) on r. Combined with the condition (iii) on U_0 , the condition (vi) says that the restriction of κ to ω equals $\kappa(X_0)$.

Define a function f_{ω} on ω by

$$f_{\omega}(U) := \psi(-\langle X_0, U_{\mathfrak{d}} \rangle)$$

for all $U \in \omega$. By the condition (i) on ω and Lemma 3.1.1), we can construct a function $f \in \mathcal{C}_c^{\infty}(\mathfrak{s}(F))$ such that

- (i) $\operatorname{Supp}(f) \subseteq \operatorname{Ad}(H(F))(\omega);$
- (ii) for all $U \in \omega$, $\kappa(U) J_G^G(\eta, U, f) = f_\omega(U)$;
- (iii) the weighted orbital integrals of f vanish for nontrivial weights.

Define a function $f'_{\omega'}$ on ω' by

$$f'_{\omega'}(V) := \psi(-\langle Y_0, V_{\mathfrak{d}'} \rangle)$$

for all $V \in \omega'$. By the condition (ii) on ω' and Lemma 3.1.2), we can construct a function $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ such that

- (i) $\operatorname{Supp}(f') \subseteq \operatorname{Ad}(H'(F))(\omega');$

(ii) for all $V \in \omega'$, $J_{H'}^{H'}(V, f') = f'_{\omega'}(V)$; (iii) the weighted orbital integrals of f' vanish for nontrivial weights.

We shall check that f and f' satisfy the conditions in the proposition. The conditions (a) and (b) result from the condition (ii) on ω and ω' , the condition (i) on f and the condition (i) on f'. The condition (c) is exactly the condition (iii) on f and the condition (iii) on f'. The condition (d) is deduced from the condition (iii) on ω and ω' , the conditions (i) and (ii) on f and the conditions (i) and (ii) on f'.

We now verify the condition (e). By Proposition 4.1.4) in Chapter 4, we write

$$J_{M}^{Q}(\eta, X_{0}, \hat{f}) = J_{M}^{M_{Q}}(\eta, X_{0}, \hat{f}_{Q}^{\eta})$$

For the same reason and the condition (iii) on f, the weighted orbital integrals of f_Q^{η} vanish for nontrivial weights. Then we have

$$J_M^{M_Q}(\eta, X_0, \hat{f}_Q^{\eta}) = I_M^{M_Q}(\eta, X_0, \hat{f}_Q^{\eta}).$$

From the Weyl integration formula (7.1.2) in Chapter 4, we deduce that

$$I_{M}^{M_{Q}}(\eta, X_{0}, \hat{f}_{Q}^{\eta}) = \sum_{\mathfrak{c}\in\mathscr{T}_{0}^{\mathfrak{m}_{Q}\cap\mathfrak{s}}} |W(M_{Q_{H}}, \mathfrak{c})|^{-1} \int_{\mathfrak{c}_{\mathrm{reg}}(F)} J_{M_{Q}}^{M_{Q}}(\eta, Z, f_{Q}^{\eta}) \hat{i}_{M}^{M_{Q}}(\eta, X_{0}, Z) dZ$$

Using Proposition 4.1.4) Chapter 4 again, we obtain

$$J_{M_Q}^{M_Q}(\eta, Z, f_Q^{\eta}) = J_{M_Q}^Q(\eta, Z, f) = J_G^G(\eta, Z, f).$$

Suppose that $J^G_G(\eta, Z, f) \neq 0$. By the condition (i) on f, there exists $x \in H(F)$ such that $\operatorname{Ad}(x^{-1})(Z) \in$ ω . For such an x, we have $\operatorname{Ad}(x)(\mathfrak{s}_{X_0}) = \mathfrak{c}$. By the condition (iv) on ω , if $M_Q \neq M$, we get $\hat{i}_M^{M_Q}(\eta, X_0, Z) = 0$ and thus

$$J_M^Q(\eta, X_0, \hat{f}) = 0.$$

Similarly, if $M_{Q'} \neq M'$, we prove the vanishing of $J_{M'}^{Q'}(Y_0, \hat{f}')$ with the help of Proposition 4.4.4) and (7.2.2) in Chapter 4, the condition (i) on f' and the condition (v) on ω' .

The condition (f) is shown as in the last paragraph of the proof of [58, Proposition 7.6] with the aide of Lemma 7.1.

8. The weighted fundamental lemma

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F. Assume that F has odd residue characteristic and that E/F is unramified. Assume that $(G', H') \simeq$ $(GL_{2n}, \operatorname{Res}_{E/F}GL_{n,E})$. Let

$$\mathfrak{s}(\mathcal{O}_F) := \left\{ \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} : A, B \in \mathfrak{gl}_n(\mathcal{O}_F) \right\}.$$

We identify $\mathfrak{s}'(F) \simeq \mathfrak{h}'(F)$ and let

$$\mathfrak{s}'(\mathcal{O}_F) := \mathfrak{gl}_n(\mathcal{O}_E).$$

Denote by $f_0 \in \mathcal{C}^{\infty}_c(\mathfrak{s}(F))$ (resp. $f'_0 \in \mathcal{C}^{\infty}_c(\mathfrak{s}'(F))$) the characteristic function of $\mathfrak{s}(\mathcal{O}_F)$ (resp. of $\mathfrak{s}'(\mathcal{O}_F)$).

LEMMA 8.1 (see Theorem 10.9 in Chapter 3). For all $M \in \mathscr{L}^{G,\omega}(M_0)$ and all $Q \in \mathscr{F}^G(M)$, we have (a) if $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ have M-matching orbits, then

$$\kappa(X)J_{M}^{Q}(\eta, X, f_{0}) = J_{M'}^{Q'}(Y, f_{0}');$$

(b) for $X \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$, we have

$$V_M^Q(\eta, X, f_0) = 0$$

unless X comes potentially from $(\mathfrak{m}_{\widetilde{O'}} \cap \mathfrak{s}'_{\mathrm{rs}})(F)$.

9. Approximation of local data by global data

Let k'/k be a quadratic extension of number fields and \mathbb{D} be a central division algebra over k. For a place v of k, denote $k'_v := k' \otimes_k k_v$ and $\mathbb{D}_v := \mathbb{D} \otimes_k k_v$.

PROPOSITION 9.1. Let E/F be a quadratic extension of non-archimedean local fields of characteristic 0.

- I) Let D be a central division algebra over F containing E. Then there exists a quadratic extension of number fields k'/k, a central division algebra \mathbb{D} over k containing k', and a finite set S of finite places of k satisfying the following conditions.
 - (a) The number field k is totally imaginary.
 - $(b) |S| \ge 2.$
 - (c) For all $v \in S$, we have $k_v \simeq F$, $k'_v \simeq E$ and $\mathbb{D}_v \simeq D$.
 - (d) For all $v \notin S$, k_v splits \mathbb{D}_v .
- II) Let D be a central division algebra over F such that $D \otimes_F E$ is a central division algebra over E. Then there exists a quadratic extension of number fields k'/k, a central division algebra \mathbb{D} over k such that $\mathbb{D} \otimes_k k'$ is a central division algebra over k', and a finite set S of finite places of k satisfying the following conditions.
 - (a) The number field k is totally imaginary.
 - (b) $|S| \ge 2$.
 - (c) For all $v \in S$, we have $k_v \simeq F$, $k'_v \simeq E$ and $\mathbb{D}_v \simeq D$.
 - (d) For all $v \notin S$, k_v splits \mathbb{D}_v .

PROOF. By [14, Proposition 9.1], there exists a number field k, a central simple algebra \mathbb{D} over k and a finite set S of finite places of k such that

- (i) the number field k is totally imaginary;
- (ii) $|S| \ge 2;$
- (iii) for all $v \in S$, we have $k_v \simeq F$ and $\mathbb{D}_v \simeq D$;
- (iv) for all $v \notin S$, k_v splits \mathbb{D}_v .

From the condition (iii), we know that \mathbb{D} is a central division algebra over k. By [20, Theorem 3.1], there exists a quadratic extension k' of k such that $k'_v \simeq E$ for all $v \in S$.

- I) We shall use [48, Theorem 1.2] to show that there exists an k-embedding of k' into \mathbb{D} . It is clear that there exists an k_v -embedding of k'_v into \mathbb{D}_v for all place v of k. Let v' be a place of k' and v be the place of k below v'. Denote by c_v (resp. d_v) the capacity (resp. index) of \mathbb{D}_v . Since D is a central division algebra over F of even degree, we deduce that \mathbb{D} is a central division algebra over k of even degree. Then $c_v d_v$ is even. Define $x_{v'} := \frac{c_v \cdot \operatorname{gcd}([k'_v: k_v], d_v)}{2} \in \mathbb{Q}_{>0}$. If v is archimedean or $v \notin S$, i.e., k_v splits \mathbb{D}_v , then $d_v = 1$ and c_v is even, so $x_{v'} \in \mathbb{Z}$. If $v \in S$, then $[k'_{v'}: k_v] = [E:F] = 2$, which implies that $x_{v'} \in \mathbb{Z}$ since $c_v d_v$ is even. We may use [48, Theorem 1.2] to conclude.
- II) Let $v \in S$ and v' be the unique place of k' over v. Since $(\mathbb{D} \otimes_k k')_{v'} \simeq \mathbb{D} \otimes_k k_v \otimes_{k_v} k'_{v'} \simeq D \otimes_F E$ is a central division algebra over E, we know that $\mathbb{D} \otimes_k k'$ is a central division algebra over k'.

10. An infinitesimal variant of Guo-Jacquet trace formulae

Let k'/k be a quadratic extension of number fields and η the quadratic character of $\mathbb{A}^{\times}/k^{\times}$ attached to it, where \mathbb{A} denotes the ring of adèles of k.

Let **G** be a reductive group over k. Fix a maximal compact subgroup K of $\mathbf{G}(\mathbb{A})$ which is admissible relative to \mathbf{M}_0 in the sense of [5, p. 9]. In this paper, we choose the standard maximal compact subgroup when $\mathbf{G}(k) = \mathbf{GL}_n(\mathbb{D})$, where \mathbb{D} is a central division algebra over a finite field extension of k. That is to say, $K := \prod_v K_v$ where each K_v is the standard maximal compact subgroup of $\mathbf{G}(k_v)$. For $\mathbf{M} \in \mathscr{L}^{\mathbf{G}}(\mathbf{M}_0)$, we define the weight function $v_{\mathbf{M}}^{\mathbf{G}}$ with respect to K as in the local case (2.1.1). We fix the Haar measure on K such that $\operatorname{vol}(K) = 1$. We fix a Haar measure on $\mathbf{G}(\mathbb{A})$ which is compatible with the Iwasawa decomposition.

Let $(\mathbf{G}, \mathbf{H}, \theta)$ be a symmetric pair defined over k. Denote by $\mathcal{S}(\mathfrak{s}(\mathbb{A}))$ the Bruhat-Schwartz space of $\mathfrak{s}(\mathbb{A})$. Let $\langle \cdot, \cdot \rangle$ be a G-invariant θ -invariant non-degenerate symmetric bilinear form on \mathfrak{g} . Fix a continuous and nontrivial unitary character $\Psi : \mathbb{A}/k \to \mathbb{C}^{\times}$. For $f \in \mathfrak{s}(\mathbb{A})$, we define its Fourier transform $\hat{f} \in \mathfrak{s}(\mathbb{A})$

by

$$\forall \widehat{X} \in \mathfrak{s}(\mathbb{A}), \widehat{f}(\widehat{X}) := \int_{\mathfrak{s}(\mathbb{A})} f(X) \Psi(\langle X, \widehat{X} \rangle) dX.$$

We fix the self-dual on $\mathfrak{s}(\mathbb{A})$ with respect to the $\Psi(\langle \cdot, \cdot \rangle)$.

We shall write (\mathbf{G}, \mathbf{H}) and $(\mathbf{G}', \mathbf{H}')$ for the global symmetric pairs over k with respect to k' defined in Sections 2.3 and 2.4 respectively. We still use a minuscule Fraktur letter to denote a global Lie algebra and write \mathfrak{s} and \mathfrak{s}' for the corresponding global symmetric spaces. We also fix minimal Levi k-subgroups of these four groups as in the local case and denote them by bold letters.

For $\phi \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$, we define

(10.0.1)
$$J^{\mathbf{G}}(\eta,\phi) = \sum_{\mathfrak{o}\in\mathcal{O}_{\mathrm{rs}}^{\mathfrak{s}}} J^{\mathbf{G}}_{\mathfrak{o}}(\eta,\phi),$$

where $\mathcal{O}_{rs}^{\mathfrak{s}}$ denotes the set of $\mathbf{H}(k)$ -orbits in $\mathfrak{s}_{rs}(k)$, and $J_{\mathfrak{o}}^{\mathbf{G}}(\eta, \phi)$ is the constant term of (5.0.1) with s = 0 in Chapter 2. From Theorem 4.11 and Corollary 5.6 in Chapter 2, we know that the right hand side of (10.0.1) is absolutely convergent. For $\mathfrak{o} \in \mathcal{O}_{rs}^{\mathfrak{s}}$, let $\mathbf{L} \in \mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)$ and $X \in \mathfrak{o} \cap (\mathfrak{l} \cap \mathfrak{s}_{rs})(k)_{ell}$. By Theorem 9.2 in Chapter 2, we have

$$J^{\mathbf{G}}_{\mathfrak{o}}(\eta,\phi) = \operatorname{vol}(A^{\infty}_{\mathbf{L}}\mathbf{H}_{X}(k) \setminus \mathbf{H}_{X}(\mathbb{A})) \int_{\mathbf{H}_{X}(\mathbb{A}) \setminus \mathbf{H}(\mathbb{A})} \phi(\operatorname{Ad}(x^{-1})(X)) \eta(\operatorname{Nrd}(x)) v^{\mathbf{G}}_{\mathbf{L}}(x) dx.$$

We denote

$$\tau(\mathbf{H}_X) := \operatorname{vol}(A_{\mathbf{L}}^{\infty}\mathbf{H}_X(k) \backslash \mathbf{H}_X(\mathbb{A}))$$

for $X \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(k)_{ell}$ and define

$$J_{\mathbf{L}}^{\mathbf{G}}(\eta, X, \phi) := \int_{\mathbf{H}_{X}(\mathbb{A}) \setminus \mathbf{H}(\mathbb{A})} \phi(\mathrm{Ad}(x^{-1})(X)) \eta(\mathrm{Nrd}(x)) v_{\mathbf{L}}^{\mathbf{G}}(x) dx$$

for all $X \in (\mathfrak{l} \cap \mathfrak{s}_{rs})(k)$. From Lemma 3.10 in Chapter 4, we obtain

(10.0.2)
$$J^{\mathbf{G}}(\eta,\phi) = \sum_{\mathbf{L}\in\mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)} |W_0^{\mathbf{L}_n}| |W_0^{\mathbf{GL}_n}|^{-1} \sum_{X\in\Gamma_{\mathrm{ell}}((\mathfrak{l}\cap\mathfrak{s}_{\mathrm{rs}})(k))} \tau(\mathbf{H}_X) J_{\mathbf{L}}^{\mathbf{G}}(\eta,X,\phi).$$

Denote by $\mathcal{S}(\mathfrak{s}'(\mathbb{A}))$ the Bruhat-Schwartz space of $\mathfrak{s}'(\mathbb{A})$. For $\phi' \in \mathcal{S}(\mathfrak{s}'(\mathbb{A}))$, we define

(10.0.3)
$$J^{\mathbf{H}'}(\phi') = \sum_{\mathfrak{o} \in \mathcal{O}_{rs}^{\mathfrak{o}'}} J^{\mathbf{H}'}_{\mathfrak{o}}(\phi'),$$

where $\mathcal{O}_{rs}^{\mathfrak{s}'}$ denotes the set of $\mathbf{H}'(k)$ -orbits in $\mathfrak{s}'_{rs}(k)$, and $J_{\mathfrak{o}}^{\mathbf{H}'}(\phi')$ is the constant term of (5.0.1) in Chapter 3. From Theorem 4.2 and Corollary 5.3 in Chapter 3, we know that the right hand side of (10.0.3) is absolutely convergent. For $\mathfrak{o} \in \mathcal{O}_{rs}^{\mathfrak{s}'}$, let $\mathbf{L}' \in \mathscr{L}^{\mathbf{H}'}(\mathbf{M}'_0)$ and $Y \in \mathfrak{o} \cap (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{rs})(k)_{ell}$. By Theorem 9.2 in Chapter 3, we have

$$J_{\mathfrak{o}}^{\mathbf{H}'}(\phi') = \operatorname{vol}(A_{\mathbf{L}'}^{\infty}\mathbf{H}'_{Y}(k)\backslash\mathbf{H}'_{Y}(\mathbb{A})) \int_{\mathbf{H}'_{Y}(\mathbb{A})\backslash\mathbf{H}'(\mathbb{A})} \phi'(\operatorname{Ad}(x^{-1})(Y))v_{\mathbf{L}'}^{\mathbf{H}'}(x)dx.$$

We denote

$$\tau(\mathbf{H}'_Y) := \operatorname{vol}(A^{\infty}_{\mathbf{L}'}\mathbf{H}'_Y(k) \backslash \mathbf{H}'_Y(\mathbb{A}))$$

for $Y \in (\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{\mathrm{rs}})(k)_{\mathrm{ell}}$ and define

$$J_{\mathbf{L}'}^{\mathbf{H}'}(Y,\phi') := \int_{\mathbf{H}'_{Y}(\mathbb{A}) \backslash \mathbf{H}'(\mathbb{A})} \phi'(\mathrm{Ad}(x^{-1})(Y)) v_{\mathbf{L}'}^{\mathbf{H}'}(x) dx$$

for all $Y \in (\tilde{\mathfrak{l}}' \cap \mathfrak{s}'_{rs})(k)$. From Lemma 3.16 in Chapter 4, we obtain

$$(10.0.4) J^{\mathbf{H}'}(\phi') = \sum_{\mathbf{L}' \in \mathscr{L}^{\mathbf{H}'}(\mathbf{M}'_0)} |W_0^{\mathbf{L}'}| |W_0^{\mathbf{H}'}|^{-1} \sum_{Y \in \Gamma_{\mathrm{ell}}((\tilde{\iota'} \cap \mathfrak{s}'_{\mathrm{rs}})(k))} \tau(\mathbf{H}'_Y) J_{\mathbf{L}'}^{\mathbf{H}'}(Y, \phi').$$

PROPOSITION 10.1 (see Theorem 7.1 in Chapter 2 and Theorem 7.1 in Chapter 3). 1) Let $\phi \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ be such that $\operatorname{Supp}(\phi) \subseteq \mathfrak{s}_{rs}(k_{v_1})$ and $\operatorname{Supp}(\hat{\phi}) \subseteq \mathfrak{s}_{rs}(k_{v_2})$ at some places v_1, v_2 of k. Then we have the equality

$$J^{\mathbf{G}}(\eta, \phi) = J^{\mathbf{G}}(\eta, \hat{\phi}).$$

2) Let $\phi' \in \mathcal{S}(\mathfrak{s}'(\mathbb{A}))$ be such that $\operatorname{Supp}(\phi') \subseteq \mathfrak{s}'_{\operatorname{rs}}(k_{v_1})$ and $\operatorname{Supp}(\hat{\phi'}) \subseteq \mathfrak{s}'_{\operatorname{rs}}(k_{v_2})$ at some places v_1, v_2 of k. Then we have the equality

$$J^{\mathbf{H}'}(\phi') = J^{\mathbf{H}'}(\hat{\phi}').$$

11. Proof of Proposition 5.5

Let F be a non-archimedean local field of characteristic zero and E be a quadratic extension of F. Fix $M' \in \mathscr{L}^{H'}(M'_0)$. Recall that its image in $\mathscr{L}^{G,\omega}(M_0)$ is denoted by M. Let $f \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}(F))$ and $f' \in \mathcal{C}^{\infty}_{c}(\mathfrak{s}'(F))$ be partially *M*-associated and satisfy the two conditions in Theorem 5.3. Fix $X_0 \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(F)$ and $Y_0 \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(F)$ with *M*-matching orbits. It is obvious from Definition 5.1 that a pair of partially M-associated functions are also partially L-associated for all $L \in \mathscr{L}^G(M)$. Then by induction, to prove Proposition 5.5, it suffices to show that for all $Q \in \mathscr{F}^G(M)$, we have the equality

(11.0.1)
$$\kappa(X_0)J_M^Q(\eta, X_0, \hat{f}) = J_{M'}^{Q'}(Y_0, \hat{f'}).$$

11.1. Global data. Fix a quadratic extension of number fields k'/k, a central division algebra $\mathbb D$ over k, and a finite set S of finite places of k satisfying the conditions in Proposition 9.1. Fix $w \in S$. Denote by V (resp. V_{∞} , V_f) the set of places (resp. archimedean places, finite places) of k.

Define the global symmetric pairs (\mathbf{G}, \mathbf{H}) and $(\mathbf{G}', \mathbf{H}')$ over k with respect to k' and \mathbb{D} as in the local case. There is a bijection $L \mapsto \mathbf{L}$ from $\mathscr{L}^G(M_0)$ to $\mathscr{L}^{\mathbf{G}}(\mathbf{M}_0)$ such that $\mathbf{L}_w \simeq L$ and we denote by \mathbf{M} the image of M under this bijection. Similarly, there is a bijection $L' \mapsto \mathbf{L}'$ from $\mathscr{L}^{H'}(M'_0)$ to $\mathscr{L}^{\mathbf{H}'}(\mathbf{M}'_0)$ such that $\mathbf{L}'_w\simeq L'$ and we denote by \mathbf{M}' the image of M' under this bijection.

11.2. Places. Fix a continuous and nontrivial unitary character $\Psi : \mathbb{A}/k \to \mathbb{C}^{\times}$ whose local component at w is ψ . Let $\langle \cdot, \cdot \rangle_{\mathfrak{s}}$ (resp. $\langle \cdot, \cdot \rangle_{\mathfrak{s}'}$) be the **H**-invariant (resp. **H**'-invariant) non-degenerate symmetric bilinear form on \mathfrak{s} (resp. on \mathfrak{s}') defined by (2.3.1) (resp. by (2.4.1)). Then we deduce local data Ψ_v , $\langle \cdot, \cdot \rangle_{\mathfrak{s}_v}$ and $\langle \cdot, \cdot \rangle_{\mathfrak{s}'_v}$ for all $v \in V$. Fix a finite set $S_1 \subseteq V$ and for $v \in V - S_1$ lattices $\mathfrak{k}_v \subseteq \mathfrak{s}(k_v)$ and $\mathfrak{k}'_v \subseteq \mathfrak{s}'(k_v)$ such that

(i) $V_{\infty} \sqcup S \subseteq S_1;$

- (ii) if $v \in V S_1$, then
 - k_v has odd residue characteristic and v is unramified in k';

•
$$\mathfrak{k}_v = \left\{ \begin{pmatrix} 0 & A \\ B & 0 \end{pmatrix} : A, B \in \mathfrak{gl}_n(\mathcal{O}_{k_v}) \right\};$$

- $\mathfrak{t}'_v = \mathfrak{t}_v$ if v splits in k', while $\mathfrak{t}'_v = \mathfrak{gl}_n(\mathcal{O}_{k'_v})$ if v is inert in k'; \mathfrak{t}_v (resp. \mathfrak{t}'_v) is self-dual respect to $\Psi_v(\langle\cdot,\cdot\rangle_{\mathfrak{s}_v})$ (resp. $\Psi_v(\langle\cdot,\cdot\rangle_{\mathfrak{s}'_v})$).

For $v \in V - S_1$, notice that $\widehat{1_{\mathfrak{k}_v}} = 1_{\mathfrak{k}_v}$ (resp. $\widehat{1_{\mathfrak{k}'_v}} = 1_{\mathfrak{k}'_v}$), where $1_{\mathfrak{k}_v}$ (resp. $1_{\mathfrak{k}'_v}$) denotes the characteristic function of \mathfrak{k}_v (resp. \mathfrak{k}'_v).

11.3. Orbits. For each $v \in S_1 - V_{\infty}$, we fix an open compact non-empty subset $\Omega_v \subseteq (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(k_v)$ such that

(i) if v = w, then $Y_0 \in \Omega_w \subseteq \operatorname{Ad}(\mathbf{M}'(k_w))((\mathfrak{s}'_{Y_0} \cap \mathfrak{s}'_{\mathrm{rs}})(k_w)), J^{Q'}_{M'}(\cdot, \hat{f'})$ is constant on Ω_w and $\kappa(\cdot)J^Q_M(\eta, \cdot, \hat{f})$ is constant on $\{X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(k_w) : \exists Y \in \Omega_w, X \text{ and } Y \text{ have } M\text{-matching orbits}\};$ (ii) if $v \in S - \{w\}$, then $\Omega_v \subseteq (\mathfrak{m}' \cap \mathfrak{s}'_{rs})(k_v)_{ell}$.

Notice that the condition (i) is achievable because of Lemma 4.7, Propositions 4.4.2) and 4.1.2) in Chapter 4 and the constancy of $J_{M'}^{Q'}(\cdot, \hat{f}')$ (resp. $\kappa(\cdot)J_{M}^{Q}(\eta, \cdot, \hat{f})$) on $\mathbf{M}'(k_w)$ (resp. $\mathbf{M}_{\mathbf{H}}(k_w)$)-orbits. Besides, the set $(\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(k_v)_{ell}$ in the condition (ii) is not empty (see [55, Lemma 2.7 and p. 14]). Since $X_0 \leftrightarrow Y_0$, the condition (i) implies that the restriction of $\kappa(\cdot)J_M^Q(\eta,\cdot,\hat{f})$ to $\{X \in (\mathfrak{m} \cap \mathfrak{s}_{\mathrm{rs}})(k_w) : \exists Y \in \Omega_w, X \leftrightarrow Y\}$ equals $\kappa(X_0) J_M^Q(\eta, X_0, \hat{f}).$

By the strong approximation theorem, there exists $Y^0 \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}')(k)$ such that

- (i) for $v \in S_1 V_{\infty}$, $Y^0 \in \Omega_v$; (ii) for $v \in V S_1$, $Y^0 \in \widetilde{\mathfrak{m}'}(k_v) \cap \mathfrak{k}'_v$.

Combined with the condition (ii) on Ω_v , the condition (i) implies that $Y^0 \in (\widetilde{\mathfrak{m}'} \cap \mathfrak{s}'_{rs})(k)_{ell}$. Choose an element $X^0 \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(k)_{ell}$ such that X^0 and Y^0 have **M**-matching orbits.

11.4. Choice of functions. For each $v \in V$, we fix functions $\phi_v \in \mathcal{S}(\mathfrak{s}(k_v))$ and $\phi'_v \in \mathcal{S}(\mathfrak{s}'(k_v))$ as follows.

- (i) If v = w, let $\phi_w = f$ and $\phi'_w = f'$.
- (ii) If $v \in S_1 V_\infty \{w\}$, then ϕ_v and ϕ'_v verify the conditions in Proposition 7.2 with respect to \mathbf{M}'_v , X^0 and Y^0 .
- (iii) If $v \in V S_1$, let $\phi_v = 1_{\mathfrak{k}_v}$ and $\phi'_v = 1_{\mathfrak{k}'_v}$.
- (iv) If $v \in V_{\infty}$, we identify $(\mathbf{H}_{v}, \mathfrak{s}_{v})$ and $(\mathbf{H}'_{v}, \mathfrak{s}'_{v})$ by the condition (a) in Proposition 9.1, and choose $\phi_{v} = \phi'_{v}$ such that
 - $J_{\mathbf{G}_{v}}^{\mathbf{G}_{v}}(\eta, X^{0}, \hat{\phi_{v}}) = J_{\mathbf{H}_{v}'}^{\mathbf{H}_{v}'}(Y^{0}, \hat{\phi_{v}'}) \neq 0;$
 - if $X \in \mathfrak{s}(k)$ is $\mathbf{H}(k_v)$ -conjugate to an element in $\operatorname{Supp}(\hat{\phi_v})$ for all $v \in V$, then X is $\mathbf{H}(k)$ conjugate to X^0 ;
 - if $Y \in \mathfrak{s}'(k)$ is $\mathbf{H}'(k_v)$ -conjugate to an element in $\operatorname{Supp}(\hat{\phi}'_v)$ for all $v \in V$, then Y is $\mathbf{H}'(k)$ conjugate to Y^0 .

Notice that the condition (ii) is easier to be satisfied if v splits in k'. In fact, we can identify $(\mathbf{H}_v, \mathfrak{s}_v)$ and $(\mathbf{H}'_v, \mathfrak{s}'_v)$ by the condition (d) in Proposition 9.1, and choose $\phi_v = \phi'_v$. Additionally, the condition (iv) is possible (see [58, p. 1874]).

In the rest of the proof of Proposition 5.5, we consider the global functions $\phi := \prod_{v \in V} \phi_v \in \mathcal{S}(\mathfrak{s}(\mathbb{A}))$ and $\phi' := \prod \phi'_v \in \mathcal{S}(\mathfrak{s}'(\mathbb{A})).$

and $\phi' := \prod_{v \in V} \phi'_v \in \mathcal{S}(\mathfrak{s}'(\mathbb{A})).$

11.5. Comparison of $J^{\mathbf{G}}(\eta, \phi)$ and $J^{\mathbf{H}'}(\phi')$.

LEMMA 11.1 (cf. [14, Lemme 10.1]). For our choice of ϕ and ϕ' , we have

$$J^{\mathbf{G}}(\eta,\phi) = \sum_{\mathbf{L}\in\mathscr{L}^{\mathbf{G}}(\mathbf{M})} |\mathrm{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n,\mathbf{L}_n)|^{-1} \sum_{X\in\Gamma_{\mathrm{ell}}((\mathfrak{l}\cap\mathfrak{s}_{\mathrm{rs}})(k))} \tau(\mathbf{H}_X) J^{\mathbf{G}}_{\mathbf{L}}(\eta,X,\phi)$$

and

$$J^{\mathbf{H}'}(\phi') = \sum_{\mathbf{L}' \in \mathscr{L}^{\mathbf{H}'}(\mathbf{M}')} |\operatorname{Tran}_{\mathbf{H}'}(\mathbf{M}', \mathbf{L}')|^{-1} \sum_{Y \in \Gamma_{\mathrm{ell}}((\widetilde{t'} \cap \mathfrak{s}'_{\mathrm{rs}})(k))} \tau(\mathbf{H}'_Y) J^{\mathbf{H}'}_{\mathbf{L}'}(Y, \phi').$$

PROOF. For the first formula, we start with (10.0.2). Fix $\mathbf{L} \in \mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)$ and $X \in \Gamma_{\text{ell}}(((\Gamma \mathfrak{s}_{rs})(k))$ such that $J^{\mathbf{G}}_{\mathbf{L}}(\eta, X, \phi) \neq 0$. By the condition (b) in Proposition 9.1, there exists $u \in S - \{w\}$. From the condition imposed on ϕ_u (see (a) in Proposition 7.2) and Lemma 4.7, we see that X is $\mathbf{H}(k_u)$ -conjugate to an element in $(\mathfrak{s}_{X^0} \cap \mathfrak{s}_{rs})(k_u)$. Choose $\mathbf{R} \in \mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)$ such that $\mathbf{R} \subseteq \mathbf{L}$ and that X is $\mathbf{L}_{\mathbf{H}}(k_u)$ -conjugate to an element in $(\mathfrak{r} \cap \mathfrak{s}_{rs})(k_u)_{\text{ell}}$. Then there exists $x \in \mathbf{H}(k_u)$ and $Z \in (\mathfrak{r} \cap \mathfrak{s}_{rs})(k_u)_{\text{ell}}$ such that $Z = \operatorname{Ad}(x)(X^0)$. But by the condition (ii) on Ω_u , we know that $X^0 \in (\mathfrak{m} \cap \mathfrak{s}_{rs})(k_u)_{\text{ell}}$. Thus by Lemma 3.10.1) in Chapter 4, there exists $w \in \left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{\mathbf{GL}_{n,u}} \right\}$ such that $\operatorname{Ad}(w)(\mathbf{M}_u) = \mathbf{R}_u$. Since we may identify $W_0^{\mathbf{GL}_{n,u}}$ with $W_0^{\mathbf{GL}_n}$, we deduce that $\operatorname{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n, \mathbf{L}_n) \neq \emptyset$. That is to say, we may restrict the sum on $\mathbf{L} \in \mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)$ in (10.0.2) to those conjugate to an element in $\mathscr{L}^{\mathbf{G}}(\mathbf{M})$ under $\left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{\mathbf{GL}_n} \right\}$.

We reindex the sum on $\mathbf{L} \in \mathscr{L}^{\mathbf{G}}(\mathbf{M})$. On the one hand, the number of elements in $\mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0)$ conjugate to \mathbf{L} under $\left\{ \begin{pmatrix} \omega_n \\ & \omega_n \end{pmatrix} : \omega_n \in W_0^{\mathbf{GL}_n} \right\}$ is

$$|\operatorname{Norm}_{W_0^{\operatorname{\mathbf{GL}}_n}}(\mathbf{L}_n) \setminus W_0^{\operatorname{\mathbf{GL}}_n}|$$

On the other hand, the number of elements in $\mathscr{L}^{\mathbf{G}}(\mathbf{M})$ conjugate to \mathbf{L} under $\left\{ \begin{pmatrix} \omega_n \\ \omega_n \end{pmatrix} : \omega_n \in W_0^{\mathbf{GL}_n} \right\}$ is

$$|W^{\mathbf{GL}_n}(\mathbf{L}_n)|^{-1}|\mathrm{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n,\mathbf{L}_n)|.$$

Since $W^{\mathbf{GL}_n}(\mathbf{L}_n) = \operatorname{Norm}_{W_0^{\mathbf{GL}_n}}(\mathbf{L}_n) / W_0^{\mathbf{L}_n}$ (see [35, (7.12.2)]), we obtain $|W_0^{\mathbf{L}_n}||W_0^{\mathbf{GL}_n}|^{-1} \cdot |\operatorname{Norm}_{W_0^{\mathbf{GL}_n}}(\mathbf{L}_n) \setminus W_0^{\mathbf{GL}_n}| \cdot |W^{\mathbf{GL}_n}(\mathbf{L}_n)||\operatorname{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n, \mathbf{L}_n)|^{-1} = |\operatorname{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n, \mathbf{L}_n)|^{-1}.$ Then the first formula of the lemma follows. The second formula can be proved in a similar way with the help of (10.0.4), the condition (b) in Proposition 7.2 of ϕ'_u and Lemma 3.16.1) in Chapter 4. The only additional ingredient is the condition (c) in Proposition 9.1, by which we may identify $W_0^{\mathbf{H}'_u}$ with $W_0^{\mathbf{H}'}$.

Recall that we have bijections $\mathscr{L}^{\mathbf{H}'}(\mathbf{M}') \to \mathscr{L}^{\mathbf{G}}(\mathbf{M})$ and $\mathscr{F}^{\mathbf{H}'}(\mathbf{M}') \to \mathscr{F}^{\mathbf{G}}(\mathbf{M})$. For all $\mathbf{L} \in \mathscr{L}^{\mathbf{G}}(\mathbf{M})$, we also have \mathbb{R} -linear isomorphisms between $\mathfrak{a}_{\mathbf{L}'}$ and $\mathfrak{a}_{\mathbf{L}}$ and, for all $v \in V$, between $\mathfrak{a}_{\mathbf{L}'_v}$ and $\mathfrak{a}_{\mathbf{L}_v}$. By modifying the scalar products on these Euclidean spaces, we may and shall suppose that these bijections are isometries. Then for all $(\mathbf{R}_v)_{v \in V} \in \prod_{v \in V} \mathscr{L}^{\mathbf{G}_v}(\mathbf{L}_v)$, we have the equality

$$d_{\mathbf{L}'}^{\mathbf{H}'}((\mathbf{R}'_v)_{v\in V}) = d_{\mathbf{L}}^{\mathbf{G}}((\mathbf{R}_v)_{v\in V}),$$

where both sides are Arthur's coefficients defined in [6, p. 356]. We shall also choose compatible sections

$$(\mathbf{R}'_v)_{v \in V} \mapsto (\mathbf{Q}_{\mathbf{R}'_v})_{v \in V} \in \prod_{v \in V} \mathscr{P}^{\mathbf{H}'_v}(\mathbf{R}'_v)$$

and

$$(\mathbf{R}_v)_{v \in V} \mapsto (\mathbf{Q}_{\mathbf{R}_v})_{v \in V} \in \prod_{v \in V} \mathscr{P}^{\mathbf{G}_v}(\mathbf{R}_v)$$

in the sense that for all $v \in V$,

$$\mathbf{Q}_{\mathbf{R}'_v} = (\mathbf{Q}_{\mathbf{R}_v})'.$$

PROPOSITION 11.2. For our choice of ϕ and ϕ' , we have the equality

$$J^{\mathbf{G}}(\eta,\phi) = J^{\mathbf{H}'}(\phi')$$

The rest of this section is devoted to the proof of Proposition 11.2.

Let $\mathbf{L} \in \mathscr{L}^{\mathbf{G}}(\mathbf{M})$. There is a canonical bijection between $\operatorname{Tran}_{\mathbf{GL}_n}(\mathbf{M}_n, \mathbf{L}_n)$ and $\operatorname{Tran}_{\mathbf{H}'}(\mathbf{M}', \mathbf{L}')$ since both of them are understood as permutations. If $X \in \Gamma_{\mathrm{ell}}((\mathfrak{l} \cap \mathfrak{s}_{\mathrm{rs}})(k))$ and $Y \in \Gamma_{\mathrm{ell}}((\widetilde{\mathfrak{l}'} \cap \mathfrak{s}'_{\mathrm{rs}})(k))$ have matching orbits, by Lemma 4.7, there is an isomorphism $\mathbf{H}_X \simeq \mathbf{H}'_Y$ over F. By choosing compatible Haar measures on them, we have $\tau(\mathbf{H}_X) = \tau(\mathbf{H}'_Y)$. Since X and Y have matching orbits at each $v \in V$, we see that $J_{\mathbf{L}}^{\mathbf{G}}(\eta, X, \phi) = J_{\mathbf{L}'}^{\mathbf{H}'}(Y, \phi')$ by our choice of ϕ and ϕ' .

Now assume that X does not come from $(\hat{\mathfrak{l}} \cap \mathfrak{s}'_{rs})(k)$ (or equivalently $\mathfrak{s}'_{rs}(k)$). By Lemma 11.1, to show Proposition 11.2, it suffices to show that $J^{\mathbf{G}}_{\mathbf{L}}(\eta, X, \phi) = 0$.

By the splitting formula of (\mathbf{G}, \mathbf{M}) -families applied to the weight function (see [6, Corollary 7.4]), we write

$$J_{\mathbf{L}}^{\mathbf{G}}(\eta, X, \phi) = \sum_{(\mathbf{R}_v)_{v \in V} \in \prod_{v \in V} \mathscr{L}^{\mathbf{G}_v}(\mathbf{L}_v)} d_{\mathbf{L}}^{\mathbf{G}}((\mathbf{R}_v)_{v \in V}) \prod_{v \in V} \kappa_v(X) J_{\mathbf{L}_v}^{\mathbf{Q}_{\mathbf{R}_v}}(\eta, X, \phi_v).$$

If $J_{\mathbf{L}}^{\mathbf{G}}(\eta, X, \phi) \neq 0$, we fix $(\mathbf{R}_{v})_{v \in V} \in \prod_{v \in V} \mathscr{L}^{\mathbf{G}_{v}}(\mathbf{L}_{v})$ such that $d_{\mathbf{L}}^{\mathbf{G}}((\mathbf{R}_{v})_{v \in V}) \prod_{v \in V} \kappa_{v}(X) J_{\mathbf{L}_{v}}^{\mathbf{Q}_{\mathbf{R}_{v}}}(\eta, X, \phi_{v}) \neq 0.$

We choose a representative of X (still denoted by X) in the form of $\begin{pmatrix} 0 & 1_n \\ A & 0 \end{pmatrix} \in (\mathfrak{l} \cap \mathfrak{s}_{rs})_{ell}(k).$

LEMMA 11.3. For all $v \in V$, we have $\eta_{\mathbf{R}_v}(X) = e_{\mathbf{R}'_v}$.

PROOF. It is trivial if $v \in V$ splits in k' (in particular, if $v \in V_{\infty}$). For $v \in V - S_1$ which does not split in k', it results from Lemma 8.1.(b). For $v \in S_1 - V_{\infty} - \{w\}$ which does not split in k', it is deduced from Proposition 7.2.(a)(c) and Corollary 4.6.(1). For v = w, it is because f satisfies the additional condition in Definition 5.2.

LEMMA 11.4. For all $v \in V$, we have $\eta_{\mathbf{L}_v}(X) = e_{\mathbf{L}'_v}$.

PROOF. Let S_2 be the subset of V_f consisting of v which does not split in k'. The assertion is trivial for $v \in V - S_2$. Consider $v \in S_2$. By Propositions 4.20 and 4.21, $\eta_{\mathbf{L}_v}(X)$ and $e_{\mathbf{L}'_v}$ are understood as $\varepsilon_X^{\mathbf{L}_v}$ and $\varepsilon'^{\mathbf{L}_v}$ respectively. By Proposition 4.19, we view $\varepsilon_X^{\mathbf{L}_v}$ and $\varepsilon'^{\mathbf{L}_v}$ as complex characters of $Z_{\widehat{\mathbf{L}}}[2]$. Denote by ζ_v the character obtained by their quotient.

For all $v \in S_2$, by Lemma 11.3, we know that the character ζ_v is trivial on $Z_{\widehat{\mathbf{R}}_v}[2]$. By the product formula of Kottwitz signs [33, the last proposition] and the fundamental exact sequence of global class

field theory for k'/k (see [42, Example 4.4.(a) in Chapter VIII] for example), we also have a global condition: the product $\prod_{v \in S_2} \zeta_v$ is trivial on $Z_{\widehat{\mathbf{L}}}[2]$.

Let $v \in S_2$. We need to show that ζ_v is trivial on $Z_{\widehat{\mathbf{L}}}[2]$. We first deduce from $d_{\mathbf{L}}^{\mathbf{G}}((\mathbf{R}_v)_{v \in V}) \neq 0$ that $Z_{\widehat{\mathbf{L}}} = A_v A^v$, where $A_v := Z_{\widehat{\mathbf{R}}_v}$ and $A^v := \bigcap_{v' \in S_2 - \{v\}} Z_{\widehat{\mathbf{R}}_{v'}}$. By [15, Lemme 4.5], we also have

 $\begin{aligned} A_v \cap A^v &= Z_{\widehat{\mathbf{G}}} \text{ since } d_{\mathbf{L}}^{\mathbf{G}}((\mathbf{R}_v)_{v \in V}) \neq 0. \\ \text{We claim that } Z_{\widehat{\mathbf{L}}}[2] &= A_v[2]A^v[2]. \text{ To see this, let } s = a_v a^v \in Z_{\widehat{\mathbf{L}}}[2], \text{ where } a_v \in A_v \text{ and } a^v \in A^v. \\ \text{Then } a_v^2 &= (a^v)^{-2} \in Z_{\widehat{\mathbf{G}}}. \text{ Since } z \mapsto z^2 \text{ is a surjective endomorphism of } Z_{\widetilde{\mathbf{G}}}, \text{ there exists } y \in Z_{\widehat{\mathbf{G}}} \text{ such that } a_v^2 &= (a^v)^{-2} = y^2. \text{ Thus } s = y^{-1}a_v \cdot (ya^v) \text{ with } y^{-1}a_v \in A_v[2] \text{ and } ya^v \in A^v[2]. \end{aligned}$ claim.

Now, let $s = a_v a^v \in Z_{\widehat{\mathbf{L}}}[2]$, where $a_v \in A_v[2]$ and $a^v \in A^v[2]$. Since the character ζ_v is trivial on $Z_{\widehat{\mathbf{R}}_{v}}[2]$, we have $\zeta_{v}(s) = \zeta_{v}(a_{v})\zeta_{v}(a^{v}) = \zeta_{v}(a^{v})$. By the global condition above, we have $\zeta_{v}(a_{v}) = \zeta_{v}(a^{v})$. $\prod_{v' \in S_2 - \{v\}} \zeta_{v'}^{-1}(a^v).$ But for $v' \in S_2 - \{v\}$, the character $\zeta_{v'}$ is trivial on $A^v[2]$. We have proved $\zeta_v(s) = 1$ for all $s \in Z_{\widehat{\mathbf{L}}}[2]$ and thus the lemma. \square

LEMMA 11.5. For all $v \in V$, X comes from $\mathfrak{s}'_{rs}(k_v)$.

PROOF. It suffices to consider $v \in V - S_1$ which does not split in k'. Notice that $e_{\mathbf{L}'_v} = (1, \dots, 1)$. We are in a similar situation as the base change for GL_n . There exists $\mathbf{L}_0 \in \mathscr{L}^{\mathbf{G},\omega}(\mathbf{M}_0), \mathbf{L}_0 \subseteq \mathbf{L}$ such that X is $(\mathbf{L}_v \cap \mathbf{H}_v)(k_v)$ conjugate to an element $Z \in (\mathfrak{l}_0 \cap \mathfrak{s}_{rs})(k_v)_{ell}$. If X does not come from $\mathfrak{s}'_{rs}(k_v)$, by [10, Lemma 1.4 in Chapter 1], there exists $\eta_{\mathbf{L}_{0,v}}(Z) \neq e_{\mathbf{L}'_{0,v}} = (1, \dots, 1)$. By Lemma 8.1.(b) and the argument of [37, Lemma III.3.4], we deduce that $J_{\mathbf{L}_{v}}^{\mathbf{Q}_{\mathbf{R}_{v}}}(\eta, X, \phi_{v}) = 0$, which contradicts our assumption.

PROPOSITION 11.6. The element X comes from $\mathfrak{s}'_{rs}(k)$.

PROOF. We start with two lemmas.

LEMMA 11.7. Let F_0 be a field. Let $Z_C \subseteq A \subseteq B \subseteq C$ be reductive groups defined over F_0 , where Z_C denotes the centre of C. Suppose that $H^1(F_0, A_\beta)$ is a singleton for all inner form A_β of A. Then the natural map

$$H^1(F_0, A/Z_C) \to H^1(F_0, B/Z_C)$$

is injective.

PROOF OF LEMMA 11.7. We begin with the following commutative diagram with exact rows.

Then we obtain the following commutative diagram of pointed sets.

$$\begin{array}{c} H^1(F_0,A/Z_C) \longrightarrow H^2(F_0,Z_C) \\ & \swarrow \\ g \\ H^1(F_0,B/Z_C) \longrightarrow H^2(F_0,Z_C) \end{array}$$

Since $H^1(F_0, A_\beta)$ is a singleton for all inner form A_β of A, the map

$$H^1(F_0, A/Z_C) \to H^2(F_0, Z_C)$$

is injective by [32, Corollary (28.13)]. As the above diagram is commutative, we deduce that g is injective.

LEMMA 11.8. Let F_0 be a global field. Let $Z_B \subseteq A \subseteq B$ be reductive groups defined over F_0 , where Z_B denotes the centre of B. Suppose that $H^1(F_0, A_\beta)$ is a singleton for all inner form A_β of A. Then the map

$$H^1(F_0, A/Z_B) \to H^1(\mathbb{A}_{F_0}, A/Z_B)$$

is injective.

PROOF OF LEMMA 11.8. We use the commutative diagram of pointed sets:

$$\begin{array}{c} H^{1}(F_{0}, A/Z_{B}) \xrightarrow{i} H^{1}(\mathbb{A}_{F_{0}}, A/Z_{B}) \\ \downarrow^{g} \qquad \qquad \downarrow^{g} \qquad \qquad \downarrow^{g} \\ H^{1}(F_{0}, B_{\mathrm{ad}}) \xrightarrow{h} H^{1}(\mathbb{A}_{F_{0}}, B_{\mathrm{ad}}) \end{array}$$

By Lemma 11.7, the map g is injective. By the Hasse principle for $B_{\rm ad}$, the map h is injective. Since the diagram is commutative, we deduce that i is injective. \square

Return to the proof of Proposition 11.6. We shall use the notation in Section 4.3 but bold letters for reductive groups defined over k. Let $t \in H^1(k, \mathbf{T}/\mathbf{R})$ be the class associated to X in Lemma 4.8. Let $u \in H^1(k, \mathbf{H}_0/Z_{\mathbf{G}})$ be the class associated to \mathfrak{g}' . Let $t' \in H^1(\mathbb{A}, \mathbf{T}/\mathbf{R})$ (resp. $u' \in H^1(\mathbb{A}, \mathbf{H}_0/Z_{\mathbf{G}})$) be the image of t (resp. u). By Lemmas 11.5 and 4.9, there exists $v' \in H^1(k, \mathbf{T}/Z_{\mathbf{G}})$ with images t' and u'.

By [38, Proposition 1.6.12], we have the following commutative diagram of pointed sets with exact rows.

Let $v'' \in H^1(\mathbb{A}, \mathbf{T}/\mathbf{R})$ (resp. $u'' \in H^1(\mathbb{A}, \mathbf{H}_0/Z_{\mathbf{G}})$) be the image of v' (resp. u'). Then u'' = 0. By a variant of Lemma 11.7 for $H^1_{ab}(\mathbb{A}/k,\cdot)$, the map g is injective. Thus v'' = 0 and there exists $v \in H^1(k, \mathbf{T}/Z_{\mathbf{G}})$ with image v'. By Lemma 11.8, the map i is injective. Since the square on the left above is commutative, the class v maps to u.

We also have the following commutative diagram.

$$\begin{array}{c} H^{1}(k,\mathbf{T}/Z_{\mathbf{G}}) \longrightarrow H^{1}(\mathbb{A},\mathbf{T}/Z_{\mathbf{G}}) \\ & \downarrow \\ & \downarrow \\ H^{1}(k,\mathbf{T}/\mathbf{R}) \xrightarrow{j} H^{1}(\mathbb{A},\mathbf{T}/\mathbf{R}) \end{array}$$

By our discussion in Section 4.3, the map j is given by

$$\prod_{i \in I_0} k_i^{\times} / N_{k_i'/k_i}({k_i'}^{\times}) \to \prod_{i \in I_0} \mathbb{A}_{k_i}^{\times} / N_{k_i'/k_i}(\mathbb{A}_{k_i'}^{\times}),$$

where $k_i = k[\lambda]/(\chi_i(\lambda))$ with χ_i being an irreducible polynomial over k, and $k'_i = k_i \otimes_k k'$. It is known to be injective. Since the diagram is commutative, the class v maps to t. \Box

By Lemma 4.9, we draw our conclusion.

PROOF OF PROPOSITION 11.2. The statement of Proposition 11.6 contradicts our assumption, so we have finished the proof.

11.6. End of the proof.

LEMMA 11.9. For our choice of ϕ and ϕ' , we have

$$J^{\mathbf{G}}(\eta, \hat{\phi}) = \tau(\mathbf{H}_{X^0}) J^{\mathbf{G}}_{\mathbf{M}}(\eta, X^0, \hat{\phi})$$

and

$$J^{\mathbf{H}'}(\hat{\phi}') = \tau(\mathbf{H}'_{Y^0}) J^{\mathbf{H}'}_{\mathbf{M}'}(Y^0, \hat{\phi}').$$

PROOF. The first formula results from (10.0.1) for $\hat{\phi}$ and the condition (iv) in Section 11.4, while the second formula results from (10.0.3) for $\hat{\phi}'$ and the condition (iv) in Section 11.4. \square

PROOF OF PROPOSITION 5.5. Combining Propositions 10.1 and 11.2 and Lemma 11.9, for our choice of ϕ and ϕ' , we have the equality

$$J_{\mathbf{M}}^{\mathbf{G}}(\eta, X^0, \hat{\phi}) = J_{\mathbf{M}'}^{\mathbf{H}'}(Y^0, \hat{\phi}').$$

But by the splitting formula and the conditions (iii) and (iv) on ϕ_v and ϕ'_v , the difference of two sides can be written as

(11.6.1)
$$\sum_{\substack{(\mathbf{L}_{v})_{v\in V}\in\prod\limits_{v\in V}\mathscr{L}^{\mathbf{G}_{v}}(\mathbf{M}_{v})}} d_{\mathbf{M}}^{\mathbf{G}}((\mathbf{L}_{v})_{v\in V}) \prod\limits_{v\in(V-S_{1})\cup V_{\infty}} \kappa_{v}(X^{0}) J_{\mathbf{M}_{v}}^{\mathbf{Q}_{\mathbf{L}_{v}}}(\eta, X^{0}, \hat{\phi_{v}}) \\ \times \left[\prod\limits_{v\in S_{1}-V_{\infty}} \kappa_{v}(X^{0}) J_{\mathbf{M}_{v}}^{\mathbf{Q}_{\mathbf{L}_{v}}}(\eta, X^{0}, \hat{\phi_{v}}) - \prod\limits_{v\in S_{1}-V_{\infty}} J_{\mathbf{M}_{v}'}^{\mathbf{Q}_{\mathbf{L}_{v}}}(Y^{0}, \hat{\phi_{v}'})\right].$$

By the condition (e) in Proposition 7.2, we may suppose that $\mathbf{L}_v = \mathbf{M}_v$ for all $v \in S_1 - V_\infty - \{w\}$. Moreover, by the condition (f) in Proposition 7.2, we have

$$\kappa_{v}(X^{0})J_{\mathbf{M}_{v}}^{\mathbf{Q}_{\mathbf{M}_{v}}}(\eta, X^{0}, \hat{\phi_{v}}) = \gamma_{\Psi_{v}}(\mathfrak{h}(k_{v}))\gamma_{\Psi_{v}}(\mathfrak{h}'(k_{v}))^{-1}J_{\mathbf{M}'_{v}}^{\mathbf{Q}'_{\mathbf{M}_{v}}}(Y^{0}, \hat{\phi_{v}'}) \neq 0$$

for all $v \in S_1 - V_\infty - \{w\}$.

Recall the product formula

$$\prod_{v \in V} \gamma_{\Psi_v}(\mathfrak{h}(k_v)) = \prod_{v \in V} \gamma_{\Psi_v}(\mathfrak{h}'(k_v)) = 1.$$

For $v \in (V - S_1) \cup V_{\infty}$, we have

$$\gamma_{\Psi_v}(\mathfrak{h}(k_v)) = \gamma_{\Psi_v}(\mathfrak{h}'(k_v)) = 1.$$

Hence,

$$\prod_{-V_{\infty}-\{w\}}\gamma_{\Psi_{v}}(\mathfrak{h}(k_{v}))\gamma_{\Psi_{v}}(\mathfrak{h}'(k_{v}))^{-1}=\gamma_{\Psi_{w}}(\mathfrak{h}(k_{w}))^{-1}\gamma_{\Psi_{w}}(\mathfrak{h}'(k_{w})).$$

Then the expression (11.6.1) equals

K

 $v \in S_1$

(11.6.2)
$$\sum_{\substack{(\mathbf{L}_{v})_{v\in V}\in\prod\limits_{v\in V}\mathscr{L}^{\mathbf{G}_{v}}(\mathbf{M}_{v})}} d_{\mathbf{M}}^{\mathbf{G}}((\mathbf{L}_{v})_{v\in V}) \prod\limits_{v\in V-\{w\}} \kappa_{v}(X^{0}) J_{\mathbf{M}_{v}}^{\mathbf{Q}_{\mathbf{L}_{v}}}(\eta, X^{0}, \hat{\phi_{v}}) \\ \times \left[\gamma_{\Psi_{w}}(\mathfrak{h}(k_{w}))^{-1} \gamma_{\Psi_{w}}(\mathfrak{h}'(k_{w})) \kappa_{w}(X^{0}) J_{\mathbf{M}_{w}}^{\mathbf{Q}_{\mathbf{L}_{w}}}(\eta, X^{0}, \hat{\phi_{w}}) - J_{\mathbf{M}_{w}'}^{\mathbf{Q}'_{\mathbf{L}_{w}}}(Y^{0}, \hat{\phi_{w}'})\right]$$

Since ϕ_w and ϕ'_w are partially \mathbf{M}_w -associated, by parabolic descent (see Propositions 4.1.4) and 4.4.4) in Chapter 4), we see that $\phi^{\eta}_{w,\mathbf{Q}_{\mathbf{L}_w}} \in \mathcal{C}^{\infty}_c((\mathfrak{l}_w \cap \mathfrak{s}_w)(k_w))$ and $\phi'_{w,\mathbf{Q}'_{\mathbf{L}_w}} \in \mathcal{C}^{\infty}_c(((\mathfrak{l}'_w \cap \mathfrak{s}'_w)(k_w)))$ are partially \mathbf{M}_w -associated. We shall prove (11.0.1) by induction on the dimension of G. Then we may suppose that for $\mathbf{L}_w \neq \mathbf{G}_w$, we have the equality

$$\kappa_w(X^0)J^{\mathbf{L}_w}_{\mathbf{M}_w}(\eta, X^0, \hat{\phi}^{\eta}_{w, \mathbf{Q}_{\mathbf{L}_w}}) = \gamma_{\Psi_w}((\mathfrak{l}_w \cap \mathfrak{h}_w)(k_w))\gamma_{\Psi_w}(\mathfrak{l}'_w(k_w))^{-1}J^{\mathbf{L}'_w}_{\mathbf{M}'_w}(Y^0, \hat{\phi}'_{w, \mathbf{Q}'_{\mathbf{L}_w}}).$$

Since the difference between the quadratic form on $\mathfrak{h}(k_w)$ (resp. $\mathfrak{h}'(k_w)$) and its restriction on $(\mathfrak{l}_w \cap \mathfrak{h}_w)(k_w)$ (resp. $\mathfrak{l}'_w(k_w)$) is a split quadratic form, we have $\gamma_{\Psi_w}((\mathfrak{l}_w \cap \mathfrak{h}_w)(k_w)) = \gamma_{\Psi_w}(\mathfrak{h}(k_w))$ (resp. $\gamma_{\Psi_w}(\mathfrak{l}'_w(k_w)) = \gamma_{\Psi_w}(\mathfrak{h}'(k_w))$). By parabolic descent again, we see that

$$\kappa_w(X^0) J^{\mathbf{Q}_{\mathbf{L}_w}}_{\mathbf{M}_w}(\eta, X^0, \hat{\phi_w}) = \gamma_{\Psi_w}(\mathfrak{h}(k_w)) \gamma_{\Psi_w}(\mathfrak{h}'(k_w))^{-1} J^{\mathbf{Q}'_{\mathbf{L}_w}}_{\mathbf{M}'_w}(Y^0, \hat{\phi_w'})$$

By the condition (i) on Ω_w , it implies (11.0.1) for all $\mathbf{Q}_{\mathbf{L}_w} \neq \mathbf{G}_w$ (actually for all $\mathbf{Q}_w \in \mathscr{L}^{\mathbf{G}_w}(\mathbf{M}_w), \mathbf{Q}_w \neq \mathbf{G}_w$). Then we may suppose that $\mathbf{L}_w = \mathbf{G}_w$ in (11.6.2). But for all $\mathbf{L} \in \mathscr{L}^{\mathbf{G}}(\mathbf{M})$, we have $\mathfrak{a}_{\mathbf{L}}^{\mathbf{G}} \simeq \mathfrak{a}_{\mathbf{L}_w}^{\mathbf{G}_w}$. Thus $d_{\mathbf{M}}^{\mathbf{G}}((\mathbf{L}_v)_{v \in V}) = 0$ unless $\mathbf{L}_v = \mathbf{M}_v$ for all $v \in V - \{w\}$, in which case $d_{\mathbf{M}}^{\mathbf{G}}((\mathbf{L}_v)_{v \in V}) = 1$. That is to say, the sum in (11.6.2) is reduced to only one term. We obtain

$$\prod_{v \in V - \{w\}} \kappa_v(X^0) J_{\mathbf{M}_v}^{\mathbf{Q}_{\mathbf{M}_v}}(\eta, X^0, \hat{\phi_v}) \\ \times \left[\gamma_{\Psi_w}(\mathfrak{h}(k_w))^{-1} \gamma_{\Psi_w}(\mathfrak{h}'(k_w)) \kappa_w(X^0) J_{\mathbf{M}_w}^{\mathbf{G}_w}(\eta, X^0, \hat{\phi_w}) - J_{\mathbf{M}'_w}^{\mathbf{G}'_w}(Y^0, \hat{\phi_w'}) \right] = 0.$$

By the condition (iv) on ϕ_v for $v \in V_\infty$, the condition (f) in Proposition 7.2, the condition (ii) on Y^0 and Lemma 8.1.(a), we know that the product on $v \in V - \{w\}$ does not vanish. Then we conclude by the condition (i) on Ω_w .

Bibliography

- Avraham Aizenbud and Dmitry Gourevitch. Generalized Harish-Chandra descent, Gelfand pairs, and an Archimedean analog of Jacquet-Rallis's theorem. *Duke Math. J.*, 149(3):509–567, 2009. With an appendix by the authors and Eitan Sayag.
- [2] James Arthur. The characters of discrete series as orbital integrals. Invent. Math., 32(3):205–261, 1976.
- [3] James Arthur. A trace formula for reductive groups. I. Terms associated to classes in $G(\mathbf{Q})$. Duke Math. J., 45(4):911–952, 1978.
- [4] James Arthur. A trace formula for reductive groups. II. Applications of a truncation operator. Compositio Math., 40(1):87–121, 1980.
- [5] James Arthur. The trace formula in invariant form. Ann. of Math. (2), 114(1):1-74, 1981.
- [6] James Arthur. The invariant trace formula. I. Local theory. J. Amer. Math. Soc., 1(2):323-383, 1988.
- [7] James Arthur. The invariant trace formula. II. Global theory. J. Amer. Math. Soc., 1(3):501-554, 1988.
- [8] James Arthur. A local trace formula. Inst. Hautes Études Sci. Publ. Math., (73):5-96, 1991.
- [9] James Arthur. An introduction to the trace formula. In Harmonic analysis, the trace formula, and Shimura varieties, volume 4 of Clay Math. Proc., pages 1–263. Amer. Math. Soc., Providence, RI, 2005.
- [10] James Arthur and Laurent Clozel. Simple algebras, base change, and the advanced theory of the trace formula, volume 120 of Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, 1989.
- [11] Roger W. Carter. Finite groups of Lie type. Pure and Applied Mathematics (New York). John Wiley & Sons, Inc., New York, 1985. Conjugacy classes and complex characters, A Wiley-Interscience Publication.
- [12] P. Cartier. Representations of p-adic groups: a survey. In Automorphic forms, representations and L-functions (Proc. Sympos. Pure Math., Oregon State Univ., Corvallis, Ore., 1977), Part 1, Proc. Sympos. Pure Math., XXXIII, pages 111–155. Amer. Math. Soc., Providence, R.I., 1979.
- [13] Pierre-Henri Chaudouard. La formule des traces pour les algèbres de Lie. Math. Ann., 322(2):347-382, 2002.
- [14] Pierre-Henri Chaudouard. Sur certaines identités endoscopiques entre transformées de Fourier. J. Reine Angew. Math., 585:1–59, 2005.
- [15] Pierre-Henri Chaudouard. Sur le changement de base stable des intégrals orbitales pondérées. Ann. Sci. École Norm. Sup. (4), 40(1):51–112, 2007.
- [16] Pierre-Henri Chaudouard. Le transfert lisse des intégrales orbitales d'après Waldspurger. In On the stabilization of the trace formula, volume 1 of Stab. Trace Formula Shimura Var. Arith. Appl., pages 145–180. Int. Press, Somerville, MA, 2011.
- [17] Pierre-Henri Chaudouard. Sur une variante des troncatures d'arthur. In Werner Müller, Sug Woo Shin, and Nicolas Templier, editors, *Geometric Aspects of the Trace Formula*, pages 85–120, Cham, 2018. Springer International Publishing.
- [18] Marion Chommaux. Distinction of the Steinberg representation and a conjecture of Prasad and Takloo-Bighash. J. Number Theory, 202:200–219, 2019.
- [19] Neil Chriss and Victor Ginzburg. Representation theory and complex geometry. Modern Birkhäuser Classics. Birkhäuser Boston, Ltd., Boston, MA, 2010. Reprint of the 1997 edition.
- [20] Brian Conrad. Global extensions approximating local extensions. Available at http://virtualmath1.stanford.edu/~conrad/248APage/handouts/localglobal.pdf.
- [21] Brooke Feigon, Kimball Martin, and David Whitehouse. Periods and nonvanishing of central L-values for GL(2n). Israel J. Math., 225(1):223–266, 2018.
- [22] Solomon Friedberg and Hervé Jacquet. Linear periods. J. Reine Angew. Math., 443:91–139, 1993.
- [23] Jiandong Guo. On a generalization of a result of Waldspurger. Canad. J. Math., 48(1):105-142, 1996.
- [24] Jiandong Guo. Uniqueness of generalized Waldspurger model for GL(2n). Pacific J. Math., 180(2):273–289, 1997.
- [25] Brian Hall. Lie groups, Lie algebras, and representations, volume 222 of Graduate Texts in Mathematics. Springer, Cham, second edition, 2015. An elementary introduction.
- [26] Harish-Chandra. Harmonic analysis on reductive p-adic groups. Lecture Notes in Mathematics, Vol. 162. Springer-Verlag, Berlin-New York, 1970. Notes by G. van Dijk.
- [27] Roger Howe. The Fourier transform and germs of characters (case of Gl_n over a *p*-adic field). Math. Ann., 208:305–322, 1974.
- [28] Atsushi Ichino and Shunsuke Yamana. Periods of automorphic forms: the case of $(GL_{n+1} \times GL_n, GL_n)$. Compos. Math., 151(4):665–712, 2015.
- [29] Hervé Jacquet. Sur un résultat de Waldspurger. Ann. Sci. École Norm. Sup. (4), 19(2):185–229, 1986.
- [30] Hervé Jacquet, Erez Lapid, and Jonathan Rogawski. Periods of automorphic forms. J. Amer. Math. Soc., 12(1):173– 240, 1999.
- [31] Hervé Jacquet and Stephen Rallis. Uniqueness of linear periods. Compositio Math., 102(1):65–123, 1996.

- [32] Max-Albert Knus, Alexander Merkurjev, Markus Rost, and Jean-Pierre Tignol. The book of involutions, volume 44 of American Mathematical Society Colloquium Publications. American Mathematical Society, Providence, RI, 1998. With a preface in French by J. Tits.
- [33] Robert E. Kottwitz. Sign changes in harmonic analysis on reductive groups. Trans. Amer. Math. Soc., 278(1):289–297, 1983.
- [34] Robert E. Kottwitz. Stable trace formula: cuspidal tempered terms. Duke Math. J., 51(3):611-650, 1984.
- [35] Robert E. Kottwitz. Harmonic analysis on reductive p-adic groups and Lie algebras. In Harmonic analysis, the trace formula, and Shimura varieties, volume 4 of Clay Math. Proc., pages 393–522. Amer. Math. Soc., Providence, RI, 2005.
- [36] Robert Edward Kottwitz. Orbital integrals on GL₃. Amer. J. Math., 102(2):327–384, 1980.
- [37] Jean-Pierre Labesse. Noninvariant base change identities. Mém. Soc. Math. France (N.S.), (61):iv+113, 1995.
- [38] Jean-Pierre Labesse. Cohomologie, stabilisation et changement de base. Astérisque, (257):vi+161, 1999. Appendix A by Laurent Clozel and Labesse, and Appendix B by Lawrence Breen.
- [39] Jean-Pierre Labesse and Jean-Loup Waldspurger. La formule des traces tordue d'après le Friday Morning Seminar, volume 31 of CRM Monograph Series. American Mathematical Society, Providence, RI, 2013. With a foreword by Robert Langlands [dual English/French text].
- [40] Jason Levy. A truncated Poisson formula for groups of rank at most two. Amer. J. Math., 117(6):1371-1408, 1995.
- [41] Qirui Li. An intersection number formula for CM cycles in Lubin-Tate towers. arXiv e-prints, page arXiv:1803.07553, Mar 2018.
- [42] J.S. Milne. Class field theory. Available at https://www.jmilne.org/math/CourseNotes/CFT.pdf, March 2013.
- [43] V. L. Popov and Eh. B. Vinberg. Invariant theory. Algebraic geometry. IV: Linear algebraic groups, invariant theory, Encycl. Math. Sci. 55, 123-278 (1994); translation from Itogi Nauki Tekh., Ser. Sovrem. Probl. Mat., Fundam. Napravleniya 55, 137-309 (1989)., 1989.
- [44] Dipendra Prasad and Ramin Takloo-Bighash. Bessel models for GSp(4). J. Reine Angew. Math., 655:189–243, 2011.
- [45] Cary Rader and Steve Rallis. Spherical characters on p-adic symmetric spaces. Amer. J. Math., 118(1):91–178, 1996.
- [46] Yiannis Sakellaridis and Akshay Venkatesh. Periods and harmonic analysis on spherical varieties. Astérisque, (396):viii+360, 2017.
- [47] Jean-Pierre Serre. Corps locaux. Hermann, Paris, 1968. Deuxième édition, Publications de l'Université de Nancago, No. VIII.
- [48] Sheng-Chi Shih, Tse-Chung Yang, and Chia-Fu Yu. Embeddings of fields into simple algebras over global fields. Asian J. Math., 18(2):365–386, 2014.
- [49] È. B. Vinberg. The Weyl group of a graded Lie algebra. Math. USSR, Izv., 10:463–495, 1977.
- [50] J.-L. Waldspurger. Sur les valeurs de certaines fonctions L automorphes en leur centre de symétrie. Compositio Math., 54(2):173–242, 1985.
- [51] J.-L. Waldspurger. Une formule des traces locale pour les algèbres de Lie p-adiques. J. Reine Angew. Math., 465:41–99, 1995.
- [52] J.-L. Waldspurger. Le lemme fondamental implique le transfert. Compositio Math., 105(2):153–236, 1997.
- [53] André Weil. Sur certains groupes d'opérateurs unitaires. Acta Math., 111:143–211, 1964.
- [54] André Weil. Basic number theory. Classics in Mathematics. Springer-Verlag, Berlin, 1995. Reprint of the second (1973) edition.
- [55] Hang Xue. Epsilon dichotomy for linear models. Available at https://www.math.arizona.edu/~xuehang/epsilon_I_v1.pdf, November 2019.
- [56] Chia-Fu Yu. Characteristic polynomials of central simple algebras. Taiwanese J. Math., 17(1):351–359, 01 2013.
- [57] Chong Zhang. On linear periods. Math. Z., 279(1-2):61-84, 2015.
- [58] Chong Zhang. On the smooth transfer for Guo-Jacquet relative trace formulae. Compos. Math., 151(10):1821–1877, 2015.
- [59] Wei Zhang. Fourier transform and the global Gan-Gross-Prasad conjecture for unitary groups. Ann. of Math. (2), 180(3):971–1049, 2014.
- [60] Michał Zydor. La variante infinitésimale de la formule des traces de Jacquet-Rallis pour les groupes unitaires. Canad. J. Math., 68(6):1382–1435, 2016.
- [61] Michał Zydor. La variante infinitésimale de la formule des traces de Jacquet-Rallis pour les groupes linéaires. J. Inst. Math. Jussieu, 17(4):735–783, 2018.